ON THE MEMBRANE MODEL AND THE DISCRETE BILAPLACIAN

DISSERTATION

zur Erlangung des Doktorgrades (Dr. rer. nat.) der Mathematisch-Naturwissenschaftlichen Fakultät der Rheinischen Friedrich-Wilhelms-Universität Bonn

vorgelegt von

FLORIAN MARTIN SCHWEIGER

aus Kempten (Allgäu)

Bonn 2020

Angefertigt mit Genehmigung der Mathematisch-Naturwissenschaftlichen Fakultät der Rheinischen Friedrich-Wilhelms-Universität Bonn

1. Gutachter: Prof. Dr. Stefan Müller 2. Gutachter: Prof. Dr. Anton Bovier Tag der Promotion: 09.12.2020 Erscheinungsjahr: 2021 Dedicated to the memory of Barbara and Josef Schweiger, and of Albert Zandtner

Acknowledgements

This thesis would not have been possible without the support of many people.

I have been very fortunate to have Stefan Müller as my advisor, and I would like to thank him for his support and the many insights he has shared with me. He has given me a lot of freedom to pursue my research and follow my interests, in particular, when the topic of this thesis shifted into initially unforeseen directions. I very much enjoyed cooperating with him on two research projects that are now part of this thesis. He has offered me a lot of very useful advice and I have learnt many things from him.

I would like to thank Anton Bovier for the advice on various topics he has given me, and for agreeing to be the second referee of this thesis. I also thank Herbert Koch for accompanying my PhD as my mentor in the BIGS, and him and Christoph Thiele from the LIMES Institute for agreeing to be members of my thesis committee.

I had the opportunity to spend three exciting months at the Courant Institute of NYU. I would like to thank Bob Kohn for hosting me there, and for many insightful discussions. During that time I also met Ofer Zeitouni, and I want to thank him for some inspiring ideas and his encouragement.

I would like to thank Endre Süli for hosting me at the University of Oxford for a few days, and for his great hospitality there. I would like to thank him as well as Jean-Dominique Deuschel and Noemi Kurt for very enjoyable cooperations on joint projects that would become parts of this thesis.

I have also benefitted from mathematical discussions with a number of other people in Bonn, in particular Simon Buchholz, Sergio Conti, Johannes Diermeier, Maximilian Fels, Behrend Heeren, Richard Höfer, Malte Lackmann, Mareike Lager, and Juan Velazquez. I would like to especially thank Simon for joining forces with me in what would result in a part of this thesis. Frauke Beeken was very helpful whenever there was an administrative issue, and I am grateful for her support. I want to thank all my colleagues at the Institute for Applied Mathematics for creating a great working environment, for regular rounds of Schafkopf and other fun activities.

I want to thank Malte and Simon for reading parts of this thesis and suggesting various improvements.

I am grateful for a scholarship from the Bonn International Graduate School in Mathematics, a scholarship from the Studienstiftung des Deutschen Volkes, a position in the Collaborative Research Center 1060, and funding from the Global Math Network, all of which have supported me financially during my PhD.

Last but not least, I would like to thank my family, especially my parents, and my friends for their support and encouragement.

Contents

1.1Overview11.2Random interface models31.2.1Motivation31.2.2Statistical mechanics51.2.3Spin systems, Gibbs measures and random interface models61.2.4Examples of random interface models71.2.5Critical dimensions and log-correlated fields121.2.6Mathematical tools to study random interface models131.2.7Sampling from the models171.3Properties of the membrane model and other interface models191.3.1Gibbs measures and scaling limits191.3.2Extrema of the field231.3.3Entropic repulsion231.3.4Pinning261.3.5Wetting291.3.6Further aspects and open questions291.4Discrete Green's functions and finite differential equations and Green's functions311.4.1Continuous elliptic partial differential equations321.4.3Estimates for finite difference schemes for the Bilaplacian371.5Notation382Estimates for the Green's function of the discrete Bilaplacian in dimensions two and three412.1Introduction472.2.2Function sands ic properties492.3.1Definitions and basic properties492.3.1Definitions and inequalities502.4Interpolation522.5Inner decay estimates for discrete biharmonic functions: special c	1	Intro	Introduction 1				
1.2 Random interface models 3 1.2.1 Motivation 3 1.2.2 Statistical mechanics 5 1.2.3 Spin systems, Gibbs measures and random interface models 6 1.2.4 Examples of random interface models 7 1.2.5 Critical dimensions and log-correlated fields 12 1.2.6 Mathematical tools to study random interface models 13 1.2.7 Sampling from the models 17 1.3 Properties of the membrane model and other interface models 19 1.3.1 Gibbs measures and scaling limits 19 1.3.2 Extrema of the field 21 1.3.3 Entropic repulsion 23 1.3.4 Pinning 26 1.3.5 Wetting 29 1.3.6 Further aspects and open questions 29 1.4.1 Continuous elliptic partial differential equations and Green's functions 31 1.4.1 Continuous elliptic partial differential equations 32 1.4.3 Estimates for discrete Green's functions 34 1.4.4 Estimates for finite difference schemes for the Bilaplacian		1.1	Overview	L			
1.2.1Motivation31.2.2Statistical mechanics51.2.3Spin systems, Gibbs measures and random interface models61.2.4Examples of random interface models71.2.5Critical dimensions and log-correlated fields121.2.6Mathematical tools to study random interface models131.2.7Sampling from the models171.3Properties of the membrane model and other interface models191.3.1Gibbs measures and scaling limits191.3.2Extrema of the field211.3.3Entropic repulsion231.3.4Pinning261.3.5Wetting291.3.6Further aspects and open questions291.3.6Further aspects and open questions291.4.1Continuous elliptic partial differential equations and Green's functions311.4.2Numerical analysis of partial differential equations321.4.3Estimates for discrete Green's functions341.4.4Estimates for finite difference schemes for the Bilaplacian371.5Notation382Estimates for the Green's function of the discrete Bilaplacian in dimensions two and three412.1Introduction472.2.2Function spaces and inequalities472.3.1Definitions and basic properties492.3.2Caccioppoli inequalities502.4Interpolation522.5Inner decay estimates f		1.2	Random interface models 3	3			
1.2.2Statistical mechanics51.2.3Spin systems, Gibbs measures and random interface models61.2.4Examples of random interface models71.2.5Critical dimensions and log-correlated fields121.2.6Mathematical tools to study random interface models131.2.7Sampling from the models171.3Properties of the membrane model and other interface models191.3.1Gibbs measures and scaling limits191.3.2Extrema of the field211.3.3Entropic repulsion231.3.4Pinning261.3.5Wetting291.3.6Further aspects and open questions291.4Discrete Green's functions and finite difference schemes311.4.1Continuous elliptic partial differential equations and Green's functions311.4.2Numerical analysis of partial differential equations321.4.3Estimates for finite difference schemes for the Bilaplacian371.5Notation412.1Introduction412.2Preliminaries472.3.1Definitions and basic properties492.3.2Caccioppoli inequalities502.4Interpolation522.5Hulf-space552.5.2Half-space613.5States for discrete biharmonic functions: special cases544.4Estimates for discrete biharmonic functions: special cases542.5.1			1.2.1 Motivation	3			
1.2.3 Spin systems, Gibbs measures and random interface models 6 1.2.4 Examples of random interface models 7 1.2.5 Critical dimensions and log-correlated fields 12 1.2.6 Mathematical tools to study random interface models 13 1.2.7 Sampling from the models 17 1.3 Properties of the membrane model and other interface models 19 1.3.1 Gibbs measures and scaling limits 19 1.3.2 Extrema of the field 21 1.3.3 Entropic repulsion 23 1.3.4 Pinning 29 1.3.5 Wetting 29 1.3.6 Further aspects and open questions 29 1.4 Discrete Green's functions and finite difference schemes 31 1.4.1 Continuous elliptic partial differential equations and Green's functions 32 1.4.3 Estimates for finite difference schemes for the Bilaplacian 37 1.5 Notation 38 2 Estimates for the Green's function of the discrete Bilaplacian in dimensions two and three 41 2.1 Introduction 47 </td <td></td> <td></td> <td>1.2.2 Statistical mechanics</td> <td>5</td>			1.2.2 Statistical mechanics	5			
1.2.4Examples of random interface models71.2.5Critical dimensions and log-correlated fields121.2.6Mathematical tools to study random interface models131.2.7Sampling from the models171.3Properties of the membrane model and other interface models191.3.1Gibbs measures and scaling limits191.3.2Extrema of the field211.3.3Entropic repulsion231.3.4Pinning261.3.5Wetting291.3.6Further aspects and open questions291.3.6Further aspects and open questions291.3.6Further aspects and open questions291.4Discrete Green's functions and finite differential equations and Green's functions311.4.2Numerical analysis of partial differential equations321.4.3Estimates for finite difference schemes for the Bilaplacian371.5Notation382Estimates for the Green's function of the discrete Bilaplacian in dimensions two and three412.1Introduction412.2Function spaces and inequalities472.3.1Definitions and basic properties492.3.2Caccioppoli inequalities502.4Interpolation522.5Inter decay estimates for discrete biharmonic functions: special cases542.5.1Full space552.5.2Half-space612.5.3Edges and vertice			1.2.3 Spin systems, Gibbs measures and random interface models 6	5			
1.2.5Critical dimensions and log-correlated fields121.2.6Mathematical tools to study random interface models131.2.7Sampling from the models171.3Properties of the membrane model and other interface models191.3.1Gibbs measures and scaling limits191.3.2Extrema of the field211.3.3Entropic repulsion231.3.4Pinning261.3.5Wetting291.3.6Further aspects and open questions291.3.6Further aspects and open questions291.4Discrete Green's functions and finite difference schemes311.4.1Continuous elliptic partial differential equations and Green's functions311.4.2Numerical analysis of partial differential equations321.4.3Estimates for discrete Green's functions341.4.4Estimates for finite difference schemes for the Bilaplacian371.5Notation38 2 Estimates for the Green's function of the discrete Bilaplacian in dimensions two and three412.1Introduction472.2.2Function spaces and inequalities472.3The discrete Bilaplacian equation492.3.1Definitions and basic properties492.3.2Caccioppoli inequalities502.4Interpolation522.5Inner decay estimates for discrete biharmonic functions: special cases542.5.1Full space55<			1.2.4 Examples of random interface models	7			
1.2.6Mathematical tools to study random interface models131.2.7Sampling from the models171.3Properties of the membrane model and other interface models191.3.1Gibbs measures and scaling limits191.3.2Extrema of the field211.3.3Entropic repulsion231.3.4Pinning261.3.5Wetting291.3.6Further aspects and open questions291.3.6Further aspects and open questions291.4Discrete Green's functions and finite difference schemes311.4.1Continuous elliptic partial differential equations and Green's functions311.4.2Numerical analysis of partial differential equations321.4.3Estimates for discrete Green's functions341.4.4Estimates for finite difference schemes for the Bilaplacian371.5Notation382Estimates for the Green's function of the discrete Bilaplacian in dimensions twoand three2.1Introduction412.2Function spaces and inequalities472.3.1Definitions and basic properties492.3.2Caccioppoli inequalities502.4Interpolation522.5.1Full space552.5.2Half-space612.5.3Edges and vertices642.6Inner and outer decay estimates for discrete biharmonic functions: special cases542.5.1Full space55 </th <th></th> <th></th> <th>1.2.5 Critical dimensions and log-correlated fields</th> <th>2</th>			1.2.5 Critical dimensions and log-correlated fields	2			
1.2.7Sampling from the models171.3Properties of the membrane model and other interface models191.3.1Gibbs measures and scaling limits191.3.2Extrema of the field211.3.3Entropic repulsion231.3.4Pinning261.3.5Wetting291.3.6Further aspects and open questions291.3.6Further aspects and open questions291.4Discrete Green's functions and finite difference schemes311.4.1Continuous elliptic partial differential equations and Green's functions311.4.2Numerical analysis of partial differential equations321.4.3Estimates for discrete Green's functions341.4.4Estimates for discrete Green's functions341.4.5Notation382Estimates for the Green's function of the discrete Bilaplacian in dimensions two and three412.1Introduction412.2Function spaces and inequalities472.3The discrete Bilaplacian equation492.3.1Definitions and basic properties492.3.2Caccioppoli inequalities502.4Interpolation522.5Inner and outer decay estimates for discrete biharmonic functions: special cases542.5.241Function space552.5.241Function space552.5.44141413.5111111<			1.2.6 Mathematical tools to study random interface models	3			
1.3 Properties of the membrane model and other interface models 19 1.3.1 Gibbs measures and scaling limits 19 1.3.2 Extrema of the field 21 1.3.3 Entropic repulsion 23 1.3.4 Pinning 29 1.3.5 Wetting 29 1.3.6 Further aspects and open questions 29 1.4.1 Continuous elliptic partial difference schemes 31 1.4.2 Numerical analysis of partial differential equations and Green's functions 31 1.4.2 Numerical analysis of partial differential equations 32 1.4.3 Estimates for discrete Green's functions 34 1.4.4 Estimates for finite difference schemes for the Bilaplacian 37 1.5 Notation 38 2 Estimates for the Green's function of the discrete Bilaplacian in dimensions two and three 2.1 Introduction 41 2.2 Function spaces and inequalities 47 2.2.1 Notation 47 2.2.2 Function sand basic properties 49 2.3.1 Definitions and basic properties			1.2.7 Sampling from the models	7			
1.3.1 Gibbs measures and scaling limits 19 1.3.2 Extrema of the field 21 1.3.3 Entropic repulsion 23 1.3.4 Pinning 26 1.3.5 Wetting 29 1.3.6 Further aspects and open questions 29 1.3.6 Further aspects and open questions 29 1.4.1 Continuous elliptic partial difference schemes 31 1.4.2 Numerical analysis of partial differential equations and Green's functions 31 32 1.4.3 Estimates for discrete Green's functions 34 1.4.4 Estimates for finite difference schemes for the Bilaplacian 37 1.5 Notation 38 2 Estimates for the Green's function of the discrete Bilaplacian in dimensions two and three 41 2.1 Introduction 41 2.2 Function spaces and inequalities 47 2.3.1 Definitions and basic properties 49 2.3.2 Caccioppoli inequalities 50 2.4 Interpolation 52 2.5 Inner decay estimates for discrete biharmonic functions: special cases 54 </th <th></th> <th>1.3</th> <th>Properties of the membrane model and other interface models</th> <th>)</th>		1.3	Properties of the membrane model and other interface models)			
1.3.2 Extrema of the field 21 1.3.3 Entropic repulsion 23 1.3.4 Pinning 26 1.3.5 Wetting 29 1.3.6 Further aspects and open questions 29 1.3.6 Further aspects and open questions 29 1.4 Discrete Green's functions and finite differential equations and Green's functions 31 1.4.1 Continuous elliptic partial differential equations and Green's functions 32 1.4.3 Estimates for discrete Green's functions 34 1.4.4 Estimates for finite difference schemes for the Bilaplacian 37 1.5 Notation 38 2 Estimates for the Green's function of the discrete Bilaplacian in dimensions two and three 41 2.1 Introduction 47 2.2.1 Notation 47 2.2.2 Function spaces and inequalities 47 2.3.1 Definitions and basic properties 49 2.3.2 Caccioppoli inequalities 50 2.4 Interpolation 52 2.5 Inner decay estimates for discrete biharmonic fu			1.3.1 Gibbs measures and scaling limits)			
1.3.3 Entropic repulsion 23 1.3.4 Pinning 26 1.3.5 Wetting 29 1.3.6 Further aspects and open questions 29 1.3.6 Further aspects and open questions 29 1.4 Discrete Green's functions and finite difference schemes 31 1.4.1 Continuous elliptic partial differential equations and Green's functions 32 1.4.2 Numerical analysis of partial differential equations 32 1.4.3 Estimates for discrete Green's functions 34 1.4.4 Estimates for finite difference schemes for the Bilaplacian 37 1.5 Notation 38 2 Estimates for the Green's function of the discrete Bilaplacian in dimensions two and three 41 2.1 Introduction 41 2.2 Function spaces and inequalities 47 2.2.1 Notation 47 2.2.2 Function spaces and inequalities 47 2.3.1 Definitions and basic properties 49 2.3.2 Caccioppoli inequalities 50 2.4 Interpolation 52			1.3.2 Extrema of the field	L			
1.3.4 Pinning 26 1.3.5 Wetting 29 1.3.6 Further aspects and open questions 29 1.4 Discrete Green's functions and finite difference schemes 31 1.4.1 Continuous elliptic partial differential equations and Green's functions 31 1.4.2 Numerical analysis of partial differential equations and Green's functions 32 1.4.3 Estimates for discrete Green's functions 34 1.4.4 Estimates for finite difference schemes for the Bilaplacian 37 1.5 Notation 38 2 Estimates for the Green's function of the discrete Bilaplacian in dimensions two and three 1.1 Introduction 41 2.2 Preliminaries 47 2.2.1 Notation 47 2.2.2 Function spaces and inequalities 47 2.3 The discrete Bilaplacian equation 49 2.3.1 Definitions and basic properties 49 2.3.2 Caccioppoli inequalities 50 2.4 Interpolation 52 2.5 Inner decay estimates for discrete biharmonic functions: special case			1.3.3 Entropic repulsion	3			
1.3.5 Wetting 29 1.3.6 Further aspects and open questions 29 1.4 Discrete Green's functions and finite difference schemes 31 1.4.1 Continuous elliptic partial differential equations and Green's functions 31 1.4.2 Numerical analysis of partial differential equations and Green's functions 32 1.4.3 Estimates for discrete Green's functions 34 1.4.4 Estimates for finite difference schemes for the Bilaplacian 37 1.5 Notation 38 2 Estimates for the Green's function of the discrete Bilaplacian in dimensions two and three 41 2.1 Introduction 41 2.2 Preliminaries 47 2.2.1 Notation 47 2.2.2 Function spaces and inequalities 47 2.3.1 Definitions and basic properties 49 2.3.2 Caccioppoli inequalities 50 2.4 Interpolation 52 2.5 Inner decay estimates for discrete biharmonic functions: special cases 54 2.5.1 Full space 55 2.5.2 Half-space <t< th=""><th></th><th></th><th>1.3.4 Pinning</th><th>5</th></t<>			1.3.4 Pinning	5			
1.3.6 Further aspects and open questions 29 1.4 Discrete Green's functions and finite difference schemes 31 1.4.1 Continuous elliptic partial differential equations and Green's functions 31 1.4.2 Numerical analysis of partial differential equations 32 1.4.3 Estimates for discrete Green's functions 34 1.4.4 Estimates for finite difference schemes for the Bilaplacian 37 1.5 Notation 38 2 Estimates for the Green's function of the discrete Bilaplacian in dimensions two and three 41 2.1 Introduction 47 2.2.1 Notation 47 2.2.2 Function spaces and inequalities 47 2.3.1 Definitions and basic properties 49 2.3.2 Caccioppoli inequalities 50 2.4 Interpolation 52 2.5 Inner decay estimates for discrete biharmonic functions: special cases 54 2.5.1 Full space 55 2.5.2 Half-space 61 2.5.3 Edges and vertices 64 2.6 Inne			1.3.5 Wetting)			
1.4 Discrete Green's functions and finite difference schemes			1.3.6 Further aspects and open questions)			
1.4.1 Continuous elliptic partial differential equations and Green's functions 31 1.4.2 Numerical analysis of partial differential equations 32 1.4.3 Estimates for discrete Green's functions 34 1.4.4 Estimates for finite difference schemes for the Bilaplacian 37 1.5 Notation 38 2 Estimates for the Green's function of the discrete Bilaplacian in dimensions two and three 41 2.1 Introduction 47 2.2.1 Notation 47 2.2.2 Function spaces and inequalities 47 2.3.1 Definitions and basic properties 49 2.3.2 Caccioppoli inequalities 50 2.4 Interpolation 52 2.5 Inner decay estimates for discrete biharmonic functions: special cases 54 2.5.1 Full space 55 2.5.2 Half-space 61 2.5.3 Edges and vertices 64 2.6 Inner and outer decay estimates for discrete biharmonic functions: special cases 64		1.4	Discrete Green's functions and finite difference schemes	l			
1.4.2 Numerical analysis of partial differential equations 32 1.4.3 Estimates for discrete Green's functions 34 1.4.4 Estimates for finite difference schemes for the Bilaplacian 37 1.5 Notation 38 2 Estimates for the Green's function of the discrete Bilaplacian in dimensions two and three 41 2.1 Introduction 41 2.2 Preliminaries 47 2.2.1 Notation 47 2.2.2 Function spaces and inequalities 47 2.3.1 Definitions and basic properties 49 2.3.2 Caccioppoli inequalities 50 2.4 Interpolation 52 2.5 Inner decay estimates for discrete biharmonic functions: special cases 54 2.5.1 Full space 55 2.5.2 Half-space 61 2.5.3 Edges and vertices 64 2.6 Inner and outer decay estimates for discrete biharmonic functions 68			1.4.1 Continuous elliptic partial differential equations and Green's functions 31	l			
1.4.3 Estimates for discrete Green's functions 34 1.4.4 Estimates for finite difference schemes for the Bilaplacian 37 1.5 Notation 38 2 Estimates for the Green's function of the discrete Bilaplacian in dimensions two 38 2 Estimates for the Green's function of the discrete Bilaplacian in dimensions two 41 2.1 Introduction 41 2.2 Preliminaries 47 2.2.1 Notation 47 2.2.2 Function spaces and inequalities 47 2.3 The discrete Bilaplacian equation 49 2.3.1 Definitions and basic properties 49 2.3.2 Caccioppoli inequalities 50 2.4 Interpolation 52 2.5 Inner decay estimates for discrete biharmonic functions: special cases 54 2.5.1 Full space 55 2.5.2 Half-space 61 2.5.3 Edges and vertices 64 2.6 Inner and outer decay estimates for discrete biharmonic functions 68			1.4.2 Numerical analysis of partial differential equations	2			
1.4.4 Estimates for finite difference schemes for the Bilaplacian 37 1.5 Notation 38 2 Estimates for the Green's function of the discrete Bilaplacian in dimensions two and three 41 2.1 Introduction 41 2.2 Preliminaries 47 2.2.1 Notation 47 2.2.2 Function spaces and inequalities 47 2.3.1 Definitions and basic properties 49 2.3.1 Definitions and basic properties 50 2.4 Interpolation 52 2.5 Inner decay estimates for discrete biharmonic functions: special cases 54 2.5.2 Half-space 61 2.5.3 Edges and vertices 64 2.6 Inner and outer decay estimates for discrete biharmonic functions 68			1.4.3 Estimates for discrete Green's functions	1			
1.5 Notation 38 2 Estimates for the Green's function of the discrete Bilaplacian in dimensions two and three 41 2.1 Introduction 41 2.2 Preliminaries 47 2.2.1 Notation 47 2.2.2 Function spaces and inequalities 47 2.3 The discrete Bilaplacian equation 49 2.3.1 Definitions and basic properties 49 2.3.2 Caccioppoli inequalities 50 2.4 Interpolation 52 2.5 Inner decay estimates for discrete biharmonic functions: special cases 54 2.5.3 Edges and vertices 61 2.5.3 Edges and vertices 64 2.6 Inner and outer decay estimates for discrete biharmonic functions: special cases 64			1.4.4 Estimates for finite difference schemes for the Bilaplacian	7			
2 Estimates for the Green's function of the discrete Bilaplacian in dimensions two and three 41 2.1 Introduction 41 2.2 Preliminaries 47 2.2.1 Notation 47 2.2.2 Function spaces and inequalities 47 2.3 The discrete Bilaplacian equation 49 2.3.1 Definitions and basic properties 49 2.3.2 Caccioppoli inequalities 50 2.4 Interpolation 52 2.5 Inner decay estimates for discrete biharmonic functions: special cases 54 2.5.2 Half-space 61 2.5.3 Edges and vertices 64 2.6 Inner and outer decay estimates for discrete biharmonic functions 68		1.5	Notation	3			
and three412.1Introduction412.2Preliminaries472.2.1Notation472.2.2Function spaces and inequalities472.3The discrete Bilaplacian equation492.3.1Definitions and basic properties492.3.2Caccioppoli inequalities502.4Interpolation522.5Inner decay estimates for discrete biharmonic functions: special cases542.5.1Full space552.5.2Half-space612.5.3Edges and vertices642.6Inner and outer decay estimates for discrete biharmonic functions68	ი	Feti	mates for the Green's function of the discrete Pilanlacian in dimensions two				
2.1Introduction412.2Preliminaries472.2.1Notation472.2.2Function spaces and inequalities472.3The discrete Bilaplacian equation492.3.1Definitions and basic properties492.3.2Caccioppoli inequalities502.4Interpolation522.5Inner decay estimates for discrete biharmonic functions: special cases542.5.2Half-space612.5.3Edges and vertices642.6Inner and outer decay estimates for discrete biharmonic functions68	2	and	three dreen's function of the discrete bilaplacian in dimensions two	•			
2.1 Inforduction 41 2.2 Preliminaries 47 2.2.1 Notation 47 2.2.2 Function spaces and inequalities 47 2.3 The discrete Bilaplacian equation 47 2.3.1 Definitions and basic properties 49 2.3.2 Caccioppoli inequalities 50 2.4 Interpolation 52 2.5 Inner decay estimates for discrete biharmonic functions: special cases 54 2.5.1 Full space 55 2.5.2 Half-space 61 2.5.3 Edges and vertices 64 2.6 Inner and outer decay estimates for discrete biharmonic functions 68		2.1	Introduction 41	י 1			
2.2 Interimination 47 2.2.1 Notation 47 2.2.2 Function spaces and inequalities 47 2.3 The discrete Bilaplacian equation 49 2.3.1 Definitions and basic properties 49 2.3.2 Caccioppoli inequalities 50 2.4 Interpolation 52 2.5 Inner decay estimates for discrete biharmonic functions: special cases 54 2.5.1 Full space 55 2.5.2 Half-space 61 2.5.3 Edges and vertices 64 2.6 Inner and outer decay estimates for discrete biharmonic functions 68		2.1		נ 7			
2.2.1 Notation 47 2.2.2 Function spaces and inequalities 47 2.3 The discrete Bilaplacian equation 49 2.3.1 Definitions and basic properties 49 2.3.2 Caccioppoli inequalities 50 2.4 Interpolation 52 2.5 Inner decay estimates for discrete biharmonic functions: special cases 54 2.5.1 Full space 55 2.5.2 Half-space 61 2.5.3 Edges and vertices 64 2.6 Inner and outer decay estimates for discrete biharmonic functions 68		2.2	2.2.1 Notation 47	7			
2.3 The discrete Bilaplacian equation 49 2.3.1 Definitions and basic properties 49 2.3.2 Caccioppoli inequalities 50 2.4 Interpolation 52 2.5 Inner decay estimates for discrete biharmonic functions: special cases 54 2.5.1 Full space 55 2.5.2 Half-space 61 2.5.3 Edges and vertices 64 2.6 Inner and outer decay estimates for discrete biharmonic functions 68			2.2.1 Notation space and inequalities	7			
2.3 1 Definitions and basic properties 49 2.3.1 Definitions and basic properties 50 2.3.2 Caccioppoli inequalities 50 2.4 Interpolation 52 2.5 Inner decay estimates for discrete biharmonic functions: special cases 54 2.5.1 Full space 55 2.5.2 Half-space 61 2.5.3 Edges and vertices 64 2.6 Inner and outer decay estimates for discrete biharmonic functions 68		23	The discrete Bilaplacian equation)			
2.3.1 Definitions and basic properties 13 2.3.2 Caccioppoli inequalities 50 2.4 Interpolation 52 2.5 Inner decay estimates for discrete biharmonic functions: special cases 54 2.5.1 Full space 55 2.5.2 Half-space 61 2.5.3 Edges and vertices 64 2.6 Inner and outer decay estimates for discrete biharmonic functions 68 2.6.1 Imper estimates 67		2.0	2.3.1 Definitions and basic properties 49	ý			
2.4 Interpolation 52 2.5 Inner decay estimates for discrete biharmonic functions: special cases 54 2.5.1 Full space 55 2.5.2 Half-space 61 2.5.3 Edges and vertices 64 2.6 Inner and outer decay estimates for discrete biharmonic functions 68			2.3.1 Deminions una suste properties))			
2.1 Interpotation · · · · · · · · · · · · · · · · · · ·		24	Interpolation 52	,			
2.5 Full space 55 2.5.1 Full space 55 2.5.2 Half-space 61 2.5.3 Edges and vertices 64 2.6 Inner and outer decay estimates for discrete biharmonic functions 68 2.6 Inner and outer decay estimates for discrete biharmonic functions 68		2.1	Inner decay estimates for discrete bibarmonic functions: special cases 54	-			
2.5.1 Half-space 61 2.5.2 Half-space 61 2.5.3 Edges and vertices 64 2.6 Inner and outer decay estimates for discrete biharmonic functions 68 2.6 Inner and outer decay estimates for discrete biharmonic functions 68		2.0	2.5.1 Full space	5			
 2.5.2 Finin space			2.5.1 Full space 61	í			
2.6 Inner and outer decay estimates for discrete biharmonic functions			2.5.2 Flan space	י 1			
2.0 Inter and outer decay estimates for discrete orian monit functions 00		26	Inner and outer decay estimates for discrete hibermonic functions	r Q			
/ h L Inner estimates		2.0	261 Inner estimates 66 46	, 2			

		2.6.2	Outer estimates via duality			. 69				
2.7 The discrete full-space Green's function						. 72				
	2.8	Proof	of the main theorem			. 74				
		2.8.1	Lower bounds for $G_h(x, x)$. 74				
		2.8.2	Upper bounds for $G_h(x, y)$. 75				
		2.8.3	Convergence of Green's functions			. 82				
3	Pro	robability to be positive for the membrane model in dimensions two and three 84								
	3.1	Introd	luction			. 85				
		3.1.1	Main results			. 85				
		3.1.2	Implications for entropic repulsion			. 86				
		3.1.3	Notation			. 87				
	3.2	Prelin	ninaries			. 87				
3.3		Lowe	r bounds			. 89				
		3.3.1	Local smallness of the field			. 89				
		3.3.2	Global smallness of the field			. 91				
		3.3.3	Change of measure			. 92				
	3.4	Uppe	r bounds			. 94				
4	The	maxim	um of the four-dimensional membrane model			97				
	4.1	Introd	luction			. 97				
		4.1.1	Main result for the membrane model			. 97				
		4.1.2	Log-correlated fields			. 98				
		4.1.3	Green's function estimates			. 99				
		4.1.4	Notation			. 102				
	4.2	Prelin	ninaries			. 103				
		4.2.1	Discrete Inequalities			. 103				
		4.2.2	Estimates for finite difference schemes			. 106				
	4.3	Estim	ates for other Green's functions			. 109				
		4.3.1	Estimates for the discrete Green's function of the full space			. 109				
		4.3.2	Estimates for continuous Green's functions			. 111				
	4.4	Proof	of the main theorems		•••	. 115				
5 Optimal order finite difference approximation of generalized solutions to the						ji-				
	harmonic equation in a cube									
	5.1	Introd	luction			. 123				
		5.1.1	Main results			. 123				
		5.1.2	Outline of the proof			. 125				
		5.1.3	Notation and preliminaries			. 126				
	5.2	Discre	ete trace and inverse trace theorems			. 127				
		5.2.1	Construction of a good extension			. 127				
		5.2.2	Estimate of the boundary values			. 128				
		5.2.3	A discrete inverse trace theorem on the cube			. 134				
	5.3	Estim	ates for the finite difference schemes			. 138				
		5.3.1	Summation-by-parts formulae and Poincaré inequalities			. 138				
		5.3.2	Proofs of the main theorems			. 141				
	5.4	Furth	er remarks			. 142				
		5.4.1	Variants and extensions			. 142				

		5.4.2	Density results	144
		5.4.3	Remarks on Interpolation	145
6	Pinn	ning for	the membrane model in dimension four and above	151
	6.1	Introd	uction	151
		6.1.1	Main results	151
		6.1.2	Heuristics: The continuous Bilaplace equation in a perforated domain	155
		6.1.3	Main ideas of the proofs	158
		6.1.4	Notation and preliminaries	162
	6.2	Struct	ure of the pinned set	163
		6.2.1	Correlation inequalities	163
		6.2.2	Estimates on the pinned set	165
	6.3	Some	inequalities	176
		6.3.1	A discrete multipolar Hardy-Rellich inequality	177
		6.3.2	An interpolation inequality	183
	6.4	Proba	bilistic decay of the L^2 -norm for biharmonic functions $\ldots \ldots \ldots \ldots$	184
		6.4.1	Bad boxes and cut-off functions	185
		6.4.2	Decay estimates on good domains	193
		6.4.3	Sparsity of bad boxes	198
	6.5	Pathw	rise bounds on the field	208
		6.5.1	Quenched estimates on the Green's function	208
		6.5.2	Estimates on variance and covariance	214
		6.5.3	Existence of the thermodynamic limit of the field	217

Bibliography

1 Introduction

1.1 Overview

This thesis is concerned with two objects from seemingly different fields: on the one hand, the membrane model from probability theory, and on the other hand, the Green's function of the discrete Bilaplacian from the theory of partial differential equations (PDEs) or numerical analysis. As it turns out, though, these two objects are closely related, and an improved understanding of one of them can also help in the study of the other.

The membrane model is an example of a random interface model. Such models arise when studying the thermal fluctuations in interfaces in physics, chemistry and biology. The membrane model, specifically, is relevant when this interface is very flexible so that the bending modes dominate. It is used, for instance, to describe the behaviour of biomembranes such as those in the walls of cells.

There are also purely mathematical reasons to be interested in the membrane model. Namely, it is a natural variant of the most fundamental example of a random interface model, the so-called discrete Gaussian free field. This model has some particularly nice features (such as random walk representations and correlation inequalities), and in the last three decades this model has been very well understood. On the other hand, there are general classes of random interface models where very little is known. The membrane model might now serve as a stepping stone to investigate these more complicated models. Namely, it shares some of the features of the discrete Gaussian free field, but not all of them. Thus, even though the membrane model is expected to behave similarly to the discrete Gaussian free field, one is forced to develop new techniques to make this rigorous.

In this thesis we study various features of the membrane model, filling several gaps in the literature. In each case the answer had previously been known for the discrete Gaussian free field, and we extend these results to the case of the membrane model.

In Chapter 3 we study the effect of a hard wall that forces the field to be positive on the interface, focussing on dimensions 2 and 3. This chapter is based on the publication [BDKS19] which is joint work with Simon Buchholz, Jean-Dominique Deuschel and Noemi Kurt and has appeared in the **Electronic Communications in Probability**.

In Chapter 4 we investigate the behaviour of the maximum of the field in dimension 4. This chapter is based on the publication [Sch20a] which has appeared in **The Annals of Probability**.

Chapter 6 is dedicated to the effects that a small attractive potential has on the interface, in dimensions 4 and above. This potential localizes the field, and we study how this localization manifests itself. This chapter is based on the preprint [Sch20b] that will be submitted for publication.

The starting point for all these results is the fact that the covariance function of the membrane model is the Green's function of the discrete Bilaplacian. Heuristically, this object should behave similarly to the Green's function of the continuous Bilaplacian. A major part of this thesis will be to make this heuristic rigorous. To that end we combine various

methods from PDE theory with methods from numerical analysis.

In fact, in Chapter 2 we prove estimates for the Green's function in dimensions 2 and 3 using a compactness argument and results for continuous elliptic equations in domains with singularities. This chapter is the basis for the results in Chapter 3. It is based on the publication [MS19], which is joint work with Stefan Müller and has appeared in the **Vietnam Journal of Mathematics**.

Furthermore, as a part of Chapter 4 we prove estimates for the Green's function in dimension 4, using estimates for finite difference schemes and preexisting results on the continuous Green's function.

For the application in Chapter 4 we do not need the full strength of the estimates for the finite difference scheme we use. Optimizing such estimates is, however, very interesting for numerical analysis itself. In Chapter 5 we improve the best known estimates on the approximation quality of the scheme we use in Chapter 4 and some other schemes. This chapter is based on the publication [MSS20] which is joint work with Stefan Müller and Endre Süli and has appeared in the **SIAM Journal on Numerical Analysis**.

In this introductory chapter we will lay the foundation for these results and discuss the necessary background, and we will give a more detailed description of the results in the following chapters.

In Section 1.2 we discuss random interface models and their basic properties. We begin by describing the motivation from physics and biochemistry for the study of these models. Each random interface model is given as a probability measure, the so-called Gibbs measure, for a certain Hamiltonian, and so we explain the physical background as well as the mathematical theory underlying these measures. We then introduce some important examples of random interface models (including the membrane model), and describe some of their basic properties. In particular, we discuss the existence of subcritical, critical and supercritical dimensions. We also survey the most important mathematical tools used to study random interface models. Finally, we describe how to simulate random interface models on a computer, and how the images throughout this introduction were generated.

In Section 1.3 we then give more details on the membrane model. We compare it with a few other random interface models, namely the discrete Gaussian free field (or gradient model), the $\nabla \varphi$ -model with strictly convex \mathcal{V} and the $\nabla \varphi$ -model with slightly non-convex \mathcal{V} . We discuss various aspects of these models, reviewing the existing results in the literature and describing the new contributions of this thesis. We begin with infinite volume limits of the interfaces, and then discuss the maximum of the fields. Afterwards, we discuss the phenomena of entropic repulsion, pinning, and wetting. Finally, we mention a few further interesting questions. As part of this section, we outline the results of Chapters 3, 4 and 6 on entropic repulsion for the subcritical membrane model, the maximum of the critical membrane model, and pinning for the critical and supercritical membrane model, respectively.

In Section 1.4 we describe the connection between the membrane model and the Green's function of the discrete Bilaplacian, and discuss discrete Green's functions more generally. We begin with a summary of some facts from elliptic PDE theory and numerical analysis. We then focus on discrete Green's functions and describe the tools available to study them. As part of this we summarize the results of Chapters 2 and 4 on the subcritical and critical Green's function of the discrete Bilaplacian. The estimates from Chapter 4 are based on estimates for a certain finite difference scheme. We explain this connection, and describe further results on such schemes that are contained in Chapter 5.

Most of the content of this introduction is an exposition of well-known results in the literature. Other than the summaries of the results of the later chapters, the only slightly original parts are the description of some algorithms to generate samples from the membrane model in Section 1.2.7, and the discussion of Hessian Gibbs measures in Section 1.3.1.

1.2 Random interface models

In this section we will give some background on random interface models, describe some examples and discuss basic properties.

1.2.1 Motivation

Macroscopic interfaces

In physics there are many systems that can form sharp interfaces. Let us discuss two main examples.

As a first example, consider a substance that can be in the solid, liquid or gaseous phase. Under certain circumstances two or more of these phases can coexist, and there will be interfaces between them. For example, for water at 0° Celsius and standard pressure both ice and liquid water can appear. More generally, a variety of materials can form stable crystals within a surrounding liquid. We assume that the system is in equilibrium. This assumption is not always reasonable (e.g. ice crystals look very different than the conjectured equilibrium shape), but for some materials such as small crystals of certain metals it aligns well with experiments [RW84]. Under this equilibrium assumption, the macroscopic theory of interfaces for crystals was pioneered by Wulff [Wul01]. He proposed that the atoms in the crystal arrange themselves in a shape $U \subset \mathbb{R}^3$ so that U minimizes the "Wulff functional"

$$\mathcal{W}(U) = \int_{\partial U} \sigma(n(x)) \, \mathrm{d}\mathcal{H}^2(x)$$

under the constraint that the volume of U is fixed. Here $n \in S^2$ is a normal vector to ∂U , and σ is the so-called surface tension. This variational problem is an anisotropic variant of the isoperimetric problem. Its minimizer (the Wulff shape) can be constructed using the so-called Wulff construction, and in practice this variational problem is well understood. Note that if we write ∂U locally as the graph of a function $u: A \subset R^2 \to \mathbb{R}$, then the integrand in the Wulff functional becomes a certain functional of ∇u .

Our second example of interfaces is from biology, and for details on the following see [Lip95]. Many biological membranes, such as the wall of a cell, are formed by bilayers of lipids. Lipids are molecules with a hydrophilic head and a hydrophobic tail, and in a solution they can arrange themselves in quite stable double layers with the hydrophobic tails pointing inwards. Such a structure is called a bilayer. The lipid molecules in the bilayer are typically in the liquid phase, meaning that the single lipids can move almost freely around in the membrane while maintaining the bilayer structure. The lipid bilayers are quite resistant to stretching (instead they rupture before stretching significantly) but have low resistance to bending. This suggests that the energy of a bilayer should depend mainly on the curvature of the bilayer. Indeed, Helfrich [Hel73] (cf. also [BWW17]) proposed a variational problem for the surface Σ occupied by a closed bilayer, namely that it minimizes the "Helfrich functional"

$$\mathbf{H}(\Sigma) = \int_{\Sigma} \frac{k_c}{2} (H - c_0)^2 + \overline{k} K \mathrm{d} \, \mathcal{H}^2(x) \, .$$

Here *H* and *K* are the mean and Gaussian curvature, respectively, k_c and k are bending moduli and c_0 is the spontaneous curvature. By the Gauss-Bonnet theorem the integral over $\bar{k}K$ evaluates to $2\pi \bar{k}\chi(\Sigma)$, where $\chi(\Sigma)$ is the Euler characteristic of Σ . Thus, for fixed topology of the bilayer we can neglect the second summand and study

$$\tilde{\mathrm{H}}(\Sigma) = \int_{\Sigma} \frac{k_c}{2} (H - c_0)^2 \mathrm{d}\mathcal{H}^2(x)$$

instead. We can again write Σ locally as the graph of a function $u: A \subset \mathbb{R}^2 \to \mathbb{R}$. If the spontaneous curvature is close to 0 and the bilayer is locally almost planar, we can neglect ∇u , and the dominant term looks like $\frac{k_c}{8} |\nabla^2 u|^2$. In particular, the functional now involves only second derivatives.

Microscopic interfaces

The next question then is to analyse the interfaces that we just described on a microscopic level, to study thermic fluctuations and to derive the respective functionals from atomistic theories. The tools for this issue are provided by statistical mechanics (cf. the next section).

For our first example of a solid-to-liquid transition this line of research was initiated by Dobrushin, Kotecký and Shlosman [DKS92], who analysed macroscopic interfaces arising from the Ising model, and since then there have been many works in that direction, cf. [BIV00]. On a macroscopic level, the Wulff shape is a deterministic subset of \mathbb{R}^3 , and we can represent its boundary locally as the graph of a function $\mathbb{R}^2 \to \mathbb{R}$. On the microscopic scale, however, the interface will not be stationary, and there will be fluctuations. We make the rather strong simplifying assumption that these fluctuations can locally be represented as the graph of a random function $\Lambda \subset \mathbb{R}^2 \to \mathbb{R}$ (and thus we in particular assume that there are no holes or overhangs in our interface). As we are interested in atomistic models, it is reasonable to assume that Λ is discrete, and we take it to be a subset of the lattice \mathbb{Z}^2 or $(h\mathbb{Z})^2$. We are thus left to study random height functions on Λ . The probability to observe a certain height function (i.e. a certain microscopic configuration) will depend on the temperature of the system and on the energy of that configuration. It turns out that this energy is (at least approximately) a certain functional of the discrete gradient. This is unsurprising when one compares it with the Wulff functional itself.

For our second example of bilayer membranes one can, in principle, proceed similarly. We are not aware of any rigourous mathematical derivation of the properties of membrane bilayers from atomistic models, but there many results in the physics literature on thermal fluctuations in bilayers and how they influence the macroscopic properties of the membrane (see e.g. [DGT06, NP87, Lip95, HL97, RCMS05]). On a microscopic level these fluctuations are once again governed by an energy consisting of discrete curvature terms, and so we arrive once again at the problem of studying the arising probability distribution of height functions.

More generally, a random interface model will be a probability distribution on functions $\Lambda \to \mathbb{R}$, where $\Lambda \subset \mathbb{R}^d$. Before we discuss the actual random interface models that we are interested in, let us discuss the physical and mathematical background necessary to choose and define these probability distributions.

1.2.2 Statistical mechanics

In our examples of interfaces we encountered the situation that the interface takes a certain deterministic macroscopic equilibrium shape. Its microscopic state, meanwhile, is hard or rather impossible to predict, and moreover it will change incredibly fast, so that it does not really make sense to speak of "a" microscopic state, but rather of an ensemble of such. This idea can best be formalized by using concepts of statistical mechanics.

More generally, statistical mechanics is concerned with the study of systems with many degrees of freedom, and with deriving their behaviour from their microscopic structure. The subject was introduced in the late 19th century by Boltzmann, Gibbs, Maxwell and others, and has since developed into an important part of modern physics. A landmark reference is [Gib02], and a comprehensive treatment can be found in [LL58]. We will briefly discuss the notions that are most important to us, mostly following [Tho72, FV18].

We would like to describe the macroscopic behaviour of a physical system consisting of a large number of constituents (e.g. the atoms in a crystal or the molecules that form a gas). In classical physics, such a system can be parametrized by the positions and momenta of all the constituents, and knowing these, the Hamilton equations describe the state of the systems for all future times. In practice, however, this is completely infeasible: there is no practical way to know the initial state of a system. Even if one did, the evolution of the system would be incredibly complicated due to the huge number of constituents. Furthermore, we are not actually interested in the detailed evolution of the constituents, but rather in the evolution of some macroscopic quantities.

The starting point of statistical mechanics is thus to replace the given microscopic initial state ω that we have no way to know with a probability distribution \mathbb{P} over the set of all microstates Ω . Of course, this probability distribution should be supported only on those microstates that are compatible with our knowledge about the macrostate. For our analysis we need a Hamiltonian $H: \Omega \to \mathbb{R}$ that gives the energy of each microstate, and an a priori measure λ on Ω (typically the Hausdorff measure). We restrict ourselves to the case that our systems are static on a macroscopic scale.

Suppose that all we know about our system is that it consists of *N* particles that are located in some $\Lambda \subset \mathbb{R}^d$ with volume $|\Lambda| = V$ and that the energy of our system is some constant *E*. Denoting by $\Omega_{\Lambda,N}$ the set of all microstates compatible with the assumptions on the number of particles and their occupied volume, it has then been postulated by Gibbs that if we have no further information the equilibrium measure on $\Omega_{\Lambda,N}$ should be given by

$$\mathbb{P}^{\mathrm{mic}}_{\Lambda, E, N}(\mathrm{d}\omega) = \frac{1}{Z_{\Lambda, E, N}} \mathbb{1}_{H(\omega) = E} \mathbb{1}_{\omega \in \Omega_{\Lambda, N}} \lambda(\mathrm{d}\omega) \,.$$

Here $\mathbb{1}_s$ is equal to 1 if *s* is true and otherwise 0. This is the so-called microcanonical ensemble, and the normalization factor $Z_{\Lambda,E,N}$ is the so-called microcanonical partition function.

Of greater interest to us, however, is a different ensemble where we do not fix the energy, but the temperature. Here we need to proceed differently as it is not clear how to define the temperature $T(\omega)$ of a microstate ω . Physically, one way to prescribe the temperature is to assume that our system is in contact with a heat reservoir, i.e. with another system with which it can exchange energy. If that other system is very large, this will lead to both systems being at approximately the temperature of the heat reservoir. Assuming that both systems together are described by the microcanical ensemble, one can, in principle, calculate the marginal distribution of the system we are interested in. When one pursues this calculation

at least on a heuristic level, one obtains the equilibrium measure

$$\mathbb{P}^{\operatorname{can}}_{\Lambda,\beta,N}(\mathrm{d}\omega) = \frac{1}{Z_{\Lambda,\beta,N}} \exp(-\beta H(\omega)) \mathbb{1}_{\omega \in \Omega_{\Lambda,N}} \lambda(\mathrm{d}\omega) \,.$$

Here β is proportional to $\frac{1}{T}$, and we choose units in such a way that the proportionality constant is equal to 1. This is the so-called canonical ensemble, and $Z_{\Lambda,\beta,N}$ is the canonical partition function. The factor $\exp(-\beta H(\omega))$ is called Boltzmann weight.

Another viewpoint on the canonical ensemble is that it is chosen in such a way as to maximize the relative entropy of $\mathbb{P}^{can}_{\Lambda,\beta,N}$ with respect to λ under the constraint that the expected value of the energy takes some fixed value.

The canonical ensemble is also called the (canonical) Gibbs measure. The Boltzmann weight is the larger, the smaller $H(\omega)$ is. If β is large (i.e. the temperature is low) then the Gibbs measure will mostly be supported on those states with small energy. On the other hand, if β is small (i.e. the temperature is high) then the Gibbs measure does not discriminate as much between states with lower and higher energy.

Often one does not know (or care about) the precise value of *N* beyond the fact that it is very large. This suggests that one should directly study the system in the limit $N \to \infty$. Then the volume *V* needs to grow simultaneously in such a way that the particle density $\frac{N}{V}$ has a finite limit. This procedure is called taking the thermodynamic limit.

1.2.3 Spin systems, Gibbs measures and random interface models

We now apply the theory of the previous section to the case of random interface models, aiming for mathematical rigor. We follow [Geo88, Bov06, FV18]. In particular, in [Geo88] the theory is described in much greater generality.

We are looking for a discretized mathematical model of an interface. This interface is formed by a set of particles in \mathbb{R}^{d+1} such that the first d coordinates of each particle are fixed, while the (d + 1)-th coordinate is free.

We thus consider an at most countable $S \subset \mathbb{R}^d$ as the parameter set (we will take $S = \mathbb{Z}^d$ or $S = (h\mathbb{Z})^d$ for some h > 0), and a set $E \subset \mathbb{R}$ (we will take $E = \mathbb{R}$ or $E = \mathbb{Z}$) with the Borel σ -algebra \mathcal{E} and a reference measure λ as the single spin space. We consider the set $\Omega = E^S$ of all possible configurations, equipped with the product σ -algebra $\mathcal{F} = \mathcal{E}^S$. Let $\mathcal{P}(\Omega, \mathcal{F})$ be the set of all probability measures on Ω . For a $\psi \in \Omega$ we write ψ_x for the value of ψ at $x \in S$, and for $\Lambda \subset S$ we let $\mathcal{F}_{\Lambda} = \sigma(\psi_x : x \in \Lambda)$.

Given a \mathcal{F} -measurable Hamiltonian $H: \Omega \to \mathbb{R} \cup \{+\infty\}$ we would now like to define the canonical ensemble as in the previous section, i.e. the measure

$$\mathbb{P}(\mathrm{d}\psi) = \frac{1}{Z_{\beta}} \exp(-\beta H(\psi)) \prod_{x \in S} \lambda(\mathrm{d}\psi_x).$$

If *S* is finite this definition works well. However, reasonable Hamiltonians will be infinite when *S* is infinite, and so we need a different construction in that case. The crucial idea here, due to Dobrushin, Lanford and Ruelle [Dob68, Dob70, LR69], is to define a measure μ as a Gibbs measure for *H*, if for every finite $\Lambda \subset S$ the law of the field under μ conditioned on the values of the field outside of Λ is the correct one.

We restrict ourselves to finite-range interactions. That is, we consider an interaction of the form $\Phi = {\Phi_A}_{A \in S}$, where $\Phi_A : S \to \mathbb{R}$ is \mathcal{F}_A -measurable (and by $A \in S$ we denote

that *A* is a compact, i.e. finite, subset of *S*). We also assume $\sup_{A \in \mathbb{Z}^d} \|\Phi_A\|_{L^{\infty}} < \infty$, and that $\Phi_A = 0$ when |A| > R for some *R*. Then for $\Lambda \in S$ we can define the Hamiltonian

$$H_\Lambda(\psi) = \sum_{\substack{A \Subset S \ A \cap \Lambda
eq arnothing
eq arnothing}} \Phi_A(\psi)$$

and for $\varphi \in \Omega$ the Gibbs specification

$$\mathbb{P}_{\Phi,\Lambda,\beta}^{(\varphi)}(\mathrm{d}\psi) = \frac{1}{Z_{\Phi,\Lambda,\beta}^{(\varphi)}} \exp(-\beta H_{\Lambda}(\psi)) \prod_{x \in \Lambda} \lambda(\mathrm{d}\psi_x) \prod_{x \in S \setminus \Lambda} \delta_{\varphi_x}(\mathrm{d}\psi_x) \,.$$

One can check that this definition is self-consistent in the sense that if $\Lambda \subset \Lambda'$ then

$$\mathbb{P}_{\Phi,\Lambda',\beta}^{(\varphi)}(\mathrm{d}\varphi')\mathbb{P}_{\Phi,\Lambda,\beta}^{(\varphi')} = \mathbb{P}_{\Phi,\Lambda',\beta}^{(\varphi)} \quad \text{for each } \varphi \in \Omega.$$
(1.2.1)

We then define that a probability measure $\mathbb{P}_{\Phi,\beta}$ on (Ω, \mathcal{F}) is a Gibbs measure for Φ if

$$\mathbb{P}_{\Phi,\beta}(E \mid \mathcal{F}_{\Lambda^c}) = \mathbb{P}_{\Phi,\Lambda,\beta}^{(\cdot)}(E) \quad \mathbb{P}_{\Phi,\beta}\text{-a.s. for each } E \in \mathcal{F} \text{ and } \Lambda \Subset \mathbb{Z}^{\mathsf{d}}.$$

One can show that $\mathbb{P}_{\Phi,\beta}$ is a Gibbs measure if and only if it satisfies the analogue of (1.2.1) in infinite volume, i.e. if and only if for any $\Lambda \subset S$

$$\mathbb{P}_{\Phi,\beta}(\mathrm{d}\varphi')\mathbb{P}_{\Phi,\Lambda,\beta}^{(\varphi')} = \mathbb{P}_{\Phi,\beta}\,. \tag{1.2.2}$$

This equation is called the DLR equation after Dobrushin, Lanford and Ruelle. The relation (1.2.2) formalizes the intuition that a system is in equilibrium if any of its subsystems is in equilibrium and thus distributed according to the canonical Gibbs measure.

Of course, this raises the question of existence and uniqueness of Gibbs measures. In our case the spin space is non-compact, and so neither question is trivial. In fact, it may happen that there is no infinite volume Gibbs measure or that there are infinitely many. One may hope to construct a Gibbs measure by choosing a specific sequence of domains (e.g. $\Lambda_N = [-N, N]^d \cap \mathbb{Z}^d$) together with a choice of boundary data (e.g. $\varphi = 0$) and considering a weak limit of the corresponding sequence of finite volume Gibbs measures (the so-called thermodynamic limit). If such a weak limit exists, it is easy to see that in our setting it will be a Gibbs measure.

We will discuss the question of existence and uniqueness of Gibbs measures in more detail once we have introduced some examples of random interface models.

1.2.4 Examples of random interface models

Now that we have laid the theoretical foundations, we can introduce and describe some important examples of random interface models. As explained in the previous section, to describe an interface model we need an interaction Φ and an inverse temperature β . From a physical point of view it would be important to treat these two objects separately. However, all the models we consider have Hamiltonians that allow an arbitrary positive prefactor (and so we can include β in that prefactor) or have a Hamiltonian that is positively homogenous of some degree (and so a change of β only scales the field by a deterministic factor). Thus, there is no loss when we set $\beta = 1$ in all of the following and omit it in our notation.

We choose $S = \mathbb{Z}^d$ as our parameter space. In most of the following, we describe continuous models in the sense that we take the single spin space $E = \mathbb{R}$. At the end we will briefly mention discrete models where $E = \mathbb{Z}$. For now, we always take the reference measure $\lambda(d\psi)$ to be the Lebesgue measure (simply denoted $d\psi$). We also write $d(x, \Lambda)$ for the (Euclidean) distance from x to Λ .

Discrete Gaussian free field

The first and most important example of a random interface model is the discrete Gaussian free field (also called gradient model or harmonic crystal). This model is given by the interaction $\Phi_A(\psi) = \frac{1}{2} \sum_{x \in A} |\nabla_1 \psi_x|^2$, where $\nabla_1 \psi_x := (D_i^1 \psi_x)_{i=1}^d := (\psi_{x+e_i} - \psi_x)_{i=1}^d$. This yields the measure

$$\mathbb{P}_{\nabla,\Lambda}^{(\varphi)}(\mathrm{d}\psi) = \frac{1}{Z_{\nabla,\Lambda}^{(\varphi)}} \exp\left(-\frac{1}{2} \sum_{\substack{x \in \mathbb{Z}^{\mathsf{d}} \\ d(x,\Lambda) \leq 1}} |\nabla_{1}\psi_{x}|^{2}\right) \prod_{x \in \Lambda} \mathrm{d}\psi_{x} \prod_{x \in \mathbb{Z}^{\mathsf{d}} \setminus \Lambda} \delta_{\varphi_{x}}(\mathrm{d}\psi_{x}) \,. \tag{1.2.3}$$

If $\varphi = 0$ we can remove the restriction on *x* in the sum, so that the measure with zero boundary values takes the form

$$\mathbb{P}_{\nabla,\Lambda}(\mathrm{d}\psi) = \frac{1}{Z_{\nabla,\Lambda}} \exp\left(-\frac{1}{2} \sum_{x \in \mathbb{Z}^{\mathsf{d}}} |\nabla_1 \psi_x|^2\right) \prod_{x \in \Lambda} \mathrm{d}\psi_x \prod_{x \in \mathbb{Z}^{\mathsf{d}} \setminus \Lambda} \delta_0(\mathrm{d}\psi_x) \tag{1.2.4}$$

where we drop the superscript (0) for brevity.

The Hamiltonian here is the discrete L^2 -norm of the gradient of ψ . It thus penalizes large slopes in ψ , in line with what we expected for the solid-liquid-interface models in Section 1.2.1. This model is particularly nice from a mathematical point of view. Namely, the Hamiltonian is a quadratic function of ψ , and so the measure is Gaussian.

The discrete Gaussian free field has a continuous analogue, the (continuous) Gaussian free field. Informally, this is the measure on functions $\psi \colon \mathbb{R}^d \to \mathbb{R}$ with density $\frac{1}{Z} \exp\left(-\frac{1}{2} \|\nabla \psi\|_{L^2}^2\right) d\psi$, but the actual definition is as a Gaussian measure on a negative Sobolev space, see e.g. [She07]. This measure appears as a scaling limit of a variety of models in probability, e.g. the dimer model in integrable probability [Ken01] or fields in random matrix theory [RV07]. It is also related to quantum field theory, where one tries to construct operator-valued Gaussian and non-Gaussian fields. This explains the name "free", since the Gaussian field corresponds to systems without interaction in that setting (cf. [GJ87]).

The discrete Gaussian free field will be one important example for us. We will mostly call it the gradient model, as this emphasizes the contrast to the membrane model (to be defined shortly).

$\nabla \varphi$ -interface models

In the context of the application to solid-to-liquid phase transitions in 1.2.1 there is no reason to assume that slopes are penalized precisely by $\frac{1}{2}|\cdot|^2$. If we instead use an arbitrary even function $\mathcal{V} \colon \mathbb{R} \to \mathbb{R}$, we obtain the (Ginzburg-Landau) $\nabla \varphi$ -model

$$\mathbb{P}_{\mathcal{V}(\nabla),\Lambda}^{(\varphi)}(\mathrm{d}\psi) = \frac{1}{Z_{\mathcal{V}(\nabla),\Lambda}^{(\varphi)}} \exp\left(-\sum_{\substack{x \in \mathbb{Z}^{\mathsf{d}} \\ d(x,\Lambda) \leq 1}} \sum_{i=1}^{\mathsf{d}} \mathcal{V}(D_{i}^{1}\psi_{x})\right) \prod_{x \in \Lambda} \mathrm{d}\psi_{x} \prod_{x \in \mathbb{Z}^{\mathsf{d}} \setminus \Lambda} \delta_{\varphi_{x}}(\mathrm{d}\psi_{x}) \quad (1.2.5)$$

and its variant with zero boundary data

$$\mathbb{P}_{\mathcal{V}(\nabla),\Lambda}(d\psi) = \frac{1}{Z_{\mathcal{V}(\nabla),\Lambda}} \exp\left(-\sum_{x\in\mathbb{Z}^d} \sum_{i=1}^d \mathcal{V}(D_i^1\psi_x)\right) \prod_{x\in\Lambda} d\psi_x \prod_{x\in\mathbb{Z}^d\setminus\Lambda} \delta_0(d\psi_x).$$
(1.2.6)

As we will explain in the following sections, this model behaves quite similarly to the discrete Gaussian free field when \mathcal{V} is strictly convex and satisfies some other mild regularity assumptions. Physically, this should not be surprising, as the convexity of \mathcal{V} ensures that mixtures of different slopes are energetically unfavourable in comparison to pure slopes. The regularity assumptions one needs for \mathcal{V} change from application to application. Typically one requires $\mathcal{V} \in C^2(\mathbb{R})$ and $c \leq \mathcal{V}''(x) \leq C$, but occasionally results in the literature require, e.g., that $\mathcal{V} \in C^{\infty}(\mathbb{R})$. In the following we will not be precise in this regard, and just speak of the $\nabla \varphi$ -model with strictly convex \mathcal{V} .

When \mathcal{V} is not convex, but in a suitable sense close to being convex, the model still behaves similarly to the Gaussian free field, although much less is known. Rigorous results on this are perturbative, and they require that \mathcal{V} is close to a strictly convex function in a sufficiently strong norm. Again we will be rather vague, and speak of the $\nabla \varphi$ -model with slightly non-convex \mathcal{V} in the following.

For the case that \mathcal{V} is far from being convex the model behaves very differently, see the discussion below.

The membrane model

Our discussion of lipid bilayers suggests that one should also study interface models involving second instead of first derivatives. The easiest such model is the membrane model where one considers the interaction $\Phi_A(\psi) = \frac{1}{2} \sum_{x \in A} |\Delta_1 \psi_x|^2$, where $\Delta_1 \psi_x := \sum_{i=1}^d \psi_{x+e_i} - 2\psi_x + \psi_{x-e_i}$. In principle $\frac{1}{2} \sum_{x \in A} |\nabla_1^2 \psi_x|^2$, where $\nabla_1^2 \psi_x := (D_i^1 D_{-j}^1 \psi_x)_{i,j=1}^d$, would be an equally natural choice, but a discrete integration by parts shows that this leads to exactly the same model. In any case, one obtains the probability measure

$$\mathbb{P}_{\Delta,\Lambda}^{(\varphi)}(\mathrm{d}\psi) = \frac{1}{Z_{\nabla,\Lambda}^{(\varphi)}} \exp\left(-\frac{1}{2} \sum_{\substack{x \in \mathbb{Z}^{\mathsf{d}} \\ d(x,\Lambda) \leq 2}} |\Delta_{1}\psi_{x}|^{2}\right) \prod_{x \in \Lambda} \mathrm{d}\psi_{x} \prod_{x \in \mathbb{Z}^{\mathsf{d}} \setminus \Lambda} \delta_{\varphi_{x}}(\mathrm{d}\psi_{x}) \tag{1.2.7}$$

or the zero boundary variant

$$\mathbb{P}_{\Delta,\Lambda}(\mathrm{d}\psi) = \frac{1}{Z_{\nabla,\Lambda}} \exp\left(-\frac{1}{2} \sum_{x \in \mathbb{Z}^{\mathsf{d}}} |\Delta_1 \psi_x|^2\right) \prod_{x \in \Lambda} \mathrm{d}\psi_x \prod_{x \in \mathbb{Z}^{\mathsf{d}} \setminus \Lambda} \delta_0(\mathrm{d}\psi_x) \,. \tag{1.2.8}$$

Once again this is a Gaussian measure. The Hamiltonian here penalizes high curvature.

This model also has a continuous relative, the (continuous) membrane model (see e.g. [CDH19] for a precise definition). In recent years this model has also been identified as the scaling limit of several fields in probability theory, e.g. the odometer in divisible sandpiles [CHR18] or certain spanning trees related to the loop-erased random walk [LSW19].

See Figure 1.1 for pictures of samples of the membrane model in various dimensions.

Other examples

Let us mention some other examples of random interface models. For simplicity we only give each model with zero boundary condition.

First of all, it remains to discuss the $\nabla \varphi$ -interface with a potential \mathcal{V} that is far from being convex. In that case the random interface can behave very differently from the models that we have mentioned so far. In particular, there can be phase transitions in the sense that there





Figure 1.1: Samples of the membrane model in dimension $d \in \{2, 3, 4, 5\}$ on the domain $\{0, \dots, 20\}^d$. The pictures show the values of the sample on the slice $\{0, \dots, 20\}^2 \times \{10\}^{d-2}$. See Section 1.2.7 for a description how the samples were generated.







Figure 1.1: Samples of the membrane model (continued)

may be more than one Gibbs measure with the same tilt. We will not discuss this further, but see [BK07, Buc19].

Next, just like the $\nabla \varphi$ -interface models are a generalization of the discrete Gaussian free field, one can consider the generalization of the membrane model where one uses an arbitrary even function $\mathcal{V} \colon \mathbb{R} \to \mathbb{R}$ instead of $\frac{1}{2} |\cdot|^2$. This yields the model

$$\mathbb{P}_{\Delta,\Lambda}(d\psi) = \frac{1}{Z_{\nabla,\Lambda}} \exp\left(-\sum_{x\in\mathbb{Z}^d} \sum_{i=1}^d \mathcal{V}(\Delta_1\psi_x)\right) \prod_{x\in\Lambda} d\psi_x \prod_{x\in\mathbb{Z}^d\setminus\Lambda} \delta_{\varphi_x}(d\psi_x).$$
(1.2.9)

This model should behave similarly to the membrane model when V is strictly convex or a small perturbation of a strictly convex function, while the case where V is far from being convex is presumably very complicated. The only work where this model is studied is [Kur12].

Of course, it is also reasonable to consider models where first and second derivatives are mixed. One such model has been studied very recently in [CDH20]. It is given by the probability measure

$$\mathbb{P}_{\Delta,\Lambda}(d\psi) = \frac{1}{Z_{\nabla,\Lambda}} \exp\left(-\sum_{x\in\mathbb{Z}^d} |\nabla_1\psi_x|^2 + a_\Lambda |\Delta_1\psi_x|^2\right) \prod_{x\in\Lambda} d\psi_x \prod_{x\in\mathbb{Z}^d\setminus\Lambda} \delta_{\varphi_x}(d\psi_x) \quad (1.2.10)$$

Here a_{Λ} is a scalar that may depend on Λ . In the limit $\Lambda \nearrow \infty$ it depends on the choice of a_{Λ} whether the behaviour of this model resembles that of the discrete Gaussian free field, that of the membrane model, or shows some genuinely new mixed behaviour.

It is also reasonable to not only use first and second order difference operators, but higher polyharmonic difference operators or various linear combinations. Such models are not studied much in the literature (an exception is [Sak03]). The reason for this probably is that these models do not add much from a mathematical point of view, while the notation becomes increasingly complicated. In particular, all the results for the membrane models to be discussed in the following should have analogues for the case of higher-order polyharmonic operators in place of Δ_1^2 .

All the models discussed so far have in common that the specification Ψ is translationinvariant. There are also interesting examples where this is not the case. In particular, one can sample the $(\Psi_{A+x})_{x\in\mathbb{Z}^d}$ from some random distribution. This leads to disordered random interface models. One can investigate their properties either for almost every realization of the specification, or averaged over the randomness of the specifications. The disorder can drastically change the behaviour of the field (see [Vel06], or e.g. [GL18] for recent work).

It is also possible to consider all the models considered so far with the continuous spin space $E = \mathbb{R}$ (with the Lebesgue measure) replaced by the discrete spin space $E = \mathbb{Z}$ (with the counting measure). The phenomenology is generally rather similar for those discrete models (at least when β is small enough), but there are also new phenomena when β is large and there is a so-called roughening transition (see e.g. [She05, Vel06]).

1.2.5 Critical dimensions and \log -correlated fields

In Figure 1.1 one can see that the membrane model becomes progressively rougher as the dimension increases. This phenomenon is not unique to the membrane model. In fact, for most of the models discussed above, there is a critical dimension so that the behaviour of

the field is very different depending on whether d is less than, equal or larger than that dimension. These regimes are called subcritical, critical and supercritical, respectively.

Typically, in the subcritical dimensions the variance of ψ_x for some $x \in \Lambda$ is unbounded and grows like a power of $d(x, \partial \Lambda)$, the distance of x to the boundary of Λ . In the critical dimension the variance of ψ_x is still unbounded but only grows like the logarithm of $d(x, \partial \Lambda)$, while in the supercritical dimensions the variance of ψ_x is bounded uniformly in xand Λ . In the following sections we describe a number of implications of this.

For now let us mention that the critical dimension of both the discrete Gaussian free field and the $\nabla \varphi$ -interface models is d = 2, while the critical dimension of the membrane model is d = 4. This already gives a hint of the heuristic that the d-dimensional membrane model behaves in many aspects like the (d – 2)-dimensional discrete Gaussian free field.

Of particular interest is the case of the critical dimension. This case is in many aspects borderline between the very different sub- and supercritical dimensions. It turns out that in the critical dimension not only the variances grow like the logarithm of the distance to the boundary, but also the covariances decay like the logarithm of the distance between the respective sites. Discrete random fields with these properties are called log-correlated fields. Of course one can also define (continuous) log-correlated fields. Both discrete and continuous log-correlated fields have been a topic of intensive study in the past years. In particular, there are various predictions of universality in this class, meaning that various properties of the field do not depend on the precise structure of the correlations, but only the fact that these decay logarithmically. See [DRSV14, DRSV17] for an overview.

The interest in log-correlated fields is fueled by the fact that these arise in a variety of contexts. Beyond the random interface models that we have already described, let us mention branching random walks and branching Brownian motion (see [Bov17] for an overview), the characteristic polynomial of certain random matrices and (closely related) the values of the Riemann ζ function on the critical line (see [FHK12, FK14] for important conjectures and [CMN18, ABB⁺19] and the references therein for recent rigorous results).

1.2.6 Mathematical tools to study random interface models

Before describing in detail what is known about the membrane model and the other random interface models, we will outline the main mathematical techniques that have been used to study these models. The message that we want to transmit here is the following: For the discrete Gaussian free field there exist many powerful techniques one can use to study it. The $\nabla \varphi$ -interface model with strictly convex \mathcal{V} and the membrane model both are more difficult than the discrete Gaussian free field. However, this difficulty manifests itself in different ways: for each of the two models only some of the tools survive while others can no longer be applied. Finally, the $\nabla \varphi$ -interface model with slightly non-convex \mathcal{V} is the most difficult model, and there are few existing techniques that can be applied. See Figure 1.2 for a schematic drawing of the relations between the models.

Markov property

The (domain) Markov property is a rather obvious, but nonetheless useful consequence of our definition of random interface models. It applies to all random interface models in the sense of our definition. For the following see [Fun05, Bis20]. Recall that we have assumed that the terms Φ_A in our interaction are 0 when diam $\Phi > R$. Then the field values in some $A \subset \Lambda$ depend only on the boundary data on the sites that are at most *R* away from *A*.



Figure 1.2: Relations between the random interface models

More formally, let $B = \{x \in \Lambda \setminus A : \operatorname{dist}(x, A) \leq R\}$. Then

$$\mathbb{P}_{\Phi,\Lambda}^{(\varphi)}(E \mid \mathcal{F}_{A^c}) = \mathbb{P}_{\Phi,\Lambda}^{(\varphi)}(E \mid \mathcal{F}_{B \setminus A}) \quad \text{for all } E \in \mathcal{F}_A \,.$$

For Gaussian measures such as the discrete Gaussian free field and the membrane model we can even give a somewhat explicit description of $\mathbb{P}_{\Phi,\Lambda}^{(\varphi)}(\cdot | \mathcal{F}_{B\setminus A})$. Suppose that ψ is distributed according to $\mathbb{P}_{\Phi,\Lambda}^{(\varphi)}$ and let $h_x = \mathbb{E}_{\Phi,\Lambda}^{(\varphi)}(\psi_x | \mathcal{F}_{B\setminus A})$ for $x \in A$. Then $\{\psi_x - h_x \colon x \in A\}$ is independent of \mathcal{F}_{A^c} and its law is given by $\mathbb{P}_{\Phi,\Lambda}$.

Random walk representations

Random walk representations exist for the discrete Gaussian free field and the $\nabla \varphi$ -interface models with strictly convex \mathcal{V} . We begin with the former case, where the situation is less complicated.

Recall that the discrete Gaussian free field is a centred Gaussian measure, and as such it is determined by its covariance matrix. It is easy to see (cf. e.g. [Fun05]) that this covariance matrix is equal to the Green's function of the symmetric random walk on \mathbb{Z}^d killed when leaving Λ . That is, when $(X_t)_{t\geq 0}$ is the path of a (continuous time) symmetric random walk on \mathbb{Z}^d that jumps at rate 2d to a uniformly chosen neighbour and \mathbb{E}^x denotes the expectation with respect to the law of $(X_t)_{t\geq 0}$ when $X_0 = x$, then

$$\mathbb{P}_{\nabla,\Lambda}(\psi_x\psi_y) = \mathbb{E}^x\left(\int_0^{T_{\Lambda^c}} \mathbb{1}_{X_t=y} \mathrm{d}t\right)$$
(1.2.11)

where $T_{\Lambda^c} = \inf\{n \ge 0 \colon X_n \notin \Lambda\}$. This representation allows to use estimates for random walks to conclude estimates on the behaviour of the covariance.

For $\nabla \varphi$ -interface models we do not have a representation as simple as (1.2.11). However, there is a more complicated version due to Helffer and Sjöstrand [HS94] that allows to write

$$\mathbb{P}_{\mathcal{V}(\nabla),\Lambda}(\psi_x\psi_y) = \mathbb{E}^{\delta_x \otimes \mathbb{P}_{\mathcal{V}(\nabla),\Lambda}}\left(\int_0^{T_{\Lambda^c}} \mathbb{1}_{X_t=y} \mathrm{d}t\right).$$
(1.2.12)

Here $(X_t)_{t\geq 0}$ describes a continuous time random walk in a time-dependent random environment given by the Langevin dynamics of the field. This random walk exists when \mathcal{V} is strictly convex. See [Fun05] for the details. Actually (1.2.12) is only a special case of the full Helffer-Sjöstrand representation, which applies to the covariances of arbitrary observables of the field. Of course, this representation is only useful if one is also able to control the right-hand side. This amounts to understanding random walks in a random environment, a subject where the methods of quantitative stochastic homogenization apply. The Helffer-Sjöstrand representation was first used in the study of $\nabla \varphi$ -interface models by Naddaf and Spencer in [NS97] with later refinements in [DGI00, GOS01]. Recently Armstrong and Wu [AW19] have made further progress by systemically using the emerging theory of quantitative stochastic homogenization as described in [AKM19].

Let us note an important fact: clearly the existence of a random walk representation for the covariance as in (1.2.11) or (1.2.12) implies that the covariance is pointwise nonnegative, and indeed this is the case for the discrete Gaussian free field as well as the $\nabla \varphi$ -interface model with strictly convex \mathcal{V} . For the membrane model, however, the covariance is, in general, not pointwise nonnegative, and so there is no random walk representation. One way to see this is to explicitly compute the covariance on some small domains. For example, for d = 2 and $\Lambda = \{0, \ldots, 20\}^2$ the Green's function is negative in the corners of the domain. Closely related to the question of nonnegativity of the Green's function of the continuous Bilaplacian. We will discuss the latter in Section 1.4.1. See [Gia01, Appendix A.2] for a more detailed discussion of random walk representations.

PDE estimates for the Green's function

For some of the Gaussian models, in particular the discrete Gaussian free field and the membrane model, one can alternatively apply estimates from PDE theory and numerical analysis. As they are Gaussian measures, they are determined by their covariance matrix. The point is that an easy calculation (cf. again [Fun05]) reveals that the covariance matrix of the discrete Gaussian free field is also given by the Green's function of $-\Delta_1$ on Λ with zero boundary data, i.e. that $\mathbb{P}_{\nabla,\Lambda}(\psi_x\psi_y) = G_{\nabla,\Lambda}(x, y)$, where for each $y \in \Lambda$ the function $G_{\nabla,\Lambda}(\cdot, y)$ is the solution of the partial differential equation

$$-\Delta_1 G_{\nabla,\Lambda}(\cdot, y) = \delta_y \qquad \text{in } \Lambda$$

$$G_{\nabla,\Lambda}(\cdot, y) = 0 \qquad \text{on } \mathbb{Z}^{\mathsf{d}} \setminus \Lambda.$$
(1.2.13)

Similarly, one can show that $\mathbb{P}_{\Delta,\Lambda}(\psi_x \psi_y) = G_{\Delta,\Lambda}(x, y)$, where $G_{\Delta,\Lambda}(\cdot, y)$ is the Green's function of Δ_1^2 , i.e. the solution of the partial differential equation

$$\Delta_1^2 G_{\Delta,\Lambda}(\cdot, y) = \delta_y \qquad \text{in } \Lambda$$

$$G_{\Delta,\Lambda}(\cdot, y) = 0 \qquad \text{on } \mathbb{Z}^d \setminus \Lambda.$$
(1.2.14)

Thus, one can hope to use methods from the theory of partial differential equations to derive results for $G_{\nabla,\Lambda}(\cdot, y)$ and $G_{\Delta,\Lambda}(\cdot, y)$. Furthermore, one can consider (1.2.13) and (1.2.14) as finite difference schemes for the Laplacian and Bilaplacian, respectively, and so apply tools from numerical analysis. In fact, these are the main approaches used in this thesis. We postpone a more detailed exposition to Section 1.4.

Correlation inequalities

Another very important tool in the study of random interface models are correlation inequalities. We describe two of them: The FKG inequality and the Gaussian correlation inequality. The former can be applied to the discrete Gaussian free field and the $\nabla \varphi$ -interface models, the latter to all Gaussian measures (and so in particular to the discrete Gaussian free field and the membrane model). These are not the only correlation inequalities important for the study of random interface models, though. Let us mention the Brascamp-Lieb inequality (cf. [Fun05, Section 4.2] or [Gia01, Appendix A.2]) and the GKS inequality (cf. [DMRR92, Appendix A]).

However, we focus on the FKG-inequality and the Gaussian correlation inequality. We begin with the former, introduced by (and named after) Fortuin, Kasteleyn and Ginibre [FKG71]. In the context of random interface models this is the statement that whenever *A*, *B* are events that are increasing (i.e. if $\psi \in A$ and $\psi' \ge \psi$ pointwise then $\psi' \in A$, and similarly for *B*) then

$$\mathbb{P}_{\Lambda}(A \cap B) \geq \mathbb{P}_{\Lambda}(A) \mathbb{P}_{\Lambda}(B) .$$

This inequality is an extremely powerful tool. One typical application is to use that conditioning on some increasing event (say, the field being large on some subset of Λ) increases the probability of some other increasing event (say, the field being large on another subset of Λ). For $\nabla \varphi$ -interface models with stricty convex \mathcal{V} the FKG inequality follows easily from the Helffer-Sjöstrand representation, cf. [Fun05]. For Gaussian interface models it is even easier to decide whether the FKG inequality holds: According to a criterion of Pitt [Pit82] this is the case if and only if the correlation matrix is elementwise nonnegative. As already mentioned that is the case for the discrete Gaussian free field but not for the membrane model. So we see that the membrane model does not satisfy the FKG inequality.

The other correlation inequality that we want to discuss, the Gaussian correlation inequality, looks quite similar. This is the statement that whenever *A*, *B* are closed events that are symmetric around 0 then

$$\mathbb{P}_{\Lambda}\left(A \cap B\right) \geq \mathbb{P}_{\Lambda}\left(A\right) \mathbb{P}_{\Lambda}\left(B\right) \,.$$

It was a longstanding open conjecture that this inequality holds for all Gaussian measures \mathbb{P}_{Λ} . This conjecture was settled in 2014 by Royen [Roy14] (see also [LM17] for an exposition of the proof). There is no reason to believe that this correlation inequality holds for general (log-concave) probability measures such as the $\nabla \varphi$ -interface models with stricty convex \mathcal{V} . For instance, a counterexample can be constructed out of a suitable Gaussian measure conditioned to be small, similar to Remark 6.2.1.

Other techniques

Let us mention a few other techniques that have been employed in the study of the membrane model. We have already briefly mentioned the Langevin dynamics associated to a $\nabla \varphi$ -interface model with strictly convex \mathcal{V} . Beyond their occurrence in the Helffer-Sjöstrand representation they can also be used directly to understand the underlying Gibbs measure. This was pioneered by Funaki and Spohn [FS97]. For that model also some other techniques such as Sheffield's cluster swapping [She05] exist.

Finally, we need to mention some techniques available for the $\nabla \varphi$ -interface model with slightly non-convex \mathcal{V} . The main technical tools used for that model are various implementations of the renormalization group from theoretical physics, ranging from a one-step renormalization scheme in [CDM09] to a very subtle multi-step renormalization scheme in [ABKM19].

1.2.7 Sampling from the models

Before discussing more detailed properties of the models, let us describe how to efficiently generate samples of a random interface model. In particular, we want to explain how the samples in Figure 1.1 and in Figures 1.3 and 1.4 were generated. The problem is that while we have an explicit expression for the Boltzmann weight exp(-H), the partition function is given by a high-dimensional integral that is very hard to calculate explicitly or numerically, so a direct computation of the probability density seems infeasible. Let us describe some alternative approaches.

Gibbs sampler

A general method for sampling from a Gibbs measure \mathbb{P} is to use a Markov chain Monte Carlo method, or, more precisely, a Gibbs sampler (see e.g [LPW09, Section 3.3]. The idea here is that while the joint density of all field heights is too complicated to understand, the conditional density of the height of a single site while all other heights are fixed is easy to compute. This suggests an algorithm where we iteratively resample the field at a (randomly chosen) site. This resampling defines a Markov chain $(\psi^{(k)})_{k=0}^{\infty}$ whose unique stationary measure is the Gibbs measure we want to sample from. Under some weak assumptions on the Markov chain, for any initial configuration $\psi^{(0)}$ the distribution of $\psi^{(k)}$ converges (as $k \to \infty$) in total variation distance to \mathbb{P} . In practice, one needs to decide how to choose k. For this one would need to know how far the distribution of $\psi^{(k)}$ still is from \mathbb{P} , i.e. how fast the Markov chain mixes. Unfortunately, this is in general a very hard problem (see again [LPW09] for a review of what is known), and so in practice one often has to make a guess.

We have implemented this algorithm in Matlab to generate samples from the membrane model on the domains $\{0, ..., N\}^d$. It is clear that we need to resample each site at least once to have a chance to see the actual behaviour of the model, and so we need at least $(N + 1)^d$ iterations. In practice, for N = 20 and d = 4 it seems that 10^2N^d iterations are not quite enough (as the maximum of the field is not as high as the theory predicts), while 10^3N^d iterations seem reasonable. This means that to sample the membrane model for N = 20 and d = 4 we should use at least $10^3N^d \approx 2 \cdot 10^8$ iterations, while for d = 5 one would need already approximately $4 \cdot 10^9$ iterations. For comparison, the author's laptop computer was able to process about 10^6 iterations per second.

We have generated Figures 1.3 and 1.4 using this method (with $2 \cdot 10^8$ iterations), as its flexibility allowed to easily include the single site potentials that will appear there. For Figure 1.1 we used another approach to be described below.

Naive direct sampling for Gaussian measures

When sampling from a Gaussian measure such as the discrete Gaussian free field or the membrane model, it is also possible to use the Gaussian structure of the measure to directly sample from the measure. This has the advantage that we can be sure that our sample has the correct distribution, and we do not need to guess how many iterations to use.

We focus on the membrane model, but the situation is similar for other Gaussian Gibbs measures. The probability distribution (1.2.8) of the membrane model can be rewritten as

$$\mathbb{P}_{\Delta,\Lambda}(\mathrm{d}\psi) = \frac{1}{Z_{\nabla,\Lambda}} \exp\left(-\frac{1}{2}(\psi_x,\Delta_1^2|_{\Lambda\times\Lambda}\psi_x)\right)_{L^2(\Lambda)} \prod_{x\in\Lambda} \mathrm{d}\psi_x \prod_{x\in\mathbb{Z}^d\setminus\Lambda} \delta_{\varphi_x}(\mathrm{d}\psi_x)$$

where $(\cdot, \cdot)_{L^2(\Lambda)}$ denotes the standard scalar product on \mathbb{R}^{Λ} . This means we need to sample a centred normal random variable with covariance $(\Delta_1^2|_{\Lambda \times \Lambda})^{-1}$. Such a sample is given by MX, where $X \in \mathbb{R}^{\Lambda}$ is a vector of i.i.d. standard Gaussians and $M \in \mathbb{R}^{\Lambda \times \Lambda}$ is a matrix such that $MM^t = (\Delta_1^2|_{\Lambda \times \Lambda})^{-1}$. An efficient way to compute a possible M is to compute the Cholesky decomposition $\Delta_1^2|_{\Lambda \times \Lambda} = LL^t$ with a lower triangular matrix L and then choosing $M = L^{-1}$.

We have implemented this algorithm in Matlab. It works well for small domains or for small dimensions, but for d = 5 and N = 20, we would have $|\Lambda| = (N + 1)^d \approx 4 * 10^6$, and computing the Cholesky decomposition of a matrix this large is infeasible (in fact, Matlab runs out of memory quickly when attempting this on a laptop computer).

Improved direct sampling for Gaussian measures

For Gaussian models there is also a better direct sampling algorithm. We again focus on the membrane model with zero boundary data. For this we use ideas of Sheffield [She07, Section 4.4] for the discrete Gaussian free field that can easily be adapted to the membrane model. The first observation is that the membrane model is much easier to sample when we first sample a complex version and our domain is a torus $(\mathbb{Z}/N'\mathbb{Z})^d$, as we can then use discrete Fourier analysis. For that end consider the scalar product $\langle \psi, \psi' \rangle := (\Delta_1 \psi, \Delta_1 \psi')_{L^2((\mathbb{Z}/N'\mathbb{Z})^d)}$ on the space *H* of lattice functions on $(\mathbb{Z}/N'\mathbb{Z})^d$ with average 0. Our goal is to sample from the Gaussian measure with density $\frac{1}{Z} \exp\left(-\frac{1}{2}\langle \psi, \psi \rangle\right) d\psi$, and one easily checks (cf. [She07, Proposition 2.1] that a sample ψ from that measure has the same law as $\sum_{j=1}^{N'^d-1} \varphi_j X_j$, where $(\varphi_j)_{j=1}^{N'^d-1}$ is a fixed orthonormal basis of *H*, and the X_j are i.i.d. standard Gaussians. We can generate such an orthonormal basis using the eigenfunctions of the discrete Bilaplacian. We take the functions

$$\psi_{k_1,\ldots,k_{\mathsf{d}}}(x) = \alpha_{k_1,\ldots,k_{\mathsf{d}}} \exp\left(i\frac{k_1x_1+\ldots+k_{\mathsf{d}}x_{\mathsf{d}}}{N'}\right)$$

for $k_l \in \{0, ..., d - 1\}$, not all 0, where

$$\alpha_{k_1,\dots,k_{\mathsf{d}}} = \frac{1}{4N'^{\mathsf{d}/2}\left(\sin^2\left(\frac{k_1\pi}{N'}\right) + \dots + \sin^2\left(\frac{k_1\pi}{N'}\right)\right)}$$

Thus, a fast way to generate a sample of the membrane model on a torus is to compute the array $(\alpha_{k_1,...,k_d} X_{k_1,...,k_d})_{k_1,...,k_d=1}^d$, where $X_{0,...,0} = 0$ and all other X are i.i.d. standard Gaussians, and then take the multidimensional discrete Fourier transform of that array. Both steps can be done extremely fast (in a few seconds even if d = 5 and N = 20).

Of course, this is not yet quite what we wanted, as we were looking for zero and not periodic boundary conditions. For this we use another observation from [She07, Section 4.4], namely, that the domain Markov property still applies. Thus, we embed our domain $\{0, \ldots, N\}^d$ into $(\mathbb{Z}/N'\mathbb{Z})^d$ for some $N' \ge N + 5$. Now, given a sample of the membrane model on $(\mathbb{Z}/N'\mathbb{Z})^d$, we only need to subtract the conditional expectation of the field given its values outside of $\{0, \ldots, N\}^d$ to obtain a sample of the membrane model with zero boundary data on $\{0, \ldots, N\}^d$. It remains to compute that conditional expectation, i.e. the biharmonic extension of the boundary values. For that purpose we need to solve the system $\Delta_1^2|_{\Lambda \times \Lambda} u = f$ for one right-hand side. This is a large system (about $4 \cdot 10^6$ unknowns if d = 5 and N = 20), but the matrix Δ_1^2 is very sparse, and we can use an iterative solver such as

Matlab's preconditioned conjugate gradient method to compute a very good approximative solution within several minutes of computation.

We have used this algorithm to generate the samples in Figure 1.1.

1.3 Properties of the membrane model and other interface models

We will now consider specific aspects of random interface models and describe what is known about them. We focus on the models described in Section 1.2.4, that is, the discrete Gaussian free field (or gradient model), the $\nabla \varphi$ -model with strictly convex \mathcal{V} , the membrane model, and the $\nabla \varphi$ -model with slightly non-convex \mathcal{V} . The reader should keep in mind the relation between these models illustrated in Figure 1.2. For most of the properties that we will mention, the answers for the gradient model are known. For quite a few of them the answer is also known for the $\nabla \varphi$ -model with strictly convex \mathcal{V} and the membrane model (and this thesis makes some further progress in the case of the membrane model). For the $\nabla \varphi$ -model with slightly non-convex \mathcal{V} , only little is known. In fact, in some of the following sections, we will not even mention it, as there are no results. It is a long-term hope that progress on models such as the membrane model or the $\nabla \varphi$ -model with strictly convex \mathcal{V} also furthers the understanding of this model.

For the following sections most of the material on the gradient model and the $\nabla \varphi$ -model with strictly convex \mathcal{V} is taken from [Fun05, Vel06].

1.3.1 Gibbs measures and scaling limits

Gibbs measures

After the discussion in Section 1.2.3 an obvious question is that of the existence of Gibbs measures for our random interface models. Because all the specifications we have considered are invariant under shifts of the field by a constant, it is clear that if there is a Gibbs measure, then there are infinitely many. Thus, a Gibbs measure is never unique, and the interesting question is whether one exists at all.

It turns out that this is the case if and only if the dimension is supercritical. Indeed, in supercritical dimensions the variance at a single site is uniformly bounded, as can be seen for the gradient model and the membrane model from straightforward estimates on the Green's function, for the $\nabla \varphi$ -model with strictly convex \mathcal{V} from the Brascamp-Lieb inequality and for $\nabla \varphi$ -model with slightly non-convex \mathcal{V} from the techniques in [ABKM19, Hil19]. Now the boundedness of the variances is easily seen to imply tightness of the finite volume Gibbs measures, and each subsequential weak limit will be a Gibbs measure. In the critical and subcritical dimensions the variances diverge as Λ grows, and so there is no hope for the existence of a Gibbs measure (cf. e.g. [Geo88] for a rigorous proof in the case of the gradient model).

Gradient Gibbs measures and Hessian Gibbs measures

It is of course quite unsatisfactory that there is no infinite volume Gibbs measure in the critical and subcritical dimensions. In the case of the gradient model and the $\nabla \varphi$ -model one alternative way to make sense of an infinite volume limit of the field is to use so-called gradient Gibbs measures. The idea here is to consider not the field heights ψ_x but their gradients $\nabla_1 \psi_x$, and to pass to a limit of the field of gradients. The limit field should then

also satisfy some variant of the DLR equations (1.2.2). Making this intuition rigorous is somewhat technical and we refer to [FS97] and in particular [She05, Section 3.1] for the details.

The point for the gradient model is that while $\operatorname{Var}_{\nabla,\Lambda}(\psi_x) = G_{\nabla,\Lambda}(x,x)$ diverges in dimensions d \leq 2, we have

$$\begin{aligned} \operatorname{Var}_{\nabla,\Lambda}(D_i^1\psi_x) &= \mathbb{E}_{\nabla,\Lambda}(\psi_{x+e_i} - \psi_x)^2 \\ &= G_{\nabla,\Lambda}(x+e_i, x+e_i) - 2G_{\nabla,\Lambda}(x, x+e_i) + G_{\nabla,\Lambda}(x, x) \\ &= D_{i,x}^1 D_{i,y}^1 G_{\nabla,\Lambda}(x, x) \end{aligned}$$

where $D_{i,x}^1$ and $D_{i,y}^1$ denote the discrete derivative with respect to the first and second variable, respectively. That is, by passing to the gradients of ψ_x we have gained two derivatives. Now, $\nabla_{1,x}\nabla_{1,y}G_{\nabla,\Lambda}(x,x)$ is uniformly bounded for all $d \ge 1$, and so one can see that there is a gradient Gibbs measure for the gradient model in all dimensions. Choosing boundary data $\varphi_x = a \cdot x$ for some $a \in \mathbb{R}^d$, we can construct a gradient Gibbs measure with given tilt (i.e. expected value of the gradient at each single site) a. One can show that every shift invariant ergodic gradient Gibbs measure \mathbb{P} which is tempered (i.e. satisfies $\mathbb{E}(D_i^1\psi_x)^2 < \infty$ for all $x \in \mathbb{Z}^d$, $i \in \{1, \dots, d\}$) is equal to one of these gradient Gibbs measures with the corresponding tilt. Indeed, this follows easily from the uniform boundedness of $\nabla_{1,x}\nabla_{1,y}G_{\nabla,\Lambda}(x,x)$ together with [Geo88, Theorem 13.24 and Theorem 13.26] and the well-known fact that every bounded discrete harmonic function is constant.

This characterization of gradient Gibbs measure for the gradient model is actually a special case of a result by Funaki and Spohn [FS97] who have extended the above considerations to the $\nabla \varphi$ -model with strictly convex \mathcal{V} . Again gradient Gibbs measures exist in every dimension, and the tempered ones are parametrised by their tilt.

For the membrane model one can proceed similarly. But in view of the fact that the Hamiltonian now involves second derivatives instead of first ones, it is more natural to consider not the gradients of the field but their Hessians, i.e. consider "Hessian Gibbs measures". While it seems that this has not been worked out in detail anywhere in the literature, the construction can proceed analogously to [She05, Section 3.1], so we only sketch the outcome. A short calculation as above shows that

$$\operatorname{Var}_{\Delta,\Lambda}(D_{i}^{1}D_{-j}^{1}\psi_{x}) = \mathbb{E}_{\nabla,\Lambda}(\psi_{x+e_{i}} - \psi_{x} - \psi_{x+e_{i}-e_{j}} + \psi_{x-e_{j}})^{2} = D_{i,x}^{1}D_{-j,x}^{1}D_{i,y}^{1}D_{-j,y}^{1}G_{\Delta,\Lambda}(x,x)$$

That is, by passing to Hessian Gibbs measures we have gained four derivatives, and so Hessian Gibbs measures can be constructed in all dimensions. For every $A \in \mathbb{R}^{d \times d}$ by choosing the boundary values $\varphi_x = \frac{1}{2}x^t Ax$ we can find a Hessian Gibbs measure with expected value of the Hessian at every site equal to A. Just as for the gradient model, one sees that all shift-invariant ergodic Hessian Gibbs measures which are tempered (i.e. satisfy $\mathbb{E}(D_i^1 D_{-j}^1 \psi_x)^2 < \infty$ for all $x \in \mathbb{Z}^d$, $i, j \in \{1, \dots, d\}$) are given this way. For the proof one needs the fact that every bounded discrete biharmonic function on \mathbb{Z}^d is constant. This fact might not be completely obvious, but it can be shown, for example, by combining Lemma 2.3.1 and Lemma 2.5.1 from Chapter 2 to see that such a function needs to be affine.

Scaling limits

A different, but closely related question is whether it is possible to extract a scaling limit of the field. To do so, we first need to define an appropriate rescaling of the field. We choose

some bounded domain $D \subset \mathbb{R}^d$ and consider the domains $\Lambda_N = ND \cap \mathbb{Z}^d$. Then, if ψ_N is distributed according to $\mathbb{P}_{\Psi,\Lambda_N}$ and $\alpha \in \mathbb{R}$, we can consider the field $\tilde{\psi}_N(x) = N^{\alpha}\psi_N(Nx)$ as a random function on $D \cap \left(\frac{1}{N}\mathbb{Z}\right)^d$. Now there are various ways to interpolate $\tilde{\psi}_N$ to a function defined on all of D, and one can wonder whether for the right choice of α these random functions on D have a scaling limit, and if so, in which topologies the convergence to the scaling limit holds.

For the gradient model the answer is straightforward. The correct choice for α here is $\alpha = \frac{d-2}{2}$, and the scaling limit is the continuum Gaussian free field that was already mentioned in Section 1.2.4. In the subcritical dimension d = 1 the continuum Gaussian free field is nothing else than Brownian motion. As *D* is bounded and connected, it is an interval. The gradient model with zero boundary data on Λ is just a random walk bridge, and it is well-known that under rescaling with factor $N^{-\frac{1}{2}}$ and piecewise linear interpolation this random walk bridge converges to a Brownian bridge on *D*, where the convergence takes place in the Hölder spaces $C^{0,\gamma}$ for any $\gamma < \frac{1}{2}$. In the critical and supercritical dimensions the convergence takes place in negative Sobolev spaces. We interpolate $\tilde{\psi}_N$ in some reasonable way (e.g. piecewise affinely on a triangulation subordinate to \mathbb{Z}^d). Then the interpolated fields converge in $H^{-s}(D)$ for any $s > \frac{d-2}{2}$, where $H^{-s}(D)$ is the dual space of the Hilbert space $H_0^s(D)$ [She07]. It is also possible to prove analogous results for non-zero boundary conditions.

The $\nabla \varphi$ -model with strictly convex \mathcal{V} behaves similarly to the gradient model. In particular, we still take $\alpha = \frac{d-2}{2}$, and the scaling limit is still (a scalar multiple of) the continuum Gaussian free field. The main focus in the literature has been to show the convergence in \mathcal{D}' , the space of distributions. This was first shown in [NS97]. Whether one can upgrade this to convergence in law in some negative Sobolev space then depends on the precise assumptions made on \mathcal{V} . In [GOS01] it is shown that under fairly general assumptions the convergence holds in $H^{-s}(D)$ for any s > d + 1. The situation with nonzero boundary conditions was investigated in [Mil11].

For the $\nabla \varphi$ -model with slightly non-convex \mathcal{V} one still has convergence to the continuum Gaussian free field [Hil16, ABKM19]. Somewhat surprisingly, this even holds for some very non-convex \mathcal{V} , at least in the zero-boundary case [BS11].

For the membrane model one observes a different scaling limit, namely the continuum membrane model that was also already mentioned in Section 1.2.4, and the correct choice for α turns out to be $\alpha = \frac{d-4}{2}$. Other than that, the situation is very similar as for the gradient model: provided that one chooses a sufficiently smooth interpolation, the convergence holds in the subcritical dimensions in the Hölder space $C^{\lfloor \gamma \rfloor, \{\gamma\}}(D)$ for any $\gamma < \frac{4-d}{2}$. This result can be found in [CDH19], with a crucial ingredient being the estimates in Chapter 2 of this thesis. In the critical and supercritical case the convergence holds in some negative Sobolev space $H^{-s}(D)$. The published result [CDH19] allows $s > s_*$ for some $s_* \approx \frac{7}{8}$ d, but it should be possible to improve their result to the optimal $s > \frac{d-4}{2}$.

1.3.2 Extrema of the field

This section includes parts of the introduction of the author's paper [Sch20a].

Existing results

We can now turn to discuss finer properties of the random interface models. The first such property is the behaviour of the extrema of the field. All of our models are invariant under

reflection around 0, and so we only need to consider the maximum. We always consider the field on the domain $\Lambda_N = [0, N]^d \cap \mathbb{Z}^d$. For $\Phi \in \{\nabla, \mathcal{V}(\nabla), \Delta\}$ we denote a sample from $\mathbb{P}_{\Phi, \Lambda_N}$ by ψ_N^{Φ} , and let $M_N^{\Phi} = \max_{x \in \Lambda_N} \psi_x$. We are interested in the asymptotics of the random variables M_N^{Φ} as $N \to \infty$. The answer here depends very much on whether the dimension is subcritical, critical or supercritical, and so we will discuss those cases separately.

In the subcritical dimensions (d = 1 for the gradient model and the $\nabla \varphi$ -models, d \leq 3 for the membrane model) we have that $\frac{M_N^{\nabla}}{N^{\frac{2-d}{2}}}$, $\frac{M_N^{\mathcal{V}(\nabla)}}{N^{\frac{2-d}{2}}}$ and $\frac{M_N^{\Delta}}{N^{\frac{4-d}{2}}}$ converge in distribution, which follows from the fact that the whole rescaled field converges weakly in C^0 , as discussed in the previous section.

In the supercritical dimensions (d \geq 3 for the gradient model and the $\nabla \varphi$ -models, d \geq 5 for the membrane model) the correlations decay rapidly, so one can expect that the maximum of the field behaves as if the ψ_N^{Φ} were independent. In the cases of the gradient model and the membrane model this can be made rigorous using Stein's method. Thereby it was shown in [CCH16b, CCH16a] that M_N^{∇} behaves as if the $(\psi_{N,x}^{\nabla})_{x \in V_N}$ were independent, i.e. that

$$\frac{\sqrt{2\mathsf{d}\log N}}{\sqrt{g_\mathsf{d}^\nabla}} \left(M_N^\nabla - \sqrt{2\mathsf{d}g_\mathsf{d}^\nabla\log N} + \frac{\sqrt{g_\mathsf{d}^\nabla}\left(\log(\mathsf{d}\log N) + \log 4\pi\right)}{\sqrt{8\mathsf{d}\log N}} \right)$$

converges in distribution to a Gumbel random variable, where x_N is a lattice point closest to the centre of $[0, N]^d$ and $g_d^{\nabla} = \lim_{N \to \infty} \operatorname{Var}(\psi_{N, x_N}^{\nabla})$. The analogous statement holds true for M_N^{Δ} . It is likely that a similar result also holds for the $\nabla \varphi$ -model with strictly convex φ , but this has not been rigorously shown yet (cf. [CCH16a, Remark 3]).

The most interesting and most subtle case is the critical one (d = 2 for the gradient model and the $\nabla \varphi$ -models, d = 4 for the membrane model). For the gradient model, in a series of papers making successive improvements [BDG01, BDZ11, BZ12, BDZ16] it was shown that $M_N^{\nabla} - m_N^{\nabla}$ converges in distribution to a randomly shifted Gumbel variable, where $m_N^{\nabla} = \sqrt{\frac{2}{\pi}} \log N - \frac{3}{\sqrt{32\pi}} \log \log N$. For the $\nabla \varphi$ -model with strictly convex φ convergence in law of the maximum is a challenging open problem, but it is known that $\frac{M_N^{V(\nabla)}}{\log N}$ converges in probability [BW20] and that there is a deterministic subsequence $(N_k)_{k=0}^{\infty}$ along which $M_{N_k}^{V(\nabla)} - \mathbb{E}_{V(\nabla),\Lambda_N} M_{N_k}^{V(\nabla)}$ is tight [WZ19]. For the membrane model previously there were only partial results. The best result [Kur09] is that $\frac{M_N^{\Lambda}}{\log N}$ converges to $\frac{1}{\pi}$ in probability. The question whether a centred version of M_N^{Λ} converges in distribution was posed for example in [Roy16, CDH19]. In Chapter 4 (that is based on the publication [Sch20a]) we prove that this is the case, i.e. that $M_N^{\Lambda} - M_N^{\Lambda}$ converges to a randomly shifted Gumbel variable, where $m_N^{\Lambda} = \frac{1}{\pi} \log N - \frac{3}{16\pi} \log \log N$.

The maximum of the critical membrane model

Let us give a few more details on the result of Chapter 4 on the maximum of the membrane model in dimension d = 4. The precise result will be the following:

Theorem 1.3.1. Let d = 4. The random variable

$$M_N^{\Delta} - m_N^{\Delta} := M_N^{\Delta} - \frac{1}{\pi} \log N + \frac{3}{16\pi} \log \log N$$

converges in distribution. The limit law is a randomly shifted Gumbel distribution μ_{∞} , given by

$$\mu_{\infty}((-\infty,t]) = \mathbb{E}e^{-\gamma^* \mathcal{Z}e^{-8\pi t}} \,\forall t$$

where γ^* is a constant and Z is a positive random variable that is the limit in law of

$$\mathcal{Z}_N = \sqrt{8} \sum_{x \in V_N} (\log N - \pi \psi_{N,x}) e^{-8(\log N - \pi \psi_{N,x})}$$

For this result it is very important that the critical membrane model is a log-correlated Gaussian field (cf. Section 1.2.5). As discussed there, it is conjectured that these form a universality class. One example of a feature that is conjectured to be universal is the behaviour of the maximum of the field, and one expects that convergence in law of the recentred maximum holds true for general log-correlated fields. However, it is a challenging problem to verify this fact for specific examples of log-correlated fields. In recent years convergence in law of the recentred maximum has been proven for the critical gradient model, as already discussed, and also for various other models. Let us mention branching Brownian motion [Bra83], branching random walks [Aïd13], and also problems from random matrix theory (see [CMN18] for partial results).

Furthermore, there have been efforts to give sufficient criteria for convergence in law of the maximum that cover a wide range of log-correlated fields. In [Mad15] this was done for so-called *-scale invariant models. Most importantly for us, in [DRZ17] Ding, Roy and Zeitouni gave some sufficient conditions on the covariances that ensure that the maximum of the field converges in distribution. Their approach is based on a very subtle comparison of the interface with a modified branched random walk. The result from [DRZ17] reduces the proof of Theorem 1.3.1 to the verification of certain estimates on the Green's function of the discrete Bilaplacian. We discuss these in Section 1.4.3.

1.3.3 Entropic repulsion

This section includes parts of the introduction of the paper [BDKS19], written jointly by Simon Buchholz, Jean-Dominique Deuschel, Noemi Kurt and the author.

Existing results

In this and the following sections we will consider the effect of various single-spin potentials to the boundary. We begin with the phenomenon of entropic repulsion. That is, we restrict the interface to be non-negative on some subset of the domain Λ . In physics, this corresponds to the presence of a hard wall that the interface cannot cross. This hard wall leads to a competition between energetic and entropic effects: on the one hand it is energetically favourable for the interface to stay flat and thus close to the hard wall, on the other hand the hard wall severely limits the possible fluctuations, so that it is entropically advantageous for the interface to keep a distance from the wall. Therefore, there will be some repulsive effect of the wall, that is, its local averages will increase. We speak of entropic repulsion if the order by which the field increases is strictly larger than the order of the square root of the variances of the original field, [LM87, Gia01]. See Figure 1.3 for a sample of the membrane model under entropic repulsion, in particular in comparison to Figure 1.1a. This and related problems have also been studied in the physics literature, e.g. in [HL97].

This leads to the question how big this repulsive effect is for our models of interest, and whether there is entropic repulsion in the sense just mentioned. As in the previous



Figure 1.3: A sample of the membrane model in dimension d = 2 on the domain $\{0, \ldots, 20\}^d$ under entropic repulsion on all of the domain. See Section 1.2.7 for a description how the sample was generated.

section, the answer strongly depends on whether the dimension is subcritical, critical or supercritical. In fact, it is known (or conjectured) that entropic repulsion happens if and only if the dimension is critical or supercritical.

We consider the field on the domain $\Lambda_N = [0, N]^d \cap \mathbb{Z}^d$, and for some $D \subset [0, 1]^d$ we let $D_N = ND \cap \mathbb{Z}^d$. We consider the event $\Omega_{D_N,+} = \{\psi: \psi_x \ge 0 \ \forall x \in D_N\}$, and we are interested in the behaviour of the fields when conditioned on being nonnegative on D_N . A first step to understand that behaviour will be to estimate the probability of $\Omega_{D_N,+}$. We focus on the two cases $D \Subset [0,1]^d$ (i.e. D is compactly contained in $[0,1]^d$), or $D = [0,1]^d$.

In the critical and supercritical dimensions there are very precise results known for the Gaussian free field. Namely, if $D \in [0, 1]^d$, then the probability $\mathbb{P}_{\nabla, \Lambda_N}(\Omega_{D_N, +})$ scales like $\exp(-C_{d,D}N^{d-2}\log N)$ if $d \geq 3$, and like $\exp(-C_{d,D}(\log N)^2)$ if d = 2, while the field is repelled to a height $C_{d,D}\sqrt{\log N}$ if $d \geq 3$, and to a height $C_{d,D}\log N$ if d = 2. Here the constants are explicitly known and they depend on d and on the capacity of D with respect to $[0,1]^d$. If $D = [0,1]^d$, then $\mathbb{P}_{\nabla,\Lambda_N}(\Omega_{D_N,+})$ scales like $\exp(-C_{d,D}N^{d-1})$ for any $d \geq 2$ (this is a boundary effect), while the field is repelled to a height of the same order as before. These results are due to [BDZ95, Deu96, BDG01]. Similar, but somewhat weaker results (namely with upper and lower bounds on the rates that differ by a constant factor) are known for the $\nabla \varphi$ -interface model with strictly convex \mathcal{V} [DG00]. For the membrane model only the case that $D \in [0,1]^d$ has been studied. There one finds that $\mathbb{P}_{\Delta,\Lambda_N}(\Omega_{D_N,+})$ scales like $\exp(-C_{d,D}N^{d-4}\log N)$ or $\exp(-C_{d,D}(\log N)^2)$ for $d \geq 5$ and d = 4, respectively, while the field is repelled to heights $C_{d,D}\sqrt{\log N}$ or $C_{d,D}\log N$, respectively [Sak03, Kur07, Kur09]. In all cases, the order of the height to which the field is repelled is larger than the square

root of the variance of the unperturbed field, and so there actually is entropic repulsion in the sense defined above.

The subcritical case is rather different. For the gradient model in dimension d = 1 we actually observe a simple random walk bridge conditioned to be positive on some part of its domain. If $D \in [0,1]^d$ this random walk has as its scaling limit a Brownian bridge conditioned to be positive on D, while if D = [0,1] the scaling limit is the Brownian excursion (cf. e.g. [CC13]). In particular, $\mathbb{P}_{\nabla,\Lambda_N}(\Omega_{D_{N,+}}) \ge c_D$ if $D \subset [0,1]$, while $\mathbb{P}_{\nabla,\Lambda_N}(\Omega_{D_{N,+}})$ scales like $\frac{c_D}{N}$ (as can easily be shown using the reflection principle). In both cases the field is repelled to the height $c_D\sqrt{N}$. These statements are also still valid for the $\nabla \varphi$ -interface model with strictly convex \mathcal{V} , as can be seen using renewal methods, cf. e.g. [FO01, Gia07]. For the membrane model in dimension d = 1 the scaling limit is an integrated Brownian bridge, and we expect that $\mathbb{P}_{\nabla,\Lambda_N}(\Omega_{D_N,+}) \ge c_D$ if $D \subset [0,1]$, while $\mathbb{P}_{\nabla,\Lambda_N}(\Omega_{D_N,+})$ scales like $\frac{c_D}{\sqrt{N}}$, and the field is repelled to the height $c_D N^{\frac{3}{2}}$. This has rigorously been shown only for the one-sided problem [DW15] (see also the earlier results [Sin92, DDG13]), but the method should carry over.

Of course, for the membrane model d = 1 is not the only subcritical dimension, and so we are left to discuss what happens when $d \in \{2,3\}$. It is likely that just as in the other subcritical cases there is no entropic repulsion here, i.e. the field is repelled only to the height $N^{\frac{4-d}{2}}$. Moreover, $\mathbb{P}_{\nabla,\Lambda_N}(\Omega_{D_N,+})$ should be bounded below for any $D \in [0,1]^d$, while it should decay at a surface rate if $D = [0,1]^d$. The only previous rigorous result on this topic, however, is [Sak16], where it is shown that for sufficiently small D we have $\mathbb{P}_{\nabla,\Lambda_N}(\Omega_{D_N,+}) \ge c_D$. In Chapter 3 (that is based on the publication [BDKS19]) we give a significant improvement and prove that $\mathbb{P}_{\nabla,\Lambda_N}(\Omega_{D_N,+})$ behaves as expected. Unfortunately, we only have partial results on the behaviour of the field when conditioned on $\Omega_{D_N,+}$.

Probability to be positive for the subcritical membrane model

We will now give a few more details on the results of Chapter 3. There we prove the following result.

Theorem 1.3.2. Let d = 2 or d = 3. For $\delta \in (0, 1)$ there is a constant $c_{\delta} > 0$ such that

$$c_{\delta} \leq \mathbb{P}_{\Delta,N}(\Omega_{\Lambda_{\delta N},+}) \leq rac{1}{2}.$$

Moreover,

$$\exp(-CN^{\mathsf{d}-1}) \le \mathbb{P}_{\Delta,N}\left(\Omega_{\Lambda_{N,\ell}+}\right) \le \exp(-cN^{\mathsf{d}-1})$$

We can even prove a result interpolating between the two estimates above, i.e. we can take $D_N = \Lambda_{N,L_N}$ for some L_N depending on N (see Theorem 3.1.1). Theorem 1.3.2 easily implies that the field still has a scaling limit in some Hölder space when conditioned on being positive on $\Omega_{\Lambda_{\delta N},+}$ for $\delta < 1$. However, this is a soft argument that relies on the lower bound on $\mathbb{P}_N(\Omega_{\Lambda_{\delta N},+})$ being uniform in N. In the case $D = [0,1]^d$ the probability $\mathbb{P}_N(\Omega_{\Lambda_{\delta N},+})$ is exponentially small in N, and so it is difficult to analyse what happens when one conditions on that event. For this we do not know yet how to proceed.

A crucial ingredient for the proof are estimates for the Green's function of the discrete Bilaplacian and its derivatives that are sharp up to the boundary. These estimates are shown in Chapter 2 (that is based on [MS19]), and we will give an outline in Section 1.4. Here we focus on the probabilistic aspects of the proof of Theorem 1.3.2. The upper bound in Theorem 1.3.2 follows easily from the estimates on the Green's function. Namely, it turns out that the correlations at sites close to the boundary decay rapidly, so we can take a sparse subset of cardinality $\geq c_d N^{d-1}$ where the correlations are very small. In fact, the correlations will be so small that we can use a Gaussian comparison lemma from [LS04] to compare to the situation where the heights at our subsets are independent, and so the upper bound immediately follows.

For the lower bound we use the fact that the field is Hölder continuous up to the boundary, with a random Hölder constant for which we have tail bounds. This means that if the field is positive at a certain site *x*, it is positive in a neighbourhood of *x* with a decent probability as well. Unfortunately, in the absence of the FKG inequality there is no direct way to patch these local results together. To solve this problem, note that we can also conclude from the Hölder continuity that the field is locally small with a decent probability, and these results we can patch together using the Gaussian correlation inequality. This is not yet the result we were looking for, but one can use a change of measure argument to bound the probability that the field is close to any given macroscopic profile from below. For a sufficiently positive profile this then implies the result.

1.3.4 Pinning

This section includes parts of the introduction of the author's preprint [Sch20b].

Existing results

In the previous section we discussed the effect of a hard wall that repels the field from 0. In this section we will do the opposite, namely consider the effect of a small attractive potential. So we add a small attractive potential that rewards the field for being equal to (or close to) 0. It is clear that this should pull the interface closer to 0, and one can explore to what extent this effect happens. The physical motivation for this is mainly that it serves as a stepping stone for understanding the phenomenon of wetting, where one considers the competition between pinning and entropic repulsion. We shall discuss that problem in the next section, and keep our focus on pinning here.

Various pinning potentials have been considered in the literature. We restrict ourselves to the mathematically easiest one, namely $\varepsilon \delta_0$ with δ_0 a point-mass at 0. That is, we consider the probability measures

$$\mathbb{P}^{\varepsilon}_{\Phi,\Lambda}(\mathrm{d}\psi) = \frac{1}{Z^{\varepsilon}_{\Phi,\Lambda}} \exp(-H_{\Lambda}(\psi)) \prod_{x \in \Lambda} (\lambda(\mathrm{d}\psi_x) + \varepsilon \delta_0(\mathrm{d}\psi_x)) \prod_{x \in \mathbb{Z}^d \setminus \Lambda} \delta_0(\mathrm{d}\psi_x)$$
(1.3.1)

for some $\varepsilon > 0$.

If λ is the Lebesgue measure on \mathbb{R} , we cannot understand this as a Gibbs measure with a priori measure λ (as δ_0 is singular with respect to the Lebesgue measure), but the formalism from Section 1.2.3 works well if we directly take $\lambda + \varepsilon \delta_0$ as the a priori measure.

The first question about pinning is whether the additional effect of the pinning measure is strong enough to actually localize the field. In the supercritical dimensions the variances are bounded already without any pinning, so one can expect that the field stays bounded when pinned. In the critical and subcritical dimensions the answer is less easy to guess. See Figure 1.4 for a sample of the pinned membrane model in comparison to Figure 1.1c.

To give a rigorous answer we first have to define what we mean by localization, or being pinned. For each sample of $\mathbb{P}_{\Phi,\Lambda}^{\varepsilon}$ there will be some $x \in \Lambda$ with $\psi_x = 0$ which we call the


Figure 1.4: A sample of the membrane model in dimension d = 4 on the domain $\{0, ..., 20\}^d$, pinned with pinning strength $\varepsilon = 1$. The picture shows the values of the sample on the slice $\{0, ..., 20\}^2 \times \{10\}^2$. See Section 1.2.7 for a description how the sample was generated.

pinned points. We call $\mathbb{P}_{\Phi,\Lambda}^{\varepsilon}$ pinned if the expected fraction of points in Λ that are pinned is bounded below uniformly in Λ . One easily checks that being pinned is a monotonic property in ε , so there will some critical value $\varepsilon_{\text{pin},*} \in [0, \infty]$ such that the field is pinned if $\varepsilon > \varepsilon_{\text{pin},*}$ but not if $\varepsilon < \varepsilon_{\text{pin},*}$. Now for the gradient model and the $\nabla \varphi$ -interface models with strictly convex \mathcal{V} it turns out that the field is always pinned, i.e. $\varepsilon_{\text{pin},*} = 0$ in any dimension. For d = 1 this follows again from renewal theory methods as in [Gia07], for d = 2 this follows from [DMRR92, DV00], and the case d ≥ 3 is almost trivial. The situation is more exciting for the membrane model. We still have $\varepsilon_{\text{pin},*} = 0$ for d ≥ 2 [Sak12, Sak18], but somewhat surprisingly $0 < \varepsilon_{\text{pin},*} < \infty$ if d = 1 [CD08].

Thus, in most cases the pinning effect manages to localize the field in the sense that it touches the 0-plane on a positive fraction of Λ . It is natural to ask whether this localization also manifests itself in some other ways. In particular, it is expected that the variance of the pinned field is bounded, and the covariance decays exponentially in the distance (i.e. a mass is generated). Physically speaking, this corresponds to a finite transverse and longitudinal correlation length, respectively. For the case of the gradient model, this was studied in [BB01, IV00, BDG01] in d \geq 2. There not only finiteness of the variance and existence of a mass is known, but even the ε -asymptotic of these quantities for small ε . The former result is also known for $\nabla \varphi$ -interface models with strictly convex \mathcal{V} [DV00]. For d = 1 and both models there are even better results [Gia07, Gia08].

For the membrane model very refined results are known if d = 1 [CD08, CD09]. These include boundedness of the variance, and should easily imply exponential decay of the

covariances. It is also known that in the supercritical dimensions $d \ge 5$ (where variances are trivially bounded) one has a positive mass [BCK17]. Other than that the only previous result is that in the critical dimension d = 4 the correlations decay stretched-exponentially [BCK16]. In Chapter 6 that is based on the preprint [Sch20b] we improve these results by showing that the variances are bounded and the covariances decay exponentially (i.e. the mass is positive) also if d = 4. We also give asymptotics for these quantities if $d \ge 4$. This leaves open the cases $d \in \{2, 3\}$. Already in [BCK17] the authors wondered whether exponential decay of correlations also holds in that dimensions. This seems likely, but we do not know how to prove this.

Pinning for the critical and supercritical membrane model

There are quite a few new results in Chapter 6. The following theorem summarizes the most important ones.

Theorem 1.3.3. Let $d \ge 4$, and $x \in \Lambda \Subset \mathbb{Z}^d$. If ε is sufficiently small, the variances of the pinned field satisfy

$$c_{\mathsf{d}} \leq \mathbb{E}^{\varepsilon}_{\Delta,\Lambda}(\psi_x^2) \leq C_{\mathsf{d}}$$

if $d \geq 5$ *, and*

$$\frac{|\log \varepsilon|}{32\pi^2} - C_4 \log |\log \varepsilon| \le \mathbb{E}_{\Delta,\Lambda}^{\varepsilon}(\psi_x^2) \le \frac{|\log \varepsilon|}{16\pi^2} + C_4 \log |\log \varepsilon|$$

if $d \ge 4$ *and* x *is sufficiently far from the boundary.*

We also have the lower bounds for the mass m_{ε} (i.e. exponential rate of decay of the covariances)

$$c_{\mathsf{d}}\varepsilon^{1/4} \leq m_{\varepsilon}$$

if $d \ge 5$ *, and*

$$c_4 \frac{\varepsilon^{1/4}}{|\log \varepsilon|^{3/8}} \le m_{\varepsilon}$$

if d = 4.

Moreover for every $\varepsilon \geq 0$ the fields have a unique thermodynamic limit as $\Lambda \nearrow \mathbb{Z}^d$.

The first step to establish the results of Theorem 1.3.3 is to understand the set of pinned points. The first important observation of Chapter 6 is that this set is positively correlated (as follows from the Gaussian correlation inequality). The heuristic is that this set behaves like a Bernoulli point process with a certain density. This is true in a rather strong sense if $d \ge 5$, and still true in a weakened sense if d = 4 (as we show following a two-scale argument from [BDG01]). If $d \le 3$ this breaks down completely, however, and so the arguments have no chance to work in that case.

Having established these estimates on the set of pinned points, the estimates on the variance follow easily. The estimates on the covariance are much more complicated. This problem resembles the classical problem of the homogenization of elliptic PDEs in perforated domains (see [CM97]), and the arguments are inspired from this connection. We proceed by using from [BCK17] the idea to use a Widman hole filler argument [Wid71] on random annuli. The details, however, are rather different. We use a multipolar Hardy-Rellich inequality for second derivatives (inspired by similar inequalities for first derivatives as e.g. in [CZ13]) to estimate the local effect of the pinned points. We also use a rather subtle multiscale construction to construct the required cut-off functions, and to prove that this construction can be done with sufficiently high probability. This is the most technical part of the chapter, and it is novel to the best of our knowledge.

1.3.5 Wetting

As already briefly mentioned in the previous section, one can consider a competition between the effects of a hard wall and an attractive potential. This problem is known as wetting. It should be plausible that, depending on the strength of the attractive potential, one or the other effect might win. That is, the interface could be repelled far away from the boundary or the interface could stay close to the wall and touch it at many points (the two phases are called dry/wet [Gia01] or partially wetted/wetted [Vel06] in the literature; we stick to the former).

The physical interpretation of this problem explains the terminology. Namely, this model arises when analyzing the coexistence of a liquid and a gas in a domain, where the liquid prefers to stick to the boundary of the domain due to certain molecular forces. Our height function then describes the interface between the liquid and the gas. It could happen that there are only a few drops of liquid at the wall with the rest of the wall being dry, or that the whole wall is covered by a liquid film. Which of the two phenomena occurs depends on the amount of liquid present (which itself depends on the attractive forces of the wall). See e.g. [Lip01] and the references therein for biophysical work on this problem.

Another physical interpretation arises when considering a biomembrane contained in a domain that experiences some attractive forces close to the wall of the domain, while entropic effects tend to keep it away from the wall, cf. [Lip95].

Mathematically, wetting consists in the analysis of the measure (1.3.1) conditioned on the event $\Omega_{\Lambda,+} = \{\psi : \psi_x \ge 0 \ \forall x \in \Lambda\}$. We call those sites where $\psi_x = 0$ under the conditioned measure the dry sets, and define the field to be dry when the expected value of the fraction of points in Λ that are dry is positive (and otherwise wet). As for pinning, one can argue that there is a critical value $\varepsilon_{wet,*} \in [0, \infty]$ such that the field is wet if $\varepsilon > \varepsilon_{wet,*}$ but not if $\varepsilon < \varepsilon_{wet,*}$. Now one can investigate whether $\varepsilon_{wet,*}$ is nontrivial and how the field behaves in the dry and wet phases.

There are only a few rigorous results on this problem: For the gradient model it is known that $\varepsilon_{wet,*} = 0$ if $d \ge 3$ [BDZ00], while $\varepsilon_{wet,*} > 0$ if $d \le 2$ [CV00]. It is unknown whether the same holds for the $\nabla \varphi$ -interface model other than in the case d = 1 [HV04]. Similarly for the membrane model it is only known that $\varepsilon_{pin,*} < \varepsilon_{wet,*} < \infty$ in d = 1 [CD08]. The only pathwise results in the literature are in [Vel04] where the main result is an estimate of the typical height of the wet interface.

As mentioned, for the membrane model it is not even known whether $\varepsilon_{wet,*} > 0$ for some $d \ge 2$, although in analogy with the gradient model one can conjecture that $\varepsilon_{wet,*} > 0$ if and only if $d \le 4$. Maybe a combination of the results in [Kur09] on pure entropic repulsion combined with the results of Chapter 6 on pure pinning can help shed light on this question.

1.3.6 Further aspects and open questions

There are many more interesting questions about random interface models that one can study. Let us describe three of them where there is active research right now.

Near-extrema and thick points

In view of the results on the maximum of the fields in Section 1.3.2, it is natural to wonder whether one can say more about the behaviour of the field at or near its maximum. This question is interesting mainly in the critical case. For the case of the gradient model, this has

been thoroughly addressed in [BL16, BL18, BL20]. There, convergence of the full extremal process (encoding height, location and local neighbourhoods of near-maxima) to some limiting process is shown, and the law of the locations of those near-maxima is identified as critical Liouville quantum gravity (cf. [Ber15] for an introduction).

One can also study the so-called λ -thick points of the field, i.e. those points whose height is approximately λ times the height of the maximum. This was done in [BL19], where the authors show that the locations of the λ -thick points converge to subcritical Liouville quantum gravity. This refines an earlier result, [Dav06] finding the Hausdorff dimension of that set.

Analogous results for the $\nabla \varphi$ -model with strictly convex \mathcal{V} seem out of reach, as one does not even know the precise height of the maximum yet. For the membrane model there is more hope in view of the results of Chapter 4. However, some of the results for the gradient model that we have just mentioned rely on its conformal invariance and thus on some special properties of the two-dimensional space, and so these probably have no replacement for the membrane model. Let us mention, though, that the results on the Hausdorff dimension of the set of thick points have already been adapted to the case of the membrane model in [Cip13].

Level surfaces

Another natural question for the random interface models is how their level surfaces look like. We focus on the zero contour, i.e. the set where the (suitably interpolated) interface intersects the 0-plane. This is mainly interesting in the critical case. For the gradient model and the $\nabla \varphi$ -model with strictly convex \mathcal{V} in d = 2 it turns out that the scaling limit of the contour surfaces (which then are contour lines) is the conformal loop ensemble CLE(4), a variant of the Schramm-Loewner evolution. This was shown in [SS09, Mil10].

For the membrane model in d = 4 the analogous question is very interesting, but probably also extremely difficult. In the absence of conformal invariance it is not even heuristically clear what the scaling limit might be.

Level set percolation

A random interface model also gives rises to a strongly correlated percolation model. This is most natural in the supercritical dimensions, as then there is an infinite volume limit of the field without rescaling. One can then consider the interface on all of \mathbb{Z}^d , and for some $t \in \mathbb{R}$ study the set $E_t = \{x : \psi_x \ge t\}$, and in particular its percolative properties. Then one can define various critical values for t. The most natural of them, t_* , is such that E_t contains almost surely an infinite connected component if $t < t_*$, but not if $t > t_*$. For the gradient model it is known that $0 < t_* < \infty$ for all $d \ge 3$ [BLM87, RS13, DPR18]. This means in particular that both the set where the gradient model is positive and its complement contain an infinite connected component. This needs to be contrasted with Bernoulli percolation, where this coexistence is almost surely impossible.

In a recent breakthrough [DCGRS20] it was shown that t_* agrees with a variety of other critical values, and so there are clearly defined subcritical and supercritical phases for percolation.

As a link to Section 1.3.3, one can consider the interface conditioned on the event that for some $D \in [0, 1]^d$ the sets $D_N = ND \cap \mathbb{Z}^d$ and $Z^d \setminus [0, N]^d$ are not connected in E_t . For $t < t_*$ this is an unlikely event, and conditioning on it will lead to a repulsion of the interface. For

the gradient model, this problem was studied in [CN20], using the result from [DCGRS20] to derive sharp estimates for the resulting entropic repulsion.

It would be very interesting to see whether one can transfer these results to the $\nabla \varphi$ -model for strictly convex \mathcal{V} or to the membrane model. For the former, a first result can be found in [Rod16].

1.4 Discrete Green's functions and finite difference schemes

We will now turn to the second important topic of this thesis, namely the discrete Bilaplace equation and its Green's function. We first provide some context and discuss the continuous counterparts of these objects, and then we turn to the new results of this thesis.

1.4.1 Continuous elliptic partial differential equations and Green's functions

Equations in smooth domains

Elliptic partial differential equations are a class of partial differential equations that generalize Poisson's equation. There exists a vast amount of theory on this subject, and we will just briefly mention a few notions important to us. The following results are classical (see e.g. [LM72a, Gia83]). We focus on constant coefficient operators $L = \sum_{|\alpha| \le 2m} a_{\alpha} \partial^{\alpha}$. Such an operator is called elliptic if its principal symbol is invertible, i.e. if $\sum_{|\alpha|=2m} a_{\alpha} \zeta^{\alpha} \neq 0$ for any $\zeta \in \mathbb{R}^d \setminus \{0\}$, where we use the usual multi-index notation. Given a domain $\Omega \subset \mathbb{R}^d$ with C^{m-1} -boundary, one can then consider the boundary value problem

$$Lu = f \quad \text{in } \Omega,$$

$$\partial_{\nu}^{k} u = 0 \quad \text{on } \partial\Omega \quad \forall 0 < k < m - 1$$
(1.4.1)

Elliptic regularity theory implies that, informally speaking, u is better than f by 2m derivatives. That is, there are interior estimates of the form $\|\nabla^{2m}u\|_X \leq C\|f\|_X$ for a variety of function spaces X. Such estimates hold locally in the interior of Ω in any case, and if the domain Ω has a sufficiently smooth boundary, they extend to global estimates. In particular, the equation is uniquely solvable for $f \in X$.

Under the stated assumptions there also exists a Green's function *G* for *L*. Formally $G(\cdot, y)$ is the solution of (1.4.1) with $f = \delta_y$, the δ -distribution at $y \in \Omega$. One can show that this is a well-defined function on $\Omega \times \Omega \setminus \{(x, x) : x \in \Omega\}$, and that for $f \in L^2(\Omega)$ one can represent the solution of (1.4.1) as

$$u = \int_{\Omega} G(\cdot, y) f(y) \, \mathrm{d} y$$

This should make it obvious that the Green's function G is closely linked to the elliptic operator L, and that it is important to understand the behaviour of G to analyse the elliptic boundary value problem associated to L. In particular, understanding the behaviour of G near its singularity at the diagonal can lead via the theory of singular integral operators to regularity estimates for the boundary value problem (1.4.1).

In the full space (i.e. $\Omega = \mathbb{R}^d$), one can analyse the Green's function using Fourier analysis and obtain an explicit expression for it. For bounded Ω one can then use some regularity theory for (1.4.1) to show that the Green's function behaves similarly to the full space Green's function. In fact, using this approach precise estimates on the Green's function and its derivatives are known in smooth domains (see e.g. [Kra67, DS04, GGS10]).

Equations in domains with singularities

If the domain Ω no longer has a smooth boundary, the situation becomes much more difficult. An important class of non-smooth domains are polyhedra (i.e. convex hulls of a finite number of points in \mathbb{R}^d), or, more generally, domains with piecewise smooth boundary. In that case there is still an existence and regularity theory for elliptic equations, although typically one has to introduce weighted function spaces, where the weight measures the distance to the singular part(s) of the boundary (cf. e.g. the series [KMR97, KMR01, MR10]). There are eigenvalue problems associated with the singularities, and these determine for what range of parameters the existence and regularity theory applies, and when one encounters a nontrivial kernel of *L*. One can still define a Green's function of (1.4.1), and derive asymptotics for its behaviour. In the interior of the domain this Green's function behaves again like the full-space one, while near the boundary it might be sensitive to the geometry of the domain.

When the boundary of Ω is no longer piecewise smooth, general results become rare, but they still exist. In particular, for general domains there is an existence and regularity theory in various spaces that are sufficiently close to the energy space $H^m(\Omega)$ (see e.g. [MM13]). There is also still a Green's function, and one can establish various estimates on it that are independent of the geometry of the domain [MM14].

The continuous Bilaplace equation

The differential operator that is most important to this thesis is the Bilaplacian operator, given as Δ^2 (where Δ is the standard Laplacian). This is a fourth-order elliptic operator, and probably the most important such operator. It arises for example in linear elasticity, fluid dynamics and the theory of phase separation.

All the general considerations from the previous paragraphs apply to the Bilaplace operator. While this operator is quite similar to the Laplacian, one important difference is the absence of a maximum principle. One can ask whether the operator is still positivity-preserving in the sense that $f \ge 0$ in (1.4.1) implies that $u \ge 0$ as well. This is equivalent to the Green's function of the Bilaplacian being nonnegative. From a physical point of view this seems quite plausible. In fact, in 1908 Hadamard [Had08] was convinced that this was the case when Ω is convex, even though he had no proof. Later it turned out that the conjecture is false. A first counterexample was found in 1949 by Duffin [Duf49], who showed that for a long thin rectangle the Green's function can become negative. Later on, many other counterexamples (including ones with smooth boundary such as certain ellipses) were found. For more on the history of this conjecture see [GGS10, Section 1.1.2].

Of course, in Section 1.2.6 we had already remarked that the Green's function of the discrete Bilaplacian can be negative, so with this in mind the failure of the conjecture should be not surprising.

1.4.2 Numerical analysis of partial differential equations

Finite differences and finite elements

While partial differential equations are a fairly universal tool to describe physical phenomena, they are ill-suited to computations. In order to actually compute an (approximate) solution, one typically wants to discretize the problem in some way, so that one obtains a finite-dimensional problem which one can then solve exactly or approximately. This is one of the main topics of numerical analysis, see e.g. [Col60, JS14, Hac17].

One way to obtain such a discretization is to use finite differences. This means that one replaces the domain Ω by some lattice, typically $\Omega_h = \Omega \cap (h\mathbb{Z})^d$, and the derivatives in the differential operators by some finite differences at the lattice points. In particular, one can replace the partial derivative $\partial_i u(x)$ by the forward difference $D_i^h u := \frac{1}{h}((u(x + he_i) - u(x)))$, the backward difference $D_{-i}^h u := \frac{1}{h}((u(x) - u(x - he_i)))$ or the central difference $D_{0,i}^h u := \frac{1}{2h}((u(x + he_i) - u(x - he_i)))$. Thus, a simple finite difference scheme for the boundary value problem (1.4.1) would be

$$L_h u_h = f_h \quad \text{in } \Omega_h,$$

$$u_h = 0 \quad \text{on } (h\mathbb{Z})^{\mathsf{d}} \setminus \Omega_h$$
(1.4.2)

where $u_h: (h\mathbb{Z})^d \to \mathbb{R}$, $f_h = f|_{(h\mathbb{Z})^d}$, $L_h = \sum_{|\alpha| \le 2m} a_\alpha D_\alpha^h$, and $D_\alpha^h = (D_1^h)^{\alpha_1} \dots (D_d^h)^{\alpha_d}$. This is a linear system with $|\Omega_h|$ unknowns, and one can solve it by using various efficient methods. One can devise other finite difference schemes using Taylor expansion, and in general one has to weigh simplicity of the scheme against its convergence properties. For our boundary problem (1.4.1) we could easily discretize the boundary condition by requiring u_h to be zero outside of Ω_h , but for other boundary conditions this might be difficult, in particular, when the shape of the boundary is complicated as well. Sometimes it might also be advantageous or necessary to choose f_h in some other way than just the restriction of f to the lattice.

While not important for the present thesis, we should also mention that another major method to discretize PDEs is to use a finite element method. Here one reformulates the PDE as a variational problem in some Hilbert space, and then picks a finite-dimensional subspace of that Hilbert space to solve the variational problem in. Typically one picks that finite-dimensional subspace as the span of a set of basis functions that have particularly nice properties (in particular, that most pairs of basis functions are orthogonal).

Consistency, stability and convergence

Whether by finite differences or finite elements, there are many ways to discretize a PDE. Of course, such schemes are only useful, if one can relate the solution of the discretized problem and the solution of the original PDE, i.e. if one understands the discretization error. This is again a classical topic of numerical analysis, well covered by the above references. Our presentation follows [Arn15].

It is a basic principle in this regard, dating back to [CFL28, Ger30, vNG47] that consistency and stability imply convergence of a scheme. Let us explain what these terms mean. We do so for the finite difference scheme (1.4.2). We define two Banach spaces V_h and W_h with norms $\|\cdot\|_{V_h}$ and $\|\cdot\|_{W_h}$ denoting the spaces in which u_h and w_h live (in our example the underlying vectorspace for both V_h and W_h is \mathbb{R}^{Ω_h}).

First of all, to compare the solutions u and u_h of (1.4.1) and (1.4.2), we need them to live in the same space. This we can achieve by using a suitable map $u \mapsto U_h \in V_h$, e.g $U_h = u|_{\Omega_h}$. Then our goal is to control the error $||U_h - u_h||_{V_h}$. Now the consistency error measures how far U_h is from being a solution of (1.4.2). We define it as $||L_hU_h - f_h||_{W_h}$. We call the system consistent, if the consistency error tends to 0 as $h \to 0$. We also define the stability constant as the operator norm of L_h^{-1} as a map $W_h \to V_h$, and we call the scheme stable if that constant is bounded uniformly in h. Now it is easy to check that consistency and stability imply convergence, and that we actually have a quantitative version of this result. Namely, one calculates

$$\|U_h - u_h\|_{V_h} = \left\|L_h^{-1}(L_h U_h - f_h)\right\|_{V_h} \le \|L_h^{-1}\|_{\mathcal{L}(W_h \to V_h)} \|(L_h U_h - f_h)\|_{W_h}$$

and so the error is bounded by the product of the stability constant and the consistency error.

In practice, it is of course a minimum requirement on a finite difference scheme that it converges. One looks for finite difference schemes whose error converges as fast as possible, while the schemes stays simple enough that the required calculations are feasible. Thus, an important topic in the field is to analyse a given finite difference schemes and to estimate its approximation error.

1.4.3 Estimates for discrete Green's functions

This section includes parts of the introduction of the paper [MS19], written jointly by Stefan Müller and the author, as well as the introduction of the author's paper [Sch20a].

Overview

We consider partial difference equations, that is systems of the form (1.4.2) with L_h some discretization of an elliptic differential operator. If that discretization is chosen in such a way that L_h is elliptic (i.e. positive definite as a linear operator on \mathbb{R}^{Ω_h}), then there is a Green's

function G_h for L_h , i.e. $G_h(\cdot, y)$ is a solution of (1.4.2) with $f_h(x) = \delta_{h,y}(x) = \begin{cases} \frac{1}{h^d} & x = y \\ 0 & \text{else} \end{cases}$. It

is cleary interesting in its own right to study the behaviour of its Green's function.

In addition to that, these Green's functions (with h = 1 and $L_h = -\Delta_h$ or $L_h = \Delta_h^2$) are the same Green' function as the ones discussed in Section 1.2.6, and in that Section we have already discussed the importance of these Green's functions for the study of the gradient and the membrane model.

Note that the Green's functions for different *h* are all related to each other via scaling, and so we can equally well study the Green's functions on $\Lambda_h := [0, 1]^d \cap (h\mathbb{Z})^d$ to conclude results for the Green's functions on $[0, N]^d \cap \mathbb{Z}^d$ that arise in the study of Gaussian interface models. In fact, the former interpretation is oftentimes better suited to the application of tools from PDE theory or numerical analysis. In view of our applications to Gaussian interface models, our main focus here is on the discrete polyharmonic operators $L_h = \Delta_h^k$ for $k \ge 1$.

In order to prove results on the behaviour of the Green's function, there are a variety of approaches one can pursue. Just like in the continuous case, for simple domains such as the full-space $(h\mathbb{Z})^d$ one can apply discrete Fourier analysis to compute somewhat explicit expressions for the Green's function, and these in turn can be used to derive precise asymptotics for the Green's function. This method has led to asymptotics for the Green's function of the discrete Laplacian (e.g. [MW40, Duf53]), and more generally, for the Green's function of discrete polyharmonic operators [DS58, Sim67, Man67]. In particular, in [Man67] Mangad gives an algorithm that allows to compute the asymptotic expansion of the Green's function of Δ_h^k up to arbitrarily high order.

Apart from that, one can try to transfer some techniques available to study continuous PDEs and continuous Green's functions to the discrete setting. For some of them this seems

hopeless, e.g. flattening the boundary, or polar coordinates for corner singularities. Some other techniques carry over well, e.g. approaches based on well-chosen test functions. For example, in the author's MSc thesis [Sch16] Campanato's approach to elliptic regularity [Cam80] was adapted to the discrete setting, leading to a regularity theory for Δ_1^2 in L^p and $L^{p,\infty}$. This was based on earlier work of Dolzmann [Dol93, Dol99], who used the same approach for error estimates for finite element schemes. This approach also works for other discrete polyharmonic operators. Also in [Kur09, Cip13] a similar approach was used to derive estimates for the Green's function of the discrete Bilaplacian in d = 4.

Alternatively, one can also try to transfer existing results for continuous Green's functions to discrete ones. For that purpose one needs a quantitative estimate that those Green's functions (or truncated versions of themselves) are close. One way to derive such estimates are estimates for the appropriate finite difference scheme, as discussed in Section 1.4.2. One can also try to use results for discrete PDEs (as in the previous paragraph) to establish this convergence.

In the following paragraphs we will describe how we have put these methods into practice.

Estimates for the discrete Bilaplacian for $d \in \{2, 3\}$ via continuous elliptic theory

As we have discussed, the subcritical membrane model is quite regular in the bulk. As a consequence, its behaviour at and near boundary is important for its global behaviour. In particular, as described in Section 1.3.3, the question of entropic repulsion is dominated by its boundary behaviour. Thus, one is interested in estimates for the Green's function of that model, i.e. the Green's function of the discrete Bilaplacian, that are sharp up to the boundary. Furthermore, to establish the Hölder continuity of the field via Kolmogorov's continuity criterion, one requires estimates for the mixed second derivatives of *G*, that again are valid up to the boundary.

In Chapter 2, that is based on the publication [MS19], we derive such estimates. In fact, the main result of that chapter is the following.

Theorem 1.4.1. Let d = 2 or d = 3, and let d(z) denote the distance of $z \in \Lambda_h = [0,1] \cap (h\mathbb{Z})^d$ to $(h\mathbb{Z}) \setminus \Lambda_h$. Then there exist c, C > 0 independent of h such that for any $x, y \in (h\mathbb{Z})^d$

$$\begin{split} |G_h(x,y)| &\leq C \min\left(d(x)^{2-\frac{d}{2}}d(y)^{2-\frac{d}{2}}, \frac{d(x)^2 d(y)^2}{(|x-y|+h)^d}\right), \\ |\nabla_{h,x}G_h(x,y)| &\leq C \min\left(d(y)^{3-\mathsf{d}}, \frac{(d(x)+h)d(y)^2}{(|x-y|+h)^d}\right), \\ |\nabla_{h,x}^2G_h(x,y)| &\leq \begin{cases} C \log\left(1 + \frac{d(y)^2}{(|x-y|+h)^2}\right) & \mathsf{d} = 2\\ C \min\left(\frac{1}{|x-y|+h}, \frac{d(y)^2}{(|x-y|+h)^3}\right) & \mathsf{d} = 3\end{cases}, \\ \nabla_{h,x}\nabla_{h,y}G_h(x,y)| &\leq \begin{cases} C \log\left(1 + \frac{(d(x)+h)(d(y)+h)}{(|x-y|+h)^2}\right) & \mathsf{d} = 2\\ C \min\left(\frac{1}{|x-y|+h}, \frac{(d(x)+h)(d(y)+h)}{(|x-y|+h)^3}\right) & \mathsf{d} = 3\end{cases}$$

Besides its application to the problem of entropic repulsion in Chapter 3, this result has also been used in [CDH19] to rigorously establish the scaling limit of the subcritical membrane model.

Let us briefly mention how we prove Theorem 1.4.1. Our main tool are Caccioppoli (or reverse Poincaré) inequalities. That is, for some *u* that is discretely biharmonic on some

large ball, we want to control the L^{∞} -norm of its Hessian on a small ball by the L^2 norm of its Hessian on a larger ball. This is an interior estimate, and combined with the corresponding exterior estimate, and the results on the full-space discrete Green's function from [Man67], one can derive the estimates in 1.4.1 using some careful reasoning.

The main challenge thus is to derive such Caccioppoli inequalities. In the interior or near flat parts of the boundary, we could, in principle, use for that purpose the classical approach based on test functions (as is done in [Dol93, Dol99]). Near the singularities of $[0, 1]^d$, this is no longer possible. There we use that the continuous theory for biharmonic functions in domains with singularities [KMR97, MR10] predicts that these functions decay rather rapidly near the singularities. Using a compactness argument based on a discrete version of the Kolmogorov-Riesz-Fréchet criterion and the Caccioppoli inequality, we can transfer this to a discrete Caccioppolli inequality near the corresponding singularity. Since we anyhow need to introduce this compactness framework, we directly use it for the estimates in the interior or near flat parts of the boundary, as well.

Estimates for the discrete Bilaplacian for d = 4 via finite difference schemes

For the critical membrane model, the boundary behaviour is less important. Instead, most relevant for the analysis of the field is the logarithmic correlation structure in the bulk. In particular, as discussed in Section 1.3.2, one can obtain the convergence of the maximum of the field provided one has very sharp estimates on the Green's function in the bulk. These estimates are derived in Chapter 4 which is based on the publication [Sch20a]. The main result on the Green's function there is too technical to state in full here, but we give some of the estimates.

Theorem 1.4.2. Let d = 4, and $\Lambda_h = [0,1]^4 \cap (h\mathbb{Z})^4$. Also let $d(x) = d(x,\partial[0,1]^d)$. Then for all $x, y \in V_h$ $\left|8\pi^2 G_{\Delta,h}(x,y) - \log\left(2 + \frac{\max(d(x),d(y))}{h+|x-y|}\right)\right| \leq C$.

Furthermore, there are a constant $\theta_0 > 0$, a continuous function $f_1: (0,1)^4 \to \mathbb{R}$ and a function $f_2: \mathbb{Z}^4 \times \mathbb{Z}^4 \to \mathbb{R}$ such that the following holds. For all $L, \varepsilon > 0$, $\theta > \theta_0$ there exists $N'_0 = N'_0(L,\varepsilon,\theta)$ such that for all $h \leq \frac{1}{N'_0}$ with $\frac{1}{h} \in \mathbb{N}$, all $x \in \Lambda_h$ such that $d(x) \geq h |\log h|^{\theta}$ and for all $u, v \in [0, L]^4 \cap \mathbb{Z}^4$ we have

$$|8\pi^2 G_{\Delta,h}(x+hu,x+hv) + \log h - f_1(x) - f_2(u,v)| < \varepsilon.$$

Similarly, there are a constant $\theta_1 > 0$ and a continuous function $f_3: \mathcal{D}^4 \to \mathbb{R}$, where $\mathcal{D}^4 = \{(x,y) : x, y \in (0,1)^4, x \neq y\}$ such that the following holds. For all $L, \varepsilon > 0, \theta > \theta_1$ there exists $N'_1 = N'_1(L,\varepsilon,\theta)$ such that for all $h \leq \frac{1}{N'_1}$ with $\frac{1}{h} \in \mathbb{N}$ and for $x, y \in \Lambda_h$ such that $\min(d(x), d(y)) \geq h |\log h|^{\theta}$ and $|x - y| \geq \frac{1}{L}$ we have

$$\left|8\pi^2 G_{\Delta,h}(x,y)-f_3(x,y)\right|<\varepsilon.$$

The compactness methods from Chapter 2 are not well-suited to be applied here for two reasons. First of all, the relevant continuous estimates have not yet been worked out for d = 4 in the literature. In addition, the estimates obtained using compactness methods and Caccioppoli inequalities are all up to a possibly large constant, while in Theorem 1.4.2 we are interested in estimates with an error that tends to 0 as $h \rightarrow 0$.

Instead we follow another of the strategies mentioned above. Namely, we use an estimate for the approximation quality of finite difference schemes to compare truncated versions of the discrete and continuous Green's function. This estimate is similar to the one in Chapter 5 to be discussed below, although it is easier than this result and can be established using textbook methods as in [JS14]. Besides this error estimates we again use the results from [Man67] on the discrete Green's function in the full-space, as well as results similar to ones in [MM13, MM14] for the continuous Green's function in $[0, 1]^4$.

1.4.4 Estimates for finite difference schemes for the Bilaplacian

This section includes parts of the introduction of the author's paper [MSS20], written jointly by Stefan Müller, Endre Süli and the author.

Overview

We have already described the importance of the Bilaplace equation, as well the relevance of finite difference schemes together with error bounds. The convergence analysis of numerical methods for the approximate solution of the biharmonic equation has therefore been of considerable interest. Some references are the early papers by Tee [Tee64], Bramble [Bra66], Smith [Smi68, Smi70], and Ehrlich [Ehr71]; see also [Col60, Ch. V, §1.5 III and Table VI in the Appendix]. For the numerical analysis of finite difference approximations of the biharmonic equation in rectangles a fast algorithm was given by Bjørstad [Bjø83]. For a modern review in the context of the approximate solution of the Navier–Stokes equations in planar domains, see [BACF13].

In these works the data and the solution to the boundary-value problems under consideration were assumed to be sufficiently smooth. This assumption, however, is quite restrictive, as in practice one often encounters right-hand sides that are rather rough. One of many examples is that of turbulence in a fluid. Another example is given by the analysis of discrete Green's function, as just discussed in Section 1.4.3.

Finite difference schemes for rough right-hand sides were considered by Lazarov [Laz81], Gavrilyuk et al. [GLMP83], and Ivanović et al. [IĬS86], for example. For a detailed survey of the relevant literature see the monograph of Jovanović and Süli [JS14], devoted to the finite difference approximation of linear partial differential equations with generalized solutions.

Consider for instance the boundary value problem

$$\Delta^2 u = f \quad \text{in } \Omega,$$

$$u = 0 \quad \text{on } \Gamma,$$

$$\partial_{\nu} u = 0 \quad \text{on } \partial\Omega,$$

(1.4.3)

where $\Omega = (0, 1)^d$. A finite difference scheme associated with this boundary value problem is given by

$$\Delta_h^2 U = T^{h,2,\dots,2} f \quad \text{in } \Lambda_h ,$$

$$U = 0 \qquad \text{on } \Gamma_h ,$$

$$D_{0,\nu}^h U = 0 \qquad \text{on } \Gamma_h .$$
(1.4.4)

where $T^{h,2,\ldots,2}f$ is a certain smoothing operator (to be defined precisely in Chapter 5), $\Lambda_h = [0,1]^d \cap (h\mathbb{Z})^d$, and $\Gamma_h = (\partial [0,1]^d) \cap (h\mathbb{Z})^d$. The operator $T^{h,2,\ldots,2}f$ regularizes the right-hand

side, and so (1.4.4) makes sense whenever $f \in H^t(\Omega)$ for $t > -\frac{3}{2}$. This scheme has been studied in [GMP83, JS14] for d = 2 and *u* in the fractional Sobolev space $H^s((0,1)^d)$ (for $s = t + 4 > \frac{5}{2}$).

In [JS14, Theorem 2.69] the error bound

$$\|u - U\|_{H^{2}_{\mu}(\Lambda_{h})} \leq Ch^{\min\{s-2,3/2\}} |\log h|^{1-|\operatorname{sgn}(s-7/2)|}) \|u\|_{H^{s}((0,1)^{d})}$$

for $\frac{5}{2} < s \le 4$ is established, and in [GMP83] the error bound

$$\|u - U\|_{H^2_h(\Lambda_h)} \le Ch^{s-2} \|u\|_{H^s((0,1)^d)}$$

is shown for $\frac{5}{2} < s \leq 4$, albeit under the additional assumption that the third normal derivative of *u* vanishes at the boundary.

These results seem suboptimal, because the operator $D_{0,h}$ has truncation error h^2 , and so one can hope that one actually has an error estimate of order h^2 unconditionally. In Chapter 5 we prove that we actually have such an estimate, and this not only if d = 2.

Improved estimates for a finite difference scheme

In fact, in Chapter 5, that is based on the publication [MSS20], we prove the following result.

Theorem 1.4.3. Let $d \ge 2$. Suppose that $\frac{1}{2} \max(5, d) < s \le 4$, and let $u \in H^s(\Omega) \cap H^2_0(\Omega)$; then, there exists a positive constant C = C(d, s), independent of h, such that

$$||u - U||_{H^2_h(\Lambda_h)} \le Ch^{s-2} ||u||_{H^s(\Omega)}.$$

The restrictions on the range of *s* in [JS14, Theorem 2.69] and on the third normal derivative of *u* in [GMP83] arise for the following reason: in order to compare the finite difference approximation with the original problem one needs an extension of the (generalized) solution *u* from Ω to $\mathbb{R}^d \setminus \Omega$ that preserves the Sobolev regularity of *u* and has, ideally, zero discrete boundary values. The assumptions in [JS14, Theorem 2.69] and in [GMP83] permit the use of the symmetric extension of *u* across $\partial \Omega$ for that purpose.

In our setting, with $\frac{1}{2} \max(5, d) < s \leq 4$, this is no longer possible. The main novelty of the proof of Theorem 1.4.3 is to use a different, carefully chosen, extension of u. This extension will no longer have zero boundary values, but we will show that they can be made small (in an appropriate norm, in terms of positive powers of the discretization parameter h), so that we can still close the argument. More precisely, we prove that the boundary values of u are small in a discrete version of the $H^{1/2}$ -norm on the boundary. We also show that this implies that there is an extension of the boundary values back into Λ_h with small L^2 -norm of the Hessian. We can then subtract this extension from u - U and apply classical energy space estimates as in [JS14] to bound the remaining error terms.

1.5 Notation

We have made an attempt to keep the notation consistent throughout the whole thesis. However, this has not always been possible, and we indicate near the beginning of each chapter when the notation there deviates. If some notation is only relevant for a particular chapter, we also only introduce it there.

Let us summarize here the notation that is relevant for all of the thesis.

• We use the convention that *c* and *C* denote generic constants whose precise value can change from occurrence to occurrence. Constants that are denoted by any other Latin or Greek letter have some fixed value and keep it. By adding subscripts to a constant we emphasize that the precise value of that constant may depend on the variables in the subscript (and typically on no others).

In a few places we use the standard Landau notation. That is, we write a = b + O(d) to denote $|a - b| \le Cd$, and a = b + o(d) to denote that $\frac{a-b}{d}$ tends to zero (in a limit that will be clear from the context). Again we add subscripts to emphasize what the implied constant or the implied convergence rate depend on.

- We denote by ℕ = {0,1,2,...}, ℕ⁺ = {1,2,...}, ℚ, ℝ and ℂ the natural numbers, positive integers, rational numbers, real numbers, and complex numbers, respectively.
- We denote the cardinality of a set A by |A|.
- We use the standard notation for multiindexes $\underline{\alpha} \in \mathbb{N}^d$. We will sometimes omit the underline, when there is no risk of confusion.
- We denote by $d \in \mathbb{N}^+$ the dimension of our space. The standard basis of R^d is denoted e_1, \ldots, e_d . We consider the lattice $Z^d \subset \mathbb{R}^d$, and for h > 0 such that $\frac{1}{h} \in \mathbb{N}^+$ also the lattices $(h\mathbb{Z})^d$.
- We write $B_r(x)$ for the open ball of radius r > 0 around $x \in \mathbb{R}^d$, and $Q_r(x) = x + (-r, r)^d$ for the open cube of sidelength 2r. We also use discrete cubes. That is, we define

$$Q_r^h(x) = \{y \in (h\mathbb{Z})^d \colon |y - x|_\infty \le r\} = \overline{Q_r(x)} \cap (h\mathbb{Z})^d$$

When x = 0 we sometimes omit the x.

As an exception to this, in Chapter 6 we will only use discrete cubes, and so we can drop the superscript in $Q_r^h(x)$ there. We will recall this in the introduction of that chapter.

- On \mathbb{R}^d we use the l^p -norms $|\cdot|_p$ for $p \in [1, \infty]$. When p = 2 we often drop the subscript p so that $|\cdot|$ denotes the Euclidean norm. For $x \in \mathbb{R}^d$ and $A, A' \subset \mathbb{R}^d$ we let $d(x, A) = \inf_{y \in A} |x y|$ be the Euclidean distance of x to A, and $d(A, A') = \inf_{y \in A, y' \in A'} |x y|$ be the Euclidean distance of A and A'. By adding a subscript to d we indicate that either we take the distance with respect to some other norm or that A is some fixed set (say $A = \Lambda_N$). This will be defined in detail in the corresponding chapters.
- Given $N \in \mathbb{N}^+$, Λ_N denotes a lattice square of sidelength comparable to N, with lattice width 1. Similarly for h > 0 such that $\frac{1}{h} \in \mathbb{N}^+$ we denote by Λ_h a lattice square of sidelength comparable to 1, with lattice width comparable $\frac{1}{h}$. The reader should think of $\Lambda_N = [0, N]^d \cap \mathbb{Z}^d$ and $\Lambda_h = [0, 1]^d \cap (h\mathbb{Z})^d$, but see the introductions of the individual chapters for the precise definitions.
- We define the forward difference quotient $D_i^h u(x) = u(x + he_i) u(x)$, the backward difference quotient $D_{-i}^h u(x) = u(x) u(x he_i)$ and the centred difference quotient $D_{0,i}^h u(x) := \frac{1}{2h}(u(x + he_i) u(x he_i))$. The discrete gradient is the vector $\nabla_h u(x) := (D_i^h u(x))_{i=1}^d$, the discrete Hessian is the tuple $\nabla_h^2 u(x) := (D_i^h D_{-j}^h u(x))_{i,j=1}^d$, the discrete Hessian is the tuple $\nabla_h^2 u(x) := (D_i^h D_{-j}^h u(x))_{i,j=1}^d$, the discrete Laplacian is $\Delta_h u(x) := \sum_{i=1}^d D_i^h D_{-i}^h u(x)$, and the discrete Bilaplacian is $\Delta_h^2 := \Delta_h \circ \Delta_h$.

For a multi-index $\underline{\alpha} \in \mathbb{N}^d$ we write $D^h_{\underline{\alpha}}u(x) = (D^h_1)^{\alpha_1} \dots (D^h_d)^{\alpha_d}u(x)$, and for $k \in \mathbb{N}$ with $k \ge 3$ we let $\nabla^k u(x)$ be the the collection of all $D^h_{\alpha}u(x)$ with $|\underline{\alpha}| = k$.

More generally, for a vector $a \in \mathbb{Z}^d$ of unit length we define the forward difference quotient $D_a^h u(x) := \frac{1}{h}(u(x+ha) - u(x))$, the backward difference quotient $D_{0,a}^h u(x) := \frac{1}{h}(u(x) - u(x-ha))$ and the centred difference quotient $D_{0,a}^h v(x) := \frac{1}{2h}(v(x+ha) - v(x-ha))$.

- We use the translation operators τ^h_{±i} defined by τ^h_{±i}u(x) = u(x ± he_i). More generally, for a ∈ Z^d we set τ^h_au(x) = u(x + ha).
- For a domain Ω ⊂ ℝ^d and p ∈ [1,∞] we use the standard L^p-norms || · ||_{L^p(Ω)}, and for k ∈ ℕ the Sobolev-norms || · ||_{W^{k,p}(Ω)}. When p = 2, we also write || · ||_{H^k(Ω)} instead of || · ||_{W^{k,2}(Ω)}. Furthermore, for α ∈ [0, 1] we use the Hölder seminorms [·]_{C^{0,α}(Ω)} and the Hölder norms || · ||_{C^{0,α}(Ω)}. We extend these norms to vector-valued functions by taking the Euclidean norm of the norms of the components. Each of these norms comes with an associated function space.
- We also use various discrete function spaces. For A ⊂ (hZ)^d and p < ∞ we define a discrete L^p_h-norm

$$||u||_{L_{h}^{p}(\Omega)}^{p} = \sum_{x \in A} h^{\mathsf{d}} |u(x)|^{p}$$

if $p < \infty$, and

$$\|u\|_{L_h^{\infty}(\Omega)} = \sup_{x \in A} h^{\mathsf{d}} |u(x)|$$

and the associated function spaces. For p = 2 this norm is induced by the scalar product

$$(u,v)_{L^2_h(\Omega)} = \sum_{x \in A} h^{\mathsf{d}} u(x) v(x)$$

When there is no risk of confusion, we drop the subscript *h*. We also use various other discrete function spaces that are defined in the individual chapters.

• For $\Lambda \in (h\mathbb{Z})^d$ we denote the gradient and membrane by $\mathbb{P}_{\nabla,\Lambda}$ and $\mathbb{P}_{\Delta,\Lambda}$ (as introduced in Section 1.2.4). We denote samples from these measures by $\psi_{\nabla,\Lambda}$ and $\psi_{\Delta,\Lambda}$, respectively, and write $G_{\nabla,\Lambda}$ and $G_{\Delta,\Lambda}$ for the associated Green's functions. When $\Lambda = \Lambda_N$ or $\Lambda = \Lambda_h$, we just write *N* or *h* instead of Λ_N or Λ_h . We drop the subscripts ∇ and Δ when there is no risk of confusion.

2 Estimates for the Green's function of the discrete Bilaplacian in dimensions two and three

This chapter is based on the paper [MS19], written jointly by Stefan Müller and the author, with only minor changes. A small part of the content of this chapter has already appeared in the author's M.Sc. thesis [Sch16], where a result similar to Theorem 2.1.1 was shown, but only for d = 2 and using a different and more complicated approach.

2.1 Introduction

In this chapter we will establish estimates for the Green's function of the subcritical discrete Bilaplacian, as described in Section 1.4.3. In particular we prove Theorem 1.4.1. Actually, we prove a slightly different statement: while we still set $\Lambda_h = [0, 1]^d \cap (h\mathbb{Z})^d$, we take the Green's function on int $\Lambda_h = [h, 1 - h]^d \cap (h\mathbb{Z})^d$ instead of Λ_h . The precise statement is Theorem 2.1.1 below. It is easy to see that this Theorem is equivalent to Theorem 1.4.1 as stated in the introduction.

We use the notation from Section 1.5. As this chapter is only concerned with the Bilaplacian, we drop all subscripts Δ right away. We define $\Lambda_N = [-N, N]^d \cap \mathbb{Z}^d$.

We first state an unrescaled version of our main result (i.e. with unit lattice width).

Theorem 2.1.1. Let d = 2 or d = 3, let G_N be the Green's function of the discrete Bilaplacian with zero boundary data outside Λ_N , and let $d(z) = \text{dist}(z, \mathbb{Z}^d \setminus \Lambda_N)$. Then there exist c, C > 0 independent of N such that G_N and its discrete derivatives satisfy the following estimates.

i) For any $x, y \in \mathbb{Z}^d$

$$|G_N(x,y)| \le C \min\left(d(x)^{2-\frac{d}{2}}d(y)^{2-\frac{d}{2}}, \frac{d(x)^2 d(y)^2}{(|x-y|+1)^d}\right),$$
(2.1.1)

$$|\nabla_x G_N(x,y)| \le C \min\left(d(y)^{3-\mathsf{d}}, \frac{(d(x)+1)d(y)^2}{(|x-y|+1)^\mathsf{d}}\right),$$
(2.1.2)

$$|\nabla_{x}^{2}G_{N}(x,y)| \leq \begin{cases} C\log\left(1 + \frac{d(y)^{2}}{(|x-y|+1)^{2}}\right) & d = 2\\ C\min\left(\frac{1}{|x-y|+1}, \frac{d(y)^{2}}{(|x-y|+1)^{3}}\right) & d = 3 \end{cases}$$
 (2.1.3)

$$|\nabla_{x}\nabla_{y}G_{N}(x,y)| \leq \begin{cases} C\log\left(1 + \frac{(d(x)+1)(d(y)+1)}{(|x-y|+1)^{2}}\right) & \mathsf{d} = 2\\ C\min\left(\frac{1}{|x-y|+1}, \frac{(d(x)+1)(d(y)+1)}{(|x-y|+1)^{3}}\right) & \mathsf{d} = 3 \end{cases}.$$
 (2.1.4)

ii) For any $x \in \mathbb{Z}^d$

$$G_N(x,x) \ge cd(x)^{4-d}$$
. (2.1.5)

 G_N is symmetric in x and y, so we also have the analogous estimates for $|\nabla_y G_N(x, y)|$ and $|\nabla_y^2 G_N(x, y)|$. For the optimality of these estimates, see the discussion after Theorem 2.1.3.

The estimates (2.1.1) and (2.1.5) immediately provide estimates for the variance and covariance of ψ under P_N . From the estimates (2.1.1) and (2.1.5) and a thinning procedure one can also deduce estimates on the probability of the membrane model to be positive. We give details on these arguments in Chapter 3.

In addition Theorem 2.1.1 implies the following continuity estimates.

Corollary 2.1.2. Let d = 2 or d = 3. Under P_N , the random field ψ satisfies

$$E_N(|\psi_x - \psi_y|^2) \le \begin{cases} C|x - y|^2 \log\left(2 + \frac{N}{|x - y|}\right) & d = 2, \\ C|x - y| & d = 3 \end{cases}.$$
 (2.1.6)

To show (2.1.6) for d = 2 one uses the identity

$$E_N(|\psi_x - \psi_y|^2) = G_N(x, x) - G_N(x, y) - G_N(y, x) + G_N(y, y), \qquad (2.1.7)$$

as well as a discrete counterpart of the identity

$$H(x,x) - H(x,y) - H(y,x) + H(y,y) = \int_0^1 \int_0^1 \partial_s \partial_t H(x + s(y - x), x + t(y - x)) \, \mathrm{d}s \, \mathrm{d}t \,,$$

valid for every smooth function *H*, and (2.1.4). For d = 3 one uses (2.1.7) and the estimates for G(x, x) - G(x, y) and G(y, y) - G(y, x) provided by (2.1.2) and its analogue for the *y*-derivative. Since ψ is a Gaussian field the estimate (2.1.6) and the Kolmogorov continuity criterion imply that the rescaled fields $\psi'_{x'} = N^{-2+d/2}\psi_{Nx'}$ are uniformly Hölder continuous with exponents $\alpha < \alpha_d$ where $\alpha_2 = 1$ and $\alpha_3 = \frac{1}{2}$. More precisely

$$P\Big(\big\{\psi': \sup_{x'\neq y'}\frac{|\psi'_{x'}-\psi'_{y'}|}{|x'-y'|^{\alpha}} \le K\big\}\Big) \ge 1 - \varepsilon_{\alpha}(K)$$

with $\lim_{K\to\infty} \varepsilon_{\alpha}(K) = 0$. After the publication of these results, Cipriani-Dan-Hazra [CDH19] completed the argument sketched above and proved that the membrane model has a Hölder-continuous scaling limit in dimensions d \leq 3.

In order to prove Theorem 2.1.1, we need regularity improving estimates for discrete biharmonic functions and optimal decay estimates for various norms in annuli around the singularity. The corresponding estimates for continuous biharmonic functions can be proved using well-established techniques. One insight of this chapter is that these estimates can be transferred to the discrete realm using two ingredients: a new compactness argument and the discrete version of the Caccioppoli (or reverse Poincaré) inequality. It should also be possible to transfer continuous estimate to discrete estimates by using error estimates in numerical analysis, see the discussion below Corollary 2.1.4.

In order to derive the estimates in detail and to highlight the similarities between the continuous and discrete setting, it is convenient to change notation. In particular, we rescale our lattice to have width h, while the domain is fixed. We also shift the boundary by h inwards.

Consider the lattice $(h\mathbb{Z})^d$, where we assume $\frac{1}{h} \in \mathbb{N}$. Let $\Lambda_h = [0, 1]^d \cap (h\mathbb{Z})^d$, int $\Lambda_h = [\frac{1}{h}, 1 - \frac{1}{h}]^d \cap (h\mathbb{Z})^d$ and let Δ_h be the discrete Laplacian on $(h\mathbb{Z})^d$. Let $G_h(x, y)$ be the Green's function for $\Delta_h^2 = (\Delta_h)^2$ on int Λ_h with zero boundary values on $(h\mathbb{Z})^d \setminus \operatorname{int} \Lambda_h$. In this setting, Theorem 2.1.1 becomes

Theorem 2.1.3. Let d = 2 or d = 3, and let d(z) denote the distance of $z \in int \Lambda_h$ to $(h\mathbb{Z})^d \setminus int \Lambda_h$. Then there exist c, C > 0 independent of h such that

i) for any $x, y \in (h\mathbb{Z})^d$

$$|G_h(x,y)| \le C \min\left(d(x)^{2-\frac{d}{2}}d(y)^{2-\frac{d}{2}}, \frac{d(x)^2 d(y)^2}{(|x-y|+h)^d}\right),$$
(2.1.8)

$$|\nabla_{h,x}G_h(x,y)| \le C \min\left(d(y)^{3-\mathsf{d}}, \frac{(d(x)+h)d(y)^2}{(|x-y|+h)^\mathsf{d}}\right),$$
(2.1.9)

$$|\nabla_{h,x}^{2}G_{h}(x,y)| \leq \begin{cases} C\log\left(1 + \frac{d(y)^{2}}{(|x-y|+h)^{2}}\right) & d = 2\\ C\min\left(\frac{1}{|x-y|+h}, \frac{d(y)^{2}}{(|x-y|+h)^{3}}\right) & d = 3 \end{cases}$$
(2.1.10)

$$|\nabla_{h,x}\nabla_{h,y}G_{h}(x,y)| \leq \begin{cases} C\log\left(1 + \frac{(d(x)+h)(d(y)+h)}{(|x-y|+h)^{2}}\right) & d = 2\\ C\min\left(\frac{1}{|x-y|+h}, \frac{(d(x)+h)(d(y)+h)}{(|x-y|+h)^{3}}\right) & d = 3 \end{cases}.$$
 (2.1.11)

ii) for any $x \in (h\mathbb{Z})^d$

$$G_h(x,x) \ge cd(x)^{4-d}$$
. (2.1.12)

Theorem 2.1.1 can be easily derived from Theorem 2.1.3 if one chooses $h = \frac{1}{2N+2}$, rescales by a factor of 2N + 2 and observes that the estimates are scale-invariant. One can also obtain estimates for higher discrete derivatives, see Remark 2.8.4 below.

Comparison with the Green's function of the continuous Bilaplacian in the ball (see [Bog05, eqn. (48)] or [GGS10, eqn. (2.65) and Thm. 4.7]), a general bounded smooth set [DS04, Thm. 3 and Thm. 12] or a half-space [GGS10, eqn. (2.66)] shows that the estimates in Theorem 2.1.3 are optimal in the interior and near the regular boundary points (edges for d = 2 and faces for d = 3).

Near the singular boundary points (corners for d = 2 and edges and corners for d = 3) the continuous regularity theory gives a more rapid decay of biharmonic functions (and their derivatives) and hence a more rapid decay for the Green's function with a decay exponent γ . Our compactness argument can be used to establish a similar decay estimate for all exponents $\gamma' < \gamma$. Since the general continuum theory provides an open interval of admissible exponents γ (due to possible logarithmic terms) there is no loss in passing to the discrete estimates.

The general statement is rather tedious, so let us look instead at an illustrative example, the corner point 0 of the square $(0,1)^2$. In this case the distance of a point *x* from the corner point is given by |x|. If $|x| < \frac{1}{4}|y|$ then $|x - y| \ge \frac{1}{2}|y| \ge \frac{1}{2}d(y)$ and the continuous theory implies that

$$|G(x,y)| \le C \left(\frac{|x|}{|y|}\right)^{2+\theta/2} d^2(y).$$
(2.1.13)

where $0 < \theta < \theta_0$, and $\theta_0 \approx 3.47918$. To see this use Lemma 2.5.13 and note that

$$\|\nabla^2 G(\cdot, y)\|_{L^2(Q_{|y|/2} \cap (0,1)^2)} \le C|y|^{-1} d^2(y)$$

(this follows from the continuous counterparts of (2.8.2) and Lemma 2.6.2). Moreover we have

$$\sup_{Q_s \cap (0,1)^2} G(\cdot, y) \le s \|\nabla^2 G(\cdot, y)\|_{L^2(Q_s \cap (0,1)^2)}$$

by the Sobolev-Poincaré inequality and scaling.

The estimate (2.1.13) is better than the estimate

$$G(x,y) \le \frac{d^2(x)d^2(y)}{|x-y|^2} \sim C\frac{d(x)^2}{|y|^2}d^2(y)$$

if

$$\frac{d(x)}{|y|} \gg \left(\frac{|x|}{|y|}\right)^{1+\theta/4}$$

Note that this condition holds in particular if |x| and d(x) are comparable and $|x| \ll |y|$. The compactness argument shows that the discrete Green's function G_h satisfies a counterpart of (2.1.13) if we replace θ by any smaller exponent θ' and C by $C_{\theta'}$.

It is also easy to show that the discrete Green's function converges to the the continuous Green's function.

Corollary 2.1.4. Let d = 2 or d = 3. Let $G(\cdot, y) \in W_0^{2,2}((0,1)^d)$ denote the continuous Green's function, i.e., the unique weak solution of $\Delta^2 G(\cdot, y) = \delta_y$. Extend $G_h(x, y)$ to $y \in (0,1)^d$ by piecewise constant interpolation in the second variable. Then for each $y \in (0,1)^d$ the following assertions hold.

i) We have

$$I_h^{pc}G_h(\cdot, y) \to G(\cdot, y)$$
 uniformly

where I_h^{pc} denotes the piecewise constant interpolation in the first variable.

- *ii)* If d = 2 then $I_h^{pc} \nabla_h G_h(\cdot, y)$ converges uniformly to $\nabla G(\cdot, y)$ and $I_h^{pc} \nabla^2 G_h(\cdot, y)$ converges to $\nabla^2 G(\cdot, y)$ in $L^p((0, 1)^2)$ for all $p < \infty$.
- *iii)* If d = 3 then $I_h^{pc} \nabla_h G_h(\cdot, y)$ is uniformly bounded and converges to $\nabla G(\cdot, y)$ in $L^p((0, 1)^3)$ for all $p < \infty$ and locally uniformly in $[0, 1]^3 \setminus \{y\}$. Moreover $I_h^{pc} \nabla_h^2 G_h(\cdot, y)$ converges to $\nabla^2 G(\cdot, y)$ in L^p for all p < 3.

A slight variant of the argument given below shows that the convergence in i) is also uniform in y, i.e., that we have uniform convergence of the piecewise constant interpolation of G_h to G in $(0,1)^d \times (0,1)^d$. The proof of assertion i) in Corollary 2.1.4 uses essentially only the elementary discrete $W^{2,2}$ estimate in Lemma 2.8.1 and the compact embedding from $W^{2,2}$ to C^0 . The other two assertion follow from Theorem 2.1.3 and the local compactness argument in Section 2.5. See Section 2.8 for the details.

For d = 2 quantitative estimates for the discrete $W^{2,2}$ norm of difference between the solutions of the discretised and the continuous biharmonic equation under weak assumptions on the regularity of the continuous solution have been obtained by Lazarov [Laz81], Gavrilyuk, Makarov and Pirnazarov [GMP83], Gavrilyuk et al. [GLMP83] and Ivanović, Jovanović and Süli [IĬS86], see also Chapter 2.7 in [JS14] which includes estimates for more general fourth order equations in divergence form with variable coefficients. More precisely, let $u \in (W_0^{2,2} \cap W^{s,2})((0,1)^2)$ and let \hat{u}_h be the solution of

$$\Delta_h^2 \hat{u}_h = K_h * \Delta^2 u$$
 in int Λ_h^2

subject to the discrete boundary conditions

$$\hat{u}_h(x) = 0$$
 and $\hat{u}_h(x+he_i) - \hat{u}_h(x-he_i) = 0$ $\forall x \in \Lambda_h^2 \setminus \operatorname{int} \Lambda_h^2$ $\forall i \in \{1, 2\}$. (2.1.14)

Here $K_h(x) = h^{-2}K(\frac{x}{h})$ and $K(z) = (1 - |z_1|)_+(1 - |z_2|)_+$. The boundary condition (2.1.14) has the advantage that it leads to a higher order of consistency compared to our boundary condition $u_h = 0$ on $(h\mathbb{Z})^2 \setminus \text{int } \Lambda_h$ (this latter condition is arguably more natural from the point of view of probability and statistical mechanics). For the discrete $W^{2,2}$ norm the optimal error estimates

$$\|u - \hat{u}_h\|_{W^{2,2}(\Lambda_h)} \le Ch^{s-2} \|u\|_{W^{s,2}((0,1)^2)}$$
(2.1.15)

were established in [GMP83] for s = 3 and in [JS14, Thm. 2.69] for $\frac{5}{2} < s < \frac{7}{2}$. In [GMP83] the estimate (2.1.15) is also proved for s = 4, but under the additional condition that that the symmetric extension \tilde{u} of u outside $(0, 1)^2$ still belongs to $W^{4,2}$. This holds only if the third normal derivatives of u (which exist in the sense of trace) vanish.

Because $K_h * \delta = \delta_h$ these estimates can be used to compare the continuous Green's function $G_y \in W_0^{2,2}$ and the discrete Green's function $\hat{G}_{h,y}$ (defined using the boundary conditions (2.1.14) rather than $G_{h,y} = 0$ on $(h\mathbb{Z})^2 \setminus \operatorname{int} \Lambda_h$) and one obtains $||G_y - \hat{G}_{h,y}||_{W^{2,2}(\Lambda_h)} \leq C_s h^{s-2} d^{3-s}(y)$ for $s \in (\frac{5}{2}, 3)$. More precise estimates can be obtained if one applies the error estimates to $u = G_y - \eta \tilde{G}_y$ where \tilde{G}_y is a suitable Green's function in \mathbb{R}^2 and η is a suitable cut-off function (see below).

One can also use Theorem 2.1.3 to obtain quantitative error estimates for $G_h - G$ and its discrete derivatives.

Let us briefly discuss some other approaches to prove Theorem 2.1.3. For d = 2 the estimates (2.1.8) and (2.1.12) as well as a discrete BMO estimate for the mixed derivative were proved in the author's M.Sc. thesis [Sch16]. There a different approach was used to obtain the estimates near the corners. One starts from a discrete biharmonic function, defines a careful interpolation to get a continuous functions which is biharmonic up to a small error and uses the continuous theory to get good estimates for that interpolation which can then be transferred back to the original discrete function. This approach can in principle be extended to d = 3, but we found the compactness argument more flexible and more convenient to use.

Hackbusch [Hac83, Thm. 2.1] has developed a very general approach to derive discrete stability estimates on a scale of Banach spaces from the corresponding continuous estimates. One advantage of the compactness method is that it avoids the construction of suitable discrete norms and restriction and prolongation operators which is a bit delicate near the singular boundary points.

Alternatively, for d = 2 and the symmetric boundary condition (2.1.14) one can use the optimal error estimates (2.1.15) in connection with the asymptotic expansion of the discrete Green's function $\tilde{G}_{h,y}$ on $(h\mathbb{Z})^2$ in [Man67] (see also Section 2.7). One applies the estimate (2.1.15) with s = 3 to $u = G_y - \eta \tilde{G}_y$ where \tilde{G}_y is a suitable Green's function in \mathbb{R}^2 . It is not difficult to estimate the additional error term $w_h = G_h - \eta \tilde{G}_h - \hat{u}_h$ in the discrete $W^{2,2}$ norm by computing $\Delta_h^2 w_h$ and testing with w_h . This yields the estimate $\|\hat{G}_{h,y} - G_y\|_{W^{2,2}(\Lambda_h^2)} \leq Ch$ and the discrete inverse estimate implies that $\|\hat{G}_{h,y} - G_y\|_{W^{2,\infty}(\Lambda_h^2)} \leq C$. Together with the known estimates for $\nabla^2 G_y$ one concludes in particular that

$$|\nabla_h^2 \hat{G}_{h,y}| \le C d^2(y) / (|x-y|+h)^2 \quad \text{for } |x-y| \le C d(y) \,. \tag{2.1.16}$$

To get the optimal estimate for $|x - y| \gg d(y)$ one may proceed as follows. From the estimate for $|x - y| \le Cd(y)$ one can obtain the crucial discrete $L^{\infty} - L^2$ estimate (2.6.1) for

the second discrete derivatives for cubes of length 2r that touch the boundary by using the identity $u(x) = \sum_{y \in \text{int } \Lambda_h} \hat{G}_h(x, y) \Delta_h^2(\eta u)(y) h^2$ for an arbitrary lattice function u and a suitable cut-off function η with $|\nabla_h^k \eta| \le C_k r^{-k}$. For cubes which do not touch the boundary one can apply the identity $v(x) = \sum_{y \in \text{int } \Lambda_h} \hat{G}_h(x, y) \Delta_h^2(\eta v)(y) h^2$ to $v(x) = u(x) - a - b \cdot x$ where a is the average of u over the cube and b is the average of $\nabla_h u$. Together with the duality argument in Lemma 2.6.2 and Theorem 2.6.3 and similar estimates for the discrete y-derivatives of $G_y - \hat{G}_{h,y}$ this yields the estimates in Theorem 2.1.3 for d = 2 for the Green's function $\hat{G}_{h,y}$ which satisfies the modified boundary conditions (2.1.14). The same argument applies to G_h .

These estimates initially hold for $\hat{G}_{h,y}$ and not for the function $G_{h,y}$ in Theorem 2.1.3. Note, however, that $\Delta_h^2(G_{h,y} - \hat{G}_{h,y}) = 0$ in int Λ_h . Using this fact as well as careful comparison of the different boundary conditions for \hat{G}_h and G_h one can show that $\|\hat{G}_{h,y} - G_{h,y}\|_{W^{2,2}(\Lambda_h)} \leq Ch$. This shows that the estimate (2.1.16) also holds for G_h . For the estimates for $|x - y| \gg d(y)$ one can then argue as for \hat{G}_h .

The remainder of this chapter is organised as follows. In Section 2.2 we introduce some notation in the discrete setting and recall discrete counterparts of the product rule as well as Sobolev and Poincaré estimates. In Section 2.3 we give the weak and strong formulation of the discrete Bilaplace equation and prove the Caccioppoli inequality (or reverse Poincaré inequality). The proof is very similar to the argument in the continuous case based on testing the equation with a cut-off function times the solution, but due to the discrete product rule some additional terms appear. In Section 2.4 we associate to each discrete function a continuous function by discrete convolution with a B-spline and prove basic estimates of the interpolation.

Sections 2.5 and 2.6 contain the key estimates. The first key ingredient is an $L^{\infty} - L^2$ estimate for the discrete second derivative of discrete biharmonic functions in cubes which may intersect the boundary (see Theorem 2.6.1). This estimate is deduced from decay estimates for the second derivative of continuous biharmonic functions using a discrete version of the Kolmogorov-Riesz-Fréchet compactness criterion and the Caccioppoli inequality. The transition from continuous to discrete decay estimates is carried out in Section 2.5 separately for interior cubes, cubes near regular boundary points and cubes near singular boundary points.

The second key estimate is an L^{∞} decay estimate for discretely biharmonic functions in the complement of a cube (see Lemma 2.6.2 and Theorem 2.6.3). This follows by duality from the $L^{\infty} - L^2$ estimate in Theorem 2.6.1. The estimates in the interior and near regular boundary points can alternatively be derived by using discrete scaled L^2 estimates, i.e., by translating the continuous Campanato regularity theory to the discrete setting (see Dolzmann [Dol93, Dol99]). For the behaviour near the singular boundary points there seems to be no argument, however, which is only based on scaled L^2 -norms and testing. For ease of exposition we use the compactness approach in all three regimes: interior points, regular boundary points and singular boundary points.

In Section 2.7 we recall Mangad's [Man67] asymptotic expansion of a Green's function \tilde{G}_h of the discrete biharmonic operator in $(h\mathbb{Z})^d$. Finally in Section 2.8 we prove Theorem 2.1.3 and Corollary 2.1.4. An L^2 estimate for the second discrete derivatives of G_h is easily obtained by testing with G_h and Poincaré's inequality. We then choose a suitable cut-off function η_h and use the fact that $G_h(\cdot, y) - \eta_h(x)\tilde{G}_h(x-y)$ is biharmonic near x = y to prove estimates for the mixed third discrete derivative $\nabla^2_{h,x}\nabla_{h,y}G_h$. The estimates for the lower derivatives now follow essentially by discrete integration over suitable paths (the relevant path are the discrete counterparts of the paths used in [DS04]). For the estimate for the first discrete derivatives for d = 3 we directly use the discrete Sobolev embedding since integration of the second derivative would generate an unnecessary additional logarithmic term.

2.2 Preliminaries

2.2.1 Notation

In the following *C* denotes a constant that may change from line to line but is independent of *h*, unless stated otherwise.

Given $a \in \mathbb{R}^d$, we define $\tau_a f = f(\cdot + a)$ for any f. This corresponds to shifting f by -a.

For a function f we denote by $[f]_{\Omega} = \frac{1}{|\Omega|} \int_{\Omega} f \, dx$ its average over the bounded open set

Ω.

For discrete quantities we choose notation in such a way that it resembles the continuous notation. Let h > 0 be the (typically small) lattice width. We consider the lattice $(h\mathbb{Z})^d \subset \mathbb{R}^d$.

For $r \in \mathbb{R}$ we define $\lfloor r \rfloor_h := h \lfloor \frac{r}{h} \rfloor$, the largest element of $h\mathbb{Z}$ less than or equal to r.

Given $A_h \subset (h\mathbb{Z})^d$, we define a corresponding subset $(A_h)_{pc} \subset \mathbb{R}^d$ as

$$(A_h)_{pc} = \operatorname{int}\left(A + \left[-\frac{h}{2}, \frac{h}{2}\right]^{\mathsf{d}}\right).$$

For example, for $x \in (h\mathbb{Z})^d$, $r \in h\mathbb{N}$, $(Q_r^h(x))_{pc} = Q_{r+\frac{h}{2}}(x)$. For a function $u_h: A_h \to \mathbb{R}$, we define its piecewise constant interpolation $I_h^{pc}u_h: A_{pc} \to \mathbb{R}$ by $I_h^{pc}u_h(y) = u_h(x)$ on each square $x + \left[-\frac{h}{2}, \frac{h}{2}\right]^d$, where $x \in A$.

For a multi-index $\alpha \in \mathbb{N}^d$ we define $D_{\pm h}^{\alpha} u_h(x) = (D_{\pm 1}^h)^{\alpha_1} \dots (D_{\pm d}^h)^{\alpha_d} u_h(x)$, and for $a \in \mathbb{N}$, a > 2 we set $\nabla_h^a u_h(x) = (D_{-i_1}^h D_{i_2}^h \dots D_{i_d}^h u_h(x))_{i_1, i_2, \dots, i_d}$.

The discrete product rule then takes the form

$$D_i^h(f_hg_h) = (D_i^hf_h)g_h + \tau_i^hf_hD_i^hg_h.$$

When dealing with functions of several variables we use a sub- or superscript to indicate the variable with respect to which a derivative is taken. So for example in $\nabla_{h,x} \nabla_{h,y} G_h(x,y)$ we take one gradient in each variable.

As mentioned in the introduction, we set $\Lambda_h = [0, 1]^{\mathsf{d}} \cap (h\mathbb{Z})^{\mathsf{d}}$ and $\operatorname{int} \Lambda_h = \left[\frac{1}{h}, 1 - \frac{1}{h}\right]^{\mathsf{d}} \cap$ $(h\mathbb{Z})^{\mathsf{d}}$. We also set $\partial \Lambda_h = \Lambda_h \setminus \operatorname{int} \Lambda_h$.

2.2.2 Function spaces and inequalities

Let $u_h, v_h: (h\mathbb{Z})^d \to \mathbb{R}$. For $\Omega \subset \mathbb{R}^d$ measurable, $p \in [1, \infty], k \in \mathbb{N}, \alpha \in [0, 1]$ we define (slightly abusing notation)

$$\|u_{h}\|_{L^{p}(\Omega)} := \|I_{h}^{pc}u_{h}\|_{L^{p}(\Omega)},$$

$$(u_{h}, v_{h})_{L^{2}(\Omega)} := (I_{h}^{pc}u_{h}, I_{h}^{pc}v_{h})_{L^{2}(\Omega)},$$

$$\|u_{h}\|_{W^{k,p}(\Omega)} := \left(\sum_{|\alpha| \le k} \|I_{h}^{pc}D_{h}^{\alpha}u_{h}\|_{L^{p}(\Omega)}^{p}\right)^{\frac{1}{p}},$$

$$[u_h]_{C_h^{0,\alpha}(\Omega)} = \sup_{\substack{x,y\in\Omega\\|x-y|\geq h}} \frac{|I_h^{pc}u_h(x) - I_h^{pc}u_h(y)|}{|x-y|^{\alpha}}.$$

For $[\cdot]_{C_h^{0,\alpha}}$ we add the index *h* to emphasize the fact that we only take the supremum over x, y with $|x - y| \ge h$.

For $A_h \subset (h\mathbb{Z})^d$ these definitions take a familiar form. For example, if $p < \infty$

$$\|u_h\|_{L^p((A_h)_{pc})} = \left(\sum_{\substack{x \in A_h}} h^{\mathsf{d}} |u_h(x)|^p\right)^{\frac{1}{p}},$$

$$[u_h]_{C_h^{0,\alpha}((A_h)_{pc})} = \sup_{\substack{x,y \in A_h \\ x \neq y}} \frac{|u_h(x) - u_h(y)|}{|x - y|^{\alpha}}.$$

We extend these definitions to vector-valued functions by taking the Euclidean norm of the norms of the components.

We also set $[u_h]_{\Omega} = [I_h^{pc} u_h]_{\Omega} = \frac{1}{|\Omega|} \int_{\Omega} I_h^{pc} u_h$. We then have the discrete analogues of Poincaré and Sobolev inequalities. All of them can be proved easily by applying their continuous counterpart to the piecewise multilinear interpolation of the function. We state the results that we will need.

Lemma 2.2.1 (Poincaré inequality on cubes with 0 boundary values). Let $p \in [1, \infty]$, let $u_h: (h\mathbb{Z})^d \to \mathbb{R}, x \in (h\mathbb{Z})^d, r \in h\mathbb{N} + \frac{h}{2}$, and suppose that $u_h = 0$ on at least one of the faces of $Q_r^h(x)$. Then

$$||u_h||_{L^p(Q_r(x))} \le Cr ||\nabla_h u_h||_{L^p(Q_r(x))}$$

where C is independent of h and r.

Lemma 2.2.2 (Poincaré inequality on annuli with 0 boundary values). Let $p \in [1, \infty]$, $u_h: (h\mathbb{Z})^d \to \mathbb{R}$, let $x \in (h\mathbb{Z})^d$, $r, s \in h\mathbb{N} + \frac{h}{2}$, s < r and suppose that $u_h = 0$ on at least one of the faces of $Q_r^h(x)$. Then

$$\|u_h\|_{L^p(Q_r(x)\setminus Q_s(x))} \leq Cr \|\nabla_h u_h\|_{L^p(Q_r(x)\setminus Q_s(x))}$$

where C only depends on $\frac{s}{r}$, p and d.

Lemma 2.2.3 (Sobolev-Poincaré inequality on cubes with 0 boundary values). Let $p \in [1, \infty]$, $u_h: (h\mathbb{Z})^d \to \mathbb{R}$, let $x \in (h\mathbb{Z})^d$, $r \in h\mathbb{N} + \frac{h}{2}$, and suppose that $u_h = 0$ on at least one of the faces of $Q_r^h(x)$.

If $q \in [1,\infty]$ is such that $\frac{d}{q} + 1 \ge \frac{d}{p}$ and $(p,q) \ne (d,\infty)$, then

$$\|u_h\|_{L^q(Q_r(x))} \le Cr^{1+\frac{d}{q}-\frac{d}{p}}\|\nabla_h u_h\|_{L^p(Q_r(x))}$$

and if $\alpha \in (0,1]$ is such that $\alpha + \frac{d}{p} \leq 1,$ then

$$[u_h]_{C_h^{0,\alpha}(Q_r(x))} \leq Cr^{1-\frac{d}{p}-\alpha} \|\nabla_h u_h\|_{L^p(Q_r(x))}.$$

2.3 The discrete Bilaplacian equation

2.3.1 Definitions and basic properties

We consider the space of functions

$$\Phi_h = \{ u_h \colon (h\mathbb{Z})^{\mathsf{d}} \to \mathbb{R} \colon u_h(x) = 0 \; \forall x \in (h\mathbb{Z})^{\mathsf{d}} \setminus \operatorname{int} \Lambda_h \}$$

The discrete Bilaplacian equation on Λ_h with 0 boundary data is the equation

$$\Delta_h^2 u_h = f_h \text{ in int } \Lambda_h \tag{2.3.1}$$

where $f_h: (h\mathbb{Z})^d \to \mathbb{R}$ is given and we are looking for a solution $u_h \in \Phi_h$.

This equation is the discrete analogue of the Bilaplace equation with clamped boundary conditions,

$$\Delta^2 u = f \quad \text{in } [0,1]^{\mathsf{d}},$$

$$u = 0 \quad \text{on } \partial [0,1]^{\mathsf{d}},$$

$$D_{\nu} u = 0 \quad \text{on } \partial [0,1]^{\mathsf{d}}.$$

If we multiply (2.3.1) with a test function $\varphi_h \in \Phi_h$ and use summation by parts, we obtain the weak form of the Bilaplace equation

$$(\nabla_h^2 u_h, \nabla_h^2 \varphi_h)_{L^2(\mathbb{R}^d)} = (f_h, \varphi_h)_{L^2(\mathbb{R}^d)} \quad \forall \varphi_h \in \Phi_h.$$
(2.3.2)

It is easy to check that (2.3.1) and (2.3.2) are equivalent.

Written as a sum over lattice points, (2.3.2) becomes

$$h^{\mathsf{d}} \sum_{x \in \Lambda_h} \nabla_h^2 u_h(x) : \nabla_h^2 \varphi_h(x) = h^{\mathsf{d}} \sum_{x \in \operatorname{int} \Lambda_h} f_h(x) \varphi_h(x) \,.$$

Observe that the sum on the left-hand side has nonzero terms for $x \in \Lambda_h$, whereas the right-hand side has nonzero terms only for $x \in \text{int } \Lambda_h$.

If we choose $\varphi_h = u_h$ in (2.3.2), we obtain

$$(\Delta_{h}^{2}u_{h}, u_{h})_{L^{2}(\mathbb{R}^{d})} = (\nabla_{h}^{2}u_{h}, \nabla_{h}^{2}u_{h})_{L^{2}(\mathbb{R}^{d})} = \|\nabla_{h}^{2}u_{h}\|_{L^{2}(\mathbb{R}^{d})}^{2}$$

Hence Δ_h^2 , seen as a linear operator on Φ_h , is positive definite and hence invertible, and so (2.3.1) has a unique solution for any right-hand side f_h .

The discrete Green's function G_h is now defined as the inverse of Δ_h^2 (considered as a matrix operating on $\mathbb{R}^{\text{int }\Lambda_h}$ with the scalar product $\langle u_h, v_h \rangle = (u_h, v_h)_{L^2(\mathbb{R}^d)}$).

Let us also give an alternative description of G_h : The discrete delta function is given as

$$\delta_{h,x}(y) = \begin{cases} \frac{1}{h^{\mathsf{d}}} & \text{if } x = y\\ 0 & \text{otherwise} \end{cases}$$

The discrete Green's function G_h of Λ_h^d is then the function $(h\mathbb{Z})^d \times (h\mathbb{Z})^d \to \mathbb{R}$ such that $G_h(x, y) = 0$ when $y \notin \text{int } \Lambda_h$ and such that $G_h(\cdot, y)$ is the unique solution in Φ_h of

$$\Delta_h^2 u_h = \delta_{h,y}$$
 in int Λ_h

when $y \in \operatorname{int} \Lambda_h$.

As in the continuous case one can easily show that G_h is symmetric in x and y. We will frequently denote $G_h(x, \cdot)$ and $G_h(\cdot, y)$ by $G_{h,x}$ and $G_{h,y}$ respectively.

Let us return our attention to (2.3.2) for a moment. If f_h is given in divergence form as $\operatorname{div}_h \operatorname{div}_{-h} g_h$, this equation takes the form

$$(
abla_h^2 u_h,
abla_h^2 \varphi_h)_{L^2(\mathbb{R}^d)} = (g_h,
abla_h^2 \varphi_h)_{L^2(\mathbb{R}^d)}$$

and if we choose $\varphi_h = u_h$, we obtain the energy estimate

$$\|\nabla_h^2 u_h\|_{L^2(\mathbb{R}^d)} \le \|g_h\|_{L^2(\mathbb{R}^d)}$$

2.3.2 Caccioppoli inequalities

We will need a discrete counterpart of the Caccioppoli (or reverse Poincaré) estimate for biharmonic functions (see e.g. [Cam80, Cap. II, Lemma 1.II]). It can be derived by testing $\Delta_h^2 u_h = 0$ with $\eta_h u_h$ for a suitable cut-off function η_h and some manipulations of the error terms.

Lemma 2.3.1. Let $d \in \mathbb{N}$, $u_h \in \Phi_h$, $x \in (h\mathbb{Z})^d$, r > 0 and assume that $\Delta_h^2 u_h(y) = 0$ for all $y \in Q_{r-h}^h(x) \cap \operatorname{int} \Lambda_h$. Then for any $0 < s \le r - 4h$ we have

$$\|\nabla_h^2 u_h\|_{L^2(Q_s(x))}^2 \le \frac{C}{(r-s)^4} \|u_h\|_{L^2(Q_r(x))}^2 + \frac{C}{(r-s)^2} \|\nabla_h u_h\|_{L^2(Q_r(x))}^2$$

The proof is similar to the continuous case. However, the fact that the discrete chain rule only holds up to translations generates additional error terms. Therefore we will give the somewhat lenghty proof in full detail. The proof is adapted from that of Lemma 2.9 in [Dol93].

Proof. By replacing r by $\lfloor r - \frac{h}{2} \rfloor_h + \frac{h}{2}$ and s by $\lfloor s - \frac{h}{2} \rfloor_h + \frac{3h}{2}$, we can assume that $r, s \in h\mathbb{Z} + \frac{h}{2}$ and $s \leq r - 3h$.

Choose a discrete cut-off function η_h with support in $Q_{r-2h}(x)$ that is 1 on $Q_{s+h}(x)$ und such that $|\nabla_h^{\kappa} \eta| \leq \frac{C}{(r-s)^{\kappa}}$ for $\kappa \leq 2$. Note that $\eta_h^4 u_h \in \Phi_h$, and $\eta_h^4 u_h = 0$ whenever $\Delta_h^2 u_h \neq 0$. Thus the weak form of (2.3.2) with $\varphi_h = \eta_h^4 u_h$ is

$$0 = \left(\Delta_h^2 u_h, \eta_h^4 u_h\right)_{L^2(\mathbb{R}^d)} = \left(\nabla_h^2 u_h, \nabla_h^2(\eta_h^4 u_h)\right)_{L^2(\mathbb{R}^d)}$$

We can expand the right-hand side and obtain

$$\begin{split} 0 &= \left(\nabla_{h}^{2} u_{h}, \nabla_{h}^{2}(\eta_{h}^{4} u_{h}) \right)_{L^{2}(\mathbb{R}^{d})} \\ &= \sum_{i,j}^{d} \left(D_{-i}^{h} D_{j}^{h} u_{h}, \eta_{h}^{4} D_{-i}^{h} D_{j}^{h} u_{h} \right)_{L^{2}(\mathbb{R}^{d})} \\ &+ \sum_{i,j}^{d} \left(D_{-i}^{h} D_{j}^{h} u_{h}, D_{j}^{h}(\eta_{h}^{4}) \tau_{j}^{h} D_{-i}^{h} u_{h} + D_{-i}^{h}(\eta_{h}^{4}) \tau_{-i}^{h} D_{j}^{h} u_{h} \right)_{L^{2}(\mathbb{R}^{d})} \\ &+ \sum_{i,j}^{d} \left(D_{-i}^{h} D_{j}^{h} u_{h}, D_{-i}^{h} D_{j}^{h}(\eta_{h}^{4}) \tau_{-i}^{h} \tau_{j}^{h} u_{h} \right)_{L^{2}(\mathbb{R}^{d})}. \end{split}$$

We can rewrite this as

$$\begin{split} \|\eta_{h}^{2}\nabla_{h}^{2}u_{h}\|_{L^{2}(\mathbb{R}^{d})}^{2} &= \sum_{i,j}^{d} \left\|\eta_{h}^{2}D_{-i}^{h}D_{j}^{h}u_{h}\right\|_{L^{2}(\mathbb{R}^{d})}^{2} \\ &\leq \left|\sum_{i,j}^{d} \left(D_{-i}^{h}D_{j}^{h}u_{h}, D_{j}^{h}(\eta_{h}^{4})\tau_{j}^{h}D_{-i}^{h}u_{h}\right)_{L^{2}(\mathbb{R}^{d})}\right| + \left|\sum_{i,j}^{d} \left(D_{-i}^{h}D_{j}^{h}u_{h}, D_{-i}^{h}(\eta_{h}^{4})\tau_{-i}^{h}D_{j}^{h}u_{h}\right)_{L^{2}(\mathbb{R}^{d})}\right| \\ &+ \left|\sum_{i,j}^{d} \left(D_{-i}^{h}D_{j}^{h}u_{h}, D_{-i}^{h}D_{j}^{h}(\eta_{h}^{4})\tau_{-i}^{h}\tau_{j}^{h}u_{h}\right)_{L^{2}(\mathbb{R}^{d})}\right|. \end{split}$$

$$(2.3.3)$$

We will estimate the terms on the right-hand side separately.

Using $\frac{a^4-b^4}{a-b} = a^3 + a^2b + ab^2 + b^3$ for $a = \eta_h^4 \circ \tau_j^h$ and $b = \eta_h^4$ we can rewrite the summands of the first term as

$$\begin{split} & \left(D_{-i}^{h} D_{j}^{h} u_{h}, D_{j}^{h} (\eta_{h}^{4}) \tau_{j}^{h} D_{-i}^{h} u_{h} \right)_{L^{2}(\mathbb{R}^{d})} \\ &= \left(D_{-i}^{h} D_{j}^{h} u_{h}, \left(\eta_{h}^{3} + \eta_{h}^{2} \tau_{j}^{h} \eta_{h} + \eta_{h} \tau_{j}^{h} \eta_{h}^{2} + \tau_{j}^{h} \eta_{h}^{3} \right) D_{j}^{h} \eta_{h} \tau_{j}^{h} D_{-i}^{h} u_{h} \right)_{L^{2}(\mathbb{R}^{d})} \\ &= \left(D_{-i}^{h} D_{j}^{h} u_{h}, 4 \eta_{h}^{3} D_{j}^{h} \eta_{h} \tau_{j}^{h} D_{-i}^{h} u_{h} \right)_{L^{2}(\mathbb{R}^{d})} \\ &+ \left(D_{-i}^{h} D_{j}^{h} u_{h}, \left(\eta_{h}^{2} (\tau_{j}^{h} \eta_{h} - \eta_{h}) + \eta_{h} (\tau_{j}^{h} \eta_{h}^{2} - \eta_{h}^{2}) + (\tau_{j}^{h} \eta_{h}^{3} - \eta_{h}^{3}) \right) D_{j}^{h} \eta_{h} \tau_{j}^{h} D_{-i}^{h} u_{h} \right)_{L^{2}(\mathbb{R}^{d})}. \end{split}$$

The second term here is problematic¹, because it does not contain a factor $\eta_h^2 D_{-i}^h D_j^h u_h$. We will control it by moving a factor $\frac{1}{h}$ from the left-hand side to the right-hand side, so that we are no longer taking second derivatives of u_h . We obtain

$$\begin{split} & \left(D_{-i}^{h} D_{j}^{h} u_{h}, D_{j}^{h} (\eta_{h}^{4}) \tau_{j}^{h} D_{-i}^{h} u_{h} \right)_{L^{2}(\mathbb{R}^{d})} \\ &= \left(\eta_{h}^{2} D_{-i}^{h} D_{j}^{h} u_{h}, 4\eta_{h} D_{j}^{h} \eta_{h} \tau_{j}^{h} D_{-i}^{h} u_{h} \right)_{L^{2}(\mathbb{R}^{d})} \\ &+ \left(\tau_{j}^{h} D_{-i}^{h} u_{h} - D_{-i}^{h} u_{h}, \left(\eta_{h}^{2} D_{j}^{h} \eta_{h} + \eta_{h} D_{j}^{h} (\eta_{h}^{2}) + D_{j}^{h} (\eta_{h}^{3}) \right) D_{j}^{h} \eta_{h} \tau_{j}^{h} D_{-i}^{h} u_{h} \right)_{L^{2}(\mathbb{R}^{d})}. \end{split}$$

Therefore, using the Cauchy-Schwarz inequality, $ab \le \delta a^2 + \frac{1}{4\delta}b^2$ and the pointwise bounds on η_h and its derivatives we get

$$\begin{split} \left| \sum_{i,j}^{d} \left(D_{-i}^{h} D_{j}^{h} u_{h}, D_{j}^{h} (\eta_{h}^{4}) \tau_{j}^{h} D_{-i}^{h} u_{h} \right)_{L^{2}(\mathbb{R}^{d})} \right| \\ &= \left| \sum_{i,j}^{d} \left(\eta_{h}^{2} D_{-i}^{h} D_{j}^{h} u_{h}, 4\eta_{h} D_{j}^{h} \eta_{h} \tau_{j}^{h} D_{-i}^{h} u_{h} \right)_{L^{2}(\mathbb{R}^{d})} \right| \\ &+ \left| \sum_{i,j}^{d} \left(\tau_{j}^{h} D_{-i}^{h} u_{h} - D_{-i}^{h} u_{h}, \left(\eta_{h}^{2} D_{j}^{h} \eta_{h} + \eta_{h} D_{j}^{h} (\eta_{h}^{2}) + D_{j}^{h} (\eta_{h}^{3}) \right) D_{j}^{h} \eta_{h} \tau_{j}^{h} D_{-i}^{h} u_{h} \right)_{L^{2}(\mathbb{R}^{d})} \right| \\ &\leq \frac{1}{4} \| \eta_{h}^{2} \nabla_{h}^{2} u_{h} \|_{L^{2}(\mathbb{R}^{d})}^{2} + \sum_{i,j}^{d} \left\| 4\eta_{h} D_{j}^{h} \eta_{h} \tau_{j}^{h} D_{-i}^{h} u_{h} \right\|_{L^{2}(Q_{r-h}(x))}^{2} \\ &+ \frac{1}{2(r-s)^{2}} \sum_{i,j}^{d} \left\| \tau_{j}^{h} D_{-i}^{h} u_{h} - D_{-i}^{h} u_{h} \right\|_{L^{2}(Q_{r-h}(x))}^{2} \end{split}$$

¹Note that in a continuous setting this term would not occur at all.

$$+ \frac{(r-s)^2}{2} \sum_{i,j}^{\mathsf{d}} \left\| \left(\eta_h^2 D_j^h \eta_h + \eta_h D_j^h (\eta_h^2) + D_j^h (\eta_h^3) \right) D_j^h \eta_h \tau_j^h D_{-i}^h u_h \right\|_{L^2(Q_{r-h}(x))}^2 \\ \leq \frac{1}{4} \| \eta_h^2 \nabla_h^2 u_h \|_{L^2(\mathbb{R}^{\mathsf{d}})}^2 + \frac{C}{(r-s)^4} \| u_h \|_{L^2(Q_r(x))}^2 + \frac{C}{(r-s)^2} \| \nabla_h u_h \|_{L^2(Q_r(x))}^2.$$

Analogously we can find the same upper bound for the other two terms on the right-hand side of (2.3.3). Then we obtain

$$\|\eta_h^2 \nabla_h^2 u_h\|_{L^2(\mathbb{R}^d)}^2 \leq \frac{3}{4} \|\eta_h^2 \nabla_h^2 u_h\|_{L^2(\mathbb{R}^d)}^2 + \frac{C}{(r-s)^4} \|u_h\|_{L^2(Q_r(x))}^2 + \frac{C}{(r-s)^2} \|\nabla_h u_h\|_{L^2(Q_r(x))}^2$$

and hence

$$\|\eta_h^2 \nabla_h^2 u_h\|_{L^2(\mathbb{R}^d)}^2 \leq \frac{C}{(r-s)^4} \|u_h\|_{L^2(Q_r(x))}^2 + \frac{C}{(r-s)^2} \|\nabla_h u_h\|_{L^2(Q_r(x))}^2$$

This implies the claim, once one notes that $\|\nabla_h^2 u_h\|_{L^2(Q_s(x))} \le \|\eta_h^2 \nabla_h^2 u_h\|_{L^2(\mathbb{R}^d)}$. \Box

2.4 Interpolation

We want to deduce discrete estimates from their continuous counterparts using compactness arguments. To do so, we need an interpolation operator that turns discrete functions into continuous functions having similar features. The most important property of this interpolation operator that we require is that the continuous derivatives of the output are comparable to the discrete derivatives of the input.

To construct such an operator we use B-splines (cf., e.g., [Sch81, §4.4]): For $m \ge 1, x \in \mathbb{R}$ the *m*-th normalized B-spline is given by

$$N^{m}(x) = m \sum_{i=0}^{m} \frac{(-1)^{i} {m \choose i} \max(x-i,0)^{m-1}}{m!}.$$

The function N^m is piecewise a polynomial of degree m - 1, has support in [0, m] and satisfies $\sum_{z \in \mathbb{Z}} N^m (x - z) = 1$ for all $x \in \mathbb{R}$. Furthermore its discrete and continuous derivatives are closely related. Indeed we have

$$\partial_x N^m(x) = N^{m-1}(x) - N^{m-1}(x-1) = D^1_{-1} N^{m-1}(x)$$
 (2.4.1)

for all $x \in \mathbb{R}$ (see [Sch81] for proofs).

We need a multidimensional version of these splines which is also adapted to the lattice $(h\mathbb{Z})^d$. So for h > 0, $\mu = (\mu_1, \ldots, \mu_n) \in \mathbb{N}^d$ with $\mu_i \ge 1$ let

$$N_h^{\mu}(x_1,\ldots,x_n) = N^{\mu_1}\left(\frac{x_1}{h}\right)\cdots N^{\mu_n}\left(\frac{x_n}{h}\right)$$

It follows easily from (2.4.1) that for any $\alpha \in \mathbb{N}^d$ with $\alpha_i < \mu_i$ for all *i* we have

$$D^{\alpha}N_{h}^{\mu} = D_{-h}^{\alpha}N_{h}^{\mu-\alpha}.$$
 (2.4.2)

Using this, we can define our interpolation operator:

Definition 2.4.1. Let h > 0, $\mu = (\mu_1, \ldots, \mu_n) \in \mathbb{N}^d$ with $\mu_i \ge 1$ for all *i*. Define $J_h^{\mu} \colon \mathbb{R}^{(h\mathbb{Z})^d} \to L^1_{loc}(\mathbb{R}^d)$ by

$$(J_h^{\mu}u_h)(x) = \sum_{z \in (h\mathbb{Z})^{\mathsf{d}}} u_h(z) N_h^{\mu}(x-z)$$

and extend J_h^{μ} to vector-valued functions component-wise.

Note that N_h^{μ} has compact support so that the above sum has only finitely many nonzero terms.

 J_h^{μ} does not interpolate the values of u_h (i.e. in general we will not have $J_h^{\mu}u_h(x) = u_h(x)$ for all $x \in (h\mathbb{Z})^d$). The maps $J_h^{\mu}u_h$ and u_h , however, share so many properties that we still call J_h^{μ} an interpolation operator.

Let us collect some properties of J_h^{μ} .

Proposition 2.4.2. Let J_h^{μ} be the family of interpolation operators that we have just defined, and let $u_h: (h\mathbb{Z})^d \to \mathbb{R}$.

- *i*) J_h^{μ} *is linear.*
- *ii)* $J_h^{\mu}u_h$ is piecewise a polynomial and is in the Sobolev space $W_{loc}^{(\min_i \mu_i)-1,2}$
- iii) J_h^{μ} is local in the sense that $(J_h^{\mu}u_h)(x)$ only depends on the values of u_h in $Q_{(\max; \mu_i)h}(x)$.
- *iv)* J_h^{μ} preserves constant functions, *i.e.* $(J_h^{\mu}c)(x) = c$ for any $c \in \mathbb{R}$ and any $x \in \mathbb{R}^d$.
- v) For every α with $\alpha_i < \mu_i$ we have $(D^{\alpha}J_h^{\mu}u_h)(x) = (J_h^{\mu-\alpha}(D_h^{\alpha}u_h))(x)$.
- *vi*) For every α with $\alpha_i < \mu_i$ and any $p \in [1, \infty]$ there is a constant $C = C(\mu, \alpha, d, p)$ such that for any $x \in \mathbb{R}^d$ and any $r \ge s + (1 + \max_i \mu_i)h$ we have

$$|D^{\alpha}J_{h}^{\mu}u_{h}\|_{L^{p}(Q_{s}(x))} \leq C \|D_{h}^{\alpha}u_{h}\|_{L^{p}(Q_{r}(x))}$$
(2.4.3)

and

$$\|D_h^{\alpha} u_h\|_{L^p(Q_s(x))} \le C \|D^{\alpha} J_h^{\mu} u_h\|_{L^p(Q_r(x))}.$$
(2.4.4)

Proof. Properties i), ii) and iii) are obvious. Property iv) easily follows from $\sum_{z \in \mathbb{Z}} N^m(x - z) = 1$ for all $x \in \mathbb{R}$, so it remains to prove v) and vi).

For v), note that we can assume that u_h is zero far away from x by iii). This means that all sums in the following calculations have only finitely many nonzero terms. Now, using (2.4.2), we can calculate that

$$(D^{\alpha}J_{h}^{\mu}u_{h})(x) = D^{\alpha}\left(\sum_{z\in(h\mathbb{Z})^{d}}u_{h}(z)N_{h}^{\mu}(x-z)\right)$$
$$= \sum_{z\in(h\mathbb{Z})^{d}}u_{h}(z)D^{\alpha}N_{h}^{\mu}(x-z)$$
$$= \sum_{z\in(h\mathbb{Z})^{d}}u_{h}(z)D_{-h}^{\alpha}N_{h}^{\mu-\alpha}(x-z)$$
$$= \sum_{z\in(h\mathbb{Z})^{d}}D_{h}^{\alpha}u_{h}(z)N_{h}^{\mu-\alpha}(x-z) = (J_{h}^{\mu-\alpha}(D_{h}^{\alpha}u_{h}))(x)$$

Finally we prove vi). In view of v) it is sufficient to consider the case $\alpha = 0$ here. We can also assume that $x \in (h\mathbb{Z})^d$ and $r, s \in h\mathbb{N} + \frac{h}{2}, r \ge s + (\max_i \mu_i)h$ (otherwise move x to the nearest lattice point, and replace r and s by $\lfloor r - \frac{h}{2} \rfloor_h + \frac{h}{2}$ and $\lfloor s - \frac{h}{2} \rfloor_h + \frac{3h}{2}$ respectively).

Let $y \in Q_s^h(x)$. The definition of J_h^{μ} immediately implies

$$\|J_{h}^{\mu}u_{h}\|_{L^{\infty}(Q_{h/2}(y))} \leq C \sup_{\substack{z \in (h\mathbb{Z})^{d} \\ |z-y| \leq (\max_{i} \mu_{i})h}} |u_{h}(z)|$$

and thus

$$\|J_h^{\mu}u_h\|_{L^p(Q_{h/2}(y))}^p \le C \sum_{\substack{z \in (h\mathbb{Z})^{\mathsf{d}} \\ |z-y| \le (\max_i \mu_i)h}} |u_h(z)|^p \le C \|u_h\|_{L^p(Q_{(\max_i \mu_i+1/2)h}(y))}^p.$$

If we sum this over all $y \in Q_s^h(x)$, we easily obtain (2.4.3).

For (2.4.4), by a similar argument it suffices to show

$$|u_h(y)| \le C \|J_h^{\mu} u_h\|_{L^p(Q_{h/2}(y))}$$
(2.4.5)

for all $y \in Q_s^h(x)$.

One can see this as follows: N_h^{μ} has support $[0, \mu_1] \times \cdots \times [0, \mu_n]$. This means that the values of $J_h^{\mu}u_h$ in $Q_{h/2}(y)$ depend on the finitely many values $\{u_h(z)\}_{z \in I_y}$, where $I_y := [y_1 - \mu_1] \times \cdots \times [y_d - \mu_d] \cap (h\mathbb{Z})^d$ and no others. Furthermore by linear independence of the B-splines (see [Sch81, Theorem 4.18] for the one-dimensional case; the d-dimensional case is analogous) $J_h^{\mu}u_h$ is identically 0 in $Q_{h/2}(y)$ only if all $\{u_h(z)\}_{z \in I_y}$ are 0. This means that $\|J_h^{\mu}u_h\|_{L^p(Q_{h/2}(y))}$ is not only a seminorm on \mathbb{R}^{I_y} but actually a norm. Now all norms on a finite-dimensional vector space are equivalent, so in particular

$$\|u_h\|_{l^2(I_y)} = \left(\sum_{z \in I_y} |u_h(z)|^2\right)^{\frac{1}{2}} \le C \|J_h^{\mu} u_h\|_{L^p(Q_{h/2}(y))}$$

for a constant *C* that is independent of *y*. This immediately implies (2.4.5).

Using these interpolation operators J_h^{μ} we define the two operators that we will actually use most often: One is $J_h := J_h^{(3,3,...,3)}$ and the other is the matrix interpolation operator \tilde{J}_h given by $(\tilde{J}_h)_{ij} = J_h^{(3,3,...,3)-e_i-e_j} \circ \tau_i^h$ (for example $(\tilde{J}_h)_{11} = J_h^{(1,3,...,3)} \circ \tau_1^h$).

One easily checks using parts ii) and v) of Proposition 2.4.2 that for any $f_h: (h\mathbb{Z})^d \to \mathbb{R}$ we have $J_h f_h \in W^{2,2}_{loc}(\mathbb{R}^d)$ and

$$\nabla^2 J_h f_h = \tilde{J}_h \nabla_h^2 f_h \,. \tag{2.4.6}$$

2.5 Inner decay estimates for discrete biharmonic functions: special cases

Our goal is to prove an $L^{\infty}-L^2$ estimate for discrete biharmonic functions (see Theorem 2.6.1): If $u_h \in \Phi_h$, $x \in \Lambda_h$, r > 0 and $\Delta_h^2 u_h(y) = 0$ for all $y \in Q_{r-h}(x) \cap \operatorname{int} \Lambda_h^d$, then, for all $z \in Q_{\frac{r}{2}}(x) \cap \Lambda_h$,

$$|\nabla_h^2 u_h(z)| \leq \frac{C}{r^{\frac{d}{2}}} \|\nabla_h^2 u_h\|_{L^2(Q_r(x))}.$$

To prove this estimate it will be necessary to distinguish where *x* lies in relation to $\partial \Lambda_h$: *x* can be far inside Λ_h , near a face, near an edge or near a vertex. In the following subsections we will study these cases separately and prove some decay estimates that we will then assemble to prove the aforementioned estimate.

2.5.1 Full space

Lemma 2.5.1. Let $d \in \mathbb{N}$, $u_h: (h\mathbb{Z})^d \to \mathbb{R}$, let $x \in (h\mathbb{Z})^d$, r > 0. Suppose $\Delta_h^2 u_h(y) = 0$ for all $y \in Q_{r-h}^h(x)$. Then

$$|
abla_h^2 u_h(x)| \leq rac{C}{r^{rac{d}{2}}} ||
abla_h^2 u_h||_{L^2(Q_r(x))}.$$

The main tool to prove this statement will be the following estimate:

Lemma 2.5.2. Let $d \in \mathbb{N}$. There exist constants $M \in \mathbb{N}$, $0 < \rho < \frac{1}{2}$ with the following property: Let $u_h: (h\mathbb{Z})^d \to \mathbb{R}$, r > 0, such that $\Delta_h^2 u_h(y) = 0$ for all $y \in Q_{r-h}^h$. Assume that $\rho r \ge Mh$. Then we have that

$$\left\|\nabla_{h}^{2}u_{h}-\left[\nabla_{h}^{2}u_{h}\right]_{Q_{\rho r}}\right\|_{L^{2}(Q_{\rho r})}^{2}\leq\rho^{n+1}\left\|\nabla_{h}^{2}u_{h}-\left[\nabla_{h}^{2}u_{h}\right]_{Q_{r}}\right\|_{L^{2}(Q_{r})}^{2}.$$

We will prove this lemma by contradiction using a compactness argument and the following decay estimate for continuous biharmonic functions:

Lemma 2.5.3. Let $d \in \mathbb{N}$, $0 < s \leq \frac{r}{2}$, $u \in W^{2,2}(Q_r)$ such that $\Delta^2 u = 0$ weakly in Q_r . Then we have

$$\left\|\nabla^{2} u - \left[\nabla^{2} u\right]_{Q_{s}}\right\|_{L^{2}(Q_{s})}^{2} \leq C\left(\frac{s}{r}\right)^{d+\frac{3}{2}} \left\|\nabla^{2} u - \left[\nabla^{2} u\right]_{Q_{r}}\right\|_{L^{2}(Q_{r})}^{2}.$$
(2.5.1)

Proof. The estimate (2.5.1) expresses the fact that the second derivatives of biharmonic functions are in the Campanato space $\mathcal{L}^{2,d+\frac{3}{2}} \simeq C^{0,3/4}$. The easiest way to show it is to use Schauder estimates for higher order elliptic equations as follows.

By scaling we can assume r = 1. By replacing u with $u - \frac{1}{2} [\nabla^2 u]_{Q_1} : x \otimes x$ we can assume that $[\nabla^2 u]_{Q_1} = 0$. Now by Schauder estimates (see e.g. [Mor66, Theorem 6.4.8] or [Cam80, Cap. II, Teorema 6.I]) we have that any $C^{0,\alpha}$ -Hölder seminorm of $\nabla^2 u$ in $Q_{1/2}$ is bounded by the L^2 -norm of $\nabla^2 u$ in Q_1 . In particular, we have

$$\left[\nabla^2 u\right]_{C^{0,\frac{3}{4}}(Q_{1/2})} \le C \left\|\nabla^2 u\right\|_{L^2(Q_1)}.$$

On the other hand, Jensen's inequality easily yields that

$$\begin{split} \left\| \nabla^2 u - \left[\nabla^2 u \right]_{Q_s} \right\|_{L^2(Q_s)}^2 &\leq \frac{1}{|Q_s|} \int_{Q_s} \int_{Q_s} |\nabla^2 u(y) - \nabla^2 u(y')|^2 \, \mathrm{d}y \, \mathrm{d}y' \\ &\leq C s^{\mathsf{d} + \frac{3}{2}} \left[\nabla^2 u \right]_{C^{0, \frac{3}{4}}(Q_{1/2})}^2 \, . \end{split}$$

Together with the previous estimate this yields the result.

We will also need a local version of the well-known Kolmogorov-Riesz-Fréchet compactness theorem.

Lemma 2.5.4. Let $d \in \mathbb{N}$, $p \in [1, \infty)$, let $U, V, W \subset \mathbb{R}^d$ be open with U compactly contained in V, and V compactly contained in W. Let A be a subset of $L^p(W)$.

i) If A is bounded in $L^p(W)$ and

$$\lim_{\delta \to 0} \sup_{f \in A} \|\tau_{\delta} f - f\|_{L^p(V)} = 0$$

then A (or rather the restriction of the elements of A to U) is precompact in $L^{p}(U)$.

ii) If A is precompact in $L^p(W)$ then

$$\lim_{\delta \to 0} \sup_{f \in A} \|\tau_{\delta} f - f\|_{L^p(V)} = 0$$

Proof. Part i) follows by applying the usual Kolmogorov-Riesz-Fréchet compactness theorem (see e.g. [Bre11, Corollary 4.27 and Exercise 4.34]) to the family $\{\eta f : f \in A\}$, where η is a smooth cut-off function that is 1 on U and 0 outside of V.

For part ii) let \tilde{V} be open such that V is compactly contained in \tilde{V} and \tilde{V} is compactly contained in W, and let ζ be a cut-off function that is 1 on \tilde{V} and 0 outside of W. Then the family { $\zeta f : f \in A$ } is precompact in $L^p(\mathbb{R}^d)$ and the statement is obtained by applying the converse of the Kolmogorov-Riesz-Fréchet compactness theorem to that family.

After these preparations we can return to the proofs of Lemma 2.5.1 and Lemma 2.5.2.

Proof of Lemma 2.5.2.

Step 1: Set-up of the compactness argument

Let the constant $\rho \leq \frac{1}{2}$ be fixed later, and suppose that the statement for that fixed ρ is wrong. Then for any $k \in \mathbb{N}$ there exist $M_k \geq k$, $h_k > 0$, $u_{h_k} \colon (h_k \mathbb{Z})^d \to \mathbb{R}$, $r_k > 0$ such that

$$\left\|\nabla_{h_{k}}^{2}u_{h_{k}}-\left[\nabla_{h_{k}}^{2}u_{h_{k}}\right]_{Q_{\rho r_{k}}}\right\|_{L^{2}(Q_{\rho r_{k}})}^{2}>\rho^{\mathsf{d}+1}\left\|\nabla_{h_{k}}^{2}u_{h_{k}}-\left[\nabla_{h_{k}}^{2}u_{h_{k}}\right]_{Q_{r_{k}}}\right\|_{L^{2}(Q_{r_{k}})}^{2}.$$
(2.5.2)

By rescaling the lattice by a factor of r_k , we can assume that all the r_k are equal to 1. Because $h_k \leq \frac{\rho}{M_k} \leq \frac{\rho}{k}$, we have that $h_k \to 0$. Omitting finitely many k, we can assume that all h_k are small (less than $\frac{1}{1000}$, say).

By replacing u_{h_k} with $u_{h_k} - \frac{1}{2} \left[\nabla_{h_k}^2 u_{h_k} \right]_{Q_1} : x \otimes x$ we can assume that $\left[\nabla_{h_k}^2 u_{h_k} \right]_{Q_1} = 0$, and by scaling we can assume that $\left\| \nabla_{h_k}^2 u_{h_k} \right\|_{L^2(Q_1)} = 1$ (note that $\nabla_{h_k}^2 u_{h_k}$ cannot be identically 0, as then u_{h_k} would be affine, and so both sides of (2.5.2) would be 0). Then (2.5.2) implies that

$$\left\|\nabla_{h_k}^2 u_{h_k} - \left[\nabla_{h_k}^2 u_{h_k}\right]_{Q_{\rho}}\right\|_{L^2(Q_{\rho})}^2 > \rho^{\mathsf{d}+1}.$$
(2.5.3)

Finally, we replace u_{h_k} by $u_{h_k} - a_k - b_k \cdot x$, where $a_k \in \mathbb{R}$, $b_k \in \mathbb{R}^d$ are constants that will be chosen below (such that equation (2.5.4) is satisfied). This leaves $\nabla_{h_k}^2 u_{h_k}$ unaffected, so all the above statements about $\nabla_{h_k}^2 u_{h_k}$ remain true.

We let $v_k = J_{h_k} u_{h_k}$, where $J_{h_k} = J_{h_k}^{(3,\dots,3)}$ is the interpolation operator introduced in Section 2.4. From $\left\| \nabla_{h_k}^2 u_{h_k} \right\|_{L^2(Q_1)} = 1$ and Proposition 2.4.2 vi) we immediately conclude that $\| \nabla^2 v_k \|_{L^2(Q_{13/14})} \leq C$.

Now we choose a_k and b_k in such a way that

$$[v_k]_{Q_{13/14}} = 0, \qquad [\nabla v_k]_{Q_{13/14}} = 0.$$
(2.5.4)

The Poincaré inequality on $Q_{13/14}$ implies that $||v_k||_{W^{2,2}(Q_{13/14})} \leq C ||\nabla^2 v_k||_{L^2(Q_{13/14})} \leq C$. Therefore the v_k are bounded in $W^{2,2}(Q_{13/14})$ and hence have a subsequence (not relabeled) that converges weakly to some $v \in W^{2,2}(Q_{13/14})$.

Step 2: $\Delta^2 v = 0$

We claim that $\Delta^2 v = 0$ weakly in $Q_{13/14}$. To prove this, let $\varphi \in C_c^{\infty}(Q_{13/14})$ be arbitrary and let φ_{h_k} be its restriction to $(h_k \mathbb{Z})^d$. We need to prove that $\int_{Q_{13/14}} \nabla^2 v : \nabla^2 \varphi \, dx = 0$.

We have by (2.4.6) that

$$\begin{split} &\int_{Q_{13/14}} \nabla^2 v_k : \nabla^2 \varphi \, \mathrm{d}x = \int_{Q_{13/14}} \nabla^2 J_{h_k} u_{h_k} : \nabla^2 \varphi \, \mathrm{d}x \\ &= \int_{Q_{13/14}} \tilde{J}_{h_k} \nabla^2_{h_k} v_k : \nabla^2 \varphi \, \mathrm{d}x \\ &= \sum_{i,j=1}^d \int_{Q_{13/14}} J_{h_k}^{(3,3,\dots,3)-e_i-e_j} \circ \tau_i^{h_k} D_{-i}^{h_k} D_j^{h_k} v_k D_i D_j \varphi \, \mathrm{d}x \\ &= \sum_{i,j=1}^d \int_{Q_{13/14}} \sum_{z \in (h_k \mathbb{Z})^d} N_{h_k}^{(3,3,\dots,3)-e_i-e_j} (x-z) D_i^{h_k} D_j^{h_k} u_{h_k}(z) D_i D_j \varphi(x) \, \mathrm{d}x \\ &= \sum_{i,j=1}^d \sum_{z \in (h_k \mathbb{Z})^d} D_i^{h_k} D_j^{h_k} u_{h_k}(z) \int_{Q_{13/14}} N_{h_k}^{(3,3,\dots,3)-e_i-e_j} (x-z) D_i D_j \varphi(x) \, \mathrm{d}x \, . \end{split}$$

Now Taylor expansion and the fact that $\int_{Q_{13/14}} N_{h_k}^{(3,3,\dots,3)-\delta_i-\delta_j} = 1$ imply that

$$\int_{Q_{13/14}} N_{h_k}^{(3,3,\dots,3)-e_i-e_j}(x-z) D_i D_j \varphi(x) \, \mathrm{d}x = D_i D_j \varphi(z) + O(h_k) = D_i^{h_k} D_j^{h_k} \varphi_{h_k}(z) + O(h_k)$$

In addition, from $\Delta_{h_k}^2 u_{h_k} = 0$ in $Q_{13/14}$ we conclude that

$$\begin{split} \sum_{i,j=1}^{d} \sum_{z \in (h_k \mathbb{Z})^d} D_i^{h_k} D_j^{h_k} u_{h_k}(z) D_i^{h_k} D_j^{h_k} \varphi_{h_k}(z) &= \sum_{i,j=1}^{d} \sum_{z \in (h_k \mathbb{Z})^d} D_{-i}^{h_k} D_j^{h_k} u_{h_k}(z) D_{-i}^{h_k} D_j^{h_k} \varphi_{h_k}(z) \\ &= (\nabla_{h_k}^2 u_{h_k}, \nabla_{h_k}^2 \varphi_{h_k})_{L^2(\mathbb{R}^d)} = 0 \end{split}$$

and so we obtain

$$\left|\int_{Q_{13/14}} \nabla^2 v_k : \nabla^2 \varphi \,\mathrm{d}x\right| \leq C \left\|\nabla_{h_k}^2 u_{h_k}\right\|_{L^2(Q_1)} h_k = Ch_k.$$

Using weak convergence of $\nabla^2 v_k$ we can pass to the limit here and get

$$\int_{Q_{13/14}} \nabla^2 v : \nabla^2 \varphi \, \mathrm{d} x = 0$$

Step 3: Strong convergence of v_k

Let $w_k = I_{h_k}^{pc} \nabla_{h_k}^2 u_{h_k}$. We claim that both $\nabla^2 v_k$ and w_k converge strongly in $L^2(Q_{1/2})$ to $\nabla^2 v$. Step 3.1: Precompactness of w_k

We first prove that $(w_k)_{k \in \mathbb{N}}$ is precompact in $L^2(Q_{4/7})$.

Because $(\nabla_{h_k}^2 u_{h_k})$ is bounded in $L^2(Q_1)$, w_k is bounded in $L^2(Q_1)$. So, according to Lemma 2.5.4 i), it suffices to verify that

$$\lim_{\substack{a \in \mathbb{R}^{d} \\ |a| \to 0}} \sup_{k \in \mathbb{N}} \|\tau_{a} w_{k} - w_{k}\|_{L^{2}(Q_{5/7})} = 0.$$
(2.5.5)

Let $a \in (h\mathbb{Z})^d$ such that $|a| \leq \frac{1}{7}$. Then $\Delta_{h_k}^2(\tau_a u_{h_k} - u_{h_k}) = 0$ in $Q_{11/14}$, so by the Caccioppoli inequality we obtain

$$\begin{aligned} \|\nabla_{h_k}^2(\tau_a u_{h_k} - u_{h_k})\|_{L^2(Q_{5/7}(x))}^2 &\leq C \|\tau_a u_{h_k} - u_{h_k}\|_{L^2(Q_{11/14}(x))}^2 \\ &+ C \|\nabla_{h_k}(\tau_a u_{h_k} - u_{h_k})\|_{L^2(Q_{11/14}(x))}^2. \end{aligned}$$

Here the left-hand side is equal to $\|\tau_a w_k - w_k\|_{L^2(Q_{5/7})}^2$, while we can use Proposition 2.4.2 vi) to bound the right-hand side. We obtain

$$\|\tau_a w_k - w_k\|_{L^2(Q_{5/7})}^2 \le C \|\tau_a v_k - v_k\|_{L^2(Q_{6/7}(x))}^2 + C \|\tau_a \nabla v_k - \nabla v_k\|_{L^2(Q_{6/7}(x))}^2$$

Recall that (v_k) is bounded in $W^{2,2}(Q_{13/14})$. Hence by the compact Sobolev embedding, (v_k) and (∇v_k) are precompact in $L^2(Q_{13/14})$. Thus by Lemma 2.5.4 ii),

$$\lim_{a \to 0} \sup_{k \in \mathbb{N}} \left(\|\tau_a v_k - v_k\|_{L^2(Q_{6/7}(x))}^2 + \|\tau_a \nabla v_k - \nabla v_k\|_{L^2(Q_{6/7}(x))}^2 \right) = 0$$

(note that this expression is defined for all a > 0, not just those in $(h\mathbb{Z})^d$).

In particular,

$$\lim_{\delta \to 0} \sup_{k \in \mathbb{N}} \sup_{\substack{a \in (h_k \mathbb{Z})^{\mathsf{d}} \\ |a| < \delta}} \left(\|\tau_a v_k - v_k\|_{L^2(Q_{6/7}(x))}^2 + \|\tau_a \nabla v_k - \nabla v_k\|_{L^2(Q_{6/7}(x))}^2 \right) = 0$$

and therefore

$$\lim_{\delta \to 0} \sup_{k \in \mathbb{N}} \sup_{\substack{a \in (h_k \mathbb{Z})^{\mathsf{d}} \\ |a| \leq \delta}} \|\tau_a w_k - w_k\|_{L^2(Q_{5/7}(x))} = 0.$$

It remains to consider shifts τ_a where $a \notin (h_k \mathbb{Z})^d$. This is possible because w_k is piecewise constant on cubes of sidelength h_k . This easily implies that for any $a \in \mathbb{R}^d$ we have

$$\|\tau_a w_k - w_k\|_{L^2(Q_{9/14}(x))} \le C \sup_{\substack{b \in (h_k \mathbb{Z})^{\mathsf{d}} \\ |b-a| \le h_k}} \|\tau_b w_k - w_k\|_{L^2(Q_{5/7}(x))}.$$

Combining this with the previous estimate we find that

$$\lim_{\delta \to 0} \sup_{k \in \mathbb{N}} \sup_{\substack{a \in \mathbb{R}^{\mathsf{d}} \\ |a| \leq \delta + h_k}} \|\tau_a w_k - w_k\|_{L^2(Q_{9/14}(x))} = 0.$$

Because $h_k \rightarrow 0$, this implies

$$\lim_{\substack{a \in \mathbb{R}^{d} \\ |a| \to 0}} \limsup_{k \to \infty} \|\tau_{a} w_{k} - w_{k}\|_{L^{2}(Q_{9/14}(x))} = 0.$$
(2.5.6)

We finally show that (2.5.6) already implies (2.5.5). It follows from (2.5.6) that for every fixed $\varepsilon > 0$ there are $\delta > 0$, $K \in \mathbb{N}$ such that $\sup_{k \ge K} \|\tau_a w_k - w_k\|_{L^2(Q_{9/14}(x))} \le \varepsilon$ for all a with $|a| \le \delta$. For the finitely many k < K, we use that $\lim_{\substack{a \in \mathbb{N}^d \\ |a| \to 0}} \|\tau_a w_k - w_k\|_{L^2(Q_{9/14}(x))} = 0$ to see that for a potentially smaller δ' we have $\sup_{k \in \mathbb{N}} \|\tau_a w_k - w_k\|_{L^2(Q_{9/14}(x))} \le \varepsilon$ for all a with $|a| \le \delta'$.

Therefore the sequence (w_k) is precompact in $L^2(Q_{4/7}(x))$. Choose a subsequence (not relabeled) converging strongly to some $w \in L^2(Q_{4/7}(x))$.

Step 3.2: Strong convergence of $(\nabla^2 v_k)$ *and* $w = \nabla^2 v$

We split w into a smooth part and a part with small L^2 -norm. Let $\varepsilon > 0$ be arbitrary, and choose a $w^{(\varepsilon)}$ in $C_c^{\infty}(Q_{4/7})$ such that $||w - w^{(\varepsilon)}||_{L^2(Q_{4/7})} \leq \varepsilon$. We denote the restriction of $w^{(\varepsilon)}$ to $(h_k \mathbb{Z})^d$ by $w_{h_k}^{(\varepsilon)}$. Using Taylor expansion, one immediately verifies that then $I_{h_k}^{pc} w_{h_k}^{(\varepsilon)}$ and $\tilde{J}_{h_k} w_{h_k}^{(\varepsilon)}$ converge to $w^{(\varepsilon)}$ in $L^2(Q_{4/7})$ and $L^2(Q_{1/2})$, respectively.

This means in particular that

$$\lim_{k\to\infty} \|w_{h_k}^{(\varepsilon)} - \nabla_{h_k}^2 u_{h_k}\|_{L^2(Q_{4/7})} = \|w^{(\epsilon)} - w\|_{L^2(Q_{4/7})} \le \varepsilon.$$

Using Proposition 2.4.2 vi), we conclude that

$$\limsup_{k\to\infty}\left\|\tilde{J}_{h_k}\left(w_{h_k}^{(\varepsilon)}-\nabla_{h_k}^2u_{h_k}\right)\right\|_{L^2(Q_{1/2})}\leq C\varepsilon$$

The left-hand side here equals $\limsup_{k\to\infty} \|w - \nabla^2 v_k\|_{L^2(Q_{1/2})}$, and so we obtain

$$\limsup_{k\to\infty} \|w-\nabla^2 v_k\|_{L^2(Q_{1/2})} \le C\epsilon.$$

Since ε was arbitrary, we conclude that $(\nabla^2 v_k)$ converges strongly in $L^2(Q_{1/2})$ to w. But we already know that $(\nabla^2 v_k)$ converges weakly in $L^2(Q_{13/14})$ to $\nabla^2 v$, so we obtain that $\nabla^2 v = w$ in $Q_{1/2}$.

Step 4: Conclusion of the argument

We proved that $w_k = I_{h_k}^{pc} \nabla_{h_k}^2 u_{h_k}$ converges strongly in $L^2(Q_{1/2})$ to $\nabla^2 v$. Because $\rho \leq \frac{1}{2}$ then also $\nabla_{h_k}^2 u_{h_k} - \left[\nabla_{h_k}^2 u_{h_k}\right]_{Q_{\rho}}$ converges strongly in $L^2(Q_{1/2})$ to $\nabla^2 v - \left[\nabla^2 v\right]_{Q_{\rho'}}$ and so from (2.5.3) we conclude that

$$\left\|\nabla^2 v - \left[\nabla^2 v\right]_{Q_{\rho}}\right\|_{L^2(Q_{\rho})}^2 \ge \rho^{\mathsf{d}+1}.$$

In addition, we know that $\|\nabla^2 v_k\|_{L^2(Q_{13/14})} \leq C$, and also that $\nabla^2 v_k$ converges weakly in $L^2(Q_{13/14})$ to $\nabla^2 v$. This implies

$$\left\|\nabla^2 v - \left[\nabla^2 v\right]_{Q_{13/14}}\right\|_{L^2(Q_{13/14})}^2 \le \left\|\nabla^2 v\right\|_{L^2(Q_{13/14})}^2 \le \liminf_{k \to \infty} \left\|\nabla^2 v_k\right\|_{L^2(Q_{13/14})}^2 \le C.$$

In summary, we have proved that there is a constant C_1 independent of ρ such that

$$\left\|\nabla^{2}v - \left[\nabla^{2}v\right]_{Q_{\rho}}\right\|_{L^{2}(Q_{\rho})}^{2} \ge \frac{\rho^{\mathsf{d}+1}}{C_{1}} \left\|\nabla^{2}v - \left[\nabla^{2}v\right]_{Q_{13/14}}\right\|_{L^{2}(Q_{13/14})}^{2}.$$
(2.5.7)

On the other hand, $\Delta^2 v = 0$ in $Q_{13/14}$, and thus Lemma 2.5.3 implies that

$$\left\|\nabla^2 v - \left[\nabla^2 v\right]_{Q_{\rho}}\right\|_{L^2(Q_{\rho})}^2 \le C_2 \left(\frac{\rho}{\frac{13}{14}}\right)^{\mathsf{d}+\frac{3}{2}} \left\|\nabla^2 v - \left[\nabla^2 v\right]_{Q_{13/14}}\right\|_{L^2(Q_{13/14})}^2$$

for a constant C_2 independent of ρ .

This is a contradiction to (2.5.7) provided that we choose ρ small enough, namely $\rho < \frac{1}{C_1^2 C_2^2} \left(\frac{13}{14}\right)^{2d+3}$. So we finally fix a ρ satisfying this condition, and proved that falsity of the claim leads to a contradiction.

Now we can return to Lemma 2.5.1.

Proof of Lemma 2.5.1. We can assume w.l.o.g. that x = 0.

We claim that for any $0 < s' \le s \le r$ we have

$$\left\|\nabla_{h}^{2}u_{h}-\left[\nabla_{h}^{2}u_{h}\right]_{Q_{s'}}\right\|_{L^{2}(Q_{s'})}^{2} \leq C\left(\frac{s'}{s}\right)^{d+1}\left\|\nabla_{h}^{2}u_{h}-\left[\nabla_{h}^{2}u_{h}\right]_{Q_{s}}\right\|_{L^{2}(Q_{s})}^{2}.$$
(2.5.8)

To prove this estimate, observe first that we can assume $s' \ge \frac{h}{2}$, as otherwise the left-hand side is 0. We can also assume $\frac{s}{s'} \ge 2M$ (where *M* is the constant from Lemma 2.5.2), as otherwise we can trivially estimate

$$\begin{split} \left\| \nabla_{h}^{2} u_{h} - \left[\nabla_{h}^{2} u_{h} \right]_{Q_{s'}} \right\|_{L^{2}(Q_{s'})}^{2} &\leq \left\| \nabla_{h}^{2} u_{h} - \left[\nabla_{h}^{2} u_{h} \right]_{Q_{s}} \right\|_{L^{2}(Q_{s})}^{2} \\ &\leq C \left(\frac{s'}{s} \right)^{\mathsf{d}+1} \left\| \nabla_{h}^{2} u_{h} - \left[\nabla_{h}^{2} u_{h} \right]_{Q_{s}} \right\|_{L^{2}(Q_{s})}^{2} , \end{split}$$

which holds for $C \ge (2M)^{d+1}$.

So we assume $s' \ge \frac{h}{2}$ and $\frac{s}{s'} \ge 2M$. Then in particular $s \ge Mh$. Consider the ρ from Lemma 2.5.2 and let κ be the largest integer such that $\rho^{\kappa}s \ge \max(s', Mh)$. We can then apply Lemma 2.5.2 repeatedly with radii $s, \rho s, \ldots, \rho^{\kappa}s$ to find

$$\left\| \nabla_{h}^{2} u_{h} - \left[\nabla_{h}^{2} u_{h} \right]_{Q_{\rho^{\kappa_{s}}}} \right\|_{L^{2}(Q_{\rho^{\kappa_{s}}})}^{2} \leq \rho^{\kappa(\mathsf{d}+1)} \left\| \nabla_{h}^{2} u_{h} - \left[\nabla_{h}^{2} u_{h} \right]_{Q_{s}} \right\|_{L^{2}(Q_{s})}^{2} .$$

Because $s' \leq \rho^{\kappa} s$, we also have

$$\left\|\nabla_{h}^{2}u_{h}-\left[\nabla_{h}^{2}u_{h}\right]_{Q_{s'}}\right\|_{L^{2}(Q_{s'})}^{2} \leq \left\|\nabla_{h}^{2}u_{h}-\left[\nabla_{h}^{2}u_{h}\right]_{Q_{\rho^{\kappa_{s}}}}\right\|_{L^{2}(Q_{\rho^{\kappa_{s}}})}^{2}$$

Here we have used the fact that $||f - [f]_{\Omega}||_{L^2(\Omega)}$ is monotone in Ω . If we combine the last two estimates and observe that $\rho^{\kappa+1}s < \max(s', Mh) \le 2Ms'$, i.e. $\rho^{\kappa} \le \frac{2M}{\rho}\frac{s'}{s}$, we indeed obtain (2.5.8) with $C = \left(\frac{2M}{\rho}\right)^{d+1}$.

Now using (2.5.8) to prove the lemma is a standard iteration argument as e.g. in [Gia93, Theorem 3.1]. For the sake of completeness we sketch the proof.

If we apply (2.5.8) with s = r and $s' = \frac{r}{2^{\lambda}}$ or $s' = \frac{r}{2^{\lambda+1}}$, we can estimate

$$\begin{split} \left\| \left[\nabla_{h}^{2} u_{h} \right]_{Q_{r/2^{\lambda+1}}} &- \left[\nabla_{h}^{2} u_{h} \right]_{Q_{r/2^{\lambda}}} \right\|_{L^{2}(Q_{r/2^{\lambda+1}})}^{2} \\ &\leq 2 \left\| \nabla_{h}^{2} u_{h} - \left[\nabla_{h}^{2} u_{h} \right]_{Q_{r/2^{\lambda}}} \right\|_{L^{2}(Q_{r/2^{\lambda}})}^{2} + 2 \left\| \nabla_{h}^{2} u_{h} - \left[\nabla_{h}^{2} u_{h} \right]_{Q_{r/2^{\lambda+1}}} \right\|_{L^{2}(Q_{r/2^{\lambda+1}})}^{2} \\ &\leq \frac{C}{2^{\lambda(d+1)}} \left\| \nabla_{h}^{2} u_{h} - \left[\nabla_{h}^{2} u_{h} \right]_{Q_{r}} \right\|_{L^{2}(Q_{r})}^{2} \end{split}$$

and hence

$$\left| \left[\nabla_h^2 u_h \right]_{\mathcal{Q}_{r/2^{\lambda+1}}} - \left[\nabla_h^2 u_h \right]_{\mathcal{Q}_{r/2^{\lambda}}} \right| \le \frac{C}{r^{\frac{d}{2}} 2^{\frac{\lambda}{2}}} \left\| \nabla_h^2 u_h - \left[\nabla_h^2 u_h \right]_{\mathcal{Q}_r} \right\|_{L^2(\mathcal{Q}_r)}$$

If we sum this for $\lambda = 0, 1, ...$ and observe that for λ small enough $\left[\nabla_h^2 u_h\right]_{Q_{r/2\lambda}} = \nabla_h^2 u_h(0)$ we obtain

$$\left|\nabla_h^2 u_h(0) - \left[\nabla_h^2 u_h\right]_{Q_r}\right| \le \frac{C}{r^{\frac{d}{2}}} \left\|\nabla_h^2 u_h - \left[\nabla_h^2 u_h\right]_{Q_r}\right\|_{L^2(Q_r)}$$

Now we can estimate

$$\begin{aligned} \left| \nabla_{h}^{2} u_{h}(0) \right|^{2} &\leq 2 \left| \nabla_{h}^{2} u_{h}(0) - \left[\nabla_{h}^{2} u_{h} \right]_{Q_{r}} \right|^{2} + 2 \left| \left[\nabla_{h}^{2} u_{h} \right]_{Q_{r}} \right|^{2} \\ &\leq \frac{C}{r^{\mathsf{d}}} \left(\left\| \nabla_{h}^{2} u_{h} - \left[\nabla_{h}^{2} u_{h} \right]_{Q_{r}} \right\|_{L^{2}(Q_{r})}^{2} + \left\| \left[\nabla_{h}^{2} u_{h} \right]_{Q_{r}} \right\|_{L^{2}(Q_{r})}^{2} \right) \\ &= \frac{C}{r^{\mathsf{d}}} \left\| \nabla_{h}^{2} u_{h} \right\|_{L^{2}(Q_{r})}^{2}, \end{aligned}$$

which proves the claim.

2.5.2 Half-space

In the half-space we want to prove the following statement, which is a slightly weaker analogue of Lemma 2.5.1:

Lemma 2.5.5. Let $d \in \mathbb{N}$, u_h : $(h\mathbb{Z})^d \to \mathbb{R}$, let $x \in (h\mathbb{Z})^d$, r > 0, $\nu \in \{e_1, -e_1, \ldots, e_n, -e_n\}$. Suppose that $u_h(y) = 0$ for all $y \in Q_r^h(x)$ such that $(y - x) \cdot \nu \leq 0$, and $\Delta_h^2 u_h(y) = 0$ for all $y \in Q_r^{h}(x)$ such that $(y - x) \cdot \nu > 0$. Then, for any $s \leq r$,

$$\|\nabla_{h}^{2}u_{h}\|_{L^{2}(Q_{s}(x))} \leq C\left(\frac{s}{r}\right)^{\frac{d}{2}} \|\nabla_{h}^{2}u_{h}\|_{L^{2}(Q_{r}(x))}.$$

The proof is mostly similar to that of Lemma 2.5.1, so we only give details where a new idea is required.

For r > 0 let $Q_{r,+} = Q_r \cap \{x_1 > 0\}$. The main step in the proof of Lemma 2.5.5 will be to prove the following estimate.

Lemma 2.5.6. Let $d \in \mathbb{N}$. There exist constants $M \in \mathbb{N}$, $0 < \rho < \frac{1}{2}$ with the following property: Let $u_h: (h\mathbb{Z})^d \to \mathbb{R}$, r > 0 be such that $u_h(y) = 0$ whenever $y \in Q_r^h$ and $y_1 \le 0$, and $\Delta_h^2 u_h(y) = 0$ for all $y \in Q_{r-h}^h$ such that $y_1 > 0$. Assume that $\rho r \ge Mh$. Then we have

$$\begin{split} \left\| \nabla_{h}^{2} u_{h} - \left[D_{-1}^{h} D_{1}^{h} u_{h} \right]_{Q_{\rho r,+}} e_{1} \otimes e_{1} \right\|_{L^{2}(Q_{\rho r,+})}^{2} \\ & \leq \rho^{\mathsf{d}+1} \left\| \nabla_{h}^{2} u_{h} - \left[D_{-1}^{h} D_{1}^{h} u_{h} \right]_{Q_{r,+}} e_{1} \otimes e_{1} \right\|_{L^{2}(Q_{r,+})}^{2} \end{split}$$

Using a compactness argument, we will deduce this estimate from the following continuous estimate.

Lemma 2.5.7. Let $d \in \mathbb{N}$, $0 < s \leq \frac{r}{2}$, $u \in W^{2,2}(Q_{r,+})$. Assume that $\Delta^2 u = 0$ weakly in $Q_{r,+}$ and that u = 0, $D_1 u = 0$ on $\partial Q_{r,+} \cap \{x_1 = 0\}$ in the sense of traces. Then we have

$$\left\|\nabla^{2}u - \left[D_{1}^{2}u\right]_{Q_{s,+}}e_{1} \otimes e_{1}\right\|_{L^{2}(Q_{s,+})}^{2} \leq C\left(\frac{s}{r}\right)^{d+\frac{3}{2}} \left\|\nabla^{2}u - \left[D_{1}^{2}u\right]_{Q_{r,+}}e_{1} \otimes e_{1}\right\|_{L^{2}(Q_{r,+})}^{2}.$$

Proof. This follows like Lemma 2.5.3 from Schauder estimates up to the boundary (cf. [Mor66, Theorem 6.4.8]).

Proof of Lemma 2.5.6. Step 1: Preparations

We follow the same strategy as in the proof of Lemma 2.5.2. That is, we assume that the

claim is wrong for some fixed ρ , and consider a sequence of counterexamples u_{h_k} and their interpolations $v_k = I_{h_k}u_{h_k}$. We can assume that $r_k = 1$.

Next observe that for $\omega_h(x) := \begin{cases} \frac{x_1(x_1+h)}{2} & x_1 \ge 0\\ 0 & x_1 < 0 \end{cases}$ we have $\omega_h(x) = 0$ if $x_1 \le 0$ and $\sum_{h=0}^{h} \sum_{h=0}^{h} \sum_{h=0}^$

 $D_{-1}^{h}D_{1}^{h}\omega_{h}(x) = \begin{cases} 1 & x_{1} \geq 0 \\ 0 & x_{1} < 0 \end{cases}$ So by replacing u_{h} with $u_{h} - \left[D_{-1}^{h}D_{1}^{h}u_{h}\right]_{Q_{1,+}}\omega_{h}$ we can also assume $\left[D_{-1}^{h}D_{1}^{h}u_{h}\right]_{Q_{1,+}} = 0$. Having normalized u_{h} on $Q_{1,+}$ in this way, we now consider Q_{1} again. We can assume

$$\left\|\nabla_{h_k}^2 u_{h_k}\right\|_{L^2(Q_1)} = 1.$$
(2.5.9)

Note that

$$\left\|\nabla_{h_{k}}^{2}u_{h_{k}}\right\|_{L^{2}(Q_{1})}^{2} = \left\|\nabla_{h_{k}}^{2}u_{h_{k}}\right\|_{L^{2}(Q_{1,+})}^{2} + \left\|\nabla_{h_{k}}^{2}u_{h_{k}}\right\|_{L^{2}((-h/2,0)\times(-1,1)^{d-1})}^{2}$$

and

$$\left\|\nabla_{h_{k}}^{2}u_{h_{k}}\right\|_{L^{2}((-h/2,0)\times(-1,1)^{d-1})}^{2}=\left\|\nabla_{h_{k}}^{2}u_{h_{k}}\right\|_{L^{2}((0,h/2)\times(-1,1)^{d-1})}^{2}$$

Now (2.5.9) combined with the last two equalities implies that $\left\|\nabla_{h_k}^2 u_{h_k}\right\|_{L^2(Q_{1,+})} \geq \frac{1}{2}$, so that

$$\left\|\nabla_{h_k}^2 u_{h_k} - \left[D_{-1}^h D_1^h u_{h_k}\right]_{Q_{\rho,+}} e_1 \otimes e_1 \right\|_{L^2(Q_{\rho,+})}^2 > \frac{\rho^{d+1}}{2}.$$
(2.5.10)

By (2.5.9), Proposition 2.4.2 and the Poincaré inequality with 0 boundary values (v_k) is bounded in $W^{2,2}(Q_{3/4})$, and so a non-relabeled subsequence converges weakly to some v in $W^{2,2}(Q_{3/4})$.

As in step 2 of the proof of Lemma 2.5.2 we can show that $\Delta^2 v = 0$ weakly in $Q_{3/4,+}$. We have $u_{h_k} = 0$ in $Q_1 \cap \{x_1 < 0\}$ and hence $v_k = 0$ in $Q_{3/4} \cap \{x_1 < -3h_k\}$. Since v_k converges to v strongly in $L^2(Q_{3/4})$, v = 0 in $\{x_1 < 0\}$, and because $v \in W^{2,2}(Q_{3/4})$, we obtain that v = 0 and $D_1v = 0$ on $Q_{3/4} \cap \{x_1 = 0\}$ in the sense of traces.

We define $w_k = I_{h_k}^{pc} \nabla_{h_k}^2 u_{h_k}$ and want to show next that $\nabla^2 v_k$ and w_k converge to $\nabla^2 v$ strongly in $L^2(Q_{1/2})$. We cannot directly reuse the argument in Step 3 of the proof of Lemma 2.5.2, as we now have to deal with boundary values. However, we can use that argument on any cube $Q_{\tilde{r}}(\tilde{x}) \subset Q_{5/8} \cap \{x_1 > 0\}$ to conclude that $\nabla^2 v_k$ and w_k converge to $\nabla^2 v$ strongly in $L^2(Q_{\tilde{r}/2})$. Since we can do this for any such cube, we conclude that $\nabla^2 v_k$ and w_k converge to $\nabla^2 v$ strongly in $L^2(Q_{\tilde{r}/2})$.

Because $u_{h_k} = 0$ in $Q_{5/8} \cap \{x_1 < -3h_k\}$, we also have that $\nabla^2 v_k$ and w_k converge to 0 strongly in $L^2_{loc}(Q_{5/8} \cap \{x_1 < 0\})$. In summary, we have proved that $\nabla^2 v_k$ and w_k converge to $\nabla^2 v$ strongly in $L^2_{loc}(Q_{5/8} \setminus \{x_1 = 0\})$.

We still have to deal with $\{x_1 = 0\}$, and for this we need a new idea.

Step 2: Nonconcentration at the boundary

We claim that for any $y \in Q_{1/2} \cap \{x_1 = 0\}$ we have

$$\lim_{\tilde{r}\to 0} \limsup_{k\to\infty} \left\| \nabla_{h_k}^2 u_{h_k} \right\|_{L^2(Q_{\tilde{r}}(y))} = 0.$$
(2.5.11)

To see this, let $\tilde{r} > 0$. For h_k small enough Lemma 2.3.1 and Proposition 2.4.2 imply that

$$\left\|\nabla_{h_{k}}^{2}u_{h_{k}}\right\|_{L^{2}(Q_{\tilde{r}}(y))}^{2} \leq \frac{C}{\tilde{r}^{2}}\left\|\nabla_{h_{k}}u_{h_{k}}\right\|_{L^{2}(Q_{2\tilde{r}}(y))}^{2} + \frac{C}{\tilde{r}^{4}}\left\|u_{h_{k}}\right\|_{L^{2}(Q_{2\tilde{r}}(y))}^{2}$$
$$\leq \frac{C}{\tilde{r}^2} \left\| \nabla v_k \right\|_{L^2(Q_{4\tilde{r}}(y))}^2 + \frac{C}{\tilde{r}^4} \left\| v_k \right\|_{L^2(Q_{4\tilde{r}}(y))}^2$$

Now v_k converges to v weakly in $W^{2,2}(Q_{3/4})$, so v_k and ∇v_k converge strongly in $L^2(Q_{3/4})$. Hence we can pass to the limit in the above inequality and find

$$\limsup_{k\to\infty} \left\| \nabla_{h_k}^2 u_{h_k} \right\|_{L^2(Q_{\bar{r}}(y))}^2 \leq \frac{C}{\tilde{r}^2} \left\| \nabla v \right\|_{L^2(Q_{4\bar{r}}(y))}^2 + \frac{C}{\tilde{r}^4} \left\| v \right\|_{L^2(Q_{4\bar{r}}(y))}^2.$$

Furthermore v is 0 in $Q_{4\tilde{r}}(y) \cap \{x_1 < 0\}$, so we can apply the Poincaré inequality to conclude

$$\limsup_{k \to \infty} \left\| \nabla_{h_k}^2 u_{h_k} \right\|_{L^2(Q_{\bar{r}}(y))}^2 \le C \left\| \nabla^2 v \right\|_{L^2(Q_{4\bar{r}}(y))}^2$$

Now $\nabla^2 v$ is a fixed L^2 -function, so if we pass to the limit $\tilde{r} \to 0$ here, we indeed obtain (2.5.11).

It is easy to see that (2.5.11) together with the fact that $w_k = I_{h_k}^{pc} \nabla_h^2 u_{h_k}$ converges to $\nabla^2 v$ strongly in $L_{loc}^2(Q_{5/8} \setminus \{x_1 = 0\})$ imply that w_k actually converges to $\nabla^2 v$ strongly in $L^2(Q_{1/2})$.

We have for any $y \in Q_{1/2} \cap \{x_1 = 0\}$ and $\tilde{r} > 0$ that

$$\limsup_{k\to\infty} \left\| \nabla^2 v_k \right\|_{L^2(Q_{\bar{r}}(y))} \le C \limsup_{k\to\infty} \left\| \nabla^2_{h_k} u_{h_k} \right\|_{L^2(Q_{2\bar{r}}(y))}$$

and so from (2.5.11) we also conclude

$$\lim_{\tilde{r}\to 0}\limsup_{k\to\infty} \|\nabla^2 v_k\|_{L^2(Q_{\tilde{r}}(y))}=0$$

This in turn implies that also $\nabla^2 v_k$ converges to $\nabla^2 v$ strongly in $L^2(Q_{1/2})$.

Step 3: Conclusion of the argument

We can now continue as in Step 4 of the proof of Lemma 2.5.2: The strong convergence of w_k to $\nabla^2 v$ allows us to conclude from (2.5.10) that

$$\left\|\nabla^2 v - \left[D_1^2 v\right]_{\mathcal{Q}_{\rho,+}} e_1 \otimes e_1\right\|_{L^2(\mathcal{Q}_{\rho,+})}^2 \geq \frac{\rho^{\mathsf{d}+1}}{2}.$$

On the other hand, we have

$$\left\| \nabla^2 v - \left[D_1^2 v \right]_{Q_{3/4,+}} e_1 \otimes e_1 \right\|_{L^2(Q_{3/4,+})}^2 \le C$$

and it is easy to check that we arrive at a contradiction to Lemma 2.5.7 once we choose ρ small enough.

Proof of Lemma 2.5.5. The proof is similar to the first half of the proof of Lemma 2.5.1: One can assume that x = 0, $\nu = e_1$. Then one first proves that, for any $0 < s' \le s \le r$,

$$\begin{split} \left\| \nabla_{h}^{2} u_{h} - \left[D_{-1}^{h} D_{1}^{h} u_{h} \right]_{Q_{s',+}} e_{1} \otimes e_{1} \right\|_{L^{2}(Q_{s',+})}^{2} \\ & \leq C \left(\frac{s'}{s} \right)^{\mathsf{d}+1} \left\| \nabla_{h}^{2} u_{h} - \left[D_{-1}^{h} D_{1}^{h} u_{h} \right]_{Q_{s,+}} e_{1} \otimes e_{1} \right\|_{L^{2}(Q_{s,+})}^{2} \end{split}$$

,

which already looks similar to the claimed estimate. We can again use this with s = r and $s' = \frac{r}{2^{\lambda}}$ or $s' = \frac{r}{2^{\lambda+1}}$ to conclude

$$\left\| \begin{bmatrix} D_{-1}^{h} D_{1}^{h} u_{h} \end{bmatrix}_{Q_{r/2\lambda+1,+}} - \begin{bmatrix} D_{-1}^{h} D_{1}^{h} u_{h} \end{bmatrix}_{Q_{r/2\lambda,+}} \right\| \\ \leq \frac{C}{r^{\frac{d}{2}} 2^{\frac{\lambda}{2}}} \left\| \nabla_{h}^{2} u_{h} - \begin{bmatrix} D_{-1}^{h} D_{1}^{h} u_{h} \end{bmatrix}_{Q_{r,+}} e_{1} \otimes e_{1} \right\|_{L^{2}(Q_{r,+})}$$

Let λ_0 be the largest integer such that $\frac{r}{2^{\lambda_0}} \ge s$. We can apply this estimate with radii $r, \frac{r}{2}, \ldots, \frac{r}{2^{\lambda_0-1}}$ and sum to conclude

$$\left| \left[D_{-1}^{h} D_{1}^{h} u_{h} \right]_{Q_{r/2}\lambda_{0,+}} - \left[D_{-1}^{h} D_{1}^{h} u_{h} \right]_{Q_{r,+}} \right| \leq \frac{C}{r^{\frac{1}{2}}} \left\| \nabla_{h}^{2} u_{h} - \left[D_{-1}^{h} D_{1}^{h} u_{h} \right]_{Q_{r,+}} e_{1} \otimes e_{1} \right\|_{L^{2}(Q_{r,+})}$$

Using all this, we can estimate

$$\begin{split} \left\| \nabla_{h}^{2} u_{h} \right\|_{L^{2}(Q_{s,+})}^{2} &\leq \left\| \nabla_{h}^{2} u_{h} \right\|_{L^{2}(Q_{r/2}\lambda_{0,+})}^{2} \\ &\leq 2 \left\| \nabla_{h}^{2} u_{h} - \left[D_{-1}^{h} D_{1}^{h} u_{h} \right]_{Q_{r/2}\lambda_{0,+}} e_{1} \otimes e_{1} \right\|_{L^{2}(Q_{r/2}\lambda_{0,+})}^{2} \\ &\quad + 2 \left\| \left[D_{-1}^{h} D_{1}^{h} u_{h} \right]_{Q_{r/2}\lambda_{0,+}} \right\|_{L^{2}(Q_{r/2}\lambda_{0,+})}^{2} \\ &\leq \left(\frac{C}{2^{\lambda_{0}(d+1)}} + \frac{C}{2^{\lambda_{0}n}} \right) \left\| \nabla_{h}^{2} u_{h} - \left[D_{-1}^{h} D_{1}^{h} u_{h} \right]_{Q_{r,+}} e_{1} \otimes e_{1} \right\|_{L^{2}(Q_{r,+})}^{2} \\ &\quad + \frac{C}{2^{\lambda_{0}n}} \left\| \left[D_{-1}^{h} D_{1}^{h} u_{h} \right]_{Q_{r,+}} \right\|_{L^{2}(Q_{r,+})}^{2} \\ &\leq \frac{C}{2^{\lambda_{0}n}} \left\| \nabla_{h}^{2} u_{h} \right\|_{L^{2}(Q_{r,+})}^{2} , \end{split}$$

which implies

$$\begin{aligned} \left\| \nabla_{h}^{2} u_{h} \right\|_{L^{2}(Q_{s,+})}^{2} &\leq C \left(\frac{r}{s} \right)^{\mathsf{d}} \left\| \nabla_{h}^{2} u_{h} \right\|_{L^{2}(Q_{r,+})}^{2} \\ &\leq C \left(\frac{r}{s} \right)^{\mathsf{d}} \left\| \nabla_{h}^{2} u_{h} \right\|_{L^{2}(Q_{r})}^{2}. \end{aligned}$$
(2.5.12)

Now by the same argument as in Step 1 of the proof of Lemma 2.5.6 we have

$$\left\| \nabla_{h}^{2} u_{h} \right\|_{L^{2}(Q_{s})}^{2} \leq 2 \left\| \nabla_{h}^{2} u_{h} \right\|_{L^{2}(Q_{s+})}^{2}$$

Combining this with (2.5.12) yields the result.

2.5.3 Edges and vertices

It remains to prove the analogue of Lemma 2.5.5 near edges (in 3D) and vertices (in 2D and 3D). The actual compactness argument requires no new idea, so we will only give a very brief sketch of the proofs. However, this time the continuous estimate require a bit more work, so we will go into detail there. Let us first state the two results:

Lemma 2.5.8. Let $u_h: (h\mathbb{Z})^3 \to \mathbb{R}$, let $x \in (h\mathbb{Z})^3$, r > 0, $v_1, v_2 \in \{e_1, -e_1, \dots, e_3, -e_3\}$ such that $v_1 \neq \pm v_2$. Suppose that $u_h(y) = 0$ for all $y \in Q_r^h(x)$ such that $(y - x) \cdot v_1 \leq 0$ or $(y - x) \cdot v_2 \leq 0$, and $\Delta_h^2 u_h(y) = 0$ for all $y \in Q_{r-h}^h(x)$ such that $(y - x) \cdot v_1 > 0$ and $(y - x) \cdot v_2 > 0$. Then, for any $s \leq r$,

$$\|\nabla_h^2 u_h\|_{L^2(Q_s(x))} \le C \left(\frac{s}{r}\right)^{\frac{3}{2}} \|\nabla_h^2 u_h\|_{L^2(Q_r(x))}.$$

Lemma 2.5.9. Let d = 2 or d = 3, $u_h: (h\mathbb{Z})^d \to \mathbb{R}$, let $x \in (h\mathbb{Z})^d$, r > 0, $v_i \in \{e_i, -e_i\}$ for $i \in \{1, ..., n\}$. Suppose that $u_h(y) = 0$ for all $y \in Q_r^h(x)$ such that $(y - x) \cdot v_i \leq 0$ for at least one i, and $\Delta_h^2 u_h(y) = 0$ for all $y \in Q_{r-h}^h(x)$ such that $(y - x) \cdot v_i > 0$ for all i. Then, for any $s \leq r$,

$$\|\nabla_h^2 u_h\|_{L^2(Q_s(x))} \le C\left(\frac{s}{r}\right)^{\frac{d}{2}} \|\nabla_h^2 u_h\|_{L^2(Q_r(x))}.$$

Proof of Lemma 2.5.8 and Lemma 2.5.9. This follows easily from the following two lemmata. \Box

Lemma 2.5.10. There are constants $M \in \mathbb{N}$, $0 < \rho < \frac{1}{2}$ with the following property: let $u_h: (h\mathbb{Z})^3 \to \mathbb{R}$, r > 0, such that $u_h(y) = 0$ for all $y \in Q_r^h$ such that $y_1 \le 0$ or $y_2 \le 0$, and $\Delta_h^2 u_h(y) = 0$ for all $y \in Q_{r-h}^h(x)$ such that $y_1 > 0$ and $y_2 > 0$. Then we have that

$$\left\| \nabla_{h}^{2} u_{h} \right\|_{L^{2}(Q_{\rho r})}^{2} \leq \rho^{\mathsf{d}} \left\| \nabla_{h}^{2} u_{h} \right\|_{L^{2}(Q_{r})}^{2}$$

Lemma 2.5.11. There are constants $M \in \mathbb{N}$, $0 < \rho < \frac{1}{2}$ with the following property: let d = 2 or d = 3, $u_h: (h\mathbb{Z})^d \to \mathbb{R}$, r > 0, such that $u_h(y) = 0$ for all $y \in Q_r^h$ such that $y_i \le 0$ for at least one $i \in \{1, ..., n\}$, and $\Delta_h^2 u_h(y) = 0$ for all $y \in Q_{r-h}^h$ such that $y_i > 0$ for all i. Assume that $\rho r \ge Mh$. Then we have that

$$\|\nabla_h^2 u_h\|_{L^2(Q_{\rho r})}^2 \le \rho^d \|\nabla_h^2 u_h\|_{L^2(Q_r)}^2.$$

We will deduce these two lemmata from the following continuous estimates. D_{ν} denotes the derivative in normal direction.

Lemma 2.5.12. There is a constant $\theta > 0$ with the following property: let d = 3, $0 < s \leq \frac{r}{2}$, $u \in W^{2,2}(Q_{r,++})$, where $Q_{r,++} = Q_r \cap \{x_1 > 0, x_2 > 0\}$. Assume that $\Delta^2 u = 0$ weakly in $Q_{r,++}$ and that u = 0, $D_v u = 0$ on $\partial Q_{r,++} \cap \{x_1 = 0 \lor x_2 = 0\}$ in the sense of traces. Assume that $\rho r \geq Mh$. Then we have

$$\left\| \nabla^2 u \right\|_{L^2(Q_{s,++})}^2 \le C \left(\frac{s}{r}\right)^{3+\theta} \left\| \nabla^2 u \right\|_{L^2(Q_{r,++})}^2$$

Lemma 2.5.13. There is a constant $\theta > 0$ with the following property: let $0 < s \leq \frac{r}{2}$, $u \in W^{2,2}(Q_{r,d+})$, where $Q_{r,d+} = Q_{r,++} = Q_r \cap \{x_1 > 0, x_2 > 0\}$ if d = 2, and $Q_{r,d+} = Q_{r,+++} = Q_r \cap \{x_1 > 0, x_2 > 0, x_3 > 0\}$ if d = 3. Assume that $\Delta^2 u = 0$ weakly in $Q_{r,d+}$ and that u = 0, $D_v u = 0$ on $\partial Q_{r,d+} \cap \{x_i = 0 \text{ for some } i\}$ in the sense of traces. Then we have

$$\left\|\nabla^2 u\right\|_{L^2(Q_{s,\mathsf{d}+})}^2 \leq C\left(\frac{s}{r}\right)^{\mathsf{d}+\theta} \left\|\nabla^2 u\right\|_{L^2(Q_{r,\mathsf{d}+})}^2.$$

The proof of Lemma 2.5.12 and Lemma 2.5.13 relies heavily on the theory of elliptic equations in domains with singularities. We use results from [KMR97] and [MR10] and refer the reader to these monographs for more background information.

Proof of Lemma 2.5.12. Let $\mathbb{R}^3_{++} = \mathbb{R}^3 \cap \{x_1 > 0, x_2 > 0\}$. For $x \in \mathbb{R}^3_{++}$ write $x = (x', x_3)$.

The statement is trivial if $s \ge \frac{r}{4}$, so assume $s < \frac{r}{4}$. Let $\eta \in C_c^{\infty}(Q_r)$ be a cut-off function that is 1 on $Q_{r/2,++}$ and such that $|\nabla^{\kappa}\eta| \le \frac{C}{r^{\kappa}}$ for $\kappa \le 4$. Then $\eta \Delta^2 u = 0$ in $\partial \mathbb{R}^3_{++}$, and we can calculate (as an identity in the sense of distributions) that

$$\Delta^2(\eta u) = (\Delta^2 \eta)u + 4\nabla \Delta \eta \cdot \nabla u + 2\Delta \eta \Delta u + 4\nabla^2 \eta : \nabla^2 u + 4\nabla \eta \cdot \nabla \Delta u.$$

In order to avoid terms with too many derivatives of *u* we rewrite the last term as

$$\nabla \eta \cdot \nabla \Delta u = \operatorname{div}(\nabla \eta \Delta u) - \Delta \eta \Delta u$$

to obtain

$$\Delta^2(\eta u) = \Delta^2 \eta u + 4\nabla \Delta \eta \cdot \nabla u - 2\Delta \eta \Delta u + 4\nabla^2 \eta : \nabla^2 u + 4\operatorname{div}(\nabla \eta \Delta u) =: f.$$

Because $u \in W^{2,2}(Q_{r,d+})$ with zero boundary values on $\partial \mathbb{R}^3_{++}$, the right-hand side f is an element of $W^{-2,2}(\mathbb{R}^3_{++})$, while ηu is in $W^{2,2}_0(\mathbb{R}^3_{++})$. Hence (cf. [MR10], Theorem 2.5.1) we can represent ηu via the Green's function of \mathbb{R}^3_{++} as

$$(\eta u)(x) = \int_{\mathbb{R}^3_{++}} G(x,\xi) f(\xi) \,\mathrm{d}\xi.$$

For $x \in Q_{s,++} \subset Q_{r/4,++}$ this implies

$$\nabla^2 u(x) = \int_{\mathbb{R}^3_{++}} \nabla^2_x G(x,\xi) f(\xi) \,\mathrm{d}\xi \,.$$

Now *f* is supported in $Q_{r,++} \setminus Q_{r/2,++}$, whereas $x \in Q_{s,++} \subset Q_{r/4,++}$. So a decay estimate for *G* will directly lead to a pointwise estimate for $\nabla^2 u$.

In fact, Theorem 2.5.4 in [MR10] states that if $|x - \xi| \ge \min(|x'|, |\xi'|)$ we have, for every $\varepsilon > 0$,

$$|D_{x'}^{\alpha} D_{x_{3}}^{j} D_{\xi'}^{\beta} D_{\xi_{3}}^{k} G(x,\xi)| \leq C_{\varepsilon} \frac{|x'|^{1+\delta_{+}-|\alpha|-\varepsilon}|\xi'|^{1+\delta_{-}-|\beta|-\varepsilon}}{|x-\xi|^{1+\delta_{+}+\delta_{-}+j+k-2\varepsilon}}.$$
(2.5.13)

Here δ_+ and δ_- are certain real parameters defined in terms of eigenvalue problems related to the Bilaplacian (see [MR10, Section 2.4] for the precise definition). According to [MR10, Section 4.3] we have that $\delta_+ = \delta_- \approx 2.73959$. In particular, $\delta_\pm > 1$, so we can choose $\theta > 0$ such that $1 + \frac{\theta}{2} < \delta_{\pm}$. Then let $\varepsilon = \delta_{\pm} - 1 - \frac{\theta}{2} > 0$.

We are interested in the case where $x \in Q_{s,++}$, $\xi \in Q_{r,++} \setminus Q_{r/2,++}$. In that case the inequality $|x - \xi| \ge \min(|x'|, |\xi'|)$ certainly holds, and we can estimate $|x'| \le s$, $|\xi'| \le r$, $|x - \xi| \ge \frac{r}{4}$, so that (2.5.13) turns into

$$\begin{aligned} |D_{x'}^{\alpha} D_{x_3}^j D_{\xi'}^{\beta} D_{\xi_3}^k G(x,\xi)| &\leq C_{\varepsilon} s^{1+\delta_+ - |\alpha| - \varepsilon} r^{\varepsilon - \delta_+ - |\beta| - j - k} \\ &= C \frac{s^{2+\frac{\theta}{2} - |\alpha|}}{r^{1+\frac{\theta}{2} + |\beta| + j + k}}. \end{aligned}$$

This estimate is sharp enough to allow us to estimate the terms of f. For example we can calculate using the Poincaré and Hölder inequality that

$$\left| \int_{\mathbb{R}^3_{++}} \nabla_x^2 G(x,\xi) \Delta^2 \eta(\xi) u(\xi) \, \mathrm{d}\xi \right| \le C \int_{Q_{r,++}} \frac{s^{2+\frac{\theta}{2}-2}}{r^{1+\frac{\theta}{2}+0+0+0}} \frac{1}{r^4} \left| u(\xi) \right| \, \mathrm{d}\xi$$

$$= C \frac{s^{\frac{\theta}{2}}}{r^{5+\frac{\theta}{2}}} \int_{Q_{r,++}} |u| \, d\xi$$

$$\leq C \frac{s^{\frac{\theta}{2}}}{r^{5+\frac{\theta}{2}}} r^{2} r^{\frac{3}{2}} \left(\int_{Q_{r,++}} |\nabla^{2}u|^{2} \, d\xi \right)^{\frac{1}{2}}$$

$$\leq C \frac{s^{\frac{\theta}{2}}}{r^{\frac{3}{2}+\frac{\theta}{2}}} \left(\int_{\mathbb{R}^{3}_{++}} |\nabla^{2}u|^{2} \, d\xi \right)^{\frac{1}{2}}$$

and that

$$\begin{split} \left| \int_{\mathbb{R}^{3}_{++}} \nabla_{x}^{2} G(x,\xi) \operatorname{div}(\nabla \eta \Delta u)(\xi) \, \mathrm{d}\xi \right| &= \left| \int_{\mathbb{R}^{3}_{++}} \nabla_{x}^{2} \nabla_{\xi} G(x,\xi) \cdot \nabla \eta(\xi) \Delta u(\xi) \, \mathrm{d}\xi \right| \\ &\leq C \int_{Q_{r,++}} \frac{s^{\frac{\theta}{2}}}{r^{2+\frac{\theta}{2}}} \frac{1}{r} \left| \Delta u(\xi) \right| \, \mathrm{d}\xi \\ &\leq C \frac{s^{\frac{\theta}{2}}}{r^{\frac{3}{2}+\frac{\theta}{2}}} \left(\int_{\mathbb{R}^{3}_{++}} \left| \nabla^{2} u \right|^{2} \, \mathrm{d}\xi \right)^{\frac{1}{2}}. \end{split}$$

We can estimate the other terms on *f* analogously. If we integrate the sum of the squares of all these inequalities with respect to *x* we immediately obtain the conclusion. \Box

Proof of Lemma 2.5.13. The proof in the case of a vertex is very similar. One can again deduce the representation

$$\nabla^2 u(x) = \int_{\mathbb{R}^d_{d+}} \nabla^2_x G(x,\xi) f(\xi) \,\mathrm{d}\xi \tag{2.5.14}$$

for $x \in Q_{r/4,d+}$, so that one only needs sharp estimates for the Green's function to complete the argument.

If d = 2, we can use for this purpose Theorem 8.4.8 in combination with Theorem 6.1.2 in [KMR97]. Theorem 8.4.8 gives a Green's function for right-hand sides in L^2 . However, according to Theorem 6.1.2, the solution operator has a continuous extension to right-hand sides in $W^{-2,2}$, so that (2.5.14) holds for this Green's function. Now Theorem 8.4.8 also gives asymptotics for *G* in terms of the eigenvalues of a certain eigenvalue problem. If we stay in the eigenvalue-free strip, this estimate reads

$$|D_x^{\alpha} D_{\xi}^{\beta} G(x,\xi)| \leq C_{\varepsilon} |x|^{1+\delta_{+}-|\alpha|-\varepsilon} |\xi|^{1-\delta_{+}-|\beta|+\varepsilon}$$

where $2|x| \le |\xi|$ and $\varepsilon > 0$ is arbitrary. Using this estimate we can continue as in the proof of Lemma 2.5.12.

The case d = 3 is slightly more complicated. We can use [MR10, Theorem 3.4.5], which states that if $2|x| \le |\xi|$, then for any $\varepsilon > 0$

$$\begin{aligned} |D_x^{\alpha} D_{\xi}^{p} G(x,\xi)| \\ &\leq C_{\varepsilon} |x|^{\Lambda_{+} - |\alpha| - \varepsilon} |\xi|^{1 - \Lambda_{+} - |\beta| + \varepsilon} \prod_{j=1}^{3} \left(\frac{r_j(x)}{|x|} \right)^{1 + \delta_{+} - |\alpha| - \varepsilon} \prod_{k=1}^{3} \left(\frac{r_k(\xi)}{|\xi|} \right)^{1 + \delta_{-} - |\beta| - \varepsilon} \end{aligned}$$

where δ_{\pm} are as before, Λ_+ is another constant defined in terms of a certain eigenvalue problem (see [MR10, Section 3.4] for the precise definition) and $r_j(x)$ denotes the distance of x to the line { $x_j = 0$ }. If we choose $\varepsilon \le \delta_+ - 1 = \delta_- - 1$, then the exponents of the terms

 $\frac{r_k(x)}{|x|}$ and $\frac{r_k(\xi)}{|\xi|}$ are non-negative whenever $|\alpha| \le 2$ and $|\beta| \le 2$. So we obtain under these assumptions

$$|D_x^{\alpha}D_{\xi}^{\beta}G(x,\xi)| \leq C_{\varepsilon}|x|^{\Lambda_+-|\alpha|-\varepsilon}|\xi|^{1-\Lambda_+-|\beta|+\varepsilon}.$$

In [MR10, Section 4.3] it is proved that $\Lambda_+ \geq 3$. This allows us to take $\theta > 0$ such that $2 + \frac{\theta}{2} \leq \Lambda_+$ and $1 + \frac{\theta}{2} < \delta_{\pm}$. By choosing $\varepsilon = \min \left(\Lambda_+ - 2 - \frac{\theta}{2}, \delta_{\pm} - 1 \right)$ we conclude

$$|D_x^{\alpha} D_{\xi}^{\beta} G(x,\xi)| \le C \frac{s^{2+\frac{\theta}{2}-|\alpha|}}{r^{1+\frac{\theta}{2}+|\beta|}}$$

for $|\alpha| \le 2$ and $|\beta| \le 2$. Now we can continue as in the proof of Lemma 2.5.12 (observe that in that proof we only needed estimates for $D_x^{\alpha} D_{\xi}^{\beta} G(x,\xi)$ with $|\alpha| \le 2$ and $|\beta| \le 1$).

Proof of Lemma 2.5.10 *and Lemma* 2.5.11. We follow the proofs of Lemma 2.5.2 and Lemma 2.5.6. The proof is slightly easier than the proof of Lemma 2.5.6 because we no longer need to worry about the subtraction of the averages of u_h . We assume that the claim is wrong for some fixed ρ , and consider a sequence of counterexamples u_{h_k} and their interpolations $v_k = I_{h_k}u_{h_k}$. We can assume that $r_k = 1$ and $\left\| \nabla_{h_k}^2 u_{h_k} \right\|_{L^2(Q_1)} = 1$, and conclude that (v_k) is bounded in $W^{2,2}(Q_{3/4})$, and so a non-relabeled subsequence converges to some v in $W^{2,2}(Q_{3/4})$.

As before we see that $\Delta^2 v = 0$ in $Q_{3/4,+++}$ and $Q_{3/4,d+}$ respectively and that v has 0 boundary values. Also we obtain strong convergence of $\nabla^2 v_k$ and $w_k := I_{pc}^{h_k} \nabla_{h_k}^2 u_{h_k}$ in $L_{loc}^2(Q_{5/8} \setminus \partial Q_{3/4,+++})$ and $L_{loc}^2(Q_{5/8} \setminus \partial Q_{3/4,d+})$, respectively. Now, as in Step 2 of the proof of Lemma 2.5.6, we find that $\nabla_{h_k}^2 u_{h_k}$ does not concentrate at the boundary, so that $\nabla^2 v_k$ and w_k actually converge strongly in $L^2(Q_{1/2})$.

This convergence allows us to pass to the limit in

$$\left\|\nabla_{h_k}^2 u_{h_k}\right\|_{L^2(Q_\rho)}^2 > \rho^{\mathsf{d}}$$

so that we easily arrive at a contradiction to Lemma 2.5.12 or Lemma 2.5.13 once we choose ρ small enough.

2.6 Inner and outer decay estimates for discrete biharmonic functions

2.6.1 Inner estimates

We can now combine the results from the previous section in one general decay estimate for biharmonic functions:

Theorem 2.6.1. Let d = 2 or d = 3, $u_h \in \Phi_h$. Let $x \in \Lambda_h$, r > 0 and suppose that $\Delta_h^2 u_h(y) = 0$ for all $y \in Q_{r-h}(x) \cap \operatorname{int} \Lambda_h^d$. Then, for all $z \in Q_{r/2}^h(x) \cap \Lambda_h^d$,

$$|\nabla_h^2 u_h(z)| \le \frac{C}{r^{\frac{d}{2}}} \|\nabla_h^2 u_h\|_{L^2(Q_r(x))}.$$
(2.6.1)

Observe that $\nabla_h^2 u_h = 0$ is zero in $(h\mathbb{Z})^d \setminus \Lambda_h$. Therefore we could equivalently only integrate over $Q_r(x) \cap (\Lambda_h)_{pc}$ on the right-hand side.

Proof. The proofs for the cases d = 2 and d = 3 are similar, but the latter is somewhat more tedious. Therefore we give the proof for d = 2 in detail and then describe how to adapt it to the case d = 3. So let d = 2.

We first prove the statement in the special case z = x. By rotating and reflecting Λ_h^2 we may assume $x_2 \le x_1 \le \frac{1}{2}$. We may also assume $r \ge \frac{h}{2}$, as otherwise we can replace r by $\frac{h}{2}$ without changing (2.6.1).

Let $x^* = (x_1, 0)$ be a point on $\partial \Lambda_h^2$ closest to x. We consider the three cases $r \leq x_2$, $x_2 < r \leq x_1$ and $r > x_1$.

Case 1: $r \le x_2$

In this case the interior estimate Lemma 2.5.1 applied to $Q_r(x)$ directly implies

$$|\nabla_h^2 u_h(x)| \leq \frac{C}{r} \|\nabla_h^2 u_h\|_{L^2(Q_r(x))}.$$

Case 2: $x_2 < r \le x_1$

Apply first Lemma 2.5.1 to $Q_{x_2+h/2}(x)$ to find

$$|\nabla_h^2 u_h(x)| \leq \frac{C}{x_2 + \frac{h}{2}} \|\nabla_h^2 u_h\|_{L^2(Q_{x_2 + h/2}(x))}.$$

If $r < 3x_2$ then this already implies (2.6.1) once we increase *C* by a factor of 3. If $r \ge 3x_2$ we have $Q_{x_2+h/2}(x) \subset Q_{2x_2+h/2}(x^*) \subset Q_r(x^*) \subset Q_r(x)$ and so, by Lemma 2.5.5,

$$\|\nabla_h^2 u_h\|_{L^2(Q_{x_2+h/2}(x))} \le \|\nabla_h^2 u_h\|_{L^2(Q_{2x_2+h/2}(x^*))} \le C \frac{2x_2+\frac{h}{2}}{r} \|\nabla_h^2 u_h\|_{L^2(Q_r(x^*))}.$$

This together with the previous equation implies (2.6.1).

*Case 3: x*¹ < *r*

As in the previous case we obtain

$$|\nabla_h^2 u_h(x)| \le \frac{C}{x_1 + \frac{h}{2}} \|\nabla_h^2 u_h\|_{L^2(Q_{x_1 + h/2}(x^*))}.$$
(2.6.2)

Now either $r < 3x_1$ and we are done, or we can continue with Lemma 2.5.9 to find

$$\|\nabla_h^2 u_h\|_{L^2(Q_{x_1+h/2}(x^*))} \le \|\nabla_h^2 u_h\|_{L^2(Q_{2x_1+h/2}(0))} \le C \frac{2x_1 + \frac{h}{2}}{r} \|\nabla_h^2 u_h\|_{L^2(Q_r(0))},$$

which in combination with (2.6.2) implies (2.6.1).

This proves (2.6.1) in the case z = x. For general z, it suffices to observe that $Q_{r/2}(z) \subset Q_r(x)$ and apply the statement we have just proved to $Q_{r/2}(z)$.

The proof for d = 3 is analogous. However there is one more case and hence we need one more intermediate step, where we deal with the case of an edge. So one applies Lemmata 2.5.1, 2.5.5, 2.5.8, 2.5.9 in order until one reaches a radius of order *r*. We omit the details. \Box

2.6.2 Outer estimates via duality

Theorem 2.6.1 states that if a discrete function is biharmonic in a subcube $Q_r(x)$ of Λ_h , then we have pointwise control over its second derivatives in a smaller subcube $Q_{r/2}(x)$. Remarkably, a dual statement is also true: If a discrete function is biharmonic outside a subcube $Q_r(x)$ of Λ_h , then we have control over its second derivatives outside of a larger subcube $Q_{2r}(x)$. The following lemma does not claim pointwise control, but only control in L^2 . However we will combine it with Theorem 2.6.1 into Theorem 2.6.3 where we actually obtain pointwise control.

Lemma 2.6.2. Let d = 2 or d = 3, let $u_h \in \Phi_h$. Let $x \in \Lambda_h$, $r \ge d(x)$ and suppose that $\Delta_h^2 u_h(x) = 0$ for all $x \in int \Lambda_h \setminus Q_r(x)$. Then, for all $s \ge r$,

$$\|\nabla_h^2 u_h\|_{L^2(\mathbb{R}^d \setminus Q_s(x))} \le C\left(\frac{r}{s}\right)^{\frac{d}{2}} \|\nabla_h^2 u_h\|_{L^2(\mathbb{R}^d \setminus Q_r(x))}.$$
(2.6.3)

Proof. Consider first the case r < h. Then d(x) = 0, i.e. $x \in \partial \Lambda_h$, and the assumptions imply $\Delta_h^2 u_h = 0$ in int Λ_h , i.e. $u_h = 0$ in int Λ_h by the uniqueness of the Bilaplacian equation. So both sides of (2.6.3) are zero and the inequality holds.

So we can assume $r \ge h$. The statement is trivial in the case that s < 23r, so we can also assume $s \ge 23r$. We can then replace r and s by $\tilde{r} = \lfloor r - \frac{h}{2} \rfloor_h + \frac{3h}{2}$ and $\tilde{s} = \lfloor s - \frac{h}{2} \rfloor_h + \frac{h}{2}$, respectively. It is easy to see that then $\tilde{r} \ge r$, $\tilde{s} \le s$ and $\tilde{s} \ge 11\tilde{r}$, and it suffices to prove the theorem for \tilde{r}, \tilde{s} . So we will directly assume $r, s \in h\mathbb{N} + \frac{h}{2}, s \ge 11r$ and $r \ge \frac{3h}{2}$.

Let $f_h = \nabla_h^2 u_h \chi_{\Lambda_h \setminus Q_s(x)}$, where χ_A is the indicator function of a set A. Let $v_h \in \Phi_h$ be the unique solution of $\Delta_h^2 v_h = \operatorname{div}_{-h} \operatorname{div}_h f_h$. Then, for any $\varphi_h \in \Phi_h$,

$$(\nabla_{h}^{2} v_{h}, \nabla_{h}^{2} \varphi_{h})_{L^{2}(\mathbb{R}^{d})} = (f_{h}, \nabla_{h}^{2} \varphi_{h})_{L^{2}(\mathbb{R}^{d})}.$$
(2.6.4)

Also let ζ_h and η_h be discrete cut-off functions such that ζ_h is 1 on $\Lambda_h \setminus Q_{5r}(x)$, 0 on $Q_{3r}(x) \cap \Lambda_h$, η_h is 1 on $Q_{7r}(x) \cap \Lambda_h$, 0 on $\Lambda_h \setminus Q_{9r}(x)$ and such that $|\nabla_h^{\kappa} \zeta_h| \leq \frac{C}{r^{\kappa}}$ and $|\nabla_h^{\kappa} \eta_h| \leq \frac{C}{r^{\kappa}}$ for $\kappa \leq 2$.

These choices ensure that

$$\nabla_h^2(\zeta_h u_h) = \nabla_h^2 u_h \text{ on the support of } f_h$$
(2.6.5)

and that

$$\eta_h = 1$$
 on the support of $\Delta_h^2(\zeta_h u_h)$. (2.6.6)

Indeed, for example the support of $\Delta_h^2(\zeta_h u_h)$ is contained in $Q_{5r+2h}(x) \setminus Q_{3r-2h}(x) \subset Q_{7r}(x)$. This implies

$$\begin{aligned} \|\nabla_{h}^{2}u_{h}\|_{L^{2}(\mathbb{R}^{d}\setminus Q_{s}(x))}^{2} &= (f_{h}, \nabla_{h}^{2}u_{h})_{L^{2}(\mathbb{R}^{d})} \\ \stackrel{(2.6.5)}{=} (f_{h}, \nabla_{h}^{2}(\zeta_{h}u_{h}))_{L^{2}(\mathbb{R}^{d})} \\ \stackrel{(2.6.4)}{=} (\nabla_{h}^{2}v_{h}, \nabla_{h}^{2}(\zeta_{h}u_{h}))_{L^{2}(\mathbb{R}^{d})} \\ &= (v_{h}, \Delta_{h}^{2}(\zeta_{h}u_{h}))_{L^{2}(\mathbb{R}^{d})} \\ \stackrel{(2.6.6)}{=} (\eta_{h}v_{h}, \Delta_{h}^{2}(\zeta_{h}u_{h}))_{L^{2}(\mathbb{R}^{d})} \\ &= (\nabla_{h}^{2}(\eta_{h}v_{h}), \nabla_{h}^{2}(\zeta_{h}u_{h}))_{L^{2}(\mathbb{R}^{d})} \\ &\leq \|\nabla_{h}^{2}(\eta_{h}v_{h})\|_{L^{2}(\mathbb{R}^{d})} \|\nabla_{h}^{2}(\zeta_{h}u_{h})\|_{L^{2}(\mathbb{R}^{d})} \,. \end{aligned}$$

$$(2.6.7)$$

Now by the product rule

$$\nabla_{h}^{2}(\eta_{h}v_{h}) = \sum_{i,j=1}^{d} D_{-i}^{h} D_{j}^{h} \eta_{h} v_{h} + \tau_{j}^{h} D_{-i}^{h} \eta_{h} D_{j}^{h} v_{h} + \tau_{-i}^{h} D_{j}^{h} \eta_{h} D_{-i}^{h} v_{h} + \tau_{-i}^{h} \tau_{j}^{h} \eta_{h} D_{-i}^{h} D_{j}^{h} v_{h}$$

and so, using the Poincaré inequality² on $Q_{9r}(x)$,

²Here we have used the assumption $r \ge d(x)$ (or rather $7r \ge d(x)$): It ensures that we have zero boundary data somewhere on $Q_{7r}^h(x) \setminus Q_r^h(x)$ so that we can indeed use the Poincaré inequality.

$$\begin{aligned} \|\nabla_{h}^{2}(\eta_{h}v_{h})\|_{L^{2}(\mathbb{R}^{d})} &\leq \frac{C}{r^{2}} \|v_{h}\|_{L^{2}(Q_{9r}(x))} + \frac{C}{r} \|\nabla_{h}v_{h}\|_{L^{2}(Q_{9r}(x))} + C \|\nabla_{h}^{2}v_{h}\|_{L^{2}(Q_{9r}(x))} \\ &\leq C \|\nabla_{h}^{2}v_{h}\|_{L^{2}(Q_{9r}(x))} \,. \end{aligned}$$

$$(2.6.8)$$

Similarly, by the Poincaré inequality on the annulus $Q_{7r}(x) \setminus Q_r(x)$,

$$\begin{split} \|\nabla_{h}^{2}(\zeta_{h}u_{h})\|_{L^{2}(\mathbb{R}^{d})} &\leq \frac{C}{r^{2}} \|u_{h}\|_{L^{2}(Q_{7r}(x)\setminus Q_{r}(x))} + \frac{C}{r} \|\nabla_{h}u_{h}\|_{L^{2}(Q_{7r}(x)\setminus Q_{r}(x))} + C \|\nabla_{h}^{2}u_{h}\|_{L^{2}(\mathbb{R}^{d}\setminus Q_{r}(x))} \\ &\leq C \|\nabla_{h}^{2}u_{h}\|_{L^{2}(\mathbb{R}^{d}\setminus Q_{r}(x))} \,. \end{split}$$

If we plug the last two estimates into (2.6.7) and then use Theorem 2.6.1 for v_h we obtain

$$\begin{split} \|\nabla_{h}^{2}u_{h}\|_{L^{2}(\Lambda_{h}\setminus Q_{s}(x))}^{2} &\leq C \|\nabla_{h}^{2}v_{h}\|_{L^{2}(Q_{9r}(x))} \|\nabla_{h}^{2}u_{h}\|_{L^{2}(\mathbb{R}^{d}\setminus Q_{r}(x))} \\ &\leq C \left(\frac{9r}{s}\right)^{\frac{d}{2}} \|\nabla_{h}^{2}(v_{h})\|_{L^{2}(\mathbb{R}^{d})} \|\nabla_{h}^{2}u_{h}\|_{L^{2}(\mathbb{R}^{d}\setminus Q_{r}(x))} \,. \end{split}$$

This implies (2.6.3) once we use the energy estimate

$$\|\nabla_{h}^{2}v_{h}\|_{L^{2}(\mathbb{R}^{d})} \leq \|f_{h}\|_{L^{2}(\mathbb{R}^{d})} = \|\nabla_{h}^{2}u_{h}\|_{L^{2}(\mathbb{R}^{d}\setminus Q_{s}(x))}.$$

Now we can combine this lemma with Theorem 2.6.1 to obtain a pointwise outer estimate.

Theorem 2.6.3. Let d = 2 or d = 3, let $u_h \in \Phi_h$. Let $x \in \Lambda_h$, r > 0 and suppose that $\Delta_h^2 u_h(x) = 0$ for all $x \in int \Lambda_h \setminus Q_r(x)$.

Then, for all $y \in \Lambda_h \setminus Q_{2r}(x)$ *,*

$$|\nabla_{h}^{2}u_{h}(y)| \leq C \frac{(\max(d(x), r))^{\frac{d}{2}}}{|x - y|^{d}} \|\nabla_{h}^{2}u_{h}\|_{L^{2}(\mathbb{R}^{d} \setminus Q_{r}(x))}.$$
(2.6.9)

Proof. As in the proof of Lemma 2.6.2 we see that d(x) = 0 implies u = 0 everywhere and (2.6.9) holds. So assume $d(x) \ge h$.

Let $y \in \Lambda_h \setminus Q_{2r}(x)$. If $y \in Q_{2d(x)}(x)$ we use Theorem 2.6.1 on $Q_{d(x)}(y) \subset \mathbb{R}^d \setminus Q_{2r}(x)$ to obtain

$$|\nabla_{h}^{2}u_{h}(y)| \leq \frac{C}{d(x)^{\frac{d}{2}}} \|\nabla_{h}^{2}u_{h}\|_{L^{2}(Q_{d(x)}(y))} \leq \frac{C}{d(x)^{\frac{d}{2}}} \|\nabla_{h}^{2}u_{h}\|_{L^{2}(\mathbb{R}^{d}\setminus Q_{2r}(x))},$$

which implies (2.6.9) because $|x - y| \le \sqrt{n} |x - y|_{\infty} \le 2\sqrt{n} d(x)$ and hence $\frac{1}{d(x)} \le 4n \frac{d(x)}{|x - y|^2}$.

If, on the other hand, $y \in \Lambda_h \setminus Q_{2d(x)}(x)$ then we use Theorem 2.6.1 on $Q_{|x-y|_{\infty}/2}(y)$ and then Lemma 2.6.2 as follows:

$$\begin{split} |\nabla_{h}^{2} u_{h}(y)| &\leq \frac{C}{\left(\frac{|x-y|_{\infty}}{2}\right)^{\frac{d}{2}}} \|\nabla_{h}^{2} u_{h}\|_{L^{2}(Q_{|x-y|_{\infty}/2}(y))} \\ &\leq \frac{C}{|x-y|_{\infty}^{\frac{d}{2}}} \|\nabla_{h}^{2} u_{h}\|_{L^{2}(\mathbb{R}^{d} \setminus Q_{|x-y|_{\infty}/2}(x))} \\ &\leq \frac{C}{|x-y|_{\infty}^{\frac{d}{2}}} \left(\frac{\max(d(x),r)}{\frac{|x-y|_{\infty}}{2}}\right)^{\frac{d}{2}} \|\nabla_{h}^{2} u_{h}\|_{L^{2}(\mathbb{R}^{d} \setminus Q_{\max(d(x),r)}(x))} \\ &\leq C \frac{(\max(d(x),r))^{\frac{d}{2}}}{|x-y|_{\infty}^{\frac{d}{2}}} \|\nabla_{h}^{2} u_{h}\|_{L^{2}(\mathbb{R}^{d} \setminus Q_{\max(d(x),r)}(x))}, \end{split}$$

which implies (2.6.9).

2.7 The discrete full-space Green's function

In order to obtain estimates for G_h , we will compare G_h with a Green's function of $(h\mathbb{Z})^d$. In the absence of boundary conditions such a Green's function is not uniquely defined. We will choose a normalization that is best suited for our application. The necessary asymptotics for the Green's function of $(h\mathbb{Z})^d$ have been derived by Mangad [Man67] using Fourier-theoretic methods.

By \mathcal{F} we denote the Fourier transform of tempered distributions (where we use the convention $(\mathcal{F}f)(x) = \int_{\mathbb{R}^d} f(\xi) e^{-2\pi i x \cdot \xi} d\xi$).

Theorem 2.7.1 ([Man67], Section 4). *Let* $d \in \mathbb{N}$. *Define* $F : \mathbb{Z}^d \times \mathbb{Z}^d \to \mathbb{R}$ *by*

$$F(x,y) = \mathcal{F}\left(\frac{V(\xi)}{\left(4\sum_{j=1}^{\mathsf{d}}\sin^2(\pi\xi_j)\right)^2}\right)(x-y)$$

where $V \in C_c^{\infty}([-1,1]^d)$ is chosen such that V = 1 near 0 and $\sum_{z \in \mathbb{Z}} V(x+z) = 1$ for all x, and $\frac{V(\xi)}{\left(4\sum_{j=1}^d \sin^2(\pi\xi_j)^2\right)^2}$ denotes the tempered distribution given by its finite part in the sense of Hadamard (see [Sch66, Chapitre II, §2 and §3]).

Then *F* is a Green's function for Δ_1^2 in the sense that $\Delta_1^2 F(\cdot, y) = \delta_y$. It satisfies the following asymptotic expansion: If d = 2 and z = x - y,

$$F(x,y) = \frac{|z|^2 \log |z|}{8\pi} + \frac{(\gamma - 1 + \log \pi)|z|^2}{8\pi} - \frac{\log |z|}{16\pi} + \frac{4(z_1^4 + z_2^4)}{|z|^4} - 12 \log \pi - 12\gamma - 3 + O\left(\frac{1}{|z|^2}\right)$$

where γ is the Euler-Mascheroni constant, and if d = 3 and z = x - y,

$$F(x,y) = -\frac{|z|}{8\pi} + \frac{z_1^4 + z_2^4 + z_3^4}{64\pi|z|^5} + \frac{1}{64\pi|z|} + O\left(\frac{1}{|z|^3}\right) \,.$$

Let us briefly sketch how to prove this theorem: Observe that $\sigma(\xi) := \left(4\sum_{j=1}^{d} \sin^2(\pi\xi_j)\right)^2$ is the symbol of Δ_1^2 , so that $\Delta_1^2 F(x, y) = \mathcal{F}(V)(x - y)$. On the other hand one easily checks that $\sum_{z \in \mathbb{Z}} V(x + z) = 1$ implies that $\mathcal{F}(V)(m) = \delta_0(m)$ for any $m \in \mathbb{Z}^d$. This proves that Fis a Green's function. To derive the asymptotic expansion, one develops a Laurent series

$$\frac{1}{\sigma(\xi)} = \frac{1}{16\pi^2 |\xi|^4} + \frac{f_{-2}(\xi)}{|\xi|^2} + f_0(\xi) + \dots + o(|\xi|^d).$$

Then one can check using the explicit formulas for the Fourier transforms of $|\xi|^m$ (see [Sch66]) and the Riemann-Lebesgue lemma that

$$\mathcal{F}\left(\frac{V(\xi)}{\sigma(\xi)} - \frac{1}{16\pi^2 |\xi|^4} + \frac{f_{-2}(\xi)}{|\xi|^2} + f_0(\xi) + \cdots\right) = o(|x|^{-\mathsf{d}-N})$$

so it suffices to compute the Fourier transform of $\frac{1}{16\pi^2|\xi|^4} + \frac{f_{-2}(\xi)}{|\xi|^2} + f_0(\xi) + \cdots$. This one can again do explicitly and thereby obtain an asymptotic expansion for *F* up to $O(|x|^d)$. For details we refer to [Man67].

By scaling the lattice we can deduce from this estimates for Green's functions on $(h\mathbb{Z})^d$. We state the estimates that we will need. **Lemma 2.7.2.** Let d = 2 or d = 3, h > 0, $r \ge 4h$. There exists a function $\tilde{G}_h \colon (h\mathbb{Z})^d \times (h\mathbb{Z})^d \to \mathbb{R}$ such that $\Delta_h^2 \tilde{G}_h(\cdot, y) = \delta_{h,y}$ and such that the following estimates are satisfied:

$$|\nabla_{h,y}\tilde{G}_{h}(x,y)| \le Cr^{3-\mathsf{d}}$$
 if $|x-y|_{\infty} \le \frac{r}{2}$, (2.7.1)

$$|\nabla_{h,x}^2 \nabla_{h,y} \tilde{G}_h(x,y)| \le \frac{C}{(|x-y|+h)^{\mathsf{d}-1}} \qquad \text{if } |x-y|_{\infty} \le \frac{r}{2},$$
(2.7.2)

$$|\nabla_{h,x}^2 \nabla_{h,y}^2 \tilde{G}_h(x,y)| \le \frac{C}{(|x-y|+h)^{\mathsf{d}}} \qquad \text{if } |x-y|_{\infty} \le \frac{r}{2}$$
(2.7.3)

and

$$|D_{h,x}^{\alpha}D_{h,y}^{\beta}\tilde{G}_{h}(x,y)| \le Cr^{4-\mathsf{d}-|\alpha|-|\beta|} \qquad \qquad \text{if } \frac{r}{2} \le |x-y|_{\infty} \le r, |\alpha|+|\beta| \le 4.$$
 (2.7.4)

For d = 2 the function \tilde{G}_h depends on *r*, but we will suppress this dependence for ease of notation.

Proof. We begin with the slightly easier case d = 3. The asymptotic expansion in Theorem 2.7.1 easily implies that

$$|D_{1,x}^{\alpha}D_{1,y}^{\beta}F(x,y)| \le C|x-y|^{1-|\alpha|-|\beta|}$$

for $|\alpha| + |\beta| \le 4$ and any x, y with $|x - y| \ge 10$, say (observe that $g = O(|x|^{-3})$ implies $D_{\pm i}^1 g(x) = O(|x|^{-3})$, so we do not need to care about the error term). On the other hand F is finite everywhere, so that

$$|D_{1,x}^{\alpha}D_{1,y}^{\beta}F(x,y)| \le C$$

for $|\alpha| + |\beta| \le 4$ and any x, y with |x - y| < 10. If we combine these two estimates we conclude that we have

$$|D_{1,x}^{\alpha}D_{1,y}^{\beta}F(x,y)| \leq C(|x-y|+1)^{1-|\alpha|-|\beta|}.$$

Now if we set $\tilde{G}_h(x, y) = hF\left(\frac{x}{h}, \frac{y}{h}\right)$ then \tilde{G}_h satisfies

$$|D_{h,x}^{\alpha}D_{h,y}^{\beta}\tilde{G}_{h}(x,y)| \leq C(|x-y|+h)^{1-|\alpha|-|\beta|},$$

which immediately implies the claimed estimates.

If d = 2 we need to take care of the logarithmic terms. So we set

$$\tilde{F}(x,y) = F(x,y) + \frac{|x-y|^2 \log\left(\frac{h}{r}\right)}{8\pi}$$

Then \tilde{F} has the asymptotic expansion

$$\begin{split} \tilde{F}(z) &= \frac{z|^2 \log |z|}{8\pi} + \frac{(\log \left(\frac{h}{r}\right) + \gamma - 1 + \log \pi)|z|^2}{8\pi} - \frac{\log |z|}{16\pi} \\ &+ \frac{4(z_1^4 + z_2^4)}{|z|^4} - 12 \log \pi - 12\gamma - 3 + O\left(\frac{1}{|z|^2}\right) \end{split}$$

and this implies

$$|D_{1,x}^{\alpha}D_{1,y}^{\beta}\tilde{F}(x,y)| \le C|x-y|^{2-|\alpha|-|\beta|} \left(\left| \log|x-y| + \log\left(\frac{h}{r}\right) \right| + 1 \right)$$

for $|\alpha| + |\beta| \le 2$ and any x, y with $|x - y| \ge 10$. Because $D_{1,x}^{\alpha} D_{1,y}^{\beta} \tilde{F}(x, y)$ is bounded by $C\left(1 + \left|\log\left(\frac{h}{r}\right)\right|\right)$ for |x - y| < 10, we conclude

$$|D_{1,x}^{\alpha}D_{1,y}^{\beta}\tilde{F}(x,y)| \leq C(|x-y|+1)^{2-|\alpha|-|\beta|} \left|\log\left(rac{h(|x-y|+1)}{r}
ight)
ight|.$$

We now set $\tilde{G}_h(x, y) = h^2 \tilde{F}\left(\frac{x}{h}, \frac{y}{h}\right)$ and obtain

$$|D_{h,x}^{\alpha}D_{h,y}^{\beta}\tilde{G}_{h}(x,y)| \leq C(|x-y|+h)^{2-|\alpha|-|\beta|} \left|\log\left(\frac{|x-y|+h}{r}\right)\right|.$$

It is easy to check that this implies (2.7.1) and (2.7.4) for $|\alpha| + |\beta| \le 2$. If $|\alpha| + |\beta| \ge 3$ we need to be slightly more careful: Observe that third discrete derivatives of $|x - y|^2$ vanish, so that we actually have

$$|\nabla_{1,x}^{\alpha}\nabla_{1,y}^{\beta}\tilde{F}(x-y)| \leq \frac{C}{|x-y|^{|\alpha|+|\beta|-2}}$$

if $|x - y| \ge 10$ from which we conclude

$$|\nabla_{1,x}^{\alpha}\nabla_{1,y}^{\beta}\tilde{F}(x-y)| \leq \frac{C}{(|x-y|+1)^{|\alpha|+|\beta|-2}}$$

for any *x*, *y*. Recalling that $\tilde{G}_h(x, y) = h^2 \tilde{F}\left(\frac{x}{h}, \frac{y}{h}\right)$ we immediately obtain (2.7.2), (2.7.3) and (2.7.4) for $|\alpha| + |\beta| \ge 3$.

2.8 Proof of the main theorem

We are now able to prove Theorem 2.1.3. We first give the straightforward proof of part ii) and then continue with part i).

2.8.1 Lower bounds for $G_h(x, x)$

The proof is rather short and based on the choice of an appropriate test function.

Proof of Theorem 2.1.3 *ii*). We can assume $d(x) \ge h$, as otherwise d(x) = 0 and hence $G_h(x, x) = 0$. If we test the equation $\Delta_h^2 G_{h,x} = \delta_{h,x}$ with $G_{h,x}$, we find

$$\|\nabla_{h}^{2}G_{h,x}\|_{L^{2}(\mathbb{R}^{d})}^{2} = (\Delta_{h}^{2}G_{h,x}, G_{h,x})_{L^{2}(\mathbb{R}^{d})} = (\delta_{h,x}, G_{h,x})_{L^{2}(\mathbb{R}^{d})} = G_{h}(x, x).$$
(2.8.1)

Now let $\varphi_h \in \Phi_h$. Then testing the equation $\Delta_h^2 G_{h,x} = \delta_{h,x}$ with φ_h and using the Cauchy-Schwarz inequality we find

$$\begin{split} \varphi_h(x) &= (\nabla_h^2 G_{h,x}, \nabla_h^2 \varphi_h)_{L^2(\mathbb{R}^d)} \\ &\leq \|\nabla_h^2 G_{h,x}\|_{L^2(\mathbb{R}^d)} \|\nabla_h^2 \varphi_h\|_{L^2(\mathbb{R}^d)} \\ &= \sqrt{G_h(x,x)} \|\nabla_h^2 \varphi_h\|_{L^2(\mathbb{R}^d)} \,. \end{split}$$

If φ_h is not identically zero this implies

$$G_h(x,x) \ge \frac{(\varphi_h(x))^2}{\|\nabla_h^2 \varphi_h\|_{L^2(\mathbb{R}^d)}^2}$$

and so it remains to find a $\varphi_h(x)$ such that $\frac{\varphi_h(x)}{\|\nabla_h^2 \varphi_h\|_{L^2(\mathbb{R}^d)}} \ge Cd(x)^{2-\frac{d}{2}}$. But this is easy: Take $\varphi_{h,x} \in \Phi_h$ supported in $Q_{d(x)}(x)$ such that $\varphi_{h,x}(x) = 1$ and such that $|\nabla_h^2 \varphi_{h,x}| \le \frac{C}{d(x)^2}$ and extend it by 0 to all of Λ_h^d .

2.8.2 Upper bounds for $G_h(x, y)$

In this section we prove part i) of Theorem 2.1.3.

We begin with a rather weak estimate for $G_h(x, y)$.

Lemma 2.8.1. Let d = 2 or d = 3 and G_h be the Green's function of Λ_h^d . Then we have

$$0 \le G_h(x, x) = \|\nabla_h^2 G_{h, x}\|_{L^2(\mathbb{R}^d)}^2 \le Cd(x)^{4-d}$$
(2.8.2)

for any $x \in \Lambda_h$ *and*

$$|G_h(x,y)| \le Cd(x)^{2-\frac{d}{2}}d(y)^{2-\frac{d}{2}}$$
(2.8.3)

for any $x, y \in \Lambda_h$ *.*

Proof. We first prove (2.8.2). By (2.8.1) we have

$$\|\nabla_h^2 G_{h,x}\|_{L^2(\mathbb{R}^d)}^2 = G_h(x,x).$$
(2.8.4)

If $x \in \partial \Lambda_h$ then $G_h(x, x) = 0$ and (2.8.2) holds. So assume $x \in \text{int } \Lambda_h$, i.e. $d(x) \ge h$. The Sobolev-Poincaré inequality implies that

$$\begin{aligned} G_h(x,x) &\leq \|G_{h,x}\|_{L^{\infty}(Q_{d(x)+h/2}(x))} \leq C \left(d(x) + \frac{h}{2} \right)^{2-\frac{d}{2}} \|\nabla_h^2 G_{h,x}\|_{L^2(Q_{d(x)+h/2}(x))} \\ &\leq C d(x)^{2-\frac{d}{2}} \|\nabla_h^2 G_{h,x}\|_{L^2(Q_{2d(x)+h/2}(x))} \,. \end{aligned}$$

If we combine this estimate with (2.8.4) we find that

$$\|\nabla_{h}^{2}G_{h,x}\|_{L^{2}(\mathbb{R}^{d})}^{2} = G_{h}(x,x) \leq Cd(x)^{2-\frac{d}{2}} \|\nabla_{h}^{2}G_{h,x}\|_{L^{2}(Q_{2d(x)+h/2}(x))} \leq Cd(x)^{2-\frac{d}{2}} \|\nabla_{h}^{2}G_{h,y}\|_{L^{2}(\mathbb{R}^{d})}$$

and hence

$$0 \le G_h(x, x) = \|\nabla_h^2 G_{h, x}\|_{L^2(\mathbb{R}^d)}^2 \le Cd(x)^{4-d}$$

This proves (2.8.2). For (2.8.3), we test $\Delta_h^2 G_{h,x} = \delta_{h,x}$ with $G_{h,y}$ and use the Cauchy-Schwarz inequality to obtain

$$|G_{h}(x,y)| = \left| (\delta_{h,x}, G_{h,y})_{L^{2}(\mathbb{R}^{d})} \right|$$

= $\left| (\nabla_{h}^{2}G_{h,x}, \nabla_{h}^{2}G_{h,y})_{L^{2}(\mathbb{R}^{d})} \right|$
 $\leq ||\nabla_{h}^{2}G_{h,x}||_{L^{2}(\mathbb{R}^{d})} ||\nabla_{h}^{2}G_{h,y}||_{L^{2}(\mathbb{R}^{d})}$
 $\stackrel{(2.8.2)}{\leq} Cd(x)^{2-\frac{d}{2}}d(y)^{2-\frac{d}{2}}.$

The next lemma gives estimates for G_h and its derivatives that are sharp when x and y are far apart. We first prove a pointwise estimate for $\nabla_{h,x}^2 \nabla_{h,y} G_h$ by applying Theorem 2.6.3 to a cut-off version of $\nabla_{h,y} G_{h,y}$. Afterwards we integrate it along suitable paths to deduce the estimates in the lemma.

Lemma 2.8.2. Let d = 2 or d = 3 and G_h be the Green's function of Λ_h^d . If $x, y \in \Lambda_h$ and $|x - y|_{\infty} > \frac{d(y)}{8}$ then

$$G_h(x,y)| \le C \frac{(d(x)+h)^2 (d(y)+h)^2}{|x-y|^{\mathsf{d}}}, \qquad (2.8.5)$$

$$|\nabla_{h,x}G_h(x,y)| \le C \frac{(d(x)+h)(d(y)+h)^2}{|x-y|^{\mathsf{d}}},$$
(2.8.6)

$$|\nabla_{h,x}^2 G_h(x,y)| \le C \frac{(d(y)+h)^2}{|x-y|^{\mathsf{d}}},$$
(2.8.7)

$$|\nabla_{h,x}\nabla_{h,y}G_h(x,y)| \le C \frac{(d(x)+h)(d(y)+h)}{|x-y|^{\mathsf{d}}}.$$
(2.8.8)

Proof.

Step 1: Pointwise estimate for $\nabla_{h,x}^2 \nabla_{h,y} G_h(x,y)$ We claim that if $x, y \in \Lambda_h$ and $|x - y|_{\infty} > \frac{d(y)}{8}$ then

$$|\nabla_{h,x}^2 \nabla_{h,y} G_h(x,y)| \le C \frac{d(y)+h}{|x-y|^{\mathsf{d}}}.$$
(2.8.9)

In the following all derivatives will be with respect to *x* unless we mark them with a sub- or superscript *y*.

If d(y) < 160h we can use a trivial estimate: From Lemma 2.8.1 we know

$$\|\nabla_h^2 G_{h,y'}\|_{L^2(\mathbb{R}^d)} \le Cd(y')^{2-\frac{d}{2}} \le Ch^{2-\frac{d}{2}}$$

if $|y' - y|_{\infty} \le h$. If we now use

$$|D_i^h f_h(y)|^2 = \left(\frac{1}{h}(f_h(y+e_i) - f_h(y))\right)^2 \le \frac{2}{h^2}(f_h(y+e_i)^2 + f_h(y)^2)$$

with $f_h = \nabla_h^2 G_h$ we get that

$$\|\nabla_{h}^{2}D_{i}^{h,y}G_{h}\|_{L^{2}(\mathbb{R}^{d})}^{2} \leq \frac{2}{h^{2}}\left(\|\nabla_{h}^{2}\tau_{i}^{h,y}G_{h}\|_{L^{2}(\mathbb{R}^{d})}^{2} + \|\nabla_{h}^{2}G_{h}\|_{L^{2}(\mathbb{R}^{d})}^{2}\right) \leq Ch^{2-d},$$

i.e.

$$\|\nabla_h^2 D_i^{h,y} G_h\|_{L^2(\mathbb{R}^d)} \le Ch^{1-\frac{d}{2}}.$$

Then Theorem 2.6.3 with r = h implies

$$|\nabla_h^2 D_i^{h,y} G_h(x,y)| \le C \frac{\max(d(y),h)^{\frac{d}{2}}}{|x-y|^{d}} \|\nabla_h^2 D_i^{h,y} G_h\|_{L^2(\mathbb{R}^d)} \le C \frac{h^{\frac{d}{2}}}{|x-y|^{d}} h^{1-\frac{d}{2}} = C \frac{h}{|x-y|^{d}},$$

which implies (2.8.9) if we choose C there large enough.

So assume $d(y) \ge 160h$. Let η_h be a discrete cut-off function that is 1 on $Q_{d(y)/32+2h}$, 0 on $(h\mathbb{Z})^d \setminus Q_{d(y)/16-2h}(x)$, and such that $|\nabla^{\kappa}\eta_h| \le \frac{C}{d(y)^{\kappa}}$ for $\kappa \le 2$. Let $H_h(x,y) = G_h(x,y) - \eta_h(x)\tilde{G}_h(x,y)$, where \tilde{G}_h is the function from Lemma 2.7.2 with $r = \frac{d(y)}{16}$. We write $H_{h,y}$ for $H_h(\cdot, y)$.

Then, for $i \in \{1, ..., n\}$, $D_i^{h,y} H_{h,y} \in \Phi_h$. Also, the singularities near y cancel out, so that $\Delta_h^2 D_i^{h,y} H_{h,y} = 0$ in $Q_{d(y)/32}(y)$ and in int $\Lambda_h \setminus Q_{d(y)/16}(y)$.

Next, we want to bound $\|\nabla_h^2 D_i^{h,y} H_{h,y}\|_{L^2(\mathbb{R}^d)}$. To do so, we introduce another cut-off function ζ_h that is 1 on int $\Lambda_h \setminus Q_{d(y)/32}(y)$, 0 on $Q_{d(y)/64}(y)$ and such that $|\nabla^{\kappa} \zeta_h| \leq \frac{C}{d(y)^{\kappa}}$ for $\kappa \leq 2$. Then we have that

$$\Delta_h^2 D_i^{h,y} H_{h,y} = \zeta_h \Delta_h^2 D_i^{h,y} H_{h,y} = -\zeta_h \Delta_h^2 D_i^{h,y} \left(\eta_h \tilde{G}_{h,y} \right) = -\zeta_h \Delta_h^2 \left(\eta_h D_i^{h,y} \tilde{G}_{h,y} \right)$$

where we have used that η_h does not depend on *y*. Thus

$$\begin{split} \|\nabla_{h}^{2}D_{i}^{h,y}H_{h,y}\|_{L^{2}(\mathbb{R}^{d})}^{2} &= (\Delta_{h}^{2}D_{i}^{h,y}H_{h,y}, D_{i}^{h,y}H_{h,y})_{L^{2}(\mathbb{R}^{d})} \\ &= -(\zeta_{h}\Delta_{h}^{2}(\eta_{h}D_{i}^{h,y}\tilde{G}_{h,y}), D_{i}^{h,y}H_{h,y})_{L^{2}(\mathbb{R}^{d})} \\ &= -(\Delta_{h}^{2}(\eta_{h}D_{i}^{h,y}\tilde{G}_{h,y}), \zeta_{h}D_{i}^{h,y}H_{h,y})_{L^{2}(\mathbb{R}^{d})} \\ &= -(\nabla_{h}^{2}(\eta_{h}D_{i}^{h,y}\tilde{G}_{h,y}), \nabla_{h}^{2}(\zeta_{h}D_{i}^{h,y}H_{h,y}))_{L^{2}(\mathbb{R}^{d})} \\ &\leq \|\nabla_{h}^{2}(\eta_{h}D_{i}^{h,y}\tilde{G}_{h,y})\|_{L^{2}(\mathbb{R}^{d})} \|\nabla_{h}^{2}(\zeta_{h}D_{i}^{h,y}H_{h,y})\|_{L^{2}(\mathbb{R}^{d})} \,. \end{split}$$
(2.8.10)

If we use the pointwise estimates for $\tilde{G}_{h,y}$ from Lemma 2.7.2, we conclude

$$|\nabla_h^2(\eta_h D_i^{h,y} \tilde{G}_{h,y})| \le C d(y)^{1-\mathsf{d}}$$

and hence

$$\|\nabla_{h}^{2}(\eta_{h}D_{i}^{h,y}\tilde{G}_{h,y})\|_{L^{2}(\mathbb{R}^{d})} \leq Cd(y)^{1-\frac{d}{2}}$$

Furthermore, as in (2.6.8), the Poincaré inequality on $Q_{d(y)+h/2}(y)$ and the pointwise estimates for ζ_h imply that

$$\begin{split} \|\nabla_{h}^{2}(\zeta_{h}D_{i}^{h,y}H_{h,y})\|_{L^{2}(\mathbb{R}^{d})} \\ &\leq \frac{C}{d(y)^{2}}\|D_{i}^{h,y}H_{h,y}\|_{L^{2}(Q_{d(y)+h/2}(y))} + \frac{C}{d(y)}\|\nabla_{h}D_{i}^{h,y}H_{h,y}\|_{L^{2}(Q_{d(y)+h/2}(y))} \\ &+ \|\nabla_{h}^{2}D_{i}^{h,y}H_{h,y}\|_{L^{2}(\mathbb{R}^{d})} \\ &\leq C\|\nabla_{h}^{2}D_{i}^{h,y}H_{h,y}\|_{L^{2}(\mathbb{R}^{d})} \,. \end{split}$$

If we combine the last two estimates with (2.8.10) we conclude that

$$\|\nabla_h^2 D_i^{h,y} H_{h,y}\|_{L^2(\mathbb{R}^d)} \le C d(y)^{1-\frac{d}{2}}.$$

We recall that $\Delta_h^2 H_h = 0$ in $\operatorname{int} \Lambda_h \setminus Q_{d(y)/16}$ and use Theorem 2.6.3 to find that, for $x \in \Lambda_h \setminus Q_{d(y)/8}(y)$,

$$|\nabla_h^2 D_i^{h,y} H_h(x)| \le C \frac{d(y)^{\frac{d}{2}}}{|x-y|^{d}} \|\nabla_h^2 D_i^{h,y} H_h\|_{L^2(\mathbb{R}^d)} \le C \frac{d(y)^{\frac{d}{2}}}{|x-y|^{d}} d(y)^{1-\frac{d}{2}} = C \frac{d(y)}{|x-y|^{d}}.$$

This implies (2.8.9) because $D_i^{h,y}H_{h,y}$ is equal to $D_i^{h,y}G_{h,y}$ in $\Lambda_h \setminus Q_{d(y)/16}(y)$ and therefore $\nabla_h^2 D_i^{h,y}H_{h,y}$ is equal to $\nabla_h^2 D_i^{h,y}G_{h,y}$ in $\Lambda_h \setminus Q_{d(y)/8}(y)$. Step 2: Proof of (2.8.8)

We can obtain (2.8.8) by integrating (2.8.9) along a well-chosen path in x. Let $(x^{(k)})_{k=0}^{L}$ be a path of length Lh from $x^{(0)} = x$ to $x^{(L)} \in (h\mathbb{Z})^{d} \setminus \Lambda_{h}$ such that $|x^{(k+1)} - x^{(k)}|_{\infty} = h$, $|x^{(k)} - y| \ge |x - y|_{\infty}$ for all k, and $L \le 2(d(x) + h)$. To construct such a path begin with the straight path from x to a closest point $x^{*} \in (h\mathbb{Z})^{d} \setminus \Lambda_{h}$ (which will have length d(x) + h).

If this path does not intersect $Q_{|x-y|_{\infty}-h}^{h}(y)$, we are done. Else we modify the path by taking a (shortest-possible) detour around $Q_{|x-y|_{\infty}-h}^{h}(y)$. This detour lengthens the path by at most $|x-y|_{\infty}$, and it is easy to check that if it is necessary then $y \in Q_{d(x)}^{h}(x)$, so that $|x-y|_{\infty} \leq d(x)$, and our path has length at most $d(x) + h + |x-y|_{\infty} \leq 2(d(x) + h)$.

Now, by (2.8.9),

$$|\nabla_{h,x}^2 \nabla_{h,y} G_h(x^{(k)},y)| \le C \frac{d(y)+h}{(|x^{(k)}-y|+h)^{\mathsf{d}}} \le C \frac{d(y)+h}{(|x^{(k)}-y|_{\infty}+h)^{\mathsf{d}}} \le C \frac{d(y)+h}{(|x-y|_{\infty}+h)^{\mathsf{d}}}.$$

Now we can perform discrete integration along $(x^{(k)})_{k=0}^L$. Note that $\nabla_{h,x}\nabla_{h,y}G_h(x^{(L)},y) = 0$ and so

$$\begin{split} |\nabla_{h,x} \nabla_{h,y} G_h(x,y)| &\leq \sum_{k=0}^{L-1} |\nabla_{h,x} \nabla_{h,y} G_h(x^{(k+1)},y) - \nabla_{h,x} \nabla_{h,y} G_h(x^{(k)},y)| \\ &\leq \sum_{k=0}^{L-1} h |\nabla_{h,x}^2 \nabla_{h,y} G_h(x^{(k)},y)| \\ &\leq L \frac{d(y) + h}{(|x-y|_{\infty} + h)^{\mathsf{d}}}, \end{split}$$

which implies (2.8.8).

Step 3: Proof of (2.8.7)

We proceed as in the previous step with the only difference that this time we integrate in *y* along a path that avoids *x*. Let $(y^{(k)})_{k=0}^{L}$ be a path of length *Lh* from $y^{(0)} = y$ to $y^{(L)} \in (h\mathbb{Z})^{d} \setminus \Lambda_{h}$ such that $|y^{(k+1)} - y^{(k)}|_{\infty} = h$, $|y^{(k)} - x|_{\infty} \ge |y - x|_{\infty}$ for all *k*, and $L \le 2(d(y) + h)$. If we construct this path as in the previous step, we can in addition ensure that $d(y^{(k)}) \le d(y)$ for all *k* (then in particular $|y^{(k)} - x|_{\infty} \ge \frac{d(y^{(k)})}{8}$, so that (2.8.9) is applicable for all $y^{(k)}$).

Now by (2.8.9)

$$|\nabla_{h,x}^2 \nabla_{h,y} G_h(x, y^{(k)})| \le C \frac{d(y^{(k)}) + h}{(|x - y^{(k)}| + h)^{\mathsf{d}}} \le C \frac{d(y^{(k)}) + h}{(|x - y^{(k)}|_{\infty} + h)^{\mathsf{d}}} \le C \frac{d(y) + h}{(|x - y|_{\infty} + h)^{\mathsf{d}}}$$

and if we integrate this along $(y^{(k)})_{k=0}^L$, we obtain (2.8.7).

Step 4: Proof of (2.8.6) and (2.8.5)

We proceed as in the previous two steps. If we integrate (2.8.7) along a path $(x^{(k)})_{k=0}^{L}$ that avoids *y* once, we obtain (2.8.6), and if we integrate once more, we obtain (2.8.5).

Now we complement this lemma with an estimate when *x* and *y* are close:

Lemma 2.8.3. Let d = 2 or d = 3 and G_h be the Green's function of Λ_h^d . If $x, y \in \Lambda_h$ and $|x - y|_{\infty} \leq \frac{d(y)}{8}$ then

$$|G_h(x,y)| \le C(d(x)+h)^{2-\frac{d}{2}}(d(y)+h)^{2-\frac{d}{2}},$$
(2.8.11)

$$|\nabla_{h,x}G_h(x,y)| \le C(d(y)+h)^{3-d}$$
, (2.8.12)

$$|\nabla_{h,x}^{2}G_{h}(x,y)| \leq \begin{cases} C\log\left(\frac{d(y)+h}{|x-y|+h}\right) & d=2\\ \frac{C}{|x-y|+h} & d=3 \end{cases}$$
 (2.8.13)

$$|\nabla_{h,x}\nabla_{h,y}G_{h}(x,y)| \leq \begin{cases} C\log\left(\frac{(d(x)+h)(d(y)+h)}{(|x-y|+h)^{2}}\right) & d=2\\ \frac{C}{|x-y|+h} & d=3 \end{cases}.$$
 (2.8.14)

Proof. Step 1: Pointwise estimate for $\nabla_{h,x}^2 \nabla_{h,y} G_h(x,y)$ We claim that if $x, y \in \Lambda_h$ and $|x - y|_{\infty} \leq \frac{d(y)}{4}$ then

$$|\nabla_{h,x}^2 \nabla_{h,y} G_h(x,y)| \le \frac{C}{(|x-y|+h)^{\mathsf{d}-1}}.$$
(2.8.15)

The fact that we prove this for $|x - y|_{\infty} \le \frac{d(y)}{4}$ will give us a bit of space to wiggle around in the following steps where we integrate (2.8.15). The proof of (2.8.15) is similar to the proof of (2.8.9). The main difference is that this time we choose the cut-off function further away from the singularity.

If d(y) < 10h we can again use a trivial estimate: By Lemma 2.8.1, $G_h(x', y')$ is bounded by $Cd(x')^{2-\frac{d}{2}}d(y')^{2-\frac{d}{2}} \le Ch^{4-d}$ if $|x'-x|_{\infty} \le h$ and $|y'-y|_{\infty} \le h$, so that

$$|\nabla_{h,x}^2 \nabla_{h,y} G_h(x,y)| \le C \frac{1}{h^3} h^{4-\mathsf{d}} = C h^{1-\mathsf{d}}.$$

Therefore (2.8.15) holds if we choose C sufficiently large.

So assume that $d(y) \ge 10h$. Let η_h be a discrete cut-off function that is 1 on $Q_{d(y)/2+2h}(y)$ and 0 on $(h\mathbb{Z})^d \setminus Q_{d(y)-2h}(y)$ and such that $|\nabla^{\kappa}\eta_h| \le \frac{C}{d(x)^k}$ for $\kappa \le 2$ and let $H_h(x,y) = G_h(x,y) - \eta_h(x)\tilde{G}_h(x,y)$, where \tilde{G}_h is the function from Lemma 2.7.2 with r = d(y).

Then, for $i \in \{1, ..., n\}$, $D_i^{h,y} H_{h,y} \in \Phi_h$ and $\Delta_h^2 D_i^{h,y} H_{h,y} = 0$ in $Q_{d(y)/2}(y)$ and in int $\Lambda_h \setminus Q_{d(y)}(y)$. We can estimate $\|\nabla_h^2 D_i^{h,y} H_{h,y}\|_{L^2(\mathbb{R}^d)}$ just as in Step 1 of the proof of Lemma 2.8.2 and obtain that

$$\|\nabla_h^2 D_i^{h,y} H_{h,y}\|_{L^2(\mathbb{R}^d)} \le Cd(y)^{1-\frac{d}{2}}.$$
(2.8.16)

Now recall that H_h is biharmonic in $Q_{d(y)/2}(y)$. So Theorem 2.6.1 implies for $x \in Q_{d(y)/4}^h(y)$

$$|\nabla_{h}^{2}D_{i}^{h,y}H_{h,y}(x)| \leq \frac{C}{d(y)^{\frac{d}{2}}} \|\nabla_{h}^{2}D_{i}^{h,y}H_{h,y}\|_{L^{2}(\mathbb{R}^{d})} \leq Cd(y)^{1-d}$$

Because $\nabla_h^2 D_i^{h,y} H_{h,y} = \nabla_h^2 D_i^{h,y} G_{h,y} - \nabla_h^2 D_i^{h,y} \tilde{G}_{h,y}$ in $Q_{d(y)/2}(y)$ we can use (2.7.2) and obtain

$$|\nabla_{h}^{2}D_{i}^{h,y}G_{h,y}(x)| \leq |\nabla_{h}^{2}D_{i}^{h,y}H_{h,y}(x)| + |\nabla_{h}^{2}D_{i}^{h,y}\tilde{G}_{h,y}(x)| \leq C\left(\frac{1}{d(y)^{\mathsf{d}-1}} + \frac{1}{(|x-y|+h)^{\mathsf{d}-1}}\right).$$

This implies (2.8.15) if we use that $|x - y|_{\infty} \leq \frac{d(y)}{4}$ and $d(y) \geq 10h$ so that $|x - y| + h \leq Cd(y)$.

Step 2: Proof of (2.8.14)

If d(y) < 9h we can repeat the trivial estimate from the previous step, so assume $d(y) \ge 9h$.

We want to integrate (2.8.15) along a suitable path. So let $(x^{(k)})_{k=0}^{L}$ be a straight path from $x^{(0)} = x$ to a closest point $x^{(L)} \in Q_{d(y)/4}(y) \setminus Q_{d(y)/4-h}(y)$. This path will have length $Lh = \left\lfloor \frac{d(y)}{4} \right\rfloor_{h} - |x - y|_{\infty}$ and we have $|y - x^{(L)}|_{\infty} \ge \frac{d(y)}{4} - h > \frac{d(y)}{8}$. By Lemma 2.8.2 we have

$$|\nabla_{h,x}\nabla_{h,y}G_{h}(x^{(L)},y)| \le C\frac{(d(x^{(L)})+h)(d(y)+h)}{|x^{(L)}-y|^{\mathsf{d}}} \le C\frac{(d(y)+h)^{2}}{|d(y)+h|^{\mathsf{d}}} \le C\frac{1}{|d(y)+h|^{\mathsf{d}-2}}.$$
(2.8.17)

Furthermore (2.8.15) implies that

$$|\nabla_{h,x}^2 \nabla_{h,y} G_h(x^{(k)}, y)| \le \frac{C}{(|x^{(k)} - y| + h)^{\mathsf{d} - 1}} \le \frac{C}{(|x - y| + (k + 1)h)^{\mathsf{d} - 1}}.$$
 (2.8.18)

Now we can integrate (2.8.18) along $(x^{(k)})_{k=0}^{L}$ and use (2.8.17), and after a short calculation we arrive at (2.8.14).

Step 3: Proof of (2.8.13)

If d(y) < 79h we can again use the trivial estimate from Step 1, so assume $d(y) \ge 79h$.

This is similar to the previous step: We choose a shortest-possible path $(y^{(k)})_{k=0}^{L}$ from $y^{(0)} = y$ to a point $y^{(L)} \in Q_{d(x)/6}(x) \setminus Q_{d(x)/6-h}(y)$. Then $|y^{(k)} - x|_{\infty} \leq \frac{d(x)}{6}$, so that $\frac{5}{6}d(x) \leq d(y^{(k)}) \leq \frac{7}{6}d(x)$ and hence

$$|y^{(k)} - x|_{\infty} \le \frac{d(x)}{6} \le \frac{d(y^{(k)})}{5}$$

Therefore we can apply (2.8.15) at the point $(x, y^{(k)})$ for each *k* and conclude

$$|\nabla_{h,x}^2 \nabla_{h,y} G_h(x, y^{(k)})| \le \frac{C}{(|x - y^{(k)}| + h)^{\mathsf{d} - 1}} \le \frac{C}{(|x - y| + (k + 1)h)^{\mathsf{d} - 1}}.$$
 (2.8.19)

On the other hand,

$$d(y^{(L)}) \ge \frac{5}{6}d(x) \ge \frac{5}{6}\frac{6}{7}d(y) \ge 56h$$

so that

$$|y^{(L)} - x|_{\infty} \ge \frac{d(x)}{6} - h \ge \frac{d(y^{(L)})}{7} - h > \frac{d(y^{(L)})}{8}.$$

This means that we can apply (2.8.8) at the point $(x, y^{(L)})$ and conclude

$$|\nabla_{h,x}^2 G_h(x, y^{(L)})| \le C \frac{(d(y^{(L)}) + h)^2}{|x - y^{(L)}|^{\mathsf{d}}} \le C \frac{(d(y) + h)^2}{|d(y) + h|^{\mathsf{d}}} \le C \frac{1}{|d(y) + h|^{\mathsf{d}-2}}.$$
 (2.8.20)

Now we can integrate (2.8.19) along the path $(y^{(k)})_{k=0}^{L}$ and use the estimate (2.8.20) for the one endpoint to obtain (2.8.13).

Step 4: Proof of (2.8.12)

We could try to prove this by integrating (2.8.13) along a path. However, this turns out to be not sharp enough at least if d = 3 (we would get a logarithmic term instead of a constant term). Instead we will use the Sobolev inequality on the function $H_{h,y}$ from Step 1. Thereby we get a bound for $\nabla_{h,y}G_h(x, y)$ if x, y are close. By the symmetry of G_h we can turn this into a bound for $\nabla_{h,x}G_h(x, y)$.

If d(y) < 10h we can again use the trivial estimate from Step 1, so assume $d(y) \ge 10h$. Recall the function $H_{h,y}$ from Step 1. If we use the Sobolev and Poincaré inequality on $Q_{d(y)+h/2}(y)$ and the estimate (2.8.16) we obtain

$$\begin{split} \|D_{i}^{h,y}H_{h,y}\|_{L^{\infty}(Q_{d(y)+h/2}(y))} &\leq C(d(y)+h/2)^{2-\frac{d}{2}} \|\nabla_{h}^{2}D_{i}^{h,y}H_{h,y}\|_{L^{2}(Q_{d(y)+h/2}(y))} \\ &\leq Cd(y)^{2-\frac{d}{2}} \|\nabla_{h}^{2}D_{i}^{h,y}H_{h,y}\|_{L^{2}(\mathbb{R}^{d})} \\ &\leq Cd(y)^{3-d} \end{split}$$

and therefore

$$|\nabla_{h,y}H_{h,y}(x)| \le Cd(y)^{3-\mathsf{d}}$$

for any $x \in Q_{d(y)}(y)$. Now we can use (2.7.2) and the fact that $D_i^{h,y}H_{h,y} = D_i^{h,y}G_{h,y} - D_i^{h,y}\tilde{G}_{h,y}$ in $Q_{d(y)/2}(y)$ and obtain

$$|D_i^{h,y}G_{h,y}(x)| \le |D_i^{h,y}H_{h,y}(x)| + |D_i^{h,y}\tilde{G}_{h,y}(x)| \le Cd(y)^{3-\mathsf{d}}$$

for any $x \in Q_{d(y)/2}(y)$ and any $i \in \{1, ..., n\}$. By the symmetry of G_h in x and y we conclude that also

$$|D_i^{h,x}G_{h,x}(y)| \le Cd(x)^{3-\mathsf{d}}$$
(2.8.21)

for any $y \in Q_{d(x)/2}(x)$.

Now in the setting of (2.8.12) we are given x, y with $|y - x|_{\infty} \leq \frac{d(y)}{4}$. These satisfy $\frac{3}{4}d(y) \leq d(x) \leq \frac{5}{4}d(y)$, so that $|y - x|_{\infty} \leq \frac{1}{3}d(x)$ and in particular $y \in Q_{d(x)/2}(x)$. Thus we can apply (2.8.21) and obtain

$$|D_i^{h,x}G_{h,x}(y)| \le Cd(x)^{3-d} \le Cd(y)^{3-d}$$

which implies (2.8.12).

Step 5: Proof of (2.8.11)

This follows immediately from (2.8.3).

Proof of Theorem 2.1.3 i). Now that we have proved Lemma 2.8.3 and Lemma 2.8.2 the proof is straightforward. First observe that it suffices to consider $x, y \in \Lambda_h$ as otherwise G_h and its relevant derivatives are trivially 0.

We claim that we can combine (2.8.8) and (2.8.14) to obtain (2.1.11). Indeed, if $|x - y|_{\infty} \le \frac{d(y)}{8}$ we have $d(y) \le \frac{8}{7}d(x)$ and $|x - y| + h \le \sqrt{n}|x - y|_{\infty} + h < d(y) + h$ which implies

$$1 \le \frac{(d(x) + h)(d(y) + h)}{(|x - y| + h)^2}$$

and we are done by (2.8.14).

If however $|x - y|_{\infty} > \frac{d(y)}{8}$, then we have in particular $|x - y| \ge h$, so that $|x - y| + h \le 2|x - y|$. We also have $d(y) \le 8|x - y|$ and $d(x) \le 9|x - y|$ and we easily see that

$$\frac{(d(x)+h)(d(y)+h)}{|x-y|^{\mathsf{d}}} \le \frac{C}{(|x-y|+h)^{\mathsf{d}-2}}$$

so we are done by (2.8.8).

Similarly, we can combine (2.8.7) and (2.8.13) into the estimate

$$|\nabla_{h,x}^2 G_h(x,y)| \le \begin{cases} C \log \left(1 + \frac{(d(y)+h)^2}{(|x-y|+h)^2}\right) & d = 2\\ C \min \left(\frac{1}{|x-y|+h}, \frac{(d(y)+h)^2}{(|x-y|+h)^3}\right) & d = 3 \end{cases}$$

This is not quite (2.1.10), but it implies (2.1.10) unless d(y) = 0. On the other hand, if d(y) = 0 then $y \in \partial \Lambda_h$. Therefore $G_{h,y}$ is identically 0, so that $\nabla_{h,x^2}G_h(x,y) = 0$ and (2.1.10) holds as well.

Similarly we can combine (2.8.6) and (2.8.12), and (2.8.5) and (2.8.11) into

$$\begin{aligned} |\nabla_{h,x}G_h(x,y)| &\leq C \min\left((d(y)+h)^{3-\mathsf{d}}, \frac{(d(x)+h)d(y)^2}{(|x-y|+h)^{\mathsf{d}}} \right), \\ |G_h(x,y)| &\leq C \min\left((d(x)+h)^{2-\frac{\mathsf{d}}{2}} (d(y)+h)^{2-\frac{\mathsf{d}}{2}}, \frac{(d(x)+h)^2(d(y)+h)^2}{(|x-y|+h)^{\mathsf{d}}} \right) \end{aligned}$$

respectively. These estimates imply (2.1.9) and (2.8.11), except in the cases d(x) = 0 or d(y) = 0, which are again trivial.

Remark 2.8.4. As a byproduct of the proofs of Lemma 2.8.3 and Lemma 2.8.2 we proved the estimates (2.8.9) and (2.8.15) which can easily be combined into the estimate

$$|\nabla_{h,x}^2 \nabla_{h,y} G_h(x,y)| \le C \min\left(\frac{1}{(|x-y|+h)^{d-1}}, \frac{d(y)+h}{(|x-y|+h)^d}\right)$$
(2.8.22)

for any $x, y \in (h\mathbb{Z})^d$.

With the same method of proof it is possible to prove an estimate for $\nabla_{h,x}^2 \nabla_{h,y}^2 G_h$ as well. One again considers $H_{h,y} = G_{h,y} - \eta_h \tilde{G}_{h,y}$ in Lemma 2.8.3 and Lemma 2.8.2 and derives estimates for $\|\nabla_{h,x}^2 \nabla_{h,y}^2 H_{h,y}\|_{L^2(\mathbb{R}^d)}$. In combination with the pointwise estimates for \tilde{G}_h (in particular (2.7.3)) these again yield estimates for $\nabla_{h,x}^2 \nabla_{h,y}^2 G_h$ in the two regimes where *x* and *y* are far away and close together, respectively. The final result is

$$|\nabla_{h,x}^2 \nabla_{h,y}^2 G_h(x,y)| \le \frac{C}{(|x-y|+h)^{\mathsf{d}}}$$
(2.8.23)

for any $x, y \in (h\mathbb{Z})^d$.

Actually it is even possible to derive estimates for higher derivatives $\nabla_{h,x}^a \nabla_{h,y}^b G_h$, at least when $a \leq 2$ or $b \leq 2$. However we cannot expect these estimates to be optimal any more, because high derivatives are increasingly divergent near the singular boundary points, and our approach does not really capture this behaviour.

2.8.3 Convergence of Green's functions

Proof of Corollary 2.1.4. We begin with the proof of assertion i). We can assume that $h \leq \frac{1}{3}$. There exists a unique $y_h \in \Lambda_h$ such that $y \in y_h + [-\frac{h}{2}, \frac{h}{2})^2$. Set $u_h(x) = G_h(x, y_h)$. We extend u_h by zero to $(h\mathbb{Z})^d \setminus \operatorname{int} \Lambda_h$.

To prove (i) we have to show that u_h converges uniformly to $G(\cdot, y)$. Testing the equation for $\Delta_h^2 u_h$ with u_h we get (see Lemma 2.8.1)

$$\|
abla_h^2 u_h\|_{L^2(\mathbb{R}^d)} \le C d^{2-\frac{d}{2}}(y_h) \le C$$

The discrete Sobolev-Poincaré inequality implies in particular that the u_h are uniformly Hölder continuous

$$[u_h]_{C_h^{0,\frac{1}{4}}(\mathbb{R}^d)} \le C.$$
(2.8.24)

Denote by J_h the interpolation operator introduced in Section 2.4. From Proposition 2.4.2 vi) and the Poincaré inequality we deduce that the sequence $J_h u_h$ is bounded in $W^{2,2}(\mathbb{R}^d)$ and $J_h u_h = 0$ in $\mathbb{R}^d \setminus (-3h, 1+3h)^d$. It follows that for a subsequence

$$J_{h_k}u_{h_k} \rightharpoonup u \quad \text{in } W^{2,2}(\mathbb{R}^d), \qquad u = 0 \text{ in } \mathbb{R}^d \setminus (0,1)^d.$$

From the uniform Hölder continuity (2.8.24) and Proposition 2.4.2 iii), iv) and vi) we deduce that, for any $x \in (-3h, 1+3h)^d$,

$$|J_{h_k}u_{h_k}(x) - I_{h_k}^{pc}u_{h_k}(x)| = |J_{h_k}(u_{h_k}(\cdot) - u_{h_k}(x))(x)| \le C ||u_{h_k} - u_{h_k}(x)||_{L^{\infty}(Q_{3h_k}(x))} \le Ch_k^{\frac{1}{4}}$$

and therefore

$$\sup_{x\in(-1,2)^{\mathsf{d}}}|J_{h_k}u_{h_k}(x)-I_{h_k}^{pc}u_{h_k}(x)|\leq Ch_k^{\frac{1}{4}}.$$

In connection with the compact embedding from $W_0^{2,2}((-1,2)^d)$ to $C^0((-1,2)^d)$ we conclude that

$$I_{h_k}^{pc} u_{h_k} \to u \quad \text{uniformly} \,. \tag{2.8.25}$$

If we can show that u(x) = G(x, y) then by uniqueness of the limit it follows that the convergences above do not only hold along a particular subsequence $h_k \rightarrow 0$ but for every subsequence $h_k \rightarrow 0$ and we are done.

To show that u(x) = G(x, y) we use that by definition of $G_h(\cdot, y_h)$ we have for each $\varphi \in C_c^5((0, 1)^d)$

$$\begin{split} \varphi(y_{h_k}) &= \sum_{x \in \operatorname{int} \Lambda_h} \Delta_{h_k}^2 u_{h_k}(x) \varphi(x) h_k^{\mathsf{d}} = \sum_{x \in \operatorname{int} \Lambda_h} u_{h_k}(x) \Delta_{h_k}^2 \varphi(x) h_k^{\mathsf{d}} \\ &= \int_{(0,1)^{\mathsf{d}}} I_{h_k}^{pc} u_{h_k} I_{h_k}^{pc} \Delta_{h_k}^2 \varphi(x) \, \mathsf{d}x \, . \end{split}$$

Now by Taylor expansion $|I_{h_k}^{pc}\Delta_{h_k}^2 \varphi - \Delta^2 \varphi| \le Ch_k$. Together with (2.8.25) we get

$$\varphi(y) = \lim_{k \to \infty} \varphi(y_{h_k}) = \lim_{k \to \infty} \int_{(0,1)^d} I_{h_k}^{pc} u_{h_k} I_{h_k}^{pc} \Delta_{h_k}^2 \varphi_h \, \mathrm{d}x = \int_{(0,1)^d} u \Delta^2 \varphi \, \mathrm{d}x \, .$$

Thus $\Delta^2 u = \delta_y$ in the sense of distributions. Since we also know that $u \in W_0^{2,2}((0,1)^d)$ we conclude that u(x) = G(x, y) as desired.

To prove ii) note that the estimates in Theorem 2.1.3 show that the second discrete derivatives are bounded in L^p for all $p < \infty$. Hence by the discrete Sobolev embedding theorem the discrete first derivatives are bounded in $C^{0,\alpha}$ for all $\alpha < 1$. This implies that

$$|I_h^{pc}\nabla_h u - \nabla J_h u_h| \le Ch^{\alpha}. \tag{2.8.26}$$

Moreover the L^p bound on the discrete second derivatives and (2.4.3) give a bound of Ju_h in $W^{2,p}$. Hence a subsequence of $J_h u_h$ converges in $C^{1,\alpha}$ to $G(\cdot, y)$. Since the limit is unique, the whole sequence converges in $C^{1,\alpha}$ to G. Together with (2.8.26) this yields uniform convergence of the discrete first derivatives.

The local compactness argument in Section 2.5 (and a diagonalisation argument) shows that a subsequence of $I_h^{pc} \nabla_h^2 u_h$ converges in $L^2_{loc}((0,1)^2 \setminus \{y\})$ to a function v. Since $I_h^{pc} \nabla_h^2 u_h$ is also bounded in L^q for some q > 2 we get strong convergence in $L^2((0,1)^2)$. Using again the L^q bound we get strong convergence in all L^p with p < q. Since we have L^q bounds for all $q < \infty$ we get strong convergence for all $p < \infty$. It remains to show that $v = \nabla^2 G(\cdot, y)$. To obtain this identity we can use discrete integration by parts and pass to the limit on both sides, as in the proof that $\Delta^2 u = \delta_y$.

The proof of (iii) is similar. Uniform boundedness of the discrete derivatives follows directly from Theorem 2.1.3. This theorem also shows that the second discrete derivatives are uniformly bounded on the complement of any cube $Q_r(y)$. It follows that the functions u_h are uniformly Lipschitz on the complement of any cube $Q_r(y)$ and we obtain locally uniform convergence of $I_h^{pc} \nabla_h u_h$ in the complement of those cubes as in the proof of (ii). Combined with the uniform boundedness we immediately conclude convergence of $I_h^{pc} \nabla_h u_h$ in L^p for all $p < \infty$.

The proof of L^p convergence of $I_h^{pc} \nabla_h^2 u_h$ for p < 3 is again analogous to the argument for d = 2.

3 Probability to be positive for the membrane model in dimensions two and three

This chapter is based on the paper [BDKS19], written jointly by Simon Buchholz, Jean-Dominique Deuschel, Noemi Kurt and the author, with only minor changes. A small part of the content of this chapter has already appeared in the author's M.Sc. thesis [Sch16], where the upper bound on the probability to be positive is shown using a very similar approach.

3.1 Introduction

In this section we consider entropic repulsion for the subcritical membrane model, as discussed in Section 1.3.3. Actually, for convenience we make a small change in comparison to the results mentioned there: instead of $\Lambda_N = \{0, ..., N\}^d$ we consider $\Lambda_N = \{-N, ..., N\}^d$. This means that we only consider boxes with odd sidelengths, and this has the small advantage that the centre of Λ_N is a lattice point. However, it is clear that our proof would also apply to boxes with even sidelengths. As we are only concerned with the membrane model, we drop the subscripts Δ .

3.1.1 Main results

Let $\Lambda = [-1, 1]^d$ and $\Lambda_N = N\Lambda \cap \mathbb{Z}^d$ with $d \in \mathbb{N}^+$ and $N \in \mathbb{N}^+$. We are interested in the event $\Omega_{D_N,+} = \{\psi: \psi_x \ge 0 \ \forall x \in D_N\}$, where $D_N \subset \Lambda_N$, as well as the behaviour of ψ conditioned on $\Omega_{D_N,+}$.

Our main result is the following.

Theorem 3.1.1. Let d = 2 or d = 3. There are constants C, c such that for all $N \in \mathbb{N}^+$, $0 \le L \le N$,

$$e^{-C\frac{N^{d-1}}{(L+1)^{d-1}}} \le \mathbb{P}_N\left(\Omega_{\Lambda_{N-L},+}\right) \le e^{-c\frac{N^{d-1}}{(L+1)^{d-1}}}.$$
(3.1.1)

A first result in this direction was already established by Sakagawa [Sak16] who proved that for every $x \in \Lambda$ there is a small neighbourhood B_x such that $\mathbb{P}_N(\Omega_{NB_{x,+}}) > c$ for some (universal) constant *c*.

Let us emphasize the two important special cases of our theorem that were already mentioned in Section 1.3.3. We first consider the case $D_N = \Lambda_{\delta N}$ for $\delta \in (0,1)$, where the hard wall stays away from the boundary. In that case the fact that the membrane model is Hölder continuous suggests that the field has a decent chance to be positive if it is uniformly positive on a sufficiently dense set of lattice points of bounded cardinality. Thus the probability that ψ is positive on $D_N = \Lambda_{\delta N}$ should be comparable to the probability of uniform positivity on that dense set, and hence bounded away from zero. Indeed, Theorem 3.1.1 implies the following corollary. **Corollary 3.1.2.** Let d = 2 or d = 3. For $\delta \in (0, 1)$ there is a constant $c_{\delta} > 0$ such that

$$c_{\delta} \leq \mathbb{P}_N(\Omega_{\Lambda_{\delta N},+}) \leq rac{1}{2}.$$

When $D_N = V_N$, the situation is somewhat different. While the Hölder continuity holds up to the boundary, the ψ_x for x near the boundary are only weakly correlated and behave almost like independent random variables. This suggests that the probability to be positive on all of V_N can at best scale like $e^{-cN^{d-1}}$ (note that the number of points of distance 1 to the boundary is of the order N^{d-1}). On the other hand, if the field is positive at all nearboundary points it gets pushed up in the interior quite a bit, and so the probability to be positive everywhere should be of the same order.

Indeed, another particular case of Theorem 3.1.1 is an estimate for $\mathbb{P}_N(\Omega_{\Lambda_{N,+}})$.

Corollary 3.1.3. Let d = 2 or d = 3. There are constants C, c such that

$$e^{-CN^{\mathsf{d}}-1} \leq \mathbb{P}_N\left(\Omega_{\Lambda_N,+}\right) \leq e^{-cN^{\mathsf{d}}-1}$$

We expect this result to be true for the membrane model and the gradient model in any dimension $d \ge 2$. For the gradient model a stronger result has been shown for $d \ge 3$ in [Deu96, Theorem 4.1]. Note that the behaviour for general $L \ge 1$ in Theorem 3.1.1 is different for the gradient model in dimension $d \ge 3$.

We give a proof of the lower and upper bound in Theorem 3.1.1 in Section 3.3 and 3.4, respectively.

3.1.2 Implications for entropic repulsion

Corollary 3.1.2 has some easy implications on the behaviour of the field when conditioned on $\Omega_{\Lambda_{\delta N,\ell}+}$. To state them precisely we need some preparation.

We define the interpolation $I_N \colon \mathbb{R}^{\Lambda_N} \to C^{0,1}([-1,1]^d)$ by $I_N f(x) = N^{-\frac{4-d}{2}} f(Nx)$ for $x \in \left(\frac{1}{N}\mathbb{Z}\right)^d \cap [-1,1]^d$, and interpolated piecewise affinely on simplices for other values of x. As the proof of [CDH19, Theorem 2.1] shows, the pushforward measures $I_N \# \mathbb{P}_N$ converge weakly in $C^{0,\alpha}([-1,1]^d)$ for any $\alpha < \frac{4-d}{2}$ to a limit law \mathbb{P}_∞ . The limit \mathbb{P}_∞ is the continuum Bilaplace field, i.e., the centred Gaussian field whose covariance is the Green's function of the continuum Bilaplace operator on Λ . Now Corollary 3.1.2 implies that the laws $I_N \# \mathbb{P}_N$ still converge when one conditions on $\Omega_{\Lambda_{\delta N},+}$. Indeed, if we introduce the event $\Omega_{D,+}^* = \{u \in C^{0,\alpha}([-1,1]^d) \colon u(x) \ge 0 \ \forall x \in D\}$ for $D \subset [-1,1]^d$, we have the following result.

Corollary 3.1.4. Let d = 2 or d = 3, and $\delta \in (0, 1)$. Then $I_N # \mathbb{P}_N(\cdot | \Omega_{\Lambda_{\delta N}, +})$ converges weakly in $C^{0,\alpha}([-1, 1]^d)$ for any $\alpha < \frac{4-d}{2}$ to $\mathbb{P}_{\infty}\left(\cdot | \Omega^*_{\delta \Lambda, +}\right)$. In particular, we have

$$\lim_{N\to\infty}\mathbb{E}_N\left(N^{-\frac{4-d}{2}}\max_{x\in\Lambda_N}\psi_x\,\middle|\,\Omega_{\Lambda_{\delta N},+}\right)<\infty\,.$$

The corollary follows from the facts that $\mathbb{P}_{\infty}\left(\Omega^*_{\delta\Lambda,+}\right)$ is a continuous function of δ and that $\Omega^{\infty}_{\delta\Lambda,+}$ is a continuity set for \mathbb{P}_{∞} (these both follow from [Bog98, Corollary 4.4.2]). Note that the second point combined with the convergence of $I_N \# \mathbb{P}_N \to \mathbb{P}_{\infty}$ and Corollary 3.1.2 implies that $\mathbb{P}_{\infty}\left(\Omega^*_{\delta\Lambda,+}\right) > 0$ so that the conditioned measure $\mathbb{P}_{\infty}\left(\cdot \mid \Omega^*_{\delta\Lambda,+}\right)$ is well-defined.

This corollary shows that there is no entropic repulsion when conditioning on $\Omega_{\Lambda_{\delta N_{\ell}}+}$.

We conjecture that a similar result remains true if we condition on $\Omega_{\Lambda_{N,+}}$. However, due to the fact that the probability of $\Omega_{\Lambda_{N,+}}$ is exponentially small this is a difficult problem even in dimension one.

Conjecture 3.1.5. For d = 2 and d = 3 the measures $I_N # \mathbb{P}_N(\cdot | \Omega_{\Lambda_N,+})$ converge weakly in $C^{0,\alpha}([-1,1]^d)$ for any $\alpha < \frac{4-d}{2}$ to some limiting measure. In particular,

$$\lim_{N o\infty} \mathbb{E}_N\left(N^{-rac{4-d}{2}}\max_{x\in\Lambda_N}\psi_x\,\bigg|\,\Omega_{\Lambda_N,+}
ight)<\infty\,.$$

As an analogue to this conjecture one can consider the gradient model in one dimension (i.e. the random walk on $\{-N, -N + 1, ..., N\}$ with i.i.d. Gaussian increments conditioned to be zero at its endpoints). It is well-known that this model, suitably rescaled, converges weakly in $C^{0,\alpha}([-1,1])$ for $\alpha < \frac{1}{2}$ to a Brownian bridge. Moreover, if one conditions the walk to be non-negative it converges weakly in $C^0([-1,1])$ to a Brownian excursion (see [CC13] and the references therein). Similar results (in particular a local limit theorem for the conditioned field) have also been shown for the membrane model in one dimension (at least if one only considers zero boundary data on one end of the interval), see [DW15].

3.1.3 Notation

For $x \in \mathbb{Z}^d$ let $d_N(x) = \text{dist}_{\infty}(x, \mathbb{Z}^d \setminus \Lambda_N)$ be the distance to the boundary of Λ_N .

In the following *c*, *C* and *C*′ denote constants that may change from line to line, but are always independent of *N* and *L*.

3.2 Preliminaries

Let us recall the relevant results that will be used in the proof of the main theorems. Let G_N be the Green's function of Δ^2 on Λ_N with 0 boundary data outside Λ_N , i.e. $G_N(\cdot, y) = 0$ if $y \notin \Lambda_N$ and

$$\Delta^2 G_N(\cdot, y) = \delta_y \quad \text{in } \Lambda_N,,$$

$$G_N(\cdot, y) = 0 \quad \text{outside } \Lambda_N$$

if $y \in \Lambda_N$. The Green's function G_N agrees with the covariance matrix of ψ , i.e. we have that $\text{Cov}_N(\psi_x, \psi_y) = G_N(x, y)$, see also [Kur09]. Our proofs are based on the estimates for the Green's function G_N from Chapter 2.

Theorem 3.2.1. Let d = 2 or d = 3. Then we have for any $x, y \in \Lambda_N$

$$cd_N(x)^{4-\mathsf{d}} \le G_N(x,x) \le Cd_N(x)^{4-\mathsf{d}}$$
, (3.2.1)

$$|
abla_{1,x}G_N(x,y)| \le Cd_N(x)^{3-\mathsf{d}}$$
, (3.2.2)

$$|G_N(x,x) - G_N(x,y)| \le Cd_N(x)^{3-d} |x-y|_{\infty},$$
(3.2.3)

$$|G_N(x,y)| \le C \frac{d_N(x)^2 d_N(y)^2}{(|x-y|_{\infty}+1)^{\mathsf{d}}},$$
(3.2.4)

where $\nabla_{1,x}$ denotes the discrete gradient with respect to *x*.

Proof. The estimates (3.2.1), (3.2.2) and (3.2.4) are taken from Theorem 2.1.1, while (3.2.3) follows from (3.2.2) by discrete integration along a path from *x* to *y*.

The lower bound relies on Dudley's inequality proved in [Dud67]. To state the inequality we introduce the following two notions. For a Gaussian process $(X_t)_{t \in T}$ we define the pseudometric d_X by

$$d_X(s,t) = \sqrt{\mathbb{E}(|X_s - X_t|^2)}$$
. (3.2.5)

The entropy number $\mathcal{N}(T, d_X, r)$ is the minimal number of open balls of radius r in the d_X metric that are needed to cover T.

Theorem 3.2.2. Let $(X_t)_{t \in T}$ be a centred Gaussian process. Then

$$\mathbb{E}\left(\sup_{t\in T}X_t\right)\leq 24\int_0^\infty\sqrt{\ln\mathcal{N}(T,d_X,r)}\,\mathrm{d}r\,.$$

Remark 3.2.3. The theorem is true for arbitrary sets T if one defines the supremum appropriately, see e.g. [Tal96]. Since we only apply it to finite index sets we do not discuss this issue here any further.

We also use the Gaussian correlation inequality due to Royen [Roy14] (see also [LM17]).

Theorem 3.2.4. Let v be a centred Gaussian measure on \mathbb{R}^m and $K, L \subset \mathbb{R}^m$ be closed, symmetric and convex. Then

$$\nu(K \cap L) \ge \nu(K)\nu(L) \,. \tag{3.2.6}$$

Finally, we recall a Gaussian correlation inequality do to Li and Shao [LS04, Lemma 5.1] that will be used in the proof of the upper bound

Lemma 3.2.5. Let $m \in \mathbb{N}$, and $X = (X_1, \ldots, X_m)$, $Y = (Y_1, \ldots, Y_m)$ be Gaussian random vectors with mean 0 and positive definite covariance matrices Σ_X , Σ_Y , and let \mathbb{P} denote their joint measure. If $\Sigma_Y \ge \Sigma_X$ (in the sense of symmetric matrices, i.e., $\Sigma_Y - \Sigma_X$ is positive semidefinite) then for every Borel set $F \subset \mathbb{R}^m$

$$\mathbb{P}(Y \in F) \geq \left(\frac{\det \Sigma_X}{\det \Sigma_Y}\right)^{\frac{1}{2}} P(X \in F).$$

For the convenience of the reader we repeat the short proof.

Proof. Let f_X , f_Y be the densities of X and Y. The assumption $\Sigma_Y \ge \Sigma_X$ implies that $\Sigma_X^{-1} \ge \Sigma_Y^{-1}$ and hence $(x, \Sigma_X^{-1}x) \ge (x, \Sigma_Y^{-1}x)$ for all $x \in \mathbb{R}^m$. Therefore:

$$f_{Y}(x) = \frac{1}{(2\pi)^{\frac{m}{2}} (\det \Sigma_{Y})^{\frac{1}{2}}} \exp\left(-\frac{1}{2}(x, \Sigma_{Y}^{-1}x)\right)$$

$$\geq \left(\frac{\det \Sigma_{X}}{\det \Sigma_{Y}}\right)^{\frac{1}{2}} \frac{1}{(2\pi)^{\frac{m}{2}} (\det \Sigma_{X})^{\frac{1}{2}}} \exp\left(-\frac{1}{2}(x, \Sigma_{X}^{-1}x)\right)$$

$$= \left(\frac{\det \Sigma_{X}}{\det \Sigma_{Y}}\right)^{\frac{1}{2}} f_{X}(x).$$

Then

$$\mathbb{P}(Y \in F) = \int_F f_Y(x) \, \mathrm{d}x \ge \left(\frac{\det \Sigma_X}{\det \Sigma_Y}\right)^{\frac{1}{2}} \int_F f_X(x) \, \mathrm{d}x = \left(\frac{\det \Sigma_X}{\det \Sigma_Y}\right)^{\frac{1}{2}} \mathbb{P}(X \in F) \, .$$

88

3.3 Lower bounds

Let

$$\Omega_{\Lambda_{N-L,\infty}} := \left\{ \psi \colon |\psi_x| \leq d_N(x)^{rac{4-d}{2}} \ orall x \in \Lambda_{N-L}
ight\}$$

be the event that ψ is uniformly small on Λ_{N-L} .

If ψ was $C^{0,\frac{4-d}{2}}$ -Hölder continuous with Hölder constant ≤ 1 with probability bounded below uniformly in N, this event would have a positive probability uniformly in N and L. Now ψ is only $C^{0,\frac{4-d}{2}-\varepsilon}$ -Hölder continuous (see Chapter 2 and [CDH19]), so we cannot expect a lower bound independent of N. Instead, we prove in Subsection 3.3.2 that the probability of $\Omega_{\Lambda_{N-L,\infty}}$ is bounded below by $e^{-c\frac{N^{d-1}}{(L+1)^{d-1}}}$. Then, using a change of measure argument, we show in Subsection 3.3.3 that, given $f: \Lambda_N \to \mathbb{R}$, we have

$$\mathbb{P}_N(f + \Omega_{\Lambda_{N-L},\infty}) \ge e^{-\frac{1}{2} \|\Delta f\|_{L^2}^2} \mathbb{P}_N(\Omega_{\Lambda_{N-L},\infty}).$$
(3.3.1)

Suppose now that we can find a function f such that $f(x) \ge d_N(x)^{\frac{4-d}{2}}$ for $x \in \Lambda_{N-L}$ and such that $\|\Delta f\|_{L^2}^2 \le C \frac{N^{d-1}}{(L+1)^{d-1}}$. Then $\Omega_{\Lambda_{N-L},+} \supset f + \Omega_{\Lambda_{N-L},\infty}$ and thus (3.3.1) will imply that

$$\begin{split} \mathbb{P}_{N}(\Omega_{\Lambda_{N-L},+}) &\geq \mathbb{P}_{N}(f + \Omega_{\Lambda_{N-L},\infty}) \\ &\geq e^{-\frac{1}{2}\|\Delta f\|_{L^{2}}^{2}} \mathbb{P}_{N}(\Omega_{\Lambda_{N-L},\infty}) \\ &\geq e^{-C\frac{N^{d-1}}{(L+1)^{d-1}}} \mathbb{P}_{N}(\Omega_{\Lambda_{N-L},\infty}) \end{split}$$

Combined with a lower bound on $\mathbb{P}_N(\Omega_{\Lambda_{N-L,\infty}})$ this implies the lower bound in Theorem 3.1.1. In Lemma 3.3.4 we construct an f with the desired properties.

3.3.1 Local smallness of the field

We first prove that locally the field is small with a positive probability. For $x_0 \in \Lambda_N$ and $\gamma > 0$ we define the set

$$A_{x_{0},\gamma} := \{ x \in \Lambda_{N} \colon |x - x_{0}|_{\infty} \le \gamma d_{N}(x_{0}) \} .$$
(3.3.2)

Lemma 3.3.1. Let d = 2 or d = 3. There is a pair of constants $\gamma, \delta > 0$ with the following property: For all $x_0 \in \Lambda_N$ the following estimate holds

$$\mathbb{P}_N\left(\psi\colon |\psi_x| \le d_N(x)^{\frac{4-d}{2}} \,\forall x \in A_{x_0,\gamma}\right) \ge \delta.$$
(3.3.3)

Proof. We apply Theorem 3.2.2 to the Gaussian process ψ distributed according to \mathbb{P}_N . We assume $\gamma < \frac{1}{2}$ so that $x \in A_{x_0,\gamma}$ implies

$$rac{d_N(x_0)}{2} \leq d_N(x) \leq rac{3d_N(x_0)}{2}$$
 .

Therefore we will always estimate distances to the boundary for $x \in A_{x_0,\gamma}$ by $d_N(x_0)$ in the following. The bound (3.2.3) implies

$$\mathbb{E}_{N}(\psi_{x} - \psi_{y})^{2} \leq |G_{N}(x, x) - G_{N}(x, y)| + |G_{N}(y, y) - G_{N}(y, x)| \leq \Theta d_{N}(x_{0})^{3-\mathsf{d}} |x - y|_{\infty}$$
(3.3.4)

for $x, y \in A_{x_0,\gamma}$ and some $\Theta > 0$. Therefore we estimate the Gaussian pseudometric defined in (3.2.5) by

$$d_{\psi}(x,y) \leq \sqrt{\Theta} d_N(x_0)^{3-\mathsf{d}} |x-y|_{\infty}.$$

This implies that for r > 0 and $x, y \in A_{x_0,\gamma}$ such that $|x - y|_{\infty} \leq \frac{r^2}{\Theta d_N(x_0)^{3-d}}$ we have

 $d_{\psi}(x,y) \leq r.$

In particular $B_{\infty}\left(x, \frac{r^2}{\Theta d_N(x_0)^{3-\mathsf{d}}}\right) \subset B_{d_{\psi}}(x, r)$ and therefore

$$\mathcal{N}(A_{x_0,\gamma}, d_{\psi}, r) \leq \left[\frac{\gamma d_N(x_0)}{\frac{r^2}{\Theta d_N(x_0)^{3-\mathsf{d}}}}\right]^{\mathsf{d}} \leq 1 \vee \left(\frac{2\gamma \Theta d_N(x_0)^{4-\mathsf{d}}}{r^2}\right)^{\mathsf{d}}.$$

Then Theorem 3.2.2 implies

$$\mathbb{E}_{N}\left(\sup_{x\in A_{x_{0},\gamma}}\psi_{x}\right) \leq 24\int_{0}^{\sqrt{2\gamma\Theta d_{N}(x_{0})^{4-d}}}\sqrt{\ln\left(\frac{2\gamma\Theta d_{N}(x_{0})^{4-d}}{r^{2}}\right)^{d}}dr$$

$$\leq 24d_{N}(x_{0})^{\frac{4-d}{2}}\sqrt{2\gamma\Theta n}\int_{0}^{1}\sqrt{-2\ln r}\,dr \leq \lambda\sqrt{\gamma}d_{N}(x_{0})^{\frac{4-d}{2}}$$
(3.3.5)

where λ only depends on d.

If we take $\gamma = (16\lambda)^{-2}$ we obtain

$$\mathbb{E}_{N}\left(\sup_{x\in A_{x_{0},\gamma}}\psi_{x}\right) \leq \frac{1}{16}d_{N}(x_{0})^{\frac{4-d}{2}}.$$
(3.3.6)

Define the oscillation of a function f on a set T as usual by

$$\operatorname{osc}_T f = \sup_T f - \inf_T f.$$

Since ψ_x is a centred process (3.3.5) implies

$$\mathbb{E}_N\left(\operatorname{osc}_{A_{x_0,\gamma}}\psi_x\right) \leq \frac{1}{8}d_N(x_0)^{\frac{4-\mathsf{d}}{2}}.$$

This implies that

$$\mathbb{P}_N\bigg(\operatorname{osc}_{A_{x_0,\gamma}}\psi_x \leq \frac{1}{4}d_N(x_0)^{\frac{4-d}{2}}\bigg) \geq \frac{1}{2}$$

Note that we have the inclusions

$$\begin{split} \left\{ \psi : |\psi_x| \le d_N(x)^{\frac{4-d}{2}} \, \forall x \in A_{x_0,\gamma} \right\} \supset \left\{ \psi : |\psi_x| \le \frac{1}{2} d_N(x_0)^{\frac{4-d}{2}} \, \forall x \in A_{x_0,\gamma} \right\} \\ \supset \left\{ \psi : \operatorname{osc}_{A_{x_0,\gamma}} \psi_x \le \frac{1}{4} d_N(x_0)^{\frac{4-d}{2}} \right\} \cap \left\{ \psi : |\psi_{x_0}| \le \frac{1}{4} d_N(x_0)^{\frac{4-d}{2}} \right\} \end{split}$$

Now the Gaussian correlation inequality (3.2.6) together with (3.2.1) imply that

$$\mathbb{P}_N\left(\left\{\psi: |\psi_x| \le d_N(x)^{\frac{4-d}{2}} \,\forall x \in A_{x_0,\gamma}\right\}\right) \ge \frac{1}{2} \mathbb{P}_N\left(|\psi_{x_0}| \le \frac{1}{4} d_N(x_0)^{\frac{4-d}{2}}\right) \ge \delta$$

for some fixed $\delta > 0$.

Remark 3.3.2. The use of the Gaussian correlation inequality could be avoided here: from (3.3.6) and (3.2.1) one easily obtains

$$\mathbb{E}_N\left(\sup_{x\in A_{x_0,\gamma}}|\psi_x|\right) \leq \mathbb{E}_N\left(\sup_{x\in A_{x_0,\gamma}}|\psi_x-\psi_{x_0}|\right) + \mathbb{E}_N(|\psi_{x_0}|) \leq \Xi d_N(x_0)^{\frac{4-d}{2}}$$

for some $\Xi > 0$ and therefore

$$\mathbb{P}_N\left(\psi\colon |\psi_x| \le 4\Xi d_N(x)^{\frac{4-d}{2}} \ \forall x \in A_{x_0,\gamma}\right) \ge \mathbb{P}_N\left(\psi\colon |\psi_x| \le 2\Xi d_N(x_0)^{\frac{4-d}{2}} \ \forall x \in A_{x_0,\gamma}\right) \ge \frac{1}{2}.$$

We could work with this estimate instead of (3.3.3) by using

$$ilde{\Omega}_{\Lambda_{N-L},\infty} := \left\{ \psi \colon |\psi_x| \leq 4 \Xi d_N(x)^{rac{4-\mathsf{d}}{2}} \ orall x \in \Lambda_{N-L}
ight\}$$

instead of $\Omega_{\Lambda_{N-L},\infty}$ in the following.

3.3.2 Global smallness of the field

From the previous we know that on small boxes the field is small with probability bounded away from zero. We can cover Λ_{N-L} with these small boxes, and then use the Gaussian correlation inequality to obtain a bound on the probability that the field is globally small.

Lemma 3.3.3. Let d = 2 or d = 3, let $\Omega_{\Lambda_{N-L},\infty}$ be as before. Then we have

$$\mathbb{P}_N(\Omega_{\Lambda_{N-L},\infty}) \ge e^{-C\frac{N^{\mathsf{d}-1}}{(L+1)^{\mathsf{d}-1}}}$$

Proof. Recall the definition of $A_{x,\gamma}$ in (3.3.2). Fix γ such that the conclusion of Lemma 3.3.1 holds and use the shorter notation $A_x := A_{x,\gamma}$.

We want to construct a subset B_N of Λ_N such that $|B_N| \leq C \frac{N^{d-1}}{(L+1)^{d-1}}$ and such that

$$\Lambda_{N-L}\subset \bigcup_{x\in B_N}A_x.$$

If we have found such a set, then the Gaussian correlation inequality (Theorem 3.2.4) and Lemma 3.3.1 imply that

$$\begin{split} \mathbb{P}_{N}(\Omega_{\Lambda_{N-L},\infty}) &\geq \mathbb{P}_{N}\bigg(\bigcap_{x\in B_{N}}\{\psi\colon |\psi_{y}| < d_{N}(y)^{\frac{4-d}{2}} \;\forall y\in A_{x}\}\bigg) \\ &\geq \prod_{x\in B_{N}}\mathbb{P}_{N}\left(\psi\colon |\psi_{y}| < d_{N}(y)^{\frac{4-d}{2}} \;\forall y\in A_{x}\right) \\ &\geq \prod_{x\in B_{N}}\delta = \delta^{|B_{N}|} \geq e^{-C\frac{N^{d-1}}{(L+1)^{d-1}}}. \end{split}$$

It remains to prove the existence of B_N . The size of the boxes A_x depends on the distance to the boundary, so in order to construct B_N it is convenient to split Λ_N into the dyadic annuli $W_{N,k} = \{x \in \Lambda_N : 2^k \le d_N(x) < 2^{k+1}\}$ for $k = 0, 1, \ldots, \lfloor \log_2 N \rfloor$. For $x \in W_{N,k}$ the cube A_x has diameter $2\gamma d_N(x) \ge \gamma 2^{k+1}$. Because $W_{N,k}$ has outer sidelength $2(N - 2^k) \le 2N$ and thickness 2^k , we can cover it by at most

$$2n\left(2\frac{2N}{\gamma 2^{k+1}}\right)^{\mathsf{d}-1}2\frac{2^{k}}{\gamma 2^{k+1}} \le C\frac{N^{\mathsf{d}-1}}{2^{k(\mathsf{d}-1)}}$$

cubes A_x , i.e. we find a set $B_{N,k}$ of at most $C \frac{N^{d-1}}{2^{k(d-1)}}$ points in Λ_N such that

$$W_{N,k} \subset \bigcup_{x \in B_{N,k}} A_x$$

Let $k_0 = \lfloor \log_2(L+1) \rfloor$ which implies that $\Lambda_{N-L} \subset \bigcup_{k \ge k_0} W_{N,k}$. Consider $B_N = \bigcup_{k=k_0}^{\log_2 N} B_{N,k}$. Then $\Lambda_{N-L} \subset \bigcup_{x \in B_N} A_x$, and we have

$$|B_N| \le \sum_{k=k_0}^{\lfloor \log_2 N \rfloor} |B_{N,k}| \le C \sum_{k=k_0}^{\infty} \frac{N^{d-1}}{2^{k(d-1)}} \le C \frac{N^{d-1}}{2^{k_0(d-1)}} \le C \frac{N^{d-1}}{(L+1)^{d-1}}.$$

3.3.3 Change of measure

We can now prove the lower bound in Theorem 3.1.1. The idea is simple: We use an explicit calculation with densities to prove that the probability of the event $\mathbb{P}_N(f + \Omega_{\Lambda_{N-L},\infty})$ is bounded below by $e^{-\|\Delta f\|_{L^2}^2} \mathbb{P}_N(\Omega_{\Lambda_{N-L},\infty})$. Then it remains to make a good choice of f.

Proof of Theorem 3.1.1, *lower bound*. Let $f: \Lambda_N \to \mathbb{R}$ be a function to be specified later, and extend it by 0 to all of \mathbb{Z}^d . We want to estimate the probability of the event $f + \Omega_{\Lambda_{N-L},\infty} = \{f + \psi : \psi \in \Omega_{\Lambda_{N-L},\infty}\}$. To do so, we calculate

$$\begin{split} \mathbb{P}_{N}(f + \Omega_{\Lambda_{N-L},\infty}) &= \int_{f + \Omega_{\Lambda_{N-L},\infty}} \frac{1}{Z_{N}} \exp\left(-\frac{1}{2} \|\Delta\psi\|_{L^{2}}^{2}\right) d\psi \\ &= \int_{\Omega_{\Lambda_{N-L},\infty}} \frac{1}{Z_{N}} \exp\left(-\frac{1}{2} \|\Delta(f + \psi)\|_{L^{2}}^{2}\right) d\psi \\ &= \int_{\Omega_{\Lambda_{N-L},\infty}} \frac{1}{Z_{N}} \exp\left(-\frac{1}{2} \|\Delta f\|_{L^{2}}^{2} - \frac{1}{2} \|\Delta\psi\|_{L^{2}}^{2} - (\Delta f, \Delta\psi)_{L^{2}}\right) d\psi. \end{split}$$
(3.3.7)

Because $\Omega_{\Lambda_{N-L},\infty}$ is symmetric around the origin, we can replace ψ by $-\psi$ and obtain that

$$\mathbb{P}_{N}(f + \Omega_{\Lambda_{N-L},\infty}) = \int_{\Omega_{\Lambda_{N-L},\infty}} \frac{1}{Z_{N}} \exp\left(-\frac{1}{2} \|\Delta f\|_{L^{2}}^{2} - \frac{1}{2} \|\Delta \psi\|_{L^{2}}^{2} + (\Delta f, \Delta \psi)_{L^{2}}\right) \, \mathrm{d}\psi. \quad (3.3.8)$$

If we add (3.3.7) and (3.3.8) and use the estimate $e^t + e^{-t} \ge 2$, we conclude

$$\mathbb{P}_{N}(f + \Omega_{\Lambda_{N-L},\infty}) = \frac{1}{2} \int_{\Omega_{\Lambda_{N-L},\infty}} \frac{e^{-\frac{1}{2} \|\Delta f\|_{L^{2}}^{2} - \frac{1}{2} \|\Delta \psi\|_{L^{2}}^{2}} \left(e^{(\Delta f, \Delta \psi)_{L^{2}}} + e^{-(\Delta f, \Delta \psi)_{L^{2}}}\right)}{Z_{N}} d\psi$$

$$\geq e^{-\frac{1}{2} \|\Delta f\|_{L^{2}}^{2}} \int_{\Omega_{\Lambda_{N-L},\infty}} \frac{e^{-\frac{1}{2} \|\Delta \psi\|_{L^{2}}^{2}}}{Z_{N}} d\psi$$

$$= e^{-\frac{1}{2} \|\Delta f\|_{L^{2}}^{2}} \mathbb{P}_{N}(\Omega_{\Lambda_{N-L},\infty}).$$
(3.3.9)

Note that the conclusion in (3.3.9) could also be derived from (3.3.7) using Jensen's inequality. We now choose f as in Lemma 3.3.4 below. Then

$$\|\Delta f\|_{L^2}^2 \le C \frac{N^{\mathsf{d}-1}}{(L+1)^{\mathsf{d}-1}} \,. \tag{3.3.10}$$



Figure 3.1: The functions f_i .

Moreover this choice of f ensures that $\Omega_{\Lambda_{N-L,+}} \supset f + \Omega_{\Lambda_{N-L,\infty}}$, and so (3.3.9), (3.3.10) and Lemma 3.3.3 imply that

$$\mathbb{P}_{N}(\Omega_{\Lambda_{N-L},+}) \geq \mathbb{P}_{N}(f + \Omega_{\Lambda_{N-L},\infty}) \geq e^{-\frac{1}{2}\|\Delta f\|_{L^{2}}^{2}} \mathbb{P}_{N}(\Omega_{\Lambda_{N-L},\infty})$$
$$\geq e^{-C\frac{N^{d-1}}{(L+1)^{d-1}}} e^{-C\frac{N^{d-1}}{(L+1)^{d-1}}} = e^{-C'\frac{N^{d-1}}{(L+1)^{d-1}}}.$$

Lemma 3.3.4. There is a constant C > 0 such that for every N and $0 \le L \le N$ there is a function $f : \mathbb{Z}^d \to \mathbb{R}$ such that supp $f \subset \Lambda_N$, $f(x) \ge d_N(x)^{\frac{4-d}{2}}$ for all $x \in \Lambda_{N-L}$ and

$$\sum_{x \in \mathbb{Z}^{\mathsf{d}}} |\Delta f(x)|^2 \le C \frac{N^{\mathsf{d}-1}}{(L+1)^{\mathsf{d}-1}}$$

Proof. We again use a dyadic construction. Recall $W_{N,k} = \{x \in \Lambda_N : 2^k \le d_N(x) < 2^{k+1}\}$ for $k = 0, 1, ..., \lfloor \log_2 N \rfloor$. Let in addition $W_{N,-1} = \mathbb{Z}^d \setminus \Lambda_N$.

Fix a smooth function η : $\mathbb{R} \to \mathbb{R}$ such that $\eta \ge 0$, $\eta = 1$ on $[1, \infty)$ and $\eta = 0$ on $(-\infty, 0]$. For $i \in \{1, 2, ..., n\}$ and $x \in \mathbb{Z}^d$ we introduce the distance $d_i(x) = \text{dist}(x, \mathbb{Z}^d \setminus (\mathbb{Z}^{i-1} \times \{-N, ..., N\} \times \mathbb{Z}^{d-i}))$ of x to the boundary in direction x_i .

For $j = 0, 1, \dots \lfloor \log_2 N \rfloor - 1$ consider the function

$$f_j(x) = 2^{\frac{j(4-d)}{2}+1} \prod_{i=1}^{d} \eta\left(\frac{d_i(x)}{2^j}\right)$$

(cf. Figure 3.1). Note that

$$f_j(x) = 2^{\frac{j(4-d)}{2}+1}$$
(3.3.11)

for all $x \in \Lambda_N$ such that $d_N(x) \ge 2^j$. Moreover

$$|\Delta f_j(x)| \le C 2^{\frac{j(4-d)}{2}+1} \|\eta''\|_{L^{\infty}} \frac{1}{2^{2j}} \le \frac{C \|\eta''\|_{L^{\infty}}}{2^{\frac{jd}{2}}}.$$
(3.3.12)

In fact $\Delta f_i(x) = 0$ if $d_N(x) > 2^j$ because f_k is constant on Λ_{N-2^j} . We define the function

$$f = \sum_{j = \lfloor \log_2(L+1) \rfloor}^{\lfloor \log_2 N \rfloor} f_j.$$

For $x \in \Lambda_{N-L}$ let now k be such that $x \in W_{N,k}$, and observe that $\lfloor \log_2(L+1) \rfloor \leq k \leq \lfloor \log_2 N \rfloor$. The estimate (3.3.11) implies

$$f(x) \ge f_k(x) \ge 2^{\frac{k(4-d)}{2}+1} \ge \left(2 \cdot 2^k\right)^{\frac{4-d}{2}} \ge d_N(x)^{\frac{4-d}{2}}$$

11 371

For an arbitrary $x \in \mathbb{Z}^d$ let again $k \in \{-1, 0, 1, ...\}$ be such that $x \in W_{N,k}$. Then (3.3.12) implies that

$$|\Delta f(x)| \leq \sum_{j=k \vee \lfloor \log_2(L+1) \rfloor}^{\lfloor \log_2 N \rfloor} |\Delta f_j| \leq \sum_{j=k \vee \lfloor \log_2(L+1) \rfloor}^{\infty} \frac{C \|\eta''\|_{L^{\infty}}}{2^{\frac{jd}{2}}} \leq \frac{C'}{2^{\frac{(k \vee \lfloor \log_2(L+1) \rfloor)d}{2}}}.$$

Using that $|W_{N,k}| \leq C2^k N^{d-1}$ for $k \geq 0$ and that $\Delta f(x)$ is zero on $W_{N,-1}$ except possibly on the set $\Lambda_{N+1} \setminus \Lambda_N$ of cardinality $CN^{d-1} \leq C'2^{-1}N^{d-1}$, the previous estimate implies that

$$\begin{split} \sum_{x \in \mathbb{Z}^{d}} |\Delta f(x)|^{2} &\leq \sum_{k=-1}^{\lfloor \log_{2} N \rfloor} \sum_{x \in W_{N,k}} |\Delta f(x)|^{2} \leq \sum_{k=-1}^{\infty} \frac{C2^{k} N^{d-1}}{2^{(k \vee \lfloor \log_{2} (L+1) \rfloor) d}} \\ &\leq \sum_{k=-1}^{\lfloor \log_{2} (L+1) \rfloor} \frac{C2^{k} N^{d-1}}{2^{\lfloor \log_{2} (L+1) \rfloor d}} + \sum_{k=\lfloor \log_{2} (L+1) \rfloor + 1} \frac{C2^{k} N^{d-1}}{2^{kd}} \\ &\leq C \frac{N^{d-1}}{(L+1)^{d-1}} + C \frac{N^{d-1}}{(L+1)^{d-1}} = C' \frac{N^{d-1}}{(L+1)^{d-1}}. \end{split}$$

3.4 Upper bounds

In order to prove the upper bound in Theorem 3.1.1, we will find a suitably sparse set $E_{N,L}$ of points at the boundary such that the random variables { $\psi_x : x \in E_{N,L}$ } are almost independent in the sense that their covariance matrix is diagonally dominant. We can then use Lemma 3.2.5 to compare them to actually independent random variables. The following argument is taken from [Sch16, Section 6.2.1].

Proof of Theorem 3.1.1, upper bound. Note that for $N \ge L > \frac{N}{2}$ the upper bound is trivial. Indeed, Λ_{N-L} is nonempty and so the symmetry of the field implies $\mathbb{P}_N(\Omega_{\Lambda_{N-L},+}) \le \frac{1}{2}$, while the right hand side of (3.1.1) exceeds $\frac{1}{2}$ if $L > \frac{N}{2}$ and $c < 2^{-d}$. We assume $L \le \frac{N}{2}$ in the following. Let $E_{N,L} = \Lambda_{N-L} \cap ((\lceil \alpha(L+1) \rceil \mathbb{Z})^{d-1} \times \{N-L\})$ where $\alpha \ge 1$ is a constant to be chosen later. This is a set of points on one face of $[-N + L, N - L]^d$ such that any two points have distance at least αL . Its cardinality satisfies

$$|E_{N,L}| = \left(2\left\lfloor\frac{N-L}{\lceil \alpha(L+1)\rceil}\right\rfloor + 1\right)^{d-1} \ge \left(\left\lfloor\frac{N-\frac{N}{2}}{\alpha(L+1)+1}\right\rfloor + 1\right)^{d-1}$$

$$\ge \left(\frac{\frac{N}{2}}{\alpha(L+1)+1}\right)^{d-1} \ge c\frac{N^{d-1}}{\alpha^{d-1}(L+1)^{d-1}}.$$
(3.4.1)

Clearly $d_N(x) = L + 1$ for any $x \in E_{N,L}$. Therefore according to (3.2.4) for $x \neq y$

$$|G_N(x,y)| \le C \frac{(L+1)^4}{(|x-y|_{\infty}+1)^d} \le C \frac{(L+1)^4}{|x-y|_{\infty}^d}.$$

If we combine this with (3.2.1) we obtain for any $x \in E_{N,L}$

$$\sum_{\substack{y \in E_{N,L} \\ y \neq x}} \frac{|G_N(x,y)|}{\sqrt{G_N(x,x)G_N(y,y)}} \le C \sum_{\substack{y \in E_{N,L} \\ y \neq x}} \frac{(L+1)^4}{(L+1)^{4-d} |x-y|_{\infty}^d}$$

$$= C \sum_{j=1}^{\infty} |\{y \in E_{N,L} \colon |y - x|_{\infty} = j \lceil \alpha(L+1) \rceil\}| \frac{(L+1)^{\mathsf{d}}}{(j \lceil \alpha(L+1) \rceil)^{\mathsf{d}}}$$
$$\leq \frac{C}{\alpha^{\mathsf{d}}} \sum_{j=1}^{\infty} \frac{a_j}{j^{\mathsf{d}}}$$

where $a_j = 2$ for d = 2 and $a_j = 8j$ for d = 3. Thus $\sum_{j=1}^{\infty} \frac{a_j}{j^d} < \infty$ and hence

$$\sum_{\substack{y \in E_{N,L} \\ y \neq x}} \frac{|G_N(x,y)|}{\sqrt{G_N(x,x)G_N(y,y)}} \le \frac{C}{\alpha^{\mathsf{d}}}.$$
(3.4.2)

We now choose α large enough that the right hand side of (3.4.2) becomes less than $\frac{1}{4}$.

We define the Gaussian random vector $(X_x)_{x \in E_{N,L}}$ by $X_x = \frac{\psi_x}{\sqrt{G_N(x,x)}}$. Let Σ_X be its covariance matrix. Then $(\Sigma_X)_{x,x} = 1$ for all x and (3.4.2) implies that

$$\sum_{\substack{y \in E_{N,L} \\ y \neq x}} |(\Sigma_X)_{x,y}| \le \frac{1}{4}.$$
(3.4.3)

Let $\{Y_x\}_{x \in E_{N,L}}$ be i.i.d. normal variables distributed according to $\mathcal{N}(0, \frac{3}{2})$ and let $\Sigma_Y = \frac{3}{2}\mathbb{1}_{E_{N,L}}$ be their joint covariance matrix, where $\mathbb{1}_{E_{N,L}}$ is a unit matrix indexed by $E_{N,L}$.

Because of (3.4.3) the matrix $\Sigma_Y - \Sigma_X$ then satisfies

$$(\Sigma_Y - \Sigma_X)_{x,x} = rac{3}{2} - 1 = rac{1}{2} > \sum_{\substack{y \in E_{N,L} \ y
eq x}} (\Sigma_X)_{x,y}$$

This means that $\Sigma_Y - \Sigma_X$ is strictly diagonally dominant and hence positive definite. Hence we can apply Lemma 3.2.5 and obtain

$$\begin{split} \left(\frac{1}{2}\right)^{|E_{N,L}|} &= \mathbb{P}(Y \in (0,\infty)^{E_{N,L}}) \\ &\geq \left(\frac{\det \Sigma_X}{\det \Sigma_Y}\right)^{\frac{1}{2}} \mathbb{P}(X \in (0,\infty)^{E_{N,L}}) \\ &= \left(\frac{\det \Sigma_X}{\det \Sigma_Y}\right)^{\frac{1}{2}} \mathbb{P}_N(\psi_x \ge 0 \; \forall x \in E_{N,L}) \\ &\geq \left(\frac{\det \Sigma_X}{\det \Sigma_Y}\right)^{\frac{1}{2}} \mathbb{P}_N(\Omega_{\Lambda_{N-L},+}) \,. \end{split}$$

It remains to estimate $\frac{\det \Sigma_X}{\det \Sigma_Y}$. Since Σ_Y is diagonal, $\det \Sigma_Y = \left(\frac{3}{2}\right)^{|E_{N,L}|}$. On the other hand, by (3.4.3) the matrix $\Sigma_X - \frac{3}{4}\mathbb{1}_{E_{N,L}}$ is still diagonally dominant and hence positive semidefinite. Hence all eigenvalues of Σ_X must be at least $\frac{3}{4}$. Therefore $\det \Sigma_X \ge \left(\frac{3}{4}\right)^{|E_{N,L}|}$.

We conclude

$$\mathbb{P}_{N}(\Omega_{\Lambda_{N-L},+}) \leq \left(\frac{1}{2}\right)^{|E_{N,L}|} \left(\frac{\det \Sigma_{Y}}{\det \Sigma_{X}}\right)^{\frac{1}{2}} \leq \left(\frac{1}{2}\right)^{|E_{N,L}|} \left(\frac{3/2}{3/4}\right)^{\frac{|E_{N,L}|}{2}} = \left(\frac{1}{\sqrt{2}}\right)^{|E_{N,L}|}.$$
 (3.4.4)

Recall that by (3.4.1) we have $|E_{N,L}| \ge c \frac{N^{d-1}}{\alpha^{d-1}(L+1)^{d-1}}$. Thus we finally obtain

$$\mathbb{P}_{N}(\Omega_{\Lambda_{N-L},+}) \leq \exp\left(-c \frac{N^{\mathsf{d}-1}}{(L+1)^{\mathsf{d}-1}}\right)$$

for $c = \frac{1}{2\alpha^{d-1}} \log 2$.

4 The maximum of the four-dimensional membrane model

This chapter is based on the author's paper [Sch20a], with only minor changes.

4.1 Introduction

As discussed in Section 1.3.2, in this chapter we will study the maximum of the fourdimensional membrane model. The main part of the proof are estimates for the Green's function of the discrete Bilaplacian in dimension 4. These were described in Section 1.4.3.

4.1.1 Main result for the membrane model

Recall that ψ_N denotes a sample of $\mathbb{P}_{\Delta,\Lambda_N}$, where $\Lambda_N = [0,N]^{\mathsf{d}} \cap \mathbb{Z}^{\mathsf{d}}$ and that $M_N^{\Delta} = \max_{x \in \Lambda_N} \psi_{N,x}$.

Our main result is the following.

Theorem 4.1.1. Let d = 4. The random variable

$$M_N^{\Delta} - m_N^{\Delta} := M_N^{\Delta} - \frac{1}{\pi} \log N + \frac{3}{16\pi} \log \log N$$

converges in distribution. The limit law is a randomly shifted Gumbel distribution μ_{∞} , given by

$$\mu_{\infty}((-\infty,x]) = \mathbb{E}e^{-\gamma^* \mathcal{Z}e^{-8\pi x}} \,\forall x$$

where γ^* is a constant and Z is a positive random variable that is the limit in law of

$$\mathcal{Z}_N = \sqrt{8} \sum_{v \in \Lambda_N} (\log N - \pi \psi_{N,v}) e^{-8(\log N - \pi \psi_{N,v})}.$$

Before we discuss our proof strategy, let us point out a generalization.

Remark 4.1.2. Our approach is not limited to the membrane model. In fact, consider for $l \in \mathbb{N}^+$ the ∇^l -model, given by the probability measure

$$\mathbb{P}_{A}^{(l)}(\mathrm{d}\psi) = \begin{cases} \frac{1}{Z_{A}^{(l)}} \exp\left(-\frac{1}{2}\sum_{v\in\mathbb{Z}^{\mathsf{d}}} |\Delta_{1}^{\frac{l}{2}}\psi_{v}|^{2}\right) \prod_{v\in A} \mathrm{d}\psi_{v} \prod_{v\in\mathbb{Z}^{\mathsf{d}}\setminus A} \delta_{0}(\mathrm{d}\psi_{v}) & l \text{ even} \\ \frac{1}{Z_{A}^{(l)}} \exp\left(-\frac{1}{2}\sum_{v\in\mathbb{Z}^{\mathsf{d}}} |\nabla_{1}\Delta_{1}^{\frac{l-1}{2}}\varphi_{v}|^{2}\right) \prod_{v\in A} \mathrm{d}\psi_{v} \prod_{v\in\mathbb{Z}^{\mathsf{d}}\setminus A} \delta_{0}(\mathrm{d}\psi_{v}) & l \text{ odd} \end{cases}$$

(note that l = 1 corresponds to the gradient model and l = 2 to the membrane model) in the critical dimension d = 2l on the cube $A = [0, N]^d \cap \mathbb{Z}^d$. Then Theorem 4.1.1 generalizes to this setting, and the maximum of the field, appropriately centred, converges in law to a randomly shifted Gumbel distribution. Our proof in the following would only require minor modifications to yield this more general result. However, since the case l = 1 is covered by [BDZ16], while the ∇^l -model for l > 2 is rarely studied, we choose to focus on the case l = 2 in the following. This allows us to avoid more complicated notation.

4.1.2 Log-correlated fields

As discussed in Section 1.3.2, one needs to consider this result in the context of log-correlated Gaussian fields, where one expects certain universality properties of features such as the maximum of the field. In particular, in [DRZ17] Ding, Roy and Zeitouni gave a set of four assumptions that ensure that the maximum of a field converges in distribution. Let us recall their result, slightly reformulated (we have changed the domain from $[0, N - 1]^d$ to $[0, N]^d$, and replaced $\log_+ |a|$ with $\log(1 + |a|)$ in (A.0) and (A.1), but it is straightforward to check that the theorem stated here is equivalent to the theorem as stated in [DRZ17]). We write $d_N(v) := \operatorname{dist}(v, \partial [0, N]^d)$ for the distance of v to the boundary of $[0, N]^d$ and $d(x) := d_1(x)$.

Theorem 4.1.3 ([DRZ17, Theorem 1.3 and Theorem 1.4]). Let $\Lambda_N = [0, N]^d \cap \mathbb{Z}^d$, and let $\varphi_N = \{\varphi_{N,v} : v \in \Lambda_N\}$ be a centred Gaussian field. Assume that

(A.0) (Logarithmically bounded fields) There is a constant $\alpha_0 > 0$ such that for all $u, v \in \Lambda_N$,

$$\operatorname{Var} \varphi_{N,v} \leq \log N + \alpha_0$$

and

$$\mathbb{E}(\varphi_{N,v} - \varphi_{N,u})^2 \le 2\log(1 + |u - v|) - |\operatorname{Var} \varphi_{N,v} - \operatorname{Var} \varphi_{N,u}| + 4\alpha_0$$

(A.1) (Logarithmically correlated fields) For any $\delta > 0$ there is a constant $\alpha^{(\delta)} > 0$ such that for all $u, v \in \Lambda_N$ with $\min(d_N(u), d_N(v)) \ge \delta N$

$$|\operatorname{Cov}(\varphi_{N,v},\varphi_{N,u}) - (\log N - \log(1 + |u - v|))| \le \alpha^{(\delta)}$$

(A.2) (Near diagonal behaviour) There are both a continuous function $f_1: (0,1)^d \to \mathbb{R}$ and a function $f_2: \mathbb{Z}^d \times \mathbb{Z}^d \to \mathbb{R}$ such that the following holds. For all $L, \varepsilon, \delta > 0$, there exists $N_0 = N_0(L, \varepsilon, \delta)$ such that for all $x \in [0, 1]^d$, $N \ge N_0$ such that $Nx \in \mathbb{Z}^d$ and $d(x) \ge \delta$, and for all $u, v \in [0, L]^d \cap \mathbb{Z}^d$ we have

$$|\operatorname{Cov}(\varphi_{N,Nx+v},\varphi_{N,Nx+u})-\log N-f_1(x)-f_2(u,v)|<\varepsilon.$$

(A.3) (Off diagonal behaviour) There is a continuous function $f_3: \mathcal{D}^d \to \mathbb{R}$, where $\mathcal{D}^d = \{(x,y): x, y \in (0,1)^d, x \neq y\}$ such that the following holds. For all $L, \varepsilon, \delta > 0$ there exists $N_1 = N_1(L,\varepsilon,\delta) > 0$ such that for all $x, y \in [0,1]^d$, $N \ge N_1$ such that $Nx, Ny \in \mathbb{Z}^d$, $\min(d(x), d(y)) \ge \delta$ and $|x - y| \ge \frac{1}{L}$ we have

$$|\operatorname{Cov}(\varphi_{N,Nx},\varphi_{N,Ny})-f_3(x,y)|<\varepsilon.$$

Let $M_N = \max_{v \in \Lambda_N} \varphi_{N,v}$ and

$$m_N = \sqrt{2\mathsf{d}} \log N - \frac{3}{2\sqrt{2\mathsf{d}}} \log \log N \,.$$

Then the sequence $M_N - m_N$ converges in distribution to a randomly shifted Gumbel distribution μ_{∞} . The limit distribution is given by

$$\mu_{\infty}((-\infty,x]) = \mathbb{E}e^{-\gamma^* \mathcal{Z}e^{-\sqrt{2}\mathsf{d}x}} \,\,\forall x$$

where γ^* is a constant and Z is a positive random variable that is the limit in law of

$$\mathcal{Z}_N = \sum_{v \in \Lambda_N} (\sqrt{2\mathsf{d}} \log N - \varphi_{N,v}) e^{-\sqrt{2\mathsf{d}}(\sqrt{2\mathsf{d}} \log N - \varphi_{N,v})}$$
This theorem easily implies Theorem 4.1.1 once we show that ψ_N^{Δ} (or rather $\sqrt{8\pi}\psi_N^{\Delta}$) satisfies assumptions (A.0), (A.1), (A.2), (A.3). In fact we can prove even slightly stronger statements than these. Let us state the precise results that we will prove. We abbreviate $\lambda = \sqrt{8\pi}$.

Theorem 4.1.4. The field $\varphi_N := \lambda \psi_N^{\Delta}$ in dimension d = 4 satisfies

(A.0') There is a constant $\alpha'_0 > 0$ such that for all $u, v \in \Lambda_N$,

 $\operatorname{Var} \varphi_{N,v} \leq \min\left(\log N + \alpha'_0, \alpha'_0 \log(2 + d_N(v))\right)$

and

$$\operatorname{Var} \varphi_{N,v} - \operatorname{Cov}(\varphi_{N,v}, \varphi_{N,u}) \leq \log(1 + |u - v|) + 2\alpha'_0.$$

(A.1') There is a constant $\alpha_0'' > 0$ such that for all $u, v \in \Lambda_N$

$$\left|\operatorname{Cov}(\varphi_{N,v},\varphi_{N,u}) - \log\left(2 + \frac{\max(d_N(u),d_N(v))}{1 + |u - v|}\right)\right| \leq \alpha_0''\,.$$

(A.2') There are a constant $\theta_0 > 0$, a continuous function $f_1: (0,1)^4 \to \mathbb{R}$ and a function $f_2: \mathbb{Z}^4 \times \mathbb{Z}^4 \to \mathbb{R}$ such that the following holds. For all $L, \varepsilon > 0, \theta > \theta_0$ there exists $N'_0 = N'_0(L, \varepsilon, \theta)$ such that for all $x \in [0,1]^4$, $N \ge N'_0$ such that $Nx \in \mathbb{Z}^4$ and $d(x) \ge \frac{(\log N)^{\theta}}{N}$, and for all $u, v \in [0,L]^4 \cap \mathbb{Z}^4$ we have

$$|\operatorname{Cov}(\varphi_{N,Nx+v},\varphi_{N,Nx+u}) - \log N - f_1(x) - f_2(u,v)| < \varepsilon.$$

(A.3') There are a constant $\theta_1 > 0$ and a continuous function $f_3: \mathcal{D}^4 \to \mathbb{R}$, where $\mathcal{D}^4 = \{(x, y) : x, y \in (0, 1)^4, x \neq y\}$ such that the following holds. For all $L, \varepsilon > 0, \theta > \theta_1$ there exists $N'_1 = N'_1(L, \varepsilon, \theta)$ such that for all $x, y \in V, N \ge N'_1$ such that $Nx, Ny \in \mathbb{Z}^4$, $\min(d(x), d(y)) \ge \frac{(\log N)^{\theta}}{N}$ and $|x - y| \ge \frac{1}{L}$ we have

$$|\operatorname{Cov}(\varphi_{N,Nx},\varphi_{N,Ny})-f_3(x,y)|<\varepsilon.$$

It is not hard to check that the assumptions (A.0'), (A.1'), (A.2'), (A.3') imply (A.0), (A.1), (A.2), (A.3) respectively, so that Theorem 4.1.1 is a straightforward corollary of Theorem 4.1.4. We give a few more details in Section 4.4.

The proof of Theorem 4.1.4 is the main contribution of this chapter. In the next section we will describe our approach.

4.1.3 Green's function estimates

The covariance function of the membrane model is the Green's function G_N^{Δ} of the discrete Bilaplacian on the grid $[0, N]^d$ with zero boundary data, and the assumptions (A.0'), (A.1'), (A.2'), (A.3') all correspond to certain estimates for this Green's function. Therefore our goal is to understand this Green's function. We are going to apply tools from PDE theory and numerical analysis, so before proceeding further it is convenient to rescale our domain to a unit box. Let $h = \frac{1}{N}$, let $\Lambda_h = [0, 1]^4 \cap (h\mathbb{Z})^4$, and let $\psi_{h,x}^{\Delta} := \psi_{N,\frac{x}{h}}^{\Delta}$. Let G_N^{Δ} and G_h^{Δ} be the covariance functions of ψ_N^{Δ} and ψ_h^{Δ} . Then also $G_h^{\Delta}(x, y) = G_N^{\Delta}(\frac{x}{h}, \frac{y}{h})$.

Using G_N^{Δ} and G_h^{Δ} , Λ_N and Λ_h , and ψ_N and ψ_h simultaneously is a slight abuse of notation. It should, however, always be clear from the context which object we are referring to. Let us also remark that from a PDE point of view it would arguably be more natural to choose $h = \frac{1}{N+2}$ and rescale $[0, N]^4$ to $[h, 1 - h]^4$, as this would give our domain a natural boundary layer of zeros, matching the continuous Dirichlet boundary data. Our choice of rescaling, however, is in line with [DRZ17].

Observation 4.1.5. Under the aforementioned rescaling, each statement (A.0'), (A.1'), (A.2'), (A.3') from Theorem 4.1.4 for $\lambda \psi_N^{\Delta}$ in dimension d = 4 is equivalent to the corresponding following statement for G_h^{Δ} .

(B.0') There is a constant $\alpha'_0 > 0$ such that for all $x, y \in \Lambda_h$,

$$\lambda^2 G_h^{\Delta}(x,x) \le \min\left(-\log h + \alpha'_0, \alpha'_0 \log\left(2 + \frac{d(x)}{h}\right)\right)$$

and

$$\lambda^2 \left(G_h^{\Delta}(x,x) - G_h^{\Delta}(x,y) \right) \le \log \left(1 + \frac{|x-y|}{h} \right) + 2\alpha'_0.$$

(B.1') There is a constant $\alpha_0'' > 0$ such that for all $x, y \in \Lambda_h$

$$\left|\lambda^2 G_h^{\Delta}(x,y) - \log\left(2 + \frac{\max(d(x),d(y))}{h+|x-y|}\right)\right| \leq \alpha_0''.$$

(B.2') There are a constant $\theta_0 > 0$, a continuous function $f_1: (0,1)^4 \to \mathbb{R}$ and a function $f_2: \mathbb{Z}^4 \times \mathbb{Z}^4 \to \mathbb{R}$ such that the following holds. For all $L, \varepsilon > 0, \theta > \theta_0$ there exists $N'_0 = N'_0(L, \varepsilon, \theta)$ such that for all $h \leq \frac{1}{N'_0}$ with $\frac{1}{h} \in \mathbb{N}$, all $x \in \Lambda_h$ such that $d(x) \geq h |\log h|^{\theta}$ and for all $u, v \in [0, L]^4 \cap \mathbb{Z}^4$ we have

$$\left|\lambda^2 G_h^{\Delta}(x+hu,x+hv)+\log h-f_1(x)-f_2(u,v)\right|<\varepsilon.$$

(B.3') There are a constant $\theta_1 > 0$ and a continuous function $f_3: \mathcal{D}^4 \to \mathbb{R}$, where $\mathcal{D}^4 = \{(x, y) : x, y \in (0, 1)^4, x \neq y\}$ such that the following holds. For all $L, \varepsilon > 0, \theta > \theta_1$ there exists $N'_1 = N'_1(L, \varepsilon, \theta)$ such that for all $h \leq \frac{1}{N'_1}$ with $\frac{1}{h} \in \mathbb{N}$ and for $x, y \in \Lambda_h$ such that $\min(d(x), d(y)) \geq h |\log h|^{\theta}$ and $|x - y| \geq \frac{1}{L}$ we have

$$\left|\lambda^2 G_h^{\Delta}(x,y) - f_3(x,y)\right| < \varepsilon.$$

Let us discuss how one might prove Theorem 4.1.4, or rather the statements (B.0'), (B.1'), (B.2'), (B.3'). We write $\Gamma_h = (h\mathbb{Z})^4 \cap ([-h, 1+h]^4 \setminus [0, h]^4)$. The function G_h^{Δ} is the Green's function associated to the discrete boundary value problem

$$\Delta_h^2 u_h = f_h \qquad \text{in } \Lambda_h$$

$$u_h = 0 \qquad \text{on } \Gamma_h$$

$$D_\nu^h u_h = 0 \qquad \text{on } \Gamma_h$$

(4.1.1)

(where $D_{\nu}^{h}u(x) = \frac{u(x+h\nu)-u(x)}{h}$ and ν is an outward unit normal vector). That is, for $y \in \Lambda_{h}$ the function $G_{h}(\cdot, y)$ is the unique solution of that equation with right hand side $f_{h} = \delta_{h}(y)$, defined as $\delta_{h,y}(x) = \begin{cases} \frac{1}{h^{4}} & \text{if } x = y \\ 0 & \text{otherwise} \end{cases}$.

One previous strategy to prove estimates for G_h^{Δ} , introduced in [Kur09] and used as well in [Cip13], was to compare G_h^{Δ} to \overline{G}_h^{Δ} , the Green's function associated to the discrete boundary value problem

$$\Delta_h^2 u_h = f_h \qquad \text{in } \Lambda_h$$

$$u_h = 0 \qquad \text{on } \Gamma_h' \qquad (4.1.2)$$

$$\Delta_h u = 0 \qquad \text{on } \Gamma_h$$

where $\Gamma'_h = (h\mathbb{Z})^4 \cap ([-2h, 1+2h]^4 \setminus [-h, 1+h]^4)$. The problem (4.1.2) can be seen as an iterated version of the discrete Poisson problem, and so many of the analytic and probabilistic tools available for the latter also have a version for (4.1.2). In particular, there are random walk representations for \overline{G}_h^{Δ} that allow to control it well. The strategy in [Kur09] then was to use PDE techniques to compare solutions of (4.1.1) and (4.1.2). This allows to estimate the difference between G_h and \overline{G}_h uniformly in compact subsets of $(0, 1)^4$. For our purposes, this is not good enough, as for (B.2') and (B.3') an error term that is only bounded is already too much. Note however that results similar to (B.0'), (B.1') can be proved using these methods. In fact, [Kur09, Proposition 1.1] and [Cip13, Lemma 2.1] are already weaker versions of (B.0') and (B.1').

In Chapter 2 we considered G_h^{Δ} in dimensions 2 and 3, and used a very different strategy, namely a compactness argument to transfer estimates for the continuous Green's function in domains with singularities to the discrete setting. This allowed us the prove discrete Caccioppoli inequalities (i.e. L^2 -based decay estimates on balls of various sizes) and to conclude from these estimates for G_h^{Δ} . In principle, this strategy can also be applied in our four-dimensional setting. One obstacle to this is that, unlike the two- or three-dimensional case, the relevant continuous estimates cannot be found in the literature. Even more importantly, the estimates in Chapter 2 are all up to a possibly large constant, and so the argument would have to be modified significantly to obtain estimates such as (B.2') and (B.3').

Instead of the aforementioned approaches to derive estimates for G_h^{Δ} we will use estimates for the approximation quality of finite difference schemes for the Bilaplacian. This idea is not completely new, as for example in [CDH19] estimates for finite difference schemes from [Tho64] were used to prove convergence of the rescaled four-dimensional membrane model in some negative Sobolev space. However, we would like to obtain a much stronger conclusion, namely pointwise estimates for the difference of the discrete and continuous Green's function. The result from [Tho64] is very general, but because of its generality it requires in our specific case very strong assumptions on the solution of the continuous Bilaplace equation to be approximated (being C^5) to yield estimates useful for us (the $W_h^{2,2}$ -approximation error decaying like $h^{\frac{1}{2}}$).

We will use a rather different estimate for the approximation quality of finite difference schemes. We will discuss the details in Section 4.2.2. Roughly speaking, the result is the following: Let $2 < s < \frac{5}{2}$, let $u \in W^{s,2} \cap W_0^{2,2}((0,1)^4)$ extended by 0 to \mathbb{R}^4 , and assume that $\Delta^2 u = f$ in $(0,1)^4$, so that u satisfies

$$\begin{aligned} \Delta^2 u &= f & \text{ in } (0,1)^4 \\ u &= 0 & \text{ on } \partial(0,1)^4 \\ \partial_\nu u &= 0 & \text{ on } \partial(0,1)^4 . \end{aligned} \tag{4.1.3}$$

Furthermore, let $u_h: (h\mathbb{Z})^4 \to \mathbb{R}$ be the solution of

$$\Delta_h^2 u_h = T^{h,3,3,3,3} f \qquad \text{in } \Lambda_h$$
$$u_h = 0 \qquad \qquad \text{on } (h\mathbb{Z})^4 \setminus \Lambda_h$$

where $T^{h,3,3,3,3}$ is a certain regularization operator. Then

$$||u - u_h||_{W^{2,2}_{h}(\Lambda_h)} \le Ch^{s-2} ||u||_{W^{s,2}((0,1)^4)}$$

where $\|\cdot\|_{W^{2,2}(\Lambda_h)}$ is a discrete Sobolev norm.

This result is inspired by closely related results in Chapter 5. However, in that chapter the focus is on obtaining estimates as above for *s* as large as possible. In the case of interest to us, $s < \frac{5}{2}$, the result can essentially be shown using the methods from [GMP83, IIS86, JS14].

We will use this result to compare solutions of (4.1.1) with solutions of (4.1.3). In particular, we will use it when *u* is the regular part of the continuous Green's function on $[0, 1]^4$. To do so, we need regularity estimates for solutions of (4.1.3). As already mentioned, optimal estimates for higher order elliptic problems on four-dimensional polyhedral domains are not yet in the literature. Instead we will use much weaker estimates (similar to ones in [MM13, MM14]) which are nonetheless sharp enough for our purposes. These estimates will allow us to place the regular part of the Green's function in $W^{2+\kappa_0,2}$ for some small $\kappa_0 > 0$, and this is good enough to apply the estimate above.

We will also need to have good estimates for the discrete Green's function on the full space $(h\mathbb{Z})^4$. These were derived in [Man67] using Fourier analysis. Furthermore, Theorem 4.2.3 gives us control over the $W_h^{2,2}$ -norm of the difference of u and u_h , while we are actually interested in the L_h^{∞} -norm and want it to decay. To achieve this, we will use a discrete Sobolev-inequality that allows us to control the L_h^{∞} -norm by the $W_h^{2,2}$ -norm at the cost of a term logarithmic in h. The presence of this term is the reason why we can prove (B.2') and (B.3') only up to distance $|\log h|^{\theta}$ to the boundary. For (B.0') and (B.1') we do not need a decaying but only a bounded error term and so we can prove these estimates on the whole domain.

We will give the details of the argument that we sketched here in the following sections. In Section 4.2 we gather various useful results: The aforementioned result on finite difference schemes, as well as some discrete inequality of Poincaré-Sobolev-type. These tools will allow us to compare G_h^{Δ} with various other Green's functions: the discrete Green's function of the full space (that we discuss in Section 4.3.1) and the continuous Green's functions of the box $[0,1]^4$ and of the full space (that we both discuss in Section 4.3.2). After all these preparations we can then turn to the proof of Theorem 4.1.4 in Section 4.4. We first prove a crucial lemma, Lemma 4.4.1 that shows that the regular part of the discrete and continuous Green's functions on the box are uniformly close, and then we use this Lemma and the results of the preceding sections to establish Theorem 4.1.4. Finally we use Theorem 4.1.3 to conclude Theorem 4.1.1 as well.

4.1.4 Notation

From now on we will only consider the membrane and not the gradient model, so there is no risk of confusion when we drop all superscripts Δ .

Occasionally we write r = s + O(t) to express $|r - s| \le Ct$.

We use the Sobolev space $W^{k,2}(\Omega)$ with the norm $\|u\|_{W^{k,2}(\Omega)}^2 = \sum_{|\alpha| \le k} \|\partial^{\alpha} u\|_{L^2(\Omega)}^2$. For s > 0 not an integer (i.e. s = k + t where $k \in \mathbb{N}$, 0 < t < 1) we will also encounter the fractional Sobolev space $W^{s,2}(\Omega)$ with norm $\|u\|_{W^{s,2}(\Omega)}^2 = \|u\|_{W^{k,2}(\Omega)}^2 + [u]_{W^{s,2}(\Omega)}^2$ and the seminorm $[u]_{W^{s,2}(\Omega)}^2 = \sum_{|\alpha|=k} \int_{\Omega} \int_{\Omega} \frac{|\partial^{\alpha} u(x) - \partial^{\alpha} u(y)|^2}{|x-y|^{4+2t}} dx dy$. For any s < 0 we define $W^{s,2}(\Omega)$ as the dual of $W_0^{-s,2}(\Omega)$. We extend these definitions to vector-valued functions by taking the Euclidean norm of the norms of the components.

For $A \subset (h\mathbb{Z})^4$ and $u_h: A \to \mathbb{R}$, we define $||u_h||_{L_h^2(A)}^2 = \sum_{x \in A} h^4 |u_h(x)|^2$, and $||u_h||_{L_h^\infty(A)}^2 = \sup_{x \in A} |u_h(x)|$. We will also use the discrete Sobolev-norm $||u_h||_{W_h^{2,2}(A)}^2 = ||u_h||_{L_h^2(A)}^2 + ||\nabla_h u_h||_{L_h^2(A)}^2 + ||\nabla_h^2 u_h||_{L_h^2(A)}^2$, where we extend the definitions to vector-valued functions as before.

Let us also fix once and for all a smooth function $\eta : \mathbb{R}^4 \to \mathbb{R}$ that is equal to 1 on $B_{\frac{1}{2}}(0)$ and 0 outside $B_1(0)$. We define $\eta^{(r)}(x) = \eta(rx)$, $\eta^{(r)}_y(x) = \eta^{(r)}(x-y)$ and let $\eta^{(r)}_{h,y}$ be the restriction of $\eta^{(r)}_y$ to $(h\mathbb{Z})^4$. Thus $\eta^{(r)}_y$ and $\eta^{(r)}_{h,y}$ are cut-off functions at scale r around y.

4.2 Preliminaries

4.2.1 Discrete Inequalities

We collect here two discrete inequalities that we will use several times in the following. We begin with a Poincaré inequality.

Lemma 4.2.1. Let $x_* \in (h\mathbb{Z})^4$, $r \ge 0$. Let $u_h: (h\mathbb{Z})^4 \to \mathbb{R}$ and suppose that u_h vanishes on at least one of the faces of $Q_r(x_*)$. Let this face be contained in a plane $x_i = c$. Then

$$\|u_h\|_{L^2_h(Q^h_r(x_*))}^2 \le Cr^2 \sum_{x: \{x, x+he_i\} \subset Q^h_r(x_*)} h^4 |D^h_i u_h(x)|^2 \le Cr^2 \|\nabla_h u_h\|_{L^2_h(Q^h_r(x_*))}^2.$$
(4.2.1)

Proof. This is a particular case of Lemma 2.2.1, but let us give a direct proof for the case at hand. The second inequality is obvious, so we only prove the first. By translating and reflecting the lattice and renaming the coordinates, we can assume i = 4, $Q_r^h(x_*) = [0, 2r]^4 \cap (h\mathbb{Z})^4$. We write $x = (x', x_4)$ where $x' \in \mathbb{R}^3$, $x_4 \in \mathbb{R}$, $u_h = 0$ if $x_4 = 0$. We will prove the one-dimensional estimate

$$\sum_{x_4 \in [0,2r] \cap h\mathbb{Z}} |u_h(x',x_4)|^2 \le Cr^2 \sum_{x_4 \in [0,2r-h] \cap h\mathbb{Z}} |D_4^h u_h(x',x_4)|^2.$$
(4.2.2)

Once we have established this, (4.2.1) follows by multiplying (4.2.2) by h^4 and summing over all $x' \in [0, 2r]^3 \cap (h\mathbb{Z})^3$. To prove (4.2.2), we use u(x', 0) = 0 and write

$$\begin{aligned} u_h(x', x_4) &| = \left| \sum_{y_4 \in [0, x_4 - h] \cap h\mathbb{Z}} u_h(x', y_4 + h) - u_h(x', y_4) \right| \\ &= \left| \sum_{y_4 \in [0, x_4 - h] \cap h\mathbb{Z}} hD_4^h u_h(x', y_4) \right| \\ &\leq h \left(\frac{x_4}{h} \right)^{\frac{1}{2}} \left(\sum_{y_4 \in [0, x_4 - h] \cap h\mathbb{Z}} |D_4^h u_h(x', y_4)|^2 \right)^{\frac{1}{2}} \end{aligned}$$

$$\leq \sqrt{2hr} \left(\sum_{y_4 \in [0,2r-h] \cap h\mathbb{Z}} |D_4^h u_h(x',y_4)|^2 \right)^{\frac{1}{2}}$$

and therefore

$$\sum_{x_4 \in [0,2r] \cap h\mathbb{Z}} |u_h(x',x_4)|^2 \le \frac{2r}{h} 2hr \sum_{y_4 \in [0,2r-h] \cap h\mathbb{Z}} |D_4^h u_h(x',y_4)|^2 \le 4r^2 \sum_{y_4 \in [0,2r-h] \cap h\mathbb{Z}} |D_4^h u_h(x',y_4)|^2.$$

This shows (4.2.2).

Next we give an inequality of Poincaré-Sobolev type. Given $u_h: (h\mathbb{Z})^4 \to \mathbb{R}$ that vanishes outside of Λ_h we would like to estimate its pointwise values by the $||u_h||_{W_h^{2,2}((h\mathbb{Z})^4)}$ -norm. We cannot hope for such an estimate to hold with a constant independent of h, as the (continuous) Sobolev space $W^{2,2}((0,1)^4)$ does not embed into $L^{\infty}((0,1)^4)$. However, by Strichartz's [Str72] version of the Moser-Trudinger inequality any $u \in W^{2,2}((0,1)^4)$ with $||u||_{W^{2,2}((0,1)^4)} = 1$ satisfies $\int_{(0,1)^4} e^{c|u(x)|^2} dx \leq C$, and this suggests that u can diverge at worst like $\sqrt{|\log |x||}$. So back in the discrete setting we can hope for an estimate with a factor scaling like $\sqrt{|\log h|}$. Indeed we have the following result:

Lemma 4.2.2. Assume that $u_h \colon (h\mathbb{Z})^4 \to \mathbb{R}$ vanishes outside of Λ_h . Then for any $x \in \Lambda_h$ we have

$$|u_h(x)| \leq C \sqrt{\log\left(2 + \frac{d(x)}{h}\right)} ||u_h||_{W_h^{2,2}((h\mathbb{Z})^4)}.$$

This lemma in combination with Theorem 4.2.3 will allow us to control the distance between the solution of a continuous Bilaplace equation and its discrete approximation at the cost of a logarithmic divergence (which we will be able to absorb in the applications in Section 4.4).

Proof of Lemma 4.2.2. We first want to localize to a ball around *x*. Let $v_h = \eta_{h,x}^{(d(x)+h)} u_h$. Then $v_h(x) = u_h(x)$. Furthermore v_h is supported on $Q_{d(x)+h}^h(x)$. The discrete chain rule implies that

$$\begin{split} |D_{i}^{h}v_{h}(y)| &\leq C \sup_{z \in Q_{h}^{h}(y)} \left| D_{i}^{h}\eta_{h,x}^{(d(x)+h)}(z) \right| \sup_{z \in Q_{h}^{h}(y)} |u_{h}(z)| \\ &+ C \sup_{z \in Q_{h}^{h}(y)} \left| \eta_{h,x}^{(d(x)+h)}(z) \right| \sup_{z \in Q_{h}^{h}(y)} |D_{i}^{h}u_{h}(z)|^{2} \\ &\leq C \sup_{z \in Q_{h}^{h}(y)} \left| D_{i}^{h}\eta_{h,x}^{(d(x)+h)}(z) \right| \left(\sum_{z \in Q_{h}^{h}(y)} |u_{h}(z)|^{2} \right)^{\frac{1}{2}} \\ &+ C \sup_{z \in Q_{h}^{h}(y)} \left| \eta_{h,x}^{(d(x)+h)}(z) \right| \left(\sum_{z \in Q_{h}^{h}(y)} |D_{i}^{h}u_{h}(z)|^{2} \right)^{\frac{1}{2}} \end{split}$$

and a similar expression for $|D_i^h D_{-j}^h v_h(y)|$. If we sum the squares of these eximates over y, we see that

$$\begin{split} \|v_{h}\|_{W_{h}^{2,2}((h\mathbb{Z})^{4})} &\leq C \|\eta_{h,x}^{(d(x)+h)}\|_{L_{h}^{\infty}((h\mathbb{Z})^{4})} \|\nabla_{h}^{2}u_{h}\|_{L_{h}^{2}(Q_{d(x)+2h}^{h}(x))} \\ &+ C \|\nabla_{h}\eta_{h,x}^{(d(x)+h)}\|_{L_{h}^{\infty}((h\mathbb{Z})^{4})} \|\nabla_{h}u_{h}\|_{L_{h}^{2}(Q_{d(x)+2h}^{h}(x))} \\ &+ C \|\nabla_{h}^{2}\eta_{h,x}^{(d(x)+h)}\|_{L_{h}^{\infty}((h\mathbb{Z})^{4})} \|u_{h}\|_{L_{h}^{2}(Q_{d(x)+2h}^{h}(x))} \\ &\leq C \|\nabla_{h}^{2}u_{h}\|_{L_{h}^{2}(Q_{d(x)+2h}^{h}(x))} + \frac{C}{d(x)+h} \|\nabla_{h}u_{h}\|_{L_{h}^{2}(Q_{d(x)+2h}^{h}(x))} \\ &+ \frac{C}{(d(x)+h)^{2}} \|u_{h}\|_{L_{h}^{2}(Q_{d(x)+2h}^{h}(x))} \,. \end{split}$$
(4.2.3)

We can apply Lemma 4.2.1 to u_h and $D_i^h u_h$ for any $i \in \{1, ..., 4\}$, because these vanish on $Q_{d(x)+2h}^h(x) \setminus [-h, 1+h]^4$ and hence in particular on a face of $Q_{d(x)+2h}^h(x)$. Thus we obtain

$$\begin{aligned} \|u_{h}\|_{L^{2}_{h}(Q^{h}_{d(x)+2h}(x))} &\leq C(d(x)+2h) \|\nabla_{h}u_{h}\|_{L^{2}_{h}(Q^{h}_{d(x)+2h}(x))} \\ &\leq C(d(x)+2h)^{2} \|\nabla^{2}_{h}u_{h}\|_{L^{2}_{h}(Q^{h}_{d(x)+2h}(x))} \,. \end{aligned}$$
(4.2.4)

If we combine this with (4.2.3) and note that $d(x) + 2h \le 2(d(x) + h)$, we obtain

$$\|v_h\|_{W_h^{2,2}((h\mathbb{Z})^4)} \le C \|u_h\|_{W_h^{2,2}((h\mathbb{Z})^4)}.$$
(4.2.5)

Furthermore, an argument analogous to the one that led to (4.2.4) shows that

$$\|v_h\|_{L^2_h((h\mathbb{Z})^4)} \le C(d(x) + h)^2 \|\nabla_h^2 v_h\|_{L^2_h((h\mathbb{Z})^4)}.$$
(4.2.6)

Now we are in a position to apply discrete Fourier analysis, similar to the proof of [Kur09, Proposition B.1]. Let

$$\widehat{v_h}(\xi) = h^4 \sum_{y \in (h\mathbb{Z})^4} v_h(y) e^{iy\cdot\xi}$$

for any $\xi \in \left[-\frac{\pi}{h}, \frac{\pi}{h}\right]^4$ be the Fourier transform of v_h . Then we also have the inverse formula

$$v_h(z) = \frac{1}{(2\pi)^4} \int_{\left[-\frac{\pi}{h}, \frac{\pi}{h}\right]^4} \widehat{v}_h(\xi) e^{-iz \cdot \xi} \, \mathrm{d}\xi$$

for any $z \in (h\mathbb{Z})^4$, and Plancherel's formula in the form

$$\int_{\left[-\frac{\pi}{h},\frac{\pi}{h}\right]^4} |\widehat{v_h}(\xi)|^2 \, \mathrm{d}\xi = (2\pi h)^4 \sum_{y \in (h\mathbb{Z})^4} |v_h(y)|^2 = (2\pi)^4 ||v_h(y)||^2_{L^2_h((h\mathbb{Z})^4)}.$$

We have

$$\widehat{D_{\alpha}^{h}v_{h}}(\xi) = (e^{-ih\xi_{1}}-1)^{\alpha_{1}}\dots(e^{-ih\xi_{4}}-1)^{\alpha_{4}}\widehat{v}_{h}(\xi)$$

for any $\alpha \in \mathbb{N}^4$. This implies

$$\left|\widehat{D_{\alpha}^{h}v_{h}}(\xi)\right| \geq \frac{1}{C}|\xi_{1}|^{\alpha_{1}}\dots|\xi_{4}|^{\alpha_{4}}|\widehat{v_{h}}(\xi)|$$

for any $\xi \in \left[-\frac{\pi}{h}, \frac{\pi}{h}\right]^4$. In combination with Plancherel's formula and (4.2.6) we conclude

$$\int_{\left[-\frac{\pi}{h},\frac{\pi}{h}\right]^4} |\xi|^4 |\hat{v}_h(\xi)|^2 \le C \|\nabla_h^2 v_h\|_{L^2_h((h\mathbb{Z})^4)}^2 \le C \|v_h\|_{W^{2,2}_h((h\mathbb{Z})^4)}^2,$$
(4.2.7)

$$\int_{\left[-\frac{\pi}{h},\frac{\pi}{h}\right]^4} |\widehat{v}_h(\xi)|^2 \le C \|v_h\|_{L^2_h((h\mathbb{Z})^4)}^2 \le C(d(x)+h)^4 \|v_h\|_{W^{2,2}_h((h\mathbb{Z})^4)}^2.$$
(4.2.8)

Next, we estimate

$$\begin{split} |v_{h}(x)| &= \frac{1}{(2\pi)^{4}} \left| \int_{\left[-\frac{\pi}{h}, \frac{\pi}{h}\right]^{4}} \widehat{v_{h}}(\xi) e^{-ix \cdot \xi} \, \mathrm{d}\xi \right| \\ &\leq C \int_{\left[-\frac{\pi}{h}, \frac{\pi}{h}\right]^{4}} |\widehat{v_{h}}(\xi)| \, \mathrm{d}\xi \\ &\leq C \left(\int_{\left[-\frac{\pi}{h}, \frac{\pi}{h}\right]^{4}} \left(|\xi|^{4} + \frac{1}{(d(x) + h)^{4}} \right) |\widehat{v_{h}}(\xi)|^{2} \, \mathrm{d}\xi \right)^{\frac{1}{2}} \\ &\times \left(\int_{\left[-\frac{\pi}{h}, \frac{\pi}{h}\right]^{4}} \left(|\xi|^{4} + \frac{1}{(d(x) + h)^{4}} \right)^{-1} \, \mathrm{d}\xi \right)^{\frac{1}{2}}. \end{split}$$

Using (4.2.7) and (4.2.8) we see that

$$\int_{\left[-\frac{\pi}{h},\frac{\pi}{h}\right]^4} \left(|\xi|^4 + \frac{1}{(d(x)+h)^4} \right) |\widehat{v}_h(\xi)|^2 \, \mathrm{d}\xi \le C \|v_h\|_{W_h^{2,2}((h\mathbb{Z})^4)}^2.$$

Furthermore we can compute using polar coordinates that

$$\begin{split} \int_{\left[-\frac{\pi}{h},\frac{\pi}{h}\right]^4} \left(|\xi|^4 + \frac{1}{(d(x)+h)^4} \right)^{-1} \mathrm{d}\xi &= \int_{\left[-\frac{\pi}{h},\frac{\pi}{h}\right]^4} \frac{(d(x)+h)^4}{1 + (d(x)+h)^4 |\xi|^4} \, \mathrm{d}\xi \\ &\leq C \int_0^{\frac{2\pi}{h}} \frac{(d(x)+h)^4 s^3}{1 + (d(x)+h)^4 s^4} \, \mathrm{d}s \\ &\leq C \log\left(1 + (d(x)+h)^4 \left(\frac{2\pi}{h}\right)^4\right) \\ &\leq C \log\left(2 + \frac{d(x)}{h}\right) \, . \end{split}$$

Putting everything together we indeed arrive at

$$egin{aligned} |u_h(x)| &= |v_h(x)| \ &\leq C \sqrt{\log\left(2+rac{d(x)}{h}
ight)} \|v_h\|_{W^{2,2}_h(h\mathbb{Z})^4)} \ &\leq C \sqrt{\log\left(2+rac{d(x)}{h}
ight)} \|u_h\|_{W^{2,2}_h((h\mathbb{Z})^4)} \,. \end{aligned}$$

4.2.2 Estimates for finite difference schemes

Let us discuss next the estimate for the approximation order of finite difference schemes that was already mentioned in the introduction.

To state it we need some definitions. These definition will be discussed in more detail in Chapter 5. For $j \ge 1$ let θ_j be the standard univariate centred B-spline of degree j - 1 (cf.

[JS14, Section 1.9.4]). Of interest to us are

$$\theta_{3}(z) := \begin{cases} \frac{3}{4} - z^{2} & |z| \leq \frac{1}{2} \\ \frac{1}{2} \left(|z| - \frac{3}{2} \right)^{2} & \frac{1}{2} < |z| \leq \frac{3}{2} \\ 0 & \text{else} \end{cases},$$
$$\theta_{1}(z) := \begin{cases} 1 & |z| \leq \frac{1}{2} \\ 0 & \text{else} \end{cases}.$$

Using this, we can define the smoothing operator $T_i^{h,j}$ for $1 \le i \le 4$ as

$$T_i^{h,j}f(x) := \frac{1}{h} \int_{\mathbb{R}} f(x_1, \dots, x_{i-1}, y_i, x_{i+1}, \dots, x_4) \theta_j\left(\frac{x_i - y_i}{h}\right) \, \mathrm{d}y_i$$

extended to distributions on \mathbb{R}^4 in the obvious way. Furthermore, we set

$$T^{h,j,\ldots,j}f := T_1^{h,j} \circ \cdots \circ T_4^{h,j}f.$$

It is important for us that $T_i^{h,j}$ maps constant functions to themselves and that

$$T_i^{h,j}\partial_i^2 f = D_i^h D_{-i}^h T_i^{h,j-2} f.$$

If we define the shorthand

$$T^{h,3,3,3,3-2e_i} := T_1^{h,3} \circ \ldots \circ T_{i-1}^{h,3} \circ T_i^{h,1} \circ T_{i+1}^{h,3} \circ \ldots \circ T_4^{h,3}$$

we also have

$$T^{h,3,3,3,3}\partial_i^2 f = D^h_i D^h_{-i} T^{h,3,3,3,2-2e_i} f.$$
(4.2.9)

Theorem 4.2.3. Let $2 < s < \frac{5}{2}$, let $u \in W_0^{s,2}((0,1)^4)$, extended by 0 to $\tilde{u} \in W^{s,2}(\mathbb{R}^4)$. Let $\Delta^2 \tilde{u} = f$ as distributions, so that in particular

$$\Delta^2 u = f \qquad in \ (0,1)^4 \,.$$

Furthermore, let $u_h: (h\mathbb{Z})^4 \to \mathbb{R}$ be the solution of

$$\begin{split} \Delta_h^2 u_h &= T^{h,3,3,3,3} f \qquad in \ \Lambda_h \\ u_h &= 0 \qquad on \ (h\mathbb{Z})^4 \setminus \Lambda_h \,. \end{split}$$

Then we have

$$\|u_h - \tilde{u}\|_{W^{2,2}_h(h\mathbb{Z})^4)} \le C_s h^{s-2} \|u\|_{W^{s,2}((0,1)^4)}$$

Note that $f = \Delta^2 \tilde{u} \in W^{s-4,2}(\mathbb{R}^4)$ is in a negative Sobolev space. The operator $T^{h,3,3,3,3}$ maps $W^{t,2}(\mathbb{R}^4)$ to $C(\mathbb{R}^4)$ for any $t > -\frac{5}{2}$ (see [JS14, Section 1.9.4]). So in particular $T^{h,3,3,3,3}f$ has pointwise values and the difference scheme in Theorem 4.2.3 makes sense.

This theorem is closely related to Theorem 5.1.2 in Chapter 5. In that theorem one takes $\frac{5}{2} < s \leq 3$, and $T^{h,3,3,3,3}$ is replaced by $T^{h,2,2,2,2}$. The novelty of that chapter lies in choosing a good extension \tilde{u} and dealing with its boundary values. In our case we can just extend u by 0 and thereby avoid many of these subtleties. In fact, all the ideas for the proof of Theorem 4.2.3 are already for example in [JS14].

To make this chapter more self-contained we give some details for a proof of Theorem 4.2.3.

Proof of Theorem 4.2.3. First of all, $s < \frac{5}{2}$ and $u \in W_0^{s,2}((0,1)^4)$ imply that \tilde{u} is actually in $W^{s,2}(\mathbb{R}^4)$ and $\|\tilde{u}\|_{W^{s,2}(\mathbb{R}^4)} = \|u\|_{W^{s,2}((0,1)^4)}$.

Let $e_h: (h\mathbb{Z})^4 \to \mathbb{R}$ be given by $e_h = \tilde{u} - u_h$. Then,

$$\begin{split} \Delta_h^2 e_h &= \Delta_h^2 \tilde{u} - \Delta_h^2 u_h = \Delta_h^2 \tilde{u} - T^{h,3,3,3,3} \Delta^2 \tilde{u} \qquad \text{on } \Lambda_h \\ e_h &= 0 \qquad \qquad \text{on } (h\mathbb{Z})^4 \setminus \Lambda_h \end{split}$$

and by summation by parts we have

$$\|\nabla_{h}^{2}e_{h}\|_{L^{2}_{h}((h\mathbb{Z})^{4})}^{2} = (e_{h}, \Delta_{h}^{2}e_{h})_{L^{2}_{h}((h\mathbb{Z})^{4})} = (e_{h}, \Delta_{h}^{2}\tilde{u} - T^{h,3,3,3}\Delta^{2}\tilde{u})_{L^{2}_{h}((h\mathbb{Z})^{4})}.$$
(4.2.10)

We can rewrite $\Delta_h^2 \tilde{u} - T^{h,3,3,3} \Delta^2 \tilde{u}$ using (4.2.9) as

$$\begin{split} \Delta_{h}^{2} \tilde{u} - T^{h,3,3,3,3} \Delta^{2} \tilde{u} &= \sum_{i=1}^{4} D_{i}^{h} D_{-i}^{h} \Delta_{h} \tilde{u} - T^{h,3,3,3,3} \partial_{i}^{2} \Delta \tilde{u} \\ &= \sum_{i=1}^{4} D_{i}^{h} D_{-i}^{h} \Delta_{h} \tilde{u} - D_{i}^{h} D_{-i}^{h} T^{h,3,3,3,3-2e_{i}} \Delta \tilde{u} \\ &= \sum_{i=1}^{4} D_{i}^{h} D_{-i}^{h} g_{i} \end{split}$$

where

$$g_i := \Delta_h \tilde{u} - T^{h,3,3,3,3-2e_i} \Delta \tilde{u} \,.$$

We can insert this into (4.2.10) and use summation-by-parts once again to obtain

$$\begin{split} \|\nabla_{h}^{2}e_{h}\|_{L^{2}_{h}((h\mathbb{Z})^{4})}^{2} &= \sum_{i=1}^{4} (e_{h}, D_{i}^{h}D_{-i}^{h}g_{i})_{L^{2}_{h}((h\mathbb{Z})^{4})} \\ &= \sum_{i=1}^{4} (D_{i}^{h}D_{-i}^{h}e_{h}, g_{i})_{L^{2}_{h}((h\mathbb{Z})^{4})} \\ &\leq \sum_{i=1}^{4} \|g_{i}\|_{L^{2}_{h}((h\mathbb{Z})^{4})} \|\nabla_{h}^{2}e_{h}\|_{L^{2}_{h}((h\mathbb{Z})^{4})} \end{split}$$

and thus

$$\|\nabla_{h}^{2}e_{h}\|_{L^{2}_{h}((h\mathbb{Z})^{4})} \leq \sum_{i=1}^{4} \|g_{i}\|_{L^{2}_{h}((h\mathbb{Z})^{4})}.$$
(4.2.11)

The summands on the right hand side can be bounded using the Bramble–Hilbert lemma (see e.g. [JS14, Theorem 2.28]): As s > 2,

$$|\Delta_h \tilde{u}(x)| \le C_h \|\tilde{u}\|_{L^{\infty}(x+(-3h/2,3h/2)^4)} \le C_{h,s} \|\tilde{u}\|_{H^s(x+(-3h/2,3h/2)^4)}.$$

Because $s > \frac{3}{2}$ and $T^{h,3,3,3-2e_i}f(x)$ only depends on $f|_{x+(-3h/2,3h/2)^4}$ we can conclude from [JS14, Theorem 1.67] and the locality of $T^{h,3,3,3-2e_i}$ that

$$|T^{h,3,3,3,3-2e_i}\Delta \tilde{u}(x)| \le C_{h,s} \|\tilde{u}\|_{H^s(x+(-3h/2,3h/2)^4)}$$

Thus $g_i(x)$ is a bounded linear functional of $\tilde{u} \in W^{s,2}(x + (-3h/2, 3h/2)^4)$. This functional vanishes when $\tilde{u}|_{x+(-3h/2, 3h/2)^4}$ is a polynomial of degree at most 2. Indeed, if that is the

case then $\Delta \tilde{u}|_{x+(-3h/2,3h/2)^4}$ is a constant function, and $\Delta_h \tilde{u}(x)$ is equal to the same constant, and the claim follows from the fact that $T_1^{h,3} \dots T_{i-1}^{h,3} T_i^{h,1} T_{i+1}^{h,3} \dots T_4^{h,3}$ maps constant functions to themselves.

We have shown that $g_i(x)$ is a bounded linear functional of $\tilde{u} \in W^{s,2}(x + (-3h/2, 3h/2)^4)$ that vanishes on polynomials of degree at most 2. By the Bramble–Hilbert lemma it is bounded by $C_{h,s}[\tilde{u}]_{W^{s,2}(x+(-3h/2,3h/2)^4)}$ for $s \leq 3$. Using a scaling argument to determine the correct prefactor of h, we obtain

$$|g_i(x)| \le C_s h^{s-4} [\tilde{u}]_{W^{s,2}(x+(-3h/2,3h/2)^4)}$$

and hence

$$\|g_i\|_{L^2_h((h\mathbb{Z})^4)}^2 \leq Ch^4 \sum_{x \in (h\mathbb{Z})^4} h^{2(s-4)} [\tilde{u}]_{W^{s,2}(x+(-3h/2,3h/2)^4)}^2$$

$$\leq C_s h^{2(s-2)} [\tilde{u}]_{W^{s,2}(\mathbb{R}^4)}^2 \leq C_s h^{2(s-2)} \|u\|_{W^{s,2}((0,1)^4)}^2$$

$$(4.2.12)$$

for those s. Now we can plug (4.2.12) into (4.2.11) and obtain

$$\|\nabla_h^2 e_h\|_{L^2_h((h\mathbb{Z})^4)} \le C_s h^{s-2} \|u\|_{W^{s,2}((0,1)^4)}$$

for $s < \frac{5}{2}$. Using the discrete Poincaré inequality completes the proof.

4.3 Estimates for other Green's functions

4.3.1 Estimates for the discrete Green's function of the full space

Our strategy will be to compare G_h with several other Green's functions, so let us introduce these first.

Recall that $\lambda = \sqrt{8\pi}$. Let *G* be the Green's function of the continuous Bilaplacian on $[0,1]^4$ with Dirichlet boundary data (i.e. of the problem (4.1.3)). We also need Green's functions on the full space. Let $\hat{G}(x,y) := -\frac{1}{\lambda^2} \log |x-y|$. It is easy to check that this is a fundamental solution of the Bilaplacian (i.e. that $\Delta^2 \left(-\frac{1}{\lambda^2} \log |\cdot-y|\right) = \delta_y$ in the sense of distributions). We also define \hat{G}_h : $(h\mathbb{Z})^4 \times (h\mathbb{Z})^4 \to \mathbb{R}$ by $\hat{G}_h(x,y) = F\left(\frac{x-y}{h}\right) - \frac{1}{\lambda^2} \log h$ where *F* is the function introduced in the following lemma. We added the summand $-\frac{1}{\lambda^2} \log h$ here to ensure that \hat{G}_h has the same asymptotic behaviour as \hat{G} . We also define shifted versions of \hat{G}_h and \hat{G} , namely for r > 0 we let $\hat{G}^{(r)} = \hat{G} + \frac{\log r}{\lambda^2}$, and $\hat{G}_h^{(r)} = \hat{G}_h + \frac{\log r}{\lambda^2}$. We occasionally write G_y for $G(\cdot, y)$, and define $G_{h,y}$, \hat{G}_y , $\hat{G}_{h,y}$, $\hat{G}_y^{(r)}$ and $\hat{G}_{h,y}^{(r)}$ analogously.

Lemma 4.3.1 ([Man67, pp. 96-97]). There is a function $F: \mathbb{Z}^4 \to \mathbb{R}$ such that $\Delta_1^2 F(x) = \begin{cases} 1 & x = 0 \\ 0 & else \end{cases}$, satisfying the asymptotics

$$F(x) = -\frac{1}{8\pi^2} \log|x| + \frac{1}{24\pi^2} \frac{x_1^4 + x_2^4 + x_3^4 + x_4^4}{|x|^6} + O\left(\frac{1}{|x|^4}\right)$$

for $x \neq 0$.

In [Man67], *F* is defined using the discrete Fourier multiplier associated to Δ_1^2 . By expanding that multiplier into a Laurent series and computing the Fourier transform termwise it is possible to give asymptotic expansions to arbitrary high order. This technique also applies to other discrete polyharmonic Green's functions. For our purposes the first two terms quoted above are sufficient.

Lemma 4.3.1 immediately gives us an asymptotic expansion of \hat{G}_h , and so we can easily obtain estimates for \hat{G}_h and $\hat{G}_h^{(r)}$.

Lemma 4.3.2. Let h > 0, and $r \ge 192h$. Let $\alpha \in \mathbb{N}^4$ with $|\alpha| \le 2$. Then for any $x, y \in (h\mathbb{Z})^4$ with $\frac{r}{64} \le |x - y|_{\infty} \le 16r$ we have

$$\left|\hat{G}_{h}^{(r)}(x,y) - \frac{1}{\lambda^{2}}\log\left(\frac{r}{|x-y|+h}\right)\right| \le C,$$
(4.3.1)

$$\left| D^h_{\alpha} \hat{G}^{(r)}_{h,y}(x) \right| \le \frac{C}{r^{|\alpha|}}, \qquad (4.3.2)$$

$$\left| D^{h}_{\alpha} \hat{G}^{(r)}_{h,y}(x) - \partial^{\alpha} \hat{G}^{(r)}_{y}(x) \right| \le C \frac{h}{r^{|\alpha|+1}}.$$
(4.3.3)

Proof. By translation invariance we may assume y = 0. The definition of $\hat{G}_h^{(r)}$ implies that

$$\begin{split} \hat{G}_{h}^{(r)}(x,0) &= F\left(\frac{x}{h}\right) - \frac{1}{\lambda^{2}}\log h + \frac{1}{\lambda^{2}}\log r \\ &= -\frac{1}{\lambda^{2}}\log\frac{|x|}{h} + \frac{h^{2}}{24\pi^{2}}\frac{x_{1}^{4} + x_{2}^{4} + x_{3}^{4} + x_{4}^{4}}{|x|^{6}} + O\left(\frac{h^{4}}{|x|^{4}}\right) - \frac{1}{\lambda^{2}}\log h + \frac{1}{\lambda^{2}}\log r \\ &= \frac{1}{\lambda^{2}}\log\frac{r}{|x|} + \frac{h^{2}}{24\pi^{2}}\frac{x_{1}^{4} + x_{2}^{4} + x_{3}^{4} + x_{4}^{4}}{|x|^{6}} + O\left(\frac{h^{4}}{|x|^{4}}\right) \,. \end{split}$$

$$(4.3.4)$$

From this we immediately conclude (4.3.1) in the case $x \neq 0$. In case x = 0 we can directly use

$$\hat{G}_{h}^{(r)}(0,0) = F(0) + \frac{1}{\lambda^{2}}\log\frac{r}{h}$$

to obtain (4.3.1).

The explicit formula for \hat{G} reveals that

$$\left|\partial^{lpha}\hat{G}_{0}^{(r)}(x)
ight|=\left|\partial^{lpha}rac{1}{\lambda^{2}}\lograc{r}{|x|}
ight|\leqrac{C}{r^{|lpha|}}$$

if $\frac{r}{64} \leq |x|_{\infty}$, and thus (4.3.2) easily follows from (4.3.3).

For (4.3.3) we want to take discrete derivatives of each summand in (4.3.4) separately. If $g = O\left(\frac{h^4}{|\cdot|^4}\right)$ then $|D^h_{\alpha}g(x)| \le \frac{C}{h^{|\alpha|}}\frac{h^4}{|x|^4} = C\frac{h^{4-|\alpha|}}{|x|^4}$ so for $|\alpha| \le 2$ we can neglect the error term. Using Taylor's theorem we can see that

$$D^h_{\alpha}\left(\frac{1}{\lambda^2}\log\frac{r}{|x|} + \frac{h^2}{24\pi^2}\frac{x_1^4 + x_2^4 + x_3^4 + x_4^4}{|x|^6}\right) = \partial^{\alpha}\frac{1}{\lambda^2}\log\frac{r}{|x|} + O\left(\frac{h}{|x|^{|\alpha|+1}}\right) \,.$$

Note that we can avoid the singularity here because $|x| \ge \frac{r}{64} \ge 3h$. This easily implies (4.3.3).

4.3.2 Estimates for continuous Green's functions

We want to compare G and G_h . This is only useful if we also have estimates for G to begin with. We will derive such estimates in this section. The following estimates are far from optimal, but sufficient for our purposes.

We obviously have a well-posedness result for the Bilaplace equation in the energy space $W^{2,2}$. The following result states that the same holds true if we raise the regularity slightly.

Theorem 4.3.3. There exists $\kappa_0 > 0$ with the following property: Let $0 \le \kappa \le \kappa_0$. Then for each $f \in W^{-2+\kappa,2}((0,1)^4)$ there is a unique $u \in W^{2+\kappa,2} \cap W_0^{2,2}((0,1)^4)$ such that $\Delta^2 u = f$ in the sense of distributions, and we have the estimate

$$\|u\|_{W^{2+\kappa,2}((0,1)^4)} \le C_{\kappa} \|f\|_{W^{-2+\kappa,2}((0,1)^4)}$$
(4.3.5)

for a constant C_{κ} depending only on κ .

For convenience we will assume in the following that $\kappa_0 < \frac{1}{2}$, and fix such a κ_0 . Note that $W^{2+\kappa,2} \cap W^{2,2}_0((0,1)^4) = W^{2+\kappa,2}_0((0,1)^4)$ if $\kappa < \frac{1}{2}$.

Proof of Theorem 4.3.3. This is a special case e.g. of [MM13, Theorem 6.32], but for the convenience of the reader we give the short argument.

We begin with the case $\kappa = 0$. In that case we can test the weak form of $\Delta^2 u = f$ with u and obtain

$$\|\nabla^2 u\|_{L^2((0,1)^4)}^2 = (u, \Delta^2 u)_{L^2((0,1)^4)} = (u, f)_{L^2((0,1)^4)} \le \|u\|_{W^{2,2}((0,1)^4)} \|f\|_{W^{-2,2}((0,1)^4)}.$$

The Poincaré inequality implies $||u||_{W^{2,2}((0,1)^4)} \leq C ||\nabla^2 u||_{L^2((0,1)^4)}$ and so we obtain (4.3.5).

For the general case we can use a stability result for analytic families of operators on Banach spaces: The spaces $W^{s,2}((0,1)^4)$ and $W^{s,2}_0((0,1)^4)$ each form an interpolation family with respect to complex interpolation, and so by [TVV88, Proposition 4.1] the set of those *s* for which Δ^2 : $W^{s,2}_0((0,1)^4) \rightarrow W^{s-4,2}((0,1)^4)$ has a bounded inverse is open. We know that this set contains 2, so the existence of κ_0 as in the theorem follows.

Next we state some estimates for *G*. We begin by estimating the regular part of *G* in certain Sobolev norms. Recall that $\hat{G}^{(r)}(x,y) = \hat{G}(x,y) + \frac{\log r}{\lambda^2}$ for any r > 0.

Lemma 4.3.4. Let κ_0 be as in Theorem 4.3.3, and let $0 \le \kappa \le \kappa_0$. Let $K \ge 2$, r > 0, $y \in (0,1)^4$ be such that $\frac{d(y)}{K} \le r \le \frac{d(y)}{2}$. Then

$$\left\| G_{y} - \eta_{y}^{(r)} \hat{G}_{y}^{(r)} \right\|_{W^{2+\kappa,2}((0,1)^{4})} \le \frac{C_{K,\kappa}}{r^{\kappa}}$$
(4.3.6)

for a constant $C_{K,\kappa}$ depending only on K and κ .

Proof. Let $H^{(r)} = G_y - \eta_y^{(r)} \hat{G}_y^{(r)}$. By Theorem 4.3.3 it suffices to show

$$\|\Delta^2 H^{(r)}\|_{W^{-2+\kappa,2}((0,1)^4)} \le \frac{C_{K,\kappa}}{r^{\kappa}}.$$
(4.3.7)

By standard interpolation theory and our assumption $\kappa \in [0, \kappa_0] \subset [0, 2]$ it suffices to establish this for $\kappa \in \{0, 2\}$.

Observe that $\Delta^2 H^{(r)}$ is zero in $(0, 1)^4 \setminus B_r(y)$ as well as in $B_{r/2}(y)$ (as the two singularities cancel out). This means that $\Delta^2 H^{(r)}$ is supported in $B_r(y) \setminus B_{r/2}(y)$ and there it is equal to

 $-\Delta^2 \left(\eta_y^{(r)} \hat{G}_y^{(r)}\right)$. We have an explicit formula for $\hat{G}_y^{(r)}$, and so it is straightforward to check that $\left|\Delta^2 \left(\eta_y^{(r)} \hat{G}_y^{(r)}\right)\right|$ is bounded by $\frac{C_K}{r^4}$ on $B_r(y) \setminus B_{r/2}(y)$. This easily implies (4.3.7) for $\kappa = 2$.

For the case $\kappa = 0$ we need to be slightly more careful: Let $\chi_y^{(r)}$ be a cut-off function that is 1 on $B_r(y) \setminus B_{r/2}(y)$ and zero outside $B_{2r}(y) \setminus B_{r/4}(y)$ (e.g. $\chi_y^{(r)} = \eta_y^{(2r)} - \eta_y^{(r/2)}$). Then we have $\Delta^2 H^{(r)} = -\chi_y^{(r)} \Delta^2 \left(\eta_y^{(r)} \hat{G}_y^{(r)} \right)$ and thus we can calculate

$$\begin{split} \|\Delta^{2}H^{(r)}\|_{W^{-2,2}((0,1)^{4})} &= \sup_{\|\varphi\|_{W^{2,2}_{0}((0,1)^{4})}=1} \int \Delta^{2}H^{(r)}\varphi \\ &= \sup_{\|\varphi\|_{W^{2,2}_{0}((0,1)^{4})}=1} \int -\Delta^{2} \left(\eta_{y}^{(r)}\hat{G}_{y}^{(r)}\right) \chi_{y}^{(r)}\varphi \\ &= \sup_{\|\varphi\|_{W^{2,2}_{0}((0,1)^{4})}=1} \int -\Delta \left(\eta_{y}^{(r)}\hat{G}_{y}^{(r)}\right) \Delta(\chi_{y}^{(r)}\varphi) \\ &\leq C \left\|\Delta \left(\eta_{y}^{(r)}\hat{G}_{y}^{(r)}\right)\right\|_{L^{2}(B_{2r}(y)\setminus B_{r/4}(y))} \sup_{\|\varphi\|_{W^{2,2}_{0}((0,1)^{4})}=1} \|\Delta(\chi_{y}^{(r)}\varphi)\|_{L^{2}((0,1)^{4})}. \end{split}$$

To estimate the second factor we proceed as in the calculation that led to (4.2.5). We have a Poincaré inequality

$$\|u\|_{L^{2}(z+(-s,s)^{4})} \leq Cs \|\nabla u\|_{L^{2}(z+(-s,s)^{4})}$$
(4.3.8)

for any $u \in W^{1,2}(z + (-s,s)^4)$ that is zero (in the sense of traces) on one of the faces of $z + (-s,s)^4$. This is the continuous analogue to Lemma 4.2.1, and the proof is very similar. Using (4.3.8) we can estimate

$$\begin{split} \|\Delta(\chi_{y}^{(r)}\varphi)\|_{L^{2}((0,1)^{4})} &\leq C \|\nabla^{2}\varphi\|_{L^{2}(B_{d(y)}(y))} + \frac{C}{r} \|\nabla\varphi\|_{L^{2}(B_{d(y)}(y))} + \frac{C}{r^{2}} \|\varphi\|_{L^{2}(B_{d(y)}(y))} \\ &\leq C \left(1 + \frac{d(y)}{r} + \frac{d(y)^{2}}{r^{2}}\right) \|\nabla^{2}\varphi\|_{L^{2}(y + (-d(y), d(y))^{4})} \\ &\leq C_{K} \|\varphi\|_{W_{0}^{2,2}((0,1)^{4})} \,. \end{split}$$

We also have that $\Delta\left(\eta_y^{(r)}\hat{G}_y^{(r)}\right)$ is bounded by $\frac{C}{r^2}$ on $B_{2r}(y) \setminus B_{r/4}(y)$ and hence

$$\left\|\Delta\left(\eta_y^{(r)}\hat{G}_y^{(r)}\right)\right\|_{L^2(B_{2r}(y)\setminus B_{r/4}(y))} \le Cr^2 \cdot \frac{1}{r^2} = C$$

Using this we obtain (4.3.7) for $\kappa = 0$.

Next we give some estimates on the local behaviour of *G*. The first two allow us to control *G* far from and close to the singularity, respectively, while the last one expresses the Hölder continuity of $G - \hat{G}$ near the diagonal.

Lemma 4.3.5. Let κ_0 be as in Theorem 4.3.3. Let $y \in (0,1)^4$. The function G_y is smooth on $(0,1)^4 \setminus \{y\}$, and $G - \hat{G}$ is symmetric and smooth on $(0,1)^4 \times (0,1)^4 \setminus \{(x,x) : x \in (0,1)^4\}$ and can be extended continuously to $(0,1)^4 \times (0,1)^4$. Slightly abusing notation, we write

$$G(y,y) - \hat{G}(y,y) := \lim_{\substack{(y',y'') \to (y,y) \\ y' \neq y''}} G(y',y'') - \hat{G}(y',y'') \,.$$

Let $K \ge 1$ *. We have the following estimates, where* $\frac{d(y)}{K} \le r \le \frac{d(y)}{2}$ *:*

$$|G(x,y)| \le C \quad if |x-y| \ge \frac{d(y)}{4},$$
 (4.3.9)

$$|G(x,y) - \hat{G}^{(r)}(x,y)| \le C_K \quad \text{if } |x-y| \le d(y).$$
(4.3.10)

Furthermore if r > 0 *is arbitrary,* $|y' - y| \le \frac{d(y)}{8}$ *and* $|y'' - y| \le \frac{d(y)}{8}$ *we have the estimate*

$$\left| G(y',y'') - \hat{G}^{(r)}(y',y'') - \left(G(y,y) - \hat{G}^{(r)}(y,y) \right) \right| \le C \frac{|y'-y|^{\kappa_0} + |y''-y|^{\kappa_0}}{d(y)^{\kappa_0}} \,. \tag{4.3.11}$$

Proof. The smoothness of *G* and $G - \hat{G}$ follows from standard regularity theory for higher order elliptic equations. The estimate (4.3.9) is given in [MM14, Theorem 8.1]. There also a variant of (4.3.10) (without the correction $\frac{\log r}{\lambda^2}$ and with slightly worse error term) is given. The results in [MM14] however are in a far more general setting, so we prefer to give an elementary proof of the specific estimates we need.

We use a standard Caccioppoli inequality (see e.g. [Cam80, Capitolo II, Teorema 3.II or Teorema 6.I]): If $u \in W^{2,2}(B_s(z))$ and $\Delta^2 u = 0$ in $B_s(z)$ then

$$\|\nabla^2 u\|_{L^{\infty}(B_{s/2}(z))} \le \frac{C}{s^2} \|\nabla^2 u\|_{L^2(B_s(z))}.$$
(4.3.12)

We will also need a special case of the Gagliardo-Nirenberg interpolation inequality, namely

$$\|u\|_{L^{\infty}(B_{s}(z))} \leq C\left(s^{2}\|\nabla^{2}u\|_{L^{\infty}(B_{s}(z))} + \frac{1}{s^{2}}\|u\|_{L^{2}(B_{s}(z))}\right).$$
(4.3.13)

To see this, observe first that by scaling we can assume s = 1. The Poincaré inequality implies that $||u - a - b \cdot (\cdot - z)||_{L^{\infty}(B_1(z))} \leq C ||\nabla^2 u||_{L^{\infty}(B_1(z))}$, where $a = \frac{1}{|B_1|} \int u$ and $b = \frac{1}{|B_1|} \int \nabla u$, and so we only have to bound *a* and *b*. We have $|a| \leq C ||u||_{L^2(B_1(z))}$, and the estimate $||u - a||_{L^2(B_1(z))} \leq ||u||_{L^2(B_1(z))}$ implies

$$\begin{split} b| &\leq C \| b \cdot (\cdot - z) \|_{L^2(B_1(z))} \\ &\leq C \left(\| u - a - b \cdot (\cdot - z) \|_{L^2(B_1(z))} + \| u - a \|_{L^2(B_1(z))} \right) \\ &\leq C \left(\| \nabla^2 u \|_{L^\infty(B_1(z))} + \| u \|_{L^2(B_1(z))} \right) \,. \end{split}$$

This completes the proof of (4.3.13).

After these preparations we can now begin with the proof of (4.3.9). We first assume that $d(x) \leq 2d(y)$. Let $H^{(d(y)/8)} = G_y - \eta_y^{(d(y)/8)} \hat{G}_y^{(d(y)/8)}$. Lemma 4.3.4 with $\kappa = 0$ implies that

$$\|\nabla^2 H^{(d(y)/8)}\|_{L^2((0,1)^4)} \le C.$$
(4.3.14)

The function $H^{(d(y)/8)}$ agrees with G_y on $(0,1)^4 \setminus B_{d(y)/8}(y)$. Because $\frac{d(x)}{16} + \frac{d(y)}{8} \le \frac{d(y)}{4} \le |x-y|_{\infty}$ we have $B_{d(x)/16}(x) \cap B_{d(y)/8}(y) = \emptyset$ and thus (4.3.14) implies

$$\|\nabla^2 G_y\|_{L^2(B_{d(x)/16}(x))} \leq C.$$

Using the Caccioppoli inequality (4.3.12) we conclude

$$\|\nabla^2 G_y\|_{L^{\infty}(B_{d(x)/32}(x))} \le \frac{C}{d(x)^2}.$$
(4.3.15)

Next, note that the Poincaré inequality (4.3.8) applied on $x + (-d(x), d(x))^4$ and (4.3.14) imply that

$$\|H^{(d(y)/8)}\|_{L^{2}(B_{d(x)}(x))} \leq Cd(x)^{2} \|\nabla^{2} H^{(d(y)/8)}\|_{L^{2}(x+(-d(x),d(x))^{4})} \leq Cd(x)^{2}$$

and therefore

$$||G_y||_{L^2(B_{d(x)/32}(x))} \le Cd(x)^2$$

Recalling (4.3.15), an application of (4.3.13) concludes the proof.

It remains to consider the case d(x) > 2d(y). In that case $|x - y| \ge d(x) - d(y) \ge \frac{d(x)}{2}$, so we can interchange the roles of *x* and *y* and repeat the above proof (using that G(x, y) = G(y, x)).

Next we give a proof of (4.3.10). This is quite similar to the preceding argument. Because $G^{(r)}$ differs from $G^{(d(y))}$ only by at most $\frac{1}{\lambda^2} \log K \leq C_K$ we can assume r = d(y). Let again $H^{(d(y))} = G_y - \eta_y^{(d(y))} \hat{G}_y^{(d(y))}$. Observe first that if $|x - y| \geq \frac{d(y)}{4}$ then (4.3.9) implies (4.3.10). Therefore we can restrict our attention to the case $|x - y| \leq \frac{d(y)}{4}$. By Lemma 4.3.4 we have that

$$\|\nabla^2 H^{(d(y))}\|_{L^2((0,1)^4)} \leq C$$
.

The function $H^{(d(y))}$ agrees with $G_y - \hat{G}_y^{(d(y))}$ on $B_{d(y)/2}(y)$. Thus, as before, the Caccioppoli inequality implies that

$$\|\nabla^2 (G_y - \hat{G}_y)\|_{L^{\infty}(B_{d(y)/4}(y))} \le \frac{C}{d(y)^2}$$

and the Poincaré inequality implies

$$\|G_y - \hat{G}_y\|_{L^2(B_{d(y)/4}(y))} \le \|H^{(d(y))}\|_{L^2(B_{d(y)}(y))} \le Cd(y)^2$$

so that the conclusion follows from the interpolation inequality (4.3.13).

For (4.3.11) observe that by Lemma 4.3.4 we control the $W^{2+\kappa_0,2}$ -norm of $G_y - \eta_y^{(d(y))} \hat{G}_y^{(d(y))}$. That Sobolev space embeds into the Hölder space C^{0,κ_0} and so we have

$$\left[G_{y} - \eta_{y}^{(d(y))}\hat{G}_{y}^{(d(y))}\right]_{C^{0,\kappa_{0}}((0,1)^{4})} \leq C \left\|G_{y} - \eta_{y}^{(d(y))}\hat{G}_{y}^{(d(y))}\right\|_{W^{2+\kappa_{0},2}((0,1)^{4})} \leq \frac{C}{d(y)^{\kappa_{0}}}$$

Because $G_y - \eta_y^{(d(y))} \hat{G}_y^{(d(y))}$ agrees with $G_y - \hat{G}_y^{(d(y))}$ on $B_{d(y)/2}(y)$ this implies

$$\left| G(y',y) - \hat{G}^{(d(y))}(y',y) - \left(G(y,y) - \hat{G}^{(d(y))}(y,y) \right) \right| \le C \frac{|y'-y|^{\kappa_0}}{d(y)^{\kappa_0}}$$

If we add and subtract $\frac{\log r - \log d(y)}{\lambda^2}$ on the left-hand side we obtain

$$\left|G(y',y) - \hat{G}^{(r)}(y',y) - \left(G(y,y) - \hat{G}^{(r)}(y,y)\right)\right| \le C \frac{|y'-y|^{\kappa_0}}{d(y)^{\kappa_0}}.$$

Similarly we obtain

$$\left| G(y'',y') - \hat{G}^{(r)}(y'',y') - \left(G(y,y') - \hat{G}^{(r)}(y,y') \right) \right| \le C \frac{|y'' - y|^{\kappa_0}}{d(y')^{\kappa_0}}$$

where we used that $d(y') \ge \frac{7}{8}d(y)$ so that $y, y'' \in B_{d(y')/2}(y')$. If we add the last two estimates and use once again that $d(y') \ge \frac{7}{8}d(y)$ we arrive at (4.3.11).

4.4 Proof of the main theorems

In this section we will finally prove that G_h satisfies (B.0'), (B.1'), (B.2'), (B.3'), which according to Observation 4.1.5 implies Theorem 4.1.4.

Recall that G_h is the Green's function of the discrete Bilaplacian on Λ_h with zero boundary data outside Λ_h , G is the Green's function of the continuous Bilaplacian on $(0, 1)^4$ with zero Dirichlet boundary data, and $\hat{G}_h^{(r)}$ and $\hat{G}^{(r)}$ are shifted versions of the discrete and continuous full space Green's function.

The main technical statement used in the proof of Theorem 4.1.4 will be the following.

Lemma 4.4.1. Let κ_0 be as in Theorem 4.3.3. Let $K \ge 2$, and $r \ge 192h$. Then for all $x, y \in \Lambda_h$ with $\frac{d(y)}{K} \le r \le \frac{d(y)}{2}$ we have

$$\begin{split} \left| \left(G_h(x,y) - \eta_{h,y}^{(r)}(x) \hat{G}_h^{(r)}(x,y) \right) - \left(G(x,y) - \eta_y^{(r)}(x) \hat{G}^{(r)}(x,y) \right) \right| \\ &\leq C_K \frac{h^{\kappa_0}}{r^{\kappa_0}} \sqrt{\log\left(2 + \frac{d(x)}{h}\right)} \,. \end{split}$$

This lemma is so useful because it simultaneously provides control over the difference between the discrete and continuous Green's function when x, y are far apart and over the difference of the regular part of the discrete and continuous Green's function when x, y are close.

Proof of Lemma 4.4.1. We define $H_h = G_{h,y} - \eta_{h,y}^{(r)} \hat{G}_{h,y}^{(r)}$ and $H = G_y - \eta_y^{(r)} \hat{G}_y^{(r)}$. Let \tilde{H}_h be the solution of $\Delta_h^2 \tilde{H}_h = T^{h,3,3,3,3} \Delta^2 H$ in Λ_h $\tilde{H}_h = 0$ on $(h\mathbb{Z})^4 \setminus \Lambda_h$.

Our goal is to estimate $|H_h(x) - H(x)|$. We will estimate $H_h - \tilde{H}_h$ and $\tilde{H}_h - H$ separately.

The estimate of the latter term is straightforward: Using Theorem 4.2.3 and Lemma 4.3.4, we obtain

$$\|\tilde{H}_h - H\|_{W_h^{2,2}((h\mathbb{Z})^4)} \le C_K h^{\kappa_0} \|H\|_{W^{2+\kappa_0,2}((0,1)^4)} \le C_K \frac{h^{\kappa_0}}{r^{\kappa_0}}.$$

Estimating $H_h - \tilde{H}_h$ is more tedious. Similarly as in the proof of Lemma 4.3.4 we let $\chi_y^{(r)} = \eta_y^{(4r)} - \eta_y^{(r/4)}$ and $\chi_{h,y}^{(r)}$ be the restriction of $\chi_y^{(r)}$ to $(h\mathbb{Z})^4$. Then we have

$$\begin{split} \Delta_{h}^{2}(H_{h} - \tilde{H}_{h}) &= \Delta_{h}^{2} \left(G_{h,y} - \eta_{h,y}^{(r)} \hat{G}_{h,y}^{(r)} \right) - T^{h,3,3,3,3} \Delta^{2} \left(G_{y} - \eta_{y}^{(r)} \hat{G}_{y}^{(r)} \right) \\ &= \chi_{h,y}^{(r)} \Delta_{h}^{2} \left(G_{h,y} - \eta_{h,y}^{(r)} \hat{G}_{h,y}^{(r)} \right) - \chi_{h,y}^{(r)} T^{h,3,3,3,3} \Delta^{2} \left(G_{y} - \eta_{y}^{(r)} \hat{G}_{y}^{(r)} \right) \\ &= -\chi_{h,y}^{(r)} \Delta_{h}^{2} \left(\eta_{h,y}^{(r)} \hat{G}_{h,y}^{(r)} \right) + \chi_{h,y}^{(r)} T^{h,3,3,3,3} \Delta^{2} \left(\eta_{y}^{(r)} \hat{G}_{y}^{(r)} \right) \\ &= -\chi_{h,y}^{(r)} \Delta_{h}^{2} \left(\eta_{h,y}^{(r)} \hat{G}_{h,y}^{(r)} \right) + \sum_{i=1}^{4} \chi_{h,y}^{(r)} T^{h,3,3,3,3} \partial_{i}^{2} \Delta \left(\eta_{y}^{(r)} \hat{G}_{y}^{(r)} \right) \\ &= -\chi_{h,y}^{(r)} \Delta_{h}^{2} \left(\eta_{h,y}^{(r)} \hat{G}_{h,y}^{(r)} \right) + \sum_{i=1}^{4} \chi_{h,y}^{(r)} T^{h,3,3,3,3} \partial_{i}^{2} \Delta \left(\eta_{y}^{(r)} \hat{G}_{y}^{(r)} \right) \\ &= -\chi_{h,y}^{(r)} \sum_{i=1}^{4} D_{i}^{h} D_{-i}^{h} \left(\Delta_{h} \left(\eta_{h,y}^{(r)} \hat{G}_{h,y}^{(r)} \right) + T^{h,3,3,3,3-2e_{i}} \Delta \left(\eta_{y}^{(r)} \hat{G}_{y}^{(r)} \right) \right) \,. \end{split}$$

Because $H_h - \tilde{H}_h$ is supported in Λ_h we have

 $\|\nabla_{h}^{2}(H_{h}-\tilde{H}_{h})\|_{L^{2}_{h}((h\mathbb{Z})^{4})}^{2} = \left(\Delta_{h}^{2}(H_{h}-\tilde{H}_{h}), H_{h}-\tilde{H}_{h}\right)_{L^{2}_{h}((h\mathbb{Z})^{4})}$

$$\leq \sup_{\substack{\varphi_{h}=0 \text{ on } (h\mathbb{Z})^{4} \setminus \Lambda_{h} \\ \|\varphi_{h}\|_{W_{h}^{2,2}((h\mathbb{Z})^{4})} = 1}} \left(\Delta_{h}^{2}(H_{h} - \tilde{H}_{h}), \varphi_{h} \right)_{L_{h}^{2}((h\mathbb{Z})^{4}))} \|H_{h} - \tilde{H}_{h}\|_{W_{h}^{2,2}((h\mathbb{Z})^{4})}$$

which together with the Poincaré inequality implies that

$$\|H_{h} - \tilde{H}_{h}\|_{W_{h}^{2,2}((h\mathbb{Z})^{4})} \leq C \sup_{\substack{\varphi_{h} = 0 \text{ on } (h\mathbb{Z})^{4} \setminus \Lambda_{h} \\ \|\varphi_{h}\|_{W_{h}^{2,2}((h\mathbb{Z})^{4})} = 1}} \left(\Delta_{h}^{2}(H_{h} - \tilde{H}_{h}), \varphi_{h} \right)_{L_{h}^{2}((h\mathbb{Z})^{4})} \right).$$

Combining this with (4.4.1), and abbreviating $T_i^* := T^{h,3,3,3,3-2e_i}$ we see that

$$\begin{split} \|H_{h} - H_{h}\|_{W_{h}^{2,2}((h\mathbb{Z})^{4})} &\leq C \sup_{\substack{\varphi_{h} = 0 \text{ on } (h\mathbb{Z})^{4} \setminus \Lambda_{h} \\ \|\varphi_{h}\|_{W_{h}^{2,2}((h\mathbb{Z})^{4})} = 1}} \sum_{i=1}^{4} \left(D_{i}^{h} D_{-i}^{h} \left(-\Delta_{h} \left(\eta_{h,y}^{(r)} \hat{G}_{h,y}^{(r)} \right) + T_{i}^{*} \Delta \left(\eta_{y}^{(r)} \hat{G}_{y}^{(r)} \right) \right), \chi_{h,y}^{(r)} \varphi_{h} \right)_{L_{h}^{2}((h\mathbb{Z})^{4}))} \\ &\leq C \sup_{\substack{\varphi_{h} = 0 \text{ on } (h\mathbb{Z})^{4} \setminus \Lambda_{h} \\ \|\varphi_{h}\|_{W_{h}^{2,2}((h\mathbb{Z})^{4})} = 1}} \sum_{i=1}^{4} \left(-\Delta_{h} \left(\eta_{h,y}^{(r)} \hat{G}_{h,y}^{(r)} \right) + T_{i}^{*} \Delta \left(\eta_{y}^{(r)} \hat{G}_{y}^{(r)} \right), D_{i}^{h} D_{-i}^{h} \chi_{h,y}^{(r)} \varphi_{h} \right)_{L_{h}^{2}((h\mathbb{Z})^{4}))} \\ &\leq C \sum_{i=1}^{4} \left\| -\Delta_{h} \left(\eta_{h,y}^{(r)} \hat{G}_{h,y}^{(r)} \right) + T_{i}^{*} \Delta \left(\eta_{y}^{(r)} \hat{G}_{y}^{(r)} \right) \right\|_{L_{h}^{2}(Q_{8r}^{h}(y) \setminus Q_{r/32}^{h}(y))} \\ &\times \sup_{\substack{\varphi_{h} = 0 \text{ on } (h\mathbb{Z})^{4} \setminus \Lambda_{h} \\ \|\varphi_{h}\|_{W_{h}^{2,2}((h\mathbb{Z})^{4})} = 1}} \left\| \nabla_{h}^{2} \left(\chi_{h,y}^{(r)} \varphi_{h} \right) \right\|_{L_{h}^{2}((h\mathbb{Z})^{4}))}, \end{split}$$

(4.4.2) where we used that $\chi_{h,y}^{(r)}$ is supported in $B_{4r}(y) \setminus B_{r/8}(y)$ so that the support of $\Delta_h \left(\chi_{h,y}^{(r)} \varphi_h \right)$ is certainly contained in $Q_{8r}^h(y) \setminus Q_{r/32}^h(y)$. The discrete product rule and the Poincaré inequality imply that

$$\begin{split} \left\| \nabla_{h}^{2} \left(\chi_{h,y}^{(r)} \varphi_{h} \right) \right\|_{L_{h}^{2}((h\mathbb{Z})^{4}))} \\ &\leq C \| \nabla_{h}^{2} \varphi_{h} \|_{L_{h}^{2}(Q_{d(y)+h}^{h}(y))} + \frac{C}{r} \| \nabla_{h} \varphi_{h} \|_{L_{h}^{2}(Q_{d(y)+h}^{h}(y))} + \frac{C}{r^{2}} \| \varphi_{h} \|_{L_{h}^{2}(Q_{d(y)+h}^{h}(y))} \\ &\leq C \left(1 + \frac{d(y) + h}{r} + \frac{(d(y) + h)^{2}}{r^{2}} \right) \| \varphi_{h} \|_{W_{h}^{2,2}((h\mathbb{Z})^{4})} \\ &\leq C_{K} \| \varphi_{h} \|_{W_{h}^{2,2}((h\mathbb{Z})^{4})} \end{split}$$

and hence

$$\sup_{\substack{\varphi_{h}=0 \text{ on } (h\mathbb{Z})^{4} \setminus \Lambda_{h} \\ \|\varphi_{h}\|_{W_{h}^{2,2}((h\mathbb{Z})^{4})} = 1}} \left\| \nabla_{h}^{2} \left(\chi_{h,y}^{(r)} \varphi_{h} \right) \right\|_{L_{h}^{2}((h\mathbb{Z})^{4}))} \leq C_{K}.$$
(4.4.3)

Let us now also estimate the first factor in (4.4.2). The operator $T^{h,3,3,3,3-2e_i}$ preserves constant functions. Therefore for any z with $|z - y|_{\infty} \ge \frac{r}{32}$

$$\begin{split} & \left(T^{h,3,3,3,3-2e_{i}}\Delta\left(\eta_{y}^{(r)}\hat{G}_{y}^{(r)}\right)\right)(z) \\ &= \Delta\left(\eta_{y}^{(r)}\hat{G}_{y}^{(r)}\right)(z) + \left(T^{h,3,3,3,3-2e_{i}}\left(\Delta\left(\eta_{y}^{(r)}\hat{G}_{y}^{(r)}\right)(\cdot) - \Delta\left(\eta_{y}^{(r)}\hat{G}_{y}^{(r)}\right)(x)\right)\right)(z) \\ &= \Delta\left(\eta_{y}^{(r)}\hat{G}_{y}^{(r)}\right)(z) + O\left(h\sup_{z+(-\frac{3h}{2},\frac{3h}{2})}\left|\nabla^{3}\left(\eta_{y}^{(r)}\hat{G}_{y}^{(r)}\right)\right|\right) \\ &= \Delta\left(\eta_{y}^{(r)}\hat{G}_{y}^{(r)}\right)(z) + O\left(\frac{h}{r^{3}}\right), \end{split}$$
(4.4.4)

where we have used that $|T^{h,3,3,3-2e_i}f(z)| \leq C \sup_{z+(-3h/2,3h/2)} |f|$ in the second step as well as the explicit formula for $\hat{G}^{(r)}(z, y)$ in the third step. From Lemma 4.3.2 and Taylor's theorem we know that for $\frac{r}{64} \leq |z-y|_{\infty} \leq 16r$

$$\begin{split} D^h_{\alpha} \hat{G}^{(r)}_{h,y}(z) &= \partial^{\alpha} \hat{G}^{(r)}_{y}(z) + O\left(\frac{h}{r^{|\alpha|+1}}\right), \\ D^h_{\alpha} \hat{G}^{(r)}_{h,y}(z) &= O\left(\frac{1}{r^{|\alpha|}}\right), \\ D^h_{\alpha} \eta^{(r)}_{h,y}(z) &= \partial^{\alpha} \eta^{(r)}_{y}(z) + O\left(\frac{h}{r^{|\alpha|+1}}\right), \\ D^h_{\alpha} \eta^{(r)}_{h,y}(z) &= O\left(\frac{1}{r^{|\alpha|}}\right). \end{split}$$

If we combine these estimates with the discrete product rule we obtain that for any z with $\frac{r}{32} \leq |z-y|_\infty \leq 8r$

$$\Delta_h \left(\eta_{h,y}^{(r)} \hat{G}_{h,y}^{(r)} \right) (z) = \Delta \left(\eta_y^{(r)} \hat{G}_y^{(r)} \right) (z) + O\left(\frac{h}{r^3}\right) .$$

$$(4.4.5)$$

Combining (4.4.4) and (4.4.5) we find that

$$\left| -\Delta_h \left(\eta_{h,y}^{(r)} \hat{G}_{h,y}^{(r)} \right) + \sum_{i=1}^4 T^{h,3,3,3,3-2e_i} \Delta \left(\eta_y^{(r)} \hat{G}_y \right) \right| \le C \frac{h}{r^3}$$

on $Q^h_{8r}(y) \setminus Q^h_{r/32}(y)$ and therefore

$$\left\| -\Delta_h \left(\eta_{h,y}^{(r)} \hat{G}_{h,y}^{(r)} \right) + \sum_{i=1}^4 T^{h,3,3,3,3-2e_i} \Delta \left(\eta_y^{(r)} \hat{G}_y \right) \right\|_{L^2_h(Q^h_{8r}(y) \setminus Q^h_{r/32}(y))} \le C \frac{h}{r} \,.$$

If we use this result and (4.4.3) in (4.4.2) we see that

$$||H_h - \tilde{H}_h||_{W_h^{2,2}((h\mathbb{Z})^4)} \le C_K \frac{h}{r}.$$

In summary,

$$\|H_h - H\|_{W_h^{2,2}((h\mathbb{Z})^4)} \le \|H_h - \tilde{H}_h\|_{W_h^{2,2}((h\mathbb{Z})^4)} + \|\tilde{H}_h - H\|_{W_h^{2,2}((h\mathbb{Z})^4)}$$

$$\leq C_K\left(rac{h^{\kappa_0}}{r^{\kappa_0}}+rac{h}{r}
ight)\leq C_Krac{h^{\kappa_0}}{r^{\kappa_0}}$$

because $\frac{h}{r} \leq 1$. Finally, Lemma 4.2.2 allows us to conclude that for any $x \in (h\mathbb{Z})^4$

$$|H_h(x) - H(x)| \le C_K \sqrt{\log\left(2 + \frac{d(x)}{h}\right)} \|H_h - H\|_{W_h^{2,2}((h\mathbb{Z})^4)} \le C_K \frac{h^{\kappa_0}}{r^{\kappa_0}} \sqrt{\log\left(2 + \frac{d(x)}{h}\right)}.$$

his completes the proof.

This completes the proof.

Before we turn to the proof of Theorem 4.1.4 let us observe that Lemma 4.2.2 already implies an upper bound on $G_h(x, y)$.

Lemma 4.4.2. For any *x*, *y* we have that

$$|G_h(x,y)| \le C \sqrt{\log\left(2 + \frac{d(x)}{h}\right)\log\left(2 + \frac{d(y)}{h}\right)}.$$
(4.4.6)

Proof. The idea is the same as in the proof of Lemma 2.8.1. We have

$$G_h(x,y) = (G_{h,x}, \delta_{h,y})_{L^2_h((h\mathbb{Z})^4)} = (G_{h,x}, \Delta^2_h G_{h,y})_{L^2_h((h\mathbb{Z})^4)} = (\nabla^2_h G_{h,x}, \nabla^2_h G_{h,y})_{L^2_h((h\mathbb{Z})^4)}.$$

This implies on the one hand

$$|G_h(x,y)| \le \|\nabla_h^2 G_{h,x}\|_{L^2_h((h\mathbb{Z})^4)} \|\nabla_h^2 G_{h,y}\|_{L^2_h((h\mathbb{Z})^4)}$$
(4.4.7)

and on the other hand (by choosing y = x) that

$$|G_h(x,x)| = \|\nabla_h^2 G_{h,x}\|_{L^2_h((h\mathbb{Z})^4)}^2$$

From Lemma 4.2.2 we know that

$$|G_h(x,x)| \leq \sqrt{\log\left(2+rac{d(x)}{h}
ight)} \|
abla_h^2 G_{h,x}\|_{L^2_h((h\mathbb{Z})^4)}.$$

Combining the last two estimates we obtain

$$|G_h(x,x)| \le C \log\left(2 + \frac{d(x)}{h}\right)$$

which is (4.4.6) in the special case x = y. For the general case we can use (4.4.7) to see that

$$|G_h(x,y)| \le \|\nabla_h^2 G_{h,x}\|_{L^2_h((h\mathbb{Z})^4)} \|\nabla_h^2 G_{h,y}\|_{L^2_h((h\mathbb{Z})^4)} \le C \sqrt{\log\left(2 + \frac{d(x)}{h}\right)\log\left(2 + \frac{d(y)}{h}\right)} .$$

Now we can turn to the proof of the main technical result of this chapter, Theorem 4.1.4.

Proof of Theorem 4.1.4. Recall that according to Observation 4.1.5 we actually have to verify (B.0'), (B.1'), (B.2') and (B.3').

Step 1: Proof of (B.1')

Let $x, y \in (h\mathbb{Z})^4$. We can assume w.l.o.g. that $d(x) \leq d(y)$ (else interchange x and

y). If d(y) < 768h we have that $\left|\log\left(2 + \frac{\max(d(x), d(y))}{h + |x - y|}\right)\right| \le C$, and by Lemma 4.4.2 also $|G_h(x, y)| \le C$, so that (B.1') holds trivially. Thus we can assume $d(y) \ge 768h$.

Consider first the case $|x - y| \le \frac{d(y)}{4}$. Then Lemma 4.4.1 with K = 2, i.e. $r = \frac{d(y)}{2} \ge 192h$ implies

$$\begin{aligned} \left| G_h(x,y) - \eta_{h,y}^{(d(y)/2)}(x) \hat{G}_h^{(d(y)/2)}(x,y) - G(x,y) + \eta_y^{(d(y)/2)}(x) \hat{G}^{(d(y)/2)}(x,y) \right| \\ &\leq C \frac{h^{\kappa_0}}{r^{\kappa_0}} \sqrt{\log\left(2 + \frac{d(x)}{h}\right)} \end{aligned}$$

which implies that

$$\left|G_{h}(x,y) - \hat{G}_{h}^{(d(y)/2)}(x,y) - G(x,y) + \hat{G}^{(d(y)/2)}(x,y)\right| \le C \frac{h^{\kappa_{0}}}{r^{\kappa_{0}}} \sqrt{\log\left(2 + \frac{2r}{h}\right)}$$

The function $s \mapsto \frac{1}{s^{x_0}} \sqrt{\log(2+2s)}$ is bounded on $[1, \infty)$, so that we actually obtain

$$\left|G_h(x,y) - \hat{G}_h^{(d(y)/2)}(x,y) - G(x,y) + \hat{G}^{(d(y)/2)}(x,y)\right| \le C.$$
(4.4.8)

From Lemma 4.3.2 we know

$$\left|\hat{G}_{h}^{(d(y)/2)}(x,y) - \frac{1}{\lambda^{2}}\log\left(\frac{d(y)}{|x-y|+h}\right)\right| \le C$$

(where we have absorbed a term $\frac{1}{\lambda^2} \log 2$ into the constant). Furthermore by Lemma 4.3.5

$$\left|G(x,y)-\hat{G}^{(d(y)/2)}(x,y)\right|\leq C.$$

If we use these estimates in (4.4.8) we obtain

$$\left|G_h(x,y)-\frac{1}{\lambda^2}\log\left(\frac{d(y)}{|x-y|+h}\right)\right|\leq C.$$

Because $|x - y| \le \frac{d(y)}{4}$, $\frac{d(y)}{|x - y| + h}$ is bounded away from 1 by a constant, and so

$$\left|\frac{1}{\lambda^2}\log\left(\frac{d(y)}{|x-y|+h}\right) - \frac{1}{\lambda^2}\log\left(2 + \frac{d(y)}{|x-y|+h}\right)\right| \le C.$$

Combining this with the preceding inequality we arrive at (B.1').

If $|x - y| \ge \frac{d(y)}{4}$ we argue similarly. We use Lemma 4.4.1 with $r = \frac{d(y)}{4} \ge 192h$ and conclude

$$|G_h(x,y)-G(x,y)|\leq C.$$

This combined with Lemma 4.3.5 implies again (B.1'), as now $\frac{d(y)}{|x-y|+h}$ is bounded above. *Step 2: Proof of (B.2')*

Recall from Lemma 4.3.5 that the term $a(x) := \lambda^2 \lim_{\substack{(x',x'') \to (x,x) \\ x' \neq x''}} (G(x',x'') - \hat{G}(x',x''))$ is

well-defined for each $x \in (0,1)^4$ and that $a: (0,1)^4 \to \mathbb{R}$ is continuous.

After this remark we can proceed similarly as in the first step. We choose $f_1(x) = a(x)$, $f_2(u, v) = \lambda^2 F(u - v)$ with the *F* from Lemma 4.3.1. Furthermore we choose $\theta_0 = \frac{1}{2\kappa_0}$. Given *L* and $\theta > \theta_0$ we take N'_0 so large that $768L \le |\log h|^{\theta}$ when $h \le \frac{1}{N'_0}$. Then $d(x) \ge 1$.

 $h|\log h|^{\theta} \ge 768Lh$. We want to apply Lemma 4.4.1 with K = 8 and $r = \frac{d(x)}{4}$ at the point (x + hu, x + hv). We have that $r = \frac{d(x)}{4} \le \frac{d(x+hv)+Lh}{4} \le \frac{d(x+hv)}{2}$ and similarly $r \ge \frac{d(x+hv)}{8}$, and also $r = \frac{d(x)}{4} \ge 192Lh \ge 192h$ so that all assumptions of the lemma are satisfied. We obtain

$$\begin{aligned} \left| G_{h}(x+hu,x+hv) - \hat{G}_{h}^{(d(x)/4)}(x+hu,x+hv) - G(x+hu,x+hv) + \hat{G}^{(d(x)/4)}(x+hu,x+hv) \right| \\ & \leq C \frac{h^{\kappa_{0}}}{r^{\kappa_{0}}} \sqrt{\log\left(2 + \frac{d(x+hu)}{h}\right)} \leq C \frac{h^{\kappa_{0}}\sqrt{|\log h|}}{r^{\kappa_{0}}} \leq C \frac{h^{\kappa_{0}}\sqrt{|\log h|}}{(h|\log h|^{\theta})^{\kappa_{0}}} \leq C|\log h|^{\frac{1}{2}-\theta\kappa_{0}}. \end{aligned}$$

$$(4.4.9)$$

Here we could omit the cut-off functions $\eta_h^{(d(x)/4)}$ and $\eta^{(d(x)/4)}$ because $|x + hu - (x + hv)| \le 4Lh \le \frac{d(x)}{8}$. Since $\theta \kappa_0 > \theta_0 \kappa_0 = \frac{1}{2}$, for N'_0 large enough the term on the right hand side will be less than $\frac{\varepsilon}{2\lambda^2}$ whenever $h \le \frac{1}{N'_0}$.

By (4.3.11) in Lemma 4.3.5 we have for $u, v \in [0, L]^4$

$$\begin{aligned} \left| G(x + hu, x + hv) - \hat{G}^{(d(x)/4)}(x + hu, x + hv) - \frac{a(x)}{\lambda^2} - \frac{1}{\lambda^2} \log \frac{d(x)}{4} \right| \\ &\leq C \left(\frac{|hu|^{\kappa_0} + |hv|^{\kappa_0}}{d(x)^{\kappa_0}} \right) \leq C_L \frac{h^{\kappa_0}}{d(x)^{\kappa_0}} \leq C_L |\log h|^{-\theta \kappa_0}. \end{aligned}$$

Thus we can choose N'_0 large enough such that for $h \leq \frac{1}{N'_0}$ we have

$$\sup_{u,v\in[0,L]^4\cap\mathbb{Z}^4} \left| G(x+hu,x+hv) - \hat{G}^{(d(x)/4)}(x+hu,x+hv) - \frac{a(x)}{\lambda^2} - \frac{1}{\lambda^2}\log\frac{d(x)}{4} \right| \le \frac{\varepsilon}{2\lambda^2}$$

uniformly in *x*. Our definition of $G_h^{(d(x)/4)}$ implies that

$$\begin{split} \hat{G}_h^{(d(x)/4)}(x+hu,x+hv) &= F\left(\frac{x+hu}{h} - \frac{x+hv}{h}\right) - \frac{1}{\lambda^2}\log h + \frac{1}{\lambda^2}\log \frac{d(x)}{4} \\ &= F(u-v) - \frac{1}{\lambda^2}\log h + \frac{1}{\lambda^2}\log \frac{d(x)}{4}. \end{split}$$

Using these results in (4.4.9) we arrive at

$$\left|G_h(x+hu,x+hv)-F(u-v)+\frac{1}{\lambda^2}\log h-\frac{a(x)}{\lambda^2}\right|\leq \frac{\varepsilon}{\lambda^2}$$

for $h \leq \frac{1}{N'_0}$, which implies (B.2').

Step 3: Proof of (B.3') This is very similar to Step 2. We set $f_3(x, y) = \lambda^2 G(x, y)$, which is continuous away from

the diagonal according to Lemma 4.3.5. We use Lemma 4.4.1 with K = L and $r = \frac{d(y)}{L} \le \frac{1}{L} \le |x - y|$. For N'_1 large enough we have $r \ge 192h$, and the lemma implies

$$|G_h(x,y) - G(x,y)| \le C_L \frac{h^{\kappa_0} \sqrt{|\log h|}}{r^{\kappa_0}} \le C_L |\log h|^{\frac{1}{2} - \theta \kappa_0}$$

and it suffices to take N'_1 so large that the right hand side is less than $\frac{\varepsilon}{\lambda^2}$ for any $h \leq \frac{1}{N'_1}$. Step 4: Proof of (B.0')

Here we actually need to prove three estimates, namely

$$\lambda^2 G_h(x,x) \le |\log h| + C \tag{4.4.10}$$

$$\lambda^2 G_h(x,x) \le C \log\left(2 + \frac{d(x)}{h}\right) \tag{4.4.11}$$

$$\lambda^{2}(G_{h}(x,x) - G_{h}(x,y)) \le \log\left(1 + \frac{|x-y|}{h}\right) + C.$$
 (4.4.12)

Now (4.4.10) follows immediately from (B.1'), and (4.4.11) is a special case of Lemma 4.4.2. Finally, (4.4.12) can be obtained from (B.1') as follows. We know that

$$\lambda^{2}(G_{h}(x,x) - G_{h}(x,y)) \leq \log\left(2 + \frac{d(x)}{h}\right) - \log\left(2 + \frac{\max(d(x),d(y))}{h + |x - y|}\right) + C$$
$$= \log\left(\frac{(d(x) + 2h)(|x - y| + h)}{h(h + |x - y| + 2\max(d(x),d(y)))}\right) + C$$

so one only has to observe that

$$\frac{d(x)+2h}{h+|x-y|+2\max(d(x),d(y))} \le C.$$

Finally we give the proof of Theorem 4.1.1.

Proof of Theorem 4.1.1. Because of Theorem 4.1.3 and Observation 4.1.5 all we have to check is that each of the statements (A.0'), (A.1'), (A.2'), (A.3') implies its counterpart without the prime.

We begin with $(A.0') \Longrightarrow (A.0)$. We know that

$$\operatorname{Var} \varphi_{N,v} \leq \min \left(\log N + \alpha'_0, \alpha'_0 \log(2 + d_N(v)) \right)$$

and this implies in particular that

$$\operatorname{Var} \varphi_{N,v} \leq \log N + \alpha'_0.$$

Furthermore, if we know

$$\operatorname{Var} \varphi_{N,v} - \operatorname{Cov}(\varphi_{N,v}, \varphi_{N,u}) \leq \log_+ |u - v| + 2\alpha_0'$$

then by symmetry this also holds with u, v interchanged, so that we actually have

 $\max \left(\operatorname{Var} \varphi_{N,v} - \operatorname{Cov}(\varphi_{N,v}, \varphi_{N,u}), \operatorname{Var} \varphi_{N,u} - \operatorname{Cov}(\varphi_{N,v}, \varphi_{N,u}) \right) \leq \log_+ |u - v| + 2\alpha'_0$ and a short calculation shows that this is the same as

$$\mathbb{E}(\varphi_{N,v} - \varphi_{N,u})^2 \le 2\log_+ |u - v| - |\operatorname{Var} \varphi_{N,v} - \operatorname{Var} \varphi_{N,u}| + C$$

For (A.1') \implies (A.1) one has to verify that $\min(d(u), d(v)) \ge \delta N$ implies

$$\left|\log\left(2+\frac{\max(d_N(u),d_N(v))}{1+|u-v|}\right)-\log\left(\frac{N}{1+|u-v|}\right)\right|\leq C_{\delta},$$

which is straightforward.

For (A.2') \Longrightarrow (A.2) we fix some $\theta > \theta_0$. Given L, ε, δ , we choose $N_0 \ge N'_0(L, \varepsilon, \theta)$ large enough such that $|\log N|^{\theta} \le \delta N$ for all $N \ge N_0$ and conclude (A.2). Analogously one sees that (A.3') \Longrightarrow (A.3).

5 Optimal order finite difference approximation of generalized solutions to the biharmonic equation in a cube

This chapter is based on the paper [MSS20], written jointly by Stefan Müller, Endre Süli and the author, with only minor changes.

5.1 Introduction

In this chapter we show some error estimates for finite difference schemes for the Bilaplacian. In Section 1.4.4 we gave some background and described some of the results. We begin by stating our results in detail.

5.1.1 Main results

We mostly follow the notation from the introduction (see, however, Section 5.1.3 for the precise definitions). Let $d \in \mathbb{N}^+$, $\Omega := (0,1)^d$, $\Gamma := \partial \Omega$. For $h \in \mathbb{R}^+$ such that $\frac{1}{h} \in \mathbb{N}$, let $\Lambda_h := \Omega \cap (h\mathbb{Z})^d$, $\Gamma_h := \Gamma \cap (h\mathbb{Z})^d$, and

$$\tilde{\Lambda}_h := [-h, 1+h]^{\mathsf{d}} \cap (h\mathbb{Z})^{\mathsf{d}} \setminus \{-h, 1+h\}^{\mathsf{d}}.$$

Consider the elliptic boundary-value problem

$$\Delta^2 u = f \quad \text{in } \Omega,$$

$$u = 0 \quad \text{on } \Gamma,$$

$$\partial_{\nu} u = 0 \quad \text{on } \Gamma,$$

(5.1.1)

where ∂_{ν} denotes the derivative in the normal direction (ν is a unit outward normal vector to Γ). We approximate the solution of this problem by the finite difference scheme (compare [JS14, Section 1.9.4])

$$\Delta_h^2 U = T^{h,2,\dots,2} f \quad \text{in } \Lambda_h ,$$

$$U = 0 \qquad \text{on } \Gamma_h ,$$

$$D_{0,\nu}^h U = 0 \qquad \text{on } \Gamma_h .$$
(5.1.2)

Here *U* is defined on $\tilde{\Lambda}_h$, $D_{0,\nu}^h U(x) := \frac{1}{2h} (U(x + h\nu) - U(x - h\nu))^{-1}$, and $T^{h,2,\dots,2}f$ is a smoothing operator acting on *f*, defined by convolving *f* with a B-spline on the scale *h* (see below for the precise definition).

¹At the singular points (i.e., at the vertices and points on the faces/edges) of Γ_h there are up to d possible boundary normal vectors. For (5.1.2) we consider all of them. Because U = 0 on Γ_h by assumption, this corresponds to setting U = 0 at all points of $\tilde{\Lambda}_h \setminus (\Lambda_h \cup \Gamma_h)$ that have distance *h* to a singular point of Γ_h .

The finite difference scheme (5.1.2) makes sense in any dimension d, as the smoothing operator $T^{h,2,\dots,2}$ maps $f \in H^{s-4}$ into a continuous function whenever $s > \frac{5}{2}$ (cf. [JS14, Theorem 1.69]).

Our objective is to prove an error bound in the discrete Sobolev norm $\|\cdot\|_{H^2_h(\Lambda_h)}$ (which is denoted by $\|\cdot\|_{W^2_2(\Lambda_h)}$ in [JS14, Section 2.2.4]).

Theorem 5.1.1. Suppose that $\frac{1}{2}\max(5,d) < s \leq 4$, and let $u \in H^s(\Omega) \cap H^2_0(\Omega)$; then, there exists a positive constant C = C(n,s), independent of h, such that

$$\|u - U\|_{H^{2}_{\mu}(\Lambda_{h})} \le Ch^{s-2} \|u\|_{H^{s}(\Omega)}.$$
(5.1.3)

This improves [JS14, Theorem 2.69] (where the above result was proved for d = 2 and $\frac{5}{2} < s \le 4$ in the more general setting of fourth-order elliptic equations with nonsmooth variable coefficients, but the order of convergence $O(h^{\min\{s-2,3/2\}} |\log h|^{1-|\operatorname{sgn}(s-7/2)|})$ established there was optimal only in the case of $\frac{5}{2} < s < \frac{7}{2}$, and is reduced to the suboptimal rate of $O(h^{\frac{3}{2}})$, instead of the optimal rate of $O(h^{s-2})$, for $\frac{7}{2} < s \le 4$) as well as the main result in [GMP83] (where the theorem was proved for d = 2 under the additional assumption that the third normal derivative of *u* vanishes at the boundary).

Our method also yields estimates for other discretizations of the boundary conditions. Consider, for instance, the finite difference scheme

$$\begin{aligned} \Delta_h^2 U^* &= T^{h,2,\dots,2} f & \text{in } \Lambda_h ,\\ U^* &= 0 & \text{on } \Gamma_h ,\\ D_v^h U^* &= 0 & \text{on } \Gamma_h . \end{aligned} \tag{5.1.4}$$

Here again U^* is defined on $\tilde{\Lambda}_h$, and $D_{\nu}^h U^*(x) := \frac{1}{h} (U^*(x + h\nu) - U^*(x))$. The conditions $U^* = 0$ and $D_{\nu}^h U^* = 0$ on Γ_h are equivalent to $U^* = 0$ on $\tilde{\Lambda}_h \setminus \Lambda_h$, so that we could equivalently consider the finite difference scheme

$$\Delta_h^2 U^* = T^{h,2,\dots,2} f \quad \text{in } \Lambda_h ,$$

$$U^* = 0 \qquad \text{on } \tilde{\Lambda}_h \setminus \Lambda_h .$$
(5.1.5)

For this difference scheme we can show the following error bound.

Theorem 5.1.2. Suppose that $\frac{1}{2} \max(5, d) < s \leq 3$, and let $u \in H^s(\Omega) \cap H^2_0(\Omega)$; then, there exists a positive constant C = C(n, s), independent of h, such that

$$\|u - U^*\|_{H^2_h(\Lambda_h)} \le Ch^{s-2} \|u\|_{H^s(\Omega)}.$$
(5.1.6)

In Theorems 5.1.1 and 5.1.2 we made the assumption $\frac{1}{2} \max(5, d) < s$. In view of the fact that the problem (5.1.1) makes sense already for $s > \frac{3}{2}$, the requirement $\frac{1}{2} \max(5, d) < s$ might seem surprising. The condition $s > \frac{n}{2}$ ensures that u is continuous. Otherwise, $||u - U||_{H_h^2(\Lambda_h)}$ and $||u - U^*||_{H_h^2(\Lambda_h)}$ would be undefined. The condition $s > \frac{5}{2}$ implies that $T^{h,2,\ldots,2}f$ is continuous so that its pointwise values are defined and the finite difference schemes (5.1.2) and (5.1.4) make sense. It should be possible to relax the assumption $s > \frac{n}{2}$ by replacing u in the expressions $||u - U||_{H_h^2(\Lambda_h)}$ and $||u - U^*||_{H_h^2(\Lambda_h)}$ with a suitably mollified version of u. Similarly, one can relax the assumption $s > \frac{5}{2}$ by replacing $T^{h,2,\ldots,2}$ with a stronger mollification operator; see also Remark 5.4.3 for additional comments in this direction.

Our results extend to more general fourth-order elliptic elliptic operators with variable coefficients, such as those treated in [JS14, Section 2.7], with similar proofs. The main difference compared to the analysis here is that in addition to terms appearing in our error bounds one encounters a variety of mixed terms. On can deal with these as in the proof of Theorem 2.68 in [JS14], using the bilinear Bramble–Hilbert lemma. It should also be possible to extend our results to other (higher order) elliptic operators, such as the polyharmonic operator Δ^k for $k \ge 3$, but the study of that question is beyond the scope of this chapter.

5.1.2 Outline of the proof

We discuss the proof of Theorem 5.1.1 only; the proof of Theorem 5.1.2 is very similar. We proceed similarly to the proof of [JS14, Theorem 2.69]. In fact, when $s < \frac{7}{2}$ we could directly use the argument in [JS14] with only minor notational changes. Let us review that argument here briefly. We begin by extending u symmetrically across Γ to a H^s -function \hat{u} on $(-1, 2)^d$ such that $\|\hat{u}\|_{H^s((-1,2)^d)} \leq C \|u\|_{H^s(\Omega)}$. Here and henceforth C signifies a generic positive constant, which may depend on the Sobolev index s and on the number of space dimensions d, but is independent of the discretization parameter h. Let $E := \hat{u} - U$. Then, E satisfies

$$E = 0 \qquad \text{on } \Gamma_h ,$$
$$D^h_{0\nu} E = 0 \qquad \text{on } \Gamma_h ,$$

and we calculate (compare [JS14, Equation (2.209)])

$$\Delta_h^2 E = \Delta_h^2 \hat{u} - \Delta_h^2 U = \Delta_h^2 \hat{u} - T^{h,2,\dots,2} f = \Delta_h^2 \hat{u} - T^{h,2,\dots,2} \Delta^2 \hat{u}.$$

Using summations by parts we obtain

$$\|
abla_h^2 E\|_{L^2_h(\Lambda_h)} \leq \|\Delta_h^2 \hat{u} - T^{h,2,\dots,2} \Delta^2 \hat{u}\|_{H^{-2}_h(\Lambda_h)}$$
 ,

where ∇_h^2 is the discrete Hessian. Now one can use the Bramble–Hilbert lemma (cf. [JS14]) to deduce that the right-hand side is bounded by $Ch^{s-2} ||u||_{H^s(\Omega)}$, which directly implies (5.1.3).

When $s \ge \frac{7}{2}$ one can no longer extend *u* symmetrically across the boundary while preserving its Sobolev regularity. This means that we cannot make $D_{0,\nu}^h u$ equal to 0 on Γ , and therefore the above argument based on summation by parts no longer works.

Our alternative approach is as follows. Although we cannot ensure that the boundary values of $D_{0,v}^h E$ are exactly zero, we will show that they can nevertheless be made small in an appropriate norm. To this end, we will first show (in Section 5.2.1) that we can take a slightly different extension \tilde{u} with $\|\tilde{u}\|_{H^s((-1,2)^d)} \leq C \|u\|_{H^s(\Omega)}, \frac{7}{2} \leq s \leq 4$, such that \tilde{u} and its derivatives vanish on the hyperplanes supporting the faces of Γ .

This will allow us to estimate the boundary values in an optimal space. In fact, in Section 5.2.2 we prove that

$$\|D_{0,\nu}^{h}\tilde{u}\|_{H_{h}^{\frac{1}{2}}(\Gamma_{h})} \leq Ch^{s-2}\|\tilde{u}\|_{H^{s}(\Omega)}.$$
(5.1.7)

Actually, we only control the $H_h^{\frac{1}{2}}$ -norm on each of the faces of Γ_h , but we ignore this issue here for the sake of simplicity and refer the reader to Section 5.2.2 for precise statements.

Then, in Section 5.2.3, we show that (5.1.7) implies the existence of a function \hat{E} that agrees with \tilde{u} on $\tilde{\Lambda}_h \setminus \Lambda_h$ and such that $\|\nabla_h^2 \hat{E}\|_{L^2(\Lambda_h)} \leq Ch^{s-2} \|u\|_{H^s(\Omega)}$. We shall construct \hat{E} by giving an explicit extension using the Fourier series representation of the boundary values,

which we then carefully cut off to comply with the boundary conditions. This is a special case of an inverse trace theorem in the following sense: for any function ψ on the boundary there is a lattice function w such that

$$w = 0$$
 on Γ_h ,
 $D^h_{0,\nu}w = \psi$ on Γ_h ,

and

$$\|\nabla_{h}^{2}w\|_{L^{2}_{h}(\Lambda_{h})} \leq C \|\psi\|_{H^{\frac{1}{2}}_{h}(\Gamma_{h})}.$$
(5.1.8)

Now that we have \hat{E} at our disposal, we can apply the argument formulated at the beginning of this subsection to $E - \hat{E}$ (which has zero boundary values) and find that

$$\|
abla_h^2(E-\hat{E})\|_{L^2(\Lambda_h)} \leq C\left(h^{s-2}\|u\|_{H^s(\Omega)} + \|\Delta_h^2 \hat{E}\|_{H_h^{-2}(\Lambda_h)}
ight).$$

Thus, by observing that

$$\|\Delta_{h}^{2} \hat{E}\|_{H_{h}^{-2}(\Lambda_{h})} \leq C \|\nabla_{h}^{2} \hat{E}\|_{L^{2}(\Lambda_{h})} \leq C \|D_{0,\nu}^{h} \tilde{u}\|_{H_{h}^{\frac{1}{2}}(\Gamma_{h})} \leq C h^{s-2} \|u\|_{H^{s}(\Omega)},$$

we directly deduce (5.1.3). The details for this argument are given in Section 5.3.2.

The heart of the matter, resulting in our main result stated in Theorem 5.1.1, are the trace estimate (5.1.7) and the inverse trace estimate (5.1.8), established in Section 5.2.2 and Section 5.2.3, respectively.

5.1.3 Notation and preliminaries

Our notation is based on that in [JS14], however we made some changes that we will review in the following.

For $s \ge 0$ and $\Xi \subset \mathbb{R}^d$ open with Lipschitz boundary we define the Sobolev space $H^s(\Xi)$ as the space of restrictions of $H^s(\mathbb{R}^d)$ -functions to Ξ . By $H_0^s(\Xi)$ we denote the closure of the set of all $C_c^{\infty}(\Xi)$ -functions in the $\|\cdot\|_{H^s(\Xi)}$ -norm.

Assume that $\Xi := I_1 \times \cdots \times I_d$, where $I_j \subset \mathbb{R}$ are (possibly unbounded) open intervals. This assumption ensures that we have \mathcal{H}^{d-1} -almost everywhere on $\partial \Xi$ an axiparallel normal vector. Given a $k \in \mathbb{N}_0$ with $k + \frac{1}{2} < s$, we denote by $H^s_{(k)}(\Xi)$ the space of all functions $u \in H^s(\Xi)$ such that the traces of $\partial^i_{\nu} u$ for $0 \le i \le k$ vanish on each face of $\partial \Xi$. We extend this definition to $k > s - \frac{1}{2}$, provided $s \notin \mathbb{N} + \frac{1}{2}$, by setting $H^s_{(k)}(\Xi) = H^s_{(\lfloor s - 1/2 \rfloor)}(\Xi)$.

There are several other equivalent definitions of $H_{(k)}^{s}(\Xi)$. Let $C_{c}^{\infty}(\overline{\Xi})$ denote the space of functions on Ξ , which are in $C^{\infty}(\Xi)$, for which all derivatives admit continuous extensions to $\overline{\Xi}$, and which are supported in $K \cap \Xi$ for some $K \subset \mathbb{R}^{d}$ compact. In other words, $C_{c}^{\infty}(\overline{\Xi})$ denotes the set of restrictions of $C_{c}^{\infty}(\mathbb{R}^{d})$ -functions to Ξ , where the equivalence follows from Whitney's extension theorem [Whi34]. Then, $H_{(k)}^{s}(\Xi)$ is also the closure in the $H^{s}(\Xi)$ -norm of the set of all functions in $C_{c}^{\infty}(\overline{\Xi})$ whose derivatives up to order k vanish on $\partial \Xi$ Furthermore, $H_{(k)}^{s}(\Xi)$ is equal to $H^{s}(\Xi) \cap H_{0}^{k+1}(\Xi)$ if $s \geq k+1$, and equal to $H_{0}^{s}(\Xi)$ if $s \leq k+1$. In particular, the space $H^{s}(\Omega) \cap H_{0}^{2}(\Omega)$ from the main theorems can now be written as $H_{(1)}^{s}(\Omega)$.

The fact that these definitions are equivalent should not be surprising. Nonetheless we could not locate a reference for this precise equivalence result, and so we present its proof in Appendix 5.4.2.

Given a $j \in \mathbb{N}$, we let θ_j be the standard univariate centered B-spline of degree j - 1, defined, for example, as the indicator function of the closed interval $\left[-\frac{1}{2}, \frac{1}{2}\right]$ convolved with itself j - 1 times (cf. [JS14, Section 1.9.4]). Using this, we define the smoothing operator $T_i^{h,j}$ for $1 \le i \le d$ as

$$T_i^{h,j}f := rac{1}{h}f *_i heta_j\left(rac{\cdot}{h}
ight)$$
 ,

where $*_i$ means convolution in the variable x_i . This is a well-defined operator on distributions on \mathbb{R}^d . Furthermore, we set

$$T^{h,j,\ldots,j}f:=T_1^{h,j}\circ\cdots\circ T_d^{h,j}f.$$

Each θ_j is in $H^t(\mathbb{R})$ for any $t < j - \frac{1}{2}$. Using this, one can verify (cf. [JS14, Section 1.9.4]) that $T^{h,j,\dots,j}$ is a bounded linear operator from $H^t(\mathbb{R}^d)$ to $C_b(\mathbb{R}^d)$ whenever $t > -j + \frac{1}{2}$.

We define the discrete Sobolev norm $||v||_{H^2_h(\Lambda_h)}$ of $v \colon \tilde{\Lambda}_h \to \mathbb{R}$ as the sum of the L^2_h -norms of $v, \nabla_h v$ and $\nabla^2_h v$, wherever they are defined; more precisely,

$$\begin{split} \|v\|_{H_{h}^{2}}^{2} &:= \sum_{x \in \tilde{\Omega}_{h}} h^{\mathsf{d}} v(x)^{2} + \sum_{i=1}^{\mathsf{d}} \sum_{\substack{x \in \tilde{\Omega}_{h}:\\ x + he_{i} \in \tilde{\Omega}_{h}}} h^{\mathsf{d}} (D_{i}^{h} v(x))^{2} \\ &+ \sum_{i,j=1}^{\mathsf{d}} \sum_{\substack{x \in \tilde{\Omega}_{h}:\\ x + he_{i}, x - he_{j}, x + he_{i} - he_{j} \in \tilde{\Omega}_{h}}} h^{\mathsf{d}} (D_{i}^{h} D_{-j}^{h} v(x))^{2} \,. \end{split}$$

Note that we have the crucial property

$$D_i^h D_{-i}^h T_i^{h,j-2} f = T_i^{h,j} \partial_i^2 f$$
(5.1.9)

for any *i* and any $j \ge 2$.

5.2 Discrete trace and inverse trace theorems

5.2.1 Construction of a good extension

Recall that $H^s_{(1)}(\Omega)$ denotes the space of functions $u \in H^s(\Omega)$ for which u and ∇u vanish on $\partial\Omega$. Our first goal is to construct an extension \tilde{u} of u that preserves its Sobolev regularity and has the additional property that \tilde{u} and $\nabla \tilde{u}$ vanish on the hyperplanes supporting the faces of Ω .

Later in our argument it will be necessary to localize the functions concerned in order to deal with the 2^d corners of $\Omega = (0, 1)^d$ separately. Actually, it is most convenient to do so right from the start. Thus we shall use a partition of unity, which allows us to split *u* into 2^d parts localized near the corners. These parts can all be dealt with in a similar way, so we focus on one of them and assume that *u* is supported in $[0, \frac{2}{3})^d$.

Lemma 5.2.1. Let $\frac{3}{2} < s \leq 4$, let $u \in H^s_{(1)}(\Omega)$ be supported in $[0, \frac{2}{3}]^d$. Then, there exists a function $\tilde{u} \in H^s_0((-1,1)^d)$ such that $\|\tilde{u}\|_{H^s((-1,1)^d)} \leq C \|u\|_{H^s(\Omega)}$, $\tilde{u}|_{\Omega} = u$, and $\tilde{u} = 0$, $\nabla \tilde{u} = 0$ on the (d-1)-dimensional hyperplanes $x_i = 0$ for $i \in \{1, \ldots, n\}$ in the sense of traces.

Because $\tilde{u} \in H_0^s((-1,1)^d)$, we can extend \tilde{u} outside $(-1,1)^d$ by zero to a function in $H^{s}(\mathbb{R}^{d})$ (that we continue to call \tilde{u}).

The construction of the extension is classical; see, e.g., [LM72a, Section 11.5]. Nonetheless we give some details, in particular because a similar construction will be used in Section 5.2.3.

Proof. We proceed by applying an extension operator similar to the one in [LM72a, Section 2.2] once across every hyperplane (or in other words by applying a tensorized version of that extension operator). There exist $\lambda_{-1}, \lambda_{-2} \in \mathbb{R}$ such that

$$\lambda_{-1} + \lambda_{-2} 2^k = (-1)^k \text{ for } k \in \{2,3\}.$$

We also let $\lambda_1 = 1$. Then we define the extension \tilde{u} of u by

$$\tilde{u}(x_1,\ldots,x_d) = \sum_{\substack{\varepsilon_1=1 \text{ if } x_1 \ge 0\\\varepsilon_1 \in \{-1,-2\} \text{ if } x_1 < 0}} \ldots \sum_{\substack{\varepsilon_d=1 \text{ if } x_d \ge 0\\\varepsilon_d \in \{-1,-2\} \text{ if } x_d < 0}} \lambda_{\varepsilon_1} \cdot \ldots \cdot \lambda_{\varepsilon_d} u(\varepsilon_1 x_1,\ldots,\varepsilon_d x_d).$$

For example, for d = 2 we have

$$\tilde{u}(x_1, x_2) = \begin{cases} u(x_1, x_2) & \text{for } x_1 \ge 0, x_2 \ge 0, \\ \lambda_{-1}u(-x_1, x_2) + \lambda_{-2}u(-2x_1, x_2) & \text{for } x_1 < 0, x_2 \ge 0, \\ \lambda_{-1}u(x_1, -x_2) + \lambda_{-2}u(x_1, -2x_2) & \text{for } x_1 \ge 0, x_2 < 0, \\ (\lambda_{-1})^2u(-x_1, -x_2) + \lambda_{-1}\lambda_{-2}u(-x_1, -2x_2) & \text{for } x_1 < 0, x_2 < 0. \end{cases}$$

One easily checks that both $\tilde{u} = 0$ and $\nabla \tilde{u} = 0$ on the face $x_i = 0$ for $i \in \{1, ..., d\}$. In addition, the support of \tilde{u} is contained in $\left(-\frac{2}{3}, \frac{2}{3}\right)^{\mathsf{d}} \subset (-1, 1)^{\mathsf{d}}$. It remains to show that $\tilde{u} \in H^{s}((-1, 1)^{\mathsf{d}})$ and

$$\|\tilde{u}\|_{H^{s}((-1,1)^{d})} \leq C \|u\|_{H^{s}(\Omega)}.$$
(5.2.1)

For this we use interpolation. If s = 4, and $u \in H^4_{(1)}(\Omega)$ observe that by the construction of \tilde{u} for $k \in \{0, 1, 2, 3\}$ the traces of $\partial_i^k \tilde{u}$ from the two sides of $\{x_i = 0\}$ agree. This implies that $\tilde{u} \in H^4((-1,1)^d)$ and $\|\tilde{u}\|_{H^4((-1,1)^d)} \leq C \|u\|_{H^4(\Omega)}$. If s = 1 and $u \in H^1_{(1)}(\Omega) = H^1_0(\Omega)$ we can use the same argument to obtain (5.2.1) once again. Now, by Lemma 5.4.9 from the Appendix, for any $\frac{3}{2} < s \le 4$ the interpolation space $\left[H_{(1)}^4(\Omega), H_{(1)}^1(\Omega)\right]_{\frac{1}{3}(4-s)}$ is equal to $H^{s}_{(1)}(\Omega)$. Thus (5.2.1) follows by standard function space interpolation theory.

5.2.2 Estimate of the boundary values

In this section we prove the estimate (5.1.7), i.e., that the discrete normal derivatives of \tilde{u} at the boundary can be estimated in the fractional discrete Sobolev space $H_h^{\frac{1}{2}}$. One can think of this result, stated in Lemma 5.2.2, as a discrete trace theorem. Before giving the precise statement we define the appropriate (semi-)norms.

Let *S* be a subset of \mathbb{R}^d that is contained in an axiparallel (d - 1)-dimensional affine subspace of \mathbb{R}^d such that $S \cap (h\mathbb{Z})^d \neq \emptyset$, and let $w \colon S \cap (h\mathbb{Z})^d \to \mathbb{R}$. We then define

$$[w]^{2}_{H^{\frac{1}{2}}_{h}(S\cap(h\mathbb{Z})^{\mathsf{d}})} := \sum_{\substack{x,y \in S\cap(h\mathbb{Z})^{\mathsf{d}} \\ x \neq y}} \frac{|w(x) - w(y)|^{2}}{|x - y|^{\mathsf{d}}} h^{2\mathsf{d}-2}$$

and

$$\|w\|_{H^{\frac{1}{2}}_{h}(S\cap(h\mathbb{Z})^{\mathsf{d}})}^{2} := [w]^{2}_{H^{\frac{1}{2}}_{h}(S\cap(h\mathbb{Z})^{\mathsf{d}})} + \sum_{x \in S\cap(h\mathbb{Z})^{\mathsf{d}}} h^{\mathsf{d}-1} |w(x)|^{2}.$$

For the discrete inverse trace theorem stated in Section 5.2.3 we will need to use the extension by zero of $D_{0,\nu}^h \tilde{u}$ and $D_{\nu}^h \tilde{u}$. Therefore we directly estimate the $H_h^{\frac{1}{2}}$ -seminorm of that extension in the following lemma.² At the first glance it might seem problematic that we are extending $D_{0,\nu}^h \tilde{u}$ by zero, because for $s \ge \frac{5}{2}$ this extension does not preserve the H^s -regularity of \tilde{u} . However it turns out that it is possible to estimate $[D_{0,\nu}^h \tilde{u}]_{H_h^{\frac{1}{2}}}$ by expressions that involve several derivatives in the direction e_d , but at most one derivative in the directions e_i for $1 \le i \le d - 1$, so our assumptions on the boundary values are sufficient.

Lemma 5.2.2. Let $s > \frac{1}{2} \max(3, d)$ and let \tilde{u} be as in Lemma 5.2.1. For $i \in \{1, ..., d\}$ let $g_{h,i}$ and $g_{h,i}^*$ be the extension by zero of $D_{0,i}^h \tilde{u}$ and $D_{-i}^h \tilde{u}$ in the hyperplane $(h\mathbb{Z})^{i-1} \times \{0\} \times (h\mathbb{Z})^{d-i}$, respectively, i.e., $g_{h,i}: (h\mathbb{Z})^{i-1} \times \{0\} \times (h\mathbb{Z})^{d-i} \to \mathbb{R}$ and $g_{h,i}^*: (h\mathbb{Z})^{i-1} \times \{0\} \times (h\mathbb{Z})^{d-i} \to \mathbb{R}$ satisfy

$$g_{h,i}(x) = \begin{cases} D_{0,i}^{h} \tilde{u}(x) & \text{when } x \in (0,\infty)^{i-1} \times \{0\} \times [0,\infty)^{d-i}, \\ 0 & \text{otherwise}, \end{cases}$$
$$g_{h,i}^{*}(0) = \begin{cases} D_{-i}^{h} \tilde{u}(x) & \text{when } x \in (0,\infty)^{i-1} \times \{0\} \times [0,\infty)^{d-i}, \\ 0 & \text{otherwise}. \end{cases}$$

We have that, if $s \leq 4$ *, then*

$$\|g_{h,i}\|_{H^{\frac{1}{2}}_{h}((h\mathbb{Z})^{i-1}\times\{0\}\times(h\mathbb{Z})^{d-i})} \leq Ch^{s-2}\|u\|_{H^{s}(\Omega)},$$
(5.2.2)

and, if $s \leq 3$, then

$$\|g_{h,i}^*\|_{H^{\frac{1}{2}}_{h}((h\mathbb{Z})^{i-1}\times\{0\}\times(h\mathbb{Z})^{\mathsf{d}-i})} \le Ch^{s-2}\|u\|_{H^{s}(\Omega)}.$$
(5.2.3)

We can assume that i = d, the other cases being analogous. For simplicity we identify \mathbb{R}^{d-1} with the hyperplane $\mathbb{R}^{d-1} \times \{0\} \subset \mathbb{R}^d$, and write $x = (x', x_d)$, with $x' := (x_1, \ldots, x_{d-1})$.

Before embarking on the proof of our main result, we state and prove two estimates that we will need.

Lemma 5.2.3. Let $s > \frac{1}{2} \max(3, d)$ and $v \in H^s(\mathbb{R}^d)$ such that v = 0 and $\partial_d v = 0$ on \mathbb{R}^{d-1} in the sense of trace. Let h > 0, let $x' \in (h\mathbb{Z})^{d-1}$, $\hat{x}' \in \mathbb{R}^{d-1} \times \{0\}$ and suppose that $|x' - \hat{x}'|_{\infty} < \frac{h}{2}$. Let further $Q_{h/2}(x') := x' + (-h/2, h/2)^{d-1}$ be the (d-1)-dimensional axiparallel cube of edgelength h centered at x'. If $s \leq 4$, we have that

$$|v(x',h) - v(x',-h) - v(\hat{x}',h) + v(\hat{x}',-h)| \le Ch^{s-\frac{d}{2}} ||v||_{H^s(Q_{h/2}(x')\times\mathbb{R})},$$
(5.2.4)

and if $s \leq 3$ we have that

$$|v(x',0) - v(x',-h) - v(\hat{x}',0) + v(\hat{x}',-h)| \le Ch^{s-\frac{d}{2}} ||v||_{H^s(Q_{h/2}(x')\times\mathbb{R})}.$$
(5.2.5)

²Alternatively one could define a discrete analogue of the $H_{00}^{\frac{1}{2}}$ -norm from [LM72a, Section 11.5]; that however leads to unnecessary technicalities in the present context.

Proof. We begin with (5.2.4). By scaling and translating we can assume that without loss of generality that h = 1 and x' = 0. Because $s > \frac{d}{2}$ the left-hand side of (5.2.4) is bounded by $C ||v||_{H^s(Q_{1/2}(0)\times(-2,2))}$. Furthermore it vanishes when v is a polynomial of degree at most 3. Indeed the boundary condition ensures that each monomial of degree at most 3 has degree at least 2 in x_d and the left-hand side vanishes for such monomials. So (5.2.4) follows from the Bramble–Hilbert lemma (applied in $H^s(Q_{1/2}(0)\times(-2,2))$). The estimate (5.2.5) can be proved analogously.

Lemma 5.2.4. Let $s > \frac{3}{2}$ and $v \in H^s((0,\infty)^{d-1} \times \mathbb{R})$. Suppose that for all $i \in \{1, \ldots, d-1\}$ we have v = 0 on $\{x_i = 0\}$ in the sense of trace, and that furthermore we have $\partial_d v = 0$ on $\{x_d = 0\}$ in the sense of trace. Let \hat{v} be the extension by zero in the first d - 1 variables of v to \mathbb{R}^d , i.e.,

$$\hat{v}(x) := \begin{cases} v(x) & x \in (0,\infty)^{\mathsf{d}-1} \times \mathbb{R}, \\ 0 & otherwise, \end{cases}$$

and let h > 0. If $s \le 4$, then we have that

$$\|\hat{v}(\cdot,h) - \hat{v}(\cdot,-h)\|_{H^{\frac{1}{2}}(\mathbb{R}^{d-1})} \le Ch^{s-1} \|v\|_{H^{s}((0,\infty)^{d-1} \times \mathbb{R})},$$
(5.2.6)

and if $s \leq 3$, then we have that

$$\|\hat{v}(\cdot,0) - \hat{v}(\cdot,-h)\|_{H^{\frac{1}{2}}(\mathbb{R}^{d-1})} \le Ch^{s-1} \|v\|_{H^{s}((0,\infty)^{d-1} \times \mathbb{R})}.$$
(5.2.7)

Proof. Let us define the function spaces $G^s((0,\infty)^{d-1} \times \mathbb{R})$ for $s \in [0,\infty) \setminus \{\frac{1}{2}, \frac{3}{2}\}$ as follows. When $s > \frac{3}{2}$, G^s is the space that is mentioned in the statement of the lemma, i.e.,

$$G^{s}((0,\infty)^{\mathsf{d}-1} \times \mathbb{R}) := H^{s}((0,\infty)^{\mathsf{d}-1} \times \mathbb{R}) \cap \{u \colon u = 0 \text{ on } \partial((0,\infty)^{\mathsf{d}-1} \times \mathbb{R})\}$$
$$\cap \{u \colon \partial_{\mathsf{d}} u = 0 \text{ on } (0,\infty)^{\mathsf{d}-1} \times \{0\}\}.$$

When $\frac{1}{2} < s < \frac{3}{2}$,

$$G^{s}((0,\infty)^{\mathsf{d}-1}\times\mathbb{R}):=H^{s}((0,\infty)^{\mathsf{d}-1}\times\mathbb{R})\cap\{u\colon u=0\text{ on }\partial((0,\infty)^{\mathsf{d}-1}\times\mathbb{R})\},$$

and if $s < \frac{1}{2}$,

$$G^{s}((0,\infty)^{\mathsf{d}-1}\times\mathbb{R}):=H^{s}((0,\infty)^{\mathsf{d}-1}\times\mathbb{R})$$

According to Lemma 5.4.10 from the Appendix we have that, for $s \notin \{\frac{1}{2}, \frac{3}{2}\}$,

$$G^{s}((0,\infty)^{\mathsf{d}-1} \times \mathbb{R}) = \left[G^{4}((0,\infty)^{\mathsf{d}-1} \times \mathbb{R}), G^{1}((0,\infty)^{\mathsf{d}-1} \times \mathbb{R}) \right]_{\frac{4-s}{3}},$$

$$G^{s}((0,\infty)^{\mathsf{d}-1} \times \mathbb{R}) = \left[G^{3}((0,\infty)^{\mathsf{d}-1} \times \mathbb{R}), G^{1}((0,\infty)^{\mathsf{d}-1} \times \mathbb{R}) \right]_{\frac{3-s}{2}}.$$

Thus it suffices to prove (5.2.6) for s = 4 and s = 1 and (5.2.7) for s = 3 and s = 1, and then the result follows by interpolation. We prove the former two statements; the proofs of the latter two are completely analogous.

If s = 1, the condition that v = 0 on $\{x_i = 0\}$ in the sense of trace ensures that $\hat{v} \in H^1(\mathbb{R}^d)$ and $\|\hat{v}\|_{H^1(\mathbb{R}^d)} \leq \|v\|_{H^1((0,\infty)^{d-1} \times \mathbb{R})}$. Now we can use standard trace theorems to bound

$$\begin{split} \|\hat{v}(\cdot,h) - \hat{v}(\cdot,-h)\|_{H^{\frac{1}{2}}(\mathbb{R}^{\mathsf{d}-1})} &\leq \|\hat{v}(\cdot,h)\|_{H^{\frac{1}{2}}(\mathbb{R}^{\mathsf{d}-1})} + \|\hat{v}(\cdot,-h)\|_{H^{\frac{1}{2}}(\mathbb{R}^{\mathsf{d}-1})} \\ &\leq 2\|v\|_{H^{1}((0,\infty)^{\mathsf{d}-1}\times\mathbb{R})} \,. \end{split}$$

If s = 4 the proof is less straightforward. The main difficulty is that \hat{v} is in general not in $H^4(\mathbb{R}^d)$. Instead we write

$$v(\cdot,h) = v(\cdot,0) + h\partial_{\mathsf{d}}v(\cdot,0) + \frac{h^2}{2}\partial_{\mathsf{d}}^2 v(\cdot,0) + \int_0^h \frac{(h-s)^2}{2}\partial_{\mathsf{d}}^3 v(\cdot,s)\,\mathsf{d}s\,.$$

This does not make sense as a pointwise equality, but we can interpret it as an equality in $H^{\frac{1}{2}}((0,\infty)^{d-1})$, with the integral on the right-hand side being understood as a Bochner integral. Similarly, we have

$$v(\cdot,-h) = v(\cdot,0) - h\partial_{\mathsf{d}}v(\cdot,0) + \frac{h^2}{2}\partial_{\mathsf{d}}^2v(\cdot,0) - \int_{-h}^0 \frac{(h+s)^2}{2}\partial_{\mathsf{d}}^3v(\cdot,s)\,\mathsf{d}s\,.$$

Because we know that $v(\cdot, 0) = 0$ and $\partial_d v(\cdot, 0) = 0$ in $H^{\frac{1}{2}}((0, \infty)^{d-1})$, we deduce from this that

$$v(\cdot,h) - v(\cdot,-h) = h^2 \int_{-h}^{h} m\left(\frac{s}{h}\right) \partial_{\mathsf{d}}^3 v(\cdot,s) \,\mathsf{d}s, \tag{5.2.8}$$

as an identity in $H^{\frac{1}{2}}((0,\infty)^{d-1})$, where $m(t) := \begin{cases} \frac{1}{2}(1-t)^2 & \text{for } t \ge 0, \\ \frac{1}{2}(1+t)^2 & \text{for } t \le 0. \end{cases}$

Let \hat{w} be the extension by zero in the first d – 1 variables of $\partial_d^3 v$ to \mathbb{R}^d , i.e.,

$$\hat{w}(x) := \begin{cases} \partial_{\mathsf{d}}^{3} v(x) & \text{for } x \in (0,\infty)^{\mathsf{d}-1} \times \mathbb{R}, \\ 0 & \text{otherwise}. \end{cases}$$

Our assumptions on v imply that $\partial_d^3 v$ belongs to $H_0^1((0,\infty)^{d-1} \times \mathbb{R})$. Therefore $\hat{w} \in H^1(\mathbb{R}^d)$ and $\|\hat{w}\|_{H^1(\mathbb{R}^d)} = \|\partial_d^3 v\|_{H_0^1((0,\infty)^{d-1} \times \mathbb{R})}$. Furthermore (5.2.8) continues to hold for the extensions by zero of both sides, so that we also have

$$\hat{v}(\cdot,h) - \hat{v}(\cdot,-h) = h^2 \int_{-h}^{h} m\left(\frac{s}{h}\right) \hat{w}(\cdot,s) \,\mathrm{d}s$$

as an identity in $H^{\frac{1}{2}}(\mathbb{R}^{d-1})$, and hence

$$\begin{aligned} \|\hat{v}(\cdot,h) - \hat{v}(\cdot,-h)\|_{H^{\frac{1}{2}}(\mathbb{R}^{d-1})} &\leq h^{2} \int_{-h}^{h} m\left(\frac{s}{h}\right) \|\hat{w}(\cdot,s)\|_{H^{\frac{1}{2}}(\mathbb{R}^{d-1})} \, \mathrm{d}s \\ &\leq h^{2} \int_{-h}^{h} \frac{1}{2} \|\hat{w}\|_{H^{1}(\mathbb{R}^{d})} \, \mathrm{d}s \\ &\leq h^{3} \|\partial_{d}^{3}v\|_{H^{1}((0,\infty)^{d-1} \times \mathbb{R})} \\ &\leq h^{3} \|v\|_{H^{4}((0,\infty)^{d-1} \times \mathbb{R})} \,, \end{aligned}$$
..6).

which is (5.2.6).

Proof of Lemma 5.2.2. We begin with (5.2.2). As before, we shall assume without loss of generality that i = d, and we identify \mathbb{R}^{d-1} with $\mathbb{R}^{d-1} \times \{0\} \subset \mathbb{R}^d$ and write $x = (x', x_d)$.

Note that $D_{0,d}^h \tilde{u}(x)$ makes sense for any $x \in [0,1)^{d-1} \times \{0\}$, not only for those in $(h\mathbb{Z})^d$. We denote by g_d the extension by zero of $D_{0,d}^h \tilde{u}$ in the hyperplane $\mathbb{R}^{d-1} \times \{0\}$, i.e., $g_d : \mathbb{R}^{d-1} \times \{0\} \to \mathbb{R}$ satisfies

$$g_{\mathsf{d}}(x) = \begin{cases} D_{0,\mathsf{d}}^{h} \tilde{u}(x) & \text{for } x \in (0,\infty)^{\mathsf{d}-1} \times \{0\}, \\ 0 & \text{otherwise}. \end{cases}$$

Then, $g_{h,d}$ is the restriction of g_d to $(h\mathbb{Z})^d$, and our goal will be to relate the discrete $H_h^{1/2}$ -norm of $g_{h,d}$ and the continuous $H^{1/2}$ -norm of g_d . We begin by estimating the latter.

Applying Lemma 5.2.4 to $hg_{h,d}$ we obtain

$$\|g_{\mathsf{d}}\|_{H^{\frac{1}{2}}(\mathbb{R}^{\mathsf{d}-1})} = \|D_{0,\mathsf{d}}^{h}\tilde{u}\|_{H^{\frac{1}{2}}(\mathbb{R}^{\mathsf{d}-1})} \le Ch^{s-2}\|\tilde{u}\|_{H^{s}(\mathbb{R}^{\mathsf{d}})}.$$
(5.2.9)

Next, let $x' \in (h\mathbb{Z})^{d-1}$, $\hat{x}' \in \mathbb{R}^{d-1}$, and suppose that $|x' - \hat{x}'|_{\infty} < \frac{h}{2}$. Recall that $Q_{h/2}(x') = x' + (-h/2, h/2)^{d-1}$ is the (d-1)-dimensional axiparallel cube of edge-length h centered at x'. Then, Lemma 5.2.3 implies that

$$|\tilde{u}(x',h)-\tilde{u}(x',-h)-\tilde{u}(\hat{x}',h)+\tilde{u}(\hat{x}',-h)| \leq Ch^{s-\frac{d}{2}} \|\tilde{u}\|_{H^s(Q_{h/2}(x')\times\mathbb{R})}$$

If $x'_i > 0$ for all i = 1, ..., d - 1, then

$$|g_{\mathsf{d}}(x') - g_{\mathsf{d}}(\hat{x}')| = \frac{1}{2h} |\tilde{u}(x',h) - \tilde{u}(x',-h) - \tilde{u}(\hat{x}',h) + \tilde{u}(\hat{x}',-h)|.$$
(5.2.10)

On the other hand, if $x_i \leq 0$ for some $i \in \{1, \dots, d-1\}$, then $\tilde{u}(x', h) = \tilde{u}(x', -h) = g_d(x', 0) = 0$ and

$$|g_{\mathsf{d}}(\hat{x}')| = \begin{cases} \frac{1}{2h} |\tilde{u}(\hat{x}',h) - \tilde{u}(\hat{x}',-h)| & \text{for } x' \in (0,\infty)^{\mathsf{d}-1}, \\ 0 & \text{otherwise}. \end{cases}$$

This, together with (5.2.10), implies that we have in any case

$$|g_{\mathsf{d}}(x') - g_{\mathsf{d}}(\hat{x}')| \le \frac{1}{2h} |\tilde{u}(x',h) - \tilde{u}(x',-h) - \tilde{u}(\hat{x}',h) + \tilde{u}(\hat{x}',-h)|.$$

Thus we get that

$$|g_{\mathsf{d}}(x') - g_{\mathsf{d}}(\hat{x}')| \le Ch^{s-1-\frac{d}{2}} [\tilde{u}]_{H^{s}(Q_{h/2}(x')\times\mathbb{R})} \le Ch^{s-1-\frac{d}{2}} \|\tilde{u}\|_{H^{s}(Q_{h/2}(x')\times\mathbb{R})}.$$
 (5.2.11)

Now let $x', y' \in (h\mathbb{Z})^{d-1}$, $\hat{x}' \in Q_{h/2}(x')$ and $\hat{y}' \in Q_{h/2}(y')$. We then have that

$$|g_{\mathsf{d}}(x') - g_{\mathsf{d}}(y')| \le |g_{\mathsf{d}}(\hat{x}') - g(\hat{y}')| + |g(x') - g(\hat{x}')| + |g(y') - g(\hat{y}')|.$$

This implies that $|g_d(x') - g_d(y')|^2 \le 3(|g_d(\hat{x}') - g_d(\hat{y}')|^2 + |g_d(x') - g_d(\hat{x}')|^2 + |g_d(y') - g_d(\hat{y}')|^2)$, and, using (5.2.11), we deduce that

$$|g_{\mathsf{d}}(x') - g_{\mathsf{d}}(y')|^{2} \leq 3|g_{\mathsf{d}}(\hat{x}') - g_{\mathsf{d}}(\hat{y}')|^{2} + Ch^{2s-2-\mathsf{d}} \|\tilde{u}\|_{H^{s}(Q_{h}(x')\times(-2h,2h))}^{2} + Ch^{2s-2-\mathsf{d}} \|\tilde{u}\|_{H^{s}(Q_{h}(y')\times(-2h,2h))}^{2} \cdot (5.2.12)$$

Thus, taking the average of (5.2.12) over all $\hat{x}' \in Q_{h/2}(x')$ and $\hat{y}' \in Q_{h/2}(y')$, we obtain

$$\begin{split} |g_{\mathsf{d}}(x') - g_{\mathsf{d}}(y')|^2 &\leq 3h^{2-2\mathsf{d}} \int_{Q_{h/2}(x')} \int_{Q_{h/2}(y')} |g_{\mathsf{d}}(\hat{x}') - g_{\mathsf{d}}(\hat{y}')|^2 \, \mathrm{d}\hat{x}' \, \mathrm{d}\hat{y}' \\ &+ Ch^{2s-2-\mathsf{d}} \left(\|\tilde{u}\|_{H^s(Q_h(x')\times\mathbb{R})}^2 + \|\tilde{u}\|_{H^s(Q_h(y')\times\mathbb{R})}^2 \right) \, . \end{split}$$

Observe that for $|x' - y'| \ge h$ we have

$$\begin{aligned} |\hat{x}' - \hat{y}'| &= |x' - y' - (x' - \hat{x}') + (y' - \hat{y}')| \\ &\leq |x' - y'| + |x' - \hat{x}'| + |y' - \hat{y}'| \end{aligned}$$

$$\leq |x'-y'|+h \leq 2|x'-y'|.$$

Using this, we deduce that

$$\begin{split} \left[g_{h,d}\right]_{H_{h}^{\frac{1}{2}}((h\mathbb{Z})^{d-1})}^{2} &= \sum_{x',y'\in(h\mathbb{Z})^{d-1}} \frac{|g_{d}(x') - g_{d}(y')|^{2}}{|x' - y'|^{d}} h^{2d-2} \\ &\leq 3 \cdot 2^{d} \int_{\mathbb{R}^{d-1}} \int_{\mathbb{R}^{d-1}} \frac{|g_{d}(\hat{x}') - g_{d}(\hat{y}')|^{2}}{|\hat{x}' - \hat{y}'|^{d}} d\hat{x}' d\hat{y}' \\ &+ Ch^{d+2s-4} \sum_{x',y'\in(h\mathbb{Z})^{d-1}} \frac{1}{|y' - x'|^{d}} \|\tilde{u}\|_{H^{s}(Q_{h}(x')\times\mathbb{R})}^{2} \\ &+ Ch^{d+2s-4} \sum_{x',y'\in(h\mathbb{Z})^{d-1}} \frac{1}{|x' - y'|^{d}} \|\tilde{u}\|_{H^{s}(Q_{h}(y')\times\mathbb{R})}^{2} . \end{split}$$
(5.2.13)

The first term on the right-hand side is a constant times $[g_d]_{H^{\frac{1}{2}}(\mathbb{R}^{d-1})}^2$. To estimate the second term, notice that

$$\sum_{\substack{y' \in (h\mathbb{Z})^{d-1}, y' \neq x'}} \frac{1}{|y' - x'|^{d}} = \frac{1}{h^{d-1}} \sum_{\substack{y' \in (h\mathbb{Z})^{d-1}, y' \neq x'}} \frac{1}{|y' - x'|^{d}} h^{d-1}$$
$$\leq \frac{C}{h^{d-1}} \int_{|y' - x'| \geq h} \frac{1}{|y' - x'|^{d}} \, \mathrm{d}y' \leq \frac{C}{h^{d}}$$

and

$$\sum_{\substack{\boldsymbol{\mathcal{U}} \in (h\mathbb{Z})^{\mathsf{d}-1}}} \|\tilde{u}\|_{H^s(\mathcal{Q}_h(\boldsymbol{x}') \times \mathbb{R})}^2 \leq \|\tilde{u}\|_{H^s(\mathbb{R}^{\mathsf{d}})}^2,$$

by superadditivity of the fractional Sobolev norm.

x

Together with the analogous estimate for the third term and (5.2.9) we arrive at

$$\begin{split} \left\| g_{h,\mathsf{d}} \right\|_{H_{h}^{\frac{1}{2}}((h\mathbb{Z})^{\mathsf{d}-1})}^{2} &\leq C[g_{\mathsf{d}}]_{H^{\frac{1}{2}}(\mathbb{R}^{\mathsf{d}-1})}^{2} + Ch^{2s-4} \| \tilde{u} \|_{H^{s}(\mathbb{R}^{\mathsf{d}})}^{2} \\ &\leq Ch^{2s-4} \| \tilde{u} \|_{H^{s}(\mathbb{R}^{\mathsf{d}})}^{2} \\ &\leq Ch^{2s-4} \| u \|_{H^{s}(\Omega)}^{2} . \end{split}$$

$$(5.2.14)$$

It remains to estimate $\|g_{h,d}\|_{L^2_h(h\mathbb{Z})^{d-1}}$. A simple way to do so is to observe that we have a Poincaré-type inequality. Indeed, $g_{h,d}$ is supported in $\left[0, \frac{2}{3}\right]^{d-1} \cap (h\mathbb{Z})^{d-1}$ and therefore

$$\begin{split} \left[g_{h,d}\right]_{H_{h}^{\frac{1}{2}}((h\mathbb{Z})^{d-1})}^{2} &= \sum_{\substack{x',y' \in (h\mathbb{Z})^{d-1} \\ x' \neq y'}} \frac{|g_{d}(x') - g_{d}(y')|^{2}}{|x' - y'|^{d}} h^{2d-2} \\ &\geq \sum_{\substack{x' \in [0,1)^{d-1} \cap (h\mathbb{Z})^{d-1} \\ x' \in [0,1)^{d-1} \cap (h\mathbb{Z})^{d-1}}} \sum_{\substack{y' \in [-2,-1)^{d-1} \cap (h\mathbb{Z})^{d-1} \\ y' \in [-2,-1)^{d-1} \cap (h\mathbb{Z})^{d-1}}} \frac{|g_{d}(x') - g_{d}(y')|^{2}}{|x' - y'|^{d}} h^{2d-2} \\ &\geq \sum_{\substack{x' \in [0,1)^{d-1} \cap (h\mathbb{Z})^{d-1} \\ x' \in [0,1)^{d-1} \cap (h\mathbb{Z})^{d-1}}} \sum_{\substack{y' \in [-2,-1)^{d-1} \cap (h\mathbb{Z})^{d-1} \\ x' \in [0,1)^{d-1} \cap (h\mathbb{Z})^{d-1}}} h^{d-1} |g_{h,d}(x')|^{2} \,. \end{split}$$

Combining this with (5.2.14) we obtain (5.2.2). The proof of (5.2.3) is similar, with the only difference that we use (5.2.5) and (5.2.7) instead of (5.2.4) and (5.2.6). \Box

5.2.3 A discrete inverse trace theorem on the cube

In the previous section we proved that the discrete normal derivative of \tilde{u} has small trace in $H_h^{\frac{1}{2}}$, and thus the same holds true for $E = \tilde{u} - U$ and $E^* = \tilde{u} - U^*$. We now want to construct a function \hat{E} such that \hat{E} and E agree on Γ_h and such that the H_h^2 -norm of \hat{E} is small (and similarly for \hat{E}^*). The existence of \hat{E} and \hat{E}^* follows from a discrete inverse trace theorem, as in (5.1.8). However we will not prove (or even state precisely) a general result, as we did in (5.1.8); instead, we shall state the result when applied directly to \hat{E} and \hat{E}^* . The following two lemmas concern the boundary conditions appearing in Theorems 5.1.1 and 5.1.2, respectively.

Lemma 5.2.5. Let $\frac{1}{2} \max(3, d) < s \leq 4$ and let \tilde{u} be as in Lemma 5.2.1. Then, there is a function \hat{E} on Λ_h such that

$$\hat{E} = 0$$
 on Γ_h ,
 $D^h_{0,\nu}\hat{E} = D^h_{0,\nu}\tilde{u}$ on Γ_h ,

and such that $\|\nabla_h^2 \hat{E}\|_{L^2(\Lambda_h)} \leq Ch^{s-2} \|u\|_{H^s(\Omega)}$.

Lemma 5.2.6. Let $\frac{1}{2} \max(3, d) < s \leq 3$ and let \tilde{u} be as in Lemma 5.2.1. Then, there is a function \hat{E}^* on Λ_h such that

$$\hat{E}^* = 0$$
 on Γ_h ,
 $D_v^h \hat{E}^* = D_v^h \tilde{u}$ on Γ_h ,

and such that $\|\nabla_h^2 \hat{E}^*\|_{L^2(\Lambda_h)} \leq Ch^{s-2} \|u\|_{H^s(\Omega)}$.

The strategy for the proof of both lemmas is the following. It suffices to consider the case when $D_{\nu}^{h}\tilde{u}$ is nonzero only on one face, say $\{x_{d} = 0\}$. We construct an extension of the boundary values there in Fourier space. This extension is constructed in such a way that we can control its H_{h}^{2} -norm by the $H_{h}^{\frac{1}{2}}$ -norm of the boundary values (at least after localizing to a bounded set). However this extension does not yet have the appropriate boundary values at $\{x_{i} = 0\}$ for i < d. To fix this we use a projection operator $H_{h}^{2} \rightarrow H_{h,0}^{2}$ on each fixed slice $\{x_{d} = c\}$ and show that we retain control of the H_{h}^{2} -norm.

Proof of Lemma 5.2.5. Step 1: Preliminaries

Because we have localized \tilde{u} , $D_{0,\nu}^{h}\tilde{u}$ has nonzero boundary values only on the faces $\Gamma_{h} \cap \{x_{i} = 0\}$. We can deal with the faces separately. In fact we will construct functions \hat{E}_{i} for $i \in \{1, ..., d\}$ such that $\hat{E}_{i} = 0$ on Γ_{h} , $D_{0,\nu}^{h}\hat{E}_{i} = D_{0,\nu}^{h}\tilde{u}$ on $x_{i} = 0$ while $D_{0,\nu}^{h}\hat{E}_{i} = 0$ on $\Gamma_{h} \setminus \{x_{i} = 0\}$, satisfying the estimate $\|\nabla_{h}^{2}\hat{E}_{i}\|_{L^{2}(\Lambda_{h})} \leq Ch^{s-2}\|u\|_{H^{s}(\Omega)}$. Then we can choose $\hat{E} = \sum_{i} \hat{E}_{i}$, which will have the desired properties. As the \hat{E}_{i} can be constructed analogously, we shall focus on \hat{E}_{d} only.

Step 2: Construction of an extension in Fourier space Recall the function $g_{h,d}$, the extension by zero of $D_{0,d}^h \hat{u}$. Thanks to our assumption, $g_{h,d}$ is supported in $[0, \frac{2}{3}]^{d-1} \cap (h\mathbb{Z})^{d-1}$. We can extend this function periodically with period 2 and represent it by its discrete Fourier series

$$g_{h,d}(x') = \sum_{k' \in \left\{-\frac{1}{h}+1,...,\frac{1}{h}\right\}^{d-1}} \gamma_{k'} e^{i\pi(k'\cdot x')},$$
where

$$\gamma_{k'} = \left(\frac{h}{2}\right)^{\mathsf{d}-1} \sum_{\xi' \in [-1,1)^{\mathsf{d}-1} \cap (h\mathbb{Z})^{\mathsf{d}-1}} g_{h,\mathsf{d}}(\xi') \, \mathrm{e}^{-i\pi k' \cdot \xi'}$$

and $k' := (k'_1, \dots, k'_{d-1}) \in \mathbb{Z}^{d-1}$.

It is easy to verify that

$$\sum_{k' \in \left\{-\frac{1}{h}+1,\ldots,\frac{1}{h}\right\}^{d-1}} (1+|k'|)\gamma_{k'}^2 \le C \|g_{h,\mathsf{d}}\|_{H_h^{\frac{1}{2}}((h\mathbb{Z})^{d-1})}^2.$$
(5.2.15)

Indeed, the Fourier norm on the left-hand side is controlled by the $H_h^{\frac{1}{2}}$ -norm on the torus $([-1,1]^{d-1} \cap (h\mathbb{Z})^{d-1})/_{\sim}$ (compare, e.g., [Hac81, Section 2.3]) and the latter is bounded by the $H_h^{\frac{1}{2}}$ -norm on $(h\mathbb{Z})^{d-1}$ because the support of $g_{h,d}$ is bounded away from $\partial [-1,1]^{d-1}$. Define

$$a(x', x_{d}) := \sum_{k' \in \left\{-\frac{1}{h}+1, \dots, \frac{1}{h}\right\}^{d-1}} \frac{\gamma_{k'}}{\cosh(|k'|h)} x_{d} e^{-|k'|x_{d}} e^{i\pi k' \cdot x'}$$

It is then easy to check that a(x',0) = 0 and $D_{0,d}^h a(x',0) = g_{h,d}(x')$ for $x' \in (-1,1)^{d-1} \cap (h\mathbb{Z})^{d-1}$. Furthermore, the H_h^2 -norm of a is controlled. Indeed, we have that

$$\nabla_h^2 a(x', x_d) = \sum_{k' \in \left\{-\frac{1}{h} + 1, \dots, \frac{1}{h}\right\}^{d-1}} \sigma(k', h, x_d) \gamma_{k'} e^{i\pi k' \cdot x'},$$

where the coefficients $\sigma(k', h, x_d)$ satisfy $|\sigma(k', h, x_d)| \le C|k'|(|k'|x_d + 1)e^{-|k'|x_d}$. This can be seen using Taylor's theorem in the form $\nabla_h^2 v(x) = \nabla^2 v(x) + O\left(h \sup_{|\hat{x} - x|_\infty \le h} |\nabla^3 v(\hat{x})|\right)$. For example,

$$D_{-d}^{h} D_{d}^{h} \left(x_{d} e^{-|k'|x_{d}} e^{i\pi k' \cdot x'} \right)$$

= $(|k'|^{2} x_{d} - 2|k'|) e^{-|k'|x_{d}} e^{i\pi k' \cdot x'} + O(h(|k'|^{2} + |k'|^{3} (x_{d} + h)) e^{-|k'|(x_{d} - h)}),$

and therefore $\left|D_{-d}^{h}D_{d}^{h}\left(x_{d}e^{-|k'|x_{d}}e^{i\pi k'\cdot x'}\right)\right| \leq C|k'|(|k'|x_{d}+1)e^{-|k'|x_{d}}$.

Now, using orthogonality in x' we get, for $x_d \ge 0$,

$$\begin{split} \sum_{x' \in [-1,1)^{d-1} \cap (h\mathbb{Z})^{d-1}} h^{d-1} |\nabla_h^2 a(x', x_d)|^2 &= 2^{d-1} \sum_{k' \in \left\{-\frac{1}{h}+1, \dots, \frac{1}{h}\right\}^{d-1}} |\sigma(k', h, x_d)|^2 |\gamma_{k'}|^2 \\ &\leq C \sum_{k' \in \left\{-\frac{1}{h}+1, \dots, \frac{1}{h}\right\}^{d-1}} |k'|^2 (|k'|^2 x_d^2 + 1) e^{-2|k'|x_d} |\gamma_{k'}|^2, \end{split}$$

and hence

$$\sum_{\substack{x \in [-1,1)^{d-1} \times [0,2] \cap (h\mathbb{Z})^d \\ \leq Ch \sum_{\substack{x_d \in [0,2] \times h\mathbb{Z} \\ k' \in \left\{-\frac{1}{h}+1, \dots, \frac{1}{h}\right\}^{d-1}}} \sum_{\substack{k' | ^2(|k'|^2 x_d^2 + 1) e^{-2|k'|x_d} |\gamma_{k'}|^2}}$$

Next, we use the estimate

$$\sum_{x_{\mathsf{d}}\in[0,2]\cap h\mathbb{Z}}hx_{\mathsf{d}}^{\alpha}\,\mathrm{e}^{-2|k'|x_{\mathsf{d}}}\leq C_{\alpha}\int_{0}^{\infty}\xi^{\alpha}\mathrm{e}^{-2|k'|\xi}\,\mathrm{d}\xi=C_{\alpha}\frac{1}{|k'|^{1+\alpha}}\int_{0}^{\infty}\theta^{2}\,\mathrm{e}^{-2\theta}\,\mathrm{d}\theta\leq\frac{C_{\alpha}}{|k'|^{1+\alpha}}$$

for $\alpha = 2$ and $\alpha = 0$ to deduce that

$$\sum_{x \in [-1,1)^{d-1} \times [0,2] \cap (h\mathbb{Z})^d} h^d |\nabla_h^2 a(x', x_d)|^2 \le C \sum_{k' \in \left\{-\frac{1}{h}+1, \dots, \frac{1}{h}\right\}^{d-1}} |k'| |\gamma_{k'}|^2$$

and thus, taking into account (5.2.15) and (5.2.2),

$$\|\nabla_{h}^{2}a\|_{L^{2}_{h}((-1,1)^{\mathsf{d}-1}\times[0,2]\cap(h\mathbb{Z})^{\mathsf{d}})} \leq C[g_{h,\mathsf{d}}]_{H^{\frac{1}{2}}_{h}((h\mathbb{Z})^{\mathsf{d}-1})} \leq Ch^{s-2}\|u\|_{H^{s}(\Omega)}.$$
(5.2.16)

Similarly, we estimate

$$\|\nabla_h a\|_{L^2_h((-1,1)^{\mathsf{d}-1} \times [0,2] \cap (h\mathbb{Z})^{\mathsf{d}})} \le Ch^{s-2} \|u\|_{H^s(\Omega)},$$
(5.2.17)

$$\|a\|_{L^{2}_{h}((-1,1)^{\mathsf{d}-1}\times[0,2]\cap(h\mathbb{Z})^{\mathsf{d}})} \le Ch^{s-2}\|u\|_{H^{s}(\Omega)}$$
(5.2.18)

(note that for these estimates we actually need control of $\|g_{h,d}\|_{H^{\frac{1}{2}}_{h}((h\mathbb{Z})^{d-1})}$, not just of

 $[g_{h,\mathsf{d}}]_{H_h^{\frac{1}{2}}((h\mathbb{Z})^{\mathsf{d}-1})}).$ Step 3: Localization

Let $\eta \in \mathbb{C}_c^{\infty}(\mathbb{R})$ be such that $\eta = 1$ in $\left[-\frac{3}{4}, \frac{3}{4}\right]$, $\eta = 0$ in $\mathbb{R} \setminus [-1, 1]$, and let

 $\tilde{a}(x) := \eta(x_1) \cdot \ldots \cdot \eta(x_d) a(x) \, .$

Because a = 0 on $\{x_i = 0\}$ for all i, we have that $\tilde{a} = 0$ on Γ_h . Furthermore, $D_{0,d}^h a = 0$ except possibly in $\left[-\frac{2}{3}, \frac{2}{3}\right]^{d-1} \times \{0\}$, and the product $\eta(x_1) \cdots \eta(x_d)$ is equal to the constant 1 in a neighborhood of that set. Therefore, $D_{0,d}^h \tilde{a} = D_{0,d}^h a = g_{h,d}$ on $\{x_d = 0\}$.

Using the estimates (5.2.16), (5.2.17), (5.2.18) and the discrete product rule, we also obtain

$$\begin{aligned} \|\nabla_{h}^{2}\tilde{a}\|_{L^{2}_{h}(\Lambda_{h})} &\leq C \bigg(\|\nabla_{h}^{2}a\|_{L^{2}_{h}((-1,1)^{d-1}\times[0,2]\cap(h\mathbb{Z})^{d})} + \|\nabla_{h}a\|_{L^{2}_{h}((-1,1)^{d-1}\times[0,2]\cap(h\mathbb{Z})^{d})} \\ &+ \|a\|_{L^{2}_{h}((-1,1)^{d-1}\times[0,2]\cap(h\mathbb{Z})^{d})} \bigg) \\ &\leq Ch^{s-2} \|u\|_{H^{s}(\Omega)} \,. \end{aligned}$$
(5.2.19)

Step 4: Correction of the boundary values

Unfortunately, \tilde{a} does not yet have the correct boundary values at $\{x_i = 0\}$ for $1 \le i \le d - 1$. To rectify this we use a discrete projection from H^2 to H_0^2 . First we define the corresponding continuous projection. It is defined in a similar way as the extension we used in the proof of Lemma 5.2.1, namely by tensorizing the restriction operator from [LM72a, Section 11.5]. Thus we choose $\lambda_{-1}, \lambda_{-2} \in \mathbb{R}$ such that

$$\lambda_{-1} + \lambda_{-2} 2^k = (-1)^{k+1}$$
 for $k \in \{0, 1\}$.

(i.e., $\lambda_{-1} = -3$, $\lambda_{-2} = 2$); we let $\lambda_1 = 1$ and define a restriction operator $R: H^2(\mathbb{R}^{d-1}) \to H^2_0((0,\infty)^{d-1})$ by

$$Rv(x) := \sum_{\varepsilon_1 \in \{1,-1,-2\}} \dots \sum_{\varepsilon_d \in \{1,-1,-2\}} \lambda_{\varepsilon_1} \cdot \dots \cdot \lambda_{\varepsilon_{d-1}} v(\varepsilon_1 x_1, \dots, \varepsilon_{d-1} x_{d-1}).$$

One can check that we indeed have $Rv \in H_0^2((0,\infty)^{d-1})$ and $||Rv||_{H^2((0,\infty)^{d-1})} \leq C ||v||_{H^2(\mathbb{R}^{d-1})}$. If we extend Rv by zero to \mathbb{R}^{d-1} we can also consider R as an operator mapping $H^2(\mathbb{R}^{d-1})$. to itself. Note that if $x' \in (h\mathbb{Z})^{d-1}$, then Rv(x') depends only on $v|_{(h\mathbb{Z})^{d-1}}$. Thus we can define $R_h: H_h^2((h\mathbb{Z})^{d-1}) \to H_h^2((h\mathbb{Z})^{d-1})$ by

$$R_{h}v(x') := \begin{cases} Rv(x') & \text{for } x' \in [0,\infty)^{d-1}, \\ Rv(x'+2he_{i}) & \text{for } x' \in [0,\infty)^{i-1} \times \{-h\} \times [0,\infty)^{d-1-i}, \\ 0 & \text{otherwise}. \end{cases}$$

We claim that

$$\|R_h v\|_{L^2_h([0,\infty)^{\mathsf{d}-1} \cap (h\mathbb{Z})^{\mathsf{d}-1})} \le C \|v\|_{L^2_h((h\mathbb{Z})^{\mathsf{d}-1})},$$
(5.2.20)

$$\|\nabla_h R_h v\|_{L^2_h([0,\infty)^{\mathsf{d}-1} \cap (h\mathbb{Z})^{\mathsf{d}-1})} \le C \|\nabla_h v\|_{L^2_h((h\mathbb{Z})^{\mathsf{d}-1})},$$
(5.2.21)

$$\|\nabla_h^2 R_h v\|_{L^2_h([0,\infty)^{\mathsf{d}-1} \cap (h\mathbb{Z})^{\mathsf{d}-1})} \le C \|\nabla_h^2 v\|_{L^2_h((h\mathbb{Z})^{\mathsf{d}-1})}.$$
(5.2.22)

Indeed, these estimates follow from the discrete chain rule. The only exception are the terms $D_i^h D_{-i}^h R_h v(x')$ in (5.2.22), which are not, a priori, controlled on $\{x_i = 0\}$. However an explicit calculation shows that for such x' one has

$$\begin{split} D_i^h D_{-i}^h R_h v(x') &= 2 \frac{R_h v(x' + he_i)}{h^2} \\ &= 2 \frac{v(x' + he_i) - 3v(x' - he_i) + 2v(x' - 2he_i)}{h^2} \\ &= 2 \frac{v(x' + he_i) - 2v(x') + v(x' - he_i)}{h^2} + 4 \frac{v(x') - 2v(x' - he_i) + v(x' - 2he_i)}{h^2} \\ &= 2 D_i^h D_{-i}^h v(x') + 4 D_i^h D_{-i}^h v(x' - he_i) \,, \end{split}$$

so that these terms, which are 'crossing the boundary', are still controlled.³

We now apply R_h along every slice $(h\mathbb{Z})^{d-1} \times \{x_d\}$, i.e., we set

$$b(x) := R_h \tilde{a}(\cdot, x_d)(x').$$

Then by construction of R_h we have b = 0 and $D_{0,i}^h b(x) = 0$ on $\{x_i = 0\}$. Furthermore, b is supported in $[-h, \frac{3}{4}]^d$ and we have b = 0 on $\{x_d = 0\}$. We know that $D_{0,d}^h \tilde{a} = g_{h,d}$ on $\{x_d = 0\}$. In addition, $R_h g_{h,d} = g_{h,d}$ on $[0, \infty)^{d-1} \times \{0\}$, and so $D_{0,d}^h b = g_{h,d}$ on $[0, \infty)^{d-1} \cap (h\mathbb{Z})^{d-1}$ follows from the fact that R_h and $D_{0,d}^h$ commute.

We next estimate $\|\nabla_h^2 b\|_{L^2(\Lambda_h)} = \|\widetilde{\nabla}_h^2 R_h \tilde{a}\|_{L^2(\Lambda_h)}$. If $i, j \leq d - 1$ then (5.2.22) implies that

$$\|D_{i}^{h}D_{-j}^{h}R_{h}\tilde{a}\|_{L^{2}(\Lambda_{h})} \leq C\|\nabla_{h}^{2}\tilde{a}\|_{L^{2}_{h}((h\mathbb{Z})^{d-1})}$$

When taking derivatives in the direction e_d we use (5.2.21) and the fact that $D_{\pm n}^h$ and R_h commute, to obtain (for i < d) that

$$\begin{split} \|D_i^h D_{-\mathsf{d}}^h R_h \tilde{a}\|_{L^2(\Lambda_h)} &= \|D_i^h R_h D_{-\mathsf{d}}^h \tilde{a}\|_{L^2(\Lambda_h)} \\ &\leq C \|D_i^h D_{-\mathsf{d}}^h \tilde{a}\|_{L^2(\Lambda_h)}, \end{split}$$

and similarly, using (5.2.20),

$$\|D^h_{\mathsf{d}}D^h_{-\mathsf{d}}R_h\tilde{a}\|_{L^2(\Lambda_h)} \leq C\|D^h_{\mathsf{d}}D^h_{-\mathsf{d}}\tilde{a}\|_{L^2(\Lambda_h)}$$

³It is of course no coincidence that we have such an identity. In fact, $\nabla^2(Ru)$ is bounded in the L^2 norm thanks to the construction of *R*, and one can therefore also expect R_h to be well-behaved at the boundary.

If we combine the last three estimates and use (5.2.19) we deduce that

$$\|
abla_h^2 b\|_{L^2(\Lambda_h)} = \|
abla_h^2 ilde{a}\|_{L^2(\Lambda_h)} \le C h^{s-2} \|u\|_{H^s(\Omega)} \,.$$

Thus we can set $b = \hat{E}_d$, and have shown that \hat{E}_d has all of the desired properties.

Proof of Lemma 5.2.6. The proof is quite similar to the proof of Lemma 5.2.5. Let us outline the differences. In Step 2 we use a different extension operator, namely

$$a^*(x', x_{\mathsf{d}}) := \sum_{k' \in \left\{-\frac{1}{h} + 1, \dots, \frac{1}{h}\right\}^{\mathsf{d}-1}} \frac{\gamma_{k'}}{\mathsf{e}^{|k'|h}} x_{\mathsf{d}} \, \mathsf{e}^{-|k'|x_{\mathsf{d}}} \, \mathsf{e}^{i\pi k' \cdot x'}.$$

so that $a^*(x', 0) = 0$ and $D^h_{-d}a(x', 0) = g^*_{h,d}(x')$ for $x' \in (-1, 1)^{d-1} \cap (h\mathbb{Z})^{d-1}$. Using Lemma 5.2.2 we then again obtain

$$\|\nabla_{h}^{2}a^{*}\|_{L^{2}_{h}((-1,1)^{d-1}\times[0,2]\cap(h\mathbb{Z})^{d})} \leq C[g_{h,d}]_{H^{\frac{1}{2}}_{h}(\mathbb{R}^{d-1})} \leq Ch^{s-2}\|u\|_{H^{s}(\Omega)}$$

for $s \leq 3$. The localization step remains unchanged. To correct the boundary values we use

$$R_h^* v(x') := \begin{cases} Rv(x') & \text{for } x' \in [0,\infty)^{d-1}, \\ 0 & \text{otherwise}, \end{cases}$$

instead of R_h . By using this projection operator we can then proceed as before.

5.3 Estimates for the finite difference schemes

5.3.1 Summation-by-parts formulae and Poincaré inequalities

For the sake of completeness we record some summation-by-parts formulae that we will use in the following. These formulae are adapted to the two boundary conditions that we encounter in (5.1.2) and (5.1.4). Zero boundary conditions are easier to deal with, so we begin with those.

Lemma 5.3.1. Let $v, \varphi \colon \tilde{\Lambda}_h \to \mathbb{R}$, and assume that $\varphi = D_v^h \varphi = 0$ on Γ_h .

We have that

$$\sum_{z\in\Lambda_h\cup\Gamma_h} h^{\mathsf{d}} \Delta_h^2 v(z)\varphi(z) = \sum_{i,j=1}^{\mathsf{d}} \sum_{z\in\Lambda_h\cup\Gamma_h} h^{\mathsf{d}} D_i^h D_{-j}^h v(z) D_i^h D_{-j}^h \varphi(z) \,. \tag{5.3.1}$$

So, if we define the scalar product $(f,g)_{L^2_{h*}(\Lambda_h\cup\Gamma_h)}$ on functions $f,g: \tilde{\Lambda}_h \to \mathbb{R}^{n\times n}$ by

$$(f,g)_{L^2_{h,*}(\Lambda_h\cup\Gamma_h)}:=\sum_{z\in\Lambda_h}\sum_{i,j=1}^{\mathsf{d}}h^{\mathsf{d}}f_{i,j}(z)g_{i,j}(z),$$

we have

$$(\Delta_h^2 v, \varphi)_{L^2_h(\Lambda_h \cup \Gamma_h)} = (\nabla_h^2 v, \nabla_h^2 \varphi)_{L^2_{h,*}(\Lambda_h \cup \Gamma_h)}.$$
(5.3.2)

Furthermore, we have, for any $i \in \{1, ..., d\}$ *, that*

$$(D_{i}^{h}D_{-i}^{h}v,\varphi)_{L_{h}^{2}(\Lambda_{h}\cup\Gamma_{h})} = (v,D_{i}^{h}D_{-i}^{h}\varphi)_{L_{h}^{2}(\Lambda_{h}\cup\Gamma_{h})}.$$
(5.3.3)

Proof. Observe that we have the summation-by-parts identity

$$\sum_{z \in (h\mathbb{Z})^d} D^h_{\pm i} f(z) g(z) = \sum_{z \in (h\mathbb{Z})^d} f(z) D^h_{\mp i} g(z)$$
(5.3.4)

for $f, g: (h\mathbb{Z})^d \to \mathbb{R}$ such that at least one of f, g has compact support, and $i \in \{1, ..., d\}$ (this follows from the one-dimensional case, where it can be easily checked). This immediately implies (5.3.3).

Next, observe that none of the terms in (5.3.1) depends on values of v or φ outside of $\tilde{\Lambda}_h$. Thus we can extend v and φ by 0 to all of $(h\mathbb{Z})^d$ and prove equivalently that

$$\sum_{z \in (h\mathbb{Z})^{\mathsf{d}}} h^{\mathsf{d}} \Delta_{h}^{2} v(z) \varphi(z) = \sum_{z \in (h\mathbb{Z})^{\mathsf{d}}} h^{\mathsf{d}} \Delta_{h} v(z) \Delta_{h} \varphi(z) = \sum_{i,j=1}^{\mathsf{d}} \sum_{z \in (h\mathbb{Z})^{\mathsf{d}}} h^{\mathsf{d}} D_{i}^{h} D_{-j}^{h} v(z) D_{i}^{h} D_{-j}^{h} \varphi(z).$$

This follows from repeated application of (5.3.4).

For the case of the boundary conditions in (5.1.2), the situation is slightly more involved. We define, for $i, j \in \{1, ..., d\}$ with $i \neq j$, the set

$$\Gamma_h^{ij}:=\{z\in\Gamma_h:z+hA_{ij}\subset [0,1]^{\mathsf{d}}\}$$
 ,

where A_{ij} is the discrete square

$$A_{ij} := \{0, e_i, -e_j, e_i - e_j\},\$$

and note that

$$\in \Gamma_h \setminus \Gamma_h^{ij} \implies (z + hA_{ij}) \cap \Lambda_h = \emptyset \text{ if } i \neq j.$$

Lemma 5.3.2. Let $v, \varphi \colon \tilde{\Lambda}_h \to \mathbb{R}$, and assume that $\varphi = D_{0,v}^h \varphi = 0$ on Γ_h . We then have that

$$\sum_{z \in \Lambda_h \cup \Gamma_h} h^{\mathsf{d}} \Delta_h^2 v(z) \varphi(z) = \sum_{i,j=1}^{\mathsf{d}} \sum_{z \in \Lambda_h} h^{\mathsf{d}} D_i^h D_{-j}^h v(z) D_i^h D_{-j}^h \tilde{\varphi}(z) + \frac{1}{2} \sum_{i=1}^{\mathsf{d}} \sum_{z \in \Gamma_h} h^{\mathsf{d}} D_i^h D_{-i}^h v(z) D_i^h D_{-i}^h \tilde{\varphi}(z) + \sum_{i,j=1}^{\mathsf{d}} \sum_{z \in \Gamma_h^{ij}} h^{\mathsf{d}} D_i^h D_{-j}^h v(z) D_i^h D_{-j}^h \tilde{\varphi}(z).$$
(5.3.5)

So, if we define the scalar product $(f,g)_{L^2_{h,\sim}(\Lambda_h\cup\Gamma_h)}$ on functions $f,g\colon \tilde{\Lambda}_h \to \mathbb{R}^{n\times n}$ by

$$(f,g)_{L^{2}_{h,\sim}(\Lambda_{h}\cup\Gamma_{h})} := \sum_{i,j=1}^{d} \sum_{z\in\Lambda_{h}} h^{d} f_{ij}(z) g_{ij}(z) + \frac{1}{2} \sum_{i=1}^{d} \sum_{z\in\Gamma_{h}} h^{d} f_{ii}(z) g_{ii}(z) + \sum_{\substack{i,j=1\\i\neq j}}^{d} \sum_{z\in\Gamma_{h}^{ij}} h^{d} f_{ij}(z) g_{ij}(z) ,$$

then we have that

$$(\Delta_h^2 v, \varphi)_{L_h^2(\Lambda_h \cup \Gamma_h)} = (\nabla_h^2 v, \nabla_h^2 \varphi)_{L_{h,\sim}^2(\Lambda_h \cup \Gamma_h)}.$$
(5.3.6)

In addition, if we also define for $f, g: \tilde{\Lambda}_h \to \mathbb{R}$ the scalar product

$$(f,g)_{L^2_{h,\sim}(\Lambda_h\cup\Gamma_h)}:=\sum_{z\in\Lambda_h}h^{\mathsf{d}}f_{ij}(z)g_{ij}(z)+\frac{1}{2}\sum_{z\in\Lambda_h}h^{\mathsf{d}}f_{ii}(z)g_{ii}(z),$$

then we have, for any $i \in \{1, ..., d\}$ *, that*

$$(D_{i}^{h}D_{-i}^{h}v,\varphi)_{L_{h}^{2}(\Lambda_{h}\cup\Gamma_{h})} = (D_{i}^{h}D_{-i}^{h}v,\varphi)_{L_{h,\sim}^{2}(\Lambda_{h}\cup\Gamma_{h})} = (v,D_{i}^{h}D_{-i}^{h}\varphi)_{L_{h,\sim}^{2}(\Lambda_{h}\cup\Gamma_{h})}.$$
(5.3.7)

Proof. Define $\tilde{\varphi}: (h\mathbb{Z})^{\mathsf{d}} \to \mathbb{R}$ as $\tilde{\varphi}(z) := \begin{cases} \varphi(z) & \text{for } z \in \Lambda_h \\ 0 & \text{otherwise} \end{cases}$. Then we can apply Lemma 5.3.1 to $v, \tilde{\varphi}$ and obtain

$$\sum_{z\in\Lambda_h\cup\Gamma_h}h^{\mathsf{d}}\Delta_h^2 v(z)\tilde{\varphi}(z) = \sum_{z\in\Lambda_h\cup\Gamma_h}h^{\mathsf{d}}\Delta_h v(z)\Delta_h\tilde{\varphi}(z) = \sum_{i,j=1}^{\mathsf{d}}\sum_{z\in\Lambda_h\cup\Gamma_h}h^{\mathsf{d}}D_i^h D_{-j}^h v(z)D_i^h D_{-j}^h\tilde{\varphi}(z).$$

We trivially have

$$\sum_{e \wedge_h \cup \Gamma_h} h^{\mathsf{d}} \Delta_h^2 v(z) ilde{arphi}(z) = \sum_{z \in \wedge_h \cup \Gamma_h} h^{\mathsf{d}} \Delta_h^2 v(z) arphi(z)$$

Furthermore, $D_i^h D_{-j}^h \tilde{\varphi}(z)$ is equal to $D_i^h D_{-j}^h \varphi(z)$ if $z \in \Lambda_h$. If $z \in \Gamma_h$ we have $D_i^h D_{-i}^h \tilde{\varphi}(z) = \frac{1}{2} D_i^h D_{-i}^h \varphi(z)$ and $D_i^h D_{-j}^h \tilde{\varphi}(z) = \begin{cases} D_i^h D_{-j}^h \varphi(z) & \text{for } z \in \Gamma_h^{ij} \\ 0 & \text{otherwise} \end{cases}$. Therefore,

$$\begin{split} &\sum_{i,j=1}^{\mathsf{d}} \sum_{z \in \Lambda_h \cup \Gamma_h} h^{\mathsf{d}} D_i^h D_{-j}^h v(z) D_i^h D_{-j}^h \tilde{\varphi}(z) = \sum_{i,j=1}^{\mathsf{d}} \sum_{z \in \Lambda_h} h^{\mathsf{d}} D_i^h D_{-j}^h v(z) D_i^h D_{-j}^h \tilde{\varphi}(z) \\ &+ \frac{1}{2} \sum_{i=1}^{\mathsf{d}} \sum_{z \in \Gamma_h} h^{\mathsf{d}} D_i^h D_{-i}^h v(z) D_i^h D_{-i}^h \tilde{\varphi}(z) + \sum_{i,j=1}^{\mathsf{d}} \sum_{z \in \Gamma_h^{ij}} h^{\mathsf{d}} D_i^h D_{-j}^h v(z) D_i^h D_{-j}^h \tilde{\varphi}(z) \,. \end{split}$$

By combining the last three displayed equalities we deduce (5.3.5). With a similar argument we can obtain (5.3.7) from (5.3.3). \Box

Next, we state Poincaré-type inequalities for the two sets of boundary conditions considered.

Lemma 5.3.3. Let $v: \tilde{\Lambda}_h \to \mathbb{R}$, and suppose that $\varphi = D_v^h \varphi = 0$ on Γ_h . Then,

$$\|v\|_{H^{2}_{h}(\Lambda_{h})} \leq C \|\nabla^{2}_{h}v\|_{L^{2}_{h,*}(\Lambda_{h})}.$$
(5.3.8)

Lemma 5.3.4. Let $v: \tilde{\Lambda}_h \to \mathbb{R}$, and suppose that $\varphi = D_{0,v}^h \varphi = 0$ on Γ_h . Then,

$$\|v\|_{H^{2}_{h}(\Lambda_{h})} \leq C \|\nabla_{h}^{2}v\|_{L^{2}_{h,\infty}(\Lambda_{h})}.$$
(5.3.9)

Proof of Lemma 5.3.3. We can extend v by 0 to $(h\mathbb{Z})^d$ without changing the statement of the lemma. Now observe that for $f: (h\mathbb{Z})^d \to \mathbb{R}$ with support contained in a cube of side-length L, and $i \in \{1, ..., d\}$, we have the Poincaré inequality

$$||f||_{L^2_h((h\mathbb{Z})^d)} \le CL ||D^h_{\pm i}f||_{L^2_h((h\mathbb{Z})^d)}.$$

Indeed this follows from the one-dimensional case, which can be proved by a straightforward summation by parts. If we apply this inequality to v and ∇v , we easily deduce (5.3.8).

Proof of Lemma 5.3.4. Let $\tilde{v}(z) := \begin{cases} v(z) & \text{for } z \in \Lambda_h \\ 0 & \text{otherwise} \end{cases}$. Then, \tilde{v} satisfies the assumptions of

Lemma 5.3.3, so that

$$\|\tilde{v}\|_{H^2_h(\Lambda_h)} \leq C \|\nabla_h^2 \tilde{v}\|_{L^2_{h,*}(\Lambda_h)}.$$

Furthermore it is easy to check that

$$\|\nabla_h^2 \tilde{v}\|_{L^2_{h,*}(\Lambda_h)} \le \|\nabla_h^2 v\|_{L^2_{h,\sim}(\Lambda_h)}$$

and

$$\|v\|_{H^2_h(\Lambda_h)} \leq 2\| ilde v\|_{H^2_h(\Lambda_h)}$$
 ,

and hence we directly deduce (5.3.9).

5.3.2 Proofs of the main theorems

We have already sketched the proof of Theorem 5.1.1 in the introduction. We now provide additional details. Parts of the following argument already appeared in the proof of Theorem 4.2.3.

Proof of Theorem 5.1.1. As was mentioned at the start of Section 5.2.1, we can assume that u is supported in $[0, \frac{2}{3})^2$. Let $E: \tilde{\Lambda}_h \to \mathbb{R}$ be defined by E := u - U. Then,

$$\begin{split} E &= 0 & \text{on } \Gamma_h \,, \\ D^h_{0,\nu} E &= D^h_{0,\nu} \tilde{u} & \text{on } \Gamma_h \,. \end{split}$$

Let \hat{E} be the function from Lemma 5.2.5. Then,

$$\begin{aligned} E - \hat{E} &= 0 \quad \text{on } \Gamma_h \,, \\ D^h_{0,\nu}(E - \hat{E}) &= 0 \quad \text{on } \Gamma_h \,. \end{aligned}$$

Therefore, using the results from Section 5.3.1 we deduce that

$$\begin{aligned} \|\nabla_{h}^{2}(E-\hat{E})\|_{L^{2}_{h,\sim}(\Lambda_{h}\cup\Gamma_{h})}^{2} &= (\Delta_{h}^{2}(E-\hat{E}), E-\hat{E})_{L^{2}_{h}(\Lambda_{h}\cup\Gamma_{h})} \\ &= (\Delta_{h}^{2}E, E-\hat{E})_{L^{2}_{h}(\Lambda_{h}\cup\Gamma_{h})} - (\nabla_{h}^{2}\hat{E}, \nabla_{h}^{2}(E-\hat{E}))_{L^{2}_{h,\sim}(\Lambda_{h}\cup\Gamma_{h})}. \end{aligned}$$
(5.3.10)

Using (5.1.9) we can rewrite $\Delta_h^2 E$ as follows

$$\begin{split} \Delta_{h}^{2} E &= \Delta_{h}^{2} \tilde{u} - \Delta_{h}^{2} U = \Delta_{h}^{2} \tilde{u} - T^{2,\dots,2} f = \Delta_{h}^{2} \tilde{u} - T^{h,2,\dots,2} \Delta^{2} \tilde{u} \\ &= \sum_{i=1}^{d} D_{i}^{h} D_{-i}^{h} \Delta_{h} \tilde{u} - T^{h,2,\dots,2} \partial_{i}^{2} \Delta \tilde{u} \\ &= \sum_{i=1}^{d} D_{i}^{h} D_{-i}^{h} \Delta_{h} \tilde{u} - D_{i}^{h} D_{-i}^{h} T_{1}^{h,2} \dots T_{i-1}^{h,2} T_{i+1}^{h,2} \dots T_{d}^{h,2} \Delta \tilde{u} \\ &= \sum_{i=1}^{d} D_{i}^{h} D_{-i}^{h} \varphi_{i} \,, \end{split}$$

where we have abbreviated

$$\varphi_i := \Delta_h \tilde{u} - T_1^{h,2} \dots T_{i-1}^{h,2} T_{i+1}^{h,2} \dots T_d^{h,2} \Delta \tilde{u} \,.$$

If we insert this into (5.3.10) and use the summation-by-parts formula (5.3.7) we arrive at

$$\|\nabla_{h}^{2}(E-\hat{E})\|_{L^{2}_{h,\sim}(\Lambda_{h}\cup\Gamma_{h})}^{2} = \sum_{i=1}^{d} (\varphi_{i}, D_{i}^{h}D_{-i}^{h}(E-\hat{E}))_{L^{2}_{h}(\Lambda_{h}\cup\Gamma_{h})} - (\nabla_{h}^{2}\hat{E}, \nabla_{h}^{2}(E-\hat{E}))_{L^{2}_{h,\sim}(\Lambda_{h}\cup\Gamma_{h})}$$

$$\leq \left(\sum_{i=1}^{\mathsf{d}} \|\varphi_i\|_{L^2_h(\Lambda_h\cup\Gamma_h)} + \|\nabla_h^2 \hat{E}\|_{L^2_h(\Lambda_h\cup\Gamma_h)}\right) \|\nabla_h^2(E-\hat{E})\|_{L^2_{h,\sim}(\Lambda_h\cup\Gamma_h)},$$

and thus

$$\|\nabla_{h}^{2}E\|_{L^{2}_{h,\sim}(\Lambda_{h}\cup\Gamma_{h})} \leq \|\nabla_{h}^{2}\hat{E}\|_{L^{2}_{h}(\Lambda_{h}\cup\Gamma_{h})} + \sum_{i=1}^{d} \|\varphi_{i}\|_{L^{2}_{h}(\Lambda_{h}\cup\Gamma_{h})}.$$
(5.3.11)

The first term on the right-hand side here is bounded by $Ch^{s-2} ||u||_{H^s(\Omega)}$ by construction of \hat{E} . The summands of the sum can be bounded using the Bramble–Hilbert lemma as in the proof of [JS14, Theorem 2.68]. Let us sketch the argument for completeness:

Recall that

$$\varphi_i(x) = \Delta_h \tilde{u}(x) - T_1^{h,2} \dots T_{i-1}^{h,2} T_{i+1}^{h,2} \dots T_d^{h,2} \Delta \tilde{u}(x) \,.$$

Because $s > \frac{d}{2}$,

$$|\Delta_h \tilde{u}(x)| \le C(h) \|\tilde{u}\|_{L^{\infty}(x+(-h,h)^d)} \le C(h) \|\tilde{u}\|_{H^s(x+(-h,h)^d)}.$$

In addition $s > \frac{5}{2}$ implies according to [JS14, Theorem 1.67] that

$$|T_1^{h,2} \dots T_{i-1}^{h,2} T_{i+1}^{h,2} \dots T_d^{h,2} \Delta \tilde{u}(x)| \le C(h) \|\tilde{u}\|_{H^s(x+(-h,h)^d)}.$$

Thus $\varphi_i(x)$ is a bounded linear functional of $\tilde{u} \in H^s(x + (-h, h)^d)$. This functional vanishes when $\tilde{u}|_{x+(-h,h)^d}$ is a polynomial of degree at most 3. Indeed, then $\Delta \tilde{u}(y)$ is equal to some affine function a(y), and $\Delta_h \tilde{u}(x) = a(x)$. On the other hand, the smoothing operators $T_j^{h,2}$ map affine functions to themselves, so that $\varphi_i(x) = 0$.

To summarize $\varphi_i(x)$ is a bounded linear functional of $\tilde{u} \in H^s(x + (-h, h)^d)$ that vanishes on polynomials of degree at most 3. Hence by the Bramble–Hilbert lemma it is bounded by $C(h)[\tilde{u}]_{H^s(x+(-h,h)^d)}$ for the range of *s* as in the statement of the theorem. Using a scaling argument to determine the correct prefactor of *h*, we obtain

$$\|\varphi_{i}\|_{L^{2}_{h,\sim}(\Lambda_{h}\cup\Gamma_{h})} \leq Ch^{s-2}[\tilde{u}]_{H^{s}(\mathbb{R}^{d})} \leq Ch^{s-2}\|u\|_{H^{s}(\Omega)}$$
(5.3.12)

for those *s*.

Now we substitute (5.3.12) into (5.3.11) and obtain the bound

$$\|
abla_h^2 E\|_{L^2_{h,\sim}(\Lambda_h\cup\Gamma_h)}\leq Ch^{s-2}\|u\|_{H^s(\Omega)}$$

for the range of *s* as in the statement of the theorem. The discrete Poincaré inequality, Lemma 5.3.4, immediately implies the asserted error bound. \Box

Proof of Theorem 5.1.2. The proof is the same as that of Theorem 5.1.1. The only differences are that we work with the inner product $(\cdot, \cdot)_{L^2_{h,*}(\Lambda_h \cup \Gamma_h)}$ instead of $(\cdot, \cdot)_{L^2_{h,\sim}(\Lambda_h \cup \Gamma_h)}$, use \hat{E}^* instead of \hat{E} , and Lemma 5.3.3 instead of Lemma 5.3.4.

5.4 Further remarks

5.4.1 Variants and extensions

Let us finally collect some miscellaneous remarks on possible variations of our results and their proofs.

Remark 5.4.1. By Section 5.2.3 we know that there are extensions of the boundary values of \tilde{u} with controlled $\|\cdot\|_{\sim}$ -norm and $\|\cdot\|_*$ -norm respectively. In fact, the optimal such extension is in both cases the biharmonic extension of the boundary values, i.e., the unique function V with the given boundary values that satisfies $\Delta_h^2 V = 0$ in Ω . Indeed, if ψ is a function such that $\psi = 0$, $D_{0,\nu}^h \psi = 0$ on Γ , then

$$\|\nabla_{h}^{2}(V+\psi)\|_{L^{2}_{h,\sim}(\Lambda_{h}\cup\Gamma_{h})}^{2} = \|\nabla_{h}^{2}V\|_{L^{2}_{h,\sim}(\Lambda_{h}\cup\Gamma_{h})}^{2} + \|\nabla_{h}^{2}\psi\|_{L^{2}_{h,\sim}(\Lambda_{h}\cup\Gamma_{h})}^{2} + 2(\nabla_{h}^{2}V,\nabla_{h}^{2}\psi)_{L^{2}_{h,\sim}(\Lambda_{h}\cup\Gamma_{h})}^{2}$$

and $(\nabla_h^2 V, \nabla_h^2 \psi)_{L^2_{h,\sim}(\Lambda_h \cup \Gamma_h)} = 0$, and similarly for $\|\cdot\|_{L^2_{h,*}(\Lambda_h \cup \Gamma_h)}$. This means that we could assume \hat{E} to be discretely biharmonic, and this would simplify the proof of Theorem 5.1.1 slightly. However, for more general fourth-order elliptic operators one cannot use this fact, so we chose to avoid it here.

Remark 5.4.2. Using function space interpolation as in Lemma 5.4.9 it is possible to deduce the intermediate cases of Theorem 5.1.1 and 5.1.2 from the borderline cases s = 4 (or s = 3) and $s = \frac{5}{2} + \varepsilon$. Our method of proof for s = 4 (or s = 3) however directly yields the desired bounds for all relevant *s*, without the need to resort to function space interpolation here.

Remark 5.4.3. Our smoothing operator $T^{h,2,...,2}$ has the advantage that it is given by convolution with a kernel with support in $[-h,h]^d$ so that, when it is applied to f, the values in Λ_h of the resulting function do not involve values of f outside Ω . However, one might want to use stronger mollification operators, as in [JIS85], for example. In particular using $T^{h,3,...,3}$ would allow one to weaken the assumptions on s to $s \ge \frac{1}{2} \max(3, \mathsf{d})$. This is possible if one extends f to a function in $H^{s-4}(\mathbb{R}^d)$ in some way or redefines the finite difference scheme appropriately near the boundary. Apart from this issue, our proof applies equally well to regularization by $T^{h,3,...,3}$. See Theorem 4.2.3 in Chapter 4 for a result of this kind.

Remark 5.4.4. In (5.1.2) and (5.1.4) we regularized the right-hand side by applying $T^{h,2,...,2}$. One might wonder whether some other choice of a regularizing operator, T'_h say, would have been equally appropriate here.

While we do not have a full answer to this question, we shall present a few necessary conditions on T'_h that will clarify why $T^{h,2,\dots,2}$ is a natural choice. We only consider T'_h defined by convolution with some kernel Θ_h , where $\Theta_h = \Theta\left(\frac{\cdot}{h}\right)$ for some $\Theta \colon \mathbb{R}^d \to \mathbb{R}$. As discussed in the previous remark, Θ should have support in $[-1,1]^d$. We want $T'_h f$ to be continuous for each $f \in H^{s-4}(\Omega)$, where $s > \frac{5}{2}$, and this requires $\Theta \in \bigcap_{s > \frac{5}{2}} H^{4-s}(\Omega)$. Furthermore, $T'_h f$ should approximate f in some sense, and thus we require $T'_h f \to f$ pointwise as $h \to 0$ for any $f \in C^{\infty}_c(\Omega)$, say.

Suppose now that the analogues of Theorem 5.1.1 and Theorem 5.1.2 hold with T'_h in place of $T^{h,2,\dots,2}$. This means that $u \mapsto U'$ is uniformly bounded in h as a map from $H^s_0(\Omega)$ into $H^2_h(\Lambda_h)$ for any $\frac{5}{2} < s \leq 4$; note that one can easily verify that $||u||_{H^2_h} \leq C||u||_{H^s(\Omega)}$, uniformly in $h \leq 1$. This means that $f \mapsto T'_h f$ is uniformly bounded in h as a map from $H^{s-4}(\Omega)$ into $H^{-2}_h(\Lambda_h)^4$. and therefore

$$(T'_h f, \varphi)_{L^2_h(\Omega_h)} \leq C \|f\|_{H^{s-4}(\Omega)} \|\varphi\|_{H^2_h(\Omega_h)} \quad \forall \varphi \in H^2_{h,0}(\Lambda_h),$$

uniformly in $h \leq 1$. After a short calculation one sees that this implies that

$$\left\|\sum_{x\in\Lambda_h}\Theta\left(\frac{x-\cdot}{h}\right)\varphi(x)\right\|_{H^{4-s}_0(\Omega)} \le C\|\varphi\|_{H^2_h(\Lambda_h)} \quad \forall \,\varphi\in H^2_{h,0}(\Omega_h)\,,\tag{5.4.1}$$

⁴Here the norm on $H_h^{-2}(\Lambda_h)$ is given as the dual of the norm on $H_{h,0}^2(\Lambda_h)$, the subspace of $H_h^2(\Lambda_h)$ consisting of those functions which are 0 outside of Λ_h .

uniformly in $h \le 1$. In particular the H^{4-s} -seminorm of the term on the left stays bounded as $h \to 0$. Choosing s > 1 and using test functions φ of the form $\varphi(x) = (a \cdot x + b)\eta(x)$ for some cut-off function η that is equal to 1 on some open set, one can show that (5.4.1) implies that

$$\sum_{x \in \mathbb{Z}^{d}} (a \cdot x + b) \Theta(x - \cdot) \quad \text{ is affine for each } a \in \mathbb{R}^{d} \text{ and } b \in \mathbb{R}.$$

This affine function needs to be the same function $y \mapsto a \cdot y + b$ as otherwise $T'_h f$ does not approximate f for functions f that are locally equal to $y \mapsto a \cdot y + b$. Therefore we actually need that

$$\sum_{x \in \mathbb{Z}^{d}} (a \cdot x + b) \Theta(x - \cdot) = a \cdot y + b \text{ is affine for each } a \in \mathbb{R}^{d} \text{ and } b \in \mathbb{R}.$$
 (5.4.2)

This is a rather strong condition on Θ . If we also recall the requirements supp $\Theta \subset [-1,1]^d$ and $\Theta \in \bigcap_{s>\frac{5}{2}} H^{4-s}(\Omega)$, then in dimension d = 1 the only remaining Θ is given by $\Theta(x) = \theta_2(x)$. In dimension d ≥ 2 there are other choices beyond $\Theta(x) = \theta_2(x_1) \cdot \ldots \cdot \theta_2(x_d)$, but that kernel is the unique one if we also demand that it factorizes into functions of the d coordinates. For further results on mollifiers in Sobolev spaces the reader is referred to [JS14, Section 1.9].

5.4.2 Density results

This section is concerned with the various definitions of the space $H_{(k)}^s$ in the introduction. Let us recall what we want to prove.

Lemma 5.4.5. Let $\text{Let } \Xi = I_1 \times \cdots \times I_d$, where $I_j \subset \mathbb{R}$ are (possibly unbounded) open intervals, $s \in \mathbb{R}$, $s \ge 0$, and $k \in \mathbb{N}_0$ such that $k + \frac{1}{2} < s$. Then, the following spaces are equal:

- *i)* $H^s_{(k)}(\Xi)$, the space of all $u \in H^s(\Xi)$ such that the traces of $\partial^i_{\nu} u$ for $0 \le i \le k$ vanish on $\partial \Xi$;
- *ii)* $\overline{\{u \in C^{\infty}(\overline{\Xi}): \partial_{\nu}^{i}u = 0 \text{ on } \partial \Xi \forall i \leq k\}}^{\|\cdot\|_{H^{s}(\Xi)}}$, the closure in the $H^{s}(\Xi)$ -norm of the set of all functions in $C^{\infty}(\overline{\Xi})$ whose derivatives up to order k vanish on $\partial \Xi$;
- *iii*) $H^{s}(\Xi) \cap H_{0}^{\min(k+1,s)}(\Xi)$.

Remark 5.4.6. This result actually holds in far more generality (with basically the same proof): on the one hand one can replace the condition $\partial_{\nu}^{i}u = 0$ for $0 \le i \le k$ by the more general condition $\partial_{\nu}^{i}u = 0$ for $i \in K$, where $K \subset \mathbb{N}$, as long as $s - \frac{1}{2} \notin K$. On the other hand one can take Ξ to be any domain with Lipschitz boundary. The only additional difficulty then is to define $\partial_{\nu}^{i}u$ in view of the fact that ν is in general only a measurable function. However if one defines $\partial_{\nu}^{i}u$ as the appropriate linear combination of the traces of $\partial^{\alpha}u$ for $|\alpha| = i$ (cf. [MM13, p. 156]) the results still hold.

Proof of Lemma 5.4.5. As was already remarked in Section 5.1.3, for the Lipschitz domain Ξ , every function in $C_c^{\infty}(\overline{\Xi})$ is the restriction of a function in $C_c^{\infty}(\mathbb{R}^d)$ to Ξ . In particular, we have that $C_c^{\infty}(\overline{\Xi}) \subset H^s(\Xi)$.

We will prove the inclusions

$$\overline{\left\{u \in C_c^{\infty}(\overline{\Xi}) : \partial_{\nu}^i u = 0 \text{ on } \partial\Xi \ \forall \ i \le k\right\}}^{\|\cdot\|_{H^s(\Xi)}} \subset H^s(\Xi) \cap H_0^{\min(k+1,s)}(\Xi) , \qquad (5.4.3)$$

$$H^{s}(\Xi) \cap H_{0}^{\min(k+1,s)}(\Xi) \subset H^{s}_{(k)}(\Xi),$$
 (5.4.4)

$$H^{s}_{(k)}(\Xi) \subset \overline{\left\{ u \in C^{\infty}_{c}(\overline{\Xi}) : \partial^{i}_{\nu}u = 0 \text{ on } \partial\Xi \,\forall \, i \leq k \right\}}^{\|\cdot\|_{H^{s}(\Xi)}}.$$
(5.4.5)

.. ..

The inclusion (5.4.4) follows immediately from the definitions and standard trace theorems. Next observe that trivially

$$\overline{\left\{u\in C^\infty_c(\overline{\Xi})\colon \partial^i_
u u=0 ext{ on } \partial\Xi ext{ } i\leq k
ight\}}^{\|\cdot\|_{H^s(\Xi)}}\subset H^s(\Xi)$$
 ,

so in order to prove (5.4.3) we only need to verify that

$$\overline{\left\{u\in C_c^{\infty}(\overline{\Xi})\colon \partial_{\nu}^{i}u=0 \text{ on } \partial\Xi \,\forall\, i\leq k\right\}}^{\|\cdot\|_{H^{s}(\Xi)}}\subset H_0^{\min(k+1,s)}(\Xi).$$

To see this, it suffices to prove that

$$\overline{\left\{u\in C_c^{\infty}(\overline{\Xi})\colon \partial_{\nu}^{i}u=0 \text{ on } \partial\Xi \,\forall i\leq k\right\}}^{\|\cdot\|_{\mathbf{H}^{\min(k+1,s)}(\Xi)}}\subset H_0^{\min(k+1,s)}(\Xi).$$

This follows from general theory (e.g. [MM13, Theorem 3.18]), but it is also easy to verify by direct calculations: we need to check that we can approximate any function $v \in \{u \in C_c^{\infty}(\overline{\Xi}) : \partial_v^i u = 0 \text{ on } \partial\Xi \ \forall i \leq k\}$ with $C_c^{\infty}(\Xi)$ -functions in the $H^{\min(k+1,s)}$ -norm.

The proof of this assertion proceeds as follows. The assumptions on v imply that the extension \bar{v} of v by 0 to \mathbb{R}^d is in $C^k(\mathbb{R}^d)$. In addition, $\bar{v} \in H^{k+1}(\mathbb{R}^d)$. To verify this one can use that all derivatives of v of order k are continuous, have zero trace, and are in $H^1(\Xi)$. Hence, their extensions by zero belong to $H^1(\mathbb{R}^d)$. This is well known for general Lipschitz domains (and is easily seen by a partition of unity argument and transformation to the half-space situation by composition with a bi-Lipschitz map). Now dilation is continuous in $H^{k+1}(\mathbb{R}^d)$, and hence v can be approximated by $H_0^{k+1}(\Xi)$ functions in the H^{k+1} norm. Thus, $v \in H_0^{k+1}(\Xi)$. Consequently, v can be approximated in the $\|\cdot\|_{H^{k+1}}$ norm, and in particular in the possibly weaker norm $\|\cdot\|_{H^{\min(s,k+1)}}$ by $C_c^{\infty}(\overline{\Xi})$ functions.

It remains to prove (5.4.5). We first observe that

$$\left\{ u \in C_c^{\infty}(\overline{\Xi}) \colon \partial_{\nu}^i u = 0 \text{ on } \partial\Xi \,\forall \, i \leq k \right\} = C_c^{\infty}(\overline{\Xi}) \cap H^s_{(k)}(\Xi).$$

Taking this into account, we need to verify that $C_c^{\infty}(\overline{\Xi}) \cap H_{(k)}^s(\Xi)$ is dense in $H_{(k)}^s(\Xi)$. It is easy to see that $C_c^{\infty}(\overline{\Xi}) \cap H_{(k)}^s(\Xi)$ is dense in $C^{\infty}(\overline{\Xi}) \cap H_{(k)}^s(\Xi)$, so it remains to prove that the latter space is dense in $H_{(k)}^s(\Xi)$. To see this we apply the criterion of Burenkov [Bur98, Theorem 2 on p. 49]. The first three assumptions of that theorem are obviously satisfied, and for the fourth we need to check that every $u \in H_{(k)}^s(\Xi)$ of compact support is continuous under translations, which is once again clear.

5.4.3 Remarks on Interpolation

In this section we shall collect and discuss various results on interpolation spaces that were used in our work. As we only consider Hilbert spaces, we do not need the theory of interpolation spaces in its full generality and can make some simplifications.

We consider two separable Hilbert spaces *X* and *Y* such that $X \subset Y$ is dense and the injection is continuous. Then, given $\theta \in [0, 1]$, we can consider the associated interpolation spaces $[X, Y]_{\theta} := D(\Lambda^{1-\theta})$ equipped with the graph norm, where Λ is a self-adjoint positive operator on *Y* with domain *X* (see [LM72a, Section 2] for details, starting with the nontrivial

fact that such a Λ always exists). Because we are considering Hilbert spaces, this definition yields up to equivalence of norms the same space as the complex interpolation space $[X, Y]_{[\theta]}$ or the real interpolation space $[X, Y]_{\theta,2}$ (see [LM72a, Section 14.2 and Section 15] for proofs). Thus we will be able to freely use results for either of these interpolation techniques from the literature.

Our first task is to study whether the spaces $H^s_{(k)}(\Xi)$ form an interpolation scale, where $\Xi \subset \mathbb{R}^d$ is open and connected. If Ξ has a smooth boundary this was shown in [Gri67] with an alternative proof in [Löf92]. However we are interested in the cases $\Xi = (0, 1)^d$ or $\Xi = (0, \infty)^d$, and the two aforementioned proofs do not easily extend to that case. On the other hand, if Ξ has Lipschitz boundary then there are results concerning the interpolation scales $H^s(\Xi)$ and $H^s_0(\Xi)$ (see e.g. [Bra95]), but not for our mixed case.

Fortunately, in our case it is possible to use the fact that our domain is a cartesian product in combination with results from [LM72a] to give a proof of the desired result by induction on the dimension.

We begin by stating a one-dimensional but vector-valued result that we will need in the proof of the following lemmas. In addition to the notation from the introduction, we define $H^s_{\#}(I)$, where $I \subset \mathbb{R}$ is an open interval such that $0 \in I$, as the closure of the linear space of functions u contained in $C^{\infty}(I) \cap H^s(I)$ with u'(0) = 0 in the $\|\cdot\|_{H^s}$ -norm.

Lemma 5.4.7. Let *E* be a separable Hilbert space, $I \subset \mathbb{R}$ a (possibly unbounded) open interval, and $k \in \mathbb{N}$. Let $s \ge t \ge 1$, and let $\theta \in (0, 1)$. If $s - \frac{1}{2} \notin \{0, 1, \dots, k\}$ and $(1 - \theta)s - \frac{1}{2} \notin \{0, 1, \dots, k\}$, then we have that

$$\left[H^{s}_{(k)}(I,E), L^{2}(I,E)\right]_{\theta} = H^{(1-\theta)s}_{(k)}(I,E).$$
(5.4.6)

Furthermore, if $0 \in I$ *,* $s \neq \frac{3}{2}$ *and* $(1 - \theta)s \neq \frac{3}{2}$ *, then*

$$\left[H_{\#}^{s}(I,E),L^{2}(I,E)\right]_{\theta} = H_{\#}^{(1-\theta)s}(I,E).$$
(5.4.7)

Proof. If $E = \mathbb{R}$ then (5.4.6) is a special case of [Gri67, Théorème 8.1]. The Hilbert-space-valued case follows from a simple general tensorization argument, see Lemma 5.4.8 below.

For (5.4.7) it suffices again to consider the case $E = \mathbb{R}$. The inclusion " \subset " is straightforward. For the converse inclusion we adapt the strategy from [Gri67]. Our goal is to construct for any given $f \in H_{(k)}^{(1-\theta)s}(I)$ some $u \in L^2(\mathbb{R}^+, H_{(k)}^s(I) \cap H^{\frac{1}{2\theta}}(\mathbb{R}^+, L^2(I))$ with $u(\cdot, 0) = f$ (cf. [Gri67, Definition 2.2]).

If $s < \frac{3}{2}$ then $H^s_{\#}(I) = H^s(I)$ and the assertion follows by standard results. Thus we may assume $s > \frac{3}{2}$.

We first assume that $s - \frac{1}{2} \notin \mathbb{N}$ and $(1 - \theta)s - \frac{1}{2} \notin \mathbb{N}$. We first define the extension *u* of *f* on $I \cap \mathbb{R}^+$ and $I \cap \mathbb{R}^-$ separately. Let $\eta \in C^{\infty}([0, 1))$ with $\eta = 1$ on $[0, \frac{1}{2}]$ and set, for $x \in I \cap \mathbb{R}^+$,

$$\begin{split} f_0^{\pm}(x) &= f(x), \\ f_k &= 0 \quad \text{for } 1 \le k < \frac{1}{2\theta} - \frac{1}{2}, \\ g_1^{\pm}(0, y) &= 0, \\ g_j^{\pm} &= (\pm 1)^j \frac{\partial^j f}{\partial \nu^j}(0) \eta(y) \quad \text{for } j < (1 - \theta)s - \frac{1}{2} \text{ and } j \ne 1, \\ g_j^{\pm}(0, y) &= 0 \qquad \text{for } (1 - \theta)s - \frac{1}{2} \le j < s - \frac{1}{2}. \end{split}$$

Then, the compatibility conditions in [Gri67, Théorème 7.2] are satisfied and thus there exist

$$u^{\pm} \in L^{2}(\mathbb{R}^{+}, H^{s}(I \cap \mathbb{R}^{\pm})) \cap H^{\frac{1}{2\theta}}(\mathbb{R}^{+}, L^{2}(I \cap \mathbb{R}^{\pm}))$$

such that

$$u^{\pm}(0,x) = f(x),$$

$$\frac{\partial^{j}}{\partial v^{j}}u^{\pm}(y,0) = g_{j}^{\pm}(y).$$

Set $u(y, x) = u^{\pm}(y, x)$ for $\pm x > 0$. Then, in particular u'(y, 0) = 0. The condition $f \in H^{(1-\theta)s}$ implies that $(-1)^j g_j^- = g_j^+$ for $j < (1-\theta)s - \frac{1}{2}$ and hence this condition holds for all j. Because $s \notin \mathbb{N}_0 + \frac{1}{2}$ it follows that $u(\cdot, y) \in H^s(I)$ for all y > 0 and in fact $u(\cdot, y) \in H^s_{\#}(I)$. Thus,

$$u \in L^2(\mathbb{R}^+, H^s_{\#}(I)) \cap H^{\frac{1}{2\theta}}(\mathbb{R}^+, L^2(I))$$

and u(0, x) = f(x). By [Gri67, Definition 2.2] we deduce that $f \in [H^s_{\#}(I), L^2(I)]_{\theta}$, and this concludes the proof of (5.4.7).

It remains to remove the assumptions $(1 - \theta)s - \frac{1}{2} \notin \mathbb{N}$ and $s - \frac{1}{2} \notin \mathbb{N}$. This can easily be handled by using [LM72a, Theorem 13.3] and reiteration. For the convenience of the reader we give the details.

Consider the case $s - \frac{1}{2} \in \mathbb{N}$, but $(1 - \theta)s - \frac{1}{2} \notin \mathbb{N}$. Let $s_* > s > \frac{3}{2}$ be such that $s_* - \frac{1}{2} \notin \mathbb{N}$ and let θ_* be such that $s = (1 - \theta_*)s_*$. By the reiteration theorem [LM72a, Theorem 6.1] we have

$$\left[H_{\#}^{s}(I), L^{2}(I)\right]_{\theta} = \left[\left[H_{\#}^{s_{*}}(I), L^{2}(I)\right]_{\theta_{*}}, L^{2}(I)\right]_{\theta} = \left[H_{\#}^{s_{*}}(I), L^{2}(I)\right]_{\theta+\theta_{*}-\theta\theta_{*}}$$

and the right-hand side equals $H_{\#}^{(1-\theta)s}(I)$ by what we have already shown (note that $(1-(\theta+\theta_*-\theta\theta_*))s_*=(1-\theta)s$).

Next consider the remaining case that $(1 - \theta)s - \frac{1}{2} \in \mathbb{N}$. Choose $\theta_- < \theta < \theta_+$ close enough to θ such that $(1 - \theta_{\pm})s - \frac{1}{2} \notin \mathbb{N}$ and $\frac{3}{2} \notin [(1 - \theta_+)s, (1 - \theta_-)s]$. Let $\tilde{\theta}$ be such that $\theta = (1 - \tilde{\theta})\theta_- + \tilde{\theta}\theta_+$. Again by reiteration and the previous results we have

$$\left[H_{\#}^{s}(I), L^{2}(I)\right]_{\theta} = \left[\left[H_{\#}^{s}(I), L^{2}(I)\right]_{\theta_{-}}, \left[H_{\#}^{s}(I), L^{2}(I)\right]_{\theta_{+}}\right]_{\tilde{\theta}} = \left[H_{\#}^{(1-\theta_{-})s}(I), H_{\#}^{(1-\theta_{+})s}(I)\right]_{\tilde{\theta}}$$

and it suffices to show that the right-hand side equals $H_{\#}^{(1-\tilde{\theta})(1-\theta_{-})s+\tilde{\theta}(1-\theta_{+})s}(I) = H_{\#}^{(1-\theta)s}(I)$. To that end, observe that

$$H^t_{\#}(I) = \begin{cases} \{f \in H^t(I) \colon f'(0) = 0\} & \text{for } t > \frac{3}{2}, \\ H^t(I) & \text{for } t < \frac{3}{2}, \end{cases}$$

is a closed subspace of $H^t(I)$ of finite codimension for any $t \neq \frac{3}{2}$. Now [LM72a, Theorem 13.3] implies that for $t < t' < \frac{3}{2}$ or $\frac{3}{2} < t < t'$ and $\hat{\theta} \in [0, 1]$ we have

$$\left[H_{\#}^{t'}(I), H_{\#}^{t}(I)\right]_{\hat{\theta}} = H_{\#}^{(1-\hat{\theta})t'+\hat{\theta}t}(I) \,.$$

In particular,

$$\left[H_{\#}^{(1-\theta_{-})s}(I), H_{\#}^{(1-\theta_{+})s}(I)\right]_{\tilde{\theta}} = H_{\#}^{(1-\tilde{\theta})(1-\theta_{-})s+\tilde{\theta}(1-\theta_{+})s}(I) \,.$$

This completes the proof.

Lemma 5.4.8. Let $X \subset Y$ be Hilbert spaces of real-valued functions, as above. Let *E* be a separable Hilbert space and denote by $X \otimes E$ and $Y \otimes E$ the corresponding spaces of *E*-valued functions. Then, for all $\theta \in (0, 1)$,

$$[X \otimes E, Y \otimes E]_{\theta} = [X, Y]_{\theta} \otimes E$$

with equivalent norms. Here for a Hilbert space Z of real-valued functions the scalar product on $Z \otimes E$ is defined as follows. If $(e_m)_{m=1}^{\infty}$ is an orthonormal basis of E and $f = \sum_{m=1}^{\infty} f_m e_m$, $g = \sum_{m=1}^{\infty} g_m e_m$, then

$$(f,g)_{X\otimes E} = \sum_{m=1}^{\infty} (f_m,g_m)_X.$$

Proof. To show the inclusion " \supset " let $a \in [X, Y]_{\theta} \otimes E$ and $\delta > 0$. Then $a_m \in [X, Y]_{\theta}$ and by [Gri67, Definition 2.2]) there exist $u_m \in L^2(\mathbb{R}^+, X) \cap H^{\frac{1}{2\theta}}(\mathbb{R}^+, Y)$ with $a_m = u_m(0)$ and

$$||u_m||_{L^2(\mathbb{R}^+,X)} + ||u_m||_{H^{\frac{1}{2\theta}}(\mathbb{R}^+,Y)} \le (1+\delta) ||a_m||_{[X,Y]_{\theta}}, \qquad m = 1,2,\ldots.$$

Taking the square and summing over *m* we see that

$$S := \sum_{m=1}^{\infty} \|u_m\|_{L^2(\mathbb{R}^+, X)}^2 + \|u_m\|_{H^{\frac{1}{2\theta}}(\mathbb{R}^+, Y)}^2 \le (1+\delta)^2 \|a\|_{[X,Y]_{\theta} \otimes E}^2.$$

Set $u := \sum_{m=1}^{\infty} u_m$. Since $S < \infty$ we see that $u \in L^2(\mathbb{R}^+, X \otimes E) \cap H^{\frac{1}{2\theta}}(\mathbb{R}^+, Y \otimes E)$. Thus $a = u(0) \in [X \otimes E, Y \otimes E]_{\theta}$ and

$$\|a\|_{[X\otimes E,Y\otimes E]_{\theta}}^{2} \leq 2S \leq 2(1+\delta)^{2} \|a\|_{[X,Y]_{\theta}\otimes E}^{2}.$$

The proof of the converse inclusion, " \subset ", is similar: Let $a \in [X \otimes E, Y \otimes E]_{\theta}$. Then there is a $u \in L^2(\mathbb{R}^+, X \otimes E) \cap H^{\frac{1}{2\theta}}(\mathbb{R}^+, Y \otimes E)$ with u(0) = a and

$$\|u\|_{L^{2}(\mathbb{R}^{+},X\otimes E)}+\|u\|_{H^{\frac{1}{2\theta}}(\mathbb{R}^{+},Y\otimes E)}\leq (1+\delta)\|a\|_{[X\otimes E,Y\otimes E]_{\theta}}$$

In particular $u_m = (u, e_m)_E$ satisfies $u_m \in L^2(\mathbb{R}^+, X) \cap H^{\frac{1}{2\theta}}(\mathbb{R}^+, Y)$, m = 1, 2, ... Thus $a_m = u_m(0) \in [X, Y]_{\theta}$, and we have

$$\begin{split} \|a\|_{[X,Y]_{\theta}\otimes E}^{2} &= \sum_{m=1}^{\infty} \|a_{m}\|_{[X,Y]_{\theta}}^{2} \\ &\leq 2\sum_{m=1}^{\infty} \|u_{m}\|_{L^{2}(\mathbb{R}^{+},X)}^{2} + \|u_{m}\|_{H^{\frac{1}{2\theta}}(\mathbb{R}^{+},Y)}^{2} \\ &= 2\|u\|_{L^{2}(\mathbb{R}^{+},X)}^{2} + 2\|u\|_{H^{\frac{1}{2\theta}}(\mathbb{R}^{+},Y)}^{2} \\ &\leq 2(1+\delta)^{2}\|a\|_{[X\otimes E,Y\otimes E]_{\theta}}^{2}. \end{split}$$

That completes the proof of the lemma.

Now we can establish the desired interpolation results in higher dimensions. For the following lemma, we are interested in the cases $\Xi = \Omega$ and k = 1 or $\Xi = (0, \infty)^d$ and k = 0.

Lemma 5.4.9. Let $\Xi = I_1 \times \cdots \times I_d$, where $I_j \subset \mathbb{R}$ are (possibly unbounded) open intervals. Let $K = \{k_1, \ldots, k_m\} \subset \mathbb{N}_0$. Let $s \ge t \ge 0$, and let $\theta \in [0, 1]$. If none of $s - \frac{1}{2}$, $t - \frac{1}{2}$ and $(1 - \theta)s + \theta t - \frac{1}{2}$ are in $\{0, 1, \ldots, k\}$, then

$$\left[H^{s}_{(k)}(\Xi), H^{t}_{(k)}(\Xi)\right]_{\theta} = H^{(1-\theta)s+\theta t}_{(k)}(\Xi), \qquad (5.4.8)$$

and, in particular, if $s \notin \{0, 1, ..., k\}$, $(1 - \theta)s - \frac{1}{2} \notin \{0, 1, ..., k\}$, then

$$\left[H^{s}_{(k)}(\Xi), L^{2}(\Xi)\right]_{\theta} = H^{(1-\theta)s}_{(k)}(\Xi).$$
(5.4.9)

Proof. The identity (5.4.8) immediately follows from (5.4.9) and reiteration, so it suffices to establish (5.4.9).

We proceed by induction on d. The case d = 1 was established in Lemma 5.4.7. Now assume that the theorem holds for d - 1 dimensions. The following argument is similar to the one in Section 2.1 in [LM72b].

Let $\Xi' = I_1 \times \cdots \times I_{d-1}$, and write $x = (x', x_d)$. If we interpret a function $\Xi \to \mathbb{R}$ as a function $I_d \to (\Xi' \to \mathbb{R})$, we claim that

$$L^{2}(\Xi) = L^{2}(I_{d}, L^{2}(\Xi'))$$
(5.4.10)

and

$$H^{s}_{(k)}(\Xi) = L^{2}(I_{\mathsf{d}}, H^{s}_{(k)}(\Xi')) \cap H^{s}_{(k)}(I_{\mathsf{d}}, L^{2}(\Xi')).$$
(5.4.11)

Indeed, (5.4.10) is obvious. For (5.4.11) one can argue as follows. It is well-known (and can be proved using the Fourier transform, for example) that

$$H^{s}(\mathbb{R}^{\mathsf{d}}) = L^{2}(\mathbb{R}, H^{s}(\mathbb{R}^{\mathsf{d}-1})) \cap H^{s}(\mathbb{R}, L^{2}(\mathbb{R}^{\mathsf{d}-1})).$$
(5.4.12)

The sets Ξ' and I_d have Lipschitz boundary, and so there exists an extension operator E, mapping $H^t(\Xi)$ continuously to $H^t(\mathbb{R}^d)$ for $t \in \{0, s\}$, that also maps $H^t(\Xi' \times \{x_d\})$ continuously to $H^t(\mathbb{R}^{d-1} \times \{x_d\})$ for any $x_d \in I_d$. One can construct such an E by first applying an appropriate extension operator on each slice $\Xi' \times \{x_d\} \subset \mathbb{R}^{d-1} \times \{x_d\}$ and then extending in the direction e_d . Using this extension operator, one can easily check that (5.4.12) implies also that

$$H^{s}(\Xi) = L^{2}(I_{\mathsf{d}}, H^{s}(\Xi')) \cap H^{s}(I_{\mathsf{d}}, L^{2}(\Xi')).$$

From this we want to deduce (5.4.11) by considering the faces of Ξ separately. We begin with " \subset " in (5.4.11). Let $u \in H^s_{(k)}(\Xi)$, and take $j \leq d - 1$. Let $\Xi'_{j,\pm} \times I_d$ be the two faces of Ξ orthogonal to e_j . By assumption the trace of $\partial^i_j u$ for $i \leq k$ vanishes on $\Xi'_{j,\pm} \times I_d$ as an element of $H^{s-j-1/2}(\Xi'_{j,\pm} \times I_d)$ and thus also as an element of $L^2(I_d, H^{s-j-1/2}(\Xi'_{j,\pm}))$. In particular, for almost every x_d the trace of $\partial^i_j u(\cdot, x_d)$ vanishes on $\Xi'_{j,\pm} \times \{x_d\}$. This holds for all $j \leq d - 1$ and all $i \leq k$, and so $u \in L^2(I_d, H^s_{(k)}(\Xi'))$. We can argue similarly for the case j = d to deduce that $u \in H^s_{(k)}(I_d, L^2(\Xi'))$ and have thus shown " \subset " in (5.4.11). The argument for " \supset " is analogous. Thus we have established (5.4.11).

We have that $L^2(I_d, H^s_{(k)}(\Xi'))$ is the domain of an unbounded positive operator Λ_1 on $L^2(I_d, L^2(\Xi'))$, and Λ_1 is an operator in x', independent of x_d . Similarly, $H^s_{(k)}(I_d, L^2(\Xi'))$ is the domain of an unbounded positive operator Λ_2 on $L^2(I_d, L^2(\Xi'))$, and Λ_2 is an operator in x', independent of x_d . In particular, Λ_1 and Λ_2 commute. Thus we can apply the criterion for the interpolation space of an intersection [LM72a, Theorem 13.1] and obtain that

$$\begin{split} \left[H^{s}_{(k)}(\Xi), L^{2}(\Xi) \right]_{\theta} \\ &= \left[L^{2}(I_{d}, H^{s}_{(k)}(\Xi')) \cap H^{s}_{(k)}(I_{d}, L^{2}(\Xi')), L^{2}(I_{d}, L^{2}(\Xi')) \right]_{\theta} \\ &= \left[L^{2}(I_{d}, (H^{s}_{(k)}(\Xi')), L^{2}(I_{d}, L^{2}(\Xi')) \right]_{\theta} \cap \left[H^{s}_{(k)}(I_{d}, L^{2}(\Xi')), L^{2}(I_{d}, L^{2}(\Xi')) \right]_{\theta} \end{split}$$

Now, according to [LM72a, Remark 14.4] and using the induction hypothesis (5.4.9) for d - 1, we have

$$\begin{split} \left[L^2(I_{\mathsf{d}}, H^s_{(k)}(\Xi')), L^2(I_{\mathsf{d}}, L^2(\Xi')) \right]_{\theta} &= L^2 \left(I_{\mathsf{d}}, \left[H^s_{(k)}(\Xi'), L^2(\Xi') \right]_{\theta} \right) \\ &= L^2(I_{\mathsf{d}}, H^{(1-\theta)s}_{(k)}(\Xi')) \,. \end{split}$$

Similarly, using (5.4.6), we find

$$\left[H^{s}_{(k)}(I_{\mathsf{d}}, L^{2}(\Xi')), L^{2}(I_{\mathsf{d}}, L^{2}(\Xi'))\right]_{\theta} = H^{(1-\theta)s}_{(k)}(I_{\mathsf{d}}, L^{2}(\Xi')).$$

If we combine the last three equalities, we deduce that

$$\begin{bmatrix} H_{(k)}^{s}(\Xi), L^{2}(\Xi) \end{bmatrix}_{\theta} = L^{2}(I_{\mathsf{d}}, H_{(k)}^{(1-\theta)s}(\Xi')) \cap H_{(k)}^{(1-\theta)s}(I_{\mathsf{d}}, L^{2}(\Xi'))$$
$$= H_{(k)}^{(1-\theta)s}(\Xi) \,.$$

That completes the proof of the lemma.

For the next lemma recall the definition of G^s from the proof of Lemma 5.2.4.

Lemma 5.4.10. Let $s \ge t \ge 0$, and let $\theta \in (0,1)$. Then, if none of s, t and $(1-\theta)s + \theta t$ are in $\{\frac{1}{2}, \frac{3}{2}\}$, we have

$$\left[G^{s}((0,\infty)^{\mathsf{d}-1}\times\mathbb{R}), G^{t}((0,\infty)^{\mathsf{d}-1}\times\mathbb{R})\right]_{\theta} = G^{(1-\theta)s+\theta t}((0,\infty)^{\mathsf{d}-1}\times\mathbb{R}), \quad (5.4.13)$$

and in particular, if $s \notin \{\frac{1}{2}, \frac{3}{2}\}, (1-\theta)s \notin \{\frac{1}{2}, \frac{3}{2}\}$, then

$$\left[G^{s}((0,\infty)^{\mathsf{d}-1}\times\mathbb{R}),L^{2}((0,\infty)^{\mathsf{d}-1}\times\mathbb{R})\right]_{\theta}=G^{(1-\theta)s}((0,\infty)^{\mathsf{d}-1}\times\mathbb{R}).$$
(5.4.14)

Proof. As in the previous lemma, (5.4.13) follows from (5.4.14) and reiteration, so we will only prove (5.4.14). Observe that

$$G^{s}((0,\infty)^{d-1} \times \mathbb{R}) = L^{2}(\mathbb{R}, H^{s}_{(0)}((0,\infty)^{d-1})) \cap H^{s}_{\#}(\mathbb{R}, L^{2}((0,\infty)^{d-1}))$$

and

$$L^{2}((0,\infty)^{d-1} \times \mathbb{R}) = L^{2}(\mathbb{R}, L^{2}((0,\infty)^{d-1})).$$

Intersection and interpolation commute by the same argument as in the proof of Lemma 5.4.9, and so we have

$$\begin{split} \left[G^{s}((0,\infty)^{\mathsf{d}-1} \times \mathbb{R}), L^{2}((0,\infty)^{\mathsf{d}-1} \times \mathbb{R}) \right]_{\theta} &= \left[L^{2}(\mathbb{R}, H^{s}_{(0)}((0,\infty)^{\mathsf{d}-1}), L^{2}(\mathbb{R}, L^{2}((0,\infty)^{\mathsf{d}-1})) \right]_{\theta} \\ &\cap \left[H^{s}_{\#}(\mathbb{R}, L^{2}((0,\infty)^{\mathsf{d}-1}), L^{2}(\mathbb{R}, L^{2}((0,\infty)^{\mathsf{d}-1})) \right]_{\theta} . \end{split}$$

Now, by Lemma 5.4.9 we have

$$\left[L^{2}(\mathbb{R}, H^{s}_{(0)}((0, \infty)^{\mathsf{d}-1}), L^{2}(\mathbb{R}, L^{2}((0, \infty)^{\mathsf{d}-1}))\right]_{\theta} = L^{2}(\mathbb{R}, H^{(1-\theta)s}_{(0)}((0, \infty)^{\mathsf{d}-1}), L^{2}(\mathbb{R}, L^{2}((0, \infty)^{\mathsf{d}-1})))$$

and Lemma 5.4.7 implies that

$$\left[H^{s}_{\#}(\mathbb{R}, L^{2}((0, \infty)^{\mathsf{d}-1}), L^{2}(\mathbb{R}, L^{2}((0, \infty)^{\mathsf{d}-1}))\right]_{\theta} = H^{(1-\theta)s}_{\#}(\mathbb{R}, L^{2}((0, \infty)^{\mathsf{d}-1}).$$

The last three equalities combined imply (5.4.14).

6 Pinning for the membrane model in dimension four and above

This chapter is based on the author's preprint [Sch20b], with only minor changes.

6.1 Introduction

In this chapter we study the pinned membrane model in dimension four and above. The problem and our main results were described in Section 1.3.4. In this chapter we are focussed on the membrane model, and so we drop all subscripts Δ .

Recall that we consider the membrane model

$$\mathbb{P}_{\Lambda}(\mathrm{d}\psi) = \frac{1}{Z_{\Lambda}} \exp\left(-\frac{1}{2} \sum_{x \in \mathbb{Z}^{\mathrm{d}}} |\Delta_{1}\psi_{x}|^{2}\right) \prod_{x \in \Lambda} \mathrm{d}\psi_{x} \prod_{x \in \mathbb{Z}^{\mathrm{d}} \setminus \Lambda} \delta_{0}(\mathrm{d}\psi_{x}) \,. \tag{6.1.1}$$

on some $\Lambda \Subset \mathbb{Z}^d$ as well as the membrane model with δ -pinning of strength $\varepsilon > 0$,

$$\mathbb{P}^{\varepsilon}_{\Lambda}(d\psi) = \frac{1}{Z^{\varepsilon}_{\Lambda}} \exp\left(-\frac{1}{2} \sum_{x \in \mathbb{Z}^{d}} |\Delta_{1}\psi_{x}|^{2}\right) \prod_{x \in \Lambda} (d\psi_{x} + \varepsilon \delta_{0}(d\psi_{x})) \prod_{x \in \mathbb{Z}^{d} \setminus \Lambda} \delta_{0}(d\psi_{x}).$$
(6.1.2)

6.1.1 Main results

Let us describe our results in detail. First of all, expanding the bracket in (6.1.2), we see that for $f \colon \mathbb{R}^{\mathbb{Z}^d} \to \mathbb{R}$ we have

$$\mathbb{E}_{\Lambda}^{\varepsilon}(f) = \frac{1}{Z_{\Lambda}^{\varepsilon}} \int \exp\left(-\frac{1}{2} \sum_{x \in \mathbb{Z}^{d}} |\Delta_{1}\psi_{x}|^{2}\right) f(\psi) \prod_{x \in \Lambda} (d\psi_{x} + \varepsilon \delta_{0}(d\psi_{x})) \prod_{x \in \mathbb{Z}^{d} \setminus \Lambda} \delta_{0}(d\psi_{x})$$

$$= \frac{1}{Z_{\Lambda}^{\varepsilon}} \sum_{A \subset \Lambda} \int \exp\left(-\frac{1}{2} \sum_{v \in \mathbb{Z}^{d}} |\Delta_{1}\psi_{x}|^{2}\right) f(\psi) \varepsilon^{|A|} \prod_{x \in \Lambda \setminus A} d\psi_{v} \prod_{x \in \mathbb{Z}^{d} \setminus (\Lambda \setminus A)} \delta_{0}(d\psi_{x}) \quad (6.1.3)$$

$$= \sum_{A \subset \Lambda} \frac{\varepsilon^{|A|} Z_{\Lambda \setminus A}}{Z_{\Lambda}^{\varepsilon}} \mathbb{E}_{\Lambda \setminus A}(f) .$$

where $\mathbb{E}^{\varepsilon}_{\Lambda}$ and $\mathbb{E}_{\Lambda \setminus A}$ denote the expectation with respect to $\mathbb{P}^{\varepsilon}_{\Lambda}$ and $\mathbb{P}_{\Lambda \setminus A}$, respectively. Thus, we have

$$\mathbb{P}^{\varepsilon}_{\Lambda}(\mathrm{d}\psi) = \sum_{A \subset \Lambda} \zeta^{\varepsilon}_{\Lambda}(A) \mathbb{P}_{\Lambda \setminus A}(\mathrm{d}\psi)$$
(6.1.4)

where

$$\zeta^{arepsilon}_{\Lambda}(A) = rac{arepsilon^{|A|}Z_{\Lambdaackslash A}}{Z^{arepsilon}_{\Lambda}}$$

so that $\zeta_{\Lambda}^{\epsilon}$ is a probability measure on $\mathfrak{P}(\Lambda)$, the powerset of Λ . It describes the set of pinned points. In fact, one easily sees that for any $A \subset \Lambda$ we have

$$A = \{x \in \Lambda \colon \psi_x = 0\} \quad \mathbb{P}_{\Lambda \setminus A}\text{-almost surely}.$$

By (6.1.4), $\mathbb{P}^{\varepsilon}_{\Lambda}$ is a mixture of the Gaussian measures $\mathbb{P}_{\Lambda \setminus A}$ for $A \subset \Lambda$. Our first goal will therefore be to understand the weights of this mixture, i.e. the measure $\zeta^{\varepsilon}_{\Lambda}$. We write $\zeta^{\varepsilon}_{\Lambda}(f)$ for $\sum_{A \subset \Lambda} f(A)\zeta^{\varepsilon}_{\Lambda}(A)$. The first result is that the measures $\zeta^{\varepsilon}_{\Lambda}$ satisfies a correlation inequality.

Theorem 6.1.1. *The measure* $\zeta^{\varepsilon}_{\Lambda}$ *satisfies the FKG inequality, i.e.*

$$\zeta^{\varepsilon}_{\Lambda}(fg) \ge \zeta^{\varepsilon}_{\Lambda}(f)\zeta^{\varepsilon}_{\Lambda}(g)$$

for any pair of increasing functions $f, g: \mathfrak{P}(\Lambda) \to \mathbb{R}$ *.*

This FKG inequality allows us to prove directly that a thermodynamic limit of the $\zeta_{\Lambda}^{\epsilon}$ exists. We can also prove that a thermodynamic limit of the $\mathbb{P}_{\Lambda}^{\epsilon}$ exists. That result relies on the estimates for the Green's function which we state in Theorem 6.1.5 below.

Theorem 6.1.2. *If* $d \ge 4$ *, the thermodynamic limit*

$$\zeta^{\varepsilon} := \lim_{\Lambda \nearrow \mathbb{Z}^{\mathsf{d}}} \zeta^{\varepsilon}_{\Lambda}$$

exists and is translation invariant.

Furthermore, there is a constant ε_d *depending on* d *only such that for any* $\varepsilon < \varepsilon_d$ *the thermody-namic limit*

$$\mathbb{P}^{\varepsilon} := \lim_{\Lambda \nearrow \mathbb{Z}^{\mathsf{d}}} \mathbb{P}^{\varepsilon}_{\Lambda}$$

exists and is translation invariant.

The convergence here is meant as weak convergence of measures on $\mathbb{R}^{\mathbb{Z}^d}$ equipped with the cylinder σ -algebra, i.e. the measures integrated against any bounded local function converge.

It is easy to see that \mathbb{P}^{ε} is an infinite volume Gibbs measure for the interaction (6.1.2) (with appropriate boundary conditions). We write \mathbb{E}^{ε} for the expectation with respect to \mathbb{P}^{ε} .

We will now state a few results on $\zeta_{\Lambda}^{\varepsilon}$ and $\mathbb{P}_{\Lambda}^{\varepsilon}$ that hold uniformly in Λ . Theorem 6.1.2 then implies that they hold for ζ^{ε} and \mathbb{P}^{ε} as well.

We begin with precise estimates on the pinned set. The heuristic is that this set behaves like a Bernoulli point process with density p_d depending on ε . It turns out that this is true in a rather strong sense if $d \ge 5$. In d = 4 this no longer holds, but fortunately we can still compare the probabilities that large sets are free of pinned points, and this is sufficient to continue with our argument. The precise result is the following. For the definition of strong stochastic domination see Definition 6.2.2. We denote by \mathcal{A} a random variable distributed according to $\zeta_{\Delta}^{\varepsilon}$.

Theorem 6.1.3. Let $d \ge 4$. There are constants c_d , C_d , $\varepsilon_{d,*}$ depending on d only with the following properties.

a) If $d \ge 5$ and $p_{d,-} = c_d \varepsilon$, then for any $\Lambda \Subset \mathbb{Z}^d$ and any $\varepsilon < \varepsilon_{d,*}$ the measure $\zeta_{\Lambda}^{\varepsilon}$ strongly dominates the Bernoulli measure on $\mathfrak{P}(\Lambda)$ with parameter $p_{d,-}$. In particular for any $E \subset \Lambda$

$$(1 - p_{\mathsf{d},-})^{|E|} \ge \zeta_{\Lambda}^{\varepsilon}(\mathcal{A} \cap E = \varnothing).$$
(6.1.5)

b) If $d \ge 5$ and $p_{d,+} = C_d \varepsilon$, then for any $\Lambda \Subset \mathbb{Z}^d$ and any $\varepsilon < \varepsilon_{d,*}$ the measure $\zeta_{\Lambda}^{\varepsilon}$ is strongly dominated by the Bernoulli measure on $\mathfrak{P}(\Lambda)$ with parameter $p_{d,+}$. In particular for any $E \subset \Lambda$

$$(1 - p_{\mathsf{d},+})^{|E|} \le \zeta^{\varepsilon}_{\Lambda}(\mathcal{A} \cap E = \varnothing).$$
(6.1.6)

c) If d = 4 and $p_{4,-} = c_4 \frac{\varepsilon}{|\log \varepsilon|^{1/2}}$, then for any $E \subset \Lambda$ and any $\varepsilon < \varepsilon_{4,*}$ we have

$$(1 - p_{4,-})^{|E|} \ge \zeta^{\varepsilon}_{\Lambda}(\mathcal{A} \cap E = \varnothing).$$
(6.1.7)

d) If d = 4, then there is for any $\alpha > 0$ a constant $C_{4,\alpha}$ depending on d and α such that with $p_{4,+,\alpha} = C_{4,\alpha} \frac{\varepsilon}{|\log \varepsilon|^{1/2}}$ for any $E \subset \Lambda$ with $d(E, \mathbb{Z}^d \setminus \Lambda) \ge \varepsilon^{-\alpha}$ and any $\varepsilon < \varepsilon_{4,*}$ we have

$$(1 - p_{4,+,\alpha})^{|E|} \le \zeta_{\Lambda}^{\varepsilon} (\mathcal{A} \cap E = \varnothing).$$
(6.1.8)

All estimates also hold with ζ^{ε} in place of $\zeta^{\varepsilon}_{\Lambda}$.

Let us warn the reader that we use the notation $p_{d,\pm}$ in the opposite way as in [BV01]. Our convention here follows [BCK17].

Note carefully that we do not claim any domination result in case d = 4. In fact, the same argument as in [BV01, Section 2] shows that neither $\zeta_{\Lambda}^{\varepsilon}$ is strongly dominated by the Bernoulli measure on $\mathfrak{P}(\Lambda)$ with parameter $p_{4,+}$, nor that $\zeta_{\Lambda}^{\varepsilon}$ strongly dominates the Bernoulli measure on $\mathfrak{P}(\Lambda)$ with parameter $p_{4,-}$.

In the subcritical dimensions d < 4 the set of pinned points is too correlated for any meaningful comparison with a Bernoulli measure. This is the reason why new techniques would be necessary to study the pinned membrane model in dimensions 2 and 3.

From Theorems 6.1.2 and 6.1.3 one immediately obtains the following corollary, which strengthens Sakagawa's result [Sak12] that the density of pinned points is positive for any $\varepsilon > 0$.

Corollary 6.1.4. *Let* $d \ge 4$ *. Consider the density of pinned points*

$$\rho_{\varepsilon} = \liminf_{\Lambda \nearrow \mathbb{Z}^{\mathsf{d}}} \frac{1}{|\Lambda|} \zeta^{\varepsilon}_{\Lambda}(|\mathcal{A}|) = \liminf_{\Lambda \nearrow \mathbb{Z}^{\mathsf{d}}} \frac{1}{|\Lambda|} \sum_{A \subset \Lambda} |A| \zeta^{\varepsilon}_{\Lambda}(A) \,.$$

For each $\varepsilon > 0$ we have $\rho_{\varepsilon} \ge c_{\mathsf{d}} p_{\mathsf{d},-} > 0$.

It is unclear whether the limit here exists in general. However, using Theorem 6.1.1 and a subadditivity argument one can show that it exists along the sequence $\Lambda_n = [-n, n]^d \cap \mathbb{Z}^d$, say.

Using the knowledge about $\zeta^{\varepsilon}_{\Lambda}$ from Theorem 6.1.3 we can now establish some more precise results on \mathbb{P}^{ε} . For any vector $\theta \in \mathbb{S}^{d-1}$ (where \mathbb{S}^{d-1} is the unit sphere in \mathbb{R}^d) define the mass

$$m_{\varepsilon}(\theta) := -\limsup_{k \to \infty} \frac{1}{k} \log \left| \mathbb{E}^{\varepsilon}(\psi_0 \psi_{\lfloor k \theta \rfloor}) \right|.$$
(6.1.9)

where we set $\log 0 = -\infty$. Note that we take the absolute value of $\mathbb{E}^{\varepsilon}(\psi_0\psi_{\lfloor k\theta \rfloor})$ in this definition. This is because for the membrane model correlations can be negative. In fact, the heuristic in Section 6.1.2 below suggests that $\mathbb{E}^{\varepsilon}(\psi_0\psi_{\lfloor k\theta \rfloor})$ behaves like an underdamped harmonic oscillator. In particular, we expect that the limit in (6.1.9) does not exist. In contrast, for the gradient model the limit in (6.1.9) exists, even without the absolute values, cf. [BV01, Appendix A].

We can show the following results on the variance, covariance and mass. Here d(x, E) denotes the distance from x to the set E.

Theorem 6.1.5. Let $d \ge 4$, and $\Lambda \Subset \mathbb{Z}^d$. There are constants c_d , C_d , $\varepsilon_{d,**}$ depending on d only with the following property.

a) Let $x \in \Lambda$. Then for $\varepsilon < \varepsilon_{d,**}$ we have the following estimates on the variance: if $d \ge 5$, then

$$c_{\mathsf{d}} \le \mathbb{E}^{\varepsilon}_{\Lambda}(\psi_x^2) \le C_{\mathsf{d}} \,, \tag{6.1.10}$$

while if d = 4 and $\alpha > 0$, then

$$\frac{|\log\varepsilon|}{32\pi^2} - C_{4,\alpha} \log|\log\varepsilon| \le \mathbb{E}^{\varepsilon}_{\Lambda}(\psi_x^2) \le \frac{|\log\varepsilon|}{16\pi^2} + C_4 \log|\log\varepsilon|, \qquad (6.1.11)$$

where the lower bound only holds if $d(x, \mathbb{Z}^d \setminus \Lambda) \ge \varepsilon^{-\alpha} + \varepsilon^{-1/4}$. The same estimates hold for $\mathbb{E}^{\varepsilon}_{\Lambda}$ (with the condition on $d(x, \mathbb{Z}^d \setminus \Lambda)$ becoming vacuous).

b) Let $x, y \in \Lambda$. Then for $\varepsilon < \varepsilon_{d,**}$ we have the following estimates on the covariance: if $d \ge 5$, then

$$|\mathbb{E}^{\varepsilon}_{\Lambda}(\psi_{x}\psi_{y})| \leq \frac{C_{\mathsf{d}}}{\varepsilon^{1/2}} \exp\left(-c_{\mathsf{d}}\varepsilon^{1/4}|x-y|\right), \qquad (6.1.12)$$

while if d = 4, then

$$|\mathbb{E}^{\varepsilon}_{\Lambda}(\psi_{x}\psi_{y})| \leq C_{4}\left(\frac{|\log\varepsilon|^{5/4}}{\varepsilon^{1/2}} + \log(1+|x-y|)\right)\exp\left(-\frac{c_{4}\varepsilon^{1/4}|x-y|}{|\log\varepsilon|^{3/8}}\right).$$
 (6.1.13)

The same estimates hold for \mathbb{E}^{ε} *instead of* $\mathbb{E}^{\varepsilon}_{\Lambda}$ *.*

In particular, we have the following estimates on the mass: if $d \ge 5$, then

$$c_{\mathsf{d}}\varepsilon^{1/4} \le m_{\varepsilon}(\theta) \qquad \forall \theta \in \mathbb{S}^{\mathsf{d}-1}$$
, (6.1.14)

while if d = 4, then

$$c_4 \frac{\varepsilon^{1/4}}{|\log \varepsilon|^{3/8}} \le m_{\varepsilon}(\theta) \qquad \forall \theta \in \mathbb{S}^3.$$
(6.1.15)

The estimates in Theorem 6.1.5 are only valid for sufficiently small $\varepsilon > 0$. However, a calculation similar to (6.1.3) reveals that for $\varepsilon' < \varepsilon$ the measure $\mathbb{P}^{(\varepsilon)}_{\Lambda}$ is a mixture of the measures $\mathbb{P}^{(\varepsilon')}_{\Lambda\setminus A}$, and so the theorem also implies that for any $\varepsilon > 0$ the measure $\mathbb{P}^{(\varepsilon)}_{\Lambda}$ has bounded variances and exponentially decaying covariances.

In the next section we describe some heuristics for the exponential decay of the correlations. These heuristics suggest that the exponential rates in (6.1.12) and (6.1.13) are optimal, but the prefactors are not. In fact, we have made no real effort to optimise these prefactors, as this would require further technicalities. Nonetheless, as we believe that the exponential rates in (6.1.12) and (6.1.13) are optimal, the same holds for the rates in (6.1.14) and (6.1.15).

The results of Theorem 6.1.5 are a far-reaching generalization of the results in [BCK16, BCK17]. In [BCK17] it is shown that for $d \ge 5$ the mass is positive for each fixed $\varepsilon > 0$. No explicit lower bound on the mass is given, but if one keeps track of the constants in their argument one can check that their proof gives the estimate

$$c_{\mathsf{d}} \varepsilon^{\frac{(2\mathsf{d}+4)(2\mathsf{d}+1)}{\mathsf{d}}} \leq m_{\varepsilon}(\theta)$$

For d = 4 in [BCK16] stretched-exponential decay of the covariances is shown, and it was unknown whether the decay is actually exponential.

6.1.2 Heuristics: The continuous Bilaplace equation in a perforated domain

Before we describe the proofs of our results in more detail, let us discuss a related problem that provides some heuristics. Namely we consider the continuous Bilaplace equation in a domain perforated by small holes. This is a well-studied problem, and the analogous problem for the Laplacian even more so, cf. [CM97, MK06]¹. If one lets the size of the holes tend to zero while keeping their capacity density constant, the problem converges (in an appropriate sense) to a Bilaplace equation with a mass term on the whole domain. The Green's function of the associated operator decays exponentially, and so it is unsurprising that the same holds true already for the Green's function in the perforated domain.

In our context, this connection gives a hint how to deduce Theorem 6.1.5 if one assumes Theorem 6.1.3. Let us explain this in detail: fix $\varepsilon > 0$, and consider a fixed large, but bounded domain $\Omega \subset \mathbb{R}^d$ with smooth boundary. Let $N \in \mathbb{N}$ be a large parameter to be chosen later. We perforate the domain $N\Omega$ with small holes of radius r > 0, centred at a subset of \mathbb{Z}^d . Theorem 6.1.3 suggests that we choose a fraction $p_{d,-}$ of the points in \mathbb{Z}^d as the centres of these holes. For now we consider the simplest case of equally-spaced holes, i.e. we place them at $(\lambda_{\text{mic}}\mathbb{Z})^d$, where $\lambda_{\text{mic}} \approx (p_{d,-})^{-\frac{1}{d}}$ is an integer. That is, we consider the equation

$$\Delta^{2} u = f \quad \text{in } N\Omega \setminus \bigcup_{x \in (\lambda_{\text{mic}}\mathbb{Z})^{d}} \overline{B_{r}(x)}, \qquad (6.1.16)$$
$$u = 0 \quad \text{else}$$

The Green's function *G* of this problem should predict the behaviour of the covariances in Theorem 6.1.5.

We can rescale (6.1.16) back to a unit domain by letting $\hat{f}(y) = \frac{1}{N^{d-4}}f(Ny)$, $\hat{u}(y) = \frac{1}{N^d}u(Ny)$, so that \hat{u} and \hat{f} solve

$$\Delta^{2} \hat{u} = \hat{f} \quad \text{in } \Omega \setminus \bigcup_{x \in ((\lambda_{\text{mic}}/N)\mathbb{Z})^{d}} \overline{B_{r/N}(x)},$$

$$\hat{u} = 0 \quad \text{else}.$$
 (6.1.17)

In order to apply now results from [CM97], we need to treat $d \ge 5$ and d = 4 separately. We begin with the former case. The collection of balls $\bigcup_{x \in ((\lambda_{mic}/N)\mathbb{Z})^d} B_{r/N}(x)$ has capacity density

$$\mu = C_{\mathsf{d}} \left(\frac{N}{\lambda_{\mathrm{mic}}}\right)^{\mathsf{d}} \left(\frac{r}{N}\right)^{\mathsf{d}-4} = C_{\mathsf{d}} N^4 r^{\mathsf{d}-4} \lambda_{\mathrm{mic}}^{-\mathsf{d}} \approx C_{\mathsf{d}} N^4 r^{\mathsf{d}-4} p_{\mathsf{d},-} = C_{\mathsf{d}} N^4 r^{\mathsf{d}-4} \varepsilon$$

¹Note that the original French and Russian works date back to the 70s and 80s.

We want to consider a limit of dense small holes where μ is constant, and so we choose $N = \varepsilon^{-1/4}$. Then, according to [CM97, Example 2.14], the solution of (6.1.17) in the limit $\varepsilon \to 0$ behaves like the solution of

$$\Delta^2 \hat{u} + \mu \hat{u} = \hat{f} \quad \text{in } \Omega,$$

$$\hat{u} = 0 \quad \text{else}.$$
 (6.1.18)

This is a Bilaplace equation with a mass term. Its Green's function \hat{G} behaves like the Green's function $\hat{G}_{\mathbb{R}^d}$ of the same equation in the full space \mathbb{R}^d (at least when we stay away from the boundary of Ω). The latter Green's function can be computed quite explicitly using separation of variables in spherical coordinates. One finds that $\hat{G}_{\mathbb{R}^d}(\hat{x}, \hat{y}) = F(\mu^{1/4}|x-y|)$, where F(r) is a linear combination of $\operatorname{Re}\left((\zeta_8 r)^{-(d-2)/2}H^{(1)}_{(d-2)/2}(\zeta_8 r)\right)$. Here $H^{(1)}_{\nu}$ is the Hankel function of the first kind, and ζ_8 runs through the primitive eighth roots of unity. A short calculation using the asymptotic expansion for these functions (cf. [AS64, Equation 9.7.2]) and the fact that $\hat{G}_{\mathbb{R}^d}$ needs to decay at infinity reveals that

$$\begin{split} \hat{G}_{\mathbb{R}^{\mathsf{d}}}(\hat{x}, \hat{y}) &= C_{\mathsf{d}} \left(\mu^{1/4} |\hat{x} - \hat{y}| \right)^{-(\mathsf{d}-1)/2} \left(\sin \left(\frac{\mu^{1/4} |\hat{x} - \hat{y}|}{2^{1/2}} - \omega_{\mathsf{d}} \right) + O(\mu^{-1/4} |\hat{x} - \hat{y}|^{-1}) \right) \\ & \times \exp \left(-\frac{\mu^{1/4} |\hat{x} - \hat{y}|}{2^{1/2}} \right) \end{split}$$

where ω_d is a phase shift depending only on d, and we used the standard Landau notation. Neglecting the error term altogether, we thus expect

$$\hat{G}(\hat{x},\hat{y}) \approx C_{\mathsf{d}} \left(\mu^{1/4} |\hat{x} - \hat{y}| \right)^{-(\mathsf{d}-1)/2} \sin\left(\frac{\mu^{1/4} |\hat{x} - \hat{y}|}{2^{1/2}} - \omega_{\mathsf{d}} \right) \exp\left(-\frac{\mu^{1/4} |\hat{x} - \hat{y}|}{2^{1/2}} \right)$$
(6.1.19)

when $|\hat{x} - \hat{y}| \gg \mu^{-1/4}$, and the Green's function of (6.1.17) should behave similarly (at least if Ω is large enough, i.e. diam $\Omega \gg \mu^{-1/4}$). Rescaling back, we thus expect for the Green's function *G* of (6.1.16) that

$$\begin{split} G(x,y) &= \frac{1}{N^{d-4}} \hat{G}\left(\frac{x}{N}, \frac{y}{N}\right) \\ &\approx \frac{1}{N^{d-4}} C_{d}\left(\frac{\mu^{1/4} |x-y|}{2^{1/2}N}\right)^{-(d-1)/2} \sin\left(-\frac{\mu^{1/4} |x-y|}{2^{1/2}N} - \omega_{d}\right) \exp\left(-\frac{\mu^{1/4} |x-y|}{2^{1/2}N}\right) \\ &\approx \frac{C_{d} \varepsilon^{(d-7)/8}}{r^{(d-1)(d-4)/8} |x-y|^{(d-1)/2}} \sin\left(-C_{d} \varepsilon^{1/4} r^{(d-4)/4} |x-y| - \omega_{d}\right) \\ &\qquad \times \exp\left(-C_{d} \varepsilon^{1/4} r^{(d-4)/4} |x-y|\right) \end{split}$$

when $|x - y| \gg N\mu^{-1/4} = \varepsilon^{-1/4}r^{-(d-4)/4}$. Thus, *G* decays exponentially, with polynomial corrections and an oscillatory term that makes *G* change sign. While the polynomial corrections and the oscillatory term are not captured in (6.1.12), the exponential decay rates in both estimates are the same (up to constant factors).

If d = 4, the argument is in principle the same, but we need to use extra care when defining the capacity density. Following [CM97, Example 2.14] we define

$$\mu = C\left(\frac{N}{\lambda_{\rm mic}}\right)^4 \frac{1}{|\log \frac{r}{N}|} \approx CN^4 p_{4,-} \frac{1}{\log N - \log r} \approx C \frac{N^4 \varepsilon}{(\log N - \log r) |\log \varepsilon|^{\frac{1}{2}}}.$$

We want $\varepsilon \to 0$ while μ is constant, and so we choose $N = \frac{|\log \varepsilon|^{3/8}}{\varepsilon^{1/4}} + O_r \left(\frac{|\log \varepsilon|^{1/8} \log |\log \varepsilon|}{\varepsilon^{1/4}}\right) = \frac{|\log \varepsilon|^{3/8} + o_r(|\log \varepsilon|^{3/8})}{\varepsilon^{1/4}}$ accordingly. Then we conclude from [CM97, Example 2.14] that the solutions of (6.1.17) and (6.1.18) are close. The Green's function of (6.1.18) in d = 4 still behaves like (6.1.19), and so, rescaling back, we find again

$$\begin{split} G(x,y) &= \hat{G}\left(\frac{x}{N}, \frac{y}{N}\right) \\ &\approx C\left(\frac{\mu^{1/4}|x-y|}{N}\right)^{-3/2} \sin\left(-\frac{\mu^{1/4}|x-y|}{2^{1/2}N} - \omega_4\right) \exp\left(-\frac{\mu^{1/4}|x-y|}{2^{1/2}N}\right) \\ &\approx \frac{C|\log\varepsilon|^{9/16}}{\varepsilon^{3/8}|x-y|^{3/2}} \sin\left(-\frac{C\varepsilon^{1/4}|x-y|}{|\log\varepsilon|^{3/8} + o_r(|\log\varepsilon|^{3/8}))} - \omega_4\right) \\ &\qquad \times \exp\left(-\frac{C\varepsilon^{1/4}|x-y|}{|\log\varepsilon|^{3/8} + o_r(|\log\varepsilon|^{3/8}))}\right). \end{split}$$

when $|x - y| \gg_r N\mu^{-1/4} \approx \varepsilon^{-1/4} |\log \varepsilon|^{3/8}$. This is again exponential decay with polynomial corrections and an oscillatory term. The exponential decay rate is again the same as in (6.1.13) (up to constants).

In summary, our heuristic predicts the same exponential decay rates as in Theorem 6.1.5. The heuristic we used is rather simplistic, though. One problem is that in the context of the membrane model a single pinned point forces the field to be zero there, but does not pose any restrictions on the gradient of the field. In contrast, in (6.1.16) we force the field and all its derivatives to be zero at the pinned balls. One way to improve the heuristics would thus be to only prescribe that *u* has average zero over each $B_r(x)$ for $x \in (\lambda_{\text{mic}}\mathbb{Z})^d$, instead of it being identically zero there. This is not a serious change, though, as a modification of [CM97, Example 2.14] or an application of the general framework in [MK06] show that the convergence of (6.1.17) to (6.1.18) still holds, albeit with a different constant prefactor in μ .

A more serious problem is that the pinned points are not distributed on a lattice, but following the probability distribution $\zeta_{\Lambda}^{\epsilon}$. If this distribution were, say, a Poisson point process, then the framework from [MK06] would still apply. Our actual $\zeta_{\Lambda}^{\epsilon}$ is possibly quite correlated (at least if d = 4), though, and so it is not clear that the heuristic still applies. On the other hand, we are not actually interested in "quenched" estimates that hold for all realizations of the sets of pinned points, but rather in "annealed" estimates where we average over the randomness of the pinned points. So there is hope to retain the heuristic.

A further question is how to rigorously show that the convergence of the boundary value problem (6.1.17) to the boundary value problem (6.1.18) implies that the Green's function of (6.1.17) already has the predicted behaviour. There are very few results on this in the literature. One exception is [NV06], where this is proved rigorously for the case of the Laplace equation in d = 3. However, that approach relies on the maximum principle, and so one cannot extend it to our situation. Instead, in [HV18] a more robust approach is used: There (in another context) exponential decay of the L^2 -norm of harmonic functions on perforated large annuli is shown, using Widman's hole filler technique [Wid71] in combination with the fact that one has a local Poincaré inequality. A similar argument is also used in [BCK17], and the authors describe that they learned it from Vladimir Maz'ya. The decay rates in [HV18] are not optimal, but a small modification of their argument leads to the optimal decay rate. These arguments are the inspiration for our proof of Theorem 6.1.5 from Theorem 6.1.3. We shall explain this in more detail in the next subsection, where we outline the proofs of our results.

6.1.3 Main ideas of the proofs

This chapter consists of two main parts. We first establish the various results on the pinned set, and then deduce from them the results on the variances and covariances. We will discuss these parts separately. Before doing so, let us remark that there are two natural lengthscales occuring. There are the average distance between pinned points

$$\lambda_{mic} pprox egin{cases} rac{1}{arepsilon^{1/d}} & \mathsf{d} \geq 5 \ rac{|\logarepsilon|^{1/8}}{arepsilon^{1/4}} & \mathsf{d} = 4 \end{cases},$$

and the lengthscale on which the correlations decay

$$\lambda_{mac} \approx \begin{cases} \frac{1}{\epsilon^{1/4}} & \mathsf{d} \geq 5\\ \frac{|\log \epsilon|^{3/8}}{\epsilon^{1/4}} & \mathsf{d} = 4 \end{cases}$$

Note that we have $1 \ll \lambda_{mic} \ll \lambda_{mac}$ as $\varepsilon \to 0$ for any $d \ge 4$.

Estimates on the pinned set

The first novel result of this chapter is the FKG inequality for the pinned set, Theorem 6.1.1. As already mentioned, it follows rather directly from the Gaussian correlation inequality [Roy14], and it is standard to deduce from the FKG inequality the existence of the thermodynamic limit of the set of pinned points, i.e. the first part of Theorem 6.1.2. We give these proofs in Section 6.2.1. Note, however, that our proof of Theorem 6.1.1 is specific to the case of δ -pinning, and we conjecture that the result is not true for other pinning potentials such as a square-well potential. The point is that conditioning a Gaussian vector on being 0 at some coordinates yields another Gaussian vector, but that is no longer true if we condition on some coordinates being small instead. We give a more detailed explanation in Remark 6.2.1.

For the proof of Theorem 6.1.3 in Section 6.2.2 we follow [BV01] rather closely. The domination results in Theorem 6.1.3 a) and b) are actually already in [BCK17]. They follow via a short calculation from the boundedness of the Green's function in $d \ge 5$. Part d) is a little more difficult. It could be proven as in [BV01, Section 3.2], but we give a slightly simpler proof. The idea is that if $x \in E$ is quite far from the pinned points we have already found, the fluctuations of ψ_x are quite big, and so the chance that x is pinned is low.

By far the most difficult part of Theorem 6.1.3 is part c), where we again mostly follow [BV01]. There we want to control the probability that $E \subset \Lambda$ is free of pinned points from above. To do so, we need to find for any configuration of pinned points that avoids *E* many others that intersect *E*. This is done using a two-scale argument. We first consider the case that *E* is a union of boxes of sidelength $C\lambda_{mic}$, and prove (6.1.7) in this case by carefully tracking how a pinned point in one of these boxes makes it likely that there are pinned points in the neighbouring boxes. Next, we pass to the larger lengthscale $C\lambda_{mac}$ and deduce from the first step that for an arbitrary *E*, most points of *E* are at a distance $\leq C\lambda_{mac}$ from a pinned point. Finally we use this knowledge together with an argument similar to the first step to construct many configurations of pinned points that intersect *E*.

The main difference to [BV01] is that one cannot use random walk estimates to see how pinning at some $x \in \Lambda$ influences the variance at $y \neq x$. Instead we use an explicit variance estimate (Lemma 6.2.4) that follows from the monotonicity of the variance in the set of pinned points. We also streamline the argument from [BV01] at some points and correct a minor mistake there.

Asymptotics for the variances and covariances

The remainder of the chapter is then concerned with proving Theorem 6.1.5 and the second part of Theorem 6.1.2. In [BV01] the random walk representation of the Green's function of the Laplacian is used for that purpose. In our case there is no such representation, so we need a completely new argument.

It turns out that the estimates for the variance follow quite easily from Theorem 6.1.3 and the variance estimate in Lemma 6.2.4. We give details in Sections 6.5.1 and 6.5.2.

The estimates on the covariance in Theorem 6.1.3 are much more difficult. The general strategy is the same as in [BCK17]. That is, we show that the L^2 -norm of the second derivative of the "quenched" Green's function decays exponentially on large annuli. These annuli have to be chosen adapted to the set of pinned points, and so we do not get an estimate valid for all realizations of A. But our estimates hold up to an exponentially small probability, so that we control $G_{\Lambda\setminus A}$ for all but exponentially few A. For these we can use a rather crude estimate. Finally, we can average these quenched estimates for the Green's function over A to deduce "annealed" bounds for the covariance.

The existence of the thermodynamic limit of the field in Theorem 6.1.2 follows then from the existence of the thermodynamic limit of the set of pinned points and the quenched decay estimates on the Green's function. The somewhat technical proof is given in Section 6.5.3.

For the remainder of this section, let us describe in more detail how we prove the quenched estimates on the covariance. Our main technical result used for that purpose is, roughly speaking, the following (see Theorem 6.4.1 for the precise statement): There is a constant \hat{N}_d such that if $k \in \mathbb{N}$ and ε is sufficiently small there is an event $\Omega_{U,k}$ with $\zeta_{\Lambda}^{\varepsilon}(\Omega_{U,k}) \ge 1 - \frac{C_U}{2^k}$ such that if $A \in \Omega_{U,k}$, and if $u: \mathbb{Z}^d \to \mathbb{R}$ is a function such that u = 0 on $A \setminus U$ and $u\Delta_1^2 u = 0$ on $\mathbb{Z}^d \setminus U$, we have the estimate

$$\left\|\nabla_{1}^{2}u\right\|_{L^{2}(\mathbb{Z}^{\mathsf{d}}\setminus(U+Q_{2k\hat{N}_{\mathsf{d}}\lambda_{\max}}(0)))}^{2} \leq \frac{1}{2^{k}}\left\|\nabla_{1}^{2}u\right\|_{L^{2}((U+Q_{2k\hat{N}_{\mathsf{d}}\lambda_{\max}}(0)\setminus U)}^{2}.$$
(6.1.20)

Here $Q_{2k\hat{N}_d\lambda_{mac}}(0)$ denotes a cube of halfdiameter $2k\hat{N}_d\lambda_{mac}$ centred at 0. This is an exterior decay estimate for biharmonic functions that holds up to exponentially small probability (and we state and prove in Theorem 6.4.1 also the analogous interior decay estimate). Applying (6.1.20) with $u = G_{\Lambda \setminus A}(\cdot, y)$ it is a bit tedious but not difficult to deduce the aforementioned quenched estimates on the Green's function, and we do so in Sections 6.5.1 and 6.5.2.

Let us describe how to prove (6.1.20). We first outline the basic strategy that was used in [BCK17] and (in another context) in [HV18], and then describe our novel ideas. For convenience we pretend in the following that u is a continuous function. Adapting the argument to the discrete setting will be somewhat technical but not hard.

We try to iterate a Widman hole filler argument [Wid71] (see, e.g., [GM12, Section 4.4] for a modern presentation). That is, given $U \subset \mathbb{R}^d$, we want to find $U' \supset U$, so that the L^2 -norm of $\nabla^2 u$ on $\mathbb{R}^d \setminus U$ is controlled by a constant less than 1 times the L^2 -norm of $\nabla^2 u$ on $U' \setminus U$. We also want dist $(U, \mathbb{R}^d \setminus U') \leq C\lambda_{\text{mac}}$. Once we have such an estimate, we can iterate it to deduce exponential decay at rate $\frac{1}{C\lambda_{\text{mac}}}$, at least on the L^2 -level.

So suppose that $U \subset U'$ are open sets and η is a smooth cut-off function such that

$$\{\Delta^2 u \neq 0\} \subset U \subset \{\eta = 0\} \subset \{\eta \neq 1\} \subset U'.$$

Then we have

$$0 = (\Delta^2 u, \eta u) = (\nabla^2 u, \nabla^2 (\eta u)) = \int \eta |\nabla^2 u|^2 + 2 \int \nabla^2 u : \nabla u \otimes \nabla \eta + \int u \nabla^2 u : \nabla^2 \eta$$
(6.1.21)

and one can rewrite this using the Cauchy-Schwarz inequality as

$$\begin{split} &\int_{\mathbb{R}^{d}\setminus U'} |\nabla^{2}u|^{2} \\ &\leq \int \eta |\nabla^{2}u|^{2} \\ &= -2\int \nabla^{2}u : \nabla u \otimes \nabla \eta - \int u \nabla^{2}u : \nabla^{2}\eta \\ &\leq \frac{1}{5}\int_{U'\setminus U} |\nabla^{2}u|^{2} + 5\int_{U'\setminus U} |\nabla u|^{2}|\nabla \eta|^{2} + \frac{1}{5}\int_{U'\setminus U} |\nabla^{2}u|^{2} + \frac{5}{4}\int_{U'\setminus U} |u|^{2}|\nabla^{2}\eta|^{2} \,. \end{split}$$
(6.1.22)

Now if we could choose η in such a way that the second and fourth summand here are both bounded by $\frac{1}{5} \int_{U'\setminus U} |\nabla^2 u|^2$ we would obtain the desired decay estimate. In fact, this is what was done in [BCK17]. However, in order to bound both the second and fourth summand, one needs to impose strong pointwise conditions on $\nabla \eta$ and $\nabla^2 \eta$, and, in particular, both need to be near zero on mesoscopic holes in the pinned set. These conditions do not allow growth of η at the optimal rate, and so using this argument one cannot obtain the optimal estimate for the decay rate (but is it comparably easy to construct an η that satisfies these conditions and grows at a non-optimal rate, cf. [BCK17]).

To solve this problem we first rewrite the right hand side of (6.1.21) so that there are no longer any terms containing $\nabla \eta$. An integration by parts shows that

$$\int \nabla^2 u : \nabla u \otimes \nabla \eta = -\int \nabla u \cdot (\nabla \cdot (\nabla u \otimes \nabla \eta)) = -\int \nabla^2 u : \nabla u \otimes \nabla \eta - \int |\nabla u|^2 \Delta \eta$$

and hence

$$\int \nabla^2 u : \nabla u \otimes \nabla \eta = -\frac{1}{2} \int |\nabla u|^2 \Delta \eta \,.$$

Plugging this into (6.1.21) we see that

$$\int \eta |\nabla^2 u|^2 = \frac{1}{2} \int |\nabla u|^2 \Delta \eta - \int u \nabla^2 u : \nabla^2 \eta .$$
(6.1.23)

Using the assumptions on η and the Cauchy-Schwarz inequality we can now estimate

$$\int_{\mathbb{R}^{d}\setminus U'} |\nabla^{2}u|^{2} \leq \int \eta |\nabla^{2}u|^{2}$$

$$= \frac{1}{2} \int |\nabla u|^{2} \Delta \eta - \int_{\mathbb{R}^{d}\setminus U} u \nabla^{2}u : \nabla^{2}\eta$$

$$\leq \frac{1}{2} \int |\nabla u|^{2} \Delta \eta + \int |u|^{2} |\nabla^{2}\eta|^{2} + \frac{1}{4} \int_{U'\setminus U} |\nabla^{2}u|^{2}.$$
(6.1.24)

If we can now arrange things in such a way that the first two summands here are each bounded by $\frac{1}{4} \int_{U'\setminus U} |\nabla^2 u|^2$, we see that

$$\int_{\mathbb{R}^d \setminus U'} |\nabla^2 u|^2 \le \frac{3}{4} \int_{U' \setminus U} |\nabla^2 u|^2 \tag{6.1.25}$$

and now we can try to iterate this estimate to obtain exponential decay of the L^2 -norm of $\nabla^2 u$. Note that unlike (6.1.22) we now only need to impose conditions on $\nabla^2 \eta$.

As it turns out, the first summand in (6.1.24) can be controlled by the second and third summand using an interpolation inequality on lengthscale λ_{mic} that we discuss in 6.3.2.

The remaining task is thus to choose η in such a way that it grows fast enough, but we nonetheless can bound the term $\int |u|^2 |\nabla^2 \eta|^2$. For that purpose we need some sort of local Poincaré inequality on scale λ_{mic} . Of course, such an estimate can only hold if there are enough pinned points close to the point of interest. In [BCK17, Lemma 4.1] this was done provided there is a nearby cube of 3^d points, which are all pinned. On that small cube we then have $u = \nabla u = 0$, and some version of the Hardy-Rellich inequality forces u to be small near that cube as well. However, this is not optimal, as cubes of 3^d points that are all pinned are very rare. We show that it is sufficient if there are d + 1 pinned points somewhere nearby that are well-spread out. The number d + 1 arises from the fact that we need to eliminate nonzero affine functions on \mathbb{R}^d . Thus, in some sense we use a multipolar Hardy-Rellich inequality instead of a unipolar one. For multipolar Hardy Rellich inequalities cf. e.g. [CZ13]; we could not find a detailed discussion of multipolar Hardy-Rellich inequalities in the literature.

The local Poincaré inequality result is, roughly speaking, the following (see Theorem 6.3.1 for the precise result): Let $A \subset \Lambda$, and $V \subset \Lambda$ be an arbitrary subset. Let $R \in \mathbb{N}$, $R \ge 2$ be a parameter. Then

$$\|u\mathbb{1}_{\in X_{R}}\|_{L^{2}(V)}^{2} \leq C_{\mathsf{d}}R^{\mathsf{d}}(1+\mathbb{1}_{\mathsf{d}=4}\log R) \|\nabla^{2}u\|_{L^{2}(V+Q_{R}(0))}^{2}$$

where X_R is the set of those points that have d + 1 well-separated pinned points at distance $\leq R\lambda_{mic}$ around them, and we write $\mathbb{1}_{\cdot \in X_R}$ for the indicator function of that set.

This result makes it clear what we need to require of η . Namely we want $|\nabla^2 \eta| \leq C \mathbb{1}_{\cdot \in X_R}$ for some *R*. If we have this relation, then our multipolar Hardy-Rellich inequality allows us to control the second term on the right hand side in (6.1.24), and we can close the argument for the exponential decay estimate.

It thus remains to choose *R* and construct η such that $|\nabla^2 \eta| \leq C \mathbb{1}_{\in X_R}$ in such a way that η grows fast enough, and the construction should work up to an exponentially small probability. This is the content of Sections 6.4.1 and 6.4.3. This is the technical heart of the present chapter, and the arguments are novel. We can think of X_R as the good set, and its complement as the bad set, and we need to construct η such that it is locally affine on the bad set, but still grows quadratically. To execute this construction, we start with one η_* that grows quadratically, and then try to modify it so that it becomes affine on the bad set. For such modifications it is necessary that the components of the bad set are well-separated from each other. In general this will not be the case, but we make a multiscale composition of the bad set into parts that live on lengthscale ℓ_j and are well-separated on that lengthscale, and then we change η_* to be affine on those parts separately. The correct choice of the lengthscales ℓ_j turns out to be the rather strange looking $\ell_j = CM^{j^3} \lambda_{mic}$, where *M* is some large integer. The construction can be carried out provided the multiscale decomposition vanishes beyond some large lengthscale. Using the results from Theorem 6.1.3 we show that this is the case up to a probability that can be made arbitrarily small.

Unfortunately "arbitrarily small" is not quite good enough, as that means that there are still exceptional pairs (U, U') on which we cannot deduce (6.1.25). But such exceptional pairs are rare, and when we iterate (6.1.25) to conclude (6.1.20) it is sufficient if we can apply

(6.1.25) on at least half of the possible (U, U'), which is possible up to an exponentially small probability.

This completes the construction of η . Once we have η at our disposal, we can complete the proof of (6.1.20). We refer to Section 6.4 for a more detailed exposition of the argument.

6.1.4 Notation and preliminaries

Recall the conventions from Section 1.5.

We will freely use various summation by part identities as in Chapter 2.

For r > 0 and $x \in \mathbb{Z}^4$ we let $Q_r(x) = x + [-r, r]^4 \cap \mathbb{Z}^4$ be the cube of diameter 2r around x. Note that this deviates from the definitions in Section 1.5. We will frequently use the Minkowski-sum of sets E, E' defined by $E + E' = \{e + e' : e \in E, e' \in E'\}$. In particular, $E + Q_r(0)$ is the set of all points at distance $\leq r$ from E.

For measures μ on $\mathfrak{P}(\Lambda)$ we write $\mu(f)$ for $\int f d\mu = \sum_{A \subset \Lambda} f(A)\mu(A)$. We denote a sample from $\zeta_{\Lambda}^{\varepsilon}$ by \mathcal{A} . We define $\tilde{\mathcal{A}} = A \cup (\mathbb{Z}^{\mathsf{d}} \setminus \Lambda)$ for $A \subset \Lambda$ and analogously $\tilde{\mathcal{A}} = \mathcal{A} \cup (\mathbb{Z}^{\mathsf{d}} \setminus \Lambda)$. We let $G_{\Lambda \setminus A}$ be the discrete Green's function of Δ_1^2 on $\Lambda \setminus A$, i.e. $G_{\Lambda \setminus A}(x, y) := \mathbb{P}_{\Lambda \setminus A}(\psi_x \psi_y)$.

We use these distances *d* with respect to $|\cdot|_1$ and $|\cdot|_{\infty}$ instead of $|\cdot|$, and in that case we write d_1 or d_{∞} instead of *d*.

We will use two different length scales λ_{mic} and λ_{mac} . The former describes the typical distance between two pinned points, which according to Theorem 6.1.3 is of the order $\frac{1}{\epsilon^{1/d}}$ if $d \ge 5$ and $\frac{|\log \epsilon|^{1/8}}{\epsilon^{1/4}}$ if d = 4. We hence define

$$\lambda_{\rm mic} = \begin{cases} \frac{1}{\varepsilon^{1/d}} + \alpha_{\rm mic,5}(\varepsilon) & \mathsf{d} \ge 5\\ \frac{|\log \varepsilon|^{1/8}}{\varepsilon^{1/4}} + \alpha_{\rm mic,4}(\varepsilon) & \mathsf{d} = 4 \end{cases}.$$

Here $\alpha_{\text{mic,d}}(\varepsilon) \in [0,2)$ is chosen in such a way that λ_{mic} is an odd integer.

The latter corresponds to the length scale on which correlations decay, and so, in line with Theorem 6.1.5 we set

$$\lambda_{\text{mac}} = \begin{cases} \frac{1}{\varepsilon^{1/4}} + \alpha_{\text{mac},5}(\varepsilon) & \mathsf{d} \ge 5\\ \frac{|\log \varepsilon|^{3/8}}{\varepsilon^{1/4}} + \alpha_{\text{mac},4}(\varepsilon) & \mathsf{d} = 4 \end{cases}$$

where $\alpha_{\text{mac,d}}(\varepsilon) \in [0, 2\lambda_{\text{mic}})$ is chosen such that λ_{mac} is an odd multiple of λ_{mic} . Note that for any d ≥ 4 we have $1 \ll \lambda_{\text{mic}} \ll \lambda_{\text{mac}}$ as $\varepsilon \to 0$.

Given an odd integer l > 0, we consider the set of *l*-boxes

$$\mathcal{Q}_{l} = \left\{ Q_{l/2}(x) \colon x \in (l\mathbb{Z})^{\mathsf{d}} \right\} = \left\{ x + \left[-\frac{l}{2}, \frac{l}{2} \right]^{\mathsf{d}} \cap \mathbb{Z}^{\mathsf{d}} \colon x \in (l\mathbb{Z})^{\mathsf{d}} \right\}$$

and the set

$$\mathcal{P}_l = \left\{ \bigcup_{Q \in I} Q \colon I \subset \mathcal{Q}_l \right\}$$

of *l*-polymers. We identify each box with the polymer consisting just of that box. We call polymers connected if they are connected as subgraphs of \mathbb{Z}^d with nearest-neighbour edges. We say that two polymers touch if they are disjoint but their union is connected.

The boxes in Q_l form a partition of \mathbb{Z}^d . Later on we will also need boxes with some overlap. Thus if l > 0 is an odd multiple of 3 we define

$$\mathcal{Q}_l^{\#} = \left\{ Q_{l/2}(x) \colon x \in \left(\frac{l}{3}\mathbb{Z}\right)^{\mathsf{d}} \right\} = \left\{ x + \left[-\frac{l}{2}, \frac{l}{2}\right]^{\mathsf{d}} \cap \mathbb{Z}^{\mathsf{d}} \colon x \in \left(\frac{l}{3}\mathbb{Z}\right)^{\mathsf{d}} \right\}.$$

Then every point of \mathbb{Z}^d is contained in precisely 3^d boxes in $\mathcal{Q}_1^{\#}$.

For some statement *s* we let $\mathbb{1}_s$ be the indicator function of *s*, that is $\mathbb{1}_s = 1$ if *s* is true, and $\mathbb{1}_s = 0$ else.

6.2 Structure of the pinned set

In this section we prove our results on the distribution of the pinned set, i.e. Theorem 6.1.1, the first part of Theorem 6.1.2, as well as Theorem 6.1.3.

6.2.1 Correlation inequalities

We want to establish the FKG inequality for the set of pinned points in Theorem 6.1.1. We begin with a useful calculation. Let $A \subset A' \subset \Lambda \Subset \mathbb{Z}^d$. Then, using that $\delta_0(d\psi)$ is a weak limit of the measures $\frac{1}{2t} \mathbb{1}_{\psi \in (-t,t)} d\psi$ as $t \to 0$, we have

$$\frac{Z_{\Lambda\setminus A'}}{Z_{\Lambda\setminus A}} = \frac{1}{Z_{\Lambda\setminus A}} \int \exp\left(-\frac{1}{2} \sum_{x\in\mathbb{Z}^{d}} |\Delta_{1}\psi_{x}|^{2}\right) \prod_{x\in\Lambda\setminus A'} d\psi_{x} \prod_{x\in\mathbb{Z}^{d}\setminus(\Lambda\setminus A')} \delta_{0}(d\psi_{x})$$

$$= \frac{1}{Z_{\Lambda\setminus A}} \lim_{t\to 0} \int \exp\left(-\frac{1}{2} \sum_{x\in\mathbb{Z}^{d}} |\Delta_{1}\psi_{v}|^{2}\right)$$

$$\times \prod_{x\in\Lambda\setminus A'} d\psi_{x} \prod_{x\in A'\setminus A} \frac{1}{2t} \mathbb{1}_{\psi_{x}\in(-t,t)} d\psi_{x} \prod_{x\in\mathbb{Z}^{d}\setminus(\Lambda\setminus A)} \delta_{0}(d\psi_{x})$$

$$= \lim_{t\to 0} \frac{1}{(2t)^{|A'\setminus A|}} \mathbb{P}_{\Lambda\setminus A}(|\psi_{x}| < t \ \forall x \in A'\setminus A).$$
(6.2.1)

We can also interpret the right hand side as the density at zero of the Gaussian vector $(\psi_x)_{x \in A' \setminus A}$ under $\mathbb{P}_{\Lambda \setminus A}$ (this observation was essentially already made in [Vel06, p. 143]). If $A' \setminus A = \{x\}$ is a singleton, the density of ψ_x at 0 is equal to $\frac{1}{\sqrt{2\pi}}$ times the inverse of its standard deviation. We thus obtain the formula

$$\frac{Z_{\Lambda \setminus (A \cup \{x\})}}{Z_{\Lambda \setminus A}} = \frac{1}{\sqrt{2\pi G_{\Lambda \setminus A}(x, x)}}.$$
(6.2.2)

Proof of Theorem 6.1.1. We will prove the FKG lattice condition

$$\zeta^{\varepsilon}_{\Lambda}(A \cup A')\zeta^{\varepsilon}_{\Lambda}(A \cap A') \ge \zeta^{\varepsilon}_{\Lambda}(A)\zeta^{\varepsilon}_{\Lambda}(A') \quad \forall A, A' \subset \Lambda.$$
(6.2.3)

It is well-known that this is a sufficient condition for the validity of the FKG inequality.

Now (6.2.3) is an easy consequence of the Gaussian correlation inequality [Roy14, LM17]. Indeed, note that by definition of $\zeta_{\Lambda}^{\epsilon}$ the estimate (6.2.3) is equivalent to

$$Z_{\Lambda \setminus (A \cup A')} Z_{\Lambda \setminus (A \cap A')} \ge Z_{\Lambda \setminus A} Z_{\Lambda \setminus A'}$$

(here we used $|A \cup A'| + |A \cap A'| = |A| + |A'|$). Dividing both sides by $(Z_{\Lambda \setminus (A \cap A')})^2$ and using (6.2.1) we only have to verify

$$\begin{split} &\lim_{t \to 0} \mathbb{P}_{\Lambda \setminus (A \cap A')}(|\psi_x| < t \; \forall x \in (A' \setminus A) \cup (A \setminus A')) \\ &\geq \lim_{t \to 0} \mathbb{P}_{\Lambda \setminus (A \cap A')}(|\psi_x| < t \; \forall x \in A' \setminus A) \mathbb{P}_{\Lambda \setminus (A \cap A')}(|\psi_x| < t \; \forall x \in A \setminus A') \end{split}$$

The sets $\{\psi : |\psi_x| < t \,\forall x \in A' \setminus A\}$ and $\{\psi : |\psi_x| < t \,\forall x \in A \setminus A'\}$ are convex and symmetric around the origin, and the measure $\mathbb{P}_{\Lambda \setminus (A \cap A')}$ is Gaussian. Thus, the claim follows from the Gaussian correlation inequality, applied for each t > 0.

Remark 6.2.1. In [BV01] it is shown that the set of pinned points for the gradient model satisfies a FKG inequality not only in the case of δ -pinning, but also in the case of pinning by a square-well potential $b\mathbb{1}_{|\cdot| < a}$. The proof in [BV01] uses the Ginibre (or GKS) inequality (as described in detail e.g. in [DMRR92, Appendix A]), and thus requires that the measure describing the field is an even fermionic measure. This is certainly not the case for the membrane model, and so that proof cannot be applied in our setting.

Our proof of Theorem 6.1.1 only used that \mathbb{P}_{Λ} is a non-degenerate Gaussian measure. However, this proof would not work if we considered pinning by a square-well potential $b\mathbb{1}_{|\psi| < a}$ instead of δ -pinning. Namely, in this case we would need to consider $\mathbb{P}_{\Lambda}(\cdot | |\psi_x| < a \forall x \in A \cap A')$ instead of $\mathbb{P}_{\Lambda \setminus (A \cap A')}$, and the former measure is not Gaussian, so that we cannot apply the Gaussian correlation inequality.

This is not a shortcoming of our proof. Namely, we conjecture that the analogue of (6.2.3) in the case of pinning by a square-well potential is false. We do not have a counterexample for the case of the membrane model. However, we can give an example of a Gaussian measure where the set of pinned points with respect to a square-well potential does not satisfy (6.2.3).

For this example, let X_1, X_2 be independent standard Gaussians, and N > 0 a large parameter, and define

$$\begin{pmatrix} Y_1 \\ Y_2 \\ Y_3 \\ Y_4 \\ Y_5 \\ Y_6 \end{pmatrix} = \begin{pmatrix} 1 & 0 \\ 0 & 1 \\ N & 0 \\ 0 & N \\ 1 & 1 \\ 1 & 1 \end{pmatrix} \begin{pmatrix} X_1 \\ X_2 \end{pmatrix}$$

Then *Y* is a multivariate Gaussian vector. It is degenerate, but one can fix this later by adding some small Gaussian noise to it, so we will ignore that point. Let also $A = \{1, 3, 5, 6\}$ and $A' = \{2, 4, 5, 6\}$.

In this setting (6.2.3) would correspond to

$$\mathbb{P}\left(|Y_i| \le t \ \forall t \in A \cup A'\right) \mathbb{P}\left(|Y_i| \le t \ \forall t \in A \cap A'\right) \ge \mathbb{P}\left(|Y_i| \le t \ \forall t \in A\right) \mathbb{P}\left(|Y_i| \le t \ \forall t \in A'\right)$$
(6.2.4)

for any t > 0. The probabilities here are equal to the Gaussian measure of certain sets in \mathbb{R}^2 (cf. Figure 6.1). As $t \to 0$, we can approximate this Gaussian measure by the Lebesgue measure, and thereby compute that

$$\begin{split} &\lim_{t\to 0} \frac{2\pi}{t^2} \mathbb{P}\left(|Y_i| \le t \; \forall t \in A \cup A'\right) = \frac{4}{N^2},\\ &\lim_{t\to 0} \frac{2\pi}{t^2} \mathbb{P}\left(|Y_i| \le t \; \forall t \in A \cap A'\right) = 2, \end{split}$$

$$\begin{split} &\lim_{t\to 0} \frac{2\pi}{t^2} \mathbb{P}\left(|Y_i| \leq t \; \forall t \in A\right) = \frac{4}{N} - \frac{2}{N^2},\\ &\lim_{t\to 0} \frac{2\pi}{t^2} \mathbb{P}\left(|Y_i| \leq t \; \forall t \in A'\right) = \frac{4}{N} - \frac{2}{N^2}. \end{split}$$

In particular, for *N* large and *t* small (6.2.4) is wrong by a factor arbitrarily close to 2.



Figure 6.1: The sets associated to the probabilities in (6.2.4). The product of the areas of the large and small square is about half the product of the two areas of the thin rectangles.

Theorem 6.1.1 directly implies the existence of a thermodynamic limit of the $\zeta_{\Lambda}^{\varepsilon}$:

Proof of Theorem 6.1.2, first part. It suffices to check that the limit $\lim_{\Lambda \nearrow \mathbb{Z}^d} \zeta_{\Lambda}^{\varepsilon}(f)$ exists for each bounded $f : \mathfrak{P}(\mathbb{Z}^d) \to \mathbb{R}$ that is a local function (i.e. depends only on finitely many points). Each such f is a linear combination of increasing functions, and so it actually suffices to check that $\lim_{\Lambda \nearrow \mathbb{Z}^d} \zeta_{\Lambda}^{\varepsilon}(f)$ exists for each local increasing f.

For that purpose note that Theorem 6.1.1 implies that for any $\Lambda \subset \Lambda' \Subset \mathbb{Z}^d$ large enough so that f only depends on the points in Λ , we have $\zeta_{\Lambda}^{\varepsilon}(f) \ge \zeta_{\Lambda'}^{\varepsilon}(f)$. Thus, $\lim_{R\to\infty} \zeta_{Q_R(0)}^{\varepsilon}(f)$ exists as a limit of a bounded decreasing sequence. Furthermore, for any $\Lambda \Subset \mathbb{Z}^d$ with $Q_r(0) \subset \Lambda \subset Q_R(0)$ we have

$$\zeta_{Q_r(0)}^{\varepsilon}(f) \ge \zeta_{\Lambda}^{\varepsilon}(f) \ge \zeta_{Q_R(0)}^{\varepsilon}(f) \ge \lim_{R \to \infty} \zeta_{Q_R(0)}^{\varepsilon}(f) \,.$$

Since $\Lambda \nearrow \mathbb{Z}^d$ allows us to take $r \to \infty$, we see from this that indeed $\lim_{\Lambda \nearrow \mathbb{Z}^d} \zeta_{\Lambda}^{\varepsilon}(f)$ exists and is equal to $\lim_{R\to\infty} \zeta_{O_R(0)}^{\varepsilon}(f)$.

Thus, the unique weak limit ζ^{ε} exists. Its translation invariance follows from the fact that

$$\zeta^{\varepsilon}(f) = \lim_{\Lambda \nearrow \mathbb{Z}^{d}} \zeta^{\varepsilon}_{\Lambda}(f) = \lim_{\Lambda \nearrow \mathbb{Z}^{d}} \zeta^{\varepsilon}_{\Lambda+x}(f(\cdot - x)) = \zeta^{\varepsilon}(f(\cdot - x))$$

for any $x \in \mathbb{Z}^d$.

6.2.2 Estimates on the pinned set

We will prove the various domination results of Theorem 6.1.3. We first show some estimates on the variance of the membrane model. We begin with the straightforward proofs of part a) and b), then show part d), and finally part c). See Section 6.1.3 for an outline of the proofs.

Let us first give the precise definition of (strong) domination, as in [BV01].

Definition 6.2.2. Let Λ be a finite set, and let ν , ν' be two probability measures on $\mathfrak{P}(\Lambda)$. We say that ν dominates ν' if we have

$$\nu(f) \ge \nu'(f)$$

for all increasing functions $f: \mathfrak{P}(\Lambda) \to \mathbb{R}$. We say that ν strongly dominates ν' , if for all $x \in \Lambda$ and for all $E \subset \Lambda \setminus \{x\}$ we have

$$\nu(A \ni x \mid A \setminus \{x\} = E) \ge \nu'(A \ni x \mid A \setminus \{x\} = E).$$

It is easy to see that strong stochastic domination implies stochastic domination, and the latter implies

$$u(A \cap E = \varnothing) \le \nu'(A \cap E = \varnothing) \quad \forall E \subset \Lambda.$$

Our proof of Theorem 6.1.3 is based on the proof of the corresponding result for the gradient model in [BV01]. We begin with some useful estimates on the variance of the membrane model.

The first one states the fact that the variance is non-increasing in the size of the pinned set.

Lemma 6.2.3. Let $A \subset A' \subset \Lambda \Subset \mathbb{Z}^d$, and let $x \in \Lambda$. Then $G_{\Lambda \setminus A'}(x, x) \leq G_{\Lambda \setminus A}(x, x)$.

Proof. This follows easily from the Markov property of the field. See e.g. [BCK17, Corollary 3.2].

The preceding lemma allows us to conclude bounds on the variances.

Lemma 6.2.4. Let $\emptyset \neq A \subset \Lambda \Subset \mathbb{Z}^d$, and let $x \in \Lambda$. If $d \ge 5$, we have

$$c_{\mathsf{d}} \le G_{\Lambda \setminus A}(x, x) \le C_{\mathsf{d}} \,. \tag{6.2.5}$$

If d = 4, we have

$$\frac{1}{8\pi^2}\log(1+d(x,A)) - C \le G_{\Lambda\setminus A}(x,x) \le \frac{1}{4\pi^2}\log(1+d(x,A)) + C.$$
(6.2.6)

Proof. We begin with the upper bound in (6.2.6). Let $a \in A$ be such that |x - a| = d(x, A). Let $N \in \mathbb{N}$. For large enough N we have $\Lambda \subset Q_N(x)$. Now Lemma 6.2.3 implies that

$$G_{\Lambda \setminus A}(x,x) \le G_{Q_N(x) \setminus \{a\}}(x,x) \tag{6.2.7}$$

for all *N* large enough. The right hand side can be computed quite explicitly: We have

$$G_{Q_N(x)\setminus\{a\}}(x,x) = G_{Q_N(x)}(x,x) - \frac{G_{Q_N(x)}(a,x)^2}{G_{Q_N(x)}(a,a)}$$

and by Theorem 4.1.4 we have

$$\left|G_{Q_N(x)}(y,y') - \frac{1}{8\pi^2}\log\left(\frac{N}{1+|y-y'|}\right)\right| \le C$$

for all $y, y' \in Q_N(x)$ with $d(y, \partial \Lambda_N) \ge cN$, $d(y', \partial \Lambda_N) \ge cN$.

Using this in (6.2.7) we find for *N* large enough

$$G_{Q_N(x)\setminus\{a\}}(x,x) \leq \frac{1}{8\pi^2} \left(\log N + C - \frac{\left(\log\left(\frac{N}{1+|x-a|}\right) - C\right)^2}{\log N - C} \right)$$

$$\leq \frac{1}{4\pi^2}\log(1+|x-a|) + C$$

and this implies the upper bound in (6.2.6). The lower bound is similar: This time we compare $G_{\Lambda\setminus A}(x, x)$ with $G_{Q_{d(x,A)-1}(x)}(x, x)$.

Finally, the proof of (6.2.5) is similar, using that $G_{Q_N(x)}(x, x)$ is bounded above and below if d \geq 5.

Proof of Theorem 6.1.3 a) and b). The two results are already proven in [BCK17, Lemma 3.4]. Nonetheless, we repeat the short argument: For $x \in \Lambda$, $E \subset \Lambda \setminus \{x\}$ we have

$$\begin{aligned} \zeta_{\Lambda}^{\varepsilon}(\mathcal{A} \ni x \mid \mathcal{A} \setminus \{x\} = E) &= \frac{\zeta_{\Lambda}^{\varepsilon}(E \cup \{x\})}{\zeta_{\Lambda}^{\varepsilon}(E \subset \mathcal{A} \subset E \cup \{x\})} \\ &= \frac{\zeta_{\Lambda}^{\varepsilon}(E \cup \{x\})}{\zeta_{\Lambda}^{\varepsilon}(E) + \zeta_{\Lambda}^{\varepsilon}(E \cup \{x\})} \\ &= \left(1 + \frac{Z_{\Lambda \setminus E}}{\varepsilon Z_{\Lambda \setminus (E \cup \{x\})}}\right)^{-1} \\ &= \left(1 + \frac{\sqrt{2\pi G_{\Lambda \setminus E}(x,x)}}{\varepsilon}\right)^{-1} \end{aligned}$$
(6.2.8)

where the last step follows from (6.2.2). Now in dimension d \geq 5 we have $c_d \leq G_{\Lambda \setminus E}(x, x) \leq C_d$ by Lemma 6.2.4, and this implies

$$c_{\mathsf{d}}\varepsilon \leq \zeta^{\varepsilon}_{\Lambda}(\mathcal{A} \ni x \mid \mathcal{A} \setminus \{x\} = E) \leq C_{\mathsf{d}}\varepsilon$$

for all ε small enough. From this we immediately conclude the strong domination results from both sides, and these easily imply (6.1.5) and (6.1.6).

Remark 6.2.5. When d = 4 the calculation (6.2.8) is still valid, but we do no longer have a uniform upper bound on $G_{\Lambda \setminus E}(x, x)$. Let us point out for future use though that (6.2.8) and Lemma 6.2.4 imply that

$$\zeta^{\varepsilon}_{\Lambda}(\mathcal{A} \ni x \mid \mathcal{A} \setminus \{x\} = E) \leq C\varepsilon$$

and thus the measure $\zeta^{\varepsilon}_{\Lambda}$ is strongly dominated by the Bernoulli measure on $\mathfrak{P}(\Lambda)$ with parameter $p'_{4,+} := C\varepsilon$.

Proof of Theorem 6.1.3 d). One could prove (6.1.8) analogously as in [BV01, Section 3.2]. We, however, give a slightly different proof in the following.

The events $\mathcal{A} \ni x$ for $x \in E$ are decreasing, and so by the FKG property of $\zeta_{\Lambda}^{\varepsilon}$ we have

$$\zeta^{\varepsilon}_{\Lambda}(\mathcal{A} \cap E = \varnothing) = \zeta^{\varepsilon}_{\Lambda}\left(\bigcap_{x \in E} \{\mathcal{A} \not\ni x\}\right) \geq \prod_{x \in E} \zeta^{\varepsilon}_{\Lambda}(\mathcal{A} \not\ni x).$$

Thus, to establish (6.1.8) it suffices to show

$$\zeta^{\varepsilon}_{\Lambda}(\mathcal{A} \not\ni x) \geq 1 - C_{\alpha} \frac{\varepsilon}{|\log \varepsilon|^{1/2}}$$

or equivalently

$$\zeta^{\varepsilon}_{\Lambda}(\mathcal{A}\ni x) \le C_{\alpha} \frac{\varepsilon}{|\log\varepsilon|^{1/2}}$$
(6.2.9)

where the constant C_{α} depends only on α .

For this we consider the box $Q := Q_{\min(\varepsilon^{-\alpha}, \varepsilon^{-1/5})}(x)$. We can write

$$\zeta_{\Lambda}^{\varepsilon}(\mathcal{A} \ni x) = \zeta_{\Lambda}^{\varepsilon}(\mathcal{A} \cap Q = \{x\}) + \zeta_{\Lambda}^{\varepsilon}(\mathcal{A} \cap Q \supsetneq \{x\})$$

$$\leq \zeta_{\Lambda}^{\varepsilon}(\mathcal{A} \ni x \mid \mathcal{A} \cap (Q \setminus \{x\}) = \emptyset) + \zeta_{\Lambda}^{\varepsilon}(\mathcal{A} \cap Q \supsetneq \{x\}).$$
(6.2.10)

By Remark 6.2.5 the second summand can be estimated as

$$\begin{aligned} \zeta_{\Lambda}^{\epsilon}(\mathcal{A} \cap Q \supsetneq \{x\}) &\leq p_{4,+}' - p_{4,+}'(1 - p_{4,+}')^{|Q|} \\ &= C\epsilon(1 - (1 - C\epsilon)^{|Q| - 1}) \\ &\leq C\epsilon^{2}|Q| \\ &\leq C\epsilon^{2} \left(\epsilon^{-1/5}\right)^{4} \\ &= C\epsilon^{6/5} \end{aligned}$$
(6.2.11)

whenever ε is small enough. For the first summand we can use the FKG property once more and then proceed as in (6.2.8) to see that

$$\begin{split} \zeta_{\Lambda}^{\varepsilon}(\mathcal{A} \ni x \mid \mathcal{A} \cap (Q \setminus \{x\}) &= \varnothing) \leq \zeta_{\Lambda}^{\varepsilon}(\mathcal{A} \ni x \mid \mathcal{A} \cap (Q \setminus \{x\}) = \varnothing, \mathcal{A} \supset \Lambda \setminus Q) \\ &= \zeta_{Q}^{\varepsilon}(\mathcal{A} \ni x \mid \mathcal{A} \subset \{x\}) \\ &= \frac{\zeta_{Q}^{\varepsilon}(\{x\})}{\zeta_{Q}^{\varepsilon}(\varnothing) + \zeta_{Q}^{\varepsilon}(\{x\})} \\ &= \left(1 + \frac{Z_{Q}}{\varepsilon Z_{Q \setminus \{x\}}}\right)^{-1} \\ &= \left(1 + \frac{\sqrt{2\pi G_{Q}(x,x)}}{\varepsilon}\right)^{-1}. \end{split}$$

From Lemma 6.2.4 we know

$$G_Q(x,x) \ge \frac{1}{C} \log \left(1 + \min(\varepsilon^{-\alpha}, \varepsilon^{-1/5}) \ge \frac{1}{C_{\alpha}} |\log \varepsilon| \right)$$

and thus

$$\zeta_{\Lambda}^{\varepsilon}(\mathcal{A} \ni x \mid \mathcal{A} \cap (Q \setminus \{x\}) \leq C_{\alpha} \frac{\varepsilon}{|\log \varepsilon|^{1/2}}.$$

When we combine this with (6.2.10) and (6.2.11) we obtain (6.2.9). This completes the proof. $\hfill \Box$

In this proof the choice of $\varepsilon^{-1/5}$ for the halfdiameter of Q might seem arbitrary. Indeed, one could also choose $\varepsilon^{-1/4} |\log \varepsilon|^{-1/8}$ and obtain the same result. This is still smaller than λ_{mic} which is the actual length scale that one expects here. However, because we have to use Remark 6.2.5 instead of a comparison with $p_{4,+}$, we lose some logarithmic factor and hence cannot use the natural length scale for the size of Q. Fortunately, this does not affect the proof as the estimate (6.2.11) shows that the second summand in (6.2.10) is of lower order.

Proof of Theorem 6.1.3 c). The following proof is based on the proof in [BV01, Section 3.3], which itself is based on [DV00, IV00]. However, that proof is a bit hard to follow as one has to refer to all three references. Furthermore, there is a small mistake in [IV00] that needs to be fixed (cf. Remark 6.2.6 below). Thus, we give a complete proof for the case at hand.

Step 1: Growing microscopic polymers

For reasons that will become clear in the next step we need a procedure to grow microscopic polymers in a controlled way. Thus, we begin with the necessary definitions.

Let *K* be an odd integer to be fixed later (in (6.2.18)). We consider the polymers in $\mathcal{P}_{K\lambda_{mic}}$. Let $E \in \mathcal{P}_{K\lambda_{mic}}$ be such a polymer. Suppose that it has *n* connected components . We want to define for any multiindex $\underline{k} \in \mathbb{N}^n$ an enlarged polymer $E^{(\underline{k})} \in \mathcal{P}_{K\lambda_{mic}}$ in such a way that we add k_i boxes to the *i*-th connected component.

To be precise, fix some enumeration of the boxes in $Q_{K\lambda_{mic}}$ by the natural numbers. Let the connected components of *E* be E_1, \ldots, E_n , named in such a way that the minimal label of a box in E_i increases with *i*.

For $i \in \{1, ..., n\}$, $j \in \{0, ..., k_i\}$ we define inductively a polymer $E^{(i,j)} \supset E$ as follows. If j = 0, we let $E^{(i,j)} = E^{(i-1,k_{i-1})}$ (and $E^{(1,0)} = E$). If j > 0, let \tilde{E}_j be the connected component of $E^{(i,j-1)}$ that contains E_j , let $Q^{(i,j)} \in Q_{K\lambda_{\text{mic}}}$ be the box of smallest index that touches \tilde{E}_j , and let $E^{(i,j)} = E^{(i,j-1)} \cup Q^{(i,j)}$. Finally we let $E^{\underline{k}} = E^{(n,k_n)}$.

Let us note some properties of $E^{\underline{k}}$. First of all, it contains precisely $|\underline{k}|_1 := k_1 + \ldots + k_n$ boxes of $Q_{K\lambda_{\text{mic}}}$ more than *E*. In other words,

$$\left| \underline{E^{\underline{k}}} \right| = |\underline{E}| + |\underline{k}|_1 K^4 \lambda_{\text{mic}}^4 \,. \tag{6.2.12}$$

Furthermore, $E^{\underline{k}}$ has at most *n* connected components. Each E_i is contained in one of the connected components of $E^{\underline{k}}$, and the latter has grown by at least k_i boxes. Also each fixed box in $Q_{K\lambda_{\text{mic}}}$ is eventually contained in $E^{\underline{k}}$ whenever $|\underline{k}|_1$ is large enough. Let us also note that each connected component of *E* consists of at least one box. Therefore we have the estimate

$$n \le \frac{|E|}{K^4 \lambda_{\rm mic}^4} \,. \tag{6.2.13}$$

Step 2: Estimate for microscopic polymers

We first prove (6.1.7) for the special case that *E* is a polymer in $\mathcal{P}_{K\lambda_{\min}}$, where *K* is a constant as in Step 1. That is, we claim that there is $\varepsilon_{4,*}$ such that for any $E \subset \Lambda$ such that $E \in \mathcal{P}_{K\lambda_{\min}}$ and any $\varepsilon < \varepsilon_{4,*}$ we have

$$\left(1 - C\frac{\varepsilon}{|\log\varepsilon|^{1/2}}\right)^{|E|} \ge \zeta_{\Lambda}^{\varepsilon}(\mathcal{A} \cap E = \emptyset).$$
(6.2.14)

Suppose that *E* has *n* connected components, and consider for $\underline{k} \in \mathbb{N}^n$ the polymers $E^{(\underline{k})}$ constructed in the previous section. For $\underline{l} \in \mathbb{N}^n$ we write $\underline{l} > \underline{k}$ to denote $l_i \ge k_i$ for all *i* and $l_i > k_i$ for at least one *i*. Recall that $\tilde{\mathcal{A}} = \mathcal{A} \cup (\mathbb{Z}^d \setminus \Lambda)$.

For $|\underline{k}|_1$ large enough we have $E^{(\underline{k})} \not\subset \Lambda$ and therefore $\tilde{\mathcal{A}} \cap E^{(\underline{k})} \neq \emptyset$ almost surely. Thus

$$\zeta_{\Lambda}^{\varepsilon}(\mathcal{A} \cap E = \emptyset) = \zeta_{\Lambda}^{\varepsilon}(\tilde{\mathcal{A}} \cap E = \emptyset) = \zeta_{\Lambda}^{\varepsilon} \left(\exists \underline{k} \in \mathbb{N}^{n} \colon \tilde{\mathcal{A}} \cap E^{(\underline{k})} = \emptyset, \tilde{\mathcal{A}} \cap E^{(\underline{l})} \neq \emptyset \; \forall \underline{l} > \underline{k} \right)$$

and so in particular

$$\begin{aligned} \zeta_{\Lambda}^{\varepsilon}(\mathcal{A} \cap E = \varnothing) &\leq \sum_{\underline{k} \in \mathbb{N}^{n}} \zeta_{\Lambda}^{\varepsilon} \left(\tilde{\mathcal{A}} \cap E^{(\underline{k})} = \varnothing, \tilde{\mathcal{A}} \cap E^{(\underline{l})} \neq \varnothing \; \forall \underline{l} > \underline{k} \right) \\ &\leq \sum_{\underline{k} \in \mathbb{N}^{n}} \zeta_{\Lambda}^{\varepsilon} \left(\tilde{\mathcal{A}} \cap E^{(\underline{k})} = \varnothing \middle| \tilde{\mathcal{A}} \cap E^{(\underline{l})} \neq \varnothing \; \forall \underline{l} > \underline{k} \right) \;. \end{aligned}$$
(6.2.15)

Note that this sum is actually a finite sum as for large enough $|\underline{k}|_1$ the conditional probability is equal to 0. Let us estimate the summands in (6.2.15) separately. We have

$$\begin{aligned} \zeta_{\Lambda}^{\varepsilon} \left(\tilde{\mathcal{A}} \cap E^{(\underline{k})} = \varnothing \middle| \tilde{\mathcal{A}} \cap E^{(\underline{l})} \neq \varnothing \forall \underline{l} > \underline{k} \right) &= \frac{\zeta_{\Lambda}^{\varepsilon} \left(\tilde{\mathcal{A}} \cap E^{(\underline{k})} = \varnothing, \tilde{\mathcal{A}} \cap E^{(\underline{l})} \neq \varnothing \forall \underline{l} > \underline{k} \right)}{\zeta_{\Lambda}^{\varepsilon} \left(\tilde{\mathcal{A}} \cap E^{(\underline{l})} \neq \varnothing \forall \underline{l} > \underline{k} \right)} \\ &= \frac{\sum_{\substack{\Lambda \subset \Lambda \setminus E^{(\underline{k})} \\ \tilde{\mathcal{A}} \cap E^{(\underline{l})} \neq \varnothing \forall \underline{l} > \underline{k} \\ \frac{\tilde{\Lambda} \cap E^{(\underline{l})} \neq \varnothing \forall \underline{l} > \underline{k}}{\sum_{\substack{\Lambda' \subset \Lambda \\ \tilde{\mathcal{A}}' \cap E^{(\underline{l})} \neq \varnothing \forall \underline{l} > \underline{k}}} \varepsilon^{|\Lambda|} Z_{\Lambda \setminus \Lambda} \\ &= \frac{\sum_{\substack{\Lambda \subset \Lambda \setminus E^{(\underline{k})} \\ \tilde{\Lambda} \cap E^{(\underline{l})} \neq \varnothing \forall \underline{l} > \underline{k}}}{\sum_{\substack{B \subset E^{(\underline{k})} \\ \tilde{\Lambda} \cap E^{(\underline{l})} \neq \varnothing \forall \underline{l} > \underline{k}}} \varepsilon^{|\Lambda| - |B|} Z_{\Lambda \setminus (\Lambda \cup B)} \\ &= \left(\sum_{\substack{B \subset E^{(\underline{k})} \\ B \subset E^{(\underline{k})} \\ \varepsilon \in E^{(\underline{$$

where we have used $\frac{\sum_{i \in I} x_i}{\sum_{i \in I} y_i} \ge \min_{i \in I} \frac{x_i}{y_i}$ in the last step. Next, we estimate this minimum from below, at least for sufficiently many sets *B*. Let $m = \frac{|E^{(\underline{k})}|}{K^4 \lambda_{\text{mic}}^4}$ be the number of boxes in $E^{(\underline{k})}$. We will consider the class of sets *B* that contain exactly one point in each box of $E^{(\underline{k})}$.

Consider some $A \subset \Lambda \setminus E^{(\underline{k})}$ such that $\tilde{A} \cap E^{(\underline{l})} \neq \emptyset$ for all l > k. The properties of A imply that each connected component of $E^{(\underline{k})}$ touches a box that contains a point of \tilde{A} , as otherwise we could still grow one of the components (by choosing a larger multiindex) without intersecting \tilde{A} . Therefore we can enumerate the boxes of $E^{(\underline{k})}$ as D_1, \ldots, D_m in such a way that each D_i touches a box that contains a point of \tilde{A} or a box D_j with j < i. As mentioned, we consider sets $B = \{b_1, \dots, b_m\}$ that contain one point b_i in each box D_i . Let $B_i = \{b_1, \ldots, b_i\}$ (and $B_0 = \emptyset$). We have that

$$\frac{Z_{\Lambda \setminus (A \cup B)}}{Z_{\Lambda \setminus A}} = \prod_{i=1}^m \frac{Z_{\Lambda \setminus (A \cup B_i)}}{Z_{\Lambda \setminus (A \cup B_{i-1})}}$$

Pick some $i \in \{1, ..., m\}$. Our construction of the D_i ensures that D_i touches a box containing a point of $\tilde{A} \cup B_{i-1}$. In particular, $b_i \in D_i$ has distance at most $\sqrt{2^2 + 1^2 + 1^2 + 1^2} K \lambda_{\text{mic}} =$ $\sqrt{7}K\lambda_{\text{mic}}$ from a point in $\tilde{A} \cup B_{i-1}$. Now, (6.2.2) and Lemma 6.2.4 imply that

$$\frac{Z_{\Lambda \setminus (A \cup B_i)}}{Z_{\Lambda \setminus (A \cup B_{i-1})}} = \frac{1}{\sqrt{2\pi G_{\Lambda \setminus (A \cup B_{i-1})}(b_i, b_i)}}$$
$$\geq \frac{1}{C\sqrt{\log\left(1+\sqrt{7}K\lambda_{\rm mic}\right)}}$$
$$\geq \frac{1}{C|\log\varepsilon|^{1/2}}$$

as soon as ε is small enough (depending on *K*). Thus,

$$\frac{Z_{\Lambda \setminus (A \cup B)}}{Z_{\Lambda \setminus A}} \geq \left(\frac{1}{C |\log \varepsilon|^{1/2}}\right)^m.$$

This estimate holds for all $A \subset \Lambda \setminus E^{(\underline{k})}$ such that $\tilde{A} \cap E^{(\underline{l})} \neq \emptyset$ for all $\underline{l} > \underline{k}$, and all *B* that contain exactly one point in each box of $E^{(\underline{k})}$. The number of such sets *B* is $(K^4 \lambda_{\text{mic}}^4)^m$, and so (6.2.16) implies that

$$\begin{aligned} \zeta_{\Lambda}^{\varepsilon} \left(\tilde{\mathcal{A}} \cap E^{(\underline{k})} = \varnothing \left| \tilde{\mathcal{A}} \cap E^{(\underline{l})} \neq \varnothing \, \forall \underline{l} > \underline{k} \right) &\leq \left((K^{4} \lambda_{\text{mic}}^{4})^{m} \varepsilon^{m} \left(\frac{1}{C |\log \varepsilon|^{1/2}} \right)^{m} \right)^{-1} \\ &\leq \left(\left(2K \frac{|\log \varepsilon|^{1/8}}{\varepsilon^{1/4}} \right)^{4} \frac{\varepsilon}{C |\log \varepsilon|^{1/2}} \right)^{-m} \quad (6.2.17) \\ &= \left(\frac{K^{4}}{\gamma} \right)^{-m} \end{aligned}$$

for a certain constant γ . We can now choose *K* as an odd integer such that

$$K \ge (e\gamma)^{1/4}$$
. (6.2.18)

Then (6.2.17) in combination with (6.2.12) implies

$$\begin{split} \zeta_{\Lambda}^{\varepsilon} \left(\tilde{\mathcal{A}} \cap E^{(\underline{k})} = \varnothing \left| \tilde{\mathcal{A}} \cap E^{(\underline{l})} \neq \varnothing \; \forall \underline{l} > \underline{k} \right) &\leq \exp(-m) \\ &= \exp\left(-\frac{|E^{(\underline{k})}|}{K^4 \lambda_{\mathrm{mic}}^4} \right) \\ &= \exp\left(-\frac{|E|}{K^4 \lambda_{\mathrm{mic}}^4} - |\underline{k}|_1 \right) \,. \end{split}$$

Now we can use this result in (6.2.15) and obtain

$$\begin{split} \zeta_{\Lambda}^{\epsilon}(\mathcal{A} \cap E = \varnothing) &\leq \sum_{\underline{k} \in \mathbb{N}^n} \exp\left(-\frac{|E|}{K^4 \lambda_{\text{mic}}^4} - |\underline{k}|_1\right) \\ &= \exp\left(-\frac{|E|}{K^4 \lambda_{\text{mic}}^4}\right) \left(\sum_{k_1=0}^{\infty} \exp(-k_1)\right) \cdot \ldots \cdot \left(\sum_{k_n=0}^{\infty} \exp(-k_1)\right) \\ &= \exp\left(-\frac{|E|}{K^4 \lambda_{\text{mic}}^4}\right) \left(\frac{e}{e-1}\right)^n \\ &= \exp\left(-\frac{|E|}{K^4 \lambda_{\text{mic}}^4} + n(1 - \log(e-1))\right) \,. \end{split}$$

Finally, we can recall (6.2.13) and conclude

$$\zeta_{\Lambda}^{\varepsilon}(\mathcal{A} \cap E = \varnothing) \le \exp\left(-\frac{|E|}{K^{4}\lambda_{\mathrm{mic}}^{4}} + \frac{|E|}{K^{4}\lambda_{\mathrm{mic}}^{4}}(1 - \log(e - 1))\right)$$

$$= \exp\left(-\log(e-1)\frac{|E|}{K^4\lambda_{\rm mic}^4}\right)$$
$$\leq \exp\left(-C\frac{\varepsilon}{|\log\varepsilon|^{1/2}}|E|\right)$$
$$\leq \left(1 - C\frac{\varepsilon}{|\log\varepsilon|^{1/2}}\right)^{|E|}$$

whenever ε is small enough, and the *K* (that is now fixed) has been absorbed into the constant. This completes the proof of (6.2.14).

Step 3: Density of pinned points on macroscopic scales We now show that on the length scale λ_{mac} most points of a set $E \subset \Lambda$ are close to a point in $\tilde{\mathcal{A}}$. To make this precise, we need to make a few definitions. Let L be an odd integer to be fixed later (in (6.2.20) and (6.2.22)). We consider polymers in $\mathcal{P}_{KL\lambda_{mac}}$. Observe that $KL\lambda_{mac}$ is an odd multiple of $K\lambda_{mic}$, the lengthscale from Step 2. For $E \subset \Lambda$ let

$$S_E = \{ Q \in \mathcal{Q}_{KL\lambda_{\max}} \colon Q \cap E \neq \emptyset \}$$

and

$$S_{E,\text{bad}}(\mathcal{A}) = \{ Q \in S_E \colon Q \cap \tilde{A} = \varnothing \}.$$

We think of the boxes in $S_{E,\text{bad}}(\mathcal{A})$ as bad boxes, as they contain points of E but no pinned point. We will show that not too many boxes are bad. Note that $|S_E| \ge \frac{|E|}{(KL)^4 \lambda_{\text{mac}}^4}$. Our claim now is that there is $\varepsilon_{4,*}$ such that for any $E \subset \Lambda$ and any $\varepsilon < \varepsilon_{4,*}$ we have

$$\zeta_{\Lambda}^{\varepsilon} \left(|S_{E,\text{bad}}(\mathcal{A})| > \frac{|E|}{2(KL)^4 \lambda_{\text{mac}}^4} \right) \le \left(1 - C \frac{\varepsilon}{|\log \varepsilon|^{1/2}} \right)^{|E|} .$$
(6.2.19)

To see this, we use the result from the previous step to estimate

$$\begin{split} \zeta_{\Lambda}^{\varepsilon} \left(|S_{E,\text{bad}}(\mathcal{A})| > \frac{|E|}{2(KL)^{4}\lambda_{\text{mac}}^{4}} \right) \\ &= \sum_{\substack{T \subseteq S_{E} \\ |T| \ge |E|/(2(KL)^{4}\lambda_{\text{mac}}^{4})}} \zeta_{\Lambda}^{\varepsilon} (S_{E,\text{bad}}(\mathcal{A}) = T) \\ &\leq \sum_{\substack{T \subseteq S_{E} \\ |T| \ge |E|/(2(KL)^{4}\lambda_{\text{mac}}^{4})}} \zeta_{\Lambda}^{\varepsilon} \left(\mathcal{A} \cap \bigcup_{Q \in T} Q = \varnothing \right) \\ &= \sum_{\substack{T \subseteq S_{E} \\ |T| \ge |E|/(2(KL)^{4}\lambda_{\text{mac}}^{4})}} \zeta_{\Lambda}^{\varepsilon} \left(\mathcal{A} \cap \bigcup_{Q \in T} Q = \varnothing \right) \\ &\leq \sum_{\substack{T \subseteq S_{E} \\ |T| \ge |E|/(2(KL)^{4}\lambda_{\text{mac}}^{4})}} \left(1 - C \frac{\varepsilon}{|\log \varepsilon|^{1/2}} \right)^{|T|(KL)^{4}\lambda_{\text{mac}}^{4}} \\ &= \sum_{j=\lceil |E|/(2(KL)^{4}\lambda_{\text{mac}}^{4})\rceil} \left(|S_{E}| \atop j \right) \left(1 - C \frac{\varepsilon}{|\log \varepsilon|^{1/2}} \right)^{j(KL)^{4}\lambda_{\text{mac}}^{4}} \\ &\leq \sum_{j=\lceil |E|/(2(KL)^{4}\lambda_{\text{mac}}^{4})\rceil} \left(|S_{E}| \atop j \right) \exp \left(-C \frac{\varepsilon}{|\log \varepsilon|^{1/2}} (KL)^{4}\lambda_{\text{mac}}^{4} \right) \end{split}$$

$$\leq \sum_{j=\lceil |E|/(2(KL)^4\lambda_{\mathrm{mac}}^4)\rceil}^{|S_E|} {|S_E| \choose j} \exp\left(-C\frac{\varepsilon}{|\log\varepsilon|^{1/2}}(KL)^4\frac{|\log\varepsilon|^{3/2}}{\varepsilon}j\right)$$
$$= \sum_{j=\lceil |E|/(2(KL)^4\lambda_{\mathrm{mac}}^4)\rceil}^{|S_E|} {|S_E| \choose j} \varepsilon^{(KL)^4\gamma'j}$$

for a certain constant γ' . We now want to apply the estimate for binomial sums that is stated in Lemma 6.2.7 below with $N = |S_E|$, $p = \varepsilon^{(KL)^4} \gamma'^j$ and $r = \frac{|E|}{2(KL)^4 \lambda_{\text{mac}}^4 |S_E|}$. To do so, we need $p \le r \le \frac{1}{2}$. Because $1 \le \frac{|E|}{|S_E|} \le (KL)^4 \lambda_{\text{mac}}^4$ we always have $r \le \frac{1}{2}$, and for $p \le r$ it suffices that $\varepsilon^{(KL)^4} \gamma'^j \le \frac{\varepsilon}{2(KL)^4 |\log \varepsilon|^{3/2}}$. To ensure the latter we choose *L* such that

$$L > \frac{\gamma'^{1/4}}{K} \tag{6.2.20}$$

and ε is small enough. Using Lemma 6.2.7 we then obtain

$$\zeta_{\Lambda}^{\varepsilon} \left(|S_{E,\text{bad}}(\mathcal{A})| > \frac{|E|}{2(KL)^4 \lambda_{\text{mac}}^4} \right) \le \left(\frac{p}{r^2}\right)^{r|S_E|} \,. \tag{6.2.21}$$

We can estimate that

$$\begin{split} \frac{p}{r^2} &= \exp\left(-(KL)^4 \gamma' |\log \varepsilon| - 2\log \frac{|E|}{2(KL)^4 \lambda_{\text{mac}}^4 |S_E|}\right) \\ &\leq \exp\left(-(KL)^4 \gamma' |\log \varepsilon| + 2\log \frac{1}{2(KL)^4 \lambda_{\text{mac}}^4}\right) \\ &\leq \exp\left(-(KL)^4 \gamma' |\log \varepsilon| + 2|\log \varepsilon| + 2\log(2(KL)^4) + \log(|\log \varepsilon|^{3/2})\right) \,. \end{split}$$

Provided that we choose

$$L > \frac{(2\gamma')^{1/4}}{K}$$
(6.2.22)

we can estimate this as

$$\frac{p}{r^2} \le \exp(-C|\log\varepsilon|)$$

whenever ε is small enough (depending on *K*, *L* that are now fixed). Returning to (6.2.21), we see that

$$\begin{split} \zeta_{\Lambda}^{\varepsilon} \left(|S_{E,\text{bad}}(\mathcal{A})| > \frac{|E|}{2(KL)^{4}\lambda_{\text{mac}}^{4}} \right) &\leq \exp\left(-C|\log\varepsilon|\frac{|E|}{2(KL)^{4}\lambda_{\text{mac}}^{4}}|S_{E}|\right) \\ &\leq \exp\left(-C\frac{\varepsilon}{|\log\varepsilon|^{1/2}}|E|\right) \end{split}$$

which implies (6.2.19).

Step 4: Estimate for arbitrary sets

We now can prove the actual result (6.1.7). So let $E \subset \Lambda$. Using the notation from the previous step, we let

$$E_{\mathrm{bad}}(\mathcal{A}) = E \cap \bigcup_{Q \in S_{E,\mathrm{bad}}(\mathcal{A})} Q$$

be the set of bad points (those which are far from a pinned point). We have the estimate $|E_{\text{bad}}(\mathcal{A})| \leq (KL)^4 \lambda_{\text{mac}}^4 |S_{E,\text{bad}}(\mathcal{A})|$ and so the previous step implies that

$$\zeta^{\varepsilon}_{\Lambda}\left(|E_{\text{bad}}(\mathcal{A})| > \frac{|E|}{2}\right) \leq \zeta^{\varepsilon}_{\Lambda}\left(|S_{E,\text{bad}}(\mathcal{A})| > \frac{|E|}{2(KL)^4\lambda_{\text{mac}}^4}\right) \leq \left(1 - C\frac{\varepsilon}{|\log\varepsilon|^{1/2}}\right)^{|E|}.$$

We can now write

$$\begin{aligned} \zeta_{\Lambda}^{\varepsilon}(\mathcal{A} \cap E = \emptyset) &\leq \zeta_{\Lambda}^{\varepsilon} \left(\mathcal{A} \cap E = \emptyset, |E_{\text{bad}}(\mathcal{A})| \leq \frac{|E|}{2}\right) + \zeta_{\Lambda}^{\varepsilon} \left(|E_{\text{bad}}(\mathcal{A})| > \frac{|E|}{2}\right) \\ &\leq \zeta_{\Lambda}^{\varepsilon} \left(\mathcal{A} \cap E = \emptyset, |E_{\text{bad}}(\mathcal{A})| \leq \frac{|E|}{2}\right) + \left(1 - C\frac{\varepsilon}{|\log \varepsilon|^{1/2}}\right)^{|E|} \end{aligned}$$

and so we only need to estimate the first term to establish (6.1.7). If $\zeta_{\Lambda}^{\varepsilon} \left(|E_{\text{bad}}(\mathcal{A})| \leq \frac{|E|}{2} \right) = 0$, that term is equal to 0 and we are trivially done. So we can assume otherwise, and estimate

$$\zeta_{\Lambda}^{\varepsilon}\left(\mathcal{A}\cap E=\varnothing, |E_{\text{bad}}(\mathcal{A})|\leq \frac{|E|}{2}\right)\leq \zeta_{\Lambda}^{\varepsilon}\left(\mathcal{A}\cap E=\varnothing\left||E_{\text{bad}}(\mathcal{A})|\leq \frac{|E|}{2}\right).$$

Next, we can apply a similar argument as in (6.2.16) to see that

$$\zeta_{\Lambda}^{\epsilon} \left(\mathcal{A} \cap E = \varnothing \left| |E_{\text{bad}}(\mathcal{A})| \leq \frac{|E|}{2} \right) = \frac{\sum\limits_{\substack{A \subset \Lambda \setminus E \\ |E_{\text{bad}}(A)| \leq |E|/2}} \varepsilon^{|A|+|B|} Z_{\Lambda \setminus (A \cup B)}}{\sum\limits_{\substack{B \subset E \\ |E_{\text{bad}}(A)| \leq |E|/2}} \varepsilon^{|A|+|B|} Z_{\Lambda \setminus (A \cup B)}}{\left| \frac{|E_{\text{bad}}(A)| \leq |E|/2}{\sum\limits_{\substack{A \subset \Lambda \setminus E \\ |E_{\text{bad}}(A)| \leq |E|/2}} \varepsilon^{|A|+|B|} Z_{\Lambda \setminus (A \cup B)}}{\sum\limits_{\substack{A \subset \Lambda \setminus E \\ |E_{\text{bad}}(A)| \leq |E|/2}} \varepsilon^{|A|} Z_{\Lambda \setminus A}} \right)^{-1}}$$
(6.2.23)
$$\leq \left(\min_{\substack{A \subset \Lambda \setminus E \\ |E_{\text{bad}}(A)| \leq |E|/2}} \sum\limits_{\substack{B \subset E \\ |E_{\text{bad}}(A)| \leq |E|/2}} \varepsilon^{|B|} \frac{Z_{\Lambda \setminus (A \cup B)}}{Z_{\Lambda \setminus A}}}{\sum\limits_{A \setminus A}} \right)^{-1}.$$

Note that unlike in (6.2.16) we interchanged the summations over *A* and *B* in an intermediate step, which allows us to have min_{*A*} \sum_{B} instead of \sum_{B} min_{*A*} in the result of this calculation.

We can estimate this further by only allowing good points for *B*, that is by estimating

$$\zeta_{\Lambda}^{\varepsilon} \left(\mathcal{A} \cap E = \varnothing \left| |E_{\text{bad}}(\mathcal{A})| \le \frac{|E|}{2} \right) \le \left(\min_{\substack{A \subset \Lambda \setminus E \\ |E_{\text{bad}}(A)| \le |E|/2}} \sum_{B \subset E \setminus E_{\text{bad}}(A)} \varepsilon^{|B|} \frac{Z_{\Lambda \setminus (A \cup B)}}{Z_{\Lambda \setminus A}} \right)^{-1}.$$
(6.2.24)

Consider some $A \subset \Lambda \setminus E$, and some $B \subset E \setminus E_{bad}(A)$. By definition of $E_{bad}(A)$, each point in *B* is in the same macroscopic box as a point of \tilde{A} . In particular, each point in *B* has distance at most $\sqrt{7}KL\lambda_{mac}$ to a point of \tilde{A} . Thus, if we let $B = \{b_1, \ldots, b_{|B|}\}$, and $B_i = \{b_1, \ldots, b_i\}$ we see as in Step 2 that

$$rac{Z_{\Lambda ackslash (A \cup B)}}{Z_{\Lambda ackslash A}} = \prod_{i=1}^{|B|} rac{Z_{\Lambda ackslash (A \cup B_i)}}{Z_{\Lambda ackslash (A \cup B_{i-1})}}
onumber \ = \prod_{i=1}^{|B|} rac{1}{\sqrt{2\pi G_{\Lambda ackslash (A \cup B_{i-1})}}(b_i, b_i)}$$

$$\geq \prod_{i=1}^{|B|} \frac{1}{C\sqrt{\log\left(1+\sqrt{7}KL\lambda_{\max}\right)}} \\ \geq \left(\frac{1}{C|\log\varepsilon|^{1/2}}\right)^{|B|}$$

where we used (6.2.2) and Lemma 6.2.4. Returning to (6.2.23) and (6.2.24), we obtain

$$\begin{split} \zeta_{\Lambda}^{\varepsilon} \left(\mathcal{A} \cap E = \varnothing \left| \left| E_{\text{bad}}(\mathcal{A}) \right| \leq \frac{|E|}{2} \right) \\ &\leq \left(\min_{\substack{A \subset \Lambda \setminus E \\ |E_{\text{bad}}(A)| \leq |E|/2}} \sum_{B \subset E \setminus E_{\text{bad}}(A)} \varepsilon^{|B|} \left(\frac{1}{C|\log \varepsilon|^{1/2}} \right)^{|B|} \right)^{-1} \\ &= \left(\min_{\substack{A \subset \Lambda \setminus E \\ |E_{\text{bad}}(A)| \leq |E|/2}} \sum_{j=0}^{|E \setminus E_{\text{bad}}(A)|} \left(\frac{|E \setminus E_{\text{bad}}(A)|}{j} \right) \left(\frac{\varepsilon}{C|\log \varepsilon|^{1/2}} \right)^{j} \right)^{-1} \\ &= \left(\min_{\substack{A \subset \Lambda \setminus E \\ |E_{\text{bad}}(A)| \leq |E|/2}} \left(1 + \frac{\varepsilon}{C|\log \varepsilon|^{1/2}} \right)^{|E|/2} \right)^{-1} \\ &\leq \left(\left(1 + \frac{\varepsilon}{C|\log \varepsilon|^{1/2}} \right)^{|E|/2} \right)^{-1} \\ &\leq \left(1 - \frac{\varepsilon}{C|\log \varepsilon|^{1/2}} \right)^{|E|} . \end{split}$$

This finally completes the proof.

Remark 6.2.6. In [BV01] a similar argument is used. However, for growing the polymers [BV01] refers to [IV00], where a construction that is different from ours is used. Unfortunately, the argument from [IV00] contains a small gap.

The problem is as follows: Take d \geq 2. In [IV00] the grown polymer $\tilde{E}^{\underline{k}}$ is only defined for certain admissible \underline{k} . Using our notation, one defines $\tilde{E}^{\underline{k}}$ by adding k_i layers of microscopic cubes to E_i , i.e. one replaces E by

$$\tilde{E}^{\underline{k}} := \bigcup_{i=1}^{n} E_i + Q_{k_i K \lambda_{\mathrm{mic}}}(0) \,.$$

However, this is only done if for each $i \in \{1, ..., n\}$ we have that $E_i + Q_{k_i K \lambda_{mic}}(0)$ and $\bigcup_{j=1}^{i-1} E_i + Q_{k_i K \lambda_{mic}}$ are disjoint or $k_i = 0$ (and the \underline{k} with this property are called admissible). Now in [IV00, p. 398] it is claimed that this construction satisfies

$$|\tilde{E}^{\underline{k}}| \ge |E| + |\underline{k}|_1 K^{\mathsf{d}} \lambda^{\mathsf{d}}_{\mathsf{mic}}, \qquad (6.2.25)$$

or in other words that we have added at least $|\underline{k}|_1$ boxes. This is not true in general: For example if *L* is a large odd number and

$$E_1 = \left[-\frac{K\lambda_{\rm mic}}{2}, \frac{K\lambda_{\rm mic}}{2}\right] \cap \mathbb{Z}^{\sf d}$$

175

$$E_2 = \left(\left[-\frac{KL\lambda_{mic}}{2}, \frac{KL\lambda_{mic}}{2} \right] \setminus \left[-\frac{3K\lambda_{mic}}{2}, \frac{3K\lambda_{mic}}{2} \right] \right) \cap \mathbb{Z}^d$$

and $E = E_1 \cup E_2$, then for any $k_1 \in \{1, \frac{L-1}{2}\}$ the multiindex $\underline{k} = (k_1, 0)$ is admissible, but to obtain $\tilde{E}^{\underline{k}}$ we only add the $3^d - 1$ cubes that form the gap between E_1 and E_2 . If L is large enough, we can take $k_1 \ge 3^d$, and we arrive at a contradiction to (6.2.25).

Note that this problem is not present in the construction that we used in Step 1 of the proof of Theorem 6.1.3 c), as our construction directly ensures that (6.2.12) holds. The same construction could also be used in [IV00] to fix the gap there.

Alternatively (as pointed out to the author by Yvan Velenik) one can also fix the gap in [IV00] by first ordering the E_k in such a way that no E_i completely surrounds an E_j with i < j.

In our proof of Theorem 6.1.3 c) we used a tail bound for certain binomial sums. We will use this estimate a few more times in Section 6.4.3, so we state and prove it separately.

Lemma 6.2.7. Let $N \in \mathbb{N}$, and $\frac{1}{2} \ge r \ge p \ge 0$. Then

$$\sum_{j=\lceil rN\rceil}^{N} \binom{N}{j} p^{j} \le \left(\frac{p}{r^{2}}\right)^{rN} .$$
(6.2.26)

This estimate is very similar to standard Chernoff tail bounds for the binomial distribution. A special case was used in [BV01, Section 3.3.2]. For the proof we will follow the proof of the Chernoff tail bound.

Proof. For any $t \ge 0$ we have the estimate

$$\sum_{j=\lceil rN\rceil}^{N} \binom{N}{j} p^{j} \le e^{-trN} \sum_{j=0}^{N} \binom{N}{j} e^{tj} p^{j}$$
$$\le e^{-trN} (1+e^{t}p)^{N}.$$

The optimal choice for *t* is $t = \log\left(\frac{r}{(1-r)p}\right)$, and this yields

$$\sum_{j=\lceil rN\rceil}^{N} \binom{N}{j} p^{j} \leq \left(\frac{(1-r)^{r-1}}{r^{r}}\right)^{N} p^{rN}$$

It remains to observe that for $0 < r \le \frac{1}{2}$ one has

$$\frac{(1-r)^{r-1}}{r^r} \le \frac{1}{r^{2r}} \,.$$

6.3 Some inequalities

In this section we provide some tools that will be used in the next two sections to establish Theorem 6.1.5, namely a discrete multipolar Hardy-Rellich inequality as well as an interpolation inequality. We begin with the former.

6.3.1 A discrete multipolar Hardy-Rellich inequality

We want to give a quantitative estimate on the strength of the pinning effect on $x \in \Lambda$. More precisely, consider a function $u: \Lambda \to \mathbb{R}$ such that u = 0 on $\tilde{A} = A \cup (\mathbb{Z}^d \setminus \Lambda)$. We want to control a weighted L^2 -norm of u by the L^2 -norm of $\nabla_1^2 u$. The weight at $x \in \Lambda$ will have to depend on the location of x with respect to \tilde{A} . If $\nabla_1^2 u$ is small, then u is (locally) close to an affine function. We need to ensure that this affine function is close to zero near x, and for this purpose we need that u is close to 0 at d + 1 points that are well-spread out, i.e. we need that x is close to d + 1 points.

To state our precise result we need some definitions. First we construct d + 1 cones of directions that are well-spread out: Let $\theta_1, \ldots, \theta_{d+1} \in \mathbb{S}^{d-1}$ be such that $\theta_i \cdot \theta_j = -\frac{1}{d}$ for $i \neq j$ (e.g. take $(\theta_i)_{i=1}^{d+1}$ to be the vertices of a regular d-dimensional simplex with circumsphere \mathbb{S}^{d-1}).

For $\kappa > 0$ let $\Theta_i = B_{\kappa}(\theta_i) \cap \mathbb{S}^{d-1}$. For κ small enough we have $\theta'_i \cdot \theta'_j < 0$ for all $\theta'_i \in \Theta_i$, $\theta'_j \in \Theta_j$ for $i \neq j$. Fix one such choice of κ . Finally let $\Xi_i = \left\{ y \in \mathbb{R}^d \setminus \{0\} : \frac{y}{|y|} \in \Theta_i \right\}$ (cf. Figure 6.2).



Figure 6.2: The sets Ξ_i for d = 2

For $x \in \Lambda$ let

$$d^{(i)}(x, \tilde{A}) = \inf_{a \in \tilde{A} \cap (x + \Xi_i)} |x - a|_1,$$

 $d_*(x, \tilde{A}) = \max_{i \in \{1, \dots, d+1\}} d^{(i)}(x, \tilde{A})$

with the convention that $\inf \emptyset = +\infty$. Thus, for each *x* there are d + 1 points in \tilde{A} which are well-spread out around *A* with distance at most $d_*(x, \tilde{A})$.

Then we have the following statement.

Theorem 6.3.1. Let $A \subset \Lambda$ be arbitrary. Let $V \subset \Lambda$ be an arbitrary subset. Let $R \in \mathbb{N}$, $R \ge 2$ be a parameter. Suppose that $u : \Lambda \to \mathbb{R}$ is such that u = 0 on \tilde{A} . Then

$$\left\| u \mathbb{1}_{d_{*}(\cdot,\tilde{A}) \leq R} \right\|_{L^{2}(V)}^{2} \leq C_{\mathsf{d}} R^{\mathsf{d}} (1 + \mathbb{1}_{\mathsf{d}=4} \log R) \left\| \nabla_{1}^{2} u \right\|_{L^{2}(V+Q_{R}(0))}^{2}$$
(6.3.1)

where the constant is independent of R, V and A.

Proof. We begin with the case $d \ge 5$. We fix an enumeration of the points in \mathbb{Z}^d .

We first establish a pointwise bound for $u \mathbb{1}_{d_*(\cdot, \tilde{A}) \leq R}$. Let $x \in V$ such that $d_*(x, \tilde{A}) \leq R$. For $i \in \{1, ..., d + 1\}$ consider the points in $\tilde{A} \cap (x + \Xi_i)$ of minimal l^1 -distance to x, and let $a_x^{(i)}$ be the one among those that comes first with respect to our fixed enumeration. By assumption $|x - a_x^{(i)}|_1 = d^{(i)}(x, \tilde{A}) \le R$.

We first claim

$$|u(x)| \leq \max_{i \in \{1, \dots, d+1\}} \left| u(a_x^{(i)}) - u(x) - \nabla_1 u(x) \cdot \left(a_x^{(i)} - x\right) \right|$$

=
$$\max_{i \in \{1, \dots, d+1\}} \left| u(x) + \nabla_1 u(x) \cdot \left(a_x^{(i)} - x\right) \right|.$$
 (6.3.2)

Indeed, we can assume $u(x) \ge 0$ (the other case is analogous). There is an index *i* such that $\nabla_1 u(x) \cdot (a_x^{(i)} - x) \ge 0$, as otherwise the d + 2 vectors

$$\nabla_1 u(x), a_x^{(1)} - x, \dots, a_x^{(d+1)} - x$$

would have pairwise negative scalar products, while it is easy to see that this is possible in \mathbb{R}^d for at most d + 1 vectors. In particular, we have

$$\max_{i \in \{1,\dots,\mathsf{d}+1\}} \nabla_1 u(x) \cdot \left(a_x^{(i)} - x\right) \ge 0$$

By assumption $u(a_x^{(i)}) = 0$ and so

$$u(x) \le u(x) + \max_{i \in \{1, \dots, d+1\}} \nabla_1 u(x) \cdot \left(a_x^{(i)} - x\right)$$

= $\max_{i \in \{1, \dots, d+1\}} u(x) - u(a_x^{(i)}) + \nabla_1 u(x) \cdot \left(a_x^{(i)} - x\right)$
 $\le \max_{i \in \{1, \dots, d+1\}} \left| u(x) - u(a_x^{(i)}) + \nabla_1 u(x) \cdot \left(a_x^{(i)} - x\right) \right|$

which implies (6.3.2).

We now want to pick a nearest neighbour path $\Psi_x^{(i)} = \left(\Psi_x^{(i)}(0), \ldots, \Psi_x^{(i)}(d^{(i)}(x, \tilde{A}))\right)$ such that $\Psi_x^{(i)}(0) = x$, $\Psi_x^{(i)}(d^{(i)}(x, \tilde{A})) = a_x^{(i)}$. We can pick this path in such a way that all of its points have distance at most \sqrt{d} from the straight line connecting x and $a_x^{(i)}$, and such that all but possible the first $\tilde{\alpha}_d$ of its vertices lie inside the widening cone $x + \Xi_i$ (cf. Figure 6.3). Here $\tilde{\alpha}_d$ is a constant depending only on d and the Ξ_i .



Figure 6.3: Choice of the path $\Psi_x^{(i)}$. We require all points to have distance at most \sqrt{d} from the straight line between *x* and $a_x^{(i)}$ (i.e. to be in the dashed strip), and all but the first $\tilde{\alpha}_d$ to be inside the cone $x + \Xi_i$.

We can now apply a discrete version of the fundamental theorem of calculus along the paths $\Psi_x^{(i)}$ to the function $v := u(\cdot) - u(x) - \nabla_1 u(x) \cdot (\cdot - x)$. Namely, we know v(x) = 0

and $\nabla_1 v(x) = 0$. The point $\Psi_x^{(i)}(1)$ is one of the 2d neighbours of x. If it happens that $\Psi_x^{(i)}(1) \in \{x + e_1, \dots, x + e_d\}$, then $v(\Psi_x^{(i)}(1)) = 0$ and we can write

$$v(a_x^{(i)}) = \sum_{t=0}^{d^{(i)}(x,A)-1} \sum_{s=1}^{t-1} v(\Psi_x^{(i)}(s+1)) - 2v(\Psi_x^{(i)}(s)) + v(\Psi_x^{(i)}(s-1)).$$

On the other hand, if $\Psi_x^{(i)}(1) \in \{x - e_1, \dots, x - e_d\}$, we can temporarily add a point $\Psi_x^{(i)}(-1) = 2x - \Psi_x^{(i)}(1)$ to our path, so that $v(\Psi_x^{(i)}(-1)) = 0$, and then write

$$v(a_x^{(i)}) = \sum_{t=0}^{d^{(i)}(x,\tilde{A})-1} \sum_{s=0}^{t-1} v(\Psi_x^{(i)}(s+1)) - 2v(\Psi_x^{(i)}(s)) + v(\Psi_x^{(i)}(s-1))$$

In both cases we can conclude that

$$\begin{aligned} \left| u(a_x^{(i)}) - u(x) - \nabla_1 u(x) \cdot \left(a_x^{(i)} - x \right) \right| &= \left| v(a_x^{(i)}) \right| \\ &\leq \sum_{t=0}^{d^{(i)}(x,\tilde{A}) - 1} \sum_{s=0}^{t-1} \left| \nabla_1^2 v(\Psi_x^{(i)}(s)) \right| \\ &= \sum_{t=0}^{d^{(i)}(x,\tilde{A}) - 1} \sum_{s=0}^{t-1} \left| \nabla_1^2 u(\Psi_x^{(i)}(s)) \right| \\ &= \sum_{s=0}^{d^{(i)}(x,\tilde{A}) - 1} (d^{(i)}(x,\tilde{A}) - s) \left| \nabla_1^2 u(\Psi_x^{(i)}(s)) \right| \end{aligned}$$

where we have changed the order of summation in the last step. Thus, (6.3.2) implies that

$$|u(x)| \leq \max_{i \in \{1, \dots, d+1\}} \left| u(a_x^{(i)}) - u(x) - \nabla_1 u(x) \cdot \left(a_x^{(i)} - x\right) \right|$$

$$= \max_{i \in \{1, \dots, d+1\}} \sum_{s=0}^{d^{(i)}(x, \tilde{A}) - 1} (d^{(i)}(x, \tilde{A}) - s) \left| \nabla_1^2 u(\Psi_x^{(i)}(s)) \right|.$$
(6.3.3)

We have this estimate for all *x* such that $d_*(x, \tilde{A}) \leq R$. Defining Ψ arbitrarily for the other *x* and summing the square of (6.3.3) over *x*, we find

$$\sum_{x \in V} |u(x)|^2 \mathbb{1}_{d_*(x,\tilde{A}) \le R} \le \sum_{x \in V} \mathbb{1}_{d_*(x,\tilde{A}) \le R} \left(\max_{i \in \{1,\dots,d\}} \sum_{s=0}^{d^{(i)}(x,\tilde{A})-1} (d^{(i)}(x,\tilde{A}) - s) \left| \nabla_1^2 u(\Psi_x^{(i)}(s)) \right| \right)^2 \le \sum_{i=1}^{d+1} \sum_{x \in V} \mathbb{1}_{d_*(x,\tilde{A}) \le R} \left(\sum_{s=0}^{d^{(i)}(x,\tilde{A})-1} (d^{(i)}(x,\tilde{A}) - s) \left| \nabla_1^2 u(\Psi_x^{(i)}(s)) \right| \right)^2.$$
(6.3.4)

Consider a nonzero summand of the outer two sums. Then $d_*(x, \tilde{A}) \leq R$, and (recalling that

 $d \ge 5$) we can apply Hölder's inequality to the innermost sum to obtain

$$\begin{pmatrix}
\binom{d^{(i)}(x,\tilde{A})-1}{\sum_{s=0}^{s=0}} (d^{(i)}(x,\tilde{A})-s) \left| \nabla_{1}^{2} u(\Psi_{x}^{(i)}(s)) \right| \\
\leq \left(\sum_{s=0}^{d^{(i)}(x,\tilde{A})-1} \frac{1}{(d^{(i)}(x,\tilde{A})-s)^{\mathsf{d}-3}} \right) \left(\sum_{s=0}^{d^{(i)}(x,\tilde{A})-1} (d^{(i)}(x,\tilde{A})-s)^{\mathsf{d}-1} \left| \nabla_{1}^{2} u(\Psi_{x}^{(i)}(s)) \right|^{2} \right) \\
\leq C_{\mathsf{d}} \sum_{s=0}^{d^{(i)}(x,\tilde{A})-1} (d^{(i)}(x,\tilde{A})-s)^{\mathsf{d}-1} \left| \nabla_{1}^{2} u(\Psi_{x}^{(i)}(s)) \right|^{2}.$$
(6.3)

(6.3.5) If $s > \tilde{\alpha}_{\mathsf{d}}$, we know $\Psi_x^{(i)}(s) \in x + \Xi_i$, and hence $\Psi_x^{(i)}(s) + \Xi_i \subset x + \Xi_i$, which implies $d^{(i)}(\Psi_x^{(i)}(s), \tilde{A}) \ge d^{(i)}(x, \tilde{A}) - s$. If $s \le \tilde{\alpha}_{\mathsf{d}}$, we can just use the estimate $(d^{(i)}(x, \tilde{A}) - s)^{\mathsf{d}-1} \le R^{\mathsf{d}-1}$.

Using this in (6.3.5) we obtain

$$\begin{split} & \left(\sum_{s=0}^{d^{(i)}(x,\tilde{A})-1} (d^{(i)}(x,\tilde{A})-s) \left| \nabla_{1}^{2} u(\Psi_{x}^{(i)}(s)) \right| \right)^{2} \\ & \leq C_{d} \sum_{s=\tilde{\alpha}_{d}+1}^{d^{(i)}(x,\tilde{A})-1} d^{(i)}(\Psi_{x}^{(i)}(s),\tilde{A})^{d-1} \left| \nabla_{1}^{2} u(\Psi_{x}^{(i)}(s)) \right|^{2} + C_{d} \sum_{s=0}^{\tilde{\alpha}_{d}} R^{d-1} \left| \nabla_{1}^{2} u(\Psi_{x}^{(i)}(s)) \right|^{2} \\ & \leq C_{d} \sum_{y \in \Psi_{x}^{(i)}} \left(\mathbb{1}_{|y-x|_{1} > \tilde{\alpha}_{d}} d^{(i)}(y,\tilde{A})^{d-1} + \mathbb{1}_{|y-x|_{1} \leq \tilde{\alpha}_{d}} R^{d-1} \right) \left| \nabla_{1}^{2} u(y) \right|^{2} \\ & \leq C_{d} \sum_{y \in \Psi_{x}^{(i)}} \left(\mathbb{1}_{|y-x|_{1} > \tilde{\alpha}_{d}} d^{(i)}(y,\tilde{A})^{d-1} + \mathbb{1}_{d^{(i)}(y,\tilde{A}) \leq \tilde{\alpha}_{d}} + \mathbb{1}_{|y-x|_{1} \leq \tilde{\alpha}_{d}} R^{d-1} \right) \left| \nabla_{1}^{2} u(y) \right|^{2} \\ & \leq C_{d} \sum_{y \in \Psi_{x}^{(i)}} \left(\mathbb{1}_{|y-x|_{1} > \tilde{\alpha}_{d}} d^{(i)}(y,\tilde{A})^{d-1} + \mathbb{1}_{|y-x|_{1} \leq \tilde{\alpha}_{d}} R^{d-1} + 1 \right) \left| \nabla_{1}^{2} u(y) \right|^{2} . \end{split}$$

We can insert this into (6.3.4) and change the order of summation once more to obtain that

$$\begin{split} \sum_{x \in V} |u(x)|^2 \mathbb{1}_{d_*(x,\tilde{A}) \le R} \\ & \le C_{\mathsf{d}} \sum_{i=1}^{\mathsf{d}+1} \sum_{x \in V} \mathbb{1}_{d_*(x,\tilde{A}) \le R} \\ & \sum_{y \in \Psi_x^{(i)}} \left(\mathbb{1}_{|y-x|_1 > \tilde{\alpha}_{\mathsf{d}}, d^{(i)}(y,\tilde{A}) > \tilde{\alpha}_{\mathsf{d}}} d^{(i)}(y,\tilde{A})^{\mathsf{d}-1} + \mathbb{1}_{|y-x|_1 \le \tilde{\alpha}_{\mathsf{d}}} R^{\mathsf{d}-1} + 1 \right) \left| \nabla_1^2 u(y) \right|^2 \\ & \le C_{\mathsf{d}} \sum_{i=1}^{\mathsf{d}+1} \sum_{y \in V + Q_R(0)} \left| \nabla_1^2 u(y) \right|^2 \\ & \sum_{x: \ y \in \Psi_x^{(i)}} \mathbb{1}_{d_*(x,\tilde{A}) \le R} \left(\mathbb{1}_{|y-x|_1 > \tilde{\alpha}_{\mathsf{d}}, d^{(i)}(y,\tilde{A}) > \tilde{\alpha}_{\mathsf{d}}} d^{(i)}(y,\tilde{A})^{\mathsf{d}-1} + \mathbb{1}_{|y-x|_1 \le \tilde{\alpha}_{\mathsf{d}}} R^{\mathsf{d}-1} + 1 \right) \end{split}$$

$$\leq C_{\mathsf{d}} \sum_{i=1}^{\mathsf{d}+1} \sum_{y \in V + Q_{R}(0)} \left| \nabla_{1}^{2} u(y) \right|^{2} \\ \left(\left| \left\{ x \colon y \in \Psi_{x}^{(i)}, |y - x|_{1} > \tilde{\alpha}_{\mathsf{d}}, d^{(i)}(y, \tilde{A}) > \tilde{\alpha}_{\mathsf{d}}, d_{*}(x, \tilde{A}) \leq R \right\} \right| d^{(i)}(y, \tilde{A})^{\mathsf{d}-1} \\ + \left| \left\{ x \colon y \in \Psi_{x}^{(i)}, |y - x|_{1} \leq \tilde{\alpha}_{\mathsf{d}} \right\} \right| R^{\mathsf{d}-1} + \left| \left\{ x \colon y \in \Psi_{x}^{(i)}, d_{*}(x, \tilde{A}) \leq R \right\} \right| \right).$$
(6.3.6)

The cardinality of the second set here is trivial to estimate and we find $|\{x: y \in \Psi_x^{(i)}, |y - x|_1 \leq \tilde{\alpha}_d\}| \leq C_d \tilde{\alpha}_d^d$. Similarly, $y \in \Psi_x^{(i)}$ and $d_*(x, \tilde{A}) \leq R$ imply $|x - y|_1 \leq R$, and so the cardinality of the third set can be estimated as $|\{x: y \in \Psi_x^{(i)}, d_*(x, \tilde{A}) \leq R\}| \leq C_d R^d$.

To estimate the cardinality of the first set we need to work a bit. The heuristic here is that the paths $\Psi_x^{(i)}$ are close to straight lines with the same endpoint passing through y, so there cannot be too many of them. To make this precise, fix y with $d^{(i)}(y, \tilde{A}) > \tilde{\alpha}_d$ and consider some x such that $y \in \Psi_x^{(i)}$, $|y - x|_1 > \tilde{\alpha}_d$ and $d_*(x, \tilde{A}) \leq R$. Because $|y - x|_1 > \tilde{\alpha}_d$ and $d^{(i)}(y, \tilde{A}) > \tilde{\alpha}_d$, we know that $a_y^{(i)} \in y + \Xi_i$ and $y \in x + \Xi_i$, and hence $a_y^{(i)} \subset x + \Xi_i$. Thus, $a_y^{(i)}$ is one of the candidates for the endpoint of the path $\Psi_x^{(i)}$, and our definition of the paths ensures that we actually have $a_x^{(i)} = a_y^{(i)}$. Because $y \in \Psi_x^{(i)}$, the point y has distance at most \sqrt{d} from the straight line connecting x and $a_x^{(i)} = a_y^{(i)}$. Therefore x is contained in some fixed cone with tip $a_y^{(i)}$ and opening angle $\leq \frac{C_d}{|a_y^{(i)} - y|_1} = \frac{C_d}{d^{(i)}(y,\tilde{A})}$. The point x is also contained in the cube around $a_y^{(i)}$ with diameter 2R, as otherwise $d_*(x, \tilde{A}) \leq d^{(i)}(x, \tilde{A}) = |x - a_y^{(i)}|_1 \geq |x - a_y^{(i)}|_{\infty} > R$. Thus, x is contained in the intersection of the aforementioned cone with that cube. This intersection contains at most $C_d R\left(\frac{R}{d^{(i)}(y,\tilde{A})}\right)^{d-1} = C_d \frac{R^d}{(d^{(i)}(y,\tilde{A}))^{d-1}}$.

$$\left|\left\{x\colon y\in\Psi_x^{(i)}, |y-x|_1>\tilde{\alpha}_{\mathsf{d}}, d^{(i)}(y,\tilde{A})>\tilde{\alpha}_{\mathsf{d}}, d_*(x,\tilde{A})\leq R\right\}\right|\leq C_{\mathsf{d}}\frac{R^{\mathsf{d}}}{(d^{(i)}(y,\tilde{A}))^{\mathsf{d}-1}}.$$

Returning now to (6.3.6), we find

$$\begin{split} &\sum_{x \in V} |u(x)|^2 \mathbb{1}_{d_*(x,\tilde{A}) \le R} \\ &\leq C_{\mathsf{d}} \sum_{i=1}^{\mathsf{d}+1} \sum_{y \in V + Q_R(0)} \left| \nabla_1^2 u(y) \right|^2 \left(\frac{R^{\mathsf{d}}}{d^{(i)}(y,\tilde{A})^{\mathsf{d}-1}} d^{(i)}(y,\tilde{A})^{\mathsf{d}-1} + \tilde{\alpha}_{\mathsf{d}} R^{\mathsf{d}-1} + R^{\mathsf{d}} \right) \\ &\leq C_{\mathsf{d}} R^{\mathsf{d}} \sum_{i=1}^{\mathsf{d}+1} \sum_{y \in V + Q_R(0)} \left| \nabla_1^2 u(y) \right|^2 \\ &\leq C_{\mathsf{d}} R^{\mathsf{d}} \sum_{y \in V + Q_R(0)} \left| \nabla_1^2 u(y) \right|^2 \,. \end{split}$$

This completes the proof in the case $d \ge 5$. The case d = 4 is very similar. The only difference is that in the estimate (6.3.5) we no longer have

$$\sum_{s=0}^{d^{(i)}(x,\tilde{A})-1} \frac{1}{(d^{(i)}(x,\tilde{A})-s)^{\mathsf{d}-3}} \le C_{\mathsf{d}}$$

but instead

$$\sum_{s=0}^{d^{(i)}(x,\tilde{A})-1} \frac{1}{d^{(i)}(x,\tilde{A})-s} \leq C \log(2+d^{(i)}(x,\tilde{A})) \leq C \log R.$$

This is the additional factor log *R* that appears on the right hand side in (6.3.1).

Later we will also use a probabilistic quenched version of this estimate.

Lemma 6.3.2. Let $d \ge 4$. There is an odd integer N_d with the following property: Let $\Lambda \Subset \mathbb{Z}^d$, and let $x \in \Lambda$, and $k \in \mathbb{N}$. Then if ε is sufficiently small (depending on d) there is an event $\Omega_{x,k}$ such that $\zeta^{\varepsilon}_{\Lambda}(\Omega_{x,k}) \ge 1 - \frac{1}{2^{k^d}}$ and such that whenever $A \in \Omega_{x,k}$ the following estimate holds: if $u: \mathbb{Z}^d \to \mathbb{R}$ is a function such that u = 0 on \tilde{A} , then

$$|u(x)| \le C_{\mathsf{d}} \frac{k^{\mathsf{d}/2} \left(1 + \mathbb{1}_{\mathsf{d}=4} \left((\log k)^{1/2} + |\log \varepsilon|^{3/4} \right) \right)}{\varepsilon^{1/2}} \|\nabla_1^2 u\|_{L^2(Q_{kN_{\mathsf{d}}\lambda_{\mathsf{mic}}}(x))}.$$
(6.3.7)

Proof. We want to apply Theorem 6.3.1 with $V = \{x\}$ and $R = kN\lambda_{\text{mic}}$. Then $R^{\mathsf{d}}(1 + \mathbb{1}_{\mathsf{d}=4}\log R) \leq C_{\mathsf{d},N} \frac{k^{\mathsf{d}}(1 + \mathbb{1}_{\mathsf{d}=4}(\log k + |\log \varepsilon|^{3/2})}{\varepsilon}$, and so (6.3.7) follows with the choice $N_{\mathsf{d}} = N$ provided that $d_*(x, \tilde{\mathcal{A}}) \leq kN\lambda_{\text{mic}}$. Thus, if we define

$$\Omega_{x,k} = \left\{ A \subset \Lambda \colon d_*(x,\tilde{A}) \le kN\lambda_{\rm mic} \right\}$$

it remains to choose *N* in such a way that we can show that $\zeta^{\varepsilon}_{\Lambda}(\Omega_{x,k}) \geq 1 - \frac{1}{2^k}$.

For an odd integer N, let $\Xi_{i,kN\lambda_{\text{mic}}}(x) = (x + \Xi_i) \cap Q_{kN\lambda_{\text{mic}}}(x)$. When $kN \ge N'_{\text{d}}$ for some dimensional constant N'_{d} (and so in particular when $N \ge N'_{\text{d}}$) the fraction of points in $Q_{kN\lambda_{\text{mic}}}(x)$ that are in $\Xi_{i,kN\lambda_{\text{mic}}}(x)$ is bounded below, i.e. we have $|\Xi_{i,kN\lambda_{\text{mic}}}(x)| \ge \frac{(kN)^d \lambda_{\text{mic}}^d}{C_{\text{d}}}$. On the other hand, $d_*(x, \tilde{A}) \le kN\lambda_{\text{mic}}$ holds if and only if all $\Xi_{i,kN\lambda_{\text{mic}}}(x)$ contain some point in \tilde{A} . Therefore, using Theorem 6.1.3 c) we see

$$\begin{split} 1 - \zeta_{\Lambda}^{\varepsilon}(\Omega_{x,k}) &= \zeta_{\Lambda}^{\varepsilon}(d_{*}(x,\tilde{\mathcal{A}}) > kN\lambda_{\mathrm{mic}}) \\ &\leq \zeta_{\Lambda}^{\varepsilon}(d_{*}(x,\mathcal{A}) > kN\lambda_{\mathrm{mic}}) \\ &= \zeta_{\Lambda}^{\varepsilon}\left(\exists i \in \{1,\ldots,\mathsf{d}+1\} \colon \mathcal{A} \cap \Xi_{i,kN\lambda_{\mathrm{mic}}}(x) = \varnothing\right) \\ &= \sum_{i=1}^{\mathsf{d}+1} \zeta_{\Lambda}^{\varepsilon}\left(\mathcal{A} \cap \Xi_{i,kN\lambda_{\mathrm{mic}}}(x) = \varnothing\right) \\ &\leq \sum_{i=1}^{\mathsf{d}+1} (1-p_{\mathsf{d},-})^{|\Xi_{i,kN\lambda_{\mathrm{mic}}}(x)|} \\ &\leq \sum_{i=1}^{\mathsf{d}+1} \exp\left(-p_{\mathsf{d},-}\frac{(kN)^{\mathsf{d}}\lambda_{\mathrm{mic}}^{\mathsf{d}}}{C_{\mathsf{d}}}\right). \end{split}$$

For any d \geq 4 we have $p_{d,-}\lambda_{\text{mic}}^{d} \geq \frac{1}{C_{d}}$, and so

$$\begin{split} 1 - \zeta_{\Lambda}^{\varepsilon}(\Omega_{x,k}) &\leq (\mathsf{d}+1)\exp\left(-\frac{(kN)^{\mathsf{d}}}{C_{\mathsf{d}}}\right) \\ &\leq \left((\mathsf{d}+1)\exp\left(-\frac{N^{\mathsf{d}}}{C_{\mathsf{d}}}\right)\right)^{k^{\mathsf{d}}} \end{split}$$

and it suffices to choose $N_d \ge N'_d$ in such a way that the right hand side is less than $\frac{1}{2^{k^d}}$ when $N \ge N_d$.

6.3.2 An interpolation inequality

Let $Q \subset \Lambda$ be a discrete cube of sidelength *R*. In the following section we will need to control $\|\nabla_1 u\|_{L^2(Q)}$ by terms involving only *u* and $\nabla_1^2 u$. Usually one would expect to do this by an interpolation inequality of the form

$$\|\nabla_1 u\|_{L^2(Q)}^2 \le C_{\mathsf{d}} \left(R^2 \|\nabla_1^2 u\|_{L^2(Q)}^2 + \frac{1}{R^2} \|u\|_{L^2(Q)}^2 \right)$$

where the factors $R^{\pm 2}$ are due to scaling. For our purposes this is not good enough, however, as we do not control *u* on all of *Q*. So it is crucial for us that a similar inequality still holds when we only control *u* on a large enough subset of *Q*. Indeed we have the following result.

Lemma 6.3.3. Let $d \in \mathbb{N}$. Let R be an odd integer and let $Q \subset \Lambda$ be a discrete cube of sidelength R (i.e. $Q = Q_{R/2}(x_*)$ for some $x_* \in \mathbb{Z}^d$), and assume $R \ge 12(\sqrt{d})^{d-1}\sqrt{d}$. Let $B \subset Q$ such that $|B| \ge \frac{1}{2}|Q|$. Let $u: \Lambda \to \mathbb{R}$. Then we have the estimate

$$\|\nabla_{1}u\|_{L^{2}(Q)}^{2} \leq C_{\mathsf{d}}\left(R^{2}\|\nabla_{1}^{2}u\|_{L^{2}(Q)}^{2} + \frac{1}{R^{2}}\|\mathbb{1}_{\in B}u\|_{L^{2}(Q)}^{2}\right).$$
(6.3.8)

Proof. By translation invariance we can assume that *Q* is centred at 0.

We first prove (6.3.8) with *u* replaced by an affine function *v*, where $v(x) = b \cdot x + a$ for some $a \in \mathbb{R}$, $b \in \mathbb{R}^d$. That is, we want to show

$$|b|^{2} \leq C \frac{1}{R^{\mathsf{d}+2}} \|\mathbb{1}_{\in B} v\|_{L^{2}(Q)}^{2}.$$
(6.3.9)

To see this, note first that we can assume $b \neq 0$ (else there is nothing to show). Let $\theta = \frac{b}{|b|} \in \mathbb{S}^{d-1}$. For $\lambda > 0$ consider the set $E = \left\{ x \in Q \colon \left| \theta \cdot x + \frac{a}{|b|} \right| \leq \lambda \right\}$. The set *E* is the intersection of *Q* with a slab of width 2λ , and so for each point $x \in E$ the cube $x + \left[-\frac{1}{2}, \frac{1}{2}\right]^d$ is contained in $\left[-\frac{R}{2}, \frac{R}{2}\right]^d$ intersected with a slab of width $2\lambda + \sqrt{d}$. Thus, we can estimate the number of points in *E* as

$$|E| \le (2\lambda + \sqrt{\mathsf{d}})(R\sqrt{\mathsf{d}})^{\mathsf{d}-1}.$$

If we assume $\lambda \ge \sqrt{d}$, we can bound this by

$$|E| \leq 3\lambda (R\sqrt{\mathsf{d}})^{\mathsf{d}-1}.$$

We want to pick $\lambda = \frac{R}{12(\sqrt{d})^{d-1}}$. This is possible, as $\frac{R}{12(\sqrt{d})^{d-1}} \ge \sqrt{d}$ by our assumption on R. Then for our choice of λ we see that $|E| \le \frac{1}{4}R^d = \frac{1}{4}|Q|$.

On the other hand, we know $|B| \ge \frac{1}{2}|Q|$, and therefore $|B \setminus E| \ge \frac{1}{4}|Q|$. Now for each $x \in B \setminus E \subset Q \setminus E$ we have $\left|\theta \cdot x + \frac{a}{|b|}\right| > \lambda$ and hence

$$|v(x)| = |b| \left| \theta \cdot x + \frac{a}{|b|} \right| \ge \lambda |b| \ge \frac{R|b|}{C_{\mathsf{d}}}$$

Summing the square of this estimate over all $x \in B \setminus E$, we see that

$$\sum_{x \in B \setminus E} |v(x)|^2 \ge \sum_{x \in B \setminus E} \frac{R^2 |b|^2}{C_{\mathsf{d}}} \ge \frac{1}{4} |Q| \frac{R^2 |b|^2}{C_{\mathsf{d}}}$$

which immediately implies (6.3.9).

Let now $(u)_Q := \frac{1}{|Q|} \sum_{x \in Q} u(x)$ and $(\nabla_1 u)_Q := \frac{1}{|Q|} \sum_{x \in Q} \nabla_1 u(x)$, and define $v(x) = (u)_Q + (\nabla_1 u)_Q \cdot x$. Then v is an affine function to which we will be able to apply (6.3.9), while u - v and $\nabla_1 (u - v)$ have average zero over Q, which allows using the discrete Poincaré inequality with zero mean. We can thus write

$$\begin{split} |\nabla_{1}u\|_{L^{2}(Q)}^{2} &\leq 2\|\nabla_{1}v\|_{L^{2}(Q)}^{2} + 2\|\nabla_{1}(u-v)\|_{L^{2}(Q)}^{2} \\ &\leq \frac{2}{R^{2}}\|\mathbbm{1}_{\cdot\in B}v\|_{L^{2}(Q)}^{2} + C_{\mathsf{d}}R^{2}\|\nabla_{1}^{2}(u-v)\|_{L^{2}(Q)}^{2} \\ &= \frac{2}{R^{2}}\|\mathbbm{1}_{\cdot\in B}v\|_{L^{2}(Q)}^{2} + C_{\mathsf{d}}R^{2}\|\nabla_{1}^{2}u\|_{L^{2}(Q)}^{2} \\ &\leq \frac{C_{\mathsf{d}}}{R^{2}}\|\mathbbm{1}_{\cdot\in B}(u-v)\|_{L^{2}(Q)}^{2} + \frac{C_{\mathsf{d}}}{R^{2}}\|\mathbbm{1}_{\cdot\in B}u\|_{L^{2}(Q)}^{2} + C_{\mathsf{d}}R^{2}\|\nabla_{1}^{2}u\|_{L^{2}(Q)}^{2} \\ &\leq \frac{C_{\mathsf{d}}}{R^{2}}\|u-v\|_{L^{2}(Q)}^{2} + \frac{C_{\mathsf{d}}}{R^{2}}\|\mathbbm{1}_{\cdot\in B}u\|_{L^{2}(Q)}^{2} + C_{\mathsf{d}}R^{2}\|\nabla_{1}^{2}u\|_{L^{2}(Q)}^{2} \\ &\leq \frac{C_{\mathsf{d}}}{R^{2}}R^{4}\|\nabla_{1}^{2}u\|_{L^{2}(Q)}^{2} + \frac{C_{\mathsf{d}}}{R^{2}}\|\mathbbm{1}_{\cdot\in B}u\|_{L^{2}(Q)}^{2} + C_{\mathsf{d}}R^{2}\|\nabla_{1}^{2}u\|_{L^{2}(Q)}^{2} \\ &\leq \frac{C_{\mathsf{d}}}{R^{2}}\|\mathbbm{1}_{\cdot\in B}u\|_{L^{2}(Q)}^{2} + C_{\mathsf{d}}R^{2}\|\nabla_{1}^{2}u\|_{L^{2}(Q)}^{2} . \end{split}$$

This is what we wanted to show.

6.4 Probabilistic decay of the L²-norm for biharmonic functions

In this section we will prove a decay estimate for the L^2 -norm of the Hessian of a discrete biharmonic function. This estimate does not hold for all realizations of A, but we prove that it holds for all but an exceptional set of realizations whose probability decays exponentially.

The precise result is the following. Recall that $\tilde{\mathcal{A}} = \mathcal{A} \cup (\mathbb{Z}^{\mathsf{d}} \setminus \Lambda)$.

Theorem 6.4.1. Let $d \ge 4$. There is an odd integer \hat{N}_d with the following property: Let $\Lambda \Subset \mathbb{Z}^d$, Let $U \in \mathcal{P}_{\hat{N}_d \lambda_{mac}}$ be a polymer consisting of $n = \frac{|U|}{\hat{N}_d^d \lambda_{mac}^d}$ boxes, and $k \in \mathbb{N}$. Then if ε is sufficiently small (depending on d only) there is an event $\Omega_{U,k}$ such that $\zeta_{\Lambda}^{\varepsilon}(\Omega_{U,k}) \ge 1 - \frac{n}{2^k}$, and such that whenever $A \in \Omega_{U,k}$ the following estimates hold:

a) If $u : \mathbb{Z}^d \to \mathbb{R}$ is a function such that u = 0 on $\tilde{A} \setminus U$ and $u\Delta_1^2 u = 0$ on $\mathbb{Z}^d \setminus U$, we have the *estimate*

$$\left\|\nabla_{1}^{2}u\right\|_{L^{2}(\mathbb{Z}^{\mathsf{d}}\setminus(U+Q_{2k\hat{N}_{\mathsf{d}}\lambda_{\max}}(0)))}^{2} \leq \frac{1}{2^{k}}\left\|\nabla_{1}^{2}u\right\|_{L^{2}((U+Q_{2k\hat{N}_{\mathsf{d}}\lambda_{\max}}(0))\setminus U)}^{2}.$$
(6.4.1)

b) If $u: \mathbb{Z}^{d} \to \mathbb{R}$ is a function such that u = 0 on $(U + Q_{2k\hat{N}_{d}\lambda_{mac}}(0)) \cap \tilde{A}$ and $u\Delta_{1}^{2}u = 0$ on $U + Q_{2k\hat{N}_{d}\lambda_{mac}}(0)$, we have the estimate

$$\left\|\nabla_{1}^{2}u\right\|_{L^{2}(U)}^{2} \leq \frac{1}{2^{k}}\left\|\nabla_{1}^{2}u\right\|_{L^{2}((U+Q_{2k\hat{N}_{\mathsf{d}}\lambda_{\max}}(0))\setminus U)}^{2}.$$
(6.4.2)

We have already outlined the strategy of the proof in Section 6.1.3. Namely, to prove (6.4.1) and (6.4.2) we want to iterate the Widman hole filler argument k times, that is we need to find k pairs U_j , U'_j on which we can apply it.

In order to make the Widman hole filler argument work, we need to be able to apply Theorem 6.3.1 and Lemma 6.3.3. We can ensure this by finding a cut-off function η_j that grows from 0 to 1 in such a way that $\nabla_1^2 \eta_j = 0$ on those points on which Theorem 6.3.1 or Lemma 6.3.3 cannot be applied.

For that purpose pick an odd integer *N* to be fixed later and decompose \mathbb{Z}^d into the boxes in $\mathcal{Q}_{N\lambda_{mac}}$. We will declare some of these boxes to be bad in such a way that on the good (i.e. non-bad) boxes we can construct an η growing from 0 to 1 on that box and satisfying the conditions on $\nabla_1^2 \eta$. If we can show that bad boxes are rare, then with high probability we can find at least *k* annuli consisting only of good boxes inbetween *U* and $\mathbb{Z}^d \setminus (U + Q_{2kN\lambda_{mac}}(0))$, and then we can iterate the Widman hole filler argument on these annuli.

In Section 6.4.1 we describe in detail how we choose the bad boxes, and we prove that on the good boxes there exist cut-off functions as required. In Section 6.4.2 we carry out the Widman hole filler argument provided all relevant boxes are good. Finally, in Section 6.4.3 we show that the bad boxes are sparse enough that with sufficiently high probability we can find enough annuli to use the hole filler argument on. Using this result we will complete the proof of Theorem 6.4.1.

6.4.1 Bad boxes and cut-off functions

The definition of the bad boxes depends on three odd integers K, L, M, where K is always a multiple of 3, and $M \ge 12$. Eventually (in Section 6.4.3), we will choose them large enough in the order M, K, L to close the argument. For now we will track all dependencies on K, L, M. The parameters K and L will play similar roles as in the proof of Theorem 6.1.3 c), albeit not quite the same. We hope this will not confuse the reader.

There will be two reasons that lead to a cube $Q \in Q_{KL\lambda_{mac}}$ being bad. The first one is related to Lemma 6.3.3. We want to be able to apply that Lemma on each subcube $Q' \in Q_{K\lambda_{mac}}$ such that $Q' \subset Q$ with $\mathbb{1}_{\cdot \in B}$ being the indicator function from Theorem 6.3.1. So we will define Q to be bad of type II if there is a subcube $Q' \subset Q$ on which the indicator function from Theorem 6.3.1 is equal to 0 too often.

The second reason is more complicated. We want to be able to modify an initial guess η_* for the cut-off function in such a way that $\nabla_1^2 \eta = 0$ on those sets on which the indicator function from Theorem 6.3.1 is equal to 0 (and we think of those points as bad as well). This is easy if the bad points are very isolated and sparse, as we then can make local adjustments to η_* that do not interfere with each other. So we start at scale $\ell_0 = K\lambda_{\text{mic}}$ and consider the bad points (or actually the bad boxes in $Q_{K\lambda_{\text{mic}}}$ that contain at least a bad point) and split them into the isolated and the clustered ones. The former ones we ignore for the moment, and the latter ones can be covered by cubes in $Q_{\ell_1}^{\#}$ such that each cube covers at least two of the small cubes. These are the bad cubes on scale ℓ_1 . Now we can once again split those cubes into the isolated ones (that we ignore for the moment) and those that are clustered and can be covered by cubes are isolated. Then we can adjust η_* on those isolated cubes, and then proceed backwards and apply our adjustment also on the isolated cubes on the smaller scales. We thus call Q bad of type I if this process terminates too late.

We have not mentioned yet how to choose the scales ℓ_j . There is a trade-off here: on the one hand, the lengthscale should grow fast so that we have enough space around each isolated bad cube on scale ℓ_j to adjust η_* there. On the other hand we want many of the cubes to be isolated, so that our process terminates soon, and this we can achieve by letting

 ℓ_i not grow too fast. It turns out that a good compromise is

$$\ell_j := M^{j^3} K \lambda_{\rm mic}$$

for each $j \ge 0$. We give a more detailed explanation of this choice later, in Remark 6.4.9.

Let us now give the precise definitions. Recall that $Q_l = \{Q_{l/2}(x) : x \in (l\mathbb{Z})^d\}$ and $Q_l^{\#} = \{Q_{l/2}(x) : x \in (\frac{1}{3}\mathbb{Z})^d\}$. We consider the microscopic cubes in $Q_{K\lambda_{mic}}$, the macroscopic cubes in $Q_{KL\lambda_{mac}}$, and the further sets of boxes Q_{ℓ_0} and $Q_{\ell_j}^{\#}$ for $j \ge 1$, associated to $\ell_j = M^{j^3}K\lambda_{mic}$ for $j \ge 0$. The reason that we use $Q_{\ell_j}^{\#}$ and not Q_{ℓ_j} for $j \ge 1$ is that we want to ensure that for any two cubes on lengthscale ℓ_{j-1} that are sufficiently close there is a cube in $Q_{\ell_j}^{\#}$ that contains both of them. We assumed $M \ge 12$, and so $\ell_{j-1} \le \frac{1}{12}\ell_j$ for each $j \ge 1$. We fix for later use an enumeration of the boxes in each $Q_{\ell_j}^{\#}$.

We define

$$S_{K,M,\text{bad}}^{(0)}(A) = \left\{ Q \in \mathcal{Q}_{K\lambda_{\text{mic}}} \colon \exists x \in Q \text{ with } d_*(x,\tilde{A}) > K\lambda_{\text{mic}} \right\}.$$

Note that $S_{K,M,\text{bad}}^{(0)}(A) \subset \mathcal{Q}_{K\lambda_{\text{mic}}} = \mathcal{Q}_{\ell_0} \subset \mathcal{Q}_{\ell_0}^{\#}$. For *M* large enough the set $S_{K,M,\text{bad}}^{(0)}(A)$ is finite (as cubes far outside of Λ will not be bad).

For $j \geq 1$ we define $S_{K,M,\text{bad}}^{(j)}(A) \subset \mathcal{Q}_{\ell_j}^{\#}$ inductively as follows: Given $S_{K,M,\text{bad}}^{(j-1)}(A) \subset \mathcal{Q}_{\ell_{j-1}}^{\#}$ such that the set $S_{K,M,\text{bad}}^{(j-1)}(A)$ is finite, we want to split it into two sets of boxes: $S_{K,M,\text{bad,clust}}^{(j-1)}(A)$ will contain those boxes that are clustered in the sense that there is another bad box at distance $\leq \frac{\ell_j}{2}$ that is disjoint from the original box, and $S_{K,M,\text{bad,isol}}^{(j-1)}(A)$ will contain the other boxes are isolated in the sense that all bad boxes that are disjoint from them are far away. Let us make this precise: we define

$$S_{K,M,\text{bad},\text{clust}}^{(j-1)}(A) = \left\{ Q \in S_{K,M,\text{bad}}^{(j-1)}(A) \colon \exists Q' \in S_{K,M,\text{bad}}^{(j-1)}(A) \text{ with } Q' \cap Q = \emptyset, d_{\infty}(Q,Q') \le \frac{\ell_j}{2} \right\},\$$

$$S_{K,M,\text{bad},\text{isol}}^{(j-1)}(A) = S_{K,M,\text{bad}}^{(j-1)}(A) \setminus S_{K,M,\text{bad},\text{clust}}^{(j-1)}(A).$$

If $S_{K,M,\text{bad,clust}}^{(j-1)}(A) = \emptyset$, we define $S_{K,M,\text{bad}}^{(j)}(A) = \emptyset$. Otherwise, if $Q \in S_{K,M,\text{bad,clust}}^{(j-1)}(A)$ and $Q' \in S_{K,M,\text{bad}}^{(j-1)}(A)$ is a witness for this in the sense that $Q' \cap Q = \emptyset$ and $d_{\infty}(Q,Q') \leq \frac{\ell_j}{2}$ then also $Q' \in S_{K,M,\text{bad,clust}}^{(j-1)}(A)$, as Q then is a witness for Q'. In particular,

$$S_{K,M,\text{bad,clust}}^{(j-1)}(A) = \left\{ Q \in S_{K,M,\text{bad}}^{(j-1)}(A) \colon \exists Q' \in S_{K,M,\text{bad,clust}}^{(j-1)}(A) \text{ with } Q' \cap Q = \varnothing, d_{\infty}(Q,Q') \le \frac{\ell_j}{2} \right\} \,.$$

Furthermore, if $Q, Q' \in S_{K,M,\text{bad,clust}}^{(j-1)}(A)$ with $Q' \cap Q = \emptyset$ and $d_{\infty}(Q,Q') \leq \frac{\ell_j}{2}$ then Q and Q' are contained in a common box of sidelength at most $\frac{\ell_j}{2} + 2\ell_{j-1} \leq \frac{2\ell_j}{3}$ (as $\ell_{j-1} \leq \frac{1}{12}\ell_j$), and so there is a cube in $Q_{\ell_j}^{\#}$ containing both of them. This means that we can cover $S_{K,M,\text{bad,clust}}^{(j-1)}(A)$ by finitely many cubes from $Q_{\ell_i}^{\#}$ in such a way that each cube from the cover contains at least

two small cubes from $S_{K,M,\text{bad,clust}}^{(j-1)}(A)$ that are disjoint. We can consider the set of subsets of $\mathcal{Q}_{\ell_i}^{\#}$ with that property,

$$\begin{aligned} \mathcal{S}_{K,M,\mathrm{bad}}^{(j)}(A) &= \left\{ S \subset \mathcal{Q}_{\ell_j}^{\#} \colon \bigcup_{\substack{Q' \in S_{K,M,\mathrm{bad,clust}}^{(j-1)}(A)}} Q' \subset \bigcup_{\substack{Q \in S}} Q, \\ \forall Q \in S \; \exists Q', Q'' \in S_{K,M,\mathrm{bad,clust}}^{(j-1)}(A) \text{ with } Q' \cap Q'' = \emptyset, Q' \cup Q'' \subset Q \right\}. \end{aligned}$$

The previous discussion implies that if $S_{K,M,\text{bad},\text{clust}}^{(j-1)}(A)$ is non-empty, then also $\mathcal{S}_{K,M,\text{bad}}^{(j)}(A)$ is non-empty. Now we consider the elements of $\mathcal{S}_{K,M,\text{bad}}^{(j)}(A)$ of minimum cardinality, and among those we define $S_{K,M,\text{bad}}^{(j)}(A)$ to be that element that is lexicographically first (with respect to the enumeration of $\mathcal{Q}_{\ell_j}^{\#}$ that we had fixed).

To summarize, $S_{K,M,\text{bad}}^{(j)}(A)$ is a finite subset of $\mathcal{Q}_{\ell_j}^{\#}$ that covers $S_{K,M,\text{bad,clust}}^{(j-1)}(A)$ in such a way that each of its boxes contains two disjoint elements from $S_{K,M,\text{bad,clust}}^{(j-1)}(A)$.

Finally we can define macroscopic bad boxes. Let $j_*(\varepsilon)$ be the largest integer j such that $\ell_j \leq \frac{KL\lambda_{\text{mac}}}{8}$ (we assume that ε is small enough so that $j_*(\varepsilon) \geq 1$). We consider the macroscopic boxes in $\mathcal{Q}_{KL\lambda_{\text{mac}}}$, and call a macroscopic box bad of type I if it contains at least one box in $S_{K,M,\text{bad}}^{(j_*(\varepsilon))}(A)$, bad of type II if one of its $K\lambda_{\text{mac}}$ -subboxes contains many boxes in $S_{K,M,\text{bad}}^{(0)}(A)$, and bad if it is bad of type I or type II. More precisely

$$S_{K,L,M,\text{bad}}^{*,I}(A) = \left\{ Q \in \mathcal{Q}_{KL\lambda_{\text{mac}}} : \exists q \in S_{K,M,\text{bad}}^{(j_*(\varepsilon))}(A) \text{ with } q \cap Q \neq \emptyset \right\},$$

$$S_{K,L,M,\text{bad}}^{*,II}(A) = \left\{ Q \in \mathcal{Q}_{KL\lambda_{\text{mac}}} : \exists Q' \in \mathcal{Q}_{K\lambda_{\text{mac}}} \text{ with } Q' \subset Q,$$

$$\left| \left\{ q \in S_{K,M,\text{bad}}^{(0)}(A) : q \subset Q' \right\} \right| \ge \frac{1}{4} \left(\frac{\lambda_{\text{mac}}}{\lambda_{\text{mic}}} \right)^{\mathsf{d}} \right\},$$

$$S_{K,L,M,\text{bad}}^{*}(A) = S_{K,L,M,\text{bad}}^{*,II}(A) \cup S_{K,L,M,\text{bad}}^{*,II}(A).$$

The point of these definitions is that we can apply our construction of a cut-off function on all cubes in $Q_{KL\lambda_{mac}} \setminus S_{K,L,M,bad}^{*,I}(A)$, while we can apply Lemma 6.3.3 on every $K\lambda_{mac}$ subcube of the cubes in $Q_{KL\lambda_{mac}} \setminus S_{K,L,M,bad}^{*,II}(A)$. Of course, this is only useful if we show that bad cubes are rare. This will be established in Section 6.4.3. For now we show that our definition of good boxes fulfils its purpose in the sense that we can construct a cut-off function growing from 0 to 1 on them in such a way that their second derivatives are 0 on the set of microscopic bad boxes.

Lemma 6.4.2. Let $d \ge 1$. Then there is a constant $M_d \ge 12$ with the following property: Let K, L, M be odd integers such that K is a multiple of 3, and $M \ge M_d$. Then for all ε sufficiently small (depending on d, K) the following holds: Let $U \in \mathcal{P}_{KL\lambda_{mac}}$ be a polymer. Suppose that none of the $KL\lambda_{mac}$ -boxes touching its boundary (in the l^{∞} -sense) are bad of type I, i.e.

$$\{Q \in \mathcal{Q}_{KL\lambda_{\max}} \colon Q \subset (U + Q_{KL\lambda_{\max}}(0)) \setminus U\} \cap S^{*,l}_{K,L,M,\mathsf{bad}}(A) = \emptyset$$

Then there is a function $\eta \colon \mathbb{Z}^d \to \mathbb{R}$ *such that*

i) $\eta(x) = 0$ for $x \in U + Q_{2K\lambda_{\text{mic}}}(0)$,

- *ii)* $\eta(x) = 1$ for $x \in \mathbb{Z}^{\mathsf{d}} \setminus (U + Q_{KL\lambda_{\max}-2K\lambda_{\min}}(0))$,
- *iii*) $|\nabla_1^2 \eta(x)| \le \frac{C_d}{(KL)^2 \lambda_{\max}^2} \mathbb{1}_{Q_1(x)} \cup_{Q \in S_{K,M,\text{bad}}^{(0)}} Q = \emptyset$

Here the constant C_d *depends only on* d.

Morally, property iii) is $|\nabla_1^2 \eta(x)| \leq \frac{C_d}{(KL)^2 \lambda_{mac}^2} \mathbb{1}_{x \notin \bigcup_{Q \in S_{K,M,bad}^{(0)}} Q}$. However, the discrete product rule lets translation operators arise, and this is why we require the slightly stronger condition iii) above.

To prove Lemma 6.4.2, we will begin with a function η_* which satisfies i) and ii), but only $|\nabla_1^2 \eta(x)| \leq \frac{C_d}{(KL)^2 \lambda_{\text{mac}}^2}$ instead of iii). Then we will modify η_* iteratively to make it affine on larger and larger subsets of $S_{K,M,\text{bad}}^{(0)}(A)$, so that eventually iii) is satisfied as well. The following lemma gives details on how to carry out a single of these modification steps.

Lemma 6.4.3. Let $d \ge 1$. There is a constant $\gamma_d > 0$ with the following property: Let $x \in \mathbb{Z}^d$, let r, R be positive integers such that $R \ge 16r$. Let $v \colon \mathbb{Z}^d \to \mathbb{R}$ be a function. Then there is a function $w \colon \mathbb{Z}^d \to \mathbb{R}$ with the following properties:

- *i*) w = 0 on $\mathbb{Z}^d \setminus Q_{R-1}(x)$,
- *ii*) $\nabla_1^2(v+w) = 0$ on $Q_r(x)$,
- *iii*) $\|\nabla_1^2(v+w)\|_{L^{\infty}(Q_R(x))} \le \left(1 + \frac{\gamma_d}{\log R \log r}\right) \|\nabla_1^2 v\|_{L^{\infty}(Q_R(x))}.$

Note that condition i) ensures that $\nabla_1^2 w = 0$ on $\mathbb{Z}^d \setminus Q_R(x)$.

Proof. By translation invariance we can assume x = 0. Suppose for the moment that there is a function $\xi \colon \mathbb{Z}^d \to \mathbb{R}$ such that $\xi = 1$ on $Q_{r+1}(0)$, $\xi = 0$ on $\mathbb{Z}^d \setminus Q_{R-1}(0)$ and

$$\begin{aligned} |\xi(y)| &\leq 1, \\ |\nabla_1 \xi(y)| &\leq \frac{C_{\mathsf{d}}}{|y|(\log R - \log r)}, \\ |\nabla_1^2 \xi(y)| &\leq \frac{C_{\mathsf{d}}}{|y|^2 (\log R - \log r)} \end{aligned}$$

for $y \in Q_R(0)$.

Let *u* be the affine function $u(y) = v(0) + y \cdot \nabla_1 v(0)$. Then we can set $w(y) = \xi(y)(u(y) - v(y))$. This choice of *w* clearly satisfies i) and ii), and so we only have to check iii). We know that u(0) - v(0) = 0 and $\nabla_1 u(0) - \nabla_1 v(0) = 0$, and so by discrete Taylor expansion, using that *u* is affine, we have

$$\begin{aligned} |\nabla_1 u(y) - \nabla_1 v(y)| &\leq C_{\mathsf{d}} |y| \|\nabla_1^2 v\|_{L^{\infty}(Q_R(0))} \\ |u(y) - v(y)| &\leq C_{\mathsf{d}} |y|^2 \|\nabla_1^2 v\|_{L^{\infty}(Q_R(0))} \end{aligned}$$

Note that $v + w = \xi u + (1 - \xi)v$. The discrete product rule allows us to write

$$D_{i}^{1}D_{-j}^{1}(v+w)(y) = D_{i}^{1}D_{-j}^{1}v(y)\tau_{i}^{1}\tau_{-j}^{1}(1-\xi)(y) + D_{i}^{1}(u-v)(y)\tau_{i}^{1}D_{-j}^{1}\xi(y) + D_{-j}^{1}(u-v)(y)\tau_{-j}^{1}D_{i}^{1}\xi(y) + (u-v)(y)D_{i}^{1}D_{-j}^{1}\xi(y)$$

and thus, using the estimates on ξ and v - u,

$$\begin{split} |D_{i}^{1}D_{-j}^{1}(v+w)(y)| &\leq |D_{i}^{1}D_{-j}^{1}v(y)| + |D_{i}^{1}(u-v)(y)| \max_{|z-y|_{\infty} \leq 1} |\nabla_{1}\xi(z)| \\ &+ |D_{-j}^{1}(u-v)(y)| \max_{|z-y|_{\infty} \leq 1} |\nabla_{1}\xi(z)| + |(u-v)(y)| \max_{|z-y|_{\infty} \leq 1} |\nabla_{1}^{2}\xi(z)| \\ &\leq |D_{i}^{1}D_{-j}^{1}v(y)| + C_{\mathsf{d}}|y| \|\nabla_{1}^{2}v\|_{L^{\infty}(Q_{R}(0))} \frac{1}{|y|(\log R - \log r)} \\ &+ C_{\mathsf{d}}|y|^{2} \|\nabla_{1}^{2}v\|_{L^{\infty}(Q_{R}(0))} \frac{1}{|y|^{2}(\log R - \log r)} \\ &\leq |D_{i}^{1}D_{-j}^{1}v(y)| + C_{\mathsf{d}} \|\nabla_{1}^{2}v\|_{L^{\infty}(Q_{R}(0))} \frac{1}{(\log R - \log r)} \,. \end{split}$$

This immediately implies that *v* satisfies iii).

It remains to show the existence of ξ with the desired properties. To do so, we choose a function $\chi \in C^{\infty}(\mathbb{R}^d)$ that is 1 on $[-1, 1]^d$, 0 outside of $[-2, 2]^d$ such that $0 \le \chi \le 1$, and for $\rho > 0$ define $\chi_{\rho} = \chi\left(\frac{\cdot}{\rho}\right)$. We define $\tilde{\xi} \colon \mathbb{R}^d \to \mathbb{R}$ by

$$\tilde{\xi}(y) = \chi_{2r}(y) + (\chi_{R/4}(y) - \chi_{2r}(y)) \frac{\log R - \log |y|}{\log R - \log r}$$

One can check that $0 \leq \tilde{\xi} \leq 1$, $\tilde{\xi} = 1$ on $[-2r, 2r]^{\mathsf{d}} \supset Q_{r+1}(0)$, $\tilde{\xi} = 0$ on $\mathbb{R}^{\mathsf{d}} \setminus \left[-\frac{R}{2}, \frac{R}{2}\right]^{\mathsf{d}} \supset \mathbb{Z}^{\mathsf{d}} \setminus Q_{R-1}R(0)$ as well as

$$|\nabla^k \xi(y)| \le \frac{C_{\mathsf{d},k}}{|y|^k (\log R - \log r)}$$

for $k \ge 1$. We can now let ξ be the restriction of $\tilde{\xi}$ to \mathbb{Z}^d . The estimates on $\tilde{\xi}$ together with Taylor's theorem then imply the corresponding estimates on ξ .

Before we turn to the proof of Lemma 6.4.2, let us investigate the structure of the $S_{K,M,\text{bad}}^{(j)}(A)$ in more detail. Note first that

$$\left|S_{K,M,\text{bad}}^{(j)}(A)\right| \ge \left|S_{K,M,\text{bad,clust}}^{(j)}(A)\right| \ge \left|S_{K,M,\text{bad}}^{(j+1)}(A)\right| + 1$$

as each cube of $S_{K,M,\text{bad}}^{(j+1)}(A)$ covers at least two cubes of $S_{K,M,\text{bad},\text{clust}}^{(j)}(A)$, and $S_{K,M,\text{bad}}^{(j+1)}(A)$ is chosen with the smallest possible cardinality. We have already noted that $S_{K,M,\text{bad}}^{(j)}(A)$ is a finite set. This implies that $S_{K,M,\text{bad}}^{(j)}(A) = \emptyset$ for j sufficiently large.

If $Q \in S_{K,M,\text{bad}}^{(j)}(A)$ for some $j \ge 0$, then either $Q \in S_{K,M,\text{bad},\text{isol}}^{(j)}(A)$ or $Q \in S_{K,M,\text{bad},\text{clust}}^{(j)}(A)$. In the latter case there is at least one $Q' \in S_{K,M,\text{bad}}^{(j+1)}(A)$ such that $Q \subset Q'$. If there is more than one such Q', we choose the one that comes first with respect to the enumeration of $Q_{\ell_j}^{\#}$ that we had fixed, and call it the parent of Q. In this manner, given $Q \in S_{K,M,\text{bad}}^{(0)}(A)$ we can find a sequence

$$Q \subset Q^{(1)} \subset \ldots \subset Q^{(j)}$$

such that each cube is the parent of the preceding cube. This sequence is necessarily finite as $S_{K,M,\text{bad}}^{(j)}(A) = \emptyset$ for *j* sufficiently large. It terminates as soon as we reach a cube $Q^{(j)} \in S_{K,M,\text{bad},\text{isol}}^{(j)}(A)$. We denote that value *j* by $j_{\text{isol},Q}$. In summary, we have a sequence

$$Q = Q^{(0)} \subset Q^{(1)} \subset \ldots \subset Q^{(j_{\text{isol},Q})}$$
(6.4.3)

where each cube is the parent of the preceding cube, the cubes $Q^{(j)}$ for $j < j_{isol,Q}$ are in $S_{K,M,bad,clust}^{(j)}(A)$, while $Q^{(j_{isol,Q})} \in S_{K,M,bad,isol}^{(j_{isol,Q})}(A)$. This allows us to find for each cube $Q \in S_{K,M,bad}^{(0)}(A)$ a lengthscale $\ell_{j_{isol,Q}}$ on which its parents become isolated. It will be that lengthscale on which we will ensure that η is locally affine on Q.

After these preparations we can turn to the proof of the main result of this section, the construction of a cut-off function.

Proof of Lemma 6.4.2.

Step 1: Construction of a function satisfying a weaker version of iii)

We assume that ε is small enough so that $\lambda_{\text{mic}} \geq 4$, say. Then also $\ell_j \geq 4$ for all $j \geq 0$. We first claim that there is a function η_* satisfying i) and ii) and such that $|\nabla_1^2 \eta_*(x)| \leq \frac{C_d}{(KL)^2 \lambda_{\text{mac}}^2}$. This should be intuitively clear, as we want to interpolate from 0 to 1 on scale $KL\lambda_{\text{mac}}$. One way to make this rigorous is as follows. Let $\tilde{U} = U + \left[-\frac{1}{2}, \frac{1}{2}\right]^d$. Choose a function $\hat{\chi} \in C^{\infty}(\mathbb{R}^d)$ that is 1 on $[-1, 1]^d$, 0 outside of $\left[-\frac{9}{7}, \frac{9}{7}\right]^d$ and such that $0 \leq \chi \leq 1$. For each $Q \in Q_{KL\lambda_{\text{mac}}}$ such that $Q \subset U$ let x_Q be its centre, and define $\tilde{\eta}_* : \mathbb{R}^d \to \mathbb{R}$ by

$$\tilde{\eta}_*(y) = \prod_{\substack{Q \in \mathcal{Q}_{KL\lambda_{\max}}\\Q \subset U}} \left(1 - \hat{\chi} \left(\frac{y - x_Q}{\frac{7}{8}KL\lambda_{\max}} \right) \right) \,.$$

This function is then equal to 0 on $\tilde{U} + \left[-\frac{3KL\lambda_{\text{mac}}}{8}, \frac{3KL\lambda_{\text{mac}}}{8} \right]^{\mathsf{d}}$ and also equal to 1 on $\mathbb{R}^{\mathsf{d}} \setminus \left(\tilde{U} + \left[-\frac{5KL\lambda_{\text{mac}}}{8}, \frac{5KL\lambda_{\text{mac}}}{8} \right]^{\mathsf{d}} \right)$. Each of the factors in the definition of $\tilde{\eta}_*$ satisfies $\left\| \nabla^k \left(1 - \hat{\chi} \left(\frac{\cdot - x_Q}{\frac{7}{8}KL\lambda_{\text{mac}}} \right) \right) \right\|_{L^{\infty}(\mathbb{R}^{\mathsf{d}})} \leq \frac{C_{\mathsf{d},k}}{(KL)^k \lambda_{\text{mac}}^k}$

for $k \ge 0$. Furthermore, for each fixed $y \in \mathbb{R}^d$ there is a neighbourhood in which at most 3^d of the factors are non-constant. Thus, if we compute $\nabla^k \tilde{\eta}_*(y)$ we get contributions only from these at most 3^d factors. Therefore,

$$\|\nabla^k \tilde{\eta}_*\|_{L^{\infty}(\mathbb{R}^d)} \le \frac{C_{\mathsf{d},k}}{(KL)^k \lambda_{\max}^k}$$

for $k \ge 0$. We can now let η_* be the restriction of $\tilde{\eta}_*$ to \mathbb{Z}^d . Then Taylor's theorem implies easily that $|\nabla_1^2 \eta(x)| \le \frac{C_d}{(KL)^2 \lambda_{\text{max}}^2}$.

Note also that η_* is equal to 0 on $U + \left[-\frac{3KL\lambda_{\text{mac}}}{8}, \frac{3KL\lambda_{\text{mac}}}{8}\right]$ and equal to 1 on $\mathbb{Z}^d \setminus (U + Q_{5KL\lambda_{\text{mac}}/8}(0))$. Therefore, $\nabla_1^2 \eta_*$ is equal to 0 except possibly on $V := (U + Q_{5KL\lambda_{\text{mac}}/8+1}(0) \setminus (U + Q_{3KL\lambda_{\text{mac}}/8-1}(0))$.

Step 2: Modification of η_*

Let $q \in S_{K,M,\text{bad}}^{(0)}(A)$ with $q \subset (U + Q_{KL\lambda_{\text{mac}}}(0)) \setminus U$, and consider the sequence of cubes (6.4.3) with Q = q. By our assumption none of the macroscopic cubes in $(U + Q_{KL\lambda_{\text{mac}}}(0)) \setminus U$ are bad of type I, i.e. there is no $Q' \in S_{K,M,\text{bad}}^{(j_*(\varepsilon))}(A)$ intersecting $(U + Q_{KL\lambda_{\text{mac}}}(0)) \setminus U$. This means that the sequence (6.4.3) necessarily terminates before $j = j_*(\varepsilon)$. In particular, we have $j_{\text{isol},q} < j_*(\varepsilon)$. We can now partition the cubes in $S_{K,M,\text{bad}}^{(0)}(A)$ according to the value of $j_{\text{isol},q}$, and define for $j \in \{0, \ldots, j_*(\varepsilon) - 1\}$ the set

$$T_{K,M,j}(A) = \left\{ q \in S_{K,M,\mathsf{bad}}^{(0)}(A) \colon q \subset (U + Q_{KL\lambda_{\mathsf{mac}}}(0)) \setminus U, j_{\mathsf{isol},q} = j \right\} \,.$$

We will now construct a sequence of functions $\eta^{(j)}$ by reverse induction in such a way that $\nabla_1^2 \eta^{(j)} = 0$ on $\bigcup_{k \ge j} \bigcup_{q \in T_{K,M,k}(A)} (q + Q_1(0))$. Eventually, we will show that the choice $\eta = \eta^{(0)}$ satisfies the properties claimed in the Lemma.

Thus, we start with $\eta^{(j_*(\varepsilon))} = \eta_*$. Let also $V^{(j_*(\varepsilon))} = V$, and note that $\operatorname{supp} \nabla_1^2 \eta^{(j_*(\varepsilon))}$ is a subset of $V^{(j_*(\varepsilon))} = V$. Suppose now that for some $j \in \{1, \ldots, j_*(\varepsilon)\}$ we have defined $\eta^{(j)}$ and $V^{(j)}$ with $\operatorname{supp} \nabla_1^2 \eta^{(j)} \subset V^{(j)}$ and $\eta^{(j)} = \eta_*$ on $\mathbb{Z}^d \setminus V^{(j)}$ such that $\nabla_1^2 \eta^{(j)} = 0$ on $\bigcup_{k \ge j} \bigcup_{q \in T_{K,M,k}(A)} (q + Q_1(0))$, and let us define $\eta^{(j-1)}$ and $V^{(j-1)}$.

Since supp $\nabla_1^2 \eta^{(j)} \subset V^{(j)}$, we trivially have $\nabla_1^2 \eta^{(j)} = 0$ on those cubes that do not intersect $V^{(j)}$, and so there is no need to change $\eta^{(j)}$ there. Let

$$Y_{K,M,V}^{(j-1)}(A) = \left\{ Q \in S_{K,M,\text{bad,isol}}^{(j-1)}(A) \colon (Q + Q_1(0)) \cap V^{(j)} \neq \emptyset \right\}$$

be the set of cubes where we will adjust $\eta^{(j)}$. By definition, this is a set of cubes on scale ℓ_{j-1} that either overlap or are far apart. That is, if $Q \in Y_{K,M,V}^{(j-1)}(A)$, then all $Q' \in S_{K,M,bad}^{(j-1)}(A)$ either satisfy $Q \cap Q' \neq \emptyset$ or $d_{\infty}(Q,Q') > \frac{\ell_j}{2}$. Let $\tilde{Y}_{K,M,V}^{(j-1)}(A)$ be a subset of $Y_{K,M,V}^{(j-1)}(A)$ of maximum cardinality such that the cubes in $\tilde{Y}_{K,M,V}^{(j-1)}(A)$ are pairwise disjoint. By construction for each $Q \in Y_{K,M,V}^{(j-1)}(A)$ there is a $Q' \in \tilde{Y}_{K,M,V}^{(j-1)}(A)$ (possibly equal to Q) with $Q \cap Q' \neq \emptyset$. In particular,

$$igcup_{Q\in Y^{(j-1)}_{K,M,V}(A)}Q\subset igcup_{Q\in ilde Y^{(j-1)}_{K,M,V}(A)}(Q+Q_{\ell_{j-1}}(0))\,.$$

Let now $Q \in \tilde{Y}_{K,M,V}^{(j-1)}(A)$. We know that there is no cube $Q' \in S_{K,M,\text{bad}}^{(j-1)}(A)$ that intersects $(Q + Q_{\ell_j/2}(0)) \setminus (Q + Q_{\ell_{j-1}}(0))$. This has several implications. An obvious one is that the cubes $Q + Q_{\ell_j/4}(0)$ for $Q \in \tilde{Y}_{K,M,V}^{(j-1)}(A)$ are pairwise disjoint. Slightly less obviously, we claim that for $Q \in \tilde{Y}_{K,M,V}^{(j-1)}(A)$ there is no $q \in \bigcup_{k \ge j} T_{K,M,k}(A)$ such that $q \cap ((Q + Q_{\ell_j/4+1}(0)) \setminus (Q + Q_{\ell_{j-1}}(0))) \neq \emptyset$. Indeed, if $q \in \bigcup_{k \ge j} T_{K,M,k}(A)$ then we have $j_{isol,q} \ge j$, so the sequence (6.4.3) contains a parent $q \subset q^{(j-1)} \in S_{K,M,\text{bad}}^{(j-1)}(A)$. Then $q^{(j-1)}$ cannot intersect $(Q + Q_{\ell_j/2}(0)) \setminus (Q + Q_{\ell_{j-1}}(0))$, and so the same holds true for q.

We would now like to apply Lemma 6.4.3 to the function $v = \eta^{(j)}$ and the cubes $Q + Q_{\ell_{j-1}+1}(0) \subset Q + Q_{\ell_j/4}(0)$. For that purpose we fix $M_d = 8 \max(16, \exp(\gamma_d))$, where γ_d is the constant from Lemma 6.4.3. If $M \ge M_d$, then

$$\frac{\left\lfloor \frac{\ell_{j-1}}{2} + \frac{\ell_j}{4} \right\rfloor}{\left\lceil \frac{\ell_{j-1}}{2} + \ell_{j-1} + 1 \right\rceil} \ge \frac{\frac{\ell_j}{4}}{2\ell_{j-1}} = \frac{1}{8} \frac{M^{j^3} K \lambda_{\rm mic}}{M^{(j-1)^3} K \lambda_{\rm mic}} = \frac{1}{8} M^{3j^2 - 3j + 1} \,. \tag{6.4.4}$$

The term on the right hand side is in particular bounded below by $\frac{M}{8} \ge 16$, and thus we can indeed apply the lemma. We obtain that there is a function w_Q such that $w_Q = 0$ on

$$Z^{\mathsf{d}} \setminus (Q + Q_{\ell_{j}/4-1}(0)), \nabla_{1}^{2} \left(w_{Q} + \eta^{(j)} \right) \text{ is zero on } Q + Q_{\ell_{j-1}+1}(0), \text{ and such that} \left\| \nabla_{1}^{2} \left(w_{Q} + \eta^{(j)} \right) \right\|_{L^{\infty}(\mathbb{Z}^{\mathsf{d}})} \leq \left(1 + \frac{\gamma_{\mathsf{d}}}{\log \left(\frac{\ell_{j-1}}{2} + \frac{\ell_{j}}{4} \right) - \log \left(\frac{\ell_{j-1}}{2} + \ell_{j-1} + 1 \right)} \right) \left\| \nabla_{1}^{2} \eta^{(j)} \right\|_{L^{\infty}(\mathbb{Z}^{\mathsf{d}})} \leq \left(1 + \frac{\gamma_{\mathsf{d}}}{\log \left(\frac{1}{8} M^{3j^{2} - 3j + 1} \right)} \right) \left\| \nabla_{1}^{2} \eta^{(j)} \right\|_{L^{\infty}(\mathbb{Z}^{\mathsf{d}})} \leq \left(1 + \frac{1}{3j^{2} - 3j + 1} \right) \left\| \nabla_{1}^{2} \eta^{(j)} \right\|_{L^{\infty}(\mathbb{Z}^{\mathsf{d}})}$$
(6.4.5)

where we have used (6.4.4) and the fact that $\frac{\gamma_d}{\log\left(\frac{M}{8}\right)} \leq \frac{\gamma_d}{\log\exp(\gamma_d)} = 1$. We set

$$\eta^{(j-1)} = \eta^{(j)} + \sum_{Q \in \tilde{Y}_{K,M,V}^{(j)}(A)} w_Q$$

and

$$V^{(j-1)} = V^{(j)} + Q_{\ell_i}(0)$$

and finally we set $\eta = \eta^{(0)}$.

Step 3: Proof that the $\eta^{(j)}$ are locally affine on the bad cubes with $j_{isol,Q} \ge j$ We prove by reverse induction that supp $\eta^{(j)} \subset V^{(j)}$, $\eta^{(j)} = \eta_*$ on $\mathbb{Z}^d \setminus V^{(j)}$ and $\nabla_1^2 \eta^{(j)} = 0$ on $\bigcup_{k \ge j} \bigcup_{q \in T_{K,M,k}(A)} (q + Q_1(0))$. This is obvious for $j = j_*(\varepsilon)$, so assume that it holds for some $j \in \{1, \ldots, j_*(\varepsilon)\}$.

We claim that then also $\nabla_1^2 \eta^{(j-1)} = 0$ on $\bigcup_{k \ge j-1} \bigcup_{q \in T_{K,M,k}(A)} (q + Q_1(0))$. To see this, let $q \in T_{K,M,k}(A)$ for $k = j_{isol,q} \ge j - 1$. We distinguish the two cases $j_{isol,q} \ge j$ and $j_{isol,q} = j - 1$.

In the former case by our inductive assumption already $\nabla_1^2 \eta^{(j)} = 0$ on $q + Q_1(0)$. Furthermore, we have argued that q does not intersect $(Q + Q_{\ell_j/4+1}(0)) \setminus (Q + Q_{\ell_{j-1}}(0))$ for any $Q \in \tilde{Y}_{K,M,V}^{(j)}(A)$. So either q does not intersect $Q + Q_{\ell_j/4+1}(0)$ for any $Q \in \tilde{Y}_{K,M,V}^{(j)}(A)$, or q is contained in $Q + Q_{\ell_{j-1}}(0)$ for exactly one $Q \in \tilde{Y}_{K,M,V}^{(j)}(A)$. In the former case, all $\nabla_1^2 w_Q$ are equal to 0 on $q + Q_1(0)$, and thus $\nabla_1^2 \eta^{(j-1)} = \nabla_1^2 \eta^{(j)} = 0$ on $q + Q_1(0)$, while in the latter case it holds that $\nabla_1^2 \eta^{(j-1)} = \nabla_1^2 (\eta^{(j)} + w_Q)$ on $q + Q_1(0)$, and thus by construction of w_Q we have $\nabla_1^2 \eta^{(j-1)} = 0$ on $q + Q_1(0)$.

It remains to consider the case that $j_{isol,q} = j - 1$. In that case the sequence (6.4.3) contains a parent $q \,\subset q^{(j-1)} \in S_{K,M,bad}^{(j-1)}(A)$. If $(q^{(j-1)} + Q_1(0)) \cap V^{(j)} = \emptyset$, then $\nabla_1^2 \eta^{(j)} = 0$ on $q^{(j-1)} + Q_1(0)$. Furthermore, by the definition of $Y_{K,M,V}^{(j-1)}(A)$, $q^{(j-1)}$ does not intersect $(Q + Q_{\ell_j/4+1}(0)) \setminus (Q + Q_{\ell_{j-1}}(0))$ for any $Q \in \tilde{Y}_{K,M,V}^{(j-1)}(A) \subset Y_{K,M,V}^{(j-1)}(A)$, and so neither does q. Arguing as in the previous case, we find that $\nabla_1^2 \eta^{(j-1)} = \nabla_1^2 \eta^{(j)}$ on $q + Q_1(0)$. On the other hand, it could be that $(q^{(j-1)} + Q_1(0)) \cap V^{(j)} \neq \emptyset$. Then $q^{(j-1)} \in Y_{K,M,V}^{(j-1)}(A)$, and so there is some $Q \in \tilde{Y}_{K,M,V}^{(j-1)}(A)$ with $q \subset q^{(j-1)} \subset Q + Q_{\ell_{j-1}}(0)$. Then it holds once again that $\nabla_1^2 \eta^{(j-1)} = \nabla_1^2 (\eta^{(j)} + w_Q) = 0$ on $q + Q_1(0)$.

This proves that $\nabla_1^2 \eta^{(j-1)} = 0$ on $\bigcup_{k \ge j-1} \bigcup_{q \in T_{K,M,k}(A)} (q + Q_1(0))$. Furthermore, each $Q \in \tilde{Y}_{K,M,V}^{(j-1)}(A)$ is contained in $V^{(j)} + Q_{2(\ell_{j-1}+2)}(0)$, so the support of the associated w_Q is contained in $V^{(j)} + Q_{2(\ell_{j-1}+2)+\ell_j/4-1}(0)$. So, $\nabla_1^2 \eta^{(j-1)}$ is supported in $V^{(j)} + Q_{2(\ell_{j-1}+2)+\ell_j/4}(0) \subset V^{(j)} + Q_{\ell_j}(0) = V^{(j-1)}$, and $\eta^{(j)} = \eta_*$ on $\mathbb{Z}^d \setminus V^{(j-1)}$. This completes the induction.

Step 4: Proof that η *satisfies i), ii) and iii)*

We define $\eta = \eta^{(0)}$. In the previous step we have shown that $\nabla_1^2 \eta$ is supported in $V^{(0)}$ and $\eta = \eta_*$ on $\mathbb{Z}^d \setminus V^{(0)}$. We have

$$V^{(0)} = V + Q_{\ell_0}(0) + \ldots + Q_{\ell_{j_*(\varepsilon)}}(0)$$

$$\subset V + Q_{2\ell_{j_*(\varepsilon)}}(0) \subset V + Q_{KL\lambda_{\max}/4}(0) \subset (U + Q_{7KL\lambda_{\max}/8+1}(0)) \setminus (U + Q_{KL\lambda_{\max}/8-1}(0)) + Q_{KL\lambda_{\max}/8-1}(0))$$

Thus, $\eta = \eta_* = 0$ on $U + Q_{KL\lambda_{mac}/8-1}(0)$, $\eta = \eta_* = 1$ on $\mathbb{Z}^d \setminus (U + Q_{7KL\lambda_{mac}/8+1}(0))$. This means that η satisfies i) and ii) as soon as ε is small enough (depending on d and K).

In Step 3 we have also seen that $\eta = \eta^{(0)}$ satisfies $\nabla_1^2 \eta = 0$ on $\bigcup_{k \ge 0} \bigcup_{q \in T_{K,M,k}(A)} (q + Q_1(0)) = \bigcup_{q \in S_{K,M,bad}^{(0)}(A)} (q + Q_1(0))$. Thus, to show that η also satisfies iii) we only have to check that $\|\nabla_1^2 \eta\|_{L^{\infty}(\mathbb{Z}^d)} \le \frac{C_d}{(KL)^2 \lambda_{mac}^2}$.

To see this, note the supports of the functions $\nabla_1^2 w_Q$ for $Q \in \tilde{Y}_{K,M,V}^{(j-1)}(A)$ are disjoint. Thus, (6.4.5) implies the bound

$$\left\|\nabla_1^2 \eta^{(j-1)}\right\|_{L^{\infty}(\mathbb{Z}^d)} \le \left(1 + \frac{1}{3j^2 - 3j + 1}\right) \left\|\nabla_1^2 \eta^{(j)}\right\|_{L^{\infty}(\mathbb{Z}^d)}$$

for $j \ge 1$. Iterating this, we find that

$$\begin{split} \|\nabla_{1}^{2}\eta\|_{L^{\infty}(\mathbb{Z}^{d})} &\leq \left(\prod_{j=1}^{j_{*}(\varepsilon)} \left(1 + \frac{1}{3j^{2} - 3j + 1}\right)\right) \|\nabla_{1}^{2}\eta_{*}\|_{L^{\infty}(\mathbb{Z}^{d})} \\ &\leq \left(\prod_{j=1}^{\infty} \left(1 + \frac{1}{3j^{2} - 3j + 1}\right)\right) \frac{C_{d}}{(KL)^{2}\lambda_{\text{mac}}^{2}} \\ &\leq \frac{C_{d}}{(KL)^{2}\lambda_{\text{mac}}^{2}} \end{split}$$
(6.4.6)

where in the last step we used that $\frac{1}{3j^2-3j+1}$ is summable and therefore $\prod_{j=1}^{\infty} \left(1 + \frac{1}{3j^2-3j+1}\right) < \infty$. This completes the proof.

6.4.2 Decay estimates on good domains

Now that we have a construction of a cut-off function at our disposal, we can execute the Widman hole filler argument.

Lemma 6.4.4. Let $d \ge 1$ and let M_d be the constant from Lemma 6.4.2. Let K, L, M be odd integers such that K is a multiple of 3 and $M \ge M_d$. Then for all ε sufficiently small (depending on d, K) the following holds. Let $U \in \mathcal{P}_{KL\lambda_{mac}}$ be a polymer. Suppose that none of the $KL\lambda_{mac}$ -boxes touching U (in the l_{∞} -sense) are bad of type I or II, i.e.

$$\{Q \in \mathcal{Q}_{KL\lambda_{\max}} \colon Q \subset (U + Q_{KL\lambda_{\max}}(0)) \setminus U\} \cap S^*_{K,L,M,\text{bad}}(A) = \emptyset.$$

Then the following holds.

a) If $u : \mathbb{Z}^d \to \mathbb{R}$ is a function such that u = 0 on $\tilde{A} \setminus U$ and $u\Delta_1^2 u = 0$ on $\mathbb{Z}^d \setminus U$, we have the estimate

$$\left\| \nabla_{1}^{2} u \right\|_{L^{2}(\mathbb{Z}^{d} \setminus (U + Q_{KL\lambda_{\max}}(0)))}^{2} \leq \left(\frac{C_{\mathsf{d}} K^{\mathsf{d}-4} \left(1 + \mathbb{1}_{\mathsf{d}=4} \log K \right)}{L^{2}} + \frac{1}{4} \right) \left\| \nabla_{1}^{2} u \right\|_{L^{2}((U + Q_{KL\lambda_{\max}}(0)) \setminus U))}^{2}$$

$$(6.4.7)$$

b) If $u: \mathbb{Z}^d \to \mathbb{R}$ is a function such that u = 0 on $(U + Q_{KL\lambda_{mac}}(0)) \cap \tilde{A}$ and $u\Delta_1^2 u = 0$ on $U + Q_{KL\lambda_{mac}}(0)$, we have the estimate

$$\left\|\nabla_{1}^{2}u\right\|_{L^{2}(U)}^{2} \leq \left(\frac{C_{\mathsf{d}}K^{\mathsf{d}-4}\left(1+\mathbb{1}_{\mathsf{d}=4}\log K\right)}{L^{2}}+\frac{1}{4}\right)\left\|\nabla_{1}^{2}u\right\|_{L^{2}\left(\left(U+Q_{KL\lambda_{\mathsf{mac}}}(0)\right)\setminus U\right)\right)}^{2}.$$
 (6.4.8)

Proof. The proof proceeds as outlined in Section 6.1.3. We begin with the proof of part a); the proof of part b) will be very similar.

Step 1: Discrete integration by parts We know that none of the cubes in $(U + Q_{KL\lambda_{mac}}(0)) \setminus U$ is bad of type I. Thus, we can apply Lemma 6.4.2. Let η be the cut-off function that we obtain from that Lemma.

We now carry out the discrete analogue of the calculation that lead to (6.1.24). Namely we see that

$$\begin{split} 0 &= (\Delta_{1}^{2}u, \eta u)_{L^{2}(\mathbb{Z}^{d})} \\ &= \sum_{i,j=1}^{d} \left(D_{i}^{1} D_{-j}^{1}u, D_{i}^{1} D_{-j}^{1}(\eta u) \right)_{L^{2}(\mathbb{Z}^{d})} \\ &= \sum_{i,j=1}^{d} \left(D_{i}^{1} D_{-j}^{1}u, \left(u D_{i}^{1} D_{-j}^{1}\eta + D_{i}^{1} u \tau_{i}^{1} D_{-j}^{1}\eta + D_{-j}^{1} u \tau_{-j}^{1} D_{i}^{1}\eta + D_{i}^{1} D_{-j}^{1} u \tau_{i}^{1} \tau_{-j}^{1}\eta \right) \right)_{L^{2}(\mathbb{Z}^{d})} \\ &= \sum_{i,j=1}^{d} \sum_{x \in \mathbb{Z}^{d}} \left| D_{i}^{1} D_{-j}^{1}u(x) \right|^{2} \tau_{i}^{1} \tau_{-j}^{1}\eta(x) + \sum_{i,j=1}^{d} \sum_{x \in \mathbb{Z}^{d}} u(x) D_{i}^{1} D_{-j}^{1}u(x) D_{i}^{1} D_{-j}^{1}\eta(x) \\ &+ \sum_{i,j=1}^{d} \sum_{x \in \mathbb{Z}^{d}} D_{i}^{1} D_{-j}^{1}u(x) D_{i}^{1}u(x) \tau_{i}^{1} D_{-j}^{1}\eta(x) + \sum_{i,j=1}^{d} \sum_{x \in \mathbb{Z}^{d}} D_{i}^{1} D_{-j}^{1}u(x) D_{-j}^{1}u(x) \tau_{-j}^{1} D_{i}^{1}\eta(x) \,. \end{split}$$

$$(6.4.9)$$

Consider the third term in this sum. We can apply summation by parts here and obtain

$$\begin{split} &\sum_{i,j=1}^{d} \sum_{x \in \mathbb{Z}^{d}} D_{i}^{1} D_{-j}^{1} u(x) D_{i}^{1} u(x) \tau_{i}^{1} D_{-j}^{1} \eta(x) \\ &= -\sum_{i,j=1}^{d} \sum_{x \in \mathbb{Z}^{d}} D_{i}^{1} u(x) D_{j}^{1} \left(D_{i}^{1} u(x) \tau_{i}^{1} D_{-j}^{1} \eta(x) \right) \\ &= -\sum_{i,j=1}^{d} \sum_{x \in \mathbb{Z}^{d}} D_{i}^{1} u(x) D_{i}^{1} u(x) \tau_{i}^{1} D_{j}^{1} D_{-j}^{1} \eta(x) - \sum_{i,j=1}^{d} \sum_{x \in \mathbb{Z}^{d}} D_{i}^{1} u(x) D_{i}^{1} D_{j}^{1} u_{-j}^{1} \eta(x) \\ &= -\sum_{i,j=1}^{d} \sum_{x \in \mathbb{Z}^{d}} |D_{i}^{1} u(x)|^{2} \tau_{i}^{1} D_{j}^{1} D_{-j}^{1} \eta(x) - \sum_{i,j=1}^{d} \sum_{x \in \mathbb{Z}^{d}} D_{i}^{1} u(x) D_{i}^{1} D_{-j}^{1} \eta(x) \end{split}$$

where in the last step we changed the index of summation from *x* to $\tau_{-j}^1 x$. This implies

$$\sum_{i,j=1}^{d} \sum_{x \in \mathbb{Z}^{d}} D_{i}^{1} D_{-j}^{1} u(x) D_{i}^{1} u(x) \tau_{i}^{1} D_{-j}^{1} \eta(x) = -\frac{1}{2} \sum_{i,j=1}^{d} \sum_{x \in \mathbb{Z}^{d}} |D_{i}^{1} u(x)|^{2} \tau_{i}^{1} D_{j}^{1} D_{-j}^{1} \eta(x).$$

Similarly, we find for the fourth summand in (6.4.9) that

$$\sum_{i,j=1}^{\mathsf{d}} \sum_{x \in \mathbb{Z}^{\mathsf{d}}} D_{i}^{1} D_{-j}^{1} u(x) D_{-j}^{1} u(x) \tau_{-j}^{1} D_{i}^{1} \eta(x) = -\frac{1}{2} \sum_{i,j=1}^{\mathsf{d}} \sum_{x \in \mathbb{Z}^{\mathsf{d}}} |D_{-j}^{1} u(x)|^{2} \tau_{-j}^{1} D_{i}^{1} D_{-i}^{1} \eta(x).$$

If we use the last two equalities in (6.4.9), we arrive at

$$\begin{split} &\sum_{i,j=1}^{\mathsf{d}} \sum_{x \in \mathbb{Z}^{\mathsf{d}}} \left| D_{i}^{1} D_{-j}^{1} u(x) \right|^{2} \tau_{i}^{1} \tau_{-j}^{1} \eta(x) \\ &= \frac{1}{2} \sum_{i,j=1}^{\mathsf{d}} \sum_{x \in \mathbb{Z}^{\mathsf{d}}} |D_{i}^{1} u(x)|^{2} \tau_{i}^{1} D_{j}^{1} D_{-j}^{1} \eta(x) + \frac{1}{2} \sum_{i,j=1}^{\mathsf{d}} \sum_{x \in \mathbb{Z}^{\mathsf{d}}} |D_{-j}^{1} u(x)|^{2} \tau_{-j}^{1} D_{i}^{1} D_{-i}^{1} \eta(x) \\ &- \sum_{i,j=1}^{\mathsf{d}} \sum_{x \in \mathbb{Z}^{\mathsf{d}}} u(x) D_{i}^{1} D_{-j}^{1} u(x) D_{i}^{1} D_{-j}^{1} \eta(x) \,. \end{split}$$

Here, in the second summand on the right-hand side, we can shift the summation from *x* to $x + e_i$ and then interchange the indices *i* and *j* to see that

$$\begin{split} &\sum_{i,j=1}^{d} \sum_{x \in \mathbb{Z}^{d}} \left| D_{i}^{1} D_{-j}^{1} u(x) \right|^{2} \tau_{i}^{1} \tau_{-j}^{1} \eta(x) \\ &= \frac{1}{2} \sum_{i,j=1}^{d} \sum_{x \in \mathbb{Z}^{d}} |D_{i}^{1} u(x)|^{2} \tau_{i}^{1} D_{j}^{1} D_{-j}^{1} \eta(x) + \frac{1}{2} \sum_{i,j=1}^{d} \sum_{x \in \mathbb{Z}^{d}} |D_{i}^{1} u(x)|^{2} D_{j}^{1} D_{-j}^{1} \eta(x) \\ &- \sum_{i,j=1}^{d} \sum_{x \in \mathbb{Z}^{d}} u(x) D_{i}^{1} D_{-j}^{1} u(x) D_{i}^{1} D_{-j}^{1} \eta(x) \\ &= \frac{1}{2} \sum_{i,j=1}^{d} \sum_{x \in \mathbb{Z}^{d}} |D_{i}^{1} u(x)|^{2} \left(\tau_{i}^{1} D_{j}^{1} D_{-j}^{1} \eta(x) + D_{j}^{1} D_{-j}^{1} \eta(x) \right) \\ &- \sum_{i,j=1}^{d} \sum_{x \in \mathbb{Z}^{d}} u(x) D_{i}^{1} D_{-j}^{1} u(x) D_{i}^{1} D_{-j}^{1} \eta(x) . \end{split}$$
(6.4.10)

This is the discrete analogue of (6.1.23). To continue, we can use the Cauchy-Schwarz inequality on the second term on the right hand side of (6.4.10) and obtain

$$\begin{split} \sum_{i,j=1}^{d} \sum_{x \in \mathbb{Z}^{d}} \left| D_{i}^{1} D_{-j}^{1} u(x) \right|^{2} \tau_{i}^{1} \tau_{-j}^{1} \eta(x) \\ &\leq \frac{1}{2} \sum_{i,j=1}^{d} \sum_{x \in \mathbb{Z}^{d}} |D_{i}^{1} u(x)|^{2} \left(\tau_{i}^{1} D_{j}^{1} D_{-j}^{1} \eta(x) + D_{j}^{1} D_{-j}^{1} \eta(x) \right) + \sum_{i,j=1}^{d} \sum_{x \in \mathbb{Z}^{d}} |u(x)|^{2} \left| D_{i}^{1} D_{-j}^{1} \eta(x) \right|^{2} \\ &+ \frac{1}{4} \sum_{i,j=1}^{d} \sum_{x \in \mathbb{Z}^{d}} \left| D_{i}^{1} D_{-j}^{1} u(x) \right|^{2} \mathbb{1}_{\nabla_{1}^{2} \eta(x) \neq 0} \end{split}$$

$$(6.4.11)$$

which is the discrete analogue of (6.1.24). Next, we use the properties of η . First, recall that $\eta = 1$ on $\mathbb{Z}^d \setminus (U + Q_{KL\lambda_{mac}-2K\lambda_{mic}}(0))$, and $\eta = 0$ on $U + Q_{2K\lambda_{mic}}(0)$. Therefore certainly $\tau_i^1 \tau_{-j}^1 \eta = 1$ on $\mathbb{Z}^d \setminus (U + Q_{KL\lambda_{mac}}(0))$, and $\nabla_1^2 \eta = 0$ on $\mathbb{Z}^d \setminus V$, where $V := (U + Q_{KL\lambda_{mac}-K\lambda_{mic}}(0)) \setminus (U + Q_{K\lambda_{mic}}(0))$, which we can use to bound the left-hand side and the third term on the right-hand side of (6.4.11). We also know that

$$|\nabla_1^2 \eta(x)| \leq \frac{C_{\mathsf{d}}}{(KL)^2 \lambda_{\max}^2} \mathbb{1}_{Q_1(x)} \cap \bigcup_{Q \in \mathcal{S}_{K,M,\mathsf{bad}}^{(0)}} Q = \emptyset$$

which implies that

$$|\tau_{\pm k}^{1} D_{i}^{1} D_{-j}^{1} \eta(x)| \leq \frac{C_{\mathsf{d}}}{(KL)^{2} \lambda_{\max}^{2}} \mathbb{1}_{x \notin \bigcup_{Q \in \mathcal{S}_{K,M,\mathsf{bad}}^{(0)}} Q}$$

for any $i, j, k \in \{1, ..., d\}$. We can use this for the first and second term on the right-hand side of (6.4.11). Putting everything together, we see that

$$\begin{split} \left\| \nabla_{1}^{2} u \right\|_{L^{2}(\mathbb{Z}^{d} \setminus (U + Q_{KL\lambda_{\text{mac}}}(0)))}^{2} \\ &= \sum_{i,j=1}^{d} \sum_{x \in \mathbb{Z}^{d} \setminus (U + Q_{KL\lambda_{\text{mac}}}(0))} \left| D_{i}^{1} D_{-j}^{1} u(x) \right|^{2} \\ &\leq \frac{C_{d}}{(KL)^{2} \lambda_{\text{mac}}^{2}} \sum_{i=1}^{d} \sum_{x \in V} |D_{i}^{1} u(x)|^{2} \mathbb{1}_{x \notin \bigcup_{Q \in S_{K,M,\text{bad}}^{(0)}} Q \\ &+ \frac{C_{d}}{(KL)^{4} \lambda_{\text{mac}}^{4}} \sum_{x \in V} |u(x)|^{2} \mathbb{1}_{x \notin \bigcup_{Q \in S_{K,M,\text{bad}}^{(0)}} Q + \frac{1}{4} \sum_{i,j=1}^{d} \sum_{x \in V} \left| D_{i}^{1} D_{-j}^{1} u(x) \right|^{2} \\ &= \frac{C_{d}}{(KL)^{2} \lambda_{\text{mac}}^{2}} \left\| \nabla_{1} u \mathbb{1}_{\cdot \notin \bigcup_{Q \in S_{K,M,\text{bad}}^{(0)}} Q \right\|_{L^{2}(V)}^{2} + \frac{C_{d}}{(KL)^{4} \lambda_{\text{mac}}^{4}} \left\| u \mathbb{1}_{\cdot \notin \bigcup_{Q \in S_{K,M,\text{bad}}^{(0)}} Q \right\|_{L^{2}(V)}^{2} \\ &+ \frac{1}{4} \left\| \nabla_{1}^{2} u \right\|_{L^{2}(V)}^{2} . \end{split}$$

$$(6.4.12)$$

Step 2: Use of Poincaré and interpolation inequalities

We continue by estimating the first term on the right hand side of (6.4.12). To do so, we want to apply Lemma 6.3.3 on each of the $K\lambda_{mac}$ -boxes in $(U + Q_{KL\lambda_{mac}}(0)) \setminus U$. Note that $(U + Q_{KL\lambda_{mac}}(0)) \setminus U \in \mathcal{P}_{KL\lambda_{mac}}$, i.e. it is the disjoint union of some cubes in $\mathcal{Q}_{KL\lambda_{mac}}$. Let Q' be one such cube. It is the disjoint union of L^d cubes in $\mathcal{Q}_{K\lambda_{mac}}$. Let q be one of them, and let $B_q := q \cap V \setminus \bigcup_{Q \in S_{K,M,bad}^{(0)}} Q$. We claim that $|B_q| \geq \frac{1}{2}|q| = \frac{1}{2}K^d\lambda_{mac}^d$. To see this, note that

$$|q \setminus V| \leq 2\mathsf{d}K^{\mathsf{d}-1}\lambda_{\mathrm{mac}}^{\mathsf{d}-1}K\lambda_{\mathrm{mic}} = 2\mathsf{d}K^{\mathsf{d}}\lambda_{\mathrm{mac}}^{\mathsf{d}-1}\lambda_{\mathrm{mic}}.$$

Furthermore, by assumption Q' is not bad of type II. Therefore, each of its $K\lambda_{\text{mac}}$ -subcubes contains at most $\frac{1}{4} \left(\frac{\lambda_{\text{mac}}}{\lambda_{\text{mic}}}\right)^{d}$ cubes in $S_{K,M,\text{bad}}^{(0)}(A)$, i.e.

$$\left| q \cap \bigcup_{Q \in S_{K,M,\text{bad}}^{(0)}(A)} Q \right| \leq \frac{1}{4} \left(\frac{\lambda_{\text{mac}}}{\lambda_{\text{mic}}} \right)^{\mathsf{d}} K^{\mathsf{d}} \lambda_{\text{mic}}^{\mathsf{d}} = \frac{1}{4} K^{\mathsf{d}} \lambda_{\text{mac}}^{\mathsf{d}} = \frac{1}{4} |q|.$$

Therefore,

$$|B_q| \ge |q| - |q \setminus V| - \left| q \cap \bigcup_{Q \in S_{K,M,\text{bad}}^{(0)}(A)} Q \right| \ge |q| - \frac{1}{4}|q| - 2\mathsf{d}K^\mathsf{d}\lambda_{\text{mac}}^{\mathsf{d}-1}\lambda_{\text{mic}} \ge \frac{1}{2}|q|$$

whenever ε is small enough (in terms of d, *K*) so that $2K^d \lambda_{mic} \leq \frac{1}{4} \lambda_{mac}$. Thus, we can apply Lemma 6.3.3 on q with $B = B_q$ and obtain

$$\left\| \nabla_{1} u \mathbb{1}_{\boldsymbol{\mathscr{I}} \in \mathcal{O}_{K,M,\text{bad}}^{(0)}} Q \right\|_{L^{2}(V \cap q)}^{2}$$

$$= \left\| \nabla_{1} u \mathbb{1}_{\boldsymbol{\mathscr{I}} \in V \setminus \bigcup_{Q \in \mathcal{S}_{K,M,\text{bad}}^{(0)}}} Q \right\|_{L^{2}(q)}^{2}$$

$$\leq C_{\mathsf{d}}K^{2}\lambda_{\mathsf{mac}}^{2} \left\|\nabla_{1}^{2}u\right\|_{L^{2}(V\cap q)}^{2} + \frac{C_{\mathsf{d}}}{K^{2}\lambda_{\mathsf{mac}}^{2}} \left\|u\mathbb{1}_{\cdot\in V\setminus\bigcup_{Q\in\mathcal{S}_{K,M,\mathsf{bad}}^{(0)}}(Q)}^{2}\right\|_{L^{2}(q)}^{2},$$

and summing this over all Q' and all $q \subset Q'$ we see that

$$\left\| \nabla_{1} u \mathbb{1}_{\mathscr{A} \cup_{Q \in S_{K,M,bad}^{(0)}}^{(0)} Q} \right\|_{L^{2}(V)}^{2} \\ \leq C_{\mathsf{d}} K^{2} \lambda_{\mathsf{mac}}^{2} \left\| \nabla_{1}^{2} u \right\|_{L^{2}(V)}^{2} + \frac{C_{\mathsf{d}}}{K^{2} \lambda_{\mathsf{mac}}^{2}} \left\| u \mathbb{1}_{\mathscr{A} \cup_{Q \in S_{K,M,bad}^{(0)}}^{(0)} Q} \right\|_{L^{2}(V)}^{2}$$

Using this estimate in (6.4.12) we arrive at

$$\begin{split} \left\| \nabla_{1}^{2} u \right\|_{L^{2}(\mathbb{Z}^{d} \setminus (U + Q_{KL\lambda_{mac}}(0)))}^{2} &\leq \frac{C_{d}}{L^{2}} \left\| \nabla_{1}^{2} u \right\|_{L^{2}(V)}^{2} + \frac{C_{d}}{K^{4}L^{2}\lambda_{mac}^{4}} \left\| u \mathbb{1}_{\mathscr{I} \notin \bigcup_{Q \in S_{K,M,bad}^{(0)}}^{Q} Q \right\|_{L^{2}(V)}^{2} \\ &+ \frac{C_{d}}{(KL)^{4}\lambda_{mac}^{4}} \left\| u \mathbb{1}_{\mathscr{I} \notin \bigcup_{Q \in S_{K,M,bad}^{(0)}}^{Q} Q \right\|_{L^{2}(V)}^{2} + \frac{1}{4} \left\| \nabla_{1}^{2} u \right\|_{L^{2}(V)}^{2} \\ &\leq \left(\frac{C_{d}}{L^{2}} + \frac{1}{4} \right) \left\| \nabla_{1}^{2} u \right\|_{L^{2}(V)}^{2} + \frac{C_{d}}{K^{4}L^{2}\lambda_{mac}^{4}} \left\| u \mathbb{1}_{\mathscr{I} \notin \bigcup_{Q \in S_{K,M,bad}^{(0)}}^{Q} Q \right\|_{L^{2}(V)}^{2} \end{split}$$
(6.4.13)

where we used $L \ge 1$ in the last step.

Theorem 6.3.1 with $R = K\lambda_{mic}$ allows to bound

Now one easily checks that for any $d \ge 4$ we have $\lambda_{\text{mic}}^{d} (1 + \mathbb{1}_{d=4} \log \lambda_{\text{mic}}) \le C_{d} \lambda_{\text{mac}}^{4}$. Thus, combining (6.4.13) and (6.4.14) we see that

$$\begin{split} \|\nabla_{1}^{2}u\|_{L^{2}(\mathbb{Z}^{d}\setminus(U+Q_{KL\lambda_{\max}}(0)))}^{2} &\leq \left(\frac{C_{d}}{L^{2}}+\frac{1}{4}\right)\|\nabla_{1}^{2}u\|_{L^{2}((U+Q_{KL\lambda_{\max}}(0))\setminus U)}^{2} \\ &+ \frac{C_{d}K^{d-4}\left(1+\mathbb{1}_{d=4}\log K\right)}{L^{2}}\|\nabla_{1}^{2}u\|_{L^{2}((U+Q_{KL\lambda_{\max}}(0))\setminus U)}^{2} \\ &\leq \left(\frac{C_{d}K^{d-4}\left(1+\mathbb{1}_{d=4}\log K\right)}{L^{2}}+\frac{1}{4}\right)\|\nabla_{1}^{2}u\|_{L^{2}((U+Q_{KL\lambda_{\max}}(0))\setminus U)}^{2} \end{split}$$

which is (6.4.8).

Step 3: Proof of part b)

We proceed completely analogously as in Step 1 and 2. The only difference is that we work with $\hat{\eta} := 1 - \eta$ instead of η . The assumptions for $\nabla_1^2 \eta$ carry over to $\nabla_1^2 \hat{\eta}$, and so the proof carries over.

Lemma 6.4.4 has the following straightforward corollary:

Lemma 6.4.5. Let $d \ge 1$ and let M_d be the constant from Lemma 6.4.2. Let K be an odd multiple of 3 and $M \ge M_d$. Then there is a constant $L_{d,K}$ depending on d and K only such that for any odd integers $L \ge L_{d,K}$, $M \ge M_d$ and for all ε sufficiently small (depending on d, K) the following holds: Let $U_0, \ldots, U_k \in \mathcal{P}_{KL\lambda_{mac}}$ be polymers. Suppose that for each $j \in \{0, \ldots, k-1\}$ we have $U_j + Q_{KL\lambda_{mac}}(0) \subset U_{j+1}$ and

 $\{Q \in \mathcal{Q}_{KL\lambda_{\max}} \colon Q \subset (U_j + Q_{KL\lambda_{\max}}(0)) \setminus U_j\} \cap S^*_{K,L,M,\mathsf{bad}}(A) = \varnothing$

Then the following holds:

a) If $u: \mathbb{Z}^d \to \mathbb{R}$ is a function such that u = 0 on $\tilde{\mathcal{A}} \setminus U_0$ and $u\Delta_1^2 u = 0$ on $\mathbb{Z}^d \setminus U_0$, we have the estimate

$$\left\|\nabla_{1}^{2}u\right\|_{L^{2}(\mathbb{Z}^{\mathsf{d}}\setminus U_{k})}^{2} \leq \frac{1}{2^{k}}\left\|\nabla_{1}^{2}u\right\|_{L^{2}((U_{0}+Q_{KL\lambda_{\max}}(0))\setminus U_{0})}^{2}.$$
(6.4.15)

b) If $u: \mathbb{Z}^{d} \to \mathbb{R}$ is a function such that u = 0 on $U_{k} \cap \tilde{\mathcal{A}}$ and $u\Delta_{1}^{2}u = 0$ on U_{k} , we have the estimate

$$\left\|\nabla_{1}^{2}u\right\|_{L^{2}(U_{0})}^{2} \leq \frac{1}{2^{k}}\left\|\nabla_{1}^{2}u\right\|_{L^{2}((U_{k-1}+Q_{KL\lambda_{\max}}(0))\setminus U_{k-1})}^{2}.$$
(6.4.16)

Proof. We choose *L* large enough so that the prefactors on the right-hand side in (6.4.7) and (6.4.8) become less than $\frac{1}{2}$, and then apply Lemma 6.4.4 iteratively on each U_j .

6.4.3 Sparsity of bad boxes

In order to conclude Theorem 6.4.1 from Lemma 6.4.5 it remains to show that with sufficiently high probability we can find sets U_j as in that Lemma. For that purpose we need to show that bad cubes are sparse enough.

In fact, we will show that we can make the probability of a cube in $Q \in Q_{KL\lambda_{mac}}$ being bad arbitrarily small. We even have a slightly stronger result, namely that for each finite $T^* \subset Q_{KL\lambda_{mac}}$ we can control the probability that all cubes in T^* are bad.

Lemma 6.4.6. Let $d \ge 4$, let p > 0 be arbitrary. Let $M \ge 12$ be an odd integer. Then there is $K_{d,M,p}$ depending on d, M and p only with the following property: let $K \ge K_{d,M,p}$ be an odd multiple of 3, let L be an odd integer, let ε be small enough (depending on d, L, M and p), and let T^* be an arbitrary finite subset of $Q_{KL\lambda_{mac}}$. Then

$$\zeta^{\varepsilon}_{\Lambda}\left(T^* \subset S^{*,I}_{K,L,M,\text{bad}}(\mathcal{A})\right) \le p^{|T^*|}, \qquad (6.4.17)$$

$$\zeta^{\varepsilon}_{\Lambda}\left(T^* \subset S^{*,II}_{K,L,M,\text{bad}}(\mathcal{A})\right) \le p^{|T^*|}, \qquad (6.4.18)$$

$$\zeta^{\varepsilon}_{\Lambda}\left(T^* \subset S^*_{K,L,M,\text{bad}}(\mathcal{A})\right) \le 2(4p)^{\frac{|T^*|}{2}}.$$
(6.4.19)

In order to prove this Lemma, we will have go to into the definition of the bad cubes of type I and II, and, in particular, we will have to understand how rare cubes in $S_{K,M,\text{bad}}^{(j)}(\mathcal{A})$ are. This is quantified in the following Lemma.

Lemma 6.4.7. Let $d \ge 4$. There is a constant K'_d with the following property: Let ε be small enough (depending on d only) so that the conclusion of Theorem 6.1.3 holds. Let $M \ge 12$ be an odd integer and assume that $K \ge K'_d$ is an odd multiple of 3. Let $j \ge 0$. Then we have the following estimates.

. (0)

a) If j = 0 and $T^{(0)} \subset Q_{\ell_0}$ is a finite subset, then

$$\zeta_{\Lambda}^{\varepsilon}\left(T^{(0)} \subset S_{K,M,\text{bad}}^{(0)}(\mathcal{A})\right) \leq \left(K^{\frac{d}{2^{\mathsf{d}}}}(\mathsf{d}+1)^{\frac{1}{2^{\mathsf{d}}}}\exp\left(-\frac{K^{\mathsf{d}}}{C_{\mathsf{d}}}\right)\right)^{|T^{(0)}|}.$$
(6.4.20)

b) If j > 0 and $T^{(j)} \subset Q^{\#}_{\ell_j}$ is a finite subset such that the elements of $T^{(j)}$ are pairwise disjoint, then

$$\zeta_{\Lambda}^{\varepsilon}\left(T^{(j)} \subset S_{K,M,\text{bad}}^{(j)}(\mathcal{A})\right) \leq \left(3^{(2^{j}-2)\mathsf{d}}M^{s_{j}\mathsf{d}}\left(K^{\frac{\mathsf{d}}{2^{\mathsf{d}}}}(\mathsf{d}+1)^{\frac{1}{2^{\mathsf{d}}}}\exp\left(-\frac{K^{\mathsf{d}}}{C_{\mathsf{d}}}\right)\right)^{2^{j}}\right)^{|T^{(j)}|} \tag{6.4.21}$$

where $s_j := j^3 + \sum_{m=0}^j m^3 2^{j-m}$.

Here part a) is rather easy to show. If $Q \in S_{K,M,\text{bad}}^{(0)}(\mathcal{A})$, then there are too few pinned points around Q, and the probability for that can be estimated using Theorem 6.1.5. The crucial point is that by choosing K large this probability can be made arbitrarily small.

Part b) then follows by induction. Each cube in $S_{K,M,\text{bad}}^{(j)}(\mathcal{A})$ contains at least two cubes in $S_{K,M,\text{bad}}^{(j-1)}(\mathcal{A})$, and so if the latter cubes are rare, an union bound will show that the former cubes will be rare as well.

Proof. We show first part a), then part b) for j = 1, and then use that result to start an induction that will yield part b) for j > 1 as well.

Step 1: Proof of part a)

This is similar to the proof of Lemma 6.3.2. However, we want a uniform estimate over the cubes in $T^{(0)}$, and so we need to be more careful.

Let $Q \in T^{(0)}$, and let $q \in Q_{\lambda_{\min}}$ be such that $q \subset Q$. Suppose that q has centre $x \in \mathbb{Z}^d$. For $i \in \{1, \dots, d+1\}$ consider the sets $\Xi_i(q) = \bigcap_{y \in q} (y + \Xi_i)$ and $\Xi_{i,K\lambda_{\min}/2}(q) = \Xi_i(q) \cap Q_{K\lambda_{\min}/2}(x)$.



Figure 6.4: A set $\Xi_i(q)$, given as the intersection of $x + \Xi_i$ for $x \in q$.

The set $\Xi_i(q)$ is an intersection of translates of the same cone, where the tips of the cone range over the set q of diameter $\leq \sqrt{d\lambda_{\text{mic}}}$ (cf. Figure 6.4). As soon as $K \geq K'_d$ for some dimensional constant K'_d the fraction of points in $Q_{K\lambda_{\text{mic}}/2}(x)$ that are in $\Xi_{i,K\lambda_{\text{mic}}/2}(q)$ is bounded below. We fix such a K'_d . In other words, we have the estimate

$$|\Xi_{i,K\lambda_{\rm mic}/2}(q)| \ge \frac{1}{C_{\rm d}} (K\lambda_{\rm mic})^{\rm d}$$
(6.4.22)

for $K \ge K'_{d}$. Furthermore, if $\Xi_{i,K\lambda_{mic}/2}(q) \cap \tilde{\mathcal{A}} \neq \emptyset$ then $d^{(i)}(y, \tilde{\mathcal{A}}) \le \frac{K}{2}\lambda_{mic} + \frac{1}{2}\lambda_{mic} \le K\lambda_{mic}$ for all $y \in q$.

The preceding discussion implies that if $\Xi_{i,K\lambda_{\text{mic}}/2}(q) \cap \tilde{\mathcal{A}} \neq \emptyset$ for all *i* then $d_*(y,\mathcal{A}) \leq K\lambda_{\text{mic}}$ for all $y \in q$. Thus, if $\Xi_{i,K\lambda_{\text{mic}}/2}(q) \cap \tilde{\mathcal{A}} \neq \emptyset$ holds for all *i* and all $q \subset Q$ then $Q \notin S_{K,M,\text{bad}}^{(0)}(\mathcal{A})$. Using this we can write

$$\begin{split} \zeta_{\Lambda}^{\varepsilon} \left(T^{(0)} \subset S_{K,M,\text{bad}}^{(0)}(\mathcal{A}) \right) \\ &= \zeta_{\Lambda}^{\varepsilon} \left(\forall Q \in T^{(0)} \; \exists q_Q \in \mathcal{Q}_{\lambda_{\text{mic}}} \; \exists i_Q \in \{1, \dots, \mathsf{d}+1\} \text{ with } q_Q \subset Q, \Xi_{i_Q, K\lambda_{\text{mic}}/2}(q_Q) \cap \tilde{\mathcal{A}} = \varnothing \right) \\ &= \zeta_{\Lambda}^{\varepsilon} \left(\forall Q \in T^{(0)} \; \exists q_Q \in \mathcal{Q}_{\lambda_{\text{mic}}} \; \exists i_Q \in \{1, \dots, \mathsf{d}+1\} \text{ with } q_Q \subset Q \\ & \text{ such that } \bigcup_{Q \in T^{(0)}} \Xi_{i_Q, K\lambda_{\text{mic}}/2}(q_Q) \cap \tilde{\mathcal{A}} = \varnothing \right). \end{split}$$

We can estimate this probability by summing over all choices $\underline{q} = (q_Q)_{Q \in T^{(0)}} \in (\mathcal{Q}_{\lambda_{\text{mic}}})^{T^{(0)}}$ and $\underline{i} = (i_Q)_{Q \in T^{(0)}} \in \{1, \dots, d+1\}^{T^{(0)}}$ to find

$$\zeta^{\varepsilon}_{\Lambda}\left(T^{(0)} \subset S^{(0)}_{K,M,\mathrm{bad}}(\mathcal{A})\right) \leq \sum_{\substack{\underline{q} \in (\mathcal{Q}_{\lambda_{\mathrm{mic}}})^{T^{(0)}} \\ q_{Q} \subset Q \ \forall Q \in T^{(0)}}} \sum_{\underline{i} \in \{1,\dots,\mathsf{d}+1\}^{T^{(0)}}} \zeta^{\varepsilon}_{\Lambda}\left(\bigcup_{Q \in T^{(0)}} \Xi_{i_{Q},K\lambda_{\mathrm{mic}}/2}(q_{Q}) \cap \tilde{\mathcal{A}} = \varnothing\right).$$

Assume for the moment that the elements of $T^{(0)}$ are well-separated in the sense that for any $Q, Q' \in T^{(0)}$ with $Q \neq Q'$ we have $d_{\infty}(Q, Q') \geq K\lambda_{\text{mic}}$. Because $\Xi_{i_Q, K\lambda_{\text{mic}}/2}(q_Q)$ is a subset of $q_Q + Q_{K\lambda_{\text{mic}}/2}(0)$, it is a subset of the cube with sidelength $2K\lambda_{\text{mic}}$ concentric to Q. In particular, by our temporary assumption, if $Q \neq Q'$, then $\Xi_{i_Q, K\lambda_{\text{mic}}/2}(q_Q)$ and $\Xi_{i_{Q'}, K\lambda_{\text{mic}}/2}(q_{Q'})$ are disjoint. Thus, (6.4.22) implies

$$\left|\bigcup_{Q\in T^{(0)}}\Xi_{i_Q,K\lambda_{\mathrm{mic}}/2}(q_Q)\right|\geq \frac{|T^{(0)}|}{C_{\mathsf{d}}}(K\lambda_{\mathrm{mic}})^{\mathsf{d}}.$$

Using Theorem 6.1.3 we now see

$$\begin{split} \zeta_{\Lambda}^{\varepsilon} \left(T^{(0)} \subset S_{K,M,\mathrm{bad}}^{(0)}(\mathcal{A}) \right) &\leq \sum_{\substack{\underline{q} \in (\mathcal{Q}_{\lambda_{\mathrm{mic}}})^{T^{(0)}} \\ q_{Q} \subset Q \ \forall Q \in T^{(0)}}} \sum_{i \in \{1,\dots,\mathsf{d}+1\}^{T^{(0)}}} (1-p_{\mathsf{d},-})^{\left| \bigcup_{Q \in T^{(0)}} \Xi_{i_{Q},K\lambda_{\mathrm{mic}}/2}(q_{Q}) \right|} \\ &\leq \sum_{\substack{\underline{q} \in (\mathcal{Q}_{\lambda_{\mathrm{mic}}})^{T^{(0)}} \\ q_{Q} \subset Q \ \forall Q \in T^{(0)}}} \sum_{\underline{i} \in \{1,\dots,\mathsf{d}+1\}^{T^{(0)}}} \exp\left(-p_{\mathsf{d},-}\frac{|T^{(0)}|(K\lambda_{\mathrm{mic}})^{\mathsf{d}}}{C_{\mathsf{d}}}\right) \\ &\leq \sum_{\substack{\underline{q} \in (\mathcal{Q}_{\lambda_{\mathrm{mic}}})^{T^{(0)}} \\ q_{Q} \subset Q \ \forall Q \in T^{(0)}}} \sum_{\underline{i} \in \{1,\dots,\mathsf{d}+1\}^{T^{(0)}}} \exp\left(-p_{\mathsf{d},-}\frac{|T^{(0)}|(K\lambda_{\mathrm{mic}})^{\mathsf{d}}}{C_{\mathsf{d}}}\right). \end{split}$$

In any dimension $d \ge 4$ we have $p_{d,-}\lambda_{\text{mic}}^d \ge \frac{1}{C_d}$. We also have $(K^d)^{|T^{(0)}|}$ choices for \underline{q} , and $(d+1)^{|T^{(0)}|}$ choices for \underline{i} , and so

$$\zeta_{\Lambda}^{\varepsilon} \left(T^{(0)} \subset S^{(0)}_{K,M,\text{bad}}(\mathcal{A}) \right) \le (K^{\mathsf{d}})^{|T^{(0)}|} (\mathsf{d}+1)^{|T^{(0)}|} \exp\left(-\frac{|T^{(0)}|K^{\mathsf{d}}}{C_{\mathsf{d}}}\right).$$
(6.4.23)

This estimate was derived under the assumption that $T^{(0)}$ is such that for any $Q, Q' \in T^{(0)}$ with $Q \neq Q'$ we have $d_{\infty}(Q, Q') \geq 2K\lambda_{\text{mic}}$. In general, this will not be the case. However, we can partition $T^{(0)}$ into 2^{d} subsets $T_i^{(0)}$ for $i \in \{1, \ldots, 2^{\mathsf{d}}\}$ such that for any i and any $Q, Q' \in T_i^{(0)}$ with $Q \neq Q'$ we have $d_{\infty}(Q, Q') \geq K\lambda_{\text{mic}}$. At least one of these subsets, say $T_{i_*}^{(0)}$, will contain at least $\frac{|T^{(0)}|}{2^{\mathsf{d}}}$ boxes. Then we can apply the estimate (6.4.23) to $T_{i_*}^{(0)}$ and obtain

$$\begin{split} \zeta_{\Lambda}^{\varepsilon} \left(T^{(0)} \subset S_{K,M,\text{bad}}^{(0)}(\mathcal{A}) \right) &\leq \zeta_{\Lambda}^{\varepsilon} \left(T_{i_*}^{(0)} \subset S_{K,M,\text{bad}}^{(0)}(\mathcal{A}) \right) \\ &\leq (K^{\mathsf{d}})^{\frac{|T^{(0)}|}{2^{\mathsf{d}}}} (\mathsf{d}+1)^{\frac{|T^{(0)}|}{2^{\mathsf{d}}}} \exp\left(-\frac{|T^{(0)}|K^{\mathsf{d}}}{2^{\mathsf{d}}C_{\mathsf{d}}}\right) \\ &\leq \left(K^{\frac{\mathsf{d}}{2^{\mathsf{d}}}} (\mathsf{d}+1)^{\frac{1}{2^{\mathsf{d}}}} \exp\left(-\frac{K^{\mathsf{d}}}{C_{\mathsf{d}}}\right) \right)^{|T^{(0)}|} \end{split}$$

which is (6.4.20).

Step 2: Proof of part b) for j = 1

We want to prove (6.4.21) by induction on j. In principle, we would want to use (6.4.20) as the base case. However, that statement is for Q_{ℓ_0} instead of $Q_{\ell_0}^{\#}$, and so we first derive (6.4.21) for j = 1 from (6.4.20) and then use this assertion to start our induction.

Let $Q \in T^{(1)}$. By construction $Q \in S^{(1)}_{K,M,\text{bad}}(\mathcal{A})$ if and only if there are at least two disjoint cubes $q, q' \in S^{(0)}_{K,M,\text{bad},\text{clust}}(\mathcal{A}) \subset S^{(0)}_{K,M,\text{bad}}(\mathcal{A})$ such that $q, q' \subset Q$, and so

As in Step 1 we can estimate this expression by the sum over all possibilities for q_Q , q'_Q to find

$$\zeta_{\Lambda}^{\varepsilon}\left(T^{(1)} \subset S^{(1)}_{K,M,\mathrm{bad}}(\mathcal{A})\right) \leq \sum_{\substack{\underline{q},\underline{q}' \in (\mathcal{Q}_{\ell_0})^{T^{(1)}} \\ q_{Q} \cap q'_{Q} = \varnothing, q_{Q} \cup q'_{Q} \subset Q \ \forall Q \in T^{(1)}}} \zeta_{\Lambda}^{\varepsilon}\left(\bigcup_{Q \in T^{(1)}} \{q_{Q},q'_{Q}\} \subset S^{(0)}_{K,M,\mathrm{bad}}(\mathcal{A})\right) \,.$$

By assumption the elements of $T^{(1)}$ are pairwise disjoint. Therefore, all q_Q and q'_Q are pairwise distinct. Hence, the set $\bigcup_{Q \in T^{(1)}} \{q_Q, q'_Q\}$ has cardinality $2|T^{(1)}|$, and so (6.4.20) implies

$$\zeta^{\varepsilon}_{\Lambda}\left(T^{(1)} \subset S^{(1)}_{K,M,\mathrm{bad}}(\mathcal{A})\right) \leq \sum_{\substack{\underline{q},\underline{q}' \in (\mathcal{Q}_{\ell_0})^{T^{(1)}} \\ q_Q \cap q'_Q = \varnothing, q_Q \cup q'_Q \subset Q \ \forall Q \in T^{(1)}}} \left(K^{\frac{\mathrm{d}}{2^{\mathrm{d}}}}(\mathrm{d}+1)^{\frac{1}{2^{\mathrm{d}}}} \exp\left(-\frac{K^{\mathrm{d}}}{C_{\mathrm{d}}}\right)\right)^{2|T^{(1)}|}$$

$$\leq \left(\left(\frac{\ell_1}{\ell_0}\right)^{2\mathsf{d}} \right)^{|T^{(1)}|} \left(K^{\frac{\mathsf{d}}{2^\mathsf{d}}} (\mathsf{d}+1)^{\frac{1}{2^\mathsf{d}}} \exp\left(-\frac{K^\mathsf{d}}{C_\mathsf{d}}\right) \right)^{2|T^{(1)}|} \\ = M^{2\mathsf{d}|T^{(1)}|} \left(K^{\frac{\mathsf{d}}{2^\mathsf{d}}} (\mathsf{d}+1)^{\frac{1}{2^\mathsf{d}}} \exp\left(-\frac{K^\mathsf{d}}{C_\mathsf{d}}\right) \right)^{2|T^{(1)}|} \\ = \left(M^{2\mathsf{d}} \left(K^{\frac{\mathsf{d}}{2^\mathsf{d}}} (\mathsf{d}+1)^{\frac{1}{2^\mathsf{d}}} \exp\left(-\frac{K^\mathsf{d}}{C_\mathsf{d}}\right) \right)^2 \right)^{|T^{(1)}|}$$

which is (6.4.21) (as $s_1 = 2$).

Step 3: Proof of part b) for j > 1

We proceed by induction on j, using the result from Step 2 as the base case. That is, we assume that (6.4.21) holds for j - 1, and we want to conclude that it also holds for j. The argument for this is analogous to the previous step. The only difference is that now the smaller cubes q, q' live in $Q_{\ell_{j-1}}^{\#}$ instead of $Q_{\ell_{j-1}}$, and so the number of possible $\underline{q}, \underline{q'}$ is now larger. Arguing as in Step 2, we obtain, using the assumption that the elements of $|T^{(j)}|$ are pairwise disjoint, that

$$\begin{split} & \zeta_{\Lambda}^{\varepsilon} \left(T^{(j)} \subset S_{K,M,\text{bad}}^{(j)}(\mathcal{A}) \right) \\ & \leq \sum_{\substack{\underline{q},\underline{q}' \in (\mathcal{Q}_{\ell_{j-1}}^{\#})^{T^{(j)}} \\ q_{Q} \cap q'_{Q} = \varnothing, q_{Q} \cup q'_{Q} \subset Q \ \forall Q \in T^{(j)}}} \zeta_{\Lambda}^{\varepsilon} \left(\bigcup_{Q \in T^{(j)}} \{ q_{Q}, q'_{Q} \} \subset S_{K,M,\text{bad}}^{(j-1)}(\mathcal{A}) \right) \\ & \leq \left(\left(\frac{3\ell_{j}}{\ell_{j-1}} \right)^{2\mathsf{d}} \right)^{|T^{(j)}|} \left(3^{(2^{j-1}-2)\mathsf{d}} M^{s_{j-1}\mathsf{d}} \left(K^{\frac{\mathsf{d}}{2\mathsf{d}}}(\mathsf{d}+1)^{\frac{1}{2\mathsf{d}}} \exp\left(-\frac{K^{\mathsf{d}}}{C_{\mathsf{d}}}\right) \right)^{2^{j-1}} \right)^{2|T^{(j)}|} \quad (6.4.24) \\ & = \left(3^{2\mathsf{d}+2(2^{j-1}-2)\mathsf{d}} M^{2(j^{3}-(j-1)^{3})\mathsf{d}+2s_{j-1}\mathsf{d}} \left(K^{\frac{\mathsf{d}}{2\mathsf{d}}}(\mathsf{d}+1)^{\frac{1}{2\mathsf{d}}} \exp\left(-\frac{M^{\mathsf{d}}}{C_{\mathsf{d}}}\right) \right)^{2^{j}} \right)^{|T^{(j)}|}. \end{split}$$

It remains to observe that $2d + 2(2^{j-1} - 2)d = (2^j - 2)d$ and

$$\begin{split} 2(j^3 - (j-1)^3) \mathrm{d} + 2s_{j-1} \mathrm{d} &= 2\mathrm{d} \left(j^3 - (j-1)^3 + (j-1)^3 + \sum_{m=0}^{j-1} m^3 2^{j-1-m} \right) \\ &= \mathrm{d} \left(2j^3 + \sum_{m=0}^{j-1} m^3 2^{j-m} \right) \\ &= \mathrm{d} \left(j^3 + \sum_{m=0}^{j} m^3 2^{j-m} \right) = s_j \mathrm{d} \,. \end{split}$$

Now we can turn to the proof of Lemma 6.4.6.

Proof of Lemma 6.4.6.

Step 1: Proof of (6.4.18)

Our main tool will be Lemma 6.4.7 a), and the argument is similar to the one in Step 3 of the proof of Theorem 6.1.3 c). We will choose $K_{d,M,p} \ge K'_d$ so that Lemma 6.4.7 can be applied.

Proceeding as in the proof of Lemma 6.4.7 we can estimate

$$\begin{split} \zeta_{\Lambda}^{\varepsilon} \left(T^* \subset S_{K,L,M,\text{bad}}^{*,II}(\mathcal{A}) \right) \\ &= \zeta_{\Lambda}^{\varepsilon} \left(\forall Q \in T^* \exists Q'_Q \in \mathcal{Q}_{K\lambda_{\text{mac}}} \text{ with } Q'_Q \subset Q, \\ & \left| \left\{ q \in S_{K,M,\text{bad}}^{(0)}(\mathcal{A}) : q \subset Q'_Q \right\} \right| \geq \frac{1}{4} \left(\frac{\lambda_{\text{mac}}}{\lambda_{\text{mic}}} \right)^d \right) \\ &\leq \sum_{\substack{Q' \in (\mathcal{Q}_{\ell_{j_*}(\varepsilon)})^{T^*} \\ Q'_Q \subset Q \forall Q \in T^*}} \zeta_{\Lambda}^{\varepsilon} \left(\forall Q \in T^* \left| \left\{ q \in S_{K,M,\text{bad}}^{(0)}(\mathcal{A}) : q \subset Q'_Q \right\} \right| \geq \frac{1}{4} \left(\frac{\lambda_{\text{mac}}}{\lambda_{\text{mic}}} \right)^d \right) \\ &\leq \sum_{\substack{Q' \in (\mathcal{Q}_{\ell_{j_*}(\varepsilon)})^{T^*} \\ Q'_Q \subset Q \forall Q \in T^*}} \zeta_{\Lambda}^{\varepsilon} \left(\left| \left\{ q \in S_{K,M,\text{bad}}^{(0)}(\mathcal{A}) : q \subset \bigcup_{Q \in T^*} Q'_Q \right\} \right| \geq \frac{1}{4} \left(\frac{\lambda_{\text{mac}}}{\lambda_{\text{mic}}} \right)^d |T^*| \right) . \end{split}$$

Let

$$\tilde{T}_{\underline{Q'}} = \{q \in \mathcal{Q}_{\ell_0} \colon q \subset \bigcup_{Q \in T^*} Q'_Q\}$$

and note that $\left|\tilde{T}_{\underline{Q'}}\right| = \left(\frac{\lambda_{\text{mac}}}{\lambda_{\text{mic}}}\right)^{d} |T^*|$. Using (6.4.20) we can now continue to estimate

$$\begin{split} \zeta_{\Lambda}^{\varepsilon} \left(T^{*} \subset S_{K,L,M,\text{bad}}^{*,II}(\mathcal{A}) \right) &\leq \sum_{\substack{\underline{Q}' \in (\mathcal{Q}_{\ell_{j_{s}(c)}})^{T^{*}} \\ Q_{Q} \subset Q \; \forall Q \in T^{*}}} \zeta_{\Lambda}^{\varepsilon} \left(\left| \tilde{T}_{\underline{Q}'} \cap S_{K,M,\text{bad}}^{(0)}(\mathcal{A}) \right| \geq \frac{1}{4} \left| \tilde{T}_{\underline{Q}'} \right| \right) \\ &= \sum_{\substack{\underline{Q}' \in (\mathcal{Q}_{\ell_{j_{s}(c)}})^{T^{*}} \\ Q_{Q} \subset Q \; \forall Q \in T^{*}} |T| \geq |\tilde{T}_{\underline{Q}'}|/4}} \zeta_{\Lambda}^{\varepsilon} \left(\tilde{T}_{\underline{Q}'} \cap S_{K,M,\text{bad}}^{(0)}(\mathcal{A}) = T \right) \\ &\leq \sum_{\substack{\underline{Q}' \in (\mathcal{Q}_{\ell_{j_{s}(c)}})^{T^{*}} \\ Q_{Q} \subset Q \; \forall Q \in T^{*}} |T| \geq |\tilde{T}_{\underline{Q}'}|/4}} \zeta_{\Lambda}^{\varepsilon} \left(T \subset S_{K,M,\text{bad}}^{(0)}(\mathcal{A}) \right) \\ &\leq \sum_{\substack{\underline{Q}' \in (\mathcal{Q}_{\ell_{j_{s}(c)}})^{T^{*}} \\ Q_{Q} \subset Q \; \forall Q \in T^{*}} |T| \geq |\tilde{T}_{\underline{Q}'}|/4}} \left(K^{\frac{1}{2^{d}}} (d+1)^{\frac{1}{2^{d}}} \exp\left(-\frac{K^{d}}{C_{d}}\right) \right)^{|T|} \\ &= \sum_{\substack{\underline{Q}' \in (\mathcal{Q}_{\ell_{j_{s}(c)}})^{T^{*}} \\ Q_{Q}' \subset Q \; \forall Q \in T^{*}} |T| \geq |\tilde{T}_{\underline{Q}'}|/4}} \left(\sum_{i=1}^{|T} \left| \tilde{T}_{\underline{Q}'} \right| \right) \left(K^{\frac{1}{2^{d}}} (d+1)^{\frac{1}{2^{d}}} \exp\left(-\frac{K^{d}}{C_{d}}\right) \right)^{i} \\ &= (L^{d})^{|T^{*}|} \sum_{i=|N/4|}^{N} \left(N \right) \left(K^{\frac{1}{2^{d}}} (d+1)^{\frac{1}{2^{d}}} \exp\left(-\frac{K^{d}}{C_{d}}\right) \right)^{i} \end{split}$$

where we have abbreviated $N := |\tilde{T}_{\underline{Q'}}| = \left(\frac{\lambda_{\max}}{\lambda_{\min}}\right)^d |T^*|$. Let $p_{d,K} := K^{\frac{d}{2d}} (d+1)^{\frac{1}{2d}} \exp\left(-\frac{K^d}{C_d}\right)$. Clearly $\lim_{K\to\infty} p_{d,K} = 0$, so we can pick $K_{d,M,p}$ (at this point independently of M) large enough such that for $K \ge K_{d,M,p}$ we have $p_{d,K} \le \frac{1}{32}$. Then Lemma 6.2.7 with that choice of p implies that

$$\begin{split} \zeta_{\Lambda}^{\varepsilon} \left(T^* \subset S_{K,L,M,\text{bad}}^{*,II}(\mathcal{A}) \right) &\leq (L^{\mathsf{d}})^{|T^*|} \left(\frac{1}{2} \right)^{\frac{N}{4}} \\ &= \left(L^{\mathsf{d}} \left(\frac{1}{2} \right)^{\frac{1}{4} \left(\frac{\lambda_{\text{mac}}}{\lambda_{\text{mic}}} \right)^{\mathsf{d}}} \right)^{|T^*|} \end{split}$$

For ε small enough (depending on d, *L*, and *p*) the term in brackets is less than *p*, and we obtain (6.4.18).

Step 2: Proof of (6.4.17)

We can use Lemma 6.4.7 b). Given $Q \in \mathcal{Q}_{KL\lambda_{mac}}$, there are at most

$$\left(\left\lceil 3 \frac{KL\lambda_{\max}}{\ell_{j_*(\varepsilon)}} \right\rceil + 3 \right)^{\mathsf{d}} \le \left(\frac{4KL\lambda_{\max}}{\ell_{j_*(\varepsilon)}} \right)^{\mathsf{d}}$$

cubes in $\mathcal{Q}_{\ell_{j_*(\varepsilon)}}^{\#}$ that intersect Q (we used that $\frac{KL\lambda_{mac}}{\ell_{j_*(\varepsilon)}} \ge 8$). We can now proceed as in the proof of Lemma 6.4.7 and obtain

$$\begin{split} \zeta_{\Lambda}^{\varepsilon} \left(T^* \subset S_{K,L,M,\text{bad}}^{*,I}(\mathcal{A}) \right) \\ &= \zeta_{\Lambda}^{\varepsilon} \left(\forall Q \in T^* \; \exists q_Q \in \mathcal{Q}_{\ell_{j_*(\varepsilon)}}^{\#} \; \text{with} \; q_Q \cap Q \neq \varnothing, q_Q \in S_{K,M,\text{bad}}^{(j_*(\varepsilon))}(\mathcal{A}) \right) \\ &= \zeta_{\Lambda}^{\varepsilon} \left(\forall Q \in T^* \; \exists q_Q \in \mathcal{Q}_{\ell_{j_*(\varepsilon)}}^{\#} \; \text{with} \; q_Q \cap Q \neq \varnothing \; \text{such that} \; \bigcup_{Q \in T^*} \{q_Q\} \subset S_{K,M,\text{bad}}^{(j_*(\varepsilon))}(\mathcal{A}) \right) \\ &\leq \sum_{\substack{\underline{q} \in (\mathcal{Q}_{\ell_{j_*(\varepsilon)}}^{\#})^{T^*} \\ q_Q \cap Q \neq \varnothing \; \forall Q \in T^*}} \zeta_{\Lambda}^{\varepsilon} \left(\bigcup_{Q \in T^*} \{q_Q\} \subset S_{K,M,\text{bad}}^{(j_*(\varepsilon))}(\mathcal{A}) \right) \; . \end{split}$$

If we assume for the moment that none of the cubes in T^* are l^{∞} -neighbours, then the q_Q are pairwise distinct, and so $\bigcup_{Q \in T^*} \{q_Q\}$ has cardinality $|T^*|$. Now (6.4.21) implies

$$\begin{aligned} \zeta_{\Lambda}^{\varepsilon} \left(T^{*} \subset S_{K,L,M,\text{bad}}^{*,I}(\mathcal{A}) \right) \\ &\leq \sum_{\substack{\underline{q} \in (\mathcal{Q}_{\ell_{j_{*}(\varepsilon)}}^{*})^{T^{*}} \\ q_{Q} \cap Q \neq \varnothing \ \forall Q \in T^{*}}}} \left(3^{(2^{j_{*}(\varepsilon)}-2)\mathsf{d}} M^{s_{j_{*}(\varepsilon)}\mathsf{d}} \left(K^{\frac{\mathsf{d}}{2\mathsf{d}}}(\mathsf{d}+1)^{\frac{1}{2\mathsf{d}}} \exp\left(-\frac{K^{\mathsf{d}}}{C_{\mathsf{d}}}\right) \right)^{2^{j_{*}(\varepsilon)}} \right)^{|T^{*}|} \\ &\leq \left(\left(\frac{4KL\lambda_{\max}}{\ell_{j_{*}(\varepsilon)}} \right)^{\mathsf{d}} \right)^{|T^{*}|} \left(3^{(2^{j_{*}(\varepsilon)}-2)\mathsf{d}} M^{s_{j_{*}(\varepsilon)}\mathsf{d}} \left(K^{\frac{\mathsf{d}}{2\mathsf{d}}}(\mathsf{d}+1)^{\frac{1}{2\mathsf{d}}} \exp\left(-\frac{K^{\mathsf{d}}}{C_{\mathsf{d}}}\right) \right)^{2^{j_{*}(\varepsilon)}} \right)^{|T^{*}|}. \end{aligned}$$

$$(6.4.25)$$

To remove the assumption that none of the the elements of T^* are neighbours we proceed as in Step 1 of the proof of Lemma 6.4.7 and find a subset $T_{i_*}^*$ of T^* of cardinality at least $\frac{|T^*|}{2^d}$

for which this is the case. Using (6.4.25) for $T_{i_*}^*$ instead of T^* we arrive at

$$\begin{aligned} \zeta_{\Lambda}^{\varepsilon} \left(T^{*} \subset S_{K,L,M,\text{bad}}^{*,I}(\mathcal{A}) \right) \\ &\leq \zeta_{\Lambda}^{\varepsilon} \left(T_{i_{*}}^{*} \subset S_{K,L,M,\text{bad}}^{*,I}(\mathcal{A}) \right) \\ &\leq \left(\left(\left(\frac{4KL\lambda_{\text{mac}}}{\ell_{j_{*}(\varepsilon)}} \right)^{\mathsf{d}} \right)^{\frac{|T^{*}|}{2^{\mathsf{d}}}} \left(3^{(2^{j_{*}(\varepsilon)}-2)\mathsf{d}} M^{s_{j_{*}(\varepsilon)}\mathsf{d}} \left(K^{\frac{\mathsf{d}}{2^{\mathsf{d}}}}(\mathsf{d}+1)^{\frac{1}{2^{\mathsf{d}}}} \exp\left(-\frac{K^{\mathsf{d}}}{C_{\mathsf{d}}}\right) \right)^{2^{j_{*}(\varepsilon)}} \right)^{\frac{|T^{*}|}{2^{\mathsf{d}}}}. \end{aligned}$$

$$(6.4.26)$$

It remains to bound the right hand side in (6.4.26). We begin by bounding s_j from above. We have

$$s_j = j^3 + \sum_{m=0}^j m^3 2^{j-m} = 2^j \left(\frac{j^3}{2^j} + \sum_{m=0}^j \frac{m^3}{2^m}\right) \le 2^j \left(4 + \sum_{m=0}^\infty \frac{m^3}{2^m}\right) = 30 \cdot 2^j$$

and so (6.4.26) implies that

$$\begin{split} \zeta_{\Lambda}^{\varepsilon} \left(T^* \subset S_{K,M,\text{bad}}^{*,I}(\mathcal{A}) \right) &\leq \left(\left(\frac{4KL\lambda_{\text{mac}}}{\ell_{j_*(\varepsilon)}} \left(3M^{30} K^{\frac{1}{2^d}} (\mathsf{d}+1)^{\frac{1}{d2^d}} \exp\left(-\frac{K^{\mathsf{d}}}{C_{\mathsf{d}}}\right) \right)^{2^{j_*(\varepsilon)}} \right)^{\frac{\mathsf{d}}{2^d}} \right)^{|T^*|} \\ &= \left(\left(\frac{4KL\lambda_{\text{mac}}}{\ell_{j_*(\varepsilon)}} p_{\mathsf{d},K,M}^{2^{j_*(\varepsilon)}} \right)^{\frac{\mathsf{d}}{2^d}} \right)^{|T^*|} \end{split}$$

where we have abbreviated $p_{d,K,M} = 3M^{30}K^{\frac{1}{2d}}(d+1)^{\frac{1}{d^{2d}}}\exp\left(-\frac{K^d}{C_d}\right)$. Note for each fixed M and d we have $\lim_{K\to\infty} p_{d,K,M} = 0$, and so we can pick $K_{d,M,p}$ such that $p_{d,K,M} \leq \frac{1}{2}$ for $K \geq K_{d,M,p}$.

For these choices of *K* we then know that

$$\zeta_{\Lambda}^{\varepsilon}\left(T^{*} \subset S_{K,L,M,\text{bad}}^{*,I}(\mathcal{A})\right) \leq \left(\left(\frac{4KL\lambda_{\max}}{\ell_{j_{*}(\varepsilon)}}2^{-2^{j_{*}(\varepsilon)}}\right)^{\frac{d}{2d}}\right)^{|T^{*}|}$$

and we only need to show that

$$\frac{4KL\lambda_{\max}}{\ell_{j_*(\varepsilon)}} 2^{-2^{j_*(\varepsilon)}} \le p^{\frac{2^d}{d}}$$
(6.4.27)

when ε is small enough. To show this, we need to bound $j_*(\varepsilon)$ from below. By definition, $j_*(\varepsilon)$ is equal to the largest integer j such that $\ell_j \leq \frac{KL\lambda_{\text{mac}}}{8}$. In particular, $\ell_{j_*(\varepsilon)+1} > \frac{KL\lambda_{\text{mac}}}{8}$, i.e. $M^{(j_*(\varepsilon)+1)^3}K\lambda_{\text{mic}} > \frac{KL\lambda_{\text{mac}}}{8}$. Estimating $(j_*(\varepsilon)+1)^3 \leq (2j_*(\varepsilon))^3$, we conclude

$$j_*(\varepsilon) \geq \frac{1}{2} \sqrt[3]{\log_M \frac{L\lambda_{\max}}{8\lambda_{\min}}}$$

Let us also abbreviate $X_{d,\epsilon,L} = \frac{L\lambda_{mac}}{8\lambda_{mic}}$, and observe that $\lim_{\epsilon \to 0} X_{d,\epsilon,L} = \infty$. For *t* sufficiently large we have $\frac{1}{2}\sqrt[3]{\log_M t} \ge \log_2 \log_2(t^2)$. This means that for $X_{d,\epsilon,L}$ sufficiently large (i.e. ϵ

sufficiently small) we have $j_*(\varepsilon) \ge \frac{1}{2} \sqrt[3]{\log_M X_{d,\varepsilon,L}} \ge \log_2 \log_2(X_{d,\varepsilon,L}^2)$. Using this and the rather crude estimate $\ell_{j_*(\varepsilon)} \ge \ell_{j_0} = K\lambda_{\text{mic}}$ for the denominator, we find

$$\frac{4KL\lambda_{\max}}{\ell_{j_*(\varepsilon)}} 2^{-2^{j_*(\varepsilon)}} \leq 32X_{\mathsf{d},\varepsilon,L} 2^{-2^{\log_2\log_2(X_{\mathsf{d},\varepsilon,L}^2)}} \leq \frac{32X_{\mathsf{d},\varepsilon,L}}{X_{\mathsf{d},\varepsilon,L}^2} = \frac{32}{X_{\mathsf{d},\varepsilon,L}}$$

for ε sufficiently small. This clearly implies that (6.4.27) holds for ε sufficiently small, which is (6.4.18).

Step 3: Proof of (6.4.19)

We can assume without loss of generality that $p \leq \frac{1}{4}$, as otherwise the estimate is trivial. Using (6.4.17) and (6.4.18) we see that

$$\begin{split} &\zeta_{\Lambda}^{\varepsilon} \left(T^{*} \subset S_{K,L,M,\text{bad}}^{*}(\mathcal{A}) \right) \\ &\leq \sum_{\substack{T_{I}^{*} \cup T_{II}^{*} = T^{*} \\ T_{I}^{*} \cup T_{II}^{*} = T^{*} \\ \zeta_{\Lambda}^{\varepsilon} \left(T_{I}^{*} \subset S_{K,L,M,\text{bad}}^{*,I}(\mathcal{A}), T_{II}^{*} \subset S_{K,L,M,\text{bad}}^{*,II}(\mathcal{A}) \right) \\ &\leq \sum_{\substack{T_{I}^{*} \subset T^{*} \\ |T_{I}^{*}| \geq |T^{*}|/2 \\ \\ &\leq 2 \sum_{i=\lceil |T^{*}|/2 \rceil}^{|T^{*}|} \binom{|T^{*}|}{i} p^{i} \\ &\leq 2(4p)^{\frac{|T^{*}|}{2}} \end{split}$$

where we have used Lemma 6.2.7 in the last step.

Using Lemma 6.4.6 we can now estimate the probability that we find sets U_j as in Lemma 6.4.3.

Lemma 6.4.8. Let $d \ge 4$, and $\Lambda \Subset \mathbb{Z}^d$. Let $M \ge 12$ be an odd integer. Then there is $K_{d,M}$ depending on d, M only with the following property: Let $K \ge K_{d,M}$ be an odd multiple of 3, let L be an odd integer. Let $U \in \mathcal{P}_{KL\lambda_{mac}}$ be a polymer consisting of $n = \frac{|U|}{(KL)^d \lambda_{mac}^d}$ boxes. Let $k \ge 0$ be an integer and let $\Omega_{U,k}$ be the event that there exist $U_0, \ldots, U_k \in \mathcal{P}_{KL\lambda_{mac}}$ such that $U \subset U_0$, for $j \in \{0, \ldots, k-1\}$ we have $U_j + Q_{KL\lambda_{mac}}(0) \subset U_{j+1}, U_k \subset U + Q_{2kKL\lambda_{mac}}(0)$, and

$$\{Q \in \mathcal{Q}_{KL\lambda_{\max}} \colon Q \subset (U_j + Q_{KL\lambda_{\max}}(0)) \setminus U_j\} \cap S^*_{K,L,M,\mathsf{bad}}(\mathcal{A}) = \varnothing$$

Then, if ε is small enough (depending on K, L and d), we have

$$\zeta^{\varepsilon}_{\Lambda}(\Omega_{U,k}) \ge 1 - \frac{n}{2^k} \,. \tag{6.4.28}$$

Proof. Let p > 0 be a constant to be chosen later (depending on d only). We pick $K_{d,M} \ge K_{d,M,p}$ with the $K_{d,M,p}$ from Lemma 6.4.6 so that this lemma can be applied.

We try to define the U_j using a greedy algorithm. That is, we define U_j as the union of all cubes $Q \in Q_{KL\lambda_{mac}}$ that can be connected to U by a non-selfintersecting l^{∞} -path of cubes in $Q_{KL\lambda_{mac}}$ that contains at most j non-bad cubes. More precisely, $Q \in Q_{KL\lambda_{mac}}$ is a subset of U if and only if there are $l \ge 0$ and $Q^{(0)} = Q, Q^{(1)}, \ldots, Q^{(l)} \subset U \in Q_{KL\lambda_{mac}}$ pairwise disjoint, with $d_{\infty}(Q^{(i)}, Q^{(i+1)}) \le 1$ for all $i \in \{0, \ldots, l-1\}$, such that at most j of $Q^{(0)}, Q^{(1)}, \ldots, Q^{(l-1)}$ are not in $S_{K,L,M,\text{bad}}^*(\mathcal{A})$.
This definition ensures that all l^{∞} -neighbouring cubes to U_j are not in $S^*_{K,L,M,\text{bad}}(\mathcal{A})$. So one sees that the U_j satisfies all the conditions from $\Omega_{U,k}$ except that we do not yet know whether $U_k \subset U + Q_{2kKL\lambda_{\text{max}}}(0)$. This means that

$$1 - \zeta_{\Lambda}^{\varepsilon}(\Omega_{U,k}) \le \zeta_{\Lambda}^{\varepsilon} \left(U_k \not\subset U + Q_{2kKL\lambda_{\max}}(0) \right)$$
(6.4.29)

and so it suffices to estimate the latter probability.

To do so, we define $\Pi_{U,k}$ to be the set of non-selfintersecting l^{∞} -nearest neighbour paths $\Psi = \left(Q^{(0)} = Q, Q^{(1)}, \dots, Q^{(l)}\right)$ of cubes, that connect a cube Q outside of $Q_{2kKL\lambda_{mac}}(0)$ with $Q^{(l)} \subset U$. For $\Psi = \left(Q^{(0)} = Q, Q^{(1)}, \dots, Q^{(l)}\right)$ let $\tilde{\Psi} = \left\{Q^{(0)}, Q^{(1)}, \dots, Q^{(l)}\right\}$ be the set of cubes in Ψ , and let $|\Psi| = |\tilde{\Psi}| = l + 1$ be the number of cubes in it.

If $U_k \not\subset U + Q_{2kKL\lambda_{mac}}(0)$, then there is some $\Psi \in \Pi_{U,k}$ that contains at most k cubes within $Q^{(0)}, \ldots, Q^{(l-1)}$ (and thus at most k + 1 cubes within the cubes in Ψ) that are not bad. Because Ψ connects U with a cube outside of $U + Q_{2kKL\lambda_{mac}}(0)$, we have $|\Psi| \ge 2k + 2$. We can now continue (6.4.29) by using a union bound over all $\Psi \in \Pi_{U,k}$, and later over all bad subsets of $\tilde{\Psi}$, and obtain using Lemma 6.4.6 that

$$\begin{split} 1 - \zeta_{\Lambda}^{\varepsilon}(\Omega_{U,k}) &\leq \zeta_{\Lambda}^{\varepsilon} \left(\exists \Psi \in \Pi_{U,k} \colon \left| \tilde{\Psi} \setminus S_{K,L,M,\text{bad}}^{*}(\mathcal{A}) \right| \leq k+1 \right) \\ &\leq \sum_{\Psi \in \Pi_{U,k}} \zeta_{\Lambda}^{\varepsilon} \left(\left| \tilde{\Psi} \setminus S_{K,L,M,\text{bad}}^{*}(\mathcal{A}) \right| \leq k+1 \right) \\ &= \sum_{\Psi \in \Pi_{U,k}} \zeta_{\Lambda}^{\varepsilon} \left(\exists T_{\Psi}^{*} \subset \tilde{\Psi} \colon T_{\Psi}^{*} \subset S_{K,L,M,\text{bad}}^{*}(\mathcal{A}), |T_{\Psi}^{*}| \geq |\Psi| - k - 1 \right) \\ &\leq \sum_{\Psi \in \Pi_{U,k}} \sum_{\substack{T_{\Psi}^{*} \subset \tilde{\Psi} \\ |T_{\Psi}^{*}| \geq |\Psi| - k - 1}} \zeta_{\Lambda}^{\varepsilon} \left(T_{\Psi}^{*} \subset S_{K,L,M,\text{bad}}^{*}(\mathcal{A}) \right) \\ &\leq \sum_{\Psi \in \Pi_{U,k}} \sum_{\substack{T_{\Psi}^{*} \subset \tilde{\Psi} \\ |T_{\Psi}^{*}| \geq |\Psi| - k - 1}} 2(4p)^{\frac{|T_{\Psi}^{*}|}{2}} \\ &\leq \sum_{\Psi \in \Pi_{U,k}} 2(4p)^{\frac{|\Psi| - k - 1}{2}} . \end{split}$$

We can reorganize this expression by summing over the lengths of Ψ . Recall that this length needs to be at least 2k + 2, and note that there are at most $n(2d)^l$ paths in $\Pi_{U,k}$ of length l + 1. Thus, we obtain

$$\begin{split} 1 - \zeta_{\Lambda}^{\varepsilon}(\Omega_{U,k}) &\leq \sum_{l=2k+1}^{\infty} \sum_{\substack{\Psi \in \Pi_{U,k} \\ |\tilde{\Psi}| = l+1}} 2(4p)^{\frac{l-k}{2}} \\ &\leq \sum_{l=2k+1}^{\infty} n(2\mathsf{d})^{l} 2(4p)^{\frac{l-k}{2}} \\ &= \frac{2n}{(2\sqrt{p})^{k}} \sum_{l=2k+1}^{\infty} (4\mathsf{d}\sqrt{p})^{l} \end{split}$$

We choose $p \le \frac{1}{144d^2}$, so that $4d\sqrt{p} \le \frac{1}{3}$. Then, in particular, the series on the right-hand side converges, and we can continue

$$1-\zeta^{\varepsilon}_{\Lambda}(\Omega_{U,k}) \leq \frac{2n}{(2\sqrt{p})^k} \frac{(4\mathsf{d}\sqrt{p})^{2k+1}}{1-4\mathsf{d}\sqrt{p}}$$

$$\leq \frac{n(4\mathsf{d}\sqrt{p})^{2k}}{(2\sqrt{p})^k}$$
$$= n(8\mathsf{d}^2\sqrt{p})^k.$$

We finalize our choice of p as $p = \frac{1}{256d^4}$. Then (6.4.28) follows.

Proof of Theorem 6.4.1. We want to combine Lemma 6.4.5 and Lemma 6.4.8. That is, we first choose *M* so large that Lemma 6.4.5 can be applied. Then we choose *K* large enough that Lemma 6.4.8 can be applied. Then Lemma 6.4.5 applies for sufficiently large *L*. We choose $\hat{N}_{d} = KL$, and note that Lemma 6.4.8 implies the bound on the probability of $\Omega_{U,k}$.

It remains to check that (6.4.15) and (6.4.16) imply (6.4.1) and (6.4.2) if $\Omega_{U,k}$ holds. This follows from the observation that $(U_0 + Q_{\hat{N}_d\lambda_{mac}}(0)) \setminus U_0 \subset (U + Q_{2k\hat{N}_d\lambda_{mac}}(0)) \setminus U$ and $(U_{k-1} + Q_{\hat{N}_d\lambda_{mac}}(0)) \setminus U_{k-1} \subset (U + Q_{2k\hat{N}_d\lambda_{mac}}(0)) \setminus U$.

Remark 6.4.9. Let us comment on why the lengthscales $\ell_j = M^{j^3} K \lambda_{\text{mic}}$ are a natural choice. For the construction in Step 2 of the proof of Lemma 6.4.2 we need that $\log \frac{\ell_j}{\ell_{j+1}}$ is summable as otherwise we could not bound $\|\nabla_1^2 \eta\|_{L^{\infty}}$ in (6.4.6). This means that ℓ_j needs to grow rather fast (e.g., $\ell_j = M^{j^2} K \lambda_{\text{mic}}$ would not be fast enough). On the other hand, for the estimate on the probability of bad cubes of type I in Lemma 6.4.6 we need that the exponent s_j d of M in (6.4.21) is at most $C_d 2^j$. This exponent arises from the combinatorial factors $\left(\frac{\ell_j}{\ell_{j-1}}\right)^{2d}$ in (6.4.24). This means that ℓ_j cannot grow too fast (e.g., $\ell_j = M^{2^j} K \lambda_{\text{mic}}$ would be too fast).

Fortunately, both requirements are compatible, and in fact, our choice $\ell_j = M^{j^3} K \lambda_{\text{mic}}$ satisfies both of them.

6.5 Pathwise bounds on the field

We can now turn to the proof of Theorem 6.1.5 and of the second part of Theorem 6.1.2. Before we actually give the proofs, however, we state and prove various quenched estimates for $G_{\Lambda\setminus A}$ that hold for all A, or at least up to exponentially small probability in A. The main tool for that will be Theorem 6.4.1.

We prove those estimates in Section 6.5.1. Then, in Sections 6.5.2 and 6.5.3 we use them to deduce Theorem 6.1.5 and the second part of Theorem 6.1.2, respectively.

6.5.1 Quenched estimates on the Green's function

We write $G_{\Lambda,y}$ for $G_{\Lambda}(\cdot, y)$. We have the following straightforward result for G_{Λ} . This is essentially the same as Lemma 2.8.1 or Lemma 4.4.2.

Lemma 6.5.1. Let $\Lambda \Subset \mathbb{Z}^d$ and $x, y \in \mathbb{Z}^d$. Then

$$G_{\Lambda}(x,y) = \left(\nabla_1^2 G_{\Lambda,x}, \nabla_1^2 G_{\Lambda,y}\right)_{L^2(\mathbb{Z}^d)} .$$
(6.5.1)

Furthermore, we have

$$|G_{\Lambda}(x,y)| \le \sqrt{G_{\Lambda}(x,x)G_{\Lambda}(y,y)}.$$
(6.5.2)

Proof. For (6.5.1) we calculate

$$G_{\Lambda}(x,y) = \left(\mathbb{1}_{\cdot=x}, G_{\Lambda,y}\right)_{L^{2}(\mathbb{Z}^{d})} = \left(\Delta_{1}^{2}G_{\Lambda,x}, G_{\Lambda,y}\right)_{L^{2}(\mathbb{Z}^{d})} = \left(\nabla_{1}^{2}G_{\Lambda,x}, \nabla_{1}^{2}G_{\Lambda,y}\right)_{L^{2}(\mathbb{Z}^{d})}.$$

The estimate (6.5.2) follows directly from the interpretation of G_{Λ} as a covariance. Alternatively, we can use (6.5.1) together with the Cauchy-Schwarz inequality to estimate

$$\begin{aligned} |G_{\Lambda}(x,y)| &= \left| \left(\nabla_{1}^{2} G_{\Lambda,x}, \nabla_{1}^{2} G_{\Lambda,y} \right)_{L^{2}(\mathbb{Z}^{d})} \right| \\ &\leq \left\| \nabla_{1}^{2} G_{\Lambda,x} \right\|_{L^{2}(\mathbb{Z}^{d})} \left\| \nabla_{1}^{2} G_{\Lambda,y} \right\|_{L^{2}(\mathbb{Z}^{d})} \\ &= \sqrt{G_{\Lambda}(x,x) G_{\Lambda}(y,y)} \,. \end{aligned}$$

Next, we establish some quenched tail estimates on $G_{\Lambda \setminus A}(x, x)$. If $d \ge 5$, then there are deterministic bounds on $G_{\Lambda \setminus A}(x, x)$ by (6.2.5), so this is only interesting if d = 4.

Lemma 6.5.2. If d = 4, there is a constant $\tilde{\gamma} > 0$ such that if $\Lambda \in \mathbb{Z}^d$, $x \in \Lambda$ and ε is small enough (depending on d only) then for any $t \geq \tilde{\gamma}$ we have

$$\zeta_{\Lambda}^{\varepsilon} \left(G_{\Lambda \setminus \mathcal{A}}(x, x) \le t \right) \ge 1 - \exp\left(-\frac{\varepsilon \exp(16\pi^2(t - \tilde{\gamma}))}{C |\log \varepsilon|^{\frac{1}{2}}} \right), \tag{6.5.3}$$

and for $\alpha > 0$, $x \in \Lambda$ with $d(x, \mathbb{Z}^d \setminus \Lambda) \ge \varepsilon^{-\alpha}$ and $0 \le t \le \frac{1}{8\pi^2} \log \left(1 + d(x, \mathbb{Z}^d \setminus \Lambda) - \varepsilon^{-\alpha}\right) - \tilde{\gamma}$ we have

$$\zeta_{\Lambda}^{\varepsilon} \left(G_{\Lambda \setminus \mathcal{A}}(x, x) \le t \right) \le 1 - \exp\left(-\frac{C_{\alpha} \varepsilon \exp(32\pi^2(t + \tilde{\gamma}))}{|\log \varepsilon|^{\frac{1}{2}}} \right)$$
(6.5.4)

for some constant C.

Furthermore, if $d \ge 4$, $k \in \mathbb{N}$ *, and* $y \in \Lambda$ *there are constants* $\tilde{\gamma}_d$ *such that*

$$\zeta_{\Lambda}^{\varepsilon} \left(G_{\Lambda \setminus \mathcal{A}}(y, y) \le \mathbb{1}_{\mathsf{d}=4} \frac{\log k + |\log \varepsilon|}{16\pi^2} + \tilde{\gamma}_{\mathsf{d}} \right) \ge 1 - \frac{1}{2^k}.$$
(6.5.5)

Proof. We begin with (6.5.3). This follows easily from Lemma 6.2.4 and Theorem 6.1.3 c). Indeed, if $x \in A$ then $G_{\Lambda \setminus A}(x, x) = 0$, while if $x \notin A$ we know from (6.2.5) that

$$G_{\Lambda\setminus\mathcal{A}}(x,x) \leq \frac{1}{4\pi^2}\log(1+d(x,\tilde{\mathcal{A}})) + C \leq \frac{1}{4\pi^2}\log(d(x,\tilde{\mathcal{A}})) + C \leq \frac{1}{4\pi^2}\log(d(x,\mathcal{A})) + C.$$

So, there is a constant $\tilde{\gamma}'$ such that $G_{\Lambda \setminus A}(x, x) > t$ for $t \geq \tilde{\gamma}'$ implies $d(x, A) \geq \exp(4\pi^2(t - \tilde{\gamma}'))$. Using (6.1.7) we can estimate that

$$\begin{split} \zeta_{\Lambda}^{\varepsilon} \left(G_{\Lambda \setminus \mathcal{A}}(x,x) \leq t \right) &\geq \zeta_{\Lambda}^{\varepsilon} \left(d(x,\mathcal{A}) \leq \exp(4\pi^2(t-\tilde{\gamma}')) \right) \\ &= 1 - \zeta_{\Lambda}^{\varepsilon} \left(\mathcal{A} \cap Q_{\exp(4\pi^2(t-\tilde{\gamma}'))}(x) = \varnothing \right) \\ &\geq 1 - (1 - p_{4,-})^{|Q_{\exp(4\pi^2(t-\tilde{\gamma}'))}(x)|} \\ &\geq 1 - \exp\left(- \frac{p_{4,-} \exp(4\pi^2(t-\tilde{\gamma}'))^4}{C} \right) \\ &\geq 1 - \exp\left(- \frac{\varepsilon \exp(16\pi^2(t-\tilde{\gamma}'))}{C|\log\varepsilon|^{\frac{1}{2}}} \right) \end{split}$$

which is (6.5.3), if we choose $\tilde{\gamma} \geq \tilde{\gamma}'$.

The argument for (6.5.4) is similar. We have that

$$G_{\Lambda \setminus \mathcal{A}}(x, x) \geq \frac{1}{8\pi^2} \log(1 + d(x, \tilde{\mathcal{A}})) - C \geq \frac{1}{8\pi^2} \log(d(x, \tilde{\mathcal{A}})) - C$$

if $x \notin A$ and $G_{\Lambda \setminus A}(x, x) = 0$ if $x \in A$. So there is a constant $\tilde{\gamma}''$ such that $G_{\Lambda \setminus A}(x, x) \leq t$ implies $d(x, \tilde{A}) \leq \exp(8\pi^2(t + \tilde{\gamma}''))$. Our assumption $t \leq \frac{1}{8\pi^2}\log(1 + d(x, \mathbb{Z}^d \setminus \Lambda) - \varepsilon^{-\alpha}) - \tilde{\gamma}$ ensures that $\exp(8\pi^2(t + \gamma'')) \leq 1 + d(x, \mathbb{Z}^d \setminus \Lambda) - \varepsilon^{-\alpha}$. This means that $Q_{\exp(8\pi^2(t + \tilde{\gamma}''))}(x)$ still has distance at least $\varepsilon^{-\alpha}$ from $\mathbb{Z}^d \setminus \Lambda$ (and in particular $d(x, \tilde{A}) < d(x, \mathbb{Z}^d \setminus \Lambda)$, so that $d(x, \tilde{A}) < d(x, A)$. Thus, we can apply (6.1.8) and obtain

$$\begin{split} \zeta_{\Lambda}^{\varepsilon} \left(G_{\Lambda \setminus \mathcal{A}}(x,x) \leq t \right) &\leq \zeta_{\Lambda}^{\varepsilon} \left(d(x,\mathcal{A}) \leq \exp(8\pi^{2}(t+\tilde{\gamma}'')) \right) \\ &= 1 - \zeta_{\Lambda}^{\varepsilon} \left(\mathcal{A} \cap Q_{\exp(8\pi^{2}(t+\tilde{\gamma}''))}(x) = \varnothing \right) \\ &\leq 1 - \exp\left(- \frac{C_{\alpha}\varepsilon \exp(32\pi^{2}(t+\tilde{\gamma}''))}{|\log \varepsilon|^{\frac{1}{2}}} \right) \end{split}$$

This is (6.5.4), if we choose $\tilde{\gamma} \geq \tilde{\gamma}''$.

Regarding (6.5.5), note that if d \geq 5 this is a trivial consequence of (6.2.5), while if d = 4 we can consider the choice $t = \frac{\log k + |\log \varepsilon|}{16\pi^2} + \tilde{\gamma}$ in (6.5.3) to obtain

$$\begin{split} \zeta^{\varepsilon}_{\Lambda} \left(G_{\Lambda \setminus \mathcal{A}}(y, y) \leq \frac{\log k + |\log \varepsilon|}{16\pi^2} + \tilde{\gamma} \right) \geq 1 - \exp\left(-\frac{\varepsilon \exp(16\pi^2 \frac{\log k + |\log \varepsilon|}{16\pi^2})}{C |\log \varepsilon|^{\frac{1}{2}}} \right) \\ \leq 1 - \exp\left(-\frac{k}{C |\log \varepsilon|^{\frac{1}{2}}} \right) \end{split}$$

and the right-hand side is at least $1 - \frac{1}{2^k}$ if ε is small enough.

Next, we prove quenched bounds on the covariance.

Lemma 6.5.3. Let $d \ge 4$, $\Lambda \Subset \mathbb{Z}^d$, and $x, y \in \Lambda$. Then, if ε is small enough (depending on d), we have

$$\zeta_{\Lambda}^{\varepsilon} \left(|G_{\Lambda \setminus \mathcal{A}}(x, y)| \le C_{\mathsf{d}} \exp\left(-\frac{|x - y|_{\infty}}{C_{\mathsf{d}} \lambda_{\max}}\right) \frac{1 + \mathbb{1}_{\mathsf{d} = 4} |\log \varepsilon|^{5/4}}{\varepsilon^{1/2}} \right) \ge 1 - \exp\left(-\frac{|x - y|_{\infty}}{C_{\mathsf{d}} \lambda_{\max}}\right) \tag{6.5.6}$$

for some constant C_d .

Proof. By translating Λ and \mathcal{A} we can assume y = 0. This ensures in particular that y is in the centre of a box in \mathcal{Q}_l for any l. Let $U = Q_{\hat{N}_d \lambda_{mac}}(0)$ with the \hat{N}_d from Theorem 6.4.1, and consider for now the case that $|x - y|_{\infty} \ge 8\hat{N}_d \lambda_{mac}$. Let $k = \left\lceil \frac{|x - y|_{\infty}}{8\hat{N}_d \lambda_{mac}} \right\rceil$, and note that $k \le \frac{|x - y|_{\infty}}{4\hat{N} \cdot 1}$.

 $k \leq \frac{|x-y|_{\infty}}{4\hat{N}_d\lambda_{mac}}$. Assume that $A \in \Omega_{U,k}$ with the $\Omega_{U,k}$ from Theorem 6.4.1. Then that theorem (applied to $G_{\Lambda\setminus A,y}$) and (6.5.1) imply that

$$\begin{aligned} \left\| \nabla_1^2 G_{\Lambda \setminus A, y} \right\|_{L^2(\mathbb{Z}^d \setminus (U + Q_{2k\hat{N}_d \lambda_{\max}}(0)))}^2 &\leq \frac{1}{2^k} \left\| \nabla_1^2 G_{\Lambda \setminus A, y} \right\|_{L^2(\mathbb{Z}^d \setminus U)}^2 \\ &= \frac{1}{2^k} G_{\Lambda \setminus A}(y, y) \,. \end{aligned}$$
(6.5.7)

Furthermore, suppose that $A \in \Omega_{x,k}$ with the $\Omega_{x,k}$ from Lemma 6.3.2. Then we can conclude

$$|G_{\Lambda\setminus A}(x,y)|^{2} = |G_{\Lambda\setminus A,y}(x)|^{2}$$

$$\leq C_{\mathsf{d}} \frac{k^{\mathsf{d}} \left(1 + \mathbb{1}_{\mathsf{d}=4}(\log k + |\log \varepsilon|^{3/2})\right)}{\varepsilon} \|\nabla_{1}^{2}G_{\Lambda\setminus A,y}\|_{L^{2}(Q_{kN_{\mathsf{d}}\lambda_{\mathsf{mic}}}(x))}^{2}.$$
(6.5.8)

For ε small enough (depending on d) we have $N_d \lambda_{mic} \le \hat{N}_d \lambda_{mac}$. Then $U + Q_{2k\hat{N}_d \lambda_{mac}}(0) = Q_{(2k+1)\hat{N}_d \lambda_{mac}}(0)$ and $Q_{kN_d \lambda_{mic}}(x)$ are disjoint, and so we can combine (6.5.7) and (6.5.8) into

$$\begin{split} |G_{\Lambda\backslash A}(x,y)|^{2} &\leq C_{\mathsf{d}} \frac{k^{\mathsf{d}} \left(1 + \mathbb{1}_{\mathsf{d}=4}(\log k + |\log \varepsilon|^{3/2})\right)}{\varepsilon} \|\nabla_{1}^{2}G_{\Lambda\backslash A,y}\|_{L^{2}(Q_{kN_{\mathsf{d}}\lambda_{\mathsf{mic}}}(x))}^{2} \\ &\leq C_{\mathsf{d}} \frac{k^{\mathsf{d}} \left(1 + \mathbb{1}_{\mathsf{d}=4}(\log k + |\log \varepsilon|^{3/2})\right)}{\varepsilon} \left\|\nabla_{1}^{2}G_{\Lambda\backslash A,y}\right\|_{L^{2}(\mathbb{Z}^{\mathsf{d}}\setminus(U+Q_{2k\hat{N}_{\mathsf{d}}\lambda_{\mathsf{mac}}}(0)))}^{2} \\ &\leq C_{\mathsf{d}} \frac{k^{\mathsf{d}} \left(1 + \mathbb{1}_{\mathsf{d}=4}(\log k + |\log \varepsilon|^{3/2})\right)}{2^{k}\varepsilon} G_{\Lambda\backslash A}(y,y) \,. \end{split}$$

$$(6.5.9)$$

Next, let $\tilde{\Omega}_{y,k}$ be the event from (6.5.5). If $A \in \tilde{\Omega}_{y,k}$, then (6.5.5) and (6.5.9) imply

$$\begin{aligned} |G_{\Lambda\setminus A}(x,y)|^{2} &\leq C_{d} \frac{k^{d} \left(1 + \mathbb{1}_{d=4} (\log k + |\log \varepsilon|^{3/2})\right)}{2^{k} \varepsilon} \left(1 + \mathbb{1}_{d=4} (\log k + |\log \varepsilon|)\right) \\ &\leq C_{d} \left(\frac{3}{4}\right)^{k} \frac{1 + \mathbb{1}_{d=4} |\log \varepsilon|^{5/2}}{\varepsilon} \\ &\leq C_{d} \exp\left(-\log \frac{3}{4} \frac{|x - y|_{\infty}}{4\hat{N}_{d}\lambda_{mac}}\right) \frac{1 + \mathbb{1}_{d=4} |\log \varepsilon|^{5/2}}{\varepsilon} \\ &\leq C_{d} \exp\left(-\frac{|x - y|_{\infty}}{C_{d}\lambda_{mac}}\right) \frac{1 + \mathbb{1}_{d=4} |\log \varepsilon|^{5/2}}{\varepsilon}. \end{aligned}$$
(6.5.10)

This estimate holds if $A \in \Omega_{U,k} \cap \Omega_{x,k} \cap \tilde{\Omega}_{y,k}$. But that probability is easy to bound:

$$\zeta^{\varepsilon}_{\Lambda}\left(\Omega_{U,k}\cap\Omega_{x,k}\cap\tilde{\Omega}_{y,k}\right) \geq 1 - \frac{1}{2^{k}} - \frac{1}{2^{k^{\mathsf{d}}}} - \frac{1}{2^{k}} \geq 1 - \exp\left(\frac{k}{C_{\mathsf{d}}}\right) \geq 1 - \exp\left(-\frac{|x-y|_{\infty}}{C_{\mathsf{d}}\lambda_{\max}}\right) \,.$$

Therefore we have shown that the set of *A* for which (6.5.10) holds has measure at least $1 - \exp\left(-\frac{|x-y|_{\infty}}{C_{d}\lambda_{mac}}\right)$, and this implies (6.5.6).

It remains to consider the case that $|x - y|_{\infty} < 8\hat{N}_{d}\lambda_{mac}$. In that case we need to show

$$\zeta^{\varepsilon}_{\Lambda}\left(|G_{\Lambda\setminus\mathcal{A}}(x,y)| \leq C_{\mathsf{d}}\frac{1+\mathbb{1}_{\mathsf{d}=4}|\log\varepsilon|^{5/2}}{\varepsilon}\right) \geq c_{\mathsf{d}}.$$

This follows immediately from (6.5.5) and (6.5.2).

We also need to quantify that for a large domain Λ the covariances far inside Λ depend only weakly on the precise shape of Λ .

Lemma 6.5.4. Let $d \ge 4$, $\Lambda' \subset \Lambda \Subset \mathbb{Z}^d$. Let ε be small enough (depending on d only). Suppose that r, R are integers with $\hat{N}_d \lambda_{mac} \le r$, $8r \le R$ and $Q_R(0) \subset \Lambda'$. we have

$$\begin{aligned} \zeta_{\Lambda}^{\varepsilon} \left(\max_{x,y \in Q_{r}(0)} \left| G_{\Lambda \setminus \mathcal{A}}(x,y) - G_{\Lambda' \setminus \mathcal{A}}(x,y) \right| &\leq C_{\mathsf{d}} \exp\left(-\frac{R-r}{C_{\mathsf{d}}\lambda_{\max}} \right) \frac{1 + \mathbb{1}_{\mathsf{d}=4} |\log \varepsilon|^{5/4}}{\varepsilon^{1/2}} \right) \\ &\geq 1 - C_{\mathsf{d}} r^{\mathsf{d}} \exp\left(-\frac{R-r}{C_{\mathsf{d}}\lambda_{\max}} \right) \,. \end{aligned}$$

$$(6.5.11)$$

Proof. The idea is that $H_y := G_{\Lambda \setminus A, y} - G_{\Lambda' \setminus A, y}$ is biharmonic in $Q_R(0)$. We will use Theorem 6.4.1 a) to conclude that the L^2 -norm of $\nabla_1^2 H$ outside of $Q_{R/2}(0)$ is exponentially small, and then use Theorem 6.4.1 b) to conclude that the L^2 -norm of $\nabla_1^2 H$ in $Q_r(0)$ is exponentially small. Of course these estimates hold not for all realizations of A, but we will estimate that they hold for sufficiently many.

Let $\tilde{r} = \left\lceil \frac{r}{\hat{N}_{d}\lambda_{mac}} \right\rceil \hat{N}_{d}\lambda_{mac}$ and $\tilde{R} = \left\lfloor \frac{R}{\hat{N}_{d}\lambda_{mac}} \right\rfloor \hat{N}_{d}\lambda_{mac}$. Then $r \leq \tilde{r} \leq 2r$, $\frac{R}{2} \leq \tilde{R} \leq R$. We let $U = Q_{\tilde{r}}(0)$ and note that $U \in \mathcal{P}_{\hat{N}_{d}\lambda_{mac}}$ is a polymer consisting of $\left(\frac{\tilde{r}}{\hat{N}_{d}\lambda_{mac}}\right)^{d}$ boxes in $\mathcal{Q}_{\hat{N}_{d}\lambda_{mac}}$. Let $k = \left\lfloor \frac{\tilde{R} - \tilde{r}}{4\hat{N}_{d}\lambda_{mac}} \right\rfloor$ and note that $k \geq \frac{R - r}{C_{d}\lambda_{mac}}$. Theorem 6.4.1 a) implies that on the event $\Omega_{U,k}$ we have

$$\begin{split} \left\| \nabla_1^2 G_{\Lambda \setminus A, y} \right\|_{L^2(\mathbb{Z}^d \setminus (U + Q_{2k\hat{N}_d \lambda_{\max}}(0)))}^2 &\leq \frac{1}{2^k} \left\| \nabla_1^2 G_{\Lambda \setminus A, y} \right\|_{L^2(\mathbb{Z}^d \setminus U)}^2 \\ &\leq \frac{1}{2^k} \left\| \nabla_1^2 G_{\Lambda \setminus A, y} \right\|_{L^2(\mathbb{Z}^d)}^2 \\ &= \frac{1}{2^k} G_{\Lambda \setminus A}(y, y) \end{split}$$

as $G_{\Lambda \setminus A,y} = 0$ on $\tilde{A} \setminus U$ and $G_{\Lambda \setminus A,y} \Delta_1^2 G_{\Lambda \setminus A,y} = 0$ on $\mathbb{Z}^d \setminus U$. Analogously we have

$$\left\|\nabla_1^2 G_{\Lambda' \setminus A, y}\right\|_{L^2(\mathbb{Z}^d \setminus (U+Q_{2k\hat{N}_d \lambda_{\max}}(0)))}^2 \leq \frac{1}{2^k} G_{\Lambda' \setminus A}(y, y)$$

as $G_{\Lambda \setminus A,y} = 0$ on $\tilde{A} \setminus U$ and $G_{\Lambda' \setminus A,y} \Delta_1^2 G_{\Lambda' \setminus A,y} = 0$ on $\mathbb{Z}^d \setminus U$ (even though $G_{\Lambda' \setminus A,y}$ is not biharmonic everywhere on $\Lambda \setminus (A \cup U)$).

If we define $H_{A,y} =: G_{\Lambda \setminus A,y} - G_{\Lambda' \setminus A,y}$, the preceding two estimates imply that

$$\begin{aligned} \left\| \nabla_{1}^{2} H_{A,y} \right\|_{L^{2}(\mathbb{Z}^{\mathsf{d}} \setminus (U+Q_{2k\hat{N}_{\mathsf{d}}\lambda_{\max}}(0)))} \\ &\leq 2 \left\| \nabla_{1}^{2} G_{\Lambda \setminus A,y} \right\|_{L^{2}(\mathbb{Z}^{\mathsf{d}} \setminus (U+Q_{2k\hat{N}_{\mathsf{d}}\lambda_{\max}}(0)))}^{2} + 2 \left\| \nabla_{1}^{2} G_{\Lambda' \setminus A,y} \right\|_{L^{2}(\mathbb{Z}^{\mathsf{d}} \setminus (U+Q_{2k\hat{N}_{\mathsf{d}}\lambda_{\max}}(0)))} \tag{6.5.12} \\ &\leq \frac{1}{2^{k-1}} \left(G_{\Lambda \setminus A}(y,y) + G_{\Lambda' \setminus A}(y,y) \right) \,. \end{aligned}$$

The polymer $U + Q_{2k\hat{N}_d\lambda_{mac}}(0)$ consists of $\left(\frac{\tilde{r}}{\tilde{N}_d\lambda_{mac}} + 2k\right)^d$ boxes in $Q_{\hat{N}_d\lambda_{mac}}$. The function H_y is biharmonic on $U + Q_{2k\hat{N}_d\lambda_{mac}}(0) \subset Q_R(0)$ as the two singularities cancel out. So we can apply Theorem 6.4.1 b) and obtain on the event $\Omega_{U+Q_{2k\hat{N}_d\lambda_{mac}}(0),k}$ that

$$\left\|\nabla_{1}^{2}H_{A,y}\right\|_{L^{2}(U+Q_{2k\hat{N}_{d}\lambda_{mac}}(0))}^{2} \leq \frac{1}{2^{k}}\left\|\nabla_{1}^{2}H_{A,y}\right\|_{L^{2}((U+Q_{4k\hat{N}_{d}\lambda_{mac}}(0))\setminus(U+Q_{2k\hat{N}_{d}\lambda_{mac}}(0)))}^{2}$$
(6.5.13)

Furthermore, we can introduce the event $\tilde{\Omega}_{y,k}$ as in (6.5.5). By definition we have

$$G_{\Lambda \setminus A}(y, y) \le C_{\mathsf{d}} \left(1 + \mathbb{1}_{\mathsf{d}=4}(\log k + |\log \varepsilon|) \right)$$
(6.5.14)

on that event. We claim that on the event $\tilde{\Omega}_{y,k}$ we also have

$$G_{\Lambda'\setminus A}(y,y) \le C_{\mathsf{d}} \left(1 + \mathbb{1}_{\mathsf{d}=4}(\log k + |\log \varepsilon|)\right) \,. \tag{6.5.15}$$

Indeed, if $d \ge 5$ this is once more a trivial consequence of (6.2.5), while if d = 4 we can use (6.2.6) to estimate

$$\begin{aligned} G_{\Lambda' \setminus A}(y, y) &\leq \frac{1}{4\pi^2} \log(1 + d(x, (A \cup (\mathbb{Z}^{\mathsf{d}} \setminus \Lambda'))) + C) \\ &\leq \frac{1}{4\pi^2} \log(1 + d(x, (A \cup (\mathbb{Z}^{\mathsf{d}} \setminus \Lambda))) + C) \\ &\leq G_{\Lambda \setminus A}(y, y) + C \end{aligned}$$

so that (6.5.15) is a consequence of (6.5.14).

Finally, if $A \in \Omega_{x,k}$ with the event $\Omega_{x,k}$ from Lemma 6.3.2, we have

$$|H_{A,y}(x)|^{2} \leq C_{\mathsf{d}} \frac{k^{\mathsf{d}} \left(1 + \mathbb{1}_{\mathsf{d}=4} (\log k + |\log \varepsilon|^{3/2})\right)}{\varepsilon} \|\nabla_{1}^{2} H_{A,y}(x)\|_{L^{2}(Q_{kN_{\mathsf{d}}\lambda_{\mathsf{mic}}}(x))}^{2}.$$
(6.5.16)

We choose ε small enough so that $N_d \lambda_{mic} \leq 2\hat{N}_d \lambda_{mac}$. Then, in particular, $Q_{kN_d \lambda_{mic}}(x) \subset U + Q_{kN_d \lambda_{mic}}(0) \subset U + Q_{r+2kN_d \lambda_{mac}}(0)$.

Now we can combine the estimates we have just collected. More precisely, assume that $A \in \Omega_{U,k} \cap \Omega_{U+Q_{2k\hat{N}_d\lambda_{mac}}(0),k} \cap \tilde{\Omega}_{y,k} \cap \Omega_{x,k}$. Then we can use (6.5.12), (6.5.13), (6.5.14), (6.5.15) and (6.5.16) to obtain

$$\begin{split} |G_{\Lambda\setminus A}(x,y) - G_{\Lambda'\setminus A}(x,y)|^{2} \\ &= |H_{A,y}(x)|^{2} \\ \leq C_{d} \frac{k^{d} \left(1 + \mathbb{1}_{d=4}(\log k + |\log \varepsilon|^{3/2})\right)}{\varepsilon} \|\nabla_{1}^{2}H_{A,y}\|_{L^{2}(Q_{kN_{d}\lambda_{mic}}(x))}^{2} \\ \leq C_{d} \frac{k^{d} \left(1 + \mathbb{1}_{d=4}(\log k + |\log \varepsilon|^{3/2})\right)}{\varepsilon} \|\nabla_{1}^{2}H_{A,y}\|_{L^{2}((U+Q_{2k\bar{N}_{d}\lambda_{mac}}(0))}^{2} \\ \leq C_{d} \frac{k^{d} \left(1 + \mathbb{1}_{d=4}(\log k + |\log \varepsilon|^{3/2})\right)}{2^{k_{\varepsilon}}} \|\nabla_{1}^{2}H_{A,y}\|_{L^{2}(Z^{d}\setminus (U+Q_{2k\bar{N}_{d}\lambda_{mac}}(0)))}^{2} \\ \leq C_{d} \frac{k^{d} \left(1 + \mathbb{1}_{d=4}(\log k + |\log \varepsilon|^{3/2})\right)}{2^{k_{\varepsilon}}} \|\nabla_{1}^{2}H_{A,y}\|_{L^{2}(Z^{d}\setminus (U+Q_{2k\bar{N}_{d}\lambda_{mac}}(0)))}^{2} \\ \leq C_{d} \frac{k^{d} \left(1 + \mathbb{1}_{d=4}(\log k + |\log \varepsilon|^{3/2})\right)}{2^{2k-1}\varepsilon} \left(G_{\Lambda\setminus A}(y,y) + G_{\Lambda'\setminus A}(y,y)\right) \\ \leq C_{d} \frac{k^{d} \left(1 + \mathbb{1}_{d=4}(\log k + |\log \varepsilon|^{3/2})\right)}{2^{2k-1}\varepsilon} \left(1 + \mathbb{1}_{d=4}(\log k + |\log \varepsilon|)\right) \\ \leq C_{d} \left(\frac{1}{2}\right)^{k} \frac{1 + \mathbb{1}_{d=4}|\log \varepsilon|^{5/2}}{\varepsilon} \\ \leq C_{d} \exp\left(-\frac{R-r}{C_{d}\lambda_{mac}}\right) \frac{1 + \mathbb{1}_{d=4}|\log \varepsilon|^{5/2}}{\varepsilon} . \end{split}$$

From (6.5.17) we see that on the event

$$\Omega := \Omega_{U,k} \cap \Omega_{U+Q_{2k\hat{N}_{d}\lambda_{\max}}(0),k} \cap \bigcap_{y \in Q_{r}(0)} \tilde{\Omega}_{y,k} \cap \bigcap_{x \in Q_{r}(0)} \Omega_{x,k}$$

we have the desired estimate. So it only remains to bound the probability of Ω from below. For this we use a union bound to see

$$\zeta_{\Lambda}^{\varepsilon}(\Omega) \geq 1 - \left(\frac{\tilde{r}}{\hat{N}_{\mathsf{d}}\lambda_{\max}}\right)^{\mathsf{d}} \frac{1}{2^{k}} - \left(\frac{\tilde{r}}{\hat{N}_{\mathsf{d}}\lambda_{\max}} + 2k\right)^{\mathsf{d}} \frac{1}{2^{k}} - (2r+1)^{\mathsf{d}} \frac{1}{2^{k^{\mathsf{d}}}} - (2r+1)^{\mathsf{d}} \frac{1}{2^{k}}$$

$$\geq 1 - C_{\mathsf{d}} r^{\mathsf{d}} \exp\left(-\frac{k}{C_{\mathsf{d}}}\right)$$
$$\geq 1 - C_{\mathsf{d}} r^{\mathsf{d}} \exp\left(-\frac{R-r}{C_{\mathsf{d}}\lambda_{\mathsf{mac}}}\right) \,.$$

This completes the proof.

6.5.2 Estimates on variance and covariance

Proof of Theorem 6.1.5. We first prove part a) and then part b).

Step 1: Estimates on the variance

We have that

$$\mathbb{E}^{\varepsilon}_{\Lambda}(\psi^2_x) = \sum_{A \subset \Lambda} \zeta^{\varepsilon}_{\Lambda}(A) \mathbb{E}_{\Lambda \setminus A}(\psi^2_x) = \sum_{A \subset \Lambda} \zeta^{\varepsilon}_{\Lambda}(A) G_{\Lambda \setminus A}(x, x) \,. \tag{6.5.18}$$

Thus, (6.1.10) follows immediately from (6.2.5). For (6.1.11) we use Lemma 6.5.2. Indeed, using Fubini's theorem and (6.5.3) we can rewrite (6.5.18) as

$$\begin{split} \mathbb{E}_{\Lambda}^{\varepsilon}(\psi_{x}^{2}) &= \int_{0}^{\infty} \zeta_{\Lambda}^{\varepsilon} \left(G_{\Lambda \setminus \mathcal{A}}(x,x) \geq t \right) \, \mathrm{d}t \\ &\leq \int_{\tilde{\gamma}}^{\infty} \zeta_{\Lambda}^{\varepsilon} \left(G_{\Lambda \setminus \mathcal{A}}(x,x) \geq t \right) \, \mathrm{d}t + \tilde{\gamma} \\ &\leq \int_{\tilde{\gamma}}^{\infty} \exp\left(-\frac{\varepsilon \exp(16\pi^{2}(t-\tilde{\gamma}))}{C|\log\varepsilon|^{\frac{1}{2}}} \right) \, \mathrm{d}t + \tilde{\gamma} \\ &\leq \int_{0}^{\infty} \exp\left(-\frac{\varepsilon \exp(16\pi^{2}t)}{C|\log\varepsilon|^{\frac{1}{2}}} \right) \, \mathrm{d}t + C \\ &\leq \int_{0}^{\frac{|\log\varepsilon|}{16\pi^{2}} + \frac{\log|\log\varepsilon|}{32\pi^{2}}} 1 \, \mathrm{d}t + \int_{\frac{|\log\varepsilon|}{16\pi^{2}} + \frac{\log|\log\varepsilon|}{32\pi^{2}}}^{\infty} \exp\left(-\frac{\varepsilon \exp(16\pi^{2}t)}{C|\log\varepsilon|^{\frac{1}{2}}} \right) \, \mathrm{d}t + C \\ &\leq \frac{|\log\varepsilon|}{16\pi^{2}} + \frac{\log|\log\varepsilon|}{32\pi^{2}} + \int_{0}^{\infty} \exp\left(-\frac{\exp(16\pi^{2}t)}{C} \right) \, \mathrm{d}t + C \\ &\leq \frac{|\log\varepsilon|}{16\pi^{2}} + \frac{\log|\log\varepsilon|}{32\pi^{2}} + C \\ &\leq \frac{|\log\varepsilon|}{16\pi^{2}} + C \log|\log\varepsilon| \end{split}$$

for ε small enough, which establishes the upper bound in (6.1.11). For the lower bound we argue similarly using (6.5.4) and obtain

$$\begin{split} \mathbb{E}^{\varepsilon}_{\Lambda}(\psi_{x}^{2}) &= \int_{0}^{\infty} \zeta^{\varepsilon}_{\Lambda} \left(G_{\Lambda \setminus \mathcal{A}}(x,x) \geq t \right) \, \mathrm{d}t \\ &\geq \int_{0}^{\frac{1}{8\pi^{2}} \log\left(1 + d(x,\mathbb{Z}^{\mathsf{d}} \setminus \Lambda) - \frac{1}{\varepsilon}\right) - \tilde{\gamma}} \exp\left(-\frac{C_{\alpha}\varepsilon \exp(32\pi^{2}(t+\tilde{\gamma}))}{|\log\varepsilon|^{\frac{1}{2}}}\right) \, \mathrm{d}t - C \\ &\geq \int_{0}^{\min\left(\frac{1}{8\pi^{2}} \log\left(1 + d(x,\mathbb{Z}^{\mathsf{d}} \setminus \Lambda) - \varepsilon^{-\alpha}\right), \frac{|\log\varepsilon|}{32\pi^{2}} - \frac{\log|\log\varepsilon|}{64\pi^{2}}\right) - \tilde{\gamma}} \exp\left(-\frac{C_{\alpha}}{|\log\varepsilon|}\right) \, \mathrm{d}t - C \end{split}$$

The assumption $d(x, \mathbb{Z}^d \setminus \Lambda) \ge \varepsilon^{-\alpha} + \varepsilon^{-1/4}$ ensures that the second term in the minimum here is smaller than the first, and so we see that indeed

$$\mathbb{E}^{\varepsilon}_{\Lambda}(\psi_x^2) \geq \frac{|\log \varepsilon|}{32\pi^2} - \frac{\log|\log \varepsilon|}{64\pi^2} \left(1 - \frac{C_{\alpha}}{|\log \varepsilon|}\right)$$

$$\geq \frac{|\log \varepsilon|}{32\pi^2} - C_\alpha \log |\log \varepsilon|$$

Step 2: Estimates on the covariance As in (6.5.19) we have

$$\left|\mathbb{E}^{\varepsilon}_{\Lambda}(\psi_{x}\psi_{y})\right| = \left|\sum_{A\subset\Lambda}\zeta^{\varepsilon}_{\Lambda}(A)\mathbb{E}_{\Lambda\setminus A}(\psi_{x}\psi_{y})\right| \le \sum_{A\subset\Lambda}\zeta^{\varepsilon}_{\Lambda}(A)|G_{\Lambda\setminus A}(x,y)|.$$
(6.5.19)

From Lemma 6.5.3 we know

$$\zeta_{\Lambda}^{\varepsilon} \left(|G_{\Lambda \setminus \mathcal{A}}(x,y)| \le C_{\mathsf{d}} \exp\left(-\frac{|x-y|_{\infty}}{C_{\mathsf{d}}\lambda_{\max}}\right) \frac{1 + \mathbb{1}_{\mathsf{d}=4} |\log\varepsilon|^{5/4}}{\varepsilon^{1/2}} \right) \ge 1 - \exp\left(-\frac{|x-y|_{\infty}}{C_{\mathsf{d}}\lambda_{\max}}\right) \cdot \frac{1 + \mathbb{1}_{\mathsf{d}=4} |\log\varepsilon|^{5/4}}{\varepsilon^{1/2}} = 1 - \exp\left(-\frac{|x-y|_{\infty}}{C_{\mathsf{d}}\lambda_{\max}}\right) \cdot \frac{1 + \mathbb{1}_{\mathsf{d}=4} |\log\varepsilon|^{5/4}}{\varepsilon^{1/2}} = 1 - \exp\left(-\frac{|x-y|_{\infty}}{C_{\mathsf{d}}\lambda_{\max}}\right) \cdot \frac{1 + \mathbb{1}_{\mathsf{d}=4} |\log\varepsilon|^{5/4}}{\varepsilon^{1/2}} = 1 - \exp\left(-\frac{|x-y|_{\infty}}{C_{\mathsf{d}}\lambda_{\max}}\right) \cdot \frac{1 + \mathbb{1}_{\mathsf{d}=4} |\log\varepsilon|^{5/4}}{\varepsilon^{1/2}} = 1 - \exp\left(-\frac{|x-y|_{\infty}}{C_{\mathsf{d}}\lambda_{\max}}\right) \cdot \frac{1 + \mathbb{1}_{\mathsf{d}=4} |\log\varepsilon|^{5/4}}{\varepsilon^{1/2}} = 1 - \exp\left(-\frac{|x-y|_{\infty}}{C_{\mathsf{d}}\lambda_{\max}}\right) \cdot \frac{1 + \mathbb{1}_{\mathsf{d}=4} |\log\varepsilon|^{5/4}}{\varepsilon^{1/2}} = 1 - \exp\left(-\frac{|x-y|_{\infty}}{C_{\mathsf{d}}\lambda_{\max}}\right) \cdot \frac{1 + \mathbb{1}_{\mathsf{d}=4} |\log\varepsilon|^{5/4}}{\varepsilon^{1/2}} = 1 - \exp\left(-\frac{|x-y|_{\infty}}{C_{\mathsf{d}}\lambda_{\max}}\right) \cdot \frac{1 + \mathbb{1}_{\mathsf{d}=4} |\log\varepsilon|^{5/4}}{\varepsilon^{1/2}} = 1 - \exp\left(-\frac{|x-y|_{\infty}}{C_{\mathsf{d}}\lambda_{\max}}\right) \cdot \frac{1 + \mathbb{1}_{\mathsf{d}=4} |\log\varepsilon|^{5/4}}{\varepsilon^{1/2}} = 1 - \exp\left(-\frac{|x-y|_{\infty}}{C_{\mathsf{d}}\lambda_{\max}}\right) \cdot \frac{1 + \mathbb{1}_{\mathsf{d}=4} |\log\varepsilon|^{5/4}}{\varepsilon^{1/2}} = 1 - \exp\left(-\frac{|x-y|_{\infty}}{C_{\mathsf{d}}\lambda_{\max}}\right) \cdot \frac{1 + \mathbb{1}_{\mathsf{d}=4} |\log\varepsilon|^{5/4}}{\varepsilon^{1/2}} = 1 - \exp\left(-\frac{|x-y|_{\infty}}{C_{\mathsf{d}}\lambda_{\max}}\right) \cdot \frac{1 + \mathbb{1}_{\mathsf{d}=4} |\log\varepsilon|^{5/4}}{\varepsilon^{1/4}} = 1 - \exp\left(-\frac{|x-y|_{\infty}}{C_{\mathsf{d}}\lambda_{\max}}\right) \cdot \frac{1 + \mathbb{1}_{\mathsf{d}=4} |\log\varepsilon|^{5/4}}{\varepsilon^{1/4}} = 1 - \exp\left(-\frac{|x-y|_{\infty}}{C_{\mathsf{d}}\lambda_{\max}}\right) \cdot \frac{1 + \mathbb{1}_{\mathsf{d}=4} |\log\varepsilon|^{5/4}}{\varepsilon^{1/4}} = 1 - \exp\left(-\frac{|x-y|_{\infty}}{C_{\mathsf{d}}\lambda_{\max}}\right) \cdot \frac{1 + \mathbb{1}_{\mathsf{d}=4} |\log\varepsilon|^{5/4}}{\varepsilon^{1/4}} = 1 - \exp\left(-\frac{|x-y|_{\infty}}{C_{\mathsf{d}}\lambda_{\max}}\right) \cdot \frac{1 + \mathbb{1}_{\mathsf{d}=4} |\log\varepsilon|^{5/4}}{\varepsilon^{1/4}} = 1 - \exp\left(-\frac{|x-y|_{\infty}}{C_{\mathsf{d}}\lambda_{\max}}\right) \cdot \frac{1 + \mathbb{1}_{\mathsf{d}=4} |\log\varepsilon|^{5/4}}{\varepsilon^{1/4}} = 1 - \exp\left(-\frac{|x-y|_{\infty}}{C_{\mathsf{d}}\lambda_{\max}}\right) \cdot \frac{1 + \mathbb{1}_{\mathsf{d}=4} |\log\varepsilon|^{5/4}}{\varepsilon^{1/4}} = 1 - \exp\left(-\frac{|x-y|_{\infty}}{C_{\mathsf{d}}\lambda_{\max}}\right) \cdot \frac{1 + \mathbb{1}_{\mathsf{d}=4} |\varepsilon|^{5/4}}{\varepsilon^{1/4}} = 1 - \exp\left(-\frac{|x-y|_{\infty}}{C_{\mathsf{d}}\lambda_{\max}}\right) \cdot \frac{1 + \mathbb{1}_{\mathsf{d}=4} |\varepsilon|^{5/4}}{\varepsilon^{1/4}} = 1 - \exp\left(-\frac{|x-y|_{\infty}}{C_{\mathsf{d}}\lambda_{\max}}\right) \cdot \frac{1 + \exp\left(-\frac{$$

Abbreviate the event described here by Ω . The decomposition (6.5.19) implies

and so we only need to bound $|G_{\Lambda \setminus A}(x, y)|$ on the rare event Ω^c .

If $d \ge 5$, we can use the bound

$$G_{\Lambda \setminus A}(x,y) \leq \max \left(G_{\Lambda \setminus A}(x,x), G_{\Lambda \setminus A}(y,y) \right) \leq C_{\mathsf{d}}$$

that follows from (6.2.5) and (6.5.2) to conclude from (6.5.20) that

$$\begin{split} |\mathbb{E}^{\varepsilon}_{\Lambda}(\psi_{x}\psi_{y})| &\leq C_{\mathsf{d}}\exp\left(-\frac{|x-y|_{\infty}}{C_{\mathsf{d}}\lambda_{\max}}\right)\frac{1}{\varepsilon^{1/2}} + C_{\mathsf{d}}\zeta^{\varepsilon}_{\Lambda}(\Omega^{c}) \\ &\leq C_{\mathsf{d}}\exp\left(-\frac{|x-y|_{\infty}}{C_{\mathsf{d}}\lambda_{\max}}\right)\frac{1}{\varepsilon^{1/2}} + C_{\mathsf{d}}\exp\left(-\frac{|x-y|_{\infty}}{C_{\mathsf{d}}\lambda_{\max}}\right) \\ &\leq \frac{C_{\mathsf{d}}}{\varepsilon^{1/2}}\exp\left(-\frac{|x-y|_{\infty}}{C_{\mathsf{d}}\lambda_{\max}}\right) \end{split}$$

which implies (6.1.12).

If d = 4, the estimate is slightly more complicated, as $G_{\Lambda \setminus A}(x, y)$ is no longer uniformly bounded. Instead we use Lemma 6.5.2 to deduce a tail bound on $G_{\Lambda \setminus A}(x, y)$. Note first that if x = y then (6.1.13) follows (6.1.11), and so we can assume $x \neq y$. By (6.5.2) and (6.5.3) we have for any $t \geq \tilde{\gamma}$ that

$$\begin{split} \zeta_{\Lambda}^{\varepsilon} \left(|G_{\Lambda \setminus \mathcal{A}}(x,y)| \geq t \right) &\leq \zeta_{\Lambda}^{\varepsilon} \left(\max \left(G_{\Lambda \setminus \mathcal{A}}(x,x), G_{\Lambda \setminus \mathcal{A}}(y,y) \right) \geq t \right) \\ &\leq \zeta_{\Lambda}^{\varepsilon} \left(G_{\Lambda \setminus \mathcal{A}}(x,x) \geq t \right) + \zeta_{\Lambda}^{\varepsilon} \left(G_{\Lambda \setminus \mathcal{A}}(y,y) \geq t \right) \\ &\leq 2 \exp \left(- \frac{\varepsilon \exp(16\pi^2(t-\tilde{\gamma}))}{C |\log \varepsilon|^{\frac{1}{2}}} \right). \end{split}$$

We can now use Fubini's theorem to estimate the second summand in (6.5.20) as

$$\begin{split} \sum_{\substack{A \subseteq \Lambda \\ A \notin \Omega}} \zeta_{\Lambda}^{\varepsilon}(A) |G_{\Lambda \setminus A}(x,y)| \\ &\leq \int_{0}^{\infty} \zeta_{\Lambda}^{\varepsilon} \left(|G_{\Lambda \setminus \mathcal{A}}(x,y)| \geq t, \mathcal{A} \notin \Omega \right) dt \\ &\leq \int_{0}^{\infty} \min \left(\zeta_{\Lambda}^{\varepsilon} \left(|G_{\Lambda \setminus \mathcal{A}}(x,y)| \geq t \right), \zeta_{\Lambda}^{\varepsilon}(\Omega^{c}) \right) dt \\ &\leq \int_{\tilde{\gamma}}^{\infty} \min \left(\zeta_{\Lambda}^{\varepsilon} \left(|G_{\Lambda \setminus \mathcal{A}}(x,y)| \geq t \right), \zeta_{\Lambda}^{\varepsilon}(\Omega^{c}) \right) dt + \int_{0}^{\tilde{\gamma}} \zeta_{\Lambda}^{\varepsilon}(\Omega^{c}) dt \\ &\leq \int_{0}^{\infty} \min \left(2 \exp \left(-\frac{\varepsilon \exp(16\pi^{2}t)}{C |\log \varepsilon|^{\frac{1}{2}}} \right), \exp \left(-\frac{|x-y|_{\infty}}{C_{d}\lambda_{mac}} \right) \right) dt + \tilde{\gamma} \exp \left(-\frac{|x-y|_{\infty}}{C\lambda_{mac}} \right). \end{split}$$
(6.5.21)

To estimate the remaining integral, note that for a, b < 1 we have $a = \exp(-b \exp(16\pi^2 t))$ for $t = t_* := \frac{1}{16\pi^2} (\log |\log a| + |\log b|)$ and so

$$\begin{split} &\int_{0}^{\infty} \min(a, \exp(-b \exp(16\pi^{2}t))) dt \\ &= \int_{0}^{t_{*}} a \, dt + \int_{t_{*}}^{\infty} \exp(-b \exp(16\pi^{2}t)) dt \\ &= t_{*}a + \int_{0}^{\infty} \exp(-b \exp(16\pi^{2}t_{*}) \exp(16\pi^{2}t)) dt \\ &\leq t_{*}a + \int_{0}^{\infty} \exp(-b \exp(16\pi^{2}t_{*})(1+16\pi^{2}t)) dt \\ &= t_{*}a + \exp(-b \exp(16\pi^{2}t_{*})) \int_{0}^{\infty} \exp(-16\pi^{2}tb \exp(16\pi^{2}t_{*})) dt \\ &= t_{*}a + \frac{\exp(-b \exp(16\pi^{2}t_{*}))}{16\pi^{2}b \exp(16\pi^{2}t_{*})} \\ &= t_{*}a + \frac{a}{16\pi^{2}|\log a|} \\ &= \frac{1}{16\pi^{2}} \left(a \log|\log a| + a|\log b| \frac{1}{|\log a|} \right) . \\ &\leq Ca(\log|\log a| + |\log b|) \end{split}$$

With the choices $a = \exp\left(-\frac{|x-y|_{\infty}}{C_{d}\lambda_{mac}}\right)$ and $b = \frac{\varepsilon}{C|\log\varepsilon|^{1/2}}$ we then obtain from (6.5.21) that

$$\begin{split} \sum_{\substack{A \subseteq \Lambda \\ A \notin \Omega}} \zeta_{\Lambda}^{\varepsilon}(A) |G_{\Lambda \setminus A}(x,y)| \\ &\leq C \exp\left(-\frac{|x-y|_{\infty}}{C\lambda_{\max}}\right) \left(\log\frac{|x-y|_{\infty}}{C\lambda_{\max}} + \left|\log\frac{\varepsilon}{C|\log\varepsilon|^{1/2}}\right| + 1\right) \\ &\leq C \exp\left(-\frac{|x-y|_{\infty}}{C\lambda_{\max}}\right) \left(\log\frac{|x-y|_{\infty}}{\lambda_{\max}} + \log\frac{|\log\varepsilon|^{1/2}}{\varepsilon} + 1\right) \\ &\leq C \exp\left(-\frac{|x-y|_{\infty}}{C\lambda_{\max}}\right) \left(\log|x-y|_{\infty} - \log|\log\varepsilon| + 1\right) \end{split}$$

where we have used that $4\log \frac{1}{\lambda_{\text{mac}}} + \log \frac{|\log \varepsilon|^{1/2}}{\varepsilon} = -\log |\log \varepsilon|$. Finally we can return to (6.5.20) and obtain

 $|\mathbb{E}^{\varepsilon}_{\Lambda}(\psi_x\psi_y)|$

$$\leq C \exp\left(-\frac{|x-y|_{\infty}}{C_{d}\lambda_{\max}}\right) \frac{|\log\varepsilon|^{5/4}}{\varepsilon^{1/2}} + C \exp\left(-\frac{|x-y|_{\infty}}{C\lambda_{\max}}\right) \left(\log|x-y|_{\infty} - \log|\log\varepsilon| + 1\right)$$

$$\leq C \left(\frac{|\log\varepsilon|^{5/4}}{\varepsilon^{1/2}} + \log|x-y|_{\infty}\right) \exp\left(-\frac{|x-y|_{\infty}}{C\lambda_{\max}}\right)$$

which implies (6.1.13).

Finally, the estimates (6.1.14) and (6.1.15) are straightforward consequences of (6.1.12) and (6.1.13), respectively. \Box

6.5.3 Existence of the thermodynamic limit of the field

It remains to prove the existence of the thermodynamic limit of the pinned field. This is significantly more difficult than the existence of the thermodynamic limit of the set of pinned points, as we do not have correlation inequalities for the field or a random walk representation. Instead we show by hand that the exponential decay of correlations implies convergence of $\mathbb{E}^{\varepsilon}_{\Lambda}(f)$ for any bounded local *f*.

Proof of Theorem 6.1.2, second part. As in the proof of the first part it suffices to check that the limit $\lim_{\Lambda \nearrow \mathbb{Z}^d} \mathbb{E}^{\varepsilon}_{\Lambda}(f)$ exists for any bounded local function $f : \mathbb{R}^{\mathbb{Z}^d} \to \mathbb{R}$. Our tail estimates on $\mathbb{P}^{\varepsilon}_{\Lambda}$ easily imply boundedness of $\mathbb{E}^{\varepsilon}_{\Lambda}(f)$, so if the limit exists it is finite.

So let a local function f be given. Suppose that f only depends on the values of ψ in $Q_r(0)$ for some r. We can assume that $r \ge \hat{N}_d$. Let $R \in \mathbb{N}$ with $R \ge 8r$. We set $\Lambda' = Q_R(0)$. Let Ω be the event described in (6.5.11). Let also $k \in \mathbb{N}$ and consider the event $\tilde{\Omega}_{0,k}$ from (6.5.5) (with y = 0). Note that if $A \in \tilde{\Omega}_{0,k}$ we have

$$G_{\Lambda\setminus A}(0,0) \leq C_{\mathsf{d}} \left(1 + \mathbb{1}_{\mathsf{d}=4}(\log k + |\log \varepsilon|)\right) \,.$$

Similarly as for (6.5.15), we see that this implies for any $x \in Q_r(0)$

$$G_{\Lambda \setminus A}(x, x) \le C_{\mathsf{d}} \left(1 + \mathbb{1}_{\mathsf{d}=4}(\log r + \log k + |\log \varepsilon|) \right) =: X_{\mathsf{d},\varepsilon,k,r}$$

and in combination with (6.5.2) also

$$\max_{x,y\in Q_r(0)} |G_{\Lambda\setminus A}(x,y)| \leq X_{\mathsf{d},\varepsilon,k,r}$$

We can now write

$$\mathbb{E}^{\varepsilon}_{\Lambda}(f) = \sum_{A \subset \Lambda} \mathbb{E}_{\Lambda \setminus A}(f) \zeta^{\varepsilon}_{\Lambda}(A) = \sum_{\substack{A \subset \Lambda \\ A \in \Omega \cap \tilde{\Omega}_{0,k}}} \mathbb{E}_{\Lambda \setminus A}(f) \zeta^{\varepsilon}_{\Lambda}(A) + \sum_{\substack{A \subset \Lambda \\ A \notin \Omega \cap \tilde{\Omega}_{0,k}}} \mathbb{E}_{\Lambda \setminus A}(f) \zeta^{\varepsilon}_{\Lambda}(A) \,.$$
(6.5.22)

The second summand here is an error term that is easy to estimate. We have

$$\left| \sum_{\substack{A \subset \Lambda \\ A \notin (\Omega \cap \tilde{\Omega}_{0,k})}} \mathbb{E}_{\Lambda \setminus A}(f) \zeta_{\Lambda}^{\varepsilon}(A) \right| \leq \|f\|_{L^{\infty}(\mathbb{Z}^{d})} \sum_{\substack{A \subset \Lambda \\ A \notin \Omega \cap \tilde{\Omega}_{0,k}}} \zeta_{\Lambda}^{\varepsilon}(A)$$

$$\leq \|f\|_{L^{\infty}(\mathbb{Z}^{d})} \left(\zeta_{\Lambda}^{\varepsilon}(\Omega^{c}) + \zeta_{\Lambda}^{\varepsilon}(\tilde{\Omega}_{0,k}^{c}) \right)$$

$$\leq \left(C_{d} r^{d} \exp\left(-\frac{R-r}{C_{d}\lambda_{\max}} \right) + \frac{1}{2^{k}} \right) \|f\|_{L^{\infty}(\mathbb{Z}^{d})}$$
(6.5.23)

and note that the right-hand side tends to 0 as $k, R \to \infty$, uniformly in $\Lambda \supset Q_R(0)$.

Next, we begin to analyse the main term in (6.5.22), i.e. the first summand. $\mathbb{P}_{\Lambda \setminus A}$ is the law of a multivariate Gaussian measure. Thus, $\mathbb{E}_{\Lambda \setminus A}(f)$ depends only on the variances and covariances of that measure. Because of the locality of f it depends only on those variances and covariances where the sites are in $Q_r(0)$. In particular, $\mathbb{E}(f)$ is a continuous function of $(G_{\Lambda \setminus A}(x, y))_{x,y \in Q_r(0)} \in \mathbb{R}^{Q_r(0) \times Q_r(0)}$. If we restrict it to the compact set $[-X_{\mathsf{d},\varepsilon,k,r}, X_{\mathsf{d},\varepsilon,k,r}]^{Q_r(0) \times Q_r(0)}$, it is uniformly continuous.

From Lemma 6.5.4 we know that for $A \in \Omega$

$$\left| \left(G_{\Lambda \setminus A}(x,y) \right)_{x,y \in Q_r(0)} - \left(G_{Q_R(0) \setminus A}(x,y) \right)_{x,y \in Q_r(0)} \right|_{\infty} \\ \leq C_{\mathsf{d}} \exp\left(-\frac{R-r}{C_{\mathsf{d}}\lambda_{\max}} \right) \frac{1 + \mathbb{1}_{\mathsf{d}=4} |\log \varepsilon|^{5/4}}{\varepsilon^{1/2}}$$

and the right-hand side tends to 0 as $R \to \infty$. Moreover, we have for $A \in \tilde{\Omega}_{0,k}$ that

$$\left(G_{\Lambda\setminus A}(x,y)\right)_{x,y\in Q_r(0)}\in \left[-X_{\mathsf{d},\varepsilon,k,r},X_{\mathsf{d},\varepsilon,k,r}\right]^{Q_r(0)\times Q_r(0)}$$

Thus, the uniform continuity of $\mathbb{E}(f)$ implies that there is a function $\omega_{\mathsf{d},\varepsilon,f,k,r}(R)$ (independent of Λ) with $\lim_{R\to\infty} \omega_{\mathsf{d},\varepsilon,f,k,r}(R) = 0$ such that for all $A \in \Omega \cap \tilde{\Omega}_{0,k}$

$$\left|\mathbb{E}_{\Lambda\setminus A}(f) - \mathbb{E}_{Q_R(0)\setminus A}(f)\right| \leq \omega_{\mathsf{d},\varepsilon,f,k,r}(R).$$

This implies for the first summand in (6.5.22) that

$$\begin{split} \sum_{\substack{A \in \Omega \cap \hat{\Omega}_{0,k}}} \mathbb{E}_{\Lambda \setminus A}(f) \zeta_{\Lambda}^{\epsilon}(A) &- \sum_{A \subset \Lambda} \mathbb{E}_{Q_{R}(0) \setminus A}(f) \zeta_{\Lambda}^{\epsilon}(A) \\ &\leq \left| \sum_{\substack{A \in \Omega \cap \hat{\Omega}_{0,k}}} \mathbb{E}_{\Lambda \setminus A}(f) \zeta_{\Lambda}^{\epsilon}(A) - \sum_{\substack{A \subset \Lambda \\ A \in \Omega \cap \hat{\Omega}_{0,k}}} \mathbb{E}_{Q_{R}(0) \setminus A}(f) \zeta_{\Lambda}^{\epsilon}(A) \right| \\ &+ \left| \sum_{\substack{A \subset \Lambda \\ A \notin \Omega \cap \hat{\Omega}_{0,k}}} \mathbb{P}_{Q_{R}(0) \setminus A}(f) \zeta_{\Lambda}^{\epsilon}(A) \right| \\ &\leq \sum_{\substack{A \subset \Lambda \\ A \in \Omega \cap \hat{\Omega}_{0,k}}} \omega_{d,\epsilon,f,k,r}(R) \zeta_{\Lambda}^{\epsilon}(A) + \|f\|_{L^{\infty}(\mathbb{Z}^{d})} \left(\zeta_{\Lambda}^{\epsilon}(\Omega^{c}) + \zeta_{\Lambda}^{\epsilon}(\tilde{\Omega}_{0,k}^{c}) \right) \\ &\leq \omega_{d,\epsilon,f,k,r}(R) + \left(C_{d}r^{d} \exp\left(-\frac{R-r}{C_{d}\lambda_{mac}}\right) + \frac{1}{2^{k}} \right) \|f\|_{L^{\infty}(\mathbb{Z}^{d})} \end{split}$$

where we have estimated the error term the same way as in (6.5.23). We also know that

$$\sum_{A \subset \Lambda} \mathbb{E}_{Q_R(0) \setminus A}(f) \zeta_{\Lambda}^{\varepsilon}(A) = \sum_{A' \subset Q_R(0)} \mathbb{E}_{Q_R(0) \setminus A}(f) \sum_{A'' \subset \Lambda \setminus Q_R(0)} \zeta_{\Lambda}^{\varepsilon}(A' \cup A'')$$

=
$$\sum_{A' \subset Q_R(0)} \mathbb{E}_{Q_R(0) \setminus A'}(f) \zeta_{\Lambda}^{\varepsilon}(A \cap Q_R(0) = A').$$
(6.5.25)

Putting (6.5.22), (6.5.23), (6.5.24), (6.5.25) together, we find

$$\left| \mathbb{E}_{\Lambda}^{\varepsilon}(f) - \sum_{A' \subset Q_{R}(0)} \mathbb{E}_{Q_{R}(0) \setminus A'}(f) \zeta_{\Lambda}^{\varepsilon}(\mathcal{A} \cap Q_{R}(0) = A') \right|$$

$$\leq \omega_{\mathsf{d},\varepsilon,f,k,r}(R) + 2 \left(C_{\mathsf{d}} r^{\mathsf{d}} \exp\left(-\frac{R-r}{C_{\mathsf{d}} \lambda_{\max}}\right) + \frac{1}{2^{k}} \right) \|f\|_{L^{\infty}(\mathbb{Z}^{\mathsf{d}})}.$$
(6.5.26)

We now want to take the limits $\Lambda \nearrow \mathbb{Z}^d$, $R \to \infty$, $k \to \infty$ in that order. For that purpose, note that the weak convergence of $\zeta_{\Lambda}^{\varepsilon}$ to ζ^{ε} implies that $\lim_{\Lambda \nearrow \mathbb{Z}^d} \zeta_{\Lambda}^{\varepsilon}(\mathcal{A} \cap Q_R(0) = A') = \zeta^{\varepsilon}(\mathcal{A} \cap Q_R(0) = A')$, and so (6.5.26) implies

$$\limsup_{k\to\infty}\limsup_{R\to\infty}\limsup_{\Lambda\nearrow\mathbb{Z}^{\mathsf{d}}}\left|\mathbb{E}^{\varepsilon}_{\Lambda}(f)-\sum_{A'\subset Q_{R}(0)}\mathbb{E}_{Q_{R}(0)\setminus A'}(f)\zeta^{\varepsilon}(\mathcal{A}\cap Q_{R}(0)=A')\right|=0.$$

From this we see that

$$\lim_{\Lambda \nearrow \mathbb{Z}^{\mathsf{d}}} \mathbb{E}^{\varepsilon}_{\Lambda}(f) = \lim_{R \to \infty} \sum_{A' \subset Q_{R}(0)} \mathbb{E}_{Q_{R}(0) \setminus A'}(f) \zeta^{\varepsilon}(\mathcal{A} \cap Q_{R}(0) = A')$$

and that in particular both limits exist. This is what we wanted to show.

The translation invariance follows as in the proof of the first part of the theorem.

Bibliography

- [ABB⁺19] L.-P. Arguin, D. Belius, P. Bourgade, M. Radziwiłł, and K. Soundararajan. Maximum of the Riemann zeta function on a short interval of the critical line. *Comm. Pure Appl. Math.*, 72(3):500–535, 2019. [Cited on page 13.]
- [ABKM19] S. Adams, S. Buchholz, R. Kotecký, and S. Müller. Cauchy-Born Rule from Microscopic Models with Non-convex Potentials, 2019. Preprint arXiv:1910.13564. [Cited on pages 16, 19, and 21.]
- [Aïd13] E. Aïdékon. Convergence in law of the minimum of a branching random walk. *Ann. Probab.*, 41(3A):1362–1426, 2013. [Cited on page 23.]
- [AKM19] S. Armstrong, T. Kuusi, and J.-C. Mourrat. *Quantitative stochastic homogenization and large-scale regularity*, volume 352 of *Grundlehren der Mathematischen Wissenschaften*. Springer, Cham, 2019. [Cited on page 15.]
- [Arn15] D. N. Arnold. Stability, consistency, and convergence of numerical discretizations. In B. E. Engquist, editor, *Encyclopedia of Applied and Computational Mathematics*, pages 1358–1364. Springer-Verlag Berlin Heidelberg, 2015. [Cited on page 33.]
- [AS64] M. Abramowitz and I. A. Stegun. Handbook of mathematical functions with formulas, graphs, and mathematical tables, volume 55 of National Bureau of Standards Applied Mathematics Series. U.S. Government Printing Office, Washington, D.C., 1964. [Cited on page 156.]
- [AW19] S. Armstrong and W. Wu. C^2 regularity of the surface tension for the $\nabla \phi$ interface model, 2019. Preprint arXiv:1909.13325. [Cited on page 15.]
- [BACF13] M. Ben-Artzi, J.-P. Croisille, and D. Fishelov. *Navier–Stokes equations in planar domains*. Imperial College Press, London, 2013. [Cited on page 37.]
- [BB01] E. Bolthausen and D. Brydges. Localization and decay of correlations for a pinned lattice free field in dimension two. In *State of the art in probability and statistics (Leiden, 1999)*, volume 36 of *IMS Lecture Notes Monogr. Ser.*, pages 134–149. Inst. Math. Statist., Beachwood, OH, 2001. [Cited on page 27.]
- [BCK16] E. Bolthausen, A. Cipriani, and N. Kurt. Fast decay of covariances under δ -pinning in the critical and supercritical membrane model, 2016. Preprint arXiv:1601.01513. [Cited on pages 28 and 155.]
- [BCK17] E. Bolthausen, A. Cipriani, and N. Kurt. Exponential decay of covariances for the supercritical membrane model. *Comm. Math. Phys.*, 353(3):1217–1240, 2017. [Cited on pages 28, 153, 155, 157, 158, 159, 160, 161, 166, and 167.]

[BDG01]	E. Bolthausen, JD. Deuschel, and G. Giacomin. Entropic repulsion and the maximum of the two-dimensional harmonic crystal. <i>Ann. Probab.</i> , 29(4):1670–1692, 2001. [Cited on pages 22, 24, 27, and 28.]
[BDKS19]	S. Buchholz, JD. Deuschel, N. Kurt, and F. Schweiger. Probability to be positive for the membrane model in dimensions 2 and 3. <i>Electron. Commun. Probab.</i> , 24:Paper No. 44, 14 pp., 2019. [Cited on pages 1, 23, 25, and 85.]
[BDZ95]	E. Bolthausen, JD. Deuschel, and O. Zeitouni. Entropic repulsion of the lattice free field. <i>Comm. Math. Phys.</i> , 170(2):417–443, 1995. [Cited on page 24.]
[BDZ00]	E. Bolthausen, J. D. Deuschel, and O. Zeitouni. Absence of a wetting transition for a pinned harmonic crystal in dimensions three and larger. <i>J. Math. Phys.</i> , 41(3):1211–1223, 2000. [Cited on page 29.]
[BDZ11]	E. Bolthausen, JD. Deuschel, and O. Zeitouni. Recursions and tightness for the maximum of the discrete, two dimensional Gaussian free field. <i>Electron. Commun. Probab.</i> , 16:114–119, 2011. [Cited on page 22.]
[BDZ16]	M. Bramson, J. Ding, and O. Zeitouni. Convergence in law of the maximum of the two-dimensional discrete Gaussian free field. <i>Comm. Pure Appl. Math.</i> , 69(1):62–123, 2016. [Cited on pages 22 and 97.]
[Ber15]	N. Berestycki. Introduction to the Gaussian Free Field and Liouville Quantum Gravity. Lecture Notes, 2015. [Cited on page 30.]
[Bis20]	M. Biskup. Extrema of the two-dimensional discrete Gaussian free field. In <i>Random graphs, phase transitions, and the Gaussian free field,</i> volume 304 of <i>Springer Proc. Math. Stat.</i> , pages 163–407. Springer, Cham, 2020. [Cited on page 13.]
[BIV00]	T. Bodineau, D. Ioffe, and Y. Velenik. Rigorous probabilistic analysis of equi- librium crystal shapes. <i>J. Math. Phys.</i> , 41(3):1033–1098, 2000. Probabilistic techniques in equilibrium and nonequilibrium statistical physics. [Cited on page 4.]
[Bjø83]	P. Bjørstad. Fast numerical solution of the biharmonic Dirichlet problem on rectangles. <i>SIAM J. Numer. Anal.</i> , 20(1):59–71, 1983. [Cited on page 37.]
[BK07]	M. Biskup and R. Kotecký. Phase coexistence of gradient Gibbs states. <i>Probab. Theory Related Fields</i> , 139(1-2):1–39, 2007. [Cited on page 12.]
[BL16]	M. Biskup and O. Louidor. Extreme local extrema of two-dimensional discrete Gaussian free field. <i>Comm. Math. Phys.</i> , 345(1):271–304, 2016. [Cited on page 30.]
[BL18]	M. Biskup and O. Louidor. Full extremal process, cluster law and freezing for the two-dimensional discrete Gaussian free field. <i>Adv. Math.</i> , 330:589–687, 2018. ^[Cited on page 30.]
[BL19]	M. Biskup and O. Louidor. On intermediate level sets of two-dimensional discrete Gaussian free field. <i>Ann. Inst. Henri Poincaré Probab. Stat.</i> , 55(4):1948–1987, 2019. [Cited on page 30.]
[BL20]	M. Biskup and O. Louidor. Conformal Symmetries in the Extremal Process of Two-Dimensional Discrete Gaussian Free Field. <i>Comm. Math. Phys.</i> , 375(1):175–235, 2020. [Cited on page 30.]

[BLM87]	J. Bricmont, J. L. Lebowitz, and C. Maes. Percolation in strongly correlated systems: the massless Gaussian field. <i>J. Statist. Phys.</i> , 48(5-6):1249–1268, 1987. [Cited on page 30.]
[Bog05]	T. Boggio. Sulle funzioni di Green d'ordine m. <i>Rend. Circ. Mat. Palermo</i> , 20:97–135, 1905. [Cited on page 43.]
[Bog98]	V. I. Bogachev. <i>Gaussian measures</i> , volume 62 of <i>Mathematical Surveys and Monographs</i> . American Mathematical Society, Providence, RI, 1998. [Cited on page 86.]
[Bov06]	A. Bovier. <i>Statistical mechanics of disordered systems. A mathematical perspect-</i> <i>ive</i> , volume 18 of <i>Cambridge Series in Statistical and Probabilistic Mathematics</i> . Cambridge University Press, Cambridge, 2006. [Cited on page 6.]
[Bov17]	A. Bovier. <i>Gaussian processes on trees. From spin glasses to branching Brownian motion,</i> volume 163 of <i>Cambridge Studies in Advanced Mathematics.</i> Cambridge University Press, Cambridge, 2017. [Cited on page 13.]
[Bra66]	J. H. Bramble. A second order finite difference analog of the first biharmonic boundary value problem. <i>Numer. Math.</i> , 9:236–249, 1966. [Cited on page 37.]
[Bra83]	M. Bramson. Convergence of solutions of the Kolmogorov equation to travel- ling waves. <i>Mem. Amer. Math. Soc.</i> , 44(285):iv+190, 1983. [Cited on page 23.]
[Bra95]	J. H. Bramble. Interpolation between Sobolev spaces in Lipschitz domains with an application to multigrid theory. <i>Math. Comp.</i> , 64(212):1359–1365, 1995. [Cited on page 146.]
[Bre11]	H. Brezis. Functional analysis, Sobolev spaces and partial differential equations. Universitext. Springer, New York, 2011. [Cited on page 56.]
[BS11]	M. Biskup and H. Spohn. Scaling limit for a class of gradient fields with nonconvex potentials. <i>Ann. Probab.</i> , 39(1):224–251, 2011. [Cited on page 21.]
[Buc19]	S. Buchholz. Phase transitions for a class of gradient fields, 2019. Preprint arXiv:1909.03021. [Cited on page 12.]
[Bur98]	V. I. Burenkov. <i>Sobolev spaces on domains,</i> volume 137 of <i>Teubner-Texte zur Mathematik [Teubner Texts in Mathematics]</i> . B. G. Teubner Verlagsgesellschaft mbH, Stuttgart, 1998. [Cited on page 145.]
[BV01]	E. Bolthausen and Y. Velenik. Critical behavior of the massless free field at the depinning transition. <i>Comm. Math. Phys.</i> , 223(1):161–203, 2001. [Cited on pages 153, 154, 158, 159, 164, 165, 166, 167, 168, 175, and 176.]
[BW20]	D. Belius and W. Wu. Maximum of the Ginzburg-Landau fields. <i>Ann. Probab.</i> , 2020. To appear. [Cited on page 22.]
[BWW17]	Y. Bernard, G. Wheeler, and VM. Wheeler. Rigidity and stability of spheres in the Helfrich model. <i>Interfaces Free Bound.</i> , 19(4):495–523, 2017. [Cited on page 3.]
[BZ12]	M. Bramson and O. Zeitouni. Tightness of the recentered maximum of the two- dimensional discrete Gaussian free field. <i>Comm. Pure Appl. Math.</i> , 65(1):1–20, 2012. [Cited on page 22.]

- [Cam80] S. Campanato. *Sistemi ellittici in forma divergenza. Regolarità all'interno*. Quaderni. Scuola Normale Superiore Pisa, Pisa, 1980. [Cited on pages 35, 50, 55, and 113.]
- [CC13] F. Caravenna and L. Chaumont. An invariance principle for random walk bridges conditioned to stay positive. *Electron. J. Probab.*, 18:no. 60, 32, 2013. [Cited on pages 25 and 87.]
- [CCH16a] A. Chiarini, A. Cipriani, and R. S. Hazra. Extremes of some Gaussian random interfaces. *J. Stat. Phys.*, 165(3):521–544, 2016. [Cited on page 22.]
- [CCH16b] A. Chiarini, A. Cipriani, and R. S. Hazra. Extremes of the supercritical Gaussian free field. *ALEA Lat. Am. J. Probab. Math. Stat.*, 13(2):711–724, 2016. [Cited on page 22.]
- [CD08] F. Caravenna and J.-D. Deuschel. Pinning and wetting transition for (1 + 1)dimensional fields with Laplacian interaction. *Ann. Probab.*, 36(6):2388–2433, 2008. [Cited on pages 27 and 29.]
- [CD09] F. Caravenna and J.-D. Deuschel. Scaling limits of (1 + 1)-dimensional pinning models with Laplacian interaction. *Ann. Probab.*, 37(3):903–945, 2009. [Cited on page 27.]
- [CDH19] A. Cipriani, B. Dan, and R. S. Hazra. The scaling limit of the membrane model. *Ann. Probab.*, 47(6):3963–4001, 2019. [Cited on pages 9, 21, 22, 35, 42, 86, 89, and 101.]
- [CDH20] A. Cipriani, B. Dan, and R. S. Hazra. Scaling limit of semiflexible polymers: a phase transition. *Comm. Math. Phys.*, 2020. To appear. [Cited on page 12.]
- [CDM09] C. Cotar, J.-D. Deuschel, and S. Müller. Strict convexity of the free energy for a class of non-convex gradient models. *Comm. Math. Phys.*, 286(1):359–376, 2009. [Cited on page 16.]
- [CFL28] R. Courant, K. Friedrichs, and H. Lewy. Über die partiellen Differenzengleichungen der mathematischen Physik. *Math. Ann.*, 100(1):32–74, 1928. [Cited on page 33.]
- [CHR18] A. Cipriani, R. S. Hazra, and W. M. Ruszel. Scaling limit of the odometer in divisible sandpiles. *Probab. Theory Related Fields*, 172(3-4):829–868, 2018. [Cited on page 9.]
- [Cip13] A. Cipriani. High points for the membrane model in the critical dimension. *Electron. J. Probab.*, 18:no. 86, 17 pp., 2013. [Cited on pages 30, 35, and 101.]
- [CM97] D. Cioranescu and F. Murat. A strange term coming from nowhere. In *Topics in the mathematical modelling of composite materials*, volume 31 of *Progr. Nonlinear Differential Equations Appl.*, pages 45–93. Birkhäuser Boston, Boston, MA, 1997. [Cited on pages 28, 155, 156, and 157.]
- [CMN18] R. Chhaibi, T. Madaule, and J. Najnudel. On the maximum of the C β E field. *Duke Math. J.*, 167(12):2243–2345, 2018. [Cited on pages 13 and 23.]
- [CN20] A. Chiarini and M. Nitzschner. Entropic repulsion for the Gaussian free field conditioned on disconnection by level-sets. *Probab. Theory Related Fields*, 177(1-2):525–575, 2020. [Cited on page 31.]

- [Col60] L. Collatz. The numerical treatment of differential equations. 3d ed. Translated from a supplemented version of the 2d German edition by P. G. Williams. Die Grundlehren der mathematischen Wissenschaften, Bd. 60. Springer-Verlag, Berlin-Göttingen-Heidelberg, 1960. [Cited on pages 33 and 37.]
- [CV00] P. Caputo and Y. Velenik. A note on wetting transition for gradient fields. Stochastic Process. Appl., 87(1):107–113, 2000. [Cited on page 29.]
- [CZ13] C. Cazacu and E. Zuazua. Improved multipolar Hardy inequalities. In Studies in phase space analysis with applications to PDEs, volume 84 of Progr. Nonlinear Differential Equations Appl., pages 35–52. Birkhäuser/Springer, New York, 2013. [Cited on pages 28 and 161.]
- [Dav06] O. Daviaud. Extremes of the discrete two-dimensional Gaussian free field. *Ann. Probab.*, 34(3):962–986, 2006. [Cited on page 30.]
- [DCGRS20] H. Duminil-Copin, S. Goswami, P.-F. Rodriguez, and F. Severo. Equality of critical parameters for percolation of Gaussian free field level-sets, 2020. Preprint arXiv:2002.07735. [Cited on pages 30 and 31.]
- [DDG13] A. Dembo, J. Ding, and F. Gao. Persistence of iterated partial sums. *Ann. Inst. Henri Poincaré Probab. Stat.*, 49(3):873–884, 2013. [Cited on page 25.]
- [Deu96] J.-D. Deuschel. Entropic repulsion of the lattice free field. II. The 0-boundary case. *Comm. Math. Phys.*, 181(3):647–665, 1996. [Cited on pages 24 and 86.]
- [DG00] J.-D. Deuschel and G. Giacomin. Entropic repulsion for massless fields. *Stochastic Process. Appl.*, 89(2):333–354, 2000. [Cited on page 24.]
- [DGI00] J.-D. Deuschel, G. Giacomin, and D. Ioffe. Large deviations and concentration properties for $\nabla \phi$ interface models. *Probab. Theory Related Fields*, 117(1):49–111, 2000. [Cited on page 15.]
- [DGT06] P. G. De Gennes and C. Taupin. Microemulsions and the flexibility of oil/water interfaces. *J. Phys. Chem.*, 86(13):2294–2304, 2006. [Cited on page 4.]
- [DKS92] R. Dobrushin, R. Kotecký, and S. Shlosman. Wulff construction. A global shape from local interaction, volume 104 of Translations of Mathematical Monographs. American Mathematical Society, Providence, RI, 1992. [Cited on page 4.]
- [DMRR92] F. Dunlop, J. Magnen, V. Rivasseau, and P. Roche. Pinning of an interface by a weak potential. *J. Statist. Phys.*, 66(1-2):71–98, 1992. [Cited on pages 16, 27, and 164.]
- [Dob68] R. L. Dobrušin. Description of a random field by means of conditional probabilities and conditions for its regularity. *Teor. Verojatnost. i Primenen*, 13:201–229, 1968. [Cited on page 6.]
- [Dob70] R. L. Dobrušin. Definition of a system of random variables by means of conditional distributions. *Teor. Verojatnost. i Primenen.*, 15:469–497, 1970. [Cited on page 6.]
- [Dol93] G. Dolzmann. *Campanato-Ungleichungen für Differenzenverfahren und finite Elemente*. Number 251 in Bonner mathematische Schriften. Rheinische Friedrich-Wilhelms-Universität Bonn, 1993. [Cited on pages 35, 36, 46, and 50.]

[Dol99] G. Dolzmann. Optimal convergence for the finite element method in Campanato spaces. Math. Comp., 68(228):1397-1427, 1999. [Cited on pages 35, 36, and 46.] [DPR18] A. Drewitz, A. Prévost, and P.-F. Rodriguez. The sign clusters of the massless Gaussian free field percolate on \mathbb{Z}^d , $d \ge 3$ (and more). *Comm. Math. Phys.*, 362(2):513-546, 2018. [Cited on page 30.] [DRSV14] B. Duplantier, R. Rhodes, S. Sheffield, and V. Vargas. Critical Gaussian multiplicative chaos: convergence of the derivative martingale. Ann. Probab., 42(5):1769–1808, 2014. [Cited on page 13.] [DRSV17] B. Duplantier, R. Rhodes, S. Sheffield, and V. Vargas. Log-correlated Gaussian fields: an overview. In *Geometry, analysis and probability,* volume 310 of *Progr.* Math., pages 191–216. Birkhäuser/Springer, Cham, 2017. [Cited on page 13.] [DRZ17] J. Ding, R. Roy, and O. Zeitouni. Convergence of the centered maximum of log-correlated Gaussian fields. Ann. Probab., 45(6A):3886–3928, 2017. [Cited on pages 23, 98, and 100.] [DS58] R. J. Duffin and E. P. Shelly. Difference equations of polyharmonic type. Duke Math. J., 25:209–238, 1958. [Cited on page 34.] [DS04] A. Dall'Acqua and G. Sweers. Estimates for Green function and Poisson kernels of higher-order Dirichlet boundary value problems. J. Differential Equations, 205(2):466-487, 2004. [Cited on pages 31, 43, and 47.] [Dud67] R. M. Dudley. The sizes of compact subsets of Hilbert space and continuity of Gaussian processes. J. Functional Analysis, 1:290–330, 1967. [Cited on page 88.] [Duf49] R. J. Duffin. On a question of Hadamard concerning super-biharmonic functions. J. Math. Physics, 27:253–258, 1949. [Cited on page 32.] [Duf53] R. J. Duffin. Discrete potential theory. Duke Math. J., 20:233-251, 1953. [Cited on page 34.] J.-D. Deuschel and Y. Velenik. Non-Gaussian surface pinned by a weak poten-[DV00] tial. Probab. Theory Related Fields, 116(3):359-377, 2000. [Cited on pages 27 and 168.] [DW15] D. Denisov and V. Wachtel. Exit times for integrated random walks. Ann. Inst. Henri Poincaré Probab. Stat., 51(1):167–193, 2015. [Cited on pages 25 and 87.] [Ehr71] L. W. Ehrlich. Solving the biharmonic equation as coupled finite difference equations. SIAM J. Numer. Anal., 8:278-287, 1971. [Cited on page 37.] [FHK12] Y. V. Fyodorov, G. A. Hiary, and J. P. Keating. Freezing transition, characteristic polynomials of random matrices, and the riemann zeta function. Phys. Rev. Lett., 108:170601, 2012. [Cited on page 13.] [FK14] Y. V. Fyodorov and J. P. Keating. Freezing transitions and extreme values: random matrix theory, and disordered landscapes. Philos. Trans. R. Soc. Lond. Ser. A Math. Phys. Eng. Sci., 372(2007):20120503, 2014. [Cited on page 13.] [FKG71] C. M. Fortuin, P. W. Kasteleyn, and J. Ginibre. Correlation inequalities on some partially ordered sets. Comm. Math. Phys., 22:89–103, 1971. [Cited on page 16.]

- [FO01] T. Funaki and S. Olla. Fluctuations for $\nabla \phi$ interface model on a wall. *Stochastic Process. Appl.*, 94(1):1–27, 2001. [Cited on page 25.]
- [FS97] T. Funaki and H. Spohn. Motion by mean curvature from the Ginzburg-Landau $\nabla \phi$ interface model. *Comm. Math. Phys.*, 185(1):1–36, 1997. [Cited on pages 16 and 20.]
- [Fun05] T. Funaki. Stochastic interface models. In *Lectures on probability theory and statistics*, volume 1869 of *Lecture Notes in Math.*, pages 103–274. Springer, Berlin, 2005. [Cited on pages 13, 14, 15, 16, and 19.]
- [FV18] S. Friedli and Y. Velenik. *Statistical mechanics of lattice systems*. Cambridge University Press, Cambridge, 2018. A concrete mathematical introduction. [Cited on pages 5 and 6.]
- [Geo88] H.-O. Georgii. *Gibbs measures and phase transitions,* volume 9 of *De Gruyter Studies in Mathematics*. Walter de Gruyter & Co., Berlin, 1988. [Cited on pages 6, 19, and 20.]
- [Ger30] S. Gerschgorin. Fehlerabschätzung für das Differenzenverfahren zur Lösung partieller Differentialgleichungen. ZAMM, 10:373–382, 1930. [Cited on page 33.]
- [GGS10] F. Gazzola, H.-C. Grunau, and G. Sweers. Polyharmonic boundary value problems. Positivity preserving and nonlinear higher order elliptic equations in bounded domains., volume 1991 of Lecture Notes in Mathematics. Springer-Verlag, Berlin, 2010. [Cited on pages 31, 32, and 43.]
- [Gia83] M. Giaquinta. *Multiple integrals in the calculus of variations and nonlinear elliptic systems*, volume 105 of *Annals of Mathematics Studies*. Princeton University Press, Princeton, NJ, 1983. [Cited on page 31.]
- [Gia93] M. Giaquinta. *Introduction to regularity theory for nonlinear elliptic systems*. Lectures in Mathematics ETH Zürich. Birkhäuser Verlag, Basel, 1993. [Cited on page 60.]
- [Gia01] G. Giacomin. Aspects of statistical mechanics of random surfaces. Notes of the lectures given at the IHP, 2001. [Cited on pages 15, 16, 23, and 29.]
- [Gia07] G. Giacomin. *Random polymer models*. Imperial College Press, London, 2007. [Cited on pages 25 and 27.]
- [Gia08] G. Giacomin. Renewal convergence rates and correlation decay for homogeneous pinning models. *Electron. J. Probab.*, 13:no. 18, 513–529, 2008. [Cited on page 27.]
- [Gib02] J. W. Gibbs. *Elementary Principles in Statistical Mechanics*. Charles Scribner's Sons, New York; Edward Arnold, London, 1902. [Cited on page 5.]
- [GJ87] J. Glimm and A. Jaffe. *Quantum physics*. Springer-Verlag, New York, second edition, 1987. A functional integral point of view. [Cited on page 8.]
- [GL18] G. Giacomin and H. Lacoin. Pinning and disorder relevance for the lattice Gaussian free field. J. Eur. Math. Soc. (JEMS), 20(1):199–257, 2018. [Cited on page 12.]

[GLMP83]	I. P. Gavrilyuk, R. D. Lazarov, V. L. Makarov, and S. P. Pirnazarov. Estimates for the rate of convergence of difference schemes for fourth-order equations of elliptic type. (Russian). <i>Zh. Vychisl. Mat. i Mat. Fiz.</i> , 23(2):355–365, 1983. [Cited on pages 37 and 44.]
[GM12]	M. Giaquinta and L. Martinazzi. <i>An introduction to the regularity theory for elliptic systems, harmonic maps and minimal graphs,</i> volume 11 of <i>Appunti. Scuola Normale Superiore di Pisa (Nuova Serie)</i> . Edizioni della Normale, Pisa, second edition, 2012. [Cited on page 159.]
[GMP83]	I. P. Gavrilyuk, V. L. Makarov, and S. P. Pirnazarov. Consistent estimates of the rate of convergence of difference solutions to generalized solutions of the first boundary value problem for fourth-order equations. (Russian). <i>Izv. Vyssh. Uchebn. Zaved. Mat.</i> , (2):15–22, 1983. [Cited on pages 38, 44, 45, 102, and 124.]
[GOS01]	G. Giacomin, S. Olla, and H. Spohn. Equilibrium fluctuations for $\nabla \phi$ interface model. <i>Ann. Probab.</i> , 29(3):1138–1172, 2001. [Cited on pages 15 and 21.]
[Gri67]	P. Grisvard. Caractérisation de quelques espaces d'interpolation. Arch. Rational Mech. Anal., 25:40–63, 1967. [Cited on pages 146, 147, and 148.]
[Hac81]	W. Hackbusch. On the regularity of difference schemes. <i>Ark. Mat.</i> , 19(1):71–95, 1981. [Cited on page 135.]
[Hac83]	W. Hackbusch. On the regularity of difference schemes. II. Regularity estimates for linear and nonlinear problems. <i>Ark. Mat.</i> , 21(1):3–28, 1983. [Cited on page 45.]
[Hac17]	W. Hackbusch. <i>Elliptic differential equations. Theory and numerical treatment,</i> volume 18 of <i>Springer Series in Computational Mathematics</i> . Springer-Verlag, Berlin, second edition, 2017. [Cited on page 33.]
[Had08]	J. Hadamard. Sur certains cas intéressants du problème biharmonique. In <i>Atti del IV Congresso Internazionale dei Matematici (Roma, 6-11 Aprile 1908),</i> volume 2, pages 12–14. R. Accademia dei Lincei, Roma, 1908. [Cited on page 32.]
[Hel73]	W. Helfrich. Elastic properties of lipid bilayers: Theory and possible experiments. <i>Zeitschrift für Naturforschung C</i> , 28(11-12):693–703, 1973. [Cited on page 3.]
[Hil16]	S. Hilger. Scaling limit and convergence of smoothed covariance for gradient models with non-convex potential, 2016. Preprint arXiv:1603.04703. [Cited on page 21.]
[Hil19]	S. Hilger. <i>Statistical mechanics of gradient models</i> . PhD Thesis, Rheinische Friedrich-Wilhelms-Universität Bonn, 2019. [Cited on page 19.]
[HL97]	C. Hiergeist and R. Lipowsky. Local contacts of membranes and strings. <i>Physica</i> A: Statistical Mechanics and its Applications, 244(1):164 – 175, 1997. [Cited on pages 4 and 23.]
[HS94]	B. Helffer and J. Sjöstrand. On the correlation for Kac-like models in the convex case. <i>J. Statist. Phys.</i> , 74(1-2):349–409, 1994. [Cited on page 14.]
[HV04]	O. Hryniv and Y. Velenik. Universality of critical behaviour in a class of recurrent random walks. <i>Probab. Theory Related Fields</i> , 130(2):222–258, 2004. [Cited on page 29.]

[HV18]	R. M. Höfer and J. J. L. Velázquez. The method of reflections, homogenization and screening for Poisson and Stokes equations in perforated domains. <i>Arch. Ration. Mech. Anal.</i> , 227(3):1165–1221, 2018. [Cited on pages 157 and 159.]
[IĬS86]	L. D. Ivanovich, B. S. Ĭovanovich, and È. È. Shili. Convergence of difference schemes for the biharmonic equation. <i>Zh. Vychisl. Mat. i Mat. Fiz.</i> , 26(5):776–779, 799, 1986. [Cited on pages 37, 44, and 102.]
[IV00]	D. Ioffe and Y. Velenik. A note on the decay of correlations under δ -pinning. <i>Probab. Theory Related Fields</i> , 116(3):379–389, 2000. [Cited on pages 27, 168, 175, and 176.]
[JIS85]	B. S. Jovanović, L. D. Ivanović, and E. Süli. Convergence of difference schemes for the equation $-\Delta u + cu = f$ on generalized solutions from $W_{2,*}^s$ ($-\infty < s < +\infty$). (Russian). <i>Publ. Inst. Math. (Beograd)</i> (<i>N.S.)</i> , 37(51):129–138, 1985. [Cited on page 143.]
[JS14]	B. S. Jovanović and E. Süli. <i>Analysis of finite difference schemes for linear partial differential equations with generalized solutions</i> , volume 46 of <i>Springer Series in Computational Mathematics</i> . Springer, London, 2014. [Cited on pages 33, 37, 38, 44, 45, 102, 107, 108, 123, 124, 125, 126, 127, 142, and 144.]
[Ken01]	R. Kenyon. Dominos and the Gaussian free field. <i>Ann. Probab.</i> , 29(3):1128–1137, 2001. [Cited on page 8.]
[KMR97]	V. A. Kozlov, V. G. Maz'ya, and J. Rossmann. <i>Elliptic boundary value problems in domains with point singularities</i> , volume 52 of <i>Mathematical Surveys and Mono-graphs</i> . American Mathematical Society, Providence, RI, 1997. [Cited on pages 32, 36, 65, and 67.]
[KMR01]	V. A. Kozlov, V. G. Maz'ya, and J. Rossmann. <i>Spectral problems associated with corner singularities of solutions to elliptic equations,</i> volume 85 of <i>Mathematical Surveys and Monographs</i> . American Mathematical Society, Providence, RI, 2001. [Cited on page 32.]
[Kra67]	J. P. Krasovskiĭ. Isolation of the singularity in Green's function. (Russian). <i>Izv. Akad. Nauk SSSR Ser. Mat.</i> , 31:977–1010, 1967. [Cited on page 31.]
[Kur07]	N. Kurt. Entropic repulsion for a class of Gaussian interface models in high dimensions. <i>Stochastic Process. Appl.</i> , 117(1):23–34, 2007. [Cited on page 24.]
[Kur09]	N. Kurt. Maximum and entropic repulsion for a Gaussian membrane model in the critical dimension. <i>Ann. Probab.</i> , 37(2):687–725, 2009. [Cited on pages 22, 24, 29, 35, 87, 101, and 105.]
[Kur12]	N. Kurt. Laplacian interface models with strictly convex potential. <i>Markov Proc. Rel. Fields</i> , 18(1):9–30, 2012. [Cited on page 12.]
[Laz81]	R. D. Lazarov. On the convergence of difference solutions to generalized solutions of a biharmonic equation in a rectangle. (Russian). <i>Differentsial'nye Uravneniya</i> , 17(7):1295–1303, 1344, 1981. [Cited on pages 37 and 44.]

[Lip95]	R. Lipowsky. Generic interactions of flexible membranes. In R. Lipowsky and E. Sackmann, editors, <i>Structure and Dynamics of Membranes</i> , volume 1 of <i>Handbook of Biological Physics</i> , pages 521 – 602. North-Holland, 1995. [Cited on pages 3, 4, and 29.]
[Lip01]	R. Lipowsky. Structured surfaces and morphological wetting transitions. <i>Inter-face Science</i> , 9(1-2):105–115, 2001. [Cited on page 29.]
[LL58]	L. D. Landau and E. M. Lifshitz. <i>Statistical physics</i> . Course of Theoretical Physics. Vol. 5. Translated from the Russian by E. Peierls and R. F. Peierls. Pergamon Press Ltd., London-Paris; Addison-Wesley Publishing Company, Inc., Reading, Mass., 1958. [Cited on page 5.]
[LM72a]	JL. Lions and E. Magenes. <i>Non-homogeneous boundary value problems and applications. Vol. I.</i> Translated from the French by P. Kenneth. Die Grundlehren der mathematischen Wissenschaften, Band 181. Springer-Verlag, New York-Heidelberg, 1972. [Cited on pages 31, 128, 129, 136, 145, 146, 147, 149, and 150.]
[LM72b]	JL. Lions and E. Magenes. <i>Non-homogeneous boundary value problems and applications. Vol. II.</i> Translated from the French by P. Kenneth. Die Grundlehren der mathematischen Wissenschaften, Band 182. Springer-Verlag, New York-Heidelberg, 1972. [Cited on page 149.]
[LM87]	J. L. Lebowitz and C. Maes. The effect of an external field on an interface, entropic repulsion. J. Statist. Phys., 46(1-2):39–49, 1987. [Cited on page 23.]
[LM17]	R. Latała and D. Matlak. Royen's proof of the Gaussian correlation inequality. In <i>Geometric aspects of functional analysis</i> , volume 2169 of <i>Lecture Notes in Math.</i> , pages 265–275. Springer, Cham, 2017. [Cited on pages 16, 88, and 163.]
[Löf92]	J. Löfström. Interpolation of boundary value problems of Neumann type on smooth domains. J. London Math. Soc. (2), 46(3):499–516, 1992. [Cited on page 146.]
[LPW09]	D. A. Levin, Y. Peres, and E. L. Wilmer. <i>Markov chains and mixing times</i> . American Mathematical Society, Providence, RI, 2009. With a chapter by James G. Propp and David B. Wilson. [Cited on page 17.]
[LR69]	O. E. Lanford, III and D. Ruelle. Observables at infinity and states with short range correlations in statistical mechanics. <i>Comm. Math. Phys.</i> , 13:194–215, 1969. ^[Cited on page 6.]
[LS04]	W. V. Li and QM. Shao. Lower tail probabilities for Gaussian processes. <i>Ann. Probab.</i> , 32(1A):216–242, 2004. [Cited on pages 26 and 88.]
[LSW19]	G. Lawler, X. Sun, and W. Wu. Four-dimensional loop-erased random walk. <i>Ann. Probab.</i> , 47(6):3866–3910, 2019. [Cited on page 9.]
[Mad15]	T. Madaule. Maximum of a log-correlated Gaussian field. <i>Ann. Inst. Henri Poincaré Probab. Stat.</i> , 51(4):1369–1431, 2015. [Cited on page 23.]
[Man67]	M. Mangad. Asymptotic expansions of Fourier transforms and discrete poly- harmonic Green's functions. <i>Pacific J. Math.</i> , 20:85–98, 1967. [Cited on pages 34, 36, 37, 45, 46, 72, 102, 109, and 110.]

- [Mil10] J. Miller. Universality for SLE(4), 2010. Preprint arXiv:1010.1356. [Cited on page 30.]
- [Mil11] J. Miller. Fluctuations for the Ginzburg-Landau $\nabla \phi$ interface model on a bounded domain. *Comm. Math. Phys.*, 308(3):591–639, 2011. [Cited on page 21.]
- [MK06] V. A. Marchenko and E. Y. Khruslov. Homogenization of partial differential equations, volume 46 of Progress in Mathematical Physics. Birkhäuser Boston, Inc., Boston, MA, 2006. [Cited on pages 155 and 157.]
- [MM13] I. Mitrea and M. Mitrea. *Multi-layer potentials and boundary problems for higherorder elliptic systems in Lipschitz domains*, volume 2063 of *Lecture Notes in Mathematics*. Springer, Heidelberg, 2013. [Cited on pages 32, 37, 102, 111, 144, and 145.]
- [MM14] S. Mayboroda and V. Maz'ya. Regularity of solutions to the polyharmonic equation in general domains. *Invent. Math.*, 196(1):1–68, 2014. [Cited on pages 32, 37, 102, and 113.]
- [Mor66] C. B. Morrey Jr. *Multiple integrals in the calculus of variations*. Die Grundlehren der mathematischen Wissenschaften, Band 130. Springer-Verlag New York, Inc., New York, 1966. [Cited on pages 55 and 61.]
- [MR10] V. Maz'ya and J. Rossmann. *Elliptic equations in polyhedral domains*, volume 162 of *Mathematical Surveys and Monographs*. American Mathematical Society, Providence, RI, 2010. [Cited on pages 32, 36, 65, 66, 67, and 68.]
- [MS19] S. Müller and F. Schweiger. Estimates for the Green's function of the discrete bilaplacian in dimensions 2 and 3. *Vietnam J. Math.*, 47(1):133–181, 2019. [Cited on pages 2, 25, 34, 35, and 41.]
- [MSS20] S. Müller, F. Schweiger, and E. Süli. Optimal-order finite difference approximation of generalized solutions to the biharmonic equation in a cube. *SIAM J. Numer. Anal.*, 58(1):298–329, 2020. [Cited on pages 2, 37, 38, and 123.]
- [MW40] W. H. McCrea and F. J. W. Whipple. Random paths in two and three dimensions. *Proc. Roy. Soc. Edinburgh*, 60:281–298, 1940. [Cited on page 34.]
- [NP87] D. R. Nelson and L. Peliti. Fluctuations in membranes with crystalline and hexatic order. *Journal de Physique*, 48(7):1085–1092, 1987. [Cited on page 4.]
- [NS97] A. Naddaf and T. Spencer. On homogenization and scaling limit of some gradient perturbations of a massless free field. *Comm. Math. Phys.*, 183(1):55–84, 1997. [Cited on pages 15 and 21.]
- [NV06] B. Niethammer and J. J. L. Velázquez. Screening in interacting particle systems. *Arch. Ration. Mech. Anal.*, 180(3):493–506, 2006. [Cited on page 157.]
- [Pit82] L. D. Pitt. Positively correlated normal variables are associated. *Ann. Probab.*, 10(2):496–499, 1982. [Cited on page 16.]
- [RCMS05] J. J. Ruiz-Lorenzo, R. Cuerno, E. Moro, and A. Sánchez. Phase transition in tensionless surfaces. *Biophysical Chemistry*, 115(2):187 – 193, 2005. BIFI 2004 International Conference Biology after the Genoma: A Physical View. [Cited on page 4.]

[Rod16]	PF. Rodriguez. Decoupling inequalities for the Ginzburg-Landau $\nabla \varphi$ models, 2016. Preprint arXiv:1612.02385. [Cited on page 31.]
[Roy14]	T. Royen. A simple proof of the Gaussian correlation conjecture extended to some multivariate gamma distributions. <i>Far East J. Theor. Stat.</i> , 48(2):139–145, 2014. [Cited on pages 16, 88, 158, and 163.]
[Roy16]	R. Roy. <i>Extreme values of log-correlated Gaussian fields</i> . PhD Thesis, University of Chicago, 2016. [Cited on page 22.]
[RS13]	PF. Rodriguez and AS. Sznitman. Phase transition and level-set percolation for the Gaussian free field. <i>Comm. Math. Phys.</i> , 320(2):571–601, 2013. [Cited on page 30.]
[RV07]	B. Rider and B. Virág. The noise in the circular law and the Gaussian free field. <i>Int. Math. Res. Not. IMRN</i> , (2):Art. ID rnm006, 33, 2007. [Cited on page 8.]
[RW84]	C. Rottman and M. Wortis. Statistical mechanics of equilibrium crystal shapes: interfacial phase diagrams and phase transitions. <i>Phys. Rep.</i> , 103(1-4):59–79, 1984. Common trends in particle and condensed matter physics (Les Houches, 1983). [Cited on page 3.]
[Sak03]	H. Sakagawa. Entropic repulsion for a Gaussian lattice field with certain finite range interaction. <i>J. Math. Phys.</i> , 44(7):2939–2951, 2003. [Cited on pages 12 and 24.]
[Sak12]	H. Sakagawa. On the free energy of a Gaussian membrane model with external potentials. <i>J. Stat. Phys.</i> , 147(1):18–34, 2012. [Cited on pages 27 and 153.]
[Sak16]	H. Sakagawa. On the probability that Laplacian interface models stay positive in subcritical dimensions. In <i>Stochastic analysis on large scale interacting systems</i> , RIMS Kôkyûroku Bessatsu, B59, pages 273–288. Res. Inst. Math. Sci. (RIMS), Kyoto, 2016. [Cited on pages 25 and 85.]
[Sak18]	H. Sakagawa. Localization of a Gaussian membrane model with weak pinning potentials. <i>ALEA Lat. Am. J. Probab. Math. Stat.</i> , 15(2):1123–1140, 2018. [Cited on page 27.]
[Sch66]	L. Schwartz. <i>Théorie des distributions</i> . Publications de l'Institut de Mathématique de l'Université de Strasbourg, No. IX-X. Nouvelle édition, entiérement corrigée, refondue et augmentée. Hermann, Paris, 1966. [Cited on page 72.]
[Sch81]	L. L. Schumaker. <i>Spline Functions: Basic Theory</i> . Pure & Applied Mathematics. Wiley-Interscience, New York, 1981. [Cited on pages 52 and 54.]
[Sch16]	F. Schweiger. Discrete Green's functions and statistical physics of membranes. Master's thesis, Rheinische Friedrich-Wilhelms-Universität Bonn, 2016. [Cited on pages 35, 41, 45, 85, and 94.]
[Sch20a]	F. Schweiger. The maximum of the four-dimensional membrane model. <i>Ann. Probab.</i> , 48(2):714–741, 2020. [Cited on pages 1, 21, 22, 34, 36, and 97.]
[Sch20b]	F. Schweiger. Pinning for the critical and supercritical membrane model, 2020. Preprint arXiv:2005.14701. [Cited on pages 1, 26, 28, and 151.]

[She05]	S. Sheffield. Random surfaces. <i>Astérisque</i> , (304):vi+175 pp., 2005. [Cited on pages 12, 16, and 20.]
[She07]	S. Sheffield. Gaussian free fields for mathematicians. <i>Probab. Theory Related Fields</i> , 139(3-4):521–541, 2007. [Cited on pages 8, 18, and 21.]
[Sim67]	R. B. Simpson. A fundamental solution for a biharmonic finite-difference operator. <i>Math. Comp.</i> , (21):321–339, 1967. [Cited on page 34.]
[Sin92]	Y. G. Sinaĭ. Distribution of some functionals of the integral of a random walk. <i>Teoret. Mat. Fiz.</i> , 90(3):323–353, 1992. [Cited on page 25.]
[Smi68]	J. Smith. The coupled equation approach to the numerical solution of the biharmonic equation by finite differences. I. <i>SIAM J. Numer. Anal.</i> , 5:323–339, 1968. [Cited on page 37.]
[Smi70]	J. Smith. The coupled equation approach to the numerical solution of the biharmonic equation by finite differences. II. <i>SIAM J. Numer. Anal.</i> , 7:104–111, 1970. [Cited on page 37.]
[SS09]	O. Schramm and S. Sheffield. Contour lines of the two-dimensional discrete Gaussian free field. <i>Acta Math.</i> , 202(1):21–137, 2009. [Cited on page 30.]
[Str72]	R. S. Strichartz. A note on Trudinger's extension of Sobolev's inequalities. <i>Indiana Univ. Math. J.</i> , 21:841–842, 1971/72. [Cited on page 104.]
[Tal96]	M. Talagrand. Majorizing measures: the generic chaining. <i>Ann. Probab.</i> , 24(3):1049–1103, 1996. [Cited on page 88.]
[Tee64]	G. J. Tee. A novel finite-difference approximation to the biharmonic operator. <i>Comput. J.</i> , 6:177–192, 1963/1964. [Cited on page 37.]
[Tho64]	V. Thomée. Elliptic difference operators and Dirichlet's problem. <i>Contributions to Differential Equations</i> , 3:301–324, 1964. [Cited on page 101.]
[Tho72]	C. J. Thompson. <i>Mathematical statistical mechanics</i> . A Series of Books in Applied Mathematics. The Macmillan Co., New York; Collier-Macmillan Ltd., London, 1972. [Cited on page 5.]
[TVV88]	A. Tabacco Vignati and M. Vignati. Spectral theory and complex interpolation. <i>J. Funct. Anal.</i> , 80(2):383–397, 1988. [Cited on page 111.]
[Vel04]	Y. Velenik. Entropic repulsion of an interface in an external field. <i>Probab. Theory Related Fields</i> , 129(1):83–112, 2004. [Cited on page 29.]
[Vel06]	Y. Velenik. Localization and delocalization of random interfaces. <i>Probab. Surv.</i> , 3:112–169, 2006. [Cited on pages 12, 19, 29, and 163.]
[vNG47]	J. von Neumann and H. H. Goldstine. Numerical inverting of matrices of high order. <i>Bull. Amer. Math. Soc.</i> , 53:1021–1099, 1947. [Cited on page 33.]
[Whi34]	H. Whitney. Functions differentiable on the boundaries of regions. <i>Ann. of Math.</i> (2), 35(3):482–485, 1934. [Cited on page 126.]

- [Wid71] K.-O. Widman. Hölder continuity of solutions of elliptic systems. *Manuscripta Math.*, 5:299–308, 1971. [Cited on pages 28, 157, and 159.]
- [Wul01] G. Wulff. XXV. Zur Frage der Geschwindigkeit des Wachsthums und der Auflösung der Krystallflächen. *Zeitschrift für Kristallographie*, 34(1):449–530, 1901. [Cited on page 3.]
- [WZ19] W. Wu and O. Zeitouni. Subsequential tightness of the maximum of two dimensional Ginzburg-Landau fields. *Electron. Commun. Probab.*, 24:Paper No. 19, 12 pp., 2019. [Cited on page 22.]